Non-perturbative aspects of gauge theories

by

Diego Delmastro

A thesis presented to the University of Waterloo in fulfillment of the thesis requirement for the degree of Doctor of Philosophy in Physics

Waterloo, Ontario, Canada, 2022

© Diego Delmastro 2022

Examining Committee Membership

The following served on the Examining Committee for this thesis. The decision of the Examining Committee is by majority vote.

External Examiner:	Erich Poppitz Professor, University of Toronto
Supervisor:	Jaume Gomis Research Faculty, Perimeter Institute for Theoretical Physics
Internal Members:	Niayesh Afshordi Associate Professor, Department of Physics and Astronomy, University of Waterloo Davide Gaiotto Research Faculty, Perimeter Institute for Theoretical Physics

Internal-External Member: Ruxandra Moraru Associate Professor, Department of Pure Mathematics, University of Waterloo

Author's Declaration

This thesis consists of material all of which I authored or co-authored: see Statement of Contributions included in the thesis. This is a true copy of the thesis, including any required final revisions, as accepted by my examiners.

I understand that my thesis may be made electronically available to the public.

Statement of Contributions

Chapters 1 and 5 correspond to the paper [1], written in collaboration with Davide Gaiotto and Jaume Gomis.

Chapter 2 corresponds to the paper [2], written in collaboration with Jaume Gomis and Matthew Yu.

Chapter 3 corresponds to the paper [3], written in collaboration with Changha Choi, Jaume Gomis, and Zohar Komargodski.

Chapter 4 corresponds to the paper [4], written in collaboration with Jaume Gomis.

Chapter 6 corresponds to the paper [5], written in collaboration with Jaume Gomis.

Abstract

The purpose of this thesis is to establish some non-perturbative results in gauge theories in $d \leq 4$ spacetime dimensions. We generically refer to such theories as *Quantum Chromodynamics* (QCD). Gauge theories are asymptotically free, which means that at short distances interactions become weak, and one can reliably use perturbation theory in order to solve the theory. Most of our current understanding of QCD stems from studying the theory at short distances; for example, this is the regime where the bulk of experimental tests of the Standard Model of particle physics has been performed. By contrast, the long distance regime is characterized by strong interactions, and here perturbation theory generically stops working. Hence, the dynamics of QCD at macroscopic scales is to a large extent still a mystery. For example, our *ab initio* understanding of the spectrum of hadrons is still very limited, most data coming from numerics or phenomenological models.

In order to study gauge theories in their strongly-interacting regime, the perturbative approach is mostly useless, and one must develop new tools. In this work we study some quantitative properties of QCD in its strongly-coupled phase. Some of our main tools are supersymmetry, 't Hooft anomalies, dualities, topological fields theories, and two-dimensional chiral algebras.

Acknowledgements

I would like to begin by expressing a big thank-you to my adviser Jaume Gomis. It goes without saying, this thesis would not have been possible without him. I am very grateful for his patient support over these years. His many insights and clear physical intuition shaped my way of thinking, and they will always stay with me.

I also want to thank my collaborators Changha Choi, Davide Gaiotto, Zohar Komargodski, and Matthew Yu, from whom I learned a lot during our many discussions. This thesis consists of articles we wrote together; any chapter written in collaboration is as much theirs as it is mine, after all.

I am also glad to have shared neighbouring worldlines with many wonderful fellow graduate students at the Perimeter Institute, from whom I also learned a lot. I would like to thank, in particular, Kasia Budzik, Justin Kulp, Ji Hoon Lee, Elise LePage, and Jingxiang Wu. And a special thank-you to Alicia Lima and Joshua Sandor for helping me stay sane during lockdown.

I also benefited from lengthy discussions with many researchers at the Perimeter Institute, especially Theo Johnson-Freyd, as well as Meng Guo, Yin-Chen He, Chong Wang, and Jon Yard, among others.

Well, now that we are at it, I would like to show my appreciation to the Perimeter Institute as a whole. It is an amazing place to do physics at, full of kind and brilliant individuals. Thank you all, and thank you Debbie Guenther. The University of Waterloo was also very kind to me. Both PI and UW made a great effort to keep our spirits high during these difficult times, and I am grateful for that.

On a similar note, I would like to thank the Galileo Galilei Institute for Theoretical Physics and the Simons Center for Geometry and Physics for their hospitality during workshops, where some results on this thesis were developed, and some others presented and discussed. Both GGI and SCGP are fantastic institutes with a very stimulating atmosphere.

Last but not least, I want to thank the members of my committee, Davide and Jaume, together with Niayesh Afshordi, Ruxandra Moraru, and Erich Poppitz, for their valuable input, and for accommodating last-minute meetings caused by my frequent absent-mindedness in regard to bureaucracy and deadlines.

And, of course, family and friends.

Table of Contents

0	Intr	roduction and summary. 1		
	0.1	Chapter 1		
	0.2	Chapter 2		
	0.3	Chapter 3		
	0.4	Chapter 4		
	0.5	Chapter 5		
	0.6	Chapter 6		
1	Glo	bal Anomalies on the Hilbert Space. 34		
	1.1	Introduction and Summary		
	1.2	Anomalies from Layers		
	1.3	Anomalies in free fermion Hilbert space		
		1.3.1 Anomalous \mathbb{Z}_2^{T} in 0+1 dimensions		
		1.3.2 Anomalous \mathbb{Z}_4^{T} in 2+1 dimensions		
		1.3.3 Anomalous \mathbb{Z}_2 in 1+1 dimensions		
	1.4	Anomalies in spin TQFT Hilbert space		
		1.4.1 $\nu = 2 \mod 4$: Arf layer		
		1.4.2 $\nu = 1 \mod 2$: fermion layer $\ldots \ldots \ldots$		
2	Infr	ared phases of 2d QCD. 75		
	2.1 Introduction			
2.2 $2d$ QCD theories \ldots \ldots \ldots \ldots \ldots \ldots \ldots		$2d$ QCD theories $\ldots \ldots \ldots$		
	2.3	Symmetries, 't Hooft Anomalies and Gaplessness		
		2.3.1 Towards Gapped QCD Theories		
	2.4	Mass spectrum and QCD Hamiltonians		
		2.4.1 Lightcone Hamiltonian		
		2.4.2 Temporal gauge Hamiltonian		
		2.4.3 Classification of gapped theories		
	2.5	Infrared dynamics of $2d$ QCD		
		2.5.1 Gapped theories		

		2.5.2	QCD with fundamental matter	16
		2.5.3	WZW models	18
		2.5.4	Minimal models	19
		2.5.5	Diagonal coset	20
		2.5.6	Kazama-Suzuki	21
	2.6	Conve	ntions and Background	21
		2.6.1	Matching Energy Momentum Tensors	24
		2.6.2	Temporal Gauge Hamiltonian commutation	26
	2.7	Infrare	ed coset CFTs $\ldots \ldots \ldots$	28
		2.7.1	Examples of topological cosets	37
		2.7.2	Example of a non-topological coset	40
	2.8	Abelia	In theories \ldots \ldots \ldots \ldots \ldots 1^4	42
		2.8.1	$U(1)$ with N charge-q Dirac fermions $\ldots \ldots \ldots$	47
		2.8.2	Vacua of gapped theories	48
3	Thr	ee-Din	nensional Gauge Theories with Two-Index Matter Fields. 15	50
	3.1	Introd	uction \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 1	51
	3.2	Phase	Diagrams	55
	3.3	Additi	onal Consistency Checks	63
		3.3.1	Special Cases	63
		3.3.2	Gravitational Counterterm Matching	65
		3.3.3	Baryon Counterterm Matching	67
		3.3.4	Self-consistency check for the dualities	68
	3.4	Domai	in Walls in Four Dimensional Gauge Theories	69
	3.5	Comm	nents on Future Directions	75
	3.6	Review	w of QCD_3 dynamics.	78
	3.7	Time-	reversal and flavor symmetries.	81
		3.7.1	Unitary group.	82
		3.7.2	Orthogonal group.	84
		3.7.3	Symplectic group	85
	3.8	Time-	reversal anomaly. \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 18	86
		3.8.1	Unitary group.	86
		3.8.2	Orthogonal group.	88
		3.8.3	Symplectic group	91
4	Syn	nmetri	es of Abelian Chern-Simons Theories and Arithmetic.	93
	4.1	Introd	uction and Summary	94
	4.2	TQFT	s and Symmetry $\ldots \ldots 1$	99
	4.3	$\mathrm{U}(1)_k$	Chern-Simons	03
		4.3.1	Symmetries of $U(1)_k$	05

		4.3.2	Minimal abelian TQFT	217
	4.4	$\mathrm{U}(1)^n$ C	Chern-Simons theory	219
		4.4.1	Symmetries of $U(1)^n$	222
		4.4.2	Dualities of $U(1)^n$	233
	4.5	Exampl	les \ldots \ldots \ldots \ldots \ldots	234
	4.6	Notatio	n and definitions. \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots	244
	4.7	Further	results	246
5	Ferr	nionic [FQFTs in three dimensions. 2	252
	5.1	Spin TO	QFTs and anyon condensation	253
		5.1.1	Boson anyon condensation	257
		5.1.2	Fermion anyon condensation	260
	5.2	Exampl	les of anyon condensation	267
		5.2.1	$\mathrm{SO}(n)_1$	267
		5.2.2	$\mathrm{SO}(n)_2$	269
		5.2.3	$\mathrm{SO}(n)_3$	273
		5.2.4	$\mathrm{U}(1)_k$	275
	5.3	Fermior	nic surgery	278
	5.4	Higher	genus. Fermionic Verlinde formula.	281
6	Don	nain Wa	alls in $4d \mathcal{N} = 1$ SYM.	291
6	Don 6.1	nain W a Domain	alls in $4d \mathcal{N} = 1$ SYM. 22 Walls in $4d \mathcal{N} = 1$ SYM	291 292
6	Don 6.1 6.2	nain Wa Domain Twisted	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM2d Witten Indices2	291 292 297
6	Don 6.1 6.2	nain Wa Domain Twisted 6.2.1	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM2d Witten Indices2 $G = SU(N)$ 2	291 292 297 301
6	Don 6.1 6.2	nain Wa Domain Twisted 6.2.1 6.2.2	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM3d Witten Indices4 $G = SU(N)$ 4 $G = Sp(N)$ 4 $G = Sp(N)$ 4	 291 292 297 301 304
6	Don 6.1 6.2	nain Wa Domain Twisted 6.2.1 6 6.2.2 6 6.2.3 6	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM2d Witten Indices2 $G = SU(N)$ 2 $G = Sp(N)$ 2 $Sp(N)$ 2 $Sp(N)$ 2 $Sp(N)$ 2 $Sp(N)$ 3 $Sp(N)$	 291 292 297 301 304 305
6	Don 6.1 6.2	nain Wa Domain Twisted 6.2.1 6 6.2.2 6 6.2.3 6 6.2.4 6	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM3d Witten Indices4 $G = SU(N)$ 4 $G = Sp(N)$ 4 $G = Sp(N)$ 4 $G = Spin(2N+1)$ 4 $G = Spin(2N)$ 4 $G = Spin(2N)$ 4 $G = Spin(2N)$ 4	 291 292 297 301 304 305 307
6	Dom 6.1 6.2	nain Wa Domain Twisted 6.2.1 6 6.2.2 6 6.2.3 6 6.2.4 6 6.2.5 6	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM	 291 292 297 301 304 305 307 310
6	Don 6.1 6.2	nain Wa Domain Twisted 6.2.1 6 6.2.2 6 6.2.3 6 6.2.4 6 6.2.5 6 6.2.6 2	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM3d Witten Indices4 $G = SU(N)$ 4 $G = Sp(N)$ 4 $G = Sp(N)$ 4 $G = Spin(2N+1)$ 4 $G = G_2$ 4 $G = G_2$ 4Minimal Wall for Arbitrary Gauge Group4	 291 292 297 301 304 305 307 310 310
6	Don 6.1 6.2 6.3	nain Wa Domain Twisted 6.2.1 6 6.2.2 6 6.2.3 6 6.2.4 6 6.2.5 6 6.2.6 1 Hilbert 1	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM	 291 292 297 301 304 305 307 310 310 311
6	Don 6.1 6.2 6.3	nain Wa Domain Twisted 6.2.1 6 6.2.2 6 6.2.3 6 6.2.4 6 6.2.5 6 6.2.6 1 Hilbert 6.3.1	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM	 291 292 297 301 304 305 307 310 311 315
6	Don 6.1 6.2 6.3 6.4	nain Wa Domain Twisted 6.2.1 6.2.2 6.2.3 6.2.4 6.2.5 6.2.6 Hilbert 6.3.1 Domain	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM	 291 292 297 301 304 305 307 310 311 315 319
6	Don 6.1 6.2 6.3 6.4	nain Wa Domain Twisted 6.2.1 6.2.2 6.2.3 6.2.4 6.2.5 6.2.6 Hilbert 6.3.1 Domain 6.4.1	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM	291 292 297 301 304 305 307 310 311 315 319 322
6	Don 6.1 6.2 6.3 6.4	nain Wa Domain Twisted 6.2.1 6.2.2 6.2.3 6.2.4 6.2.5 6.2.6 Hilbert 6.3.1 Domain 6.4.1	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM2a Witten Indices2 $G = SU(N)$ 2 $G = Sp(N)$ 2 $G = Sp(N)$ 2 $G = Spin(2N+1)$ 2 $G = Spin(2N)$ 2 $G = G_2$ 2Minimal Wall for Arbitrary Gauge Group2Space of Spin TQFTs2Partition Function of Spin TQFTs2 $G = Sp(n)$ 2 $G = U(n)$ 2	 291 292 297 301 304 305 307 310 311 315 319 322 323
6	Don 6.1 6.2 6.3 6.4	nain Wa Domain Twisted 6.2.1 6.2.2 6.2.3 6.2.4 6.2.5 6.2.6 Hilbert 6.3.1 Domain 6.4.1 6.4.3	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM	 291 292 297 301 304 305 307 310 311 3115 319 322 323 323 329
6	Don 6.1 6.2 6.3 6.4	nain Wa Domain Twisted 6.2.1 6.2.2 6.2.3 6.2.4 6.2.5 6.2.6 Hilbert 6.3.1 Domain 6.4.1 6.4.3 6.4.4	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM	291 292 297 301 304 305 307 310 311 315 319 322 323 322 323 329 3335
6	Don 6.1 6.2 6.3 6.4	nain Wa Domain Twisted $6.2.1$ 6 $6.2.2$ 6 $6.2.3$ 6 $6.2.4$ 6 $6.2.5$ 6 $6.2.6$ 1 $6.2.6$ 1 $6.2.6$ 1 $6.3.1$ 2 $6.4.1$ 6 $6.4.3$ 6 $6.4.4$ 6 $6.4.5$ 2	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM	291292297301304305307310311315319322323329335
6	Don 6.1 6.2 6.3 6.4 6.5	nain Wa Domain Twisted 6.2.1 6.2.2 6.2.3 6.2.4 6.2.5 6.2.6 Hilbert 6.3.1 Domain 6.4.1 6.4.3 6.4.4 6.4.5 Conclude	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM	291 292 297 301 304 305 307 310 310 311 315 312 322 323 322 3323 323 323 335 335
6	Don 6.1 6.2 6.3 6.4 6.5 6.6	nain Wa Domain Twisted 6.2.1 6.2.2 6.2.3 6.2.4 6.2.5 6.2.6 Hilbert 6.3.1 Domain 6.4.1 6.4.3 6.4.4 6.4.5 Conclude Chern-S	alls in $4d \ \mathcal{N} = 1$ SYM.2a Walls in $4d \ \mathcal{N} = 1$ SYM	291 292 2297 301 304 305 307 310 310 311 315 322 323 322 335 335 336 337

		6.6.2 $G = \operatorname{Spin}(2n+1)$	338
		6.6.3 $G = \operatorname{Spin}(2n)$	340
	6.7	The Exceptional Groups	341
Re	efere	nces	343
A	ppen	dices	371
\mathbf{A}	Rev	iew and background.	372
Α	Rev A.1	riew and background. Fermions in various dimensions.	372 372
Α	Rev A.1 A.2	iew and background. Fermions in various dimensions	372 372 376
A B	Rev A.1 A.2 Use	iew and background. Fermions in various dimensions	 372 372 376 384
A B	Rev A.1 A.2 Use B.1	iew and background. Fermions in various dimensions. Fermion kinetic term. Fermion kinetic term. ful data for Lie groups and representations. Summary of representations.	 372 372 376 384 386

Chapter 0 Introduction and summary.

Say we are given a quantum field theory. An extremely useful tool to understand its dynamics is *perturbation theory*: we expand observables in powers of a coupling constant $g(\mu)$, where μ is the energy scale of interest. In general, this expansion is reliable if and only if $g(\mu)$ is sufficiently small, a situation that is controlled by the beta function $\beta(g) := \mu \partial_{\mu} g(\mu)$. (See e.g. [6] for the textbook analysis.) If $\beta < 0$ one has $g \to 0$ as $\mu \to \infty$, which means that the theory is ultraviolet free, and perturbation theory works at large energies. Conversely, if $\beta > 0$, one has $g \to 0$ as $\mu \to 0$, which means that the theory is infrared free, and perturbation theory works at low energies. In either case, perturbation theory does *not* work in the opposite limit, and the theory becomes strongly interacting, in which case new tools are required.

The leading behaviour of the beta function is controlled by dimensional analysis. For example, let us look at a gauge theory. Let A be the gauge field and F = dA + [A, A]. The Lagrangian reads

$$\mathcal{L} = -\frac{1}{2g^2} \operatorname{tr}(F \wedge \star F) + \cdots$$
 (0.0.1)

By dimensional analysis it is easy to see that g^2 has units of μ^{4-d} where d is the number of spacetime dimensions, which means that

$$\beta(g^2) = (d-4)g^2 + \mathcal{O}(g^3) \tag{0.0.2}$$

Therefore, for d > 4 the theory is infrared free, and the ultraviolet is ill-behaved. Similarly, for d < 4, the theory is ultraviolet free and the infrared is non-perturbative. The edge case d = 4 is of special interest because it is the physical dimension, but also because g is classically marginal. A one-loop computation is necessary, and one learns that the sign of β depends on the details of the matter fields. For realistic field content, such as the Standard Model of particle physics, the beta function is negative, and the theory behaves as in the d < 4 case. So, to summarize, in $d \leq 4$ the coupling constant of gauge theories can be as small as we want, just by looking at shorter and shorter distances, which means that the high energy regime can be studied reliably using perturbation theory.

Understanding the behaviour of the theory at low energies is, thus, a very non-trivial yet important task: first, it is a non-perturbative problem, thus requiring new tools; and second, it is of practical interest, because it describes the real world at real (macroscopic) energy scales. For example, the phenomenological spectrum of hadrons looks nothing like what one would expect from weakly-coupled quarks, a reflection of the fact that the dynamics are strongly interacting. While lattice QCD has been able to shed much light on the problem, understanding it analytically is still one of the long-standing open problems in QFT.

Our goal in this thesis is to say a few things about some non-perturbative properties of gauge theories in $d \leq 4$. In the reminder of this chapter we summarize some of our main findings. For the sake of readability, our arguments here are more heuristic that in the main text, and we include fewer references to the literature. We generically organize the the document from low dimensions to high dimensions.

0.1 Chapter 1.

In chapter 1 we discuss some aspects of 't Hooft anomalies [7] in QFT. Anomalies are arguably the most systematic and powerful tool that we have in order to analyze non-perturbative dynamics (other than numerics). They are one of the few properties of strongly interacting theories that can in fact be determined explicitly and reliably, although sometimes the computation can require substantial mathematical skill.

Consider a generic QFT with some algebra of operators. The symmetry group of the theory is, by definition, the group of transformations on this algebra that leaves the dynamics invariant. Importantly, the group of symmetries of the theory is typically realized on the Hilbert space \mathcal{H} projectively [6]. The physical reason is that only transition amplitudes are observable, so the physical states are rays on \mathcal{H} : their phase is unobservable and therefore symmetries need only compose properly modulo a phase. Mathematically, operators live in $End(\mathcal{H}) \cong \mathcal{H} \otimes \mathcal{H}^*$ and an action on $\mathcal{H} \otimes \mathcal{H}^*$ fixes an action on \mathcal{H} only up to a phase.

What determines the projective representation of the symmetry group on the Hilbert space? How can we compute this phase in practice? And what consequences does it have for the properties of the theory? The main idea of chapter 1 is that, as it turns out, the projective phase is entirely determined by the anomalies of the symmetry group. In this way, if one understands the anomalies of the system, one can determine the projective action. Often enough, the converse is also true, in the sense that if one is able to compute the projective action independently, one can use that information to recover (at least some partial) information about the anomalies of the theory. Therefore, this insight not only simplifies the calculations of some anomalies, but it also gives a somewhat transparent picture of what the anomaly represents and what its consequences are.

The simplest illustration of this idea is in a theory with a bunch of fermions and timereversal symmetry. Being fermionic, the system has a \mathbb{Z}_2 symmetry denoted by $(-1)^F$, which measures whether a given operator is bosonic (if even) or fermionic (if odd). Time-reversal, denoted by T, will be chosen to satisfy the simplest symmetry group there is, namely \mathbb{Z}_2 , i.e., $T^2 = 1$. All in all, in this example we take the symmetry group to be $\mathbb{Z}_2 \times \mathbb{Z}_2$, represented as

$$((-1)^F)^2 = 1, \qquad \mathsf{T}^2 = 1, \qquad (-1)^F \mathsf{T} = \mathsf{T}(-1)^F$$
 (0.1.1)

An important property to keep in mind is that time-reversal is an anti-unitary operator, meaning that it conjugates complex numbers.

Given the symmetry group above, how will it be represented on the Hilbert space? In full generality, the group structure need only be realized up to a phase, which means that, as a matter of principle, we can have the following:

$$((-1)^F)^2 = \alpha, \qquad \mathsf{T}^2 = \beta, \qquad (-1)^F \mathsf{T} = \gamma \mathsf{T}(-1)^F$$
 (0.1.2)

where $\alpha, \beta, \gamma \in U(1)$ are three phases. We can always redefine $(-1)^F \mapsto \alpha^{-1/2}(-1)^F$ such that α disappears from the first equation, hence we can always fix fermion parity to square to 1. Importantly, we cannot do this for time-reversal, since T is anti-unitary and therefore T^2 is invariant under rephasing. Explicitly, if we let $T \to \xi T$ for some phase ξ , then $T^2 \to \xi T \xi T = |\xi|^2 T^2$, which is still equal to T^2 . In this sense, it seems that the phase β cannot be redefined away.

An important condition on β can be obtained by looking at T³. This can be parenthesized in two ways,

$$\mathsf{T}^2\mathsf{T} = \mathsf{T}\mathsf{T}^2 \tag{0.1.3}$$

which means that β commutes with T and therefore it has to be real, i.e., $\beta = \pm 1$. Finally, a similar constraint on γ can be obtained by multiplying both sides of $(-1)^F T = \gamma T(-1)^F$ by $(-1)^F$, once from the left and once from the right, and comparing the result. This yields $\gamma^2 T = T$, i.e., $\gamma = \pm 1$. So, to summarize, the most general projective action consistent with this symmetry group is

$$((-1)^F)^2 = 1, \qquad \mathsf{T}^2 = \pm 1, \qquad (-1)^F \mathsf{T} = \pm \mathsf{T}(-1)^F$$
 (0.1.4)

where the two signs are independent. (Note that if we considered a unitary symmetry instead of an anti-unitary one, then the second sign β could be redefined away too, and only the third sign γ would remain.)

A key property of this result is that the space of possible projective phases is disconnected. Consequently, if we deform the system continuously, while preserving the symmetry, we cannot change the two signs above – they are *invariants* of the system. Any continuous map from $\beta = +1$ to $\beta = -1$ would have to go through a point where β is a complex phase, which is inconsistent with the symmetry group, as we just showed. In this way, the two signs \pm above are a property of the theory that is protected under smooth deformations. This gives a convenient way to compute them: if we are able to find a deformation from whatever theory we are interested in, into a theory we already understand, we can use that information to determine the two signs. We just deform into the simpler theory, which will preserve the signs, and compute them there. We can then deform back into the original theory, with the guarantee that the signs will stay the same. For example, one can show that a system of d = 1 dimensional fermions, with arbitrary self-interactions, has $\gamma = -1$ if the number of fermions is odd; this can be shown by turning off the interactions, which leaves us with a system of free fermions, where the result $\gamma = -1$ can be established in a few lines of straightforward algebra.

The exact same phenomenon occurs for many other symmetry groups beyond time-reversal, and with any group structure beyond \mathbb{Z}_2 . Generically, the space of possible projective phases is disconnected, and therefore the specific phase chosen by a system becomes an invariant of the same. As such, the phase can be computed by deforming the theory into a simpler one and determining the phase there. For example, many theories we are interested in can be deformed into free theories, such as QCD, where $g \to 0$ takes us to a system of free quarks and free gluons. Free systems can be solved exactly and therefore any projective phase can be calculated explicitly. The projective phase of the original system is therefore determined too.

That being said, and in order to not give the impression that these features are specific to systems of free fields, in chapter 1 we also illustrate them by looking at topological quantum field theories. These are not continuously connected to a free phase (that is, not connected via a weakly-coupled path), inasmuch as their coupling constants are discrete and hence cannot be smoothly turned off. The study of fermionic TQFTs requires some machinery that we shall develop more systematically in chapter 5, although some basic notions are reviewed in chapter 1. In any event, the final conclusion is that these theories too acquire projective phases, exactly like the theories that can be deformed into free phases, which is of course unsurprising since, as we argue, these projective phases are a universal features of QFT.

These projective phases are not a mere curiosity, but they have important implications. For example, if a given system has time-reversal as above, but the group is realized as $T^2 = -1$, then one can immediately conclude that the spectrum of the Hamiltonian is necessarily degenerate, with at least two-fold degeneracy. This is the well-known Kramers degeneracy [8].¹ The same conclusion can be reached if the anomalous sign appears in $(-1)^F T = -T(-1)^F$, with the added property that the degeneracy is now of the supersymmetric type: the degenerate states have opposite fermion parity.²

¹The proof is straightforward: given an energy eigenstate $|E\rangle$, we can construct another state $\mathsf{T}|E\rangle$; as T is by definition a symmetry, it commutes with the Hamiltonian, and therefore the new state has the same energy as the old one. And these two states are clearly linearly independent, because a putative relation of the form $\mathsf{T}|E\rangle = c|E\rangle$ for some constant c is incompatible with $\mathsf{T}^2 = -1$, inasmuch as $-|E\rangle = \mathsf{T}^2|E\rangle = |c|^2|E\rangle$ is inconsistent.

²The proof is analogous. The two states $|E\rangle$ and $\mathsf{T}|E\rangle$ have opposite signs under $(-1)^F$ as T by assumption anti-commutes with $(-1)^F$. This automatically precludes a linear dependence of the form $\mathsf{T}|E\rangle \propto |E\rangle$ as both sides are eigenvectors of $(-1)^F$ with different eigenvalue.

Other similar conclusions can also be established for other symmetry groups. For example, in chapter 1 we argue that any theory in d = 1 + 1 that is invariant under (chiral) parity and an odd number of fermions necessarily has a two-fold degenerate supersymmetric spectrum. Similarly, any theory in d = 2 + 1 invariant under time-reversal and an odd number of fermions also has a supersymmetric spectrum. The method we use to prove these claims is to deform the theory into a system of free fermions, and analyzing the realization of the symmetry group in the free case. We also argue that this result is consistent with some dynamical scenarios about the strongly-coupled regime of these theories that had previously been suggested in the literature. For example, it had been conjectured that some gauge theories flow to certain topological theories at large distances; we check that these topological theories are supersymmetric precisely when the original gauge theory has an odd number of fermions, as predicted by the general argument above.

To summarize the discussion so far,

We have argued that the symmetry group of a given quantum theory can and often is realized projectively, and that this projective action is protected under small deformations. The invariance under deformations allows us to explicitly compute these projective phases, by deforming into a simpler theory. And these projective phases, when present, imply non-trivial predictions about the properties of the theory, such as degeneracy in the spectrum of states. These predictions can then be tested against any potential dynamical scenario about the non-perturbative regime of the theory.

The second main result of chapter 1 is a discussion of how this relates to 't Hooft anomalies. The claim is that the projective phases are entirely determined by the anomalies of the symmetry group, and therefore they can be fixed in terms of the anomaly data, if the data is known. In practice, we can turn this around and use the projective phases, if they are known, to figure out the anomaly itself, which is often a more subtle task. Let us summarize the main points here.

In general, given a theory in d dimensions and symmetry group G, the anomalies of the theory take values in $\operatorname{Tor}(\Omega_{d+1}(G)) \oplus \operatorname{Free}(\Omega_{d+2}(G))$ (see e.g. [9] for a nice review). Here Tor and Free denote the torsion and free parts of Ω , respectively, and where $\Omega_n(G)$ denotes the n-dimensional bordism group of G, namely the space of n-dimensional manifolds that carry a G-structure, modulo bordism, where two manifolds are considered *bordant* if they can be obtained as the boundary of an n + 1-dimensional manifold that extends the G-structure. The bordism group $\Omega_n(G)$ is a generalized cohomology theory, similar in spirit to the classical group cohomology $H^n(G)$ but that requires a more refined mathematical treatment.

The factor $\operatorname{Free}(\Omega_{d+2}(G))$ is generically well-understood by now: it classifies the *local* anomalies, namely those that follow from triangle diagrams. In other words, the local anomalies are those that are captured by perturbation theory. As such, their computation and consequences are textbook material. By contrast, the factor $\operatorname{Tor}(\Omega_{d+1}(G))$, which

classifies *global* anomalies, is much harder to get a grip on. On the one hand, computing this bordism group is a rather non-trivial task by itself, requiring advanced mathematical tools; and on the other hand, even if we understood this group, establishing which element is chosen by a specific quantum theory is also very hard, since these anomalies are not captured by perturbation theory.

At the formal level, a possible path is the following. Consider a certain quantum theory with symmetry G and anomaly $\alpha \in \Omega_{d+1}(G)$. If we put this theory on a compact spatial manifold X_{d-1} (equipped with whatever structures are required to properly define the system, such as boundary conditions), we can associate a Hilbert space $\mathcal{H}(X_{d-1})$ to it via canonical quantization. This compactification of the theory yields an effective two-dimensional class $\tilde{\alpha}$ obtained by integrating the original class α over X_{d-1} ,

$$\tilde{\alpha} = \int_{X_{d-1}} \alpha \tag{0.1.5}$$

The two-dimensional class $\tilde{\alpha}$ has a simple interpretation: it describes the anomaly of the effective d = 1 theory obtained after compactification. But in d = 1, i.e., point-particle quantum mechanics, anomalies are essentially a synonym for a projective representation. In this sense, the projective realization of G on X_{d-1} is determined by the image $\tilde{\alpha}$ of α under the reduction along X_{d-1} . This does not fix α uniquely, since some information is lost after integration, but it does contain some partial information. We may be able to recover more information if we perform the same procedure for several different manifolds X_{d-1} and, more optimistically, it is conceivable that the full class α can be determined if we understand $\tilde{\alpha}$ for sufficiently many distinct spatial manifolds. At the very least, if $\tilde{\alpha}$ is non-trivial for at least one X_{d-1} (i.e., if the symmetry is actually realized projectively), then we can automatically conclude that α has to be non-trivial too, and thus the symmetry must have an anomaly.

Another interesting piece of information is contained in the isometries of X_{d-1} . In general, when we compactify a theory on a given spatial manifold, the isometries of the same descend to internal symmetries of the compact system. Therefore, the Hilbert space $\mathcal{H}(X_{d-1})$ actually realizes a representation of the extended group $G \times \text{Iso}(X_{d-1})$. This extended group can be realized projectively too, and this contains more information that just the projective realization of G alone. These extra projective phases typically give us more information about the original class α than the pure G part.

There is more to the story. There are several techniques to calculate the group $\Omega_{d+1}(G)$ in the mathematics literature. One that seems to be particularly useful in physics is the so-called *Atiyah-Hirzebruch spectral sequence* [10]. This spectral sequence has a nice physical interpretation in terms of decorating defects and junctions with invertible theories that keep track of the different obstructions that go into a proper definition of how the symmetry acts on the theory [11].

In very broad terms, the end result of the spectral sequence is not quite $\Omega_{d+1}(G)$ but rather the associated graded of a filtration thereof. Specifically, the sequence converges to quotients of the form $F_i\Omega(G)/F_{i-1}\Omega(G)$, where $0 \subset F_0\Omega(G) \subset F_1\Omega(G) \subset \cdots \subset F_d\Omega(G) \subset \Omega(G)$ is a filtration of $\Omega(G)$. The different successive quotients of this filtration turn out to have specific implications for the projective realization of the symmetry. In other words, the different projective phases that appear on the Hilbert space have a natural layer structure, and each layer has a specific image in the filtration of $\Omega_{d+1}(G)$.

To give a very concrete example, we can look at the time-reversal symmetry discussed at the beginning of this section, whose projective realization was argued to be

$$((-1)^F)^2 = 1, \qquad \mathsf{T}^2 = \pm 1, \qquad (-1)^F \mathsf{T} = \pm \mathsf{T}(-1)^F$$
 (0.1.6)

The bordism group predicts anomalies classified by $\Omega_2^{\text{pin}^-} = \mathbb{Z}_8$. The Atiyah-Hirzebruch spectral sequence yields three quotients isomorphic to \mathbb{Z}_2 , coming from the filtration $0 \subset \mathbb{Z}_2 \subset \mathbb{Z}_4 \subset \mathbb{Z}_8$ (i.e., $\mathbb{Z}_8/\mathbb{Z}_4 = \mathbb{Z}_4/\mathbb{Z}_2 = \mathbb{Z}_2/0 = \mathbb{Z}_2$). The three \mathbb{Z}_2 layers have specific effects on the projective realization of the time-reversal symmetry: two of them are just the two signs in (0.1.6), while the third \mathbb{Z}_2 is a more subtle phenomenon where the operator $(-1)^F$ happens to become ill-defined (in the sense that we cannot even realize it on the Hilbert space. This third sign does not tell us whether the representation is projective, but rather whether a representation is possible at all. We discuss this in more detail in chapter 1 but also give a very down to earth description in appendix A.)

For another example with a new twist, we can look at the (chiral) parity symmetry in d = 1 + 1 dimensions that we mentioned in passing earlier. The bordism group predicts that this symmetry has anomalies classified by $\Omega_3^{\text{spin}}(\mathbb{Z}_2) = \mathbb{Z}_8$. Interestingly, the projective realization of the parity symmetry on the Hilbert space only admits a single sign ± 1 in its group structure (as discussed right below (0.1.4)), and therefore it can only detect a single \mathbb{Z}_2 inside the full \mathbb{Z}_8 group. In this case, the isometry group of the spatial manifold turns out to be non-trivial, and when one looks at the projective realization of the extended symmetry group, the full \mathbb{Z}_8 anomaly emerges. This example highlights that in order to reconstruct the full anomaly given the projective representation of a symmetry, it is not always enough to look at the symmetry alone, but rather at how it interacts with the emergent symmetry of the compactified theory.

To summarize this second part of the discussion,

We have argued that the projective realization of a given symmetry group is entirely encoded in its 't Hooft anomaly. Anomalies are classified by a certain generalized cohomology theory, namely group bordism. This group has a natural layer structure, and this structure is inherited by the projective representation. If one already knows the anomaly of a system, one can use that knowledge to determine the projective realization. Conversely, if one understands the projective realization instead, one can use that information to reconstruct the anomaly.

This gives a convenient method to analyze some anomalies that could be hard to compute

otherwise. A projective phase in a symmetry group is a very concrete object, which can be fixed using very elementary tools, while an anomaly can be rather abstract and hard to approach given more traditional definitions. It also gives a somewhat explicit picture of what the anomalies imply, namely the failure for symmetries to compose as expected in the presence of a given background. This has the advantage that some consequences of the anomaly are easy to extract, since for example an extra sign in a group relation can immediately imply degeneracy in the spectrum, as argued above. Finally, this story also gives a window towards the intricacies of bordism theory, since some of its aspects sometimes have a simple physical interpretation.

We would like to make one final comment here. The fact that the class $\tilde{\alpha}$ that encodes the projective representation is obtained by integrating over X_{d-1} makes it clear that, if X_{d-1} is a boundary, then $\tilde{\alpha}$ is trivial. This means that if whatever structures are required to define the theory (such as a *G*-connection or spin structures) extend to one higher dimension, the symmetry group will not be realized projectively. This does not mean that the symmetry is non-anomalous, since $\tilde{\alpha} = 0$ does not imply that $\alpha = 0$, but rather that the chosen background was not able to detect the anomaly. In this sense, in order to be able to say anything about anomalies, we must always choose structures that do not extend. This is also clear from the physics point of view, since the non-bounding structures are precisely those that allow for zero-modes. In a compactified theory only these matter, and therefore in the absence of zero-modes the vacuum is trivial. Indeed, there are no non-trivial unitary onedimensional projective representations, so a unique ground state always realizes symmetries non-projectively.

This explains why we only give examples up to dimension d = 3. In d = 4 any orientable spatial manifold is bounding, hence unless we turn on non-trivial holonomies we will not observe any projective phases. One could try to find simple examples in d = 4 by looking at non-orientable spatial manifolds (as $\Omega_3^{\text{pin}_+} = \mathbb{Z}_2$ is non-trivial), which would detect anomalies for time-reversal symmetry in four dimensions. We do not explore this possibility in this thesis.

0.2 Chapter 2.

In chapter 2 we discuss some aspects of gauge theories in d = 1 + 1 dimensions. As $d \leq 4$, these theories are asymptotically free, which means that the perturbative loop expansion in terms of Feynman diagrams only gives useful information at high energies (i.e., short distances). By contrast, the behaviour of the theory at low energies (i.e., long distances) is much harder to explore, since the coupling constant grows and the perturbative expansion breaks down. The goal of chapter 2 is to try and say a few things about the low energy regime of QCD using non-perturbative tools.

The most basic question one may ask about a non-perturbative QFT is whether it is

gapped or not. This question controls the leading macroscopic behaviour of the system, since a gapless theory contains non-trivial phenomena at any energy scale, while a gapped theory has rather boring dynamics at energies below the gap. More specifically, a gapless system always flows to a conformal field theory at large distances, while a gapped one to a (possibly trivial) topological field theory. In this sense, the first question we would like to ask is whether QCD is gapless or not; and, in either case, perhaps more ambitiously, which specific CFT or TQFT describes its macroscopic dynamics.

A situation where the infrared can be studied reliably is when the mass of the quarks m is very large. Integrating out a massive field generates a local effective action for the remaining fields. In particular, what is left in QCD is an effective action for the gauge field; by gauge invariance and dimensional analysis, the only local term that survives the limit $m \to \infty$ is the gauge kinetic term.³ In other words, if the mass of the fermion is sufficiently large, the dynamics are given by those of pure Yang-Mills, up to a renormalization of the coupling constant and irrelevant corrections suppressed by powers of 1/m. Importantly, two-dimensional Yang-Mills is exactly solvable [12, 13] and, in particular, it is a gapped theory, with a unique vacuum.⁴ In conclusion, QCD with very massive fermions is a gapped theory with unique vacuum, and the infrared TQFT is trivial.

Given this observation, we learn that the really hard case is that of massless quarks.⁵ Are such theories gapless or not? And what is their infrared behaviour?

The main idea of chapter 2, somewhat oversimplified, is the following. First, we compactify the spatial direction, i.e., we put the theory on a cylinder of finite radius L. In this situation, and after some careful considerations regarding composite operators and gauge invariance, one can show that the Hamiltonian of QCD takes the form

$$H = g^2 L^3 \sum_{n>0} \frac{1}{n^2} |J_n^a|^2$$
(0.2.1)

where J_n are the Fourier modes of the gauge current $J^a = \bar{\psi} t^a \psi$,

$$J^{a}(x) = \sum_{n \ge 0} J^{a}_{n} e^{inx/L}$$
(0.2.2)

The conclusion is that the space of massless states, defined by H = 0, translates to the condition $J_n^a = 0$, as H is expressed as a sum of squares. Consequently, the massless part of the spectrum is obtained by setting $J^a(x) \equiv 0$:

³There is also theta terms for abelian factors, if any. One can also generate terms for background fields if they couple to the fermions, such as the Arf invariant for gravity.

⁴Abelian factors at $\theta = \pi$ lead to two-fold vacuum degeneracy.

⁵Note that the massless point is well-defined since there are symmetries that only exist at that point, such as time-reversal and (chiral) fermion parity. This shall not be the case for d = 2 + 1 that we will discuss later on.

Space of massless	_	Free algebra generated by ψ	(0, 2, 2)
operators in QCD	=	subject to the relation $J^a \equiv 0$.	(0.2.3)

Note that this is a purely algebraic problem. If we are given some gauge-invariant operator \mathcal{O} constructed out of the quark fields ψ , and such that this operator can be factorized as $\mathcal{O} = J^a \mathcal{O}^a$ for some other operator \mathcal{O}^a , then the observable \mathcal{O} creates massive excitations only, and it decouples at large distances. Conversely, if \mathcal{O} cannot be factorized in terms of gauge currents J^a , then that operator creates massless excitations and thus survives the infrared limit. With this in mind, a given theory is gapped if and only if there are finitely-many massless operators \mathcal{O} . Indeed, a non-trivial CFT always has an infinite number of operators, while a TQFT has a finite number of operators.

Importantly, this problem can be solved. One can show that a QCD theory defined by some gauge group G and quarks in some representation R has finitely-many massless operators (that is, finitely-many operators that do not contain the gauge current J^a) if and only if it is one of these:

- Any gauge group G and quarks in the adjoint representation R = adj.
- $G = S(U(N) \times U(M)), G = SO(N) \times SO(M), G = Sp(N) \times Sp(M)$, and quarks in the bifundamental representation $R = (\Box, \Box)$.
- G = U(N), G = SO(N), G = Sp(N), and quarks in the rank-2 representations $R = \Box \Box, \Box$.
- A finite list of isolated theories (e.g., G = Spin(9) and quarks in the spinor representation).
- Combinations of the above.

Any other choice of (G, R) will lead to a gapless theory (i.e., one which contains an infinite number of operators that do not contain gauge currents J^a).

The idea behind this classification is as follows. One can show that the space of operators described by (0.2.3) coincides with the space of operators of a well-known rational CFT [14–18], namely the theory whose chiral algebra is the Goddard-Kent-Olive coset⁶

$$\frac{\operatorname{SO}(\dim R)_1}{G_{T(R)}}\tag{0.2.4}$$

Here SO(dim R)₁ denotes the free chiral algebra generated by the fermion field ψ , while $G_{T(R)}$ the Wess-Zumino-Witten subalgebra generated by the currents $J^a = \bar{\psi} t^a \psi$. The quotient of

 $^{{}^{6}}T(R)$ is the Dynkin index of R, a certain group theory factor; we define it properly and give its value for several important representations in appendix B.

these two algebras precisely enumerates operators of the numerator (i.e., combinations of quark fields ψ) modulo operators of the denominator (i.e., combinations of gauge currents J^a). Let us stress that, at this point, we are not claiming that the coset (0.2.4) is the low-energy effective theory of QCD. We are only claiming that the infrared of QCD, and the coset (0.2.4), have the same spectrum of operators, as sets (i.e., without assuming that they describe the same dynamics). As the two theories have the same number of operators, they are either both gapped (if finite) or both gapless (if infinite).

With this in mind, a QCD theory is gapped if and only if its associated GKO coset $SO(\dim R)_1/G_{T(R)}$ has a finite spectrum. Luckily, the classification of finite GKO models is known, and it gives rise to the list above. Therefore, this completes the problem of figuring out whether a given gauge theory in two-dimensions is gapless or gapped.

Needless to say, this is not the end of the story. In order to understand the large distance dynamics of QCD, it is not enough to known whether it is gapless or not, but we would also like to know which degrees of freedom describe the infrared physics. At this point we know the space of massless operators, but not their properties at the infrared fixed-point. In other words, we know whether a given operator defined in the ultraviolet survives the infrared limit, but we do not known what are its dynamical properties in the infrared. For example, we do not know the scaling dimensions of operators at large distances.

A very natural conjecture for the infrared CFT is precisely the GKO coset (0.2.4). In other words, this chiral algebra not only classifies massless operators, but it also contains the CFT data thereof. We are thus lead to consider the conjecture [19]

Infrared of QCD
$$\equiv \frac{\text{SO}(\dim R)_1}{G_{T(R)}}$$
 (0.2.5)

Note that this conjecture contains our previous result (0.2.3), and extends it. The claim in (0.2.3) is reliable – it follows from the explicit analysis of the Hamiltonian of QCD – while the conjecture (0.2.5) is not entirely justified from first principles. That being said, the conjecture is the simplest CFT whose spectrum coincides with the reliable result (0.2.3) and, as we argue in chapter 2, it also matches all the 't Hooft anomalies of QCD, which makes the conjecture particularly attractive.

The conjecture (0.2.5), if correct, completely solves the problem of figuring out the macroscopic behaviour of QCD in two dimensions. If a given theory is in the list above, the infrared is gapped, while if it is not, it is gapless. And, moreover, in either case, the long distance degrees of freedom are those contained in the GKO coset $SO(\dim R)_1/G_{T(R)}$. If the theory is gapped, this coset is a TQFT, with a finite-dimensional Hilbert space that describes the vacua of the theory. Instead, if the theory is gapless, the coset describes a CFT, whose chiral algebra takes the form of a specific gauged-WZW model.

To give a few examples of what this looks like, consider the following two prototypical QCD theories:

• Adjoint QCD, namely one quark in the adjoint representation. This is in the list above, and therefore the theory is gapped. The coset that describes its space of vacua is

$$\frac{\operatorname{SO}(\dim G)_1}{G_h} \tag{0.2.6}$$

where h = T(adj). This coset describes a TQFT; by working out its properties one learns that its Hilbert space has dimension $2^{\operatorname{rank}(G)}$, so adjoint QCD has these many discrete vacuum states. The properties of this TQFT, such as the line operators and their correlators, can be extracted from the coset SO(dim G)₁/ G_h .

• Fundamental QCD, namely quarks in the fundamental representation. This is (generically) not in the list above, and therefore the theory is gapless. The coset that describes its massless degrees of freedom is

$$\frac{\mathrm{SO}(\nu N N_F)_1}{G_{N_F}} \tag{0.2.7}$$

where N is the number of colors, N_F the number of flavors, and $\nu = 1, 2, 4$ depending on whether the representation is real, complex, or pseudo-real, respectively. This coset describes a CFT; by level-rank duality, it turns out that it is in fact isomorphic to an ungauged WZW model for the flavor symmetry, at level N. (For example, QCD with G = SU(N) and fundamental quarks has $U(N_F)$ flavor symmetry, and therefore the infrared CFT is an $U(N_F)_N$ WZW model.)

This also partially solves the problem of light quarks, since there is a well-defined method to identify the image of the mass operator $m\bar{\psi}\psi$ in the coset (0.2.5). The mass operator is mapped into a certain operator in this coset, and therefore turning on small masses is equivalent to deforming the coset by that certain operator. The fate of the theory under such deformation (or any other deformation one may be interested in, such as four-fermi interactions), depends on the specific situation but is at least a well-defined problem, which can be worked out explicitly in a case-by-case analysis.

There are several consistency checks on this picture. For example, we can see that the gapped theories all have quarks in dimensions that grow, at most, quadratically with the number of colors. To some extent this was to be expected, since if the number of quarks grows faster than the number of colors, then for sufficiently large N the gluon dynamics are subleading, and the gap of the theory is controlled by the mass of the fermions. In particular, for massless fermions the theory is gapless, and therefore we should only find gapped theories for finitely many choices of N; these are precisely the isolated cases in the list of gapped theories.

Another simple consistency check regards the fact that none of the theories in the list above have continuous flavor symmetries. This was also to be expected since in two dimensions, any continuous (chiral) symmetry necessarily has an 't Hooft anomaly, a situation that is incompatible with a mass gap. More generally, one can show that the coset (0.2.5) has in fact the exact same symmetries (both discrete and continuous, zero-form and one-form) and anomalies (both local and global) as the original QCD system, which is a necessary condition for the latter to describe the macroscopic limit of the former.

$0.3 \quad \text{Chapter } \mathbf{3}.$

Our next object of study is gauge theories in d = 2 + 1 dimensions. As before, in three dimensions the theory is still asymptotically free, so the strongly-coupled regime corresponds to low energies, i.e., large distances. Our goal is the same as before, namely to try and say a few things about the non-perturbative regime of QCD. Some aspects are shared with the previous section, but some are new.

One of the most important elements of three-dimensional gauge theories is the existence of Chern-Simons terms [20–22]. This means that, on top of the standard gluon kinetic term $tr(F^2)$, one can also add terms of the form tr(FA). Three key properties of this last term are that: First, it has one fewer derivatives than the Yang-Mills kinetic term, hence it is more relevant at large distances. Second, it depends on A rather than F, so gauge invariance is not entirely manifest; in fact, checking invariance under large transformations yields the condition that the coefficient k of the Chern-Simons term must be integrally quantized. And, third, it turns out that the pure Chern-Simons theory, where the only term in the Lagrangian is tr(FA), is actually exactly solvable; the resulting theory is topological, in the sense that it has a finite-dimensional Hilbert space and observables only depend on the topology of spacetime. All of these properties will play an essential role below.

With this in mind, we are set to study the low energy dynamics of three dimensional QCD. The theory is specified by some gauge group G, which carries a coupling constant and a Chern-Simons term with coefficient k, plus some fermions in some representation R and some mass m. We shall denote the corresponding theory as $G_k + \psi_R$.

As in two-dimensions, a regime where one can study the theory reliably is the case of very large m. Integrating out massive matter fields yields an effective action for the gauge fields. By gauge invariance and dimensional analysis, the only local terms one can generate are the gauge kinetic term and a Chern-Simons term. Therefore, up to corrections of order 1/m, massive fermions only generate a renormalization of the coupling constant and the Chern-Simons coefficient. Importantly, the Chern-Simons coefficient is quantized, and therefore it cannot depend on the gauge coupling constant. This means that the renormalization, if any, can be computed in perturbation theory. By performing the corresponding one-loop computation, one learns [23, 24] that the Chern-Simons coefficient gets renormalized as $k \mapsto k + \operatorname{sign}(m)T(R)$, where $\operatorname{sign}(m) = m/|m|$ is the sign of the quark mass, and T(R) is the Dynkin index of the representation R (see appendix B for the definition of T(R) and its value for several important representations). Once the fermion has been integrated out, we can take the low-energy limit explicitly: since the gluon kinetic term is less relevant than the Chern-Simons one, we can drop the former,⁷ and we are left with a pure Chern-Simons theory, which is topological (so it does not flow). This is the end result: the macroscopic limit of the theory is a pure Chern-Simons theory at level $k \pm T(R)$, which is now exactly solvable.

So, to summarize,

A QCD theory with gauge group G, Chern-Simons level k, and quarks in a representation R with sufficiently large mass m, flows at long distances to a Chern-Simons TQFT:

$$\lim_{m \to \pm \infty} G_k + \psi_R = G_{k \pm T(R)} \tag{0.3.1}$$

Unlike in two-dimensions, here the large-mass regime is not entirely trivial, but rather it is given by a certain TQFT. In any case, we are still faced with the same situation, where the really hard case is that of light quarks.⁸ We cannot really follow the same strategy as before (for example, the 3d Hamiltonian is not as simple as the 2d one; and, moreover, 3d CFTs are not nearly as well-understood and constrained as 2d ones, so the gauged Wess-Zumino-Witten formulation would not be particularly useful anyway). So we must try something new.

The approach we take in chapter 3 is the following. To be concrete we choose the gauge group G = SU(N), since this is the group we study in that chapter, but the story for other groups is qualitatively identical. We begin by looking at the large mass case, where the infrared is given by $SU(N)_{k\pm T(R)}$, and ask ourselves what happens as we decrease the value of the mass. Clearly, there must be at least one phase transition, since the two large-mass phases $SU(N)_{k-T(R)}$ and $SU(N)_{k+T(R)}$ are distinct. One can in fact show that, for sufficiently large k, or large N_F , this is all there is [25, 26] (see also [27]): the two large-mass phases continue

⁷Needless to say, we can drop the gluon kinetic term in favor of the Chern-Simons term only when the latter has a non-zero coefficient. If for some reason we are faced with a theory with zero Chern-Simons level, then the gluon kinetic term is the most relevant interaction. This is a rare situation but it does play a role below, when we look at a $U(1)_0$ theory.

⁸Unlike in d = 2, here the massless point is in general not well-defined, since there are no special symmetries at that point (so the choice m = 0 depends on the renormalization scheme). The reason time-reversal does not exist is that the Chern-Simons term, being proportional to $\epsilon^{\mu\nu\rho}$, is not invariant. This does have an exception: given that time-reversal maps $k \to -k$, the case k = 0 does have a well-defined massless point m = 0, since at that point the theory becomes invariant under time-reversal, and hence m = 0 is protected from radiative corrections.

all the way down to some finite value of m, at which there is a unique phase transition:

$$\underbrace{\frac{\mathrm{SU}(N)_k + \psi_R}{m \to -\infty}}_{\mathrm{SU}(N)_{k-T(R)}} \xrightarrow{m \to +\infty} (0.3.2)$$

The degrees of freedom at the transition point can be studied reliably using 1/k or $1/N_F$ expansions.

What happens for small k and small N_F ? The most immediate possibility is that the diagram above continues to hold in this regime. In this case, the degrees of freedom at the transition point must still be non-trivial (e.g., in order to match several 't Hooft anomalies), but we can no longer study them in any controlled fashion. This seems like a roadblock. A more attractive possibility is that, for small k, N_F , a new phase opens up, and there are two transition points. The advantage of this scenario is that, if correct, it gives a consistent method to reliably understand the degrees of freedom at small mass. When doing so, one learns that these degrees of freedom pass many non-trivial consistency checks, which a posteriori justifies the assumptions that allowed us to obtain them in the first place.

With this in mind, let us assume that for small k and small N_F the phase diagram of the theory looks something like this:

$$\underbrace{\frac{\mathrm{SU}(N)_k + \psi_R}{m \to -\infty}}_{\mathrm{SU}(N)_{k-T(R)}} \xrightarrow{m \to +\infty} (0.3.3)$$

where the intermediate phase is to be determined.

A hint of what the dynamics at the transition point are can be obtained thanks to the so-called level-rank dualities of Chern-Simons (see e.g. [28]). The two asymptotic phases $SU(N)_{k\pm T(R)}$ have in fact an alternative description: these TQFTs can also be expressed in terms of a unitary gauge group:

$$\operatorname{SU}(N)_{k\pm T(R)} \longleftrightarrow \operatorname{U}(T(R)\pm k)_{\mp N}$$
 (0.3.4)

Under level-rank duality of two groups $G \leftrightarrow G'$, representations are exchanged as $R \leftrightarrow R^t \times f^r$, where t denotes the transpose of the Young diagram, f the transparent fermion (an invisible anyon that carries spin 1/2), and r is the rank of R (i.e., the number of boxes in its Young diagram). This invites us to guess that, perhaps, the full gauge theory at the transition point may also enjoy a non-trivial duality of the form

$$\operatorname{SU}(N)_k + \psi_R \quad \longleftrightarrow \quad \operatorname{U}(T(R) \pm k)_{k_1, k_2} + \mathcal{O}_{R^t}$$
 (0.3.5)

where \mathcal{O} is a boson if r is odd, and a fermion if even. This is clearly a very optimistic guess, and with little motivation so far, but it turns out to be surprisingly self-consistent as we shall see. At this point, the dual levels k_1, k_2 are unknown. The first level k_1 denotes the coefficient of the SU($T(R) \pm k$) subgroup, while the second level k_2 denotes the coefficient of the U(1) part. For very large mass, U(1) levels shift as $k_2 \mapsto k_2 \pm \frac{2}{T(R)\pm k} \dim(R^t)$.

The case of rank r = 1 fermions was analyzed in quite detail in [29], where the dual field \mathcal{O} is a boson (since r is odd). The analysis of fermions in rank r = 2 representations was initiated in [30], where the dual field \mathcal{O} is a fermion (since r is even). That paper studied the case of $N_F = 1$ copies of the adjoint representation, and also the symmetric and anti-symmetric representations for G = SO(N) and G = SU(N). The remaining rank-2 representations, namely the symmetric and anti-symmetric of SU(N), is the content of chapter 3. Higher rank representations are not expected to follow a similar pattern, since the number of fermionic degrees of freedom grows faster than the number of gluons, and therefore the gauge interactions are subdominant, so to leading order the system behaves more like weakly coupled fermions (and in particular it is unlikely that the system has a quantum phase similar to that of $r \leq 2$).

Let us, then, focus on the symmetric and anti-symmetric representations of SU(N). These are denoted as $R = \square$ and $R = \square$, respectively. The working hypothesis is that, close to the two transition points in (0.3.3), the system can also be described using dual variables

$$\operatorname{SU}(N)_k + \psi_R \quad \longleftrightarrow \quad \operatorname{U}(T(R) \pm k)_{k_1, k_2} + \psi_{R^t}$$
 (0.3.6)

where the dual field is a fermion since r is even. Here, the transposition R^t interchanges the symmetric and anti-symmetric representations. The dual levels k_1, k_2 can be fixed as follows. If we send $m \to \pm \infty$, then the original gauge theory flows to $SU(N)_{k\pm T(R)}$. If we assume that the dual mass is m' = -m, then sending $m \to \pm \infty$ is equivalent to $m' \to \mp \infty$, and therefore the dual theory flows to $U(T(R) \pm k)_{k_1 \mp T(R^t), k_2 \mp \frac{2}{T(R) \pm k} \dim(R^t)}$. As these two theories are supposed to describe the same phase, and given the level-rank duality in (0.3.4), we conclude that the dual level must be

$$k_{1} = \pm (T(R^{t}) - N)$$

$$k_{2} = \mp N \pm \frac{2}{T(R) \pm k} \dim(R^{t})$$
(0.3.7)

And now comes the key point: by sending the mass m' to infinity in the opposite direction, namely $m' \to \pm \infty$, we must land on the intermediate phase denoted by "???" in (0.3.3). The intermediate phase predicted by the two duals is

$$U(T(R) \pm k)_{\pm(2T(R^t) - N), \mp N \pm \frac{4}{T(R) + k} \dim(R^t)}$$
(0.3.8)

Apparently, the two duals predict a different phase, which would be an inconsistent situation. Luckily, there is another non-trivial equivalence of Chern-Simons theories, this time taking the form

$$U(A)_{B,B\pm A} \quad \longleftrightarrow \quad U(|B|)_{-A\operatorname{sign}(B),(-A\mp B)\operatorname{sign}(B)} \tag{0.3.9}$$

In order for the two theories in (0.3.8) to be dual to each other we must have

$$T(R) + k = -2T(R^{t}) + N$$

$$\frac{4}{T(R) + k} \dim(R^{t}) = T(R) + k + 2T(R^{t})$$
(0.3.10)

Using $T(\Box) = \frac{1}{2}(M-2)$, dim $(\Box) = \frac{1}{2}M(M-1)$, together with $T(\Box) = \frac{1}{2}(M+2)$, dim $(\Box) = \frac{1}{2}M(M+1)$, where M is the number of colors (either N or $T(R) \pm k$, depending on whether we are looking at the original fermion R or its dual R^t), one can check that these conditions are indeed satisfied! This is a very non-trivial coincidence of group theory factors, and it works out thanks to an independent level-rank duality of TQFTs, which is a very good sign that the conjecture is on the right track.

One thing that should be noticed is that the dual theories are $U(T(R) \pm k)$; of course, this only makes sense for $|k| \leq T(R)$, so this picture can hold at most in the range $k \in (-T(R), T(R))$. We conjecture that it does in fact hold in the whole range, while for any $|k| \geq T(R)$, the theory is in the semi-classical regime with only two phases (0.3.2). This is consistent with our general expectations, since we know that for sufficiently large k the theory definitely is in the two-phase regime. Note also that the dual theory is also given by a unitary group with a rank 2 representation, and in constructing the full phase diagram we assumed that it was in its two-phase regime. This is only self-consistent if |k| < T(R)holds if and only if $|k_1| \geq T(R^t)$ which, using the expressions above, is indeed correct. So this constitutes another non-trivial consistency check of the whole picture.⁹

To summarize, the proposed phase diagram looks like this:



where at each phase transition we write a weakly-coupled dual gauge theory, and in the intermediate phase the TQFT that this dual theory predicts, namely (0.3.8). The two duals

⁹Unfortunately, this also means that the dual theories are never both weakly-coupled simultaneously: one is in the semi-classical phase when and only when the other is in the quantum phase. So we cannot really test the duality using perturbative tools.

could in principle predict distinct intermediate phases, but they in fact predict the same, thanks to a level-rank duality. Otherwise the picture would not be self-consistent. Here k_1, k_2 are given by (0.3.7).

An interesting fact about the conjecture just described is the following. If we plug in the values of $T(R^t)$ and $\dim(R^t)$ into (0.3.8), we get the following intermediate quantum phase for the theory:

$$U(T(R) \pm k)_{\mp T(R)+k,2k}$$
 (0.3.12)

Recall that the original gauge theory is not in general invariant under time-reversal. Such operation takes $k \to -k$, so this is not a symmetry for non-zero k. Of course, we could formally restore time-reversal if, on top of reversing the orientation of spacetime, we also manually take $k \to -k$; in other words, we think of k as something like a spurion. We now look at the intermediate phase in (0.3.12). If we reverse the orientation of spacetime, this flips the sign of the two levels. If we also manually change $k \to -k$, we end up with $U(T(R) \mp k)_{\pm T(R)+k,2k}$, which is the exact same TQFT we began with. So the intermediate phase is consistent with time-reversal. In particular, for k = 0, the operation of time-reversal becomes an actual symmetry of the system, and the intermediate TQFT becomes timereversal invariant as well, which is a non-trivial fact since most TQFTs are not invariant under time reversal.¹⁰

Let us look at the time-reversal invariant point k = 0 in more detail. For this value of the level the TQFT in (0.3.12) becomes $U(T(R))_{\mp T(R),0} \equiv PSU(T(R))_{\mp T(R)} \times U(1)_0$. An interesting aspect of this theory is the factor of $U(1)_0$. This just describes a free, massless photon, with gauge group U(1) and no Chern-Simons coefficient. In three dimensions, a free photon is equivalent to a compact scalar (by Hodge duality, where a two-form is dual to a zero-form, i.e., a scalar). This massless scalar that appears at large distances is nothing but the Goldstone boson associated to the U(1) flavor symmetry of the original system, which acts as $\psi \mapsto e^{i\alpha}\psi$. So this conjecture predicts that this global symmetry is spontaneously broken, a phenomenon that is very hard to see semi-classically. So this constitutes a very concrete, non-trivial prediction about this gauge theory.

Needless to say, this whole story hinges on several educated guesses, so it is by no means a rigorous derivation. In order to gain confidence in the conjecture we must perform as many consistency checks as possible. We already mentioned some simple ones above. Another set of consistency checks come from embedding the d = 3 theories above into d = 4 gauge theories, whose large distance behaviour must be compatible with the dynamics predicted by the conjecture above. We will perform several quantitative tests of the four-dimensional picture in chapter 6, and also some more qualitative ones in chapter 3.

Going back to the pure d = 3 dynamics, another large set of tests is obtained by turning on background fields for the continuous symmetries. Given a continuous symmetry with

¹⁰In general G_k is not dual to G_{-k} , with $U(k)_{k,0}$ being an exception thanks to the level-rank duality (0.3.9), as can be seen by taking A = B = k and the bottom sign therein.

group K, one may add to the original Lagrangian any term constructed out of the background K connection. Non-trivial tests come from looking a Chern-Simons terms for the group K. As the fermions transform non-trivially under K, integrating them out will in general shift the the coefficient of the Chern-Simons terms for K. Furthermore, such coefficient also generically shifts under level-rank duality, and also under integrating out the fermions in the dual theory. All in all, if we follow the Chern-Simons term for K as we go around the conjectured phase diagram, its coefficient will change in the different phases. In particular, if we reach the intermediate phase from either the negative mass (left) or the positive mass (right) side of the diagram, we go through different shifts, and the accumulated shift may be different. Of course, the intermediate phase is unique, so consistency of the picture requires that the shift computed through the left part of the diagram must be identical to the shift computed through the right part of the diagram, which is a non-trivial consistency condition. Furthermore, given the spurion argument above, this accumulated shift must be in fact an odd function of k, since the coefficient of a Chern-Simons term is odd under time-reversal, i.e., under $k \to -k$. These two conditions – that the left and right shifts are equal, and that they are odd functions of k – are rather non-trivial tests of the phase diagram, and they can be checked for any continuous symmetry of the system. In chapter 3 we show that these conditions check out for the proposed phase diagram, and we also include the analysis of other phase diagrams that appeared in the literature, for completeness.

A final consistency check that we discuss is the matching of 't Hooft anomalies for discrete symmetries. Such anomalies are global in the sense of chapter 1, so they are somewhat subtle to analyze. We focus on the case of time-reversal for the k = 0 theory, which can be argued to have anomalies classified by $\Omega_4^{\text{pin}^+} = \mathbb{Z}_{16}$. The anomaly computed in the original ultraviolet variables must be equal to the anomaly of the intermediate TQFT, which is a very constraining condition for the phase diagram. In the theory studied above, the time-reversal invariant TQFT also includes a free scalar, which in a sense trivializes the test; but similar theories, where we look at other rank 2 representations, do not have a compact scalar and here the test is much more stringent. Therefore, in chapter 3 we also include an analysis of the \mathbb{Z}_{16} time-reversal anomaly and check for its matching.

0.4 Chapter 4.

As discussed in the previous section, three-dimensional gapped QFTs flow at large distances – in their strongly-coupled regime – to topological field theories. It is therefore worthwhile to study TQFTs in detail, since they describe the macroscopic dynamics of realistic systems. Conversely, an a priori understanding of the properties of TQFTs (such as dualities, and symmetries) is of much help in figuring out which TQFT appears in the low-energy limit of a given QFT.

There are other reasons one may be interested in TQFTs in their own right. For example,

it is possible to realize them in actual experiments, such as the Hall effect [31]. Moreover, they are an example of an exactly solvable QFT [22], and therefore they are a great playground to analyze generic features of QFTs, such as symmetries, anomalies, etc.

With these motivations in mind, the goal of chapter 4 is to determine the symmetries of abelian TQFTs. A TQFT is said to be *abelian* if it can be described by a Chern-Simons theory over an abelian gauge group. Equivalently, it is abelian if and only if every Wilson line admits an inverse, i.e., another line such that their combination is the identity [32]. Abelian theories thus form an abelian group, as opposed to a fusion category for non-abelian theories. Groups are more constrained than general fusion categories, and therefore the space of symmetries is much more manageable and, in a sense, rich. Generic TQFTs have very few symmetries, most often they are "trivial" symmetries associated to permutations of a Dynkin diagram, and the non-trivial symmetries are exceptional and hard to determine. By contrast, the symmetries of abelian theories are more accessible since they are (a subset of) the automorphisms of the abelian group above, which is a classic problem in group theory. The result is a very rich set of symmetries, often with fun connections to other areas such as number theory.

Let us summarize the main points of chapter 4. We begin by quickly reviewing some background material. One way to characterize an abelian TQFT is by its Chern-Simons presentation. The most general Lagrangian over an abelian gauge group is

$$\mathcal{L} = \frac{1}{4\pi} K_{ij} a^i \wedge \mathrm{d}a^j \tag{0.4.1}$$

where a^i is a tuple of U(1) gauge fields and K_{ij} is an integral symmetric matrix. This matrix specifies the TQFT. While in chapter 4 we deal with the general case, in this section we will simplify the discussion by assuming that all diagonal components of K are even; this ensures that the TQFT is bosonic, i.e., it does not depend on the spin structure of spacetime. The case of spin theories is almost identical, with some extra factors of 2 here and there.

The observables of a TQFT are the Wilson lines. In the abelian case, such a line is specified by a tuple of integers $\alpha_i \in \mathbb{Z}$, namely

$$W_{\alpha}(\mathsf{c}) = e^{i\alpha_i \oint_{\mathsf{c}} a^i} \tag{0.4.2}$$

where **c** is some cycle in spacetime. There are two crucial properties of $W_{\alpha}(\mathbf{c})$ that we need to describe:

• The operator W does not only depend on c, but there is also a dependence on the choice of trivalization of the normal bundle thereof [22]. The idea is that short-distance divergences force us to introduce a regulator. A choice that preserves topological invariance is to specify a vector normal to the line, which dictates the direction we are to deform it in case of coinciding singularities. The line thus becomes a ribbon. One

can show that the dependence on this choice of normal vector is rather mild: it merely introduces a (computable) phase:

$$= \theta(\alpha)$$

$$\alpha$$

$$(0.4.3)$$

where

$$\theta(\alpha) := e^{i\pi\alpha^t K^{-1}\alpha} \tag{0.4.4}$$

• Not every choice of $\alpha_i \in \mathbb{Z}$ leads to a distinct operator. Two Wilson lines W_{α} and $W_{\alpha'}$ have identical correlation functions if and only if $\alpha \equiv \alpha' \mod K$. The idea is that the propagator of two gauge fields is $2\pi K^{-1}$ (this being the inverse of the gauge kinetic term in (0.4.1)). Therefore, correlation functions of W_{α} can be expressed as a certain function of $e^{2\pi i \alpha K^{-1}\beta}$, where β is the charge of any other Wilson line. If $\alpha \equiv \alpha' \mod K$, then $e^{2\pi i \alpha K^{-1}\beta} \equiv e^{2\pi i \alpha' K^{-1}\beta}$ for all β , and $W_{\alpha}, W_{\alpha'}$ have the same correlators with respect to any other line. Thus, there are finitely-many distinct Wilson lines, which take values in the coset $\mathbb{Z}/K\mathbb{Z}$. (This coset is finite and has $|\det(K)|$ elements)

If we have two lines $W_{\alpha}(\mathbf{c})$ and $W_{\alpha'}(\mathbf{c}')$ such that \mathbf{c}, \mathbf{c}' are homologous (and the normal bundles are in the same homotopy class), then we can bring the cycles \mathbf{c}, \mathbf{c}' together, since the theory is topological. The end result is a single Wilson line, with charge $\alpha + \alpha'$:



Abelian theories are fully specified by these pieces of data we just described: the coset $\mathcal{A} := \mathbb{Z}/K\mathbb{Z}$, the operation $\mathcal{A} \times \mathcal{A} \to \mathcal{A}$ given by entry-wise addition $\alpha + \alpha'$, and the quadratic function $\theta \colon \mathcal{A} \to \mathrm{U}(1)$. Indeed, the space of all such structures is isomorphic to the space of integral symmetric matrices K (up to stable equivalence [33]). Any correlation function of a configuration of Wilson lines can be computed if we know the abelian group $(\mathcal{A}, +)$ and the

function θ . For example, using the two moves just described (0.4.3), (0.4.5), one can show that

where $link(c_1, c_2)$ is the linking number of the cycles c_1, c_2 (in this example, $link(c_1, c_2) = 2$). We can summarize this review of abelian theories as follows:

A general abelian TQFT is characterized by an integral symmetric matrix K. From this matrix one can extract a finite abelian group $(\mathcal{A}, +)$, with $\mathcal{A} = \mathbb{Z}/K\mathbb{Z}$, and a quadratic, homogeneous function on this group $\theta(\alpha) = \exp(i\pi\alpha^t K^{-1}\alpha)$, with $\alpha \in \mathcal{A}$. Any correlation function in the theory can be calculated in terms of these basic pieces of data, namely undoing self-twists (0.4.3) and fusing lines (0.4.5).

We can now move on to the main point of chapter 4, to wit, classifying all symmetries of such TQFTs. Given the discussion above, we can characterize symmetries as follows. A symmetry is, by definition, an operation on the observables of the theory that leaves correlation functions invariant. In the case at hand, the observables are the Wilson lines, and therefore a symmetry is a permutation $\rho: \mathcal{A} \to \mathcal{A}$ such that

$$\langle W_{\rho(\alpha)}(\mathsf{c})\cdots\rangle \equiv \langle W_{\alpha}(\mathsf{c})\cdots\rangle$$
 (0.4.7)

where \cdots denotes any insertion of other Wilson lines. As we just reviewed, this correlation function can be determined entirely in terms of fusion and θ , and therefore ρ will leave correlation functions invariant if and only if

$$\rho(\alpha + \alpha') = \rho(\alpha) + \rho(\alpha') \mod K, \qquad \theta(\rho(\alpha)) = \theta(\alpha) \tag{0.4.8}$$

for all $\alpha, \alpha' \in \mathcal{A}$.

Similarly, anti-unitary symmetries by definition reverse the orientation of spacetime, i.e., they are permutations that satisfy

$$\langle W_{\rho(\alpha)}(\bar{\mathbf{c}})\cdots\rangle \equiv \langle W_{\alpha}(\mathbf{c})\cdots\rangle^*$$
 (0.4.9)

where $\bar{\mathbf{c}}$ is the cycle \mathbf{c} with the opposite orientation, and the complex conjugation * is due to the symmetry being anti-unitary. Thus, we learn that ρ is an anti-unitary symmetry if and only if

$$\rho(\alpha + \alpha') = \rho(\alpha) + \rho(\alpha') \mod K, \qquad \theta(\rho(\alpha)) = \theta(\alpha)^* \tag{0.4.10}$$

for all $\alpha, \alpha' \in \mathcal{A}$.

In more abstract terms, an abelian TQFT is specified by its data $(\mathcal{A}, +, \theta)$, and a symmetry is an automorphism of the group $(\mathcal{A}, +)$ that leaves the quadratic function θ invariant, up to complex conjugation for anti-unitary symmetries. This formulation makes the problem straightforward: for a given abelian group, we first identify all the automorphisms, and then we check which of these leave the quadratic form invariant. The abelian group is finite, and so in the worst-case-scenario we brute-force through finitely many permutations. As usual, it is enough to check the generators, on which an action of a homomorphism determines an action on the whole group. The recipe is now clear: if we are given some abelian TQFT, we must first determine its abelian group and quadratic form data, and then we list all automorphisms thereof. We explain how to do this in practice for arbitrary theories, paying special attention to the case of spin (a.k.a. fermionic) TQFTs, that is, theories that depend on the spin structure of the underlying manifold. QCD with quarks is fermionic, and therefore the infrared TQFT most often is, too.

Let us illustrate this with the most basic abelian theory, $U(1)_k$, namely a Chern-Simons theory with gauge group U(1) and level k. This theory admits both a bosonic and a fermionic version. In order to simplify the presentation, we do not make a distinction between these two options here, and also drop some factors of 2. The abelian group is $\mathcal{A} = \mathbb{Z}_k$, and the quadratic form is $\theta(\alpha) = \exp(\pi i \alpha^2/k)$ for $\alpha \in \mathbb{Z}_k$, where we use additive notation for the group law. A general automorphism of \mathbb{Z}_k is a map of the form $\alpha \mapsto q\alpha$ where q is an integer coprime to k, and insisting that this map leaves the quadratic form invariant yields the condition $q^2 = \pm 1 \mod k$, where the upper sign corresponds to unitary symmetries and the lower one to anti-unitary ones. Solving the congruence $q^2 = \pm 1 \mod k$ yields all the symmetries of the theory.

Unitary symmetries are given by the solutions to $q^2 = +1 \mod k$. There is always at least one non-trivial solution, namely q = -1. This corresponds to the usual charge-conjugation symmetry of gauge theories, which more traditionally acts as $a^i \mapsto -a^i$. The equation $q^2 = +1$ mod k may or may not admit solutions other than $q = \pm 1$; when it does, the permutation $\alpha \mapsto q\alpha$ will not be a symmetry of the Lagrangian \mathcal{L} , since the only transformations that leave $\frac{k}{4\pi}a \wedge da$ invariant are $a \mapsto \pm a$. Therefore, charge-conjugation is the only symmetry of the classical theory, and other symmetries, if any, will only be symmetries of the quantum theory.

Anti-unitary symmetries, on the other hand, are given by the solutions to $q^2 = -1 \mod k$. This equation does not always admit solutions. One can show that this congruence equation admits solutions if and only if all prime factors of k are congruent to 1 modulo 4, i.e., only for those values of the level is the theory time-reversal invariant. Furthermore, one can argue that the number of unitary – and anti-unitary, if any – symmetries is $2^{\omega(k)}$, where $\omega(k)$ is the number of distinct prime factors of k.

The first few values of k for which $U(1)_k$ is time-reversal invariant are $k = 1, 2, 5, 10, 13, \ldots$. Indeed, these are the first few integers for which the congruence $q^2 = -1 \mod k$ admits solutions. As before, this symmetry is somewhat unusual since it again does not leave the Lagrangian invariant. This means that, for those special values of k above, the symmetry is a quantum symmetry, which is not present in the classical theory. It is a non-trivial self-duality of the theory. Another exotic aspect of this symmetry is that, except for k = 2, it is an order-four operation, i.e., it satisfies $T^4 = 1$ as opposed to the more standard algebra $T^2 = 1$.

We thus learn that $U(1)_k$ generically has lots of symmetries, and their type and number depends on the arithmetic properties of the level k. Most of these symmetries are quantum symmetries, i.e., they are transformations that leave correlation functions invariant, even if they do not leave the Lagrangian invariant in any meaningful sense. That being said, it is interesting to note that TQFTs often have lots of equivalent descriptions: theories described by different matrices K_1, K_2 may in fact give rise to the same theory, via a non-trivial duality. This is so if the matrices K_1, K_2 are associated to the same topological data $(\mathcal{A}, +, \theta)$. A symmetry that is non-Lagrangian with respect to the matrix K_1 may become Lagrangian with respect to the matrix K_2 . In other words, whether a symmetry is classical or quantum depends on the duality frame we choose to describe the theory, and is not intrinsic to the theory itself. To give a very concrete example, consider the theory $U(1)_k$. This theory is in fact equivalent to another Chern-Simons theory, described by three gauge fields coupled via

$$K = \begin{pmatrix} k & 0 & 0 \\ 0 & 0 & 1 \\ 0 & 1 & 0 \end{pmatrix} \tag{0.4.11}$$

In the duality frame of K, the symmetry $\alpha \to q\alpha$ becomes a Lagrangian symmetry, since it can be effected by the change of variables

$$\begin{pmatrix} a^{1} \\ a^{2} \\ a^{3} \end{pmatrix} \mapsto Q \begin{pmatrix} a^{1} \\ a^{2} \\ a^{3} \end{pmatrix}, \qquad Q := \begin{pmatrix} q & -p & 1 \\ pk & \frac{1}{2}p(-q\pm 1) & \frac{1}{2}(q\pm 1) \\ -k & \frac{1}{2}(q\pm 1) & \frac{1}{2p}(-q\pm 1) \end{pmatrix}$$
(0.4.12)

where p is an integer such that $q^2 = 1 + 2pk$. Although it may not be obvious at first glance, the matrix Q is integral and unimodular, which means that it is a valid change of variables (it respects the periodicity conditions on the U(1) gauge fields a^i). It is easily checked that the matrix K satisfies $Q^t KQ \equiv K$, and therefore the change of variables (0.4.12) leaves the Lagrangian $a^t K da$ invariant. On the first gauge field a^1 , the change of variables acts as $a^1 \rightarrow qa^1 + \cdots$, and therefore this redefinition implements the symmetry $\alpha \rightarrow q\alpha$, as required. The conclusion is that, while this unitary symmetry is non-Lagrangian with respect to U(1)_k, it becomes Lagrangian with respect to a dual description $K = k \oplus \sigma_x$.

The general case is conceptually very similar. The abelian group for arbitrary abelian TQFTs is always a product of cyclic group, and symmetries are defined by polynomial congruences on these groups, roughly of the form $Q^2 = \pm 1 \mod K$, with matrices Q, K. Solving this system of congruences classifies all symmetries; the end result depends sensitively on the number-theoretic properties of the levels of the Chern-Simons terms. Generically, these symmetries are quantum in the sense of the previous paragraph: they are symmetries of the theory but not of the classical Lagrangian. In a suitable dual description, some symmetries

may become Lagrangian. And, as above, the symmetries often satisfy exotic algebras, such as $T^4 = 1$, or even non-abelian algebras. See chapter 4 for a long list of examples.

This completes the problem of determining the symmetries of abelian theories. While interesting in its own right, it is also useful to try and apply the conclusions to more general QCD theories. We can list a couple of examples here. Two abelian theories that appear in the low-energy regime of QCD are U(1)₂ and U(1)₅, which describe the macroscopic dynamics of SU(2) plus an adjoint fermion, and PSp(5) plus an anti-symmetric fermion, both at the m = k = 0 point [30] (see also section 3.6 for a review). As the gauge theory is time-reversal invariant, so must the low-energy TQFT. And, using the results of this chapter, this is indeed seen to be the case, since k = 2, 5 both satisfy the condition for time-reversal invariance of $U(1)_k$. Moreover, $U(1)_2$ has order-two time-reversal invariance while $U(1)_5$ has order-four (i.e., $T^2 = 1$ and $T^4 = 1$, respectively). This is precisely what one expects from the gauge theory point of view, since PSp(5) is not simply-connected, and the corresponding magnetic symmetry modifies the time-reversal algebra from $T^2 = 1$ to $T^2 = \mathcal{M}$, where $\mathcal{M}^2 = 1$ [34]. On the other hand, SU(2) is simply-connected and thus there is no magnetic symmetry, and the time-reversal algebra is the standard order-two algebra. We also discuss other possible applications of our results to other gauge theories in chapter 4.

0.5 Chapter **5**.

In chapter 5 we continue our study of topological theories in three dimensions. The motivation is the same as in the previous section: these theories describe the macroscopic regime of gapped QFTs, so any knowledge we can extract from the former will potentially teach us something about the latter. Furthermore, these theories share many of the features of more complex QFTs while being exactly solvable, which makes them a great arena to explore general properties of quantum systems.

To be concrete, the topic of chapter 5 is the dependence of TQFTs on the *spin structure* of spacetime [35]. Gauge theories that contain quarks require spin structures to be defined (i.e., a choice of boundary conditions for fermions), and therefore their macroscopic effective description often require a spin structure too. Topological theories that depend on the spin structure – known as *spin* TQFTs or *fermionic* TQFTs – go beyond the scope of traditional TQFTs – known as *bosonic* TQFTs – which depend only on the topology of spacetime. In other words, spin TQFTs are more sensitive to the underlying manifold than bosonic ones. This means that their construction requires a more refined treatment than the traditional approach.

In chapter 5 we describe how spin TQFTs are constructed, with special emphasis to their Hilbert space. This space is a super-vector space, i.e., it is \mathbb{Z}_2 -graded, owing to the fact that the states in fermionic theories can be labelled by their fermion parity, i.e., by whether they are a boson or a fermion. This super-vector space only depends on the topology of the spatial manifold and the associated spin structure, up to isomorphism. Large diffeomorphisms that leave the topological space invariant induce isomorphisms of the associated Hilbert spaces, which must be compatible with the \mathbb{Z}_2 grading. The specific form of these isomorphisms – known as the *modular data* of the theory – is crucial for solving the theory on arbitrary manifolds (which can always be reconstructed by cutting it into simpler manifolds, and pasting them back together via a suitable diffeomorphism [22]).

The degrees of freedom of three-dimensional TQFTs are line operators, also known as *anyons*. The distinguishing feature of a fermionic TQFT, as opposed to a bosonic one, is that the line operators come in two different types. The "regular" anyons, which we refer to as *Neveu-Schwarz* lines, behave quite similarly to the anyons of a bosonic theory. They are topological lines, in the sense that correlation functions only depend on the global structure of the configuration of operators. By contrast, fermionic theories also contain a different type of anyons, which we call *Ramond* lines, which are not genuine line operators but rather they exist at the end of the topological surface that implements fermion parity $(-1)^F$:



We denote the set of Neveu-Schwarz anyons as \mathcal{A}_{-} and the set of Ramond anyons as \mathcal{A}_{+} . Similarly to bosonic TQFTs, these anyons have associated topological data, such as S and T matrices, which measure the response of the theory to large diffeomorphisms.

A nice example of a spin three-manifold whose partition function can be evaluated explicitly is the lens space $L(a, 1) \equiv S^3/\mathbb{Z}_a$. For a even this space has two spin structures, and the corresponding partition functions read

$$Z[L(a,1);\pm] \equiv \sum_{\alpha \in \mathcal{A}_{\pm}} (T_{\alpha})^{-a} (S_{0,\alpha})^2$$
(0.5.2)

This shows that, if we understand the sets \mathcal{A}_{\pm} and the associated topological data S, T, we can compute the partition function on the lens space with spin structure \pm . More generally, the same is true for any three-dimensional spin manifold: any observable can be determined in terms of the anyons \mathcal{A}_{\pm} and the topological data of the theory (i.e., the action of large diffeomorphisms).

How can we determine the topological data of a fermionic theory, in practice? The strategy we follow in order to construct fermionic theories is the following [36, 37]. If we have
one such theory to begin with, one can always sum over all possible spin structures, which yields a TQFT that no longer depends on the spin structure. In other words, summing over spin structures turns a fermionic TQFT into a bosonic TQFT. Importantly, this process can be reversed: one can also turn the bosonic TQFT into the fermionic one back. The way one does this is by gauging a suitable one-form symmetry, an operation that is dual to summing over spin structures. Specifically, this gauging means summing over all possible two-form gauge fields for the symmetry. In conclusion, any fermionic TQFT can be obtained by a suitable sum of bosonic TQFTs. As the latter are very well-understood, we can use this approach to give an explicit construction of the former. For example, we show how all the modular data of the fermionic theory can be written entirely in terms of the modular data of the dual bosonic theory. Similarly, the fermionic Hilbert space is a direct sum of suitable bosonic Hilbert spaces.

From this point of view, the Neveu-Schwarz lines of the fermionic theory are the anyons of the bosonic theory that are uncharged under the one-form symmetry, and the Ramond line those that are charged. This explains why, after gauging, the latter appear at the end of the $(-1)^{F}$ -surface.

To summarize,

Fermionic TQFTs always have an associated bosonic TQFT, and one can go back and forth between these two theories by suitable gaugings. Any object on either side of the duality can be expressed as a linear combination of similar objects on the other side of the duality. As bosonic theories are better understood than fermionic ones, one can use this correspondence to solve the latter using known facts about the former.

To given an explicit example of what the duality looks like, consider the following. Let $\mathcal{H}(\Sigma^{\alpha_1\cdots\alpha_n})$ denote the Hilbert space of the bosonic theory, quantized on an arbitrary (orientable) Riemann surface Σ with external punctures $\alpha_1, \cdots, \alpha_n$ (i.e., labelled boundary components). Let also $\hat{\mathcal{H}}(\Sigma_s^{\alpha_1\cdots\alpha_n})$ denote the super-Hilbert space, on the same surface, with spin structure s, as defined by the fermionic theory. Then these two spaces are related by

$$\bigoplus_{\boldsymbol{\varepsilon}\in H_1(\Sigma,\mathbb{Z}_2)} P_{\boldsymbol{\varepsilon},s} \left[\mathcal{H}(\Sigma^{\alpha_1\cdots\alpha_n}) \oplus \mathcal{H}(\Sigma^{\alpha_1\cdots\alpha_n\psi}) \right] = \hat{\mathcal{H}}(\Sigma_s^{\alpha_1\cdots\alpha_n})$$

$$\bigoplus_{s\in H^1(\Sigma,\mathbb{Z}_2)} \hat{\mathcal{H}}(\Sigma_s^{\alpha_1\cdots\alpha_n}) = \mathcal{H}(\Sigma^{\alpha_1\cdots\alpha_n}) \oplus \mathcal{H}(\Sigma^{\alpha_1\cdots\alpha_n\psi})$$

$$(0.5.3)$$

where $P_{c,s}$ are projectors of the form $P_{c,s} = s^c W_{\psi}^c$, with $s^c = \pm 1$ the spin structure around the cycle **c** and W_{ψ}^c the Wilson line around **c** associated to the transparent anyon ψ (the generator of the one-form symmetry). The projectors are defined in such a way that $W_{\psi}^c \equiv s^c$ on $\hat{\mathcal{H}}$, i.e., such that the transparent fermion around a cycle measures precisely the boundary condition around the same. In any case, these expressions (0.5.3) demonstrate the idea that summing over spin structures $H^1(\Sigma, \mathbb{Z}_2)$ "undoes" the sum over \mathbb{Z}_2 gauge fields $H_1(\Sigma, \mathbb{Z}_2)$, and vice versa. The modular data of the fermionic theory can be obtained by acting with the projectors $P_{c,s}$ on the data of the bosonic theory. This data is very important since it allows us to compute observables on arbitrary spin 3-manifolds, via surgery. In order to illustrate the procedure we explicitly calculate this data for several examples of common TQFTs, and then use it to compute the path integral on lens spaces, a non-trivial family of spin 3-manifolds.

While the construction of the Hilbert space as above is complete and self-contained, it might sometimes be time-consuming for high-genus surfaces. If the explicit basis is not needed, but just the total number of states, then there is a simple formula that computes that number directly for us. In the bosonic case it is the so-called Verlinde formula [38, 39]. In chapter 5 we also discuss the fermionic version of this formula, which gives us the number of bosons and fermions of an arbitrary spin TQFT, on an arbitrary surface, in terms of the torus modular data of the bosonic parent, which is much more accessible than the data for higher genus. (In particular, we only need the vacuum S-matrix, so the F-symbols are not required.)

As in the previous section, these results are definitely useful in their own right, since spin TQFTs are non-trivial examples of QFTs and therefore any progress towards their understanding is welcome. But, of course, these considerations become much more interesting when applied to other problems. We can in fact make contact with other chapters in this thesis. For example, by using our explicit construction of the Hilbert space of fermionic TQFTs, in chapter 1 we show that these spaces often realize projective representations of the symmetry group, signaling an 't Hooft anomaly for the symmetry, as reviewed in section 0.1. Similarly, in chapter 4 we determine the symmetries of abelian TQFTs, including fermionic theories. One aspect that was left unanswered in that chapter is how the symmetry group is extended by fermion parity, if at all, since this symmetry does not act on the anyons and hence cannot be detected by the methods in that chapter (cf. footnote 92 on p. 209). Using our refined understanding of fermionic TQFTs from this chapter, one could in principle answer this question, since the Hilbert space does in fact detect fermion parity (although we do not do this in this thesis). Finally, the strongest connection to other topics in this thesis is that with chapter 6. As we shall review momentarily, that chapter deals with four dimensional gauge theories and their domain walls. The Witten index of these walls is computed by counting states in the Hilbert space of certain three-dimensional fermionic TQFTs, a calculation that crucially utilizes the results of this present chapter.

0.6 Chapter 6.

The final chapter of this thesis concerns the concept of domain walls in d = 3 + 1 gauge theories. These walls are non-perturbative codimension-1 dynamical objects in a QFT, which generically support degrees of freedom in one lower dimension. Understanding the dynamics of these walls, and the localized degrees of freedom they host, is a strong-coupling problem which is in general very hard to address. The goal of chapter 6 is to try and say a few things about them in d = 3 + 1 dimensions, and make contact with dynamics of gauge theories in d = 2 + 1 dimensions as discussed above.

Consider a four-dimensional theory that has some discrete symmetry Γ that is spontaneously broken. As such, the system has $|\Gamma|$ vacuum sectors, labeled by elements $g \in \Gamma$; assume further that all these vacua are trivial, i.e., that the system is gapped and the vacua do not carry topological degrees of freedom. In this situation, one can define boundary conditions such that the fields approach different vacua in opposite asymptotic directions, say at $z \to \pm \infty$. This configuration is topologically protected since there is no finite-energy process through which it could decay everywhere to the same vacuum. From very far away, i.e., at low energies, all the energy density of the configuration is confined to a codimension-1 region of spacetime, orthogonal to the z direction. This defines a domain wall, that is, a dynamical surface that separates two vacuum states.

What are the dynamics of this surface? To leading order the surface will tend to minimize its tension, i.e., its area. In other words, the overall configuration of the surface is determined by the Nambu-Goto action, S = -TA. This center-of-mass mode can be interpreted as the Goldstone boson associated to the spontaneously broken translation symmetry along the z direction caused by the wall. The hard question is, what is there beyond this mode. Specifically, what are the degrees of freedom that are localized on the wall? Note that, at low energies, these degrees of freedom define a purely d = 2 + 1 dimensional theory, since they do not have enough energy to leave the wall, and the vacua on both sides of the wall do not have topological degrees of freedom which could interact with the d = 2 + 1 dimensional dynamics. We would like to identify this d = 2 + 1 theory explicitly.

We can begin by doing some kinematics [40]. A general wall is specified by the two vacua it separates, say $\gamma_1, \gamma_2 \in \Gamma$. In fact, the wall only depends on the combination $\gamma_1 \gamma_2^{-1}$, since we can use the affinely-realized symmetry Γ to relabel the vacua. So the wall really only depends on a single group element, and we can denote the wall as W_{γ} . Another simple fact is that the parity-reversal of the wall, denoted by $\overline{W_{\gamma}}$, can be obtained by $z \to -z$, i.e., by interchanging the elements γ_1, γ_2 . In other words, $\overline{W_{\gamma}} \equiv W_{\gamma^{-1}}$. So whatever the 3*d* theory on W_{γ} is, it must be such that, under time-reversal, it is mapped into $W_{\gamma^{-1}}$. This is as much as we can say without invoking actual dynamical arguments.

In order to make progress, let us be more concrete. Consider a gauge theory with gauge group G and massless quarks in some representation R. The case of fundamental quarks, $R = \Box$, was analyzed in [41]; there, it is assumed that the time-reversal symmetry was spontaneously broken, leading two two vacua and one domain wall. The 3d degrees of freedom were identified. Here we consider, instead, the case of rank-2 quarks, i.e., the adjoint representation or the symmetric and anti-symmetric representations $R = \Box , \Box$. Generally speaking, we do not expect non-trivial infrared dynamics for higher-rank representations, since the beta function is only negative for finitely many values of N.

We assume that the gauge group G is simply-connected, for otherwise the vacua will generically have topological degrees of freedom and domain walls are not well-defined as 3d theories. The 4d theory has an axial symmetry $\Gamma = \mathbb{Z}_{4T(R)}$, where T(R) is the Dynkin index of R (see appendix B for the definition of T(R) and its value for several important representations); this symmetry is what remains of the classical axial U(1) symmetry $\psi \mapsto e^{i\alpha\gamma_*}\psi$, which is explicitly broken – down to a discrete subgroup – in the quantum theory by a mixed anomaly with the gauge group.

We now make the following key assumption: the fermion bilinear $\psi^t \psi$ condenses. This is definitely true for the adjoint case [42–46], and this also implies the condensation for the other rank-2 representations for sufficiently large N, since all rank-2 theories are roughly the same to leading order in 1/N [47]. We assume that the condensation continues to hold for all finite N. The condensation of the fermion bilinear spontaneously breaks $\mathbb{Z}_{4T(R)} \to \mathbb{Z}_2$, where \mathbb{Z}_2 acts as $\psi \to -\psi$; this leads to 2T(R) discrete, trivial vacua, labelled by elements of $\mathbb{Z}_{4T(R)}/\mathbb{Z}_2 = \mathbb{Z}_{2T(R)}$. We thus have 2T(R) domain walls, labelled by this cyclic group.

What is the d = 2 + 1 theory that lives on these walls? We propose that the threedimensional theory supported on the worldvolume of W_n , where $n \in \mathbb{Z}_{2T(R)}$, is QCD with gauge group G and fermions in the representation R, together with a Chern-Simons term at (bare) level $k_0 = n$. This value of the Chern-Simons level can be motivated as follows (see diagram below). The domain wall W_n is defined by the condition that the fermions have nmore units of phase on one side than the other; we can therefore perform an axial rotation on half space, with parameter n, to bring both sides to the same phase. Due to the axial anomaly, the phase rotation induces a theta term, which is a total derivative. On half space, the total derivative localizes to the boundary; but the theta term is in fact a derivative of the Chern-Simons interaction, $F \wedge F = d(F \wedge A)$, and therefore the theta term with phase ninduces a Chern-Simons term with coefficient n on the boundary, that is, on the domain wall. In other words, the configuration that defines W_n is equivalent to having the same vacuum everywhere, plus a Chern-Simons term localized on the wall.



The first step defines the domain wall as the interface between vacua 1 and 2. On this configuration (second step), we perform an axial rotation on the bottom region only, with n units of phase, equal to the difference between vacua 2 and 1. This axial rotation generates a theta term, which is a total derivative and hence integrates to the boundary, namely the

interface. This yields a configuration where we have the same vacuum everywhere, but with a Chern-Simons term on a surface (third step).

To summarize the discussion so far,

A d = 3 + 1 gauge theory with simply-connected gauge group G and fermions in the rank-2 representation R has 2T(R) vacua labelled by $\mathbb{Z}_{2T(R)}$. The domain wall separating vacua differing by $n \in \mathbb{Z}_{2T(R)}$ units of phase supports a d = 2 + 1 dimensional gauge theory with gauge group G, fermions in the representation R, and a Chern-Simons term with bare coefficient n.

We often label the 3*d* gauge theory by the renormalized level, defined as the bare level minus T(R). In this convention, the claim is that $W_n \equiv G_{n-T(R)} + \psi_R$.

A simple consistency check of this proposal is the following. Recall that the kinematics of domain walls require that time-reversal acting on W_{γ} is equivalent to $W_{\gamma^{-1}}$; here $\gamma = n$ and $\gamma^{-1} = 2T(R) - n$, as $n \in \mathbb{Z}_{2T(R)}$. So we better have $\overline{W}_n \equiv W_{2T(R)-n}$. This is indeed satisfied by the conjecture $W_n \equiv G_{n-T(R)} + \psi_R$, inasmuch as time-reversal flips the sign of the (renormalized) Chern-Simons level, i.e., $\overline{G_k} \equiv G_{-k}$. So at this level of the analysis, the overall picture is seen to be self-consistent.

In chapter 3 we give some qualitative arguments that support the conjecture for the symmetric and anti-symmetric representations of the gauge group. In the case of the adjoint representation we can be much more explicit and give some quantitative evidence. The property that makes the adjoint case more tractable is that the theory is, in fact, supersymmetric – a gauge field plus an adjoint fermion define an $\mathcal{N} = 1$ vector multiplet. Being supersymmetric, there are some observables that can be computed exactly, which can then be used to test the proposal above.

The observable we shall use is the Witten index [42, 46]. This is defined as $I := tr(-1)^F e^{-\beta H}$, and it has the nice property that it is invariant under smooth (supersymmetry preserving) deformations. Indeed, in a supersymmetric theory all the excited states are paired-up in bose-fermi doublets, and therefore they cancel out in I, as $(-1)^F$ weights them with opposite sign. Therefore, the index is sensitive to the vacua only. Furthermore, any reasonable deformation can only add or remove vacuum states in bose-fermi pairs, so the index does not change under it. All in all, the Witten index is an observable of supersymmetric theories that does not change under small deformations, and this makes the index computable, since we can often deform the theory into one that we know how to solve. One can also study twisted Witten indices, where one turns on fugacities for flavor symmetries of the theory.

In chapter 6 we compute the (regular and twisted) Witten indices of the domain walls of adjoint QCD using both the 4d variables and the (conjectured) 3d variables. Consistency of the proposed 3d theory requires these two computations to agree. This matching also lends some evidence to the pure 3d dynamics discussed in the previous sections, since the computation of the Witten index requires knowledge of the infrared (the vacua) of these

theories, which we previously conjectured to be certain TQFTs via alternative methods. So the agreement of the Witten index for the domain walls not only tests the proposal that the domain wall supports specific 3d QFTs, but it also tests the proposal that these QFTs flow, at large distances, to specific TQFTs.

The matching of Witten indices is our quantitative check on the domain wall proposal for adjoint quarks. In chapter 3 we also describe the domain walls in the case of symmetric and anti-symmetric quarks. These theories are not supersymmetric, and therefore we cannot compute any partition function reliably and check for matching. That being said, in some cases we can give some milder evidence that the proposal is correct. We close this introductory chapter with a rough description of the argument presented in chapter 3.

Consider a 4d gauge theory with group SU(N) and a quark in the (anti-)symmetric representation. The domain walls are labelled by $n \in \mathbb{Z}_{N\pm 2}$, with the upper sign referring to the symmetric representation and the lower sign to the anti-symmetric. The TQFT that we derive in chapter 3 for this gauge theory is $U(n)_{N\pm 2-n,N\pm 2-2n}$. The most interesting wall is the time-reversal invariant one, $n = (N \pm 2)/2$, since for this value of n the TQFT becomes $U(n)_{n,0} \equiv PSU(n)_n \times U(1)_0$. The factor of $U(1)_0$ gives a U(1) gauge field with zero topological mass, which in three-dimensions is equivalent (by Hodge duality) to a compact scalar, namely a free, massless, circle-valued scalar field. This massless field is to be thought of as the Goldstone mode of the U(1) baryon symmetry. Hence, in this case the TQFT predicts that, while the baryon symmetry is unbroken in the bulk, it breaks on the wall!

As a matter of fact, there is a situation where this spontaneous breaking can be seen explicitly, directly in the 4d theory. In general, a semi-classical computation will not see a topological sector such as $PSU(n)_n$, which means that we can only hope to obtain a reliable semi-classical description if this topological sector is trivial. This TQFT is only trivial for n = 2, since $PSU(2)_2 \equiv SO(3)_1$ is an SPT. Recalling that $n = (N \pm 2)/2$, we are invited to look more carefully at the theories $SU(2) + \square$ and $SU(4) + \square$. We claim that in these two cases, the compact scalar $U(1)_0$ is visible on the wall in a reliable effective analysis.

What makes the two theories $SU(2) + \Box$ and $SU(4) + \Box$ special? The answer is that, in these two cases, the flavor symmetry is enhanced, since SU(2) = Spin(3) and SU(4) = Spin(6), and \Box, \Box are both equivalent to two copies of the fundamental representation. Hence, for these two values of N, the flavor symmetry is SU(2) instead of U(1). Importantly, a mass term preserves U(1) but not SU(2), and therefore the former cannot break spontaneously while the latter may. We assume that, indeed, SU(2) breaks to U(1), which leads to a vacuum manifold isomorphic to a sphere, $S^2 = SU(2)/U(1)$. In this description, the two vacua separated by the n = 2 domain wall are just two antipodal points on the sphere, and the domain wall itself is a geodesic connecting them. Importantly, this geodesic is not unique: all half circles connecting antipodal points have the same length, and thus there is an S^1 -worth of possible geodesics. This means that U(1) is spontaneously broken, the compact scalar corresponding to motion along the azimuthal direction:



(0.6.1)

Chapter 1 Global Anomalies on the Hilbert Space.

Authorship. The content of this chapter is taken almost verbatim from the paper [1], written in collaboration with Davide Gaiotto and Jaume Gomis.

Abstract. We show that certain global anomalies can be detected in an elementary fashion by analyzing the way the symmetry algebra is realized on the torus Hilbert space of the anomalous theory. Distinct anomalous behaviours imprinted in the Hilbert space are identified with the distinct cohomology "layers" that appear in the classification of anomalies in terms of cobordism groups. We illustrate the manifestation of the layers in the Hilbert for a variety of anomalous symmetries and spacetime dimensions, including time-reversal symmetry, and both in systems of fermions and in anomalous topological quantum field theories (TQFTs) in 2+1d. We argue that anomalies can imply an exact bose-fermi degeneracy in the Hilbert space, thus revealing a supersymmetric spectrum of states; we provide a sharp characterization of when this phenomenon occurs and give nontrivial examples in various dimensions, including in strongly coupled QFTs.

Acknowledgments. We would like to thank Theo Johnson-Freyd, Nathan Seiberg, Ryan Thorngren, and Alex Turzillo for useful discussions.

1.1	Introduction and Summary	35
1.2	Anomalies from Layers	43
1.3	Anomalies in free fermion Hilbert space	45
1.4	Anomalies in spin TQFT Hilbert space	57

1.1 Introduction and Summary

Consider a system with a classical global symmetry group G. Powerful constraints on the dynamics can be derived by coupling the system to a background connection A for the symmetry G. The system has an 't Hooft anomaly [7] if the non-invariance of the partition function under background gauge transformations generated by $g \in G$

$$Z[A] \mapsto e^{i\alpha(g,A)} Z[A] \tag{1.1.1}$$

cannot be cancelled by a local counterterm constructed out of the background fields. This is the physics that endows anomalies with a cohomological formulation [6, 48, 49].

The anomaly $\alpha(g, A)$ is a local functional of the background connection and of the transformation $g \in G$. An 't Hooft anomaly is captured via anomaly inflow [50] from a topological term in one dimension higher. Each such topological term can be thought of as the effective action characterizing a symmetry protected topological (SPT) phase [51–53] with symmetry G in one higher dimension. The topological term is gauge invariant on a closed manifold and reproduces the anomaly on a manifold with a boundary. Being topological, an 't Hooft anomaly is robust under deformations that preserve the symmetry, including renormalization group transformations. 't Hooft anomalies give physicists some of the very few clues into the nonpertubative dynamics of a quantum system.

A combination of insights from condensed matter physics, particle physics, quantum information and mathematics has culminated in a conjecturally complete answer to the problem of classifying the possible anomalies in various dimensions [52, 54–60]. This includes anomalies in bosonic as well as fermionic systems, for discrete and continuous internal symmetry groups as well as discrete spacetime symmetries such as time-reversal and parity.¹¹ This has led to the topological classification of anomalies in terms of cobordism theory and generalized cohomology theories [11, 36, 54, 56, 58–60, 65–70].

Consider first a bosonic system, one which can be defined without a choice of spin structure of the underlying manifold. By Wigner's theorem, symmetries come in two flavours: linear and unitary, or antilinear and antiunitary, with time-reversal being the prototypical example of an antiunitary symmetry. Thus, the symmetry data of a bosonic system is specified by the pair

$$(G, w_1),$$
 (1.1.2)

where G is a group and $w_1 \in H^1(G, \mathbb{Z}_2)$ a certain cohomology class $w_1 \colon G \to \mathbb{Z}_2$ that encodes the unitarity/antiunitarity of the group elements in G. The anomalies of a bosonic system with symmetry data (G, w_1) in D spacetime dimensions are classified by the twisted cobordism group [58]

$$\Omega_{\rm so}^{D+1}(G; w_1) \,. \tag{1.1.3}$$

¹¹Coupling to a time-reversal background requires defining the system on unoriented manifolds [58, 61-64].

In low spacetime dimensions, for $D \leq 2$, the anomaly classification reduces to group cohomology: $\Omega_{so}^{D+1}(G; w_1) = H^{D+1}(G, U(1))$, extending the classic result that anomalies in quantum mechanics (i.e. D = 1) are classified by $H^2(G, U(1))$, that is, by the projective representations of G.¹² In higher dimensions, $\Omega_{so}^{D+1}(G; w_1)$ can be reconstructed (losing some information about the addition law) from the Atiyah-Hirzebruch spectral sequence [10], that combines $H^{D+1}(G, U(1))$ with other cohomology groups of lower degrees.

Now recall the characterization of symmetries and classification anomalies of a fermionic system, which requires the choice of a (G-twisted) spin structure to be defined. A fermionic system has a universal and unbreakable \mathbb{Z}_2^F unitary symmetry generated by fermion parity, denoted by $(-1)^F$. This symmetry induces a \mathbb{Z}_2 -grading in the Hilbert space \mathcal{H} of fermionic systems, which become super-vector spaces. Since (classically) symmetries cannot change the fermion parity, that is $[g, (-1)^F] = 0$, the symmetry group G_f acting on the local operators of a fermionic system is necessarily a \mathbb{Z}_2^F central extension of a group G, such that $G = G_f/\mathbb{Z}_2^F$. Also, by virtue of Wigner's theorem, a symmetry can be either unitary or antiunitary. Therefore, the symmetries of a fermionic system are characterized by a cocycle $w_2 \in H^2(G, \mathbb{Z}_2^F)$ specifying the \mathbb{Z}_2^F central extension and by a cocycle $w_1 \in H^1(G, \mathbb{Z}_2)$ encoding the unitarity/antiunitarity of group elements. The anomalies of a fermionic system with symmetry data¹³

$$(G; w_1, w_2) \tag{1.1.4}$$

in D spacetime dimensions are classified by the twisted cobordism group [11, 59]

$$\Omega_{\rm spin}^{D+1}(G; w_1, w_2). \tag{1.1.5}$$

State-of-the-art mathematical techniques allow for the computation of these twisted cobordism groups; see [9, 71-73] for many relevant examples together with reviews aimed at physicists. A particularly convenient computational tool is again the Atiyah-Hirzebruch spectral sequence. The different ingredients that go into the computation of (1.1.5) in this spectral sequence can be given a nice physical interpretation in terms of layers in various dimensions (see below).

While the topological classification of anomalies is rather well understood, *detecting* whether a physical system is anomalous can be a difficult task. Intuitively, one has to keep track of all the arbitrary choices required for a sharp definition of the system on a general background and then quantify the topological obstruction to the trivialization of these choices. A concrete calculation may involve hard-to-determine data characterizing the system.¹⁴ While

¹²When w_1 is nontrivial the cocycle condition defining $H^*(G, K)$ is twisted by the action of w_1 , which acts as an involution on K. This action is nontrivial for K = U(1) and $K = \mathbb{Z}$ but trivial for $K = \mathbb{Z}_2$. In order to avoid clutter we do not write the twisting by w_1 explicitly.

¹³For example, for $G = \mathbb{Z}_2$, and taking w_1 and w_2 to be the nontrivial $\mathbb{Z}_2 = \{0, 1\}$ element in $H^1(\mathbb{Z}_2, \mathbb{Z}_2)$ and $H^2(\mathbb{Z}_2, \mathbb{Z}_2^F)$ yields the symmetry group generated by time-reversal T obeying $\mathsf{T}^2 = (-1)^F$, sometimes denoted by $\mathbb{Z}_4^{\mathsf{T}}$. This is the relevant symmetry group of the celebrated topological superconductors.

¹⁴For example, detecting anomalies in bosonic topological quantum field theories (TQFTs) requires knowing the F-symbols [74].

detecting the anomalies induced by transformations connected to the identity of a Lie group G is textbook material, the detection of *global* anomalies, which includes anomalies for all discrete symmetries, is more subtle [75, 76].¹⁵ The approach is often indirect, for example by embedding some global anomalies into perturbative ones (see for example the recent work [85] and references therein).

In this paper we exhibit an elementary method for detecting *some* anomalies, based on constructing the Hilbert space of the theory on a flat (spatial) torus T^{D-1} as well as determining how the algebra of symmetries is realized on the Hilbert space. This can be given the following physical interpretation. The anomaly of a *D*-dimensional theory can be represented by the class

$$\alpha_{D+1} \in \Omega^{D+1}_{\text{spin}}(G; w_1, w_2).$$
 (1.1.6)

Studying the Hilbert space of the D-dimensional anomalous theory on a spatial torus¹⁶ produces upon integration a class

$$\tilde{\alpha}_2 = \int_{T^{D-1}} \alpha_{D+1} \,. \tag{1.1.7}$$

The class $\tilde{\alpha}_2$ can be viewed as the effective anomaly class of a quantum mechanical theory in 0 + 1d, which we recognize from the properties of the Hilbert space. As a result, we expect to be able to detect this way all anomalies whose cobordism class can be recognized from the values on manifolds of the form $T^{D-1} \times \Sigma_2$, equipped with generic flat connections, spin structures, etc.

This perspective also shows that a torus compactification can provide useful anomaly information only if the relevant structures – either the background G connection or spin structure – do not extend to one higher dimension (i.e. if they are not the boundary of a manifold in one higher dimension). Indeed, if these structures were all bounding such that $T^{D-1} = \partial M^D$, then $\int_{T^{D-1}} \alpha_{D+1} = \int_{M^D} d\alpha_{D+1} = 0$, and the effective anomaly in 0 + 1dvanishes. This means that in order to detect the anomaly in the torus Hilbert space we must either turn on non-trivial holonomies for the symmetry G or we must consider periodic boundary conditions on the torus for fermionic theories – or both.¹⁷

In practice, we find that this method captures a surprisingly large amount of anomaly information. This is especially true for fermionic systems.

In order to illustrate how various anomalies are manifested in the Hilbert space, it is useful to recall some ingredients of the (partial) reconstruction of $\Omega_{\text{spin}}^{D+1}(G; w_1, w_2)$ via the

¹⁵In the case of antiunitary symmetries it requires, for example, learning how to define spin TQFTs on unoriented manifolds, which is an open problem. Numerous interesting partial results have been obtained, however [63, 77–84].

¹⁶One could also study the reduction of the anomaly class on more general manifolds, potentially detecting more anomalies.

¹⁷Turning on non-trivial holonomies for the symmetry G defines a G-twisted Hilbert spaces, which are the Hilbert spaces where to detect anomalies if the spin structures are bounding.

Atiyah-Hirzebruch spectral sequence. The starting point is a collection of layers in various degrees¹⁸ (see e.g. [11, 66, 67])

$$\begin{array}{l}
\vdots\\
\nu_{D-2} \in H^{D-2}(G,\mathbb{Z}) & p_x + ip_y \text{ layer}\\
\nu_{D-1} \in H^{D-1}(G,\mathbb{Z}_2) & \text{Arf layer} \\
\nu_D \in H^D(G,\mathbb{Z}_2) & \psi \text{ layer}\\
\nu_{D+1} \in H^{D+1}(G, \mathrm{U}(1)) & \text{Bosonic layer}
\end{array}$$
(1.1.8)

with nontrivial differentials connecting the various classes. Each layer has a physical and geometric interpretation (see section 1.2 for more details). In particular, the groups which appear in the second slot of $H^{D-k}(G, \cdot)$ are the groups of k-dimensional SPT phases with no symmetries. We summarize them in table 1.1.

	0 + 1d	1 + 1d	2 + 1d
SPTs	\mathbb{Z}_2	\mathbb{Z}_2	Z
generator	ψ	Arf	$p_x + i p_y$ aka SO(1) ₁
Z	$S_{\pm} \mapsto \pm 1$	$\Sigma \mapsto (-1)^{\operatorname{Arf}(\Sigma)}$	$M_3 \mapsto e^{i \mathrm{CS}_{\mathrm{grav}}[M_3]}$

Table 1.1: The first row gives the classification of SPT phases with no symmetries, the second the generators of the SPT classes, and the last the partition functions of the generators. S_{\pm} denotes a circle with periodic/antiperiodic (R/NS) boundary conditions; Σ is a compact Riemann surface, and Arf(Σ) is the Arf-invariant of the surface with spin structure, which evaluates to 0 on even and to 1 on odd spin structures; and M_3 is a three-manifold, with $CS_{grav} = \frac{1}{4\pi} \frac{1}{48} \int_{M_3} tr(\omega \, d\omega + \frac{2}{3}\omega^3).$

The endpoint of the Atiyah-Hirzebruch spectral sequence calculation is the associated graded of a filtration of $\Omega_{\text{spin}}^{D+1}(G; w_1, w_2)$: the addition law on the k-th layer is modified by unknown carry-overs from lower layers, which are somewhat tricky to compute. Physically, that means that even if the non-trivial differentials vanish, we can only really assign a specific value to ν_{D-k} if all $\nu_{D-k'}$ with k' > k vanish, or we can only discuss the difference in the ν_{D-k} anomaly of two theories for which all $\nu_{D-k'}$ with k' > k are the same.

We now demonstrate the Hilbert space manifestation of the layers in the anomalies of 0 + 1d fermionic systems with an antiunitary time-reversal symmetry T with $\mathsf{T}^2 = 1$, so that $G_f = \mathbb{Z}_2^{\mathsf{T}} \times \mathbb{Z}_2^{\mathsf{F}}$. The anomalies of such a system are classified by $\Omega^2_{\text{spin}}(\mathbb{Z}_2; 1, 0) = \Omega^2_{\text{pin}^-} = \mathbb{Z}_8$

¹⁸Recall that the action of w_1 is trivial on \mathbb{Z}_2 coefficients.

[86]. The anomaly arises from three layers

$$\nu_{0} \in H^{0}(\mathbb{Z}_{2}^{\mathsf{T}}, \mathbb{Z}_{2}) \simeq \mathbb{Z}_{2} \qquad \text{Arf layer}$$

$$\nu_{1} \in H^{1}(\mathbb{Z}_{2}^{\mathsf{T}}, \mathbb{Z}_{2}) \simeq \mathbb{Z}_{2} \qquad \psi \text{ layer} \qquad (1.1.9)$$

$$\nu_{2} \in H^{2}(\mathbb{Z}_{2}^{\mathsf{T}}, \mathrm{U}(1)) \simeq \mathbb{Z}_{2} \qquad \text{Bosonic layer}$$

We can think about these three groups as compiling into \mathbb{Z}_8 , corresponding to the binary expansion

$$\nu = \nu_0 + 2\nu_1 + 4\nu_2 \mod 8. \tag{1.1.10}$$

The simplest 0+1d system with anomaly $\nu \in \mathbb{Z}_8$ is a set of ν free massless Majorana fermions, with time-reversal acting as $\mathsf{T}(\psi(t)) = \psi(-t)$ on all ν fermions. At the level of operators, this system has symmetries generated by T and $(-1)^F$, these two operations commuting and being both of order two, i.e., $G_f = \mathbb{Z}_2^{\mathsf{T}} \times \mathbb{Z}_2^F$. At the level of the Hilbert space, the anomaly ν is manifested through the following anomalous pattern:

- $\nu = 1 \mod 2$: There is no graded Hilbert space \mathcal{H} . This arises from the Arf layer ν_0 .
- $\nu = 2 \mod 4$: There is a graded Hilbert space \mathcal{H} but the symmetry generators on \mathcal{H} do not commute. Instead, they anti-commute:

$$\{\mathsf{T}, (-1)^F\} = 0. \tag{1.1.11}$$

This arises from the fermion ψ layer ν_1 .

• $\underline{\nu} = 4 \mod 8$: There is a graded Hilbert space \mathcal{H} with $[\mathsf{T}, (-1)^F] = 0$ on it, but the symmetry algebra $\mathsf{T}^2 = 1$ is realized projectively on \mathcal{H} , that is

$$\mathsf{T}^2 = -1 \qquad \text{on} \qquad \mathcal{H}. \tag{1.1.12}$$

This arises from the bosonic layer ν_2 .

As we compactify higher-dimensional systems on tori, we will use this characterization to recognize the image of various anomalies.¹⁹

Let us explain our approach to detecting anomalies in the celebrated example of topological superconductors. Consider the anomalies of 2 + 1d fermionic systems with time-reversal symmetry T obeying $T^2 = (-1)^F$. The symmetry group is $G_f = \mathbb{Z}_4^T$, with $G = G_f/\mathbb{Z}_2^F = \mathbb{Z}_2^T$ and the symmetry is twisted by the nontrivial \mathbb{Z}_2 classes w_1 and w_2 in $H^1(\mathbb{Z}_2^T, \mathbb{Z}_2)$ and

¹⁹An elegant instance of this general idea is Witten's SU(2) anomaly [75], which is described by a cobordism class $\eta \wedge c_2(F)$ [73], that when integrated over a four sphere with background gauge fields with minimal instanton number, yields the SPT class η in 0 + 1d with no symmetries, which describes the ψ -phase (see table 1.1). Therefore the SU(2) global anomaly is detected as an anomaly in $(-1)^F$ due to a fermion zero mode in the instanton background and arises from the ψ -layer.

 $H^2(\mathbb{Z}_2^{\mathsf{T}},\mathbb{Z}_2^F)$. The anomalies are classified by $\Omega_{\text{spin}}^4(\mathbb{Z}_2,1,1) = \Omega_{\text{pin}^+}^4 = \mathbb{Z}_{16}$ [59, 87–89]. By anomaly inflow, this is the same as the classification of topological superconductors in 3 + 1d. The anomalies are constructed from the following layers

$$\nu_{1} \in H^{1}(\mathbb{Z}_{2}^{\mathsf{T}}, \mathbb{Z}) \simeq \mathbb{Z}_{2} \qquad p_{x} + ip_{y} \text{ layer}$$

$$\nu_{2} \in H^{2}(\mathbb{Z}_{2}^{\mathsf{T}}, \mathbb{Z}_{2}) \simeq \mathbb{Z}_{2} \qquad \text{Arf layer}$$

$$\nu_{3} \in H^{3}(\mathbb{Z}_{2}^{\mathsf{T}}, \mathbb{Z}_{2})) \simeq \mathbb{Z}_{2} \qquad \psi \text{ layer}$$

$$\nu_{4} \in H^{4}(\mathbb{Z}_{2}^{\mathsf{T}}, \mathrm{U}(1)) \simeq \mathbb{Z}_{2} \qquad \text{Bosonic layer}$$

$$(1.1.13)$$

These four groups compile into \mathbb{Z}_{16} , corresponding to the binary expansion

$$\nu = \nu_1 + 2\nu_2 + 4\nu_3 + 8\nu_4 \mod 16. \tag{1.1.14}$$

Anomalies $\nu \in \mathbb{Z}_{16}$ can be detected by studying the Hilbert spaces \mathcal{H}_{XY} of the theory on the two-torus T^2 , which depend on the choice of spin structure on T^2 , where $X, Y \in \{NS, R\}$. This gives rise to the Hilbert spaces associated to even spin structures $\mathcal{H}_{NS-NS}, \mathcal{H}_{NS-R}, \mathcal{H}_{R-NS}$, and to the odd spin structure \mathcal{H}_{R-R} . As explained above, anomalies can only appear in \mathcal{H}_{R-R} , as the other three spin structures are bounding.

The following anomalies can be detected on the Hilbert space, as we show in both the study of spin TQFTs and fermions in 2 + 1d:

• $\underline{\nu} = 1 \mod 2$: In $\mathcal{H}_{\text{R-R}}$ the classically $(-1)^F$ -even time-reversal symmetry generator T becomes $(-1)^F$ -odd, thus changing the parity of the states in $\mathcal{H}_{\text{R-R}}$. This corresponds to T anticommuting with $(-1)^F$ instead of commuting in $\mathcal{H}_{\text{R-R}}$:

$$\{\mathsf{T}, (-1)^F\} = 0. \tag{1.1.15}$$

This anomalous behaviour is associated with the $p_x + ip_y$ layer in (1.1.13). For ν even $[\mathsf{T}, (-1)^F] = 0$ on \mathcal{H}_{XY} .

• $\underline{\nu} = 2 \mod 4$: The $\mathbb{Z}_4^{\mathsf{T}}$ symmetry algebra on \mathcal{H}_{XY} is

$$\mathsf{T}^2 = (-1)^F \times (-1)^{\operatorname{Arf}(T^2)}$$
 on \mathcal{H}_{XY} . (1.1.16)

The symmetry algebra is undeformed on the Hilbert spaces with even spin structure and deformed in the Hilbert space with odd spin structure. This anomalous behaviour is associated with the Arf layer in (1.1.13). For $\nu = 0 \mod 4$, $\mathsf{T}^2 = (-1)^F$ on \mathcal{H}_{XY} .

• The next two layers ν_3 and ν_4 , corresponding to $\nu = 4 \mod 8$ and $\nu = 8 \mod 16$, are not visible on the torus Hilbert space and require other observables to detect them.

The analysis of anomalies for time-reversal symmetry $T^2 = (-1)^F$ in the Hilbert space of spin TQFTs [35] requires constructing \mathcal{H}_{XY} in the first place, and also learning how to compute the action of the operators (Wilson lines) on \mathcal{H}_{XY} . A systematic construction of the Hilbert space of fermionic TQFTs is presented in chapter 5, although we also quickly summarize the essential points in this chapter in section 1.4.

Another interesting example where the anomaly layers can be detected on the Hilbert space is in 1 + 1d fermionic systems with a unitary \mathbb{Z}_2 symmetry²⁰. The overall symmetry is $G_f = \mathbb{Z}_2 \times \mathbb{Z}_2^F$ and the anomalies are classified by $\Omega_{\text{spin}}^3(\mathbb{Z}_2, 0, 0) = \mathbb{Z}_8$ [59, 66, 92, 93], constructed from the layers

$$\nu_{1} \in H^{1}(\mathbb{Z}_{2}, \mathbb{Z}_{2}) \simeq \mathbb{Z}_{2} \qquad \text{Arf layer}$$

$$\nu_{2} \in H^{2}(\mathbb{Z}_{2}, \mathbb{Z}_{2}) \simeq \mathbb{Z}_{2} \qquad \psi \text{ layer} \qquad (1.1.17)$$

$$\nu_{3} \in H^{3}(\mathbb{Z}_{2}, \mathrm{U}(1)) \simeq \mathbb{Z}_{2} \qquad \text{Bosonic layer}$$

These three groups compile into \mathbb{Z}_8 , corresponding to the binary expansion

$$\nu = \nu_1 + 2\nu_2 + 4\nu_3 \mod 8. \tag{1.1.18}$$

The simplest example of a 1 + 1d theory with symmetry $\mathbb{Z}_2 \times \mathbb{Z}_2^F$ that realizes the $\nu \in \mathbb{Z}_8$ anomaly is a system of ν Majorana fermions. The generator of \mathbb{Z}_2 is the chiral symmetry $g = (-1)^{F_L}$, which acts trivially on the right-moving fermions and negates the left-moving fermions.

The anomaly $\nu \in \mathbb{Z}_8$ can be detected by studying the untwisted Hilbert space \mathcal{H}_X and the \mathbb{Z}_2 -twisted Hilbert space \mathcal{H}_X^g of the ν Majorana fermions, where $X \in \{NS, R\}$ labels the spin structure on the (spatial) circle. We observe the following pattern:

• $\underline{\nu = 1 \mod 2}$: The theory does not have proper graded twisted Hilbert spaces \mathcal{H}_{NS}^{g} and \mathcal{H}_{R}^{g} .

Also, while the untwisted Hilbert spaces \mathcal{H}_X are well-defined, $(-1)^{F_L}$ and $(-1)^F$ do not commute on \mathcal{H}_R

$$\{(-1)^{F_L}, (-1)^F\} = 0 \quad \text{on } \mathcal{H}_R.$$
(1.1.19)

For ν even \mathcal{H}_X and \mathcal{H}_X^g are properly graded and $[(-1)^{F_L}, (-1)^F] = 0.$

• The $\nu = 2 \mod 4$ and $\nu = 4 \mod 8$ layers are not visible on the Hilbert space as an anomalous realization of symmetry or a projective representation. Indeed reducing the H^3 class in (1.1.17) on the circle produces a trivial class in $H^2(\mathbb{Z}_2, U(1))$, signaling that there are no nontrivial projective representations of \mathbb{Z}_2 in \mathcal{H}_X or \mathcal{H}_X^g . We note, however, that the anomaly can be detected by measuring the spin of states in the twisted Hilbert spaces \mathcal{H}_X^g (see e.g. [94] for a similar discussion for bosonic systems). In an anomalous theory this spin has a fractional part, which means that the rotation symmetry is realized projectively.

 $^{^{20}}$ This symmetry is related to the two previous examples via the Smith map [90, 91]. The results below can also be derived using this perspective.

An interesting application of these results is the following. As explained above, some anomalies imply that the symmetry generator is fermion-odd in the Hilbert space \mathcal{H} with the appropriate (non-bounding) structure: the operator that implements the symmetry anticommutes with $(-1)^F$ instead of commuting. This immediately implies that the spectrum of the theory is supersymmetric, namely for any state in \mathcal{H} there is a partner with the same energy and with opposite fermion parity. This property of the theory is rather surprising: the bose-fermi degeneracy is a consequence of an anomaly instead of a conventional supersymmetry. This provides a unified perspective on several observations in the literature, and it leads to generalizations and new predictions:

- Any 0 + 1d theory with an antiunitary $\mathbb{Z}_2^{\mathsf{T}}$ symmetry and an odd number of Dirac fermions has exact bose-fermi degeneracy. The supersymmetric spectrum of an odd number of free Dirac fermions was described in [95], and has been studied more recently in [96–98]. From our perspective, any theory with a $\mathbb{Z}_2^{\mathsf{T}}$ anomaly $\nu = 2 \mod 4$ has a supersymmetric spectrum.
- Any 1 + 1d theory with a unitary chiral \mathbb{Z}_2 symmetry and an odd number of Majorana fermions has exact bose-fermi degeneracy in the untwisted Ramond Hilbert space \mathcal{H}_{R} . This includes the supersymmetric spectrum of SU(N) adjoint QCD with N even in 1 + 1d recently discussed in [99] (the spectrum is supersymmetric in spite of the fact that Lagrangian of adjoint QCD is not supersymmetric). The fact that a \mathbb{Z}_2 symmetric theory with an odd number of Majorana fermions has $\{(-1)^{F_L}, (-1)^F\} = 0$ in \mathcal{H}_R implies that any such a theory will have a supersymmetric spectrum. This includes examples in Yang-Mills with Spin(N) gauge group, e.g. in the fundamental representation for N odd and in the traceless symmetric representation for N = 0, 3mod 4. It would be interesting to exhibit this bose-fermi degeneracy explicitly.
- Any 2 + 1d theory with antiunitary $\mathbb{Z}_4^{\mathsf{T}}$ symmetry and an odd number of Majorana fermions has exact bose-fermi degeneracy in the odd spin structure Hilbert space $\mathcal{H}_{\mathsf{R-R}}$. This is a nontrivial prediction for the spectrum of gauge theories in 2 + 1d theories, which are strongly coupled in the infrared. An instance of a theory that should have a supersymmetric spectrum is $\mathrm{SO}(N)$ gauge theory (with vanishing Chern-Simons coupling) with a fermion in the traceless symmetric representation. Time-reversal invariance requires that N is even, and ν odd further requires that $N = 0 \mod 4$. While the Lagrangian of this theory is not supersymmetric, the anomaly implies that the spectrum is nonetheless supersymmetric. We can provide nontrivial evidence for this claim. In [30] the infrared dynamics of this theory was proposed to be captured by $\mathrm{SO}(\frac{N+2}{2})_{\frac{N+2}{2}}$ Chern-Simons theory. Using the formulae in [5] for the number of bosonic and fermionic states in $\mathcal{H}_{\mathrm{R-R}}$ of this Chern-Simons theory²¹ we find that the spectrum

²¹In [5] it was shown that SO(2n+1)_{2k+1} Chern-Simons theory has $\binom{n+k-1}{k-1}$ bosons and $\binom{n+k-1}{k}$ fermions

is indeed supersymmetric! Our argument applies to other gauge theories with higher rank real representations and is a nontrivial prediction of their spectrum.

The next layer, measuring the projectivity of the symmetry algebra on \mathcal{H} , also has nontrivial implications, the most famous being Kramers theorem. From our analysis one can conclude that any theory in 2 + 1d with $\mathbb{Z}_4^{\mathsf{T}}$ symmetry and anomaly $\nu = 2 \mod 4$ has (at least) two-fold degeneracy in the fermionic part of the even-spin-structure Hilbert spaces, and in the bosonic part of the odd-spin-structure Hilbert space. When $\nu = 0 \mod 4$ there is (at least) two-fold degeneracy for all the fermionic states, in any of the spin structures.

Finally, we should stress that our analysis may not yet capture all the information about anomalies which is encoded in the torus Hilbert spaces. Isometries of the internal space will act on the Hilbert space of a compactified theory. As a result, one could study anomalies for the combination of the original symmetries and the new internal symmetries of the compactified system. We leave this to future work.

The plan for the rest of the chapter is as follows. In section 1.3 we study free fermions in various dimensions and illustrate how anomalies manifest themselves at the level of their Hilbert space. We consider antiunitary time-reversal symmetry in 0 + 1 and 2 + 1 dimensions, where the algebra is $T^2 = 1$ and $T^2 = (-1)^F$, respectively; and we also consider unitary chiral symmetry in 1 + 1 dimensions with algebra $g^2 = 1$. After that, in section 1.4 we consider the same problem in 2 + 1d spin TQFTs. We study how their anomalies are seen by constructing their Hilbert spaces. Here we revisit the algebra $T^2 = (-1)^F$ and find the same behaviour as in the case of free fermions.

1.2 Anomalies from Layers

The classification of SPT phases in terms of generalized cohomology/cobordism and associated "layers" is somewhat forbidding, but has a rather transparent physical meaning. Ultimately, we want to have a procedure to associate a partition function to a manifold equipped with appropriate structures. First, we can triangulate the manifold, equipping it with a discretization of the various structures we want to endow it with: a flat connection along the edges of the triangulation, some discrete version of the spin structure and orientation, etc. Next, we can take the cell decomposition C dual to the triangulation, and place on the facets of C some collection of invertible TFTs (meaning here SPTs with no symmetries) of appropriate dimension, following some rules which take into account the discrete data we put on the manifold. The partition function is then defined as the partition function of the collection of invertible TFTs.

in the Hilbert space \mathcal{H}_{R-R} . The spectrum is supersymmetric for n = k, when the theory is time-reversal invariant.

The "layers" of the cohomology theory are simply a way to encode which rules we use to place invertible TFTs on facets. The differential in the generalized cohomology theory imposes the constraint that the final answer should be independent of the choice of triangulation as well as any other choices made at intermediate steps of the construction. It also identifies pairs of rules which give the same final answer.

As an example, consider orientable, spin SPTs for unitary symmetries. A discretized G flat connection is given as a collection of G elements on the edges of the triangulation.

- 1. If we were to include only the bottom layer, we would leave all facets bare and only focus on vertices of C. At each vertex we place some complex phase (aka elements of U(1)) determined by the group elements along the edges of the dual simplex. This is literally the cochain ν_{D+1} representing an element in the group cohomology $H^{D+1}(G, U(1))$. The cocycle condition ensures that the partition function defined as the product of all the phases is independent of the choice of triangulation and gauge. Coboundaries give partition functions which evaluate to 1 in a trivial manner.
- 2. Following [54], the next refinement of the story involves placing a fermionic onedimensional Hilbert space along some of the edges of C. The choice is the cochain ν_D representing an element in the group cohomology $H^D(G, \mathbb{Z}_2)$. The cocycle condition ensures that each vertex is connected to an even number of fermionic edges. At each vertex we now get to pick a vector in the (one-dimensional, Grassmann even) tensor product of these vector spaces. This is roughly the same as a choice of ν_{D+1} , but not canonically, because of sign ambiguities in the tensor product. The Grassmann combinatorics needed to rearrange the tensor products when contracting states at the endpoints of fermionic edges, as well as the (spin structure dependent) signs arising from fermion loops contribute to the overall sign of the partition function.
- 3. At the next level of refinement, we can place Arf theories on some two-dimensional facets according to some ν_{D-1} . The cocycle condition ensures that we have an even number of Arf facets impinging on an edge, but the edge must now carry a specific choice of how to gap the corresponding Majorana modes. The two possible choices have opposite Grassmann parity, so this choice is similar but not canonically equivalent to a choice of ν_D , etc. The evaluation of the partition function will now require a careful manipulation of the Majorana modes.
- 4. Next, we can place $SO(n)_{\pm 1}$ Chern-Simon theories on three-dimensional facets according to some ν_{D-2} . The cocycle condition insures that we have the same number of chiral and anti-chiral fermions at two-dimensional facets, but facets must now carry a specific choice of how to gap these 2*d* fermions. Two inequivalent choices differ by a factor of Arf. The evaluation of the partition function must now cope with this extra level of complication.

5. In principle, we could continue, selecting some invertible fermionic theories to place on four-dimensional facets, etc. In practice, no non-trivial invertible theories are expected to exist up to dimension 7, so we can safely stop here for most physical systems.

On general grounds, the differential in the generalized cohomology theory takes a triangular form, with the diagonal being the standard differential for $H^{D+1-k}(G, T_k)$, where T_k is the group of invertible theories we can place on the k-th dimensional facets. The off-diagonal components of the differential are non-trivial and somewhat tricky to compute far from the diagonal. Furthermore, the "stacking" operation on generalized cohomology classes, i.e. the sum of anomalies, is also defined in a triangular manner, with the diagonal being the usual operation of stacking invertible theories.

As one compactifies an SPT on, say, a circle, one can take a triangulation of the *D*dimensional manifold M and refine it to a triangulation of $M \times S^1$ in a systematic way. Applying the rules above to $M \times S^1$ and reducing them to some evaluation on the triangulation of M one can figure out the resulting SPT theory in one dimension lower. This was done for the Gu-Wen layer in [100], but has not been done in full generality.

1.3 Anomalies in free fermion Hilbert space

In this section we demonstrate how the Hilbert space on the torus detects a variety of anomalies in systems of free fermions in various dimensions.

1.3.1 Anomalous $\mathbb{Z}_2^{\mathsf{T}}$ in 0+1 dimensions

The anomalies of a fermionic system in 0+1 dimensions with an antiunitary time-reversal symmetry T with $T^2 = 1$, so that $G_f = \mathbb{Z}_2^T \times \mathbb{Z}_2^F$, are classified by $\Omega^2_{\text{spin}}(\mathbb{Z}_2; 1, 0) = \Omega^2_{\text{pin}^-} = \mathbb{Z}_8$. These anomalies arise from three layers

$$\nu_0 \in H^0(\mathbb{Z}_2^\mathsf{T}, \mathbb{Z}_2) \simeq \mathbb{Z}_2$$

$$\nu_1 \in H^1(\mathbb{Z}_2^\mathsf{T}, \mathbb{Z}_2) \simeq \mathbb{Z}_2$$

$$\nu_2 \in H^2(\mathbb{Z}_2^\mathsf{T}, \mathrm{U}(1)) \simeq \mathbb{Z}_2,$$
(1.3.1)

which generate the \mathbb{Z}_8 anomaly.

We shall study the \mathbb{Z}_8 anomaly in a system of free fermions. Related considerations can be found in [101–103].

Consider ν Majorana fermions in 0 + 1 dimensions

$$\mathcal{L} = \sum_{a=1}^{\nu} \frac{i}{4} \psi^a \partial_t \psi^a \,. \tag{1.3.2}$$

The theory has a $\mathbb{Z}_2^{\mathsf{T}}$ time-reversal symmetry which acts as²²

$$\mathsf{T}\psi^a(t) = \psi^a(-t)\mathsf{T} \tag{1.3.3}$$

and \mathbb{Z}_2^F fermion parity

$$\{(-1)^F, \psi^a(t)\} = 0.$$
(1.3.4)

It is known that a $\mathbb{Z}_2^{\mathsf{T}}$ -symmetric quartic interaction that gaps out the fermions can be written for $\nu = 8$ [86]. This realizes in the fermion system the \mathbb{Z}_8 anomaly expected from the cobordism classification.

Canonical quantization of (1.3.2) leads to a Clifford algebra of rank ν

$$\{\psi^a, \psi^b\} = 2\delta^{ab} \qquad a, b = 1, 2, \dots, \nu.$$
 (1.3.5)

We now proceed to identifying the anomaly layers (1.3.1). Each layer is implemented in a characteristic way in the fashion that symmetries are realized on the Hilbert space \mathcal{H} .

• $\nu = 1 \mod 2$. There is a rather severe anomaly for ν odd as the operator $(-1)^F$ generating the \mathbb{Z}_2^F symmetry does not exist. The theory does not admit a proper graded Hilbert space of states. Equivalently stated, the Clifford algebra of odd rank has two irreducible representations, and $(-1)^F$ exchanges them, instead of acting within an irreducible representation. This anomaly is associated with the $H^0(\mathbb{Z}_2^T, \mathbb{Z}_2) = \mathbb{Z}_2$ layer, the Arf layer.

This anomaly due to the lack of proper Hilbert space can also be detected by studying partition functions. Consider the partition function on the circle with antiperiodic (NS) and periodic (R) boundary conditions. The partition function with NS boundary conditions is²³

$$Z_{\rm NS} = 2^{\nu/2} \,. \tag{1.3.6}$$

Nominally, this partition function should count the number of states in \mathcal{H} , that is $Z_{\rm NS} = \operatorname{tr}_{\mathcal{H}}(\mathbf{1}) = \dim(\mathcal{H})$. The answer (1.3.6) mirrors the statement that there is no proper Hilbert space for ν odd, as $2^{\nu/2}$ is not an integer. Likewise, while the partition

²²For the purposes of studying anomalies it suffices to take all fermions to transform with the same sign under T. If a fermion is assigned the transformation $\mathsf{T}\psi_{-}(t) = -\psi_{-}(-t)\mathsf{T}$, we can then write a $\mathbb{Z}_{2}^{\mathsf{T}}$ -invariant mass term $i\psi_{+}\psi_{-}$ that couples a pair of fermions which transform with opposite signs under T. This lifts both fermions and therefore without loss of generality we can focus on a collection of fermions that transform with the same sign under T.

²³This partition function can be evaluated by taking the square root of partition function of 2ν Majorana fermions, which has a 2^{ν} -dimensional Hilbert space. It can also be computed by zeta-regularizing $Z \equiv Pf(i\partial_t)^{\nu} = \prod_{n \in \mathbb{Z}} \lambda_n^{\nu}$, where the eigenvalues of the 0 + 1d Dirac operator are $\lambda_n = n + 1/2$ in the NS sector, and $\lambda_n = n$ in the R sector.

function with R boundary conditions vanishes due to the presence of zero-modes, i.e. $Z_{\rm R} = {\rm tr}_{\mathcal{H}}(-1)^F = 0$, the correlator

$$\langle \psi^1 \psi^2 \cdots \psi^\nu \rangle_{\mathbf{R}} \tag{1.3.7}$$

is non-vanishing, as the insertions compensate the zero-modes. This observable (1.3.7) changes sign under the action of $(-1)^F$, signaling that $(-1)^F$ is anomalous as the \mathbb{Z}_2^F Ward identities are violated.

• $\nu = 2 \mod 4$. For ν even the theory has a well-defined Hilbert space and operator $(-1)^F$ acting on it. The Clifford algebra of even rank ν has a unique irreducible representation of dimension $2^{\nu/2}$, thus all representations are unitarily equivalent, and we can study the implementation of symmetries in any choice of basis. We can construct \mathcal{H} by defining the creation and annihilation operators

$$\psi_{\pm}^{A} = \frac{1}{2} \left(\psi^{2A-1} \pm i \psi^{2A} \right) \qquad A = 1, \dots, \nu/2,$$
 (1.3.8)

which obey

$$\{\psi_{+}^{A},\psi_{-}^{B}\} = \delta^{AB}, \ \{\psi_{+}^{A},\psi_{+}^{B}\} = \{\psi_{-}^{A},\psi_{-}^{B}\} = 0 \qquad A,B = 1,\dots,\nu/2,$$
(1.3.9)

We define the vacuum by

$$\psi_{-}^{A}|0\rangle = 0 \qquad A = 1, \dots, \nu/2,$$
 (1.3.10)

and create the whole module by acting with the different ψ_{+}^{A} on it. Time-reversal acts by exchanging the creation and annihilation operators (see (1.3.3) and recall that T is antilinear)

$$\mathsf{T}\psi_{\pm}^{A} = \psi_{\mp}^{A}\mathsf{T}\,.\tag{1.3.11}$$

This allows us to determine the action of T on the vacuum $|0\rangle$ by considering the most general state

$$\mathsf{T}|0\rangle = \alpha|0\rangle + \alpha_A \psi_+^A|0\rangle + \dots + \alpha_{12\dots\nu/2} \psi_+^1 \psi_+^2 \cdots \psi_+^{\nu/2}|0\rangle, \qquad (1.3.12)$$

for some yet-to-be-fixed coefficients $\{\alpha\}$. Acting on both sides with ψ_{-}^{A} and using (1.3.9), (1.3.10) and (1.3.11) we conclude that all but the last coefficient vanish, namely²⁴

$$\mathsf{T}|0\rangle = \alpha_{12...\nu/2}\psi_{+}^{1}\psi_{+}^{2}\cdots\psi_{+}^{\nu/2}|0\rangle, \qquad (1.3.13)$$

²⁴This corresponds to what is usually referred to as *particle-hole symmetry*: time-reversal exchanges ψ_+ and ψ_- , so a state full of ψ_+ is mapped to a state full of ψ_- , and vice-versa.

with $|\alpha_{12...\nu/2}| = 1$ so that state is normalized. Note that T adds $\nu/2$ fermionic modes, so it changes the fermion parity of the state if $\nu/2$ is odd. This implies that the $\mathbb{Z}_2^{\mathsf{T}} \times \mathbb{Z}_2^F$ symmetry generators on \mathcal{H} obey

$$\{\mathsf{T}, (-1)^F\} = 0 \quad \text{for } \nu = 2 \mod 4.$$
 (1.3.14)

and

$$[\mathsf{T}, (-1)^F] = 0 \quad \text{for } \nu = 0 \mod 4,$$
 (1.3.15)

Therefore, the anomaly corresponding to the $\nu = 2 \mod 4$ layer is detected by virtue of the operators T and $(-1)^F$ anticommuting in \mathcal{H} . This anomaly is associated with the $H^1(\mathbb{Z}_2^{\mathsf{T}}, \mathbb{Z}_2) = \mathbb{Z}_2$ layer, the ψ -layer.

ν = 4 mod 8. For ν = 4 mod 8 the theory has a proper Hilbert space and [T, (-1)^F] = 0. We now proceed to study how the Z₂^T symmetry is realized on the Hilbert space. Acting with T again in (1.3.12) yields

$$\mathsf{T}^{2}|0\rangle = |\alpha_{12\dots\nu/2}|^{2}\psi_{-}^{1}\psi_{-}^{2}\cdots\psi_{-}^{\nu/2}\psi_{+}^{1}\psi_{+}^{2}\cdots\psi_{+}^{\nu/2}|0\rangle = (-1)^{\nu/4(\nu/2-1)}|0\rangle.$$
(1.3.16)

Therefore, for $\nu = 4 \mod 8$ the $\mathbb{Z}_2^{\mathsf{T}}$ symmetry is realized projectively on the Hilbert space, that is

$$\mathsf{T}^2 = -1. \tag{1.3.17}$$

Therefore, the anomaly corresponding to the $\nu = 4 \mod 8$ layer is detected by virtue of the $\mathbb{Z}_2^{\mathsf{T}}$ symmetry being realized projectively on the Hilbert space.²⁵ This anomaly is associated with the $H^2(\mathbb{Z}_2^{\mathsf{T}}, \mathrm{U}(1)) = \mathbb{Z}_2$ layer, the bosonic layer.

SYK model. To close this section we can consider a very simple application of these results. One of the most well-known systems in 0 + 1d is the celebrated SYK model [104, 105], which consists of a system of N Majorana fermions interacting via four-fermi terms

$$\mathcal{L} = \sum_{a} \frac{i}{4} \psi^a \partial_t \psi^a - \sum_{abcd} J_{abcd} \psi^a \psi^b \psi^c \psi^d , \qquad (1.3.18)$$

where the coupling constants J are real. This Lagrangian is invariant under $\mathsf{T}(\psi^i(t)) = \psi^i(-t)$, and therefore all the conclusions from our previous discussion hold. The time-reversal anomaly of the system is $\nu = N \mod 8$. We immediately conclude that,

²⁵We note that $T^2 = -1$ for $\nu = 4, 6 \mod 8$ and $T^2 = 1$ for $\nu = 0, 2 \mod 8$. This follows from the properties of the antilinear involution acting on the Clifford algebra as $T^{-1}\gamma^a T = \gamma^a$, or equivalently $U\gamma^a U^{-1} = (\gamma^a)^*$, where we have written T = UK, with K denoting complex conjugation and U a unitary. Therefore, $T^2 = U^*U = \pm 1$ with $T^2 = 1$ corresponding to a real involution and $T^2 = -1$ to a pseudoreal/quaternionic involution. Using the explicit form of the gamma matrices we constructed we find the signs as discussed, implying that for $\nu = 0, 2 \mod 8$ and $\nu = 4, 6 \mod 8$ the Clifford algebra admits a real and quaternionic involution respectively.

- If N is odd, the SYK model does not admit a satisfactory (\mathbb{Z}_2 -graded) Hilbert space.
- If N is even, $N = 2 \mod 4$, then T is fermion-odd, and therefore the spectrum of the Hamiltonian is (at least) two-fold degenerate, with energy doublets having opposite fermion parity (the Hilbert space is supersymmetric).
- If N is even, $N = 4 \mod 8$, then T squares to -1, and therefore the spectrum of the Hamiltonian is (at least) two-fold degenerate, with energy doublets having the same fermion parity (they are Kramers doublets).
- If N is even, $N = 0 \mod 8$, the symmetry is non-anomalous and we cannot conclude anything about the spectrum of the Hamiltonian. Unless we tune the coefficients J to have some special symmetry, we do not expect any degeneracy in the Hilbert space.

It is a rather entertaining exercise to explicitly check these claims by numerically diagonalizing the SYK Hamiltonian for small values of N. We also note that similar ideas can be found in e.g. [106, 107].

1.3.2 Anomalous $\mathbb{Z}_4^{\mathsf{T}}$ in 2+1 dimensions

The anomalies of a fermionic system in 2 + 1d with an antiunitary time-reversal symmetry $\mathsf{T}^2 = (-1)^F$ are classified by $\Omega^4(\mathbb{Z}_2; 1, 1) = \Omega^4_{\mathrm{pin}^+} = \mathbb{Z}_{16}$. These anomalies arise from four layers (1.1.13)

$$\nu_{1} \in H^{1}(\mathbb{Z}_{2}^{\mathsf{T}}, \mathbb{Z}) \simeq \mathbb{Z}_{2}
\nu_{2} \in H^{2}(\mathbb{Z}_{2}^{\mathsf{T}}, \mathbb{Z}_{2}) \simeq \mathbb{Z}_{2}
\nu_{3} \in H^{3}(\mathbb{Z}_{2}^{\mathsf{T}}, \mathbb{Z}_{2})) \simeq \mathbb{Z}_{2}
\nu_{4} \in H^{4}(\mathbb{Z}_{2}^{\mathsf{T}}, \mathrm{U}(1)) \simeq \mathbb{Z}_{2},$$
(1.3.19)

which compile into \mathbb{Z}_{16} .

In this section we study this anomaly in a system of free Majorana fermions, and in section 1.4 we will do the same in time-reversal symmetric spin TQFTs.

Consider a system of ν Majorana fermions ψ . We shall work in the Majorana basis where the gamma matrices are real,

$$\gamma^0 = i\sigma^2, \qquad \gamma^1 = \sigma^1, \qquad \gamma^2 = \sigma^3. \tag{1.3.20}$$

In this basis the Majorana condition is simply $\psi^* = \psi$ so ψ is a real two-component Grassmann-odd spinor. We can without loss of generality take time-reversal to act as

$$T(\psi(t)) = \pm \gamma^0 \psi(-t)$$
. (1.3.21)

Given a pair of fermions transforming with opposite signs, we can write down a T-invariant mass term, which means that such a pair does not contribute to anomalies. Therefore, as far

as anomalies is concerned, we can take all fermions to transform with the same sign, say +1. It is known that a T-invariant interaction exists with 16 fermions that lifts all of them [56, 87–89, 108–110].

We now construct the torus Hilbert space of the system and study how the time-reversal anomaly manifests itself on it. A subtle but important difference in 2 + 1d as opposed to the examples in 0 + 1d and 1 + 1d is that $\Omega_4^{\text{spin}} = \mathbb{Z}$ contains a free part: in 2 + 1d there exists a purely gravitational SPT. This invertible theory is intertwined with time-reversal in an interesting way, which we review next. The generator of SPTs with no symmetry in 2 + 1d is given by the spin TQFT denoted by SO(1)₁, corresponding to the super Ising category [63, 111–113]. The partition function of this theory is $e^{-iCS_{\text{grav}}}$, where locally

$$CS_{grav} = \frac{1}{4\pi} \int_{M_3} tr\left(\omega \,\mathrm{d}\omega + \frac{2}{3}\omega^3\right) \,, \qquad (1.3.22)$$

where ω is the spin connection for the gravitational background of M_3 . An arbitrary SPT with no symmetry is given by a number $n \in \mathbb{Z}$ of copies of the generator, namely $\mathrm{SO}(n)_1 := \mathrm{SO}(1)_1^n$, whose partition function is $e^{-in\mathrm{CS}_{\mathrm{grav}}}$. As a spin TQFT, $\mathrm{SO}(n)_1$ can be obtained by condensing a certain fermion in the bosonic TQFT $\mathrm{Spin}(n)_1$, that is by gauging a certain \mathbb{Z}_2 one-form symmetry (see section 1.4). Note that the Chern-Simons form $\mathrm{CS}_{\mathrm{grav}}$ is a volume form, so it is odd under time-reversal.

If the manifold is non-trivial, the fermions automatically couple to the Chern-Simons term for the background gravitational field, because the Dirac operator contains a piece proportional to the spin connection. In 2 + 1d time-reversal acts both on the fermions and on the Chern-Simons interactions, and the combined system is only time-reversal invariant if the coefficient of the latter is adjusted appropriately. This behaviour should be thought of as a mixed time-reversal-gravitational anomaly, and it can be ascribed to a controlled non-invariance of the fermion path-integral measure $D\psi$. This non-invariance is a topological phase, the eta invariant η , and we can summarize the anomaly as the statement that each massless Majorana fermion ψ transforms as

$$\mathsf{T} \colon D\psi \mapsto e^{-i\pi\eta/2} D\psi \,. \tag{1.3.23}$$

In absence of other background fields, the eta invariant is precisely the gravitational Chern-Simons term,

$$\frac{1}{2}\pi\eta = \mathrm{CS}_{\mathrm{grav}} \mod 2\pi\mathbb{Z}.$$
(1.3.24)

In this sense, time-reversal does not map the QFT of a single massless fermion into itself, but rather into itself tensored with a copy of the SPT $SO(1)_1$; schematically

$$\mathsf{T}(\text{massless }\psi) = \text{massless }\psi \times \mathrm{SO}(1)_1. \tag{1.3.25}$$

In order to compensate for the anomalous phase $e^{-i\pi\eta/2}$, we formally need to attach to each massless Majorana fermion a copy of $\frac{1}{2}CS_{grav}$, i.e., to a copy of a "square root" of SO(1)₁. The combined object $e^{\frac{1}{2}CS_{\text{grav}}}D\psi$ is now time-reversal invariant. In the notation of (1.3.25), we formally need to move "half" of SO(1)₁ to the left, so as to have T mapping a QFT into itself instead of into a second QFT.

The discussion above is equivalent to the statement that a massless Majorana fermion carries chiral central charge c = 1/4 (this is also known as the *framing anomaly* [22]; recall that c measures the coupling of the theory to CS_{grav}). As c is odd under time-reversal, a system with $c \neq 0$ is not invariant by itself, but must be coupled to a suitable SPT, whose central charge is -c, in order to make the total central charge zero. The generator of SPTs $SO(1)_1$ has c = 1/2, so in order to compensate for the c = 1/4 of the fermion we formally need to couple it to a square root of $SO(1)_1$. More generally, given an arbitrary number ν of massless Majorana fermions, the system ψ^{ν} is not actually time-reversal invariant, but the combined system $\psi^{\nu} \times SO(\nu/2)_{-1}$ is. Naturally, if the number of fermions ν is odd, the coefficient of CS_{grav} is not properly normalized, and the system does not make sense as a purely 2 + 1d object: we either give up time-reversal invariance and drop the gravitational counterterm, or we keep the symmetry and regard the system as the boundary of a 3 + 1dtheory. For ν even, we can maintain time-reversal invariance and still have a conventional 2 + 1d theory, but only after coupling the fermions to $SO(\nu/2)_{-1}$. For now, we will consider the ν fermions alone, and later on we will study the effect of turning on SO($\nu/2$)₋₁ for ν even.

With this in mind, let us go back to studying the system of $2 + 1d \nu$ massless Majorana fermions on the torus T^2 . Anomalies, being renormalization-group invariant, always arise in the realization of the symmetry on the low energy states; therefore, in order to detect the anomalies, it suffices to look at the vacuum sector. For even spin structure on T^2 there are no zero modes and no anomalies; this agrees with the general discussion in section 1.1 where we argued that anomalies can only be detected on manifolds that do not bound.

For odd spin structure there are zero modes and potential anomalies. Roughly, the system with odd spin structure on T^2 behaves as 2ν copies of the 0 + 1d system of Majoranas we analyzed earlier, the factor of 2 being due to the fact that each ψ has two real components instead of one. In this sense, the analogous to the first layer in 0 + 1d is never activated in 2 + 1d, because the number of Majorana components is always even. In other words, the Hilbert space \mathcal{H}_{XY} of 2 + 1d Majorana fermions is always well-defined, regardless of the parity of the number of fermions. But the other two layers, those measured by the fermion parity of T and the sign in $T^2 = \pm 1$, are potentially activated. The first one is measured by the parity of ν , and the second one by the parity of $\nu/2$. We will exhibit the following anomalous behavior in the Hilbert space:

• $\nu = 1 \mod 2$: In \mathcal{H}_{R-R} time-reversal is fermion-odd, it anticommutes with $(-1)^F$

$$\{\mathsf{T}, (-1)^F\} = 0. \tag{1.3.26}$$

This anomaly is associated to the $p_x + ip_y$ layer ν_1 . For ν even $[\mathsf{T}, (-1)^F] = 0$.

• $\nu = 2 \mod 4$: In the even spin structure Hilbert spaces $\mathcal{H}_{\text{NS-NS}}, \mathcal{H}_{\text{NS-R}}, \mathcal{H}_{\text{R-NS}}$ timereversal satisfies the standard algebra $\mathsf{T} = (-1)^F$, but in the odd spin structure Hilbert space $\mathcal{H}_{\text{R-R}}$ this algebra is realized projectively, namely $\mathsf{T}^2 = -(-1)^F$. In other words, the time-reversal symmetry on \mathcal{H}_{XY} satisfies

$$\mathsf{T}^2 = (-1)^F \times (-1)^{\operatorname{Arf}(T^2)}. \tag{1.3.27}$$

This anomaly is associated to the Arf layer ν_2 .

The next two layers, ν_3 , ν_4 , which measure $\nu \mod 8$ and $\nu \mod 16$, respectively, are invisible on the torus Hilbert spaces.

The discussion regarding the first two layers is essentially identical to the 0 + 1d case, so we only highlight the differences. The fermions now depend on both time t and the spatial coordinate \boldsymbol{x} , which we take to coordinatize a torus T^2 . The Hilbert space associated to this spatial slice is built by acting with the spatial modes on the vacuum sector. If $\operatorname{Arf}(T^2) = 0$, then there are no zero-modes, and the vacuum Hilbert space is trivial: there is a unique vacuum state $|0\rangle$. Therefore, here time-reversal acts quite trivially: T is fermion-even and satisfies $T^2 = (-1)^F$ on the nose: neither layer is activated. In order to detect the anomaly we have to look at the non-bounding torus, i.e., where both boundary conditions are periodic such that $\operatorname{Arf}(T^2) = 1$. Here there is a single zero-mode for each Majorana fermion, which is spatially constant. In what follows we shall study this vacuum module generated by these zero-modes in \mathcal{H}_{R-R} .

First of all, since $T(\psi) = \gamma^0 \psi$ where $\gamma^0 = i\sigma^2$, time-reversal acts on the two components of the Majorana fermion as

$$T(\psi^1) = +\psi^2, \qquad T(\psi^2) = -\psi^1.$$
 (1.3.28)

In terms of the complex spinor $\Psi = \frac{1}{\sqrt{2}}(\psi^1 + i\psi^2)$ this becomes

$$\mathsf{T}(\Psi) = i\Psi^* \,. \tag{1.3.29}$$

The Hilbert space is built by declaring that $\Psi_i |0\rangle = 0$ for all $i = 1, 2, ..., \nu$, and by repeatedly acting with Ψ_i^* on $|0\rangle$. The action of time-reversal on the whole vacuum Hilbert space is uniquely fixed in terms of its action on $|0\rangle$, which again reads

$$\mathsf{T}|0\rangle = \Psi_1^* \Psi_2^* \cdots \Psi_\nu^* |0\rangle \tag{1.3.30}$$

up to an inconsequential phase. We thus see that, indeed, if ν is odd T anticommutes with $(-1)^F$. Now, if we act with T twice we get

$$\begin{aligned}
\mathsf{T}^{2}|0\rangle &= \mathsf{T}\Psi_{1}^{*}\Psi_{2}^{*}\cdots\Psi_{\nu}^{*}|0\rangle \\
&= (-i)^{\nu}\Psi_{1}\Psi_{2}\cdots\Psi_{\nu}\mathsf{T}|0\rangle \\
&= (-i)^{\nu}\Psi_{1}\Psi_{2}\cdots\Psi_{\nu}\Psi_{1}^{*}\Psi_{2}^{*}\cdots\Psi_{\nu}^{*}|0\rangle \\
&= i^{-\nu^{2}}|0\rangle.
\end{aligned}$$
(1.3.31)

When ν is even we get $\mathsf{T}^2|0\rangle = +|0\rangle$. More generally, as $\mathsf{T}^2 = (-1)^F$ when acting on the creation operators, the relation $\mathsf{T}^2|0\rangle = +|0\rangle$ lifts to $\mathsf{T}^2 = (-1)^F$ on the whole Hilbert space.²⁶

We now return to the effect of the gravitational SPT $SO(\nu/2)_{-1}$ for ν even that is needed in order to have a time-reversal symmetric theory. This SPT has a unique state on any spin structure Hilbert space \mathcal{H}_{XY} . The fermion parity of this state is known to be $(-1)^F = (-1)^{Arf(T^2)\nu/2}$ (see [5, 36] and section 5.2.2). Therefore, the T-invariant combined system $\psi^{\nu} \times SO(\nu/2)_{-1}$ has a time-reversal algebra

$$\mathsf{T}^{2} = (-1)^{F} \times (-1)^{\operatorname{Arf}(T^{2})\nu/2}.$$
(1.3.32)

This means that the operator algebra $\mathsf{T}^2 = (-1)^F$ is undeformed for $\nu = 0 \mod 4$, while it gets deformed by the Arf theory for $\nu = 2 \mod 4$ in $\mathcal{H}_{\text{R-R}}$, as claimed.

1.3.3 Anomalous \mathbb{Z}_2 in 1+1 dimensions

The anomalies of a fermionic system in 1 + 1d with a unitary \mathbb{Z}_2 symmetry such that $G_f = \mathbb{Z}_2 \times \mathbb{Z}_2^F$ are classified by $\Omega^2_{\text{spin}}(\mathbb{Z}_2; 0, 0) = \mathbb{Z}_8$. These anomalies arise from three layers

$$\nu_1 \in H^1(\mathbb{Z}_2, \mathbb{Z}_2) \simeq \mathbb{Z}_2$$

$$\nu_2 \in H^2(\mathbb{Z}_2, \mathbb{Z}_2) \simeq \mathbb{Z}_2$$

$$\nu_3 \in H^3(\mathbb{Z}_2, \mathrm{U}(1)) \simeq \mathbb{Z}_2,$$

(1.3.33)

which generate the \mathbb{Z}_8 anomaly.

Consider ν Majorana fermions in $1 + 1d^{27}$

$$\mathcal{L} = \sum_{a=1}^{\nu} i\psi_L^a \partial_+ \psi_L^a + i\psi_R^a \partial_- \psi_R^a \,, \qquad (1.3.34)$$

where $\partial_{\pm} = \partial_t \pm \partial_x$. This system has a chiral \mathbb{Z}_2 unitary symmetry generated by $g = (-1)^{F_L}$ which combines with the nonchiral \mathbb{Z}_2^F symmetry generated by $(-1)^F$ to yield the symmetry group $G_f = \mathbb{Z}_2 \times \mathbb{Z}_2^F$. These symmetries act on the fermions as

$$\{(-1)^{F_L}, \psi_L^a\} = [(-1)^{F_L}, \psi_R^a] = 0$$

$$\{(-1)^F, \psi_L^a\} = \{(-1)^F, \psi_R^a\} = 0.$$
 (1.3.35)

²⁶Here we are fixing the fermion parity of the odd-spin-structure ground state to be +1, the same as in the even-spin-structures case. We could declare instead that the ground state has fermion parity -1. This would eliminate the anomalous sign in the time-reversal algebra below. The anomaly would manifest itself, instead, in the presence of an Arf-dependent anomalous sign in the fermion parity of the ground state, $(-1)^F|0\rangle = (-1)^{\operatorname{Arf}(T^2)\nu/2}|0\rangle$. This is simply a redefinition $(-1)^F \mapsto (-1)^{\operatorname{Arf}(T^2)\nu/2}(-1)^F$.

²⁷The anomalies in 1 + 1d are actually $\mathbb{Z}_8 \times \mathbb{Z}$, the second factor being the gravitational anomaly. We take $\nu_L = \nu_R = \nu$ to cancel this gravitational anomaly and focus directly on the \mathbb{Z}_8 factor.

It is known that a \mathbb{Z}_2 -symmetric interaction that gaps out the fermions can be written for $\nu = 8$ [92, 114–116]. This realizes in the fermion system the \mathbb{Z}_8 anomaly expected from the cobordism classification.

We now analyze the anomaly layers that can be detected in the Hilbert space. We discuss in turn the Hilbert space \mathcal{H}_X and the \mathbb{Z}_2 -twisted Hilbert space \mathcal{H}_X^g , where $X \in \{NS, R\}$ denotes the spin structure on the spatial circle. The twisted Hilbert space \mathcal{H}_X^g is defined by quantizing in the presence of a nontrivial \mathbb{Z}_2 (flat) connection around the circle for the \mathbb{Z}_2 symmetry.

In order to detect the anomalies we proceed to study the implementation of symmetries on the zero-mode operators in \mathcal{H}_X and \mathcal{H}_X^g in turn.

Anomalies in \mathcal{H}_X

Since in the NS sector there are no fermion zero-modes, there is a unique, trivial vacuum and symmetries are realized on \mathcal{H}_{NS} in a non-anomalous fashion. In the R sector there are fermion zero-modes which upon quantization furnish a Clifford algebra of rank 2ν

$$\{\psi_L^a, \psi_L^b\} = \{\psi_R^a, \psi_R^b\} = 2\delta^{ab}, \qquad \{\psi_L^a, \psi_R^b\} = 0 \qquad a, b = 1, 2, \dots, \nu.$$
(1.3.36)

This Clifford algebra has a unique irreducible representation of dimension 2^{ν} , thus all representations are unitarily equivalent, and we can study the implementation of symmetries in any choice of basis. We can construct $\mathcal{H}_{\rm R}$ by defining the creation and annihilation operators $\psi^a_+ = \frac{1}{2}(\psi^a_R + i\psi^a_L)$ and $\psi^a_- = \frac{1}{2}(\psi^a_R - i\psi^a_L)$, such that $\psi^a_-|0\rangle = 0$. It follows from (1.3.35) that

$$(-1)^{F_L}\psi^a_+ = \psi^a_-(-1)^{F_L}. \tag{1.3.37}$$

The \mathbb{Z}_2 symmetry generator thus maps the empty vacuum to the completely filled state

$$(-1)^{F_L}|0\rangle = \alpha \psi_+^1 \psi_+^2 \cdots \psi_+^{\nu}|0\rangle.$$
 (1.3.38)

for some phase α . This implies that the $\mathbb{Z}_2 \times \mathbb{Z}_2^F$ symmetry generators on \mathcal{H}_R obey

$$\{(-1)^{F_L}, (-1)^F\} = 0 \quad \text{for } \nu \text{ odd}, \qquad (1.3.39)$$

and

$$[(-1)^{F_L}, (-1)^F] = 0 \qquad \text{for } \nu \text{ even}.$$
 (1.3.40)

Therefore, the anomaly corresponding to the ν odd layer is detected by virtue of the operators $(-1)^{F_L}$ and $(-1)^F$ anticommuting in \mathcal{H}_R . This has also been noticed in [11, 117].

The anomaly associated to the ν odd layer can also be detected in the torus partition function with periodic boundary conditions around both the spatial circle and temporal circle, that is with (R, R) boundary conditions along the two cycles of the torus. The zero-modes in \mathcal{H}_R imply that the partition function vanishes, but the partition function with fermion zero-modes saturated is nonvanishing:

$$\langle \psi_L^1 \psi_L^2 \cdots \psi_L^\nu \psi_R^1 \psi_R^2 \cdots \psi_R^\nu \rangle \neq 0.$$
(1.3.41)

This implies that the $(-1)^{F_L}$ Ward identities are violated for ν odd, that is, there is an anomaly for the chiral \mathbb{Z}_2 symmetry.

Anomalies in \mathcal{H}^g_X

This Hilbert space is constructed by imposing boundary conditions twisted by $(-1)^{F_L}$ when fermions are transported around the spatial circle. This yields different boundary conditions for the left-moving and right-moving fermions, which we will denote by $[X_L, X_R]$, where $X_{L/R} \in \{NS, R\}$.²⁸

Let us consider \mathcal{H}_X^g in turn:

 \mathcal{H}_{NS}^{g} . This corresponds to [R, NS] boundary conditions on the fermions. There are ν zero modes from the left movers and none from the right movers. The zero mode algebra is thus a Clifford algebra of rank ν

$$\{\psi_L^a, \psi_L^b\} = 2\delta^{ab} \qquad a, b = 1, 2, \dots, \nu.$$
(1.3.42)

• ν odd. There is a rather severe anomaly for ν odd as the operator $(-1)^F$ generating \mathbb{Z}_2^F obeying $\{(-1)^F, \psi_L^a\} = 0$ does not exist. The theory does not admit a proper graded Hilbert space of states. Equivalently stated, the Clifford algebra of odd rank has two irreducible representations, and $(-1)^F$ exchanges them, instead of acting within an irreducible representation. This anomaly is associated with the $H^1(\mathbb{Z}_2, \mathbb{Z}_2) = \mathbb{Z}_2$ layer, the Arf layer.

The \mathbb{Z}_2 anomaly associated to the Arf layer and the corresponding lack of a Hilbert space can also be detected in the torus partition function with antiperiodic boundary conditions around both the spatial circle and the temporal circle, that is, with (NS, NS) boundary conditions along the two cycles of the torus. This partition function is given by

$$Z_{(\text{NS,NS})}^{(g,1)} = (\sqrt{2}\chi_{\sigma})^{\nu} (\overline{\chi}_1 + \overline{\chi}_{\epsilon})^{\nu}, \qquad (1.3.43)$$

where χ_1, χ_{σ} and χ_{ϵ} are the Virasoro characters with weight 0, 1/16 and 1/2. For ν odd, the partition function indeed does not have an integral expansion, and thus there is no suitable Hilbert space. The dimensions of the modules are integers times a factor of $\sqrt{2}$, consistent with the fact that the anomaly is due to a dangling $C\ell(1)$, whose dimension is formally $\sqrt{2}$.

²⁸This is to be contrasted with the boundary condition in the untwisted Hilbert space, where $X_L = X_R$.

• ν even. For ν even the theory has a well-defined Hilbert space \mathcal{H}_{NS}^g and well defined $(-1)^F$ and $(-1)^{F_L}$ operators obeying $[(-1)^{F_L}, (-1)^F] = 0$ in the Hilbert space. The \mathbb{Z}_2 symmetry generator maps the empty vacuum in \mathcal{H}_X^g to itself up to phase $(-1)^{F_L}|0\rangle = \alpha|0\rangle$.

 $\mathcal{H}^{g}_{\mathbf{R}}$. This corresponds to [NS, R] boundary conditions on the fermions. There are ν zero modes from the right-movers and none from the left-movers. The zero mode algebra is thus a Clifford algebra of rank ν

$$\{\psi_R^a, \psi_R^b\} = 2\delta^{ab} \qquad a, b = 1, 2, \dots, \nu.$$
(1.3.44)

ν odd. Verbatim our discussion above: this system has a Z₂ anomaly associated to the Arf layer, diagnosed by the lack of a proper graded Hilbert space of states. This can also be seen from the lack of integral expansion of the torus partition function with (R, NS) boundary conditions along the two cycles of the torus

$$Z_{(R,NS)}^{(g,1)} = (\chi_1 + \chi_{\epsilon})^{\nu} (\sqrt{2}\overline{\chi}_{\sigma})^{\nu}$$
(1.3.45)

which again does not have a properly quantized expansion.

• ν even. For ν even the theory has a well-defined Hilbert space \mathcal{H}^g_R and well defined $(-1)^F$ and $(-1)^{F_L}$ operators obeying $[(-1)^{F_L}, (-1)^F] = 0$ in the Hilbert space.

Twisting by the symmetry eliminates the zero modes of one chirality. So effectively the twisting halves the number of zero-modes, and the system moves one step up the ladder of layers. In the untwisted Hilbert spaces \mathcal{H}_X , $\nu = 1 \mod 2$ means that g anticommutes with $(-1)^F$. For twisted Hilbert spaces \mathcal{H}_X^g , $\nu = 1 \mod 2$ means lack of graded Hilbert spaces.

Projective rotations. In the discussion so far, we argued that the $\nu = 1 \mod 2$ layer of the anomaly can be detected by the symmetry algebra of g and $(-1)^F$. It is noteworthy that, if we look at a more general group of symmetries, the *full* anomaly $\nu \mod 8$ can be detected.

The detection of the $\nu \in \mathbb{Z}_8$ anomaly is well-understood by now. This anomaly measures the anomalous spin of operators on the twisted Hilbert spaces \mathcal{H}_X^g . From our point of view, this statement is understood as follows. When we compactify the theory on a circle, the model acquires an extra flavor symmetry: translations along the compact direction become internal symmetries of the effective 0 + 1 dimensional quantum mechanical model. Therefore, the Hilbert spaces on the circle realize a representation of a larger symmetry group: the initial $G_f = \mathbb{Z}_2 \times \mathbb{Z}_2^F$ symmetry is enhanced to $G_f \times U(1)$, with U(1) the group of rotations around the spatial circle. While the bottom layer $\nu \mod 2$ only measures a projective action of G_f , the full anomaly $\nu \mod 8$ measures a projective realization of the extended group $G_f \times U(1)$. In a non-anomalous theory, the Hilbert space realizes a double-cover of the symmetry group U(1), because of the presence of fermions — the system is invariant under 4π rotations. In an anomalous theory, the Hilbert space realizes a higher cover of U(1), because of the presence of operators with fractional spin. We claim that in a system with anomaly $\nu \in \mathbb{Z}_8$, the twisted Hilbert spaces \mathcal{H}_X^g realize a $2 \times (8/\gcd(\nu, 8))$ -cover of U(1).

This claim can be established by looking again at a free fermion system. In this system, the anomalous spin comes from the "zero-point momentum" of the fermions, namely 1/16 units for each Ramond zero-mode. In the untwisted Hilbert space there is no fractional momentum: if both chiralities are NS, there are no zero-modes, and if they are both R, the zero-point momenta cancel out. In the twisted Hilbert space, one of the two chiralities is NS and the other is R, so there is no cancellation, and there are $\nu/16$ units of fractional momentum. Therefore, the eigenvalues of the momentum operator are half-integers (from the oscillator modes), plus $\nu/16$ (from the zero-modes). In other words, $2P \pm \nu/8 \in \mathbb{Z}$. The momentum operator P is precisely the generator of U(1) rotations; therefore, instead of two circles, we need to perform $2 \times (8/ \gcd(\nu, 8))$ full turns instead in order to compensate for this fractional momentum. This proves the claim.

This also admits a path-integral interpretation. In a system with $\nu \in \mathbb{Z}_8$ units of anomaly, the partition function twisted by the g symmetry picks up a phase $e^{\pm 2\pi i\nu/8}$ under a 4π rotation, which signals the presence of operators with fractional momentum. We show this by looking at the free fermion system, whose partition function over the twisted Hilbert space \mathcal{H}_X^g is (1.3.43), (1.3.45)

$$\begin{aligned}
\mathcal{H}_{\rm NS}^g : \quad Z_{\rm (NS,NS)}^{(g,1)} &= (\sqrt{2}\chi_{\sigma})^{\nu} (\overline{\chi}_1 + \overline{\chi}_{\epsilon})^{\nu} \\
\mathcal{H}_{\rm R}^g : \quad Z_{\rm (R,NS)}^{(g,1)} &= (\chi_1 + \chi_{\epsilon})^{\nu} (\sqrt{2}\overline{\chi}_{\sigma})^{\nu}
\end{aligned} \tag{1.3.46}$$

A 4π rotation corresponds to a T^2 modular transformation, which induces the phase $e^{\pm 4\pi i\nu h_{\sigma}} \equiv e^{\pm 2\pi i\nu/8}$, with $h_{\sigma} = 1/16$ the conformal weight of χ_{σ} .

To summarize, the effect of the anomaly $\nu \in \mathbb{Z}_8$ is that the twisted Hilbert spaces realize a projective representation of the rotation symmetry group U(1), and from this projective action one can read off the anomaly. For a given value of ν , the Hilbert space realizes a $2 \times (8/\gcd(\nu, 8))$ cover of U(1), as opposed to a double-cover, and the generator of rotations has fractional momentum $\nu/16$. In a conformal theory, this can also be detected by performing a T^2 modular transformation on the twisted partition function. A very similar philosophy was used in [63] to derive the time-reversal anomaly in a 2 + 1d dimensional system.

1.4 Anomalies in spin TQFT Hilbert space

In this section we demonstrate the existence of anomalies in fermionic TQFTs by looking directly at their Hilbert space. We follow the construction of the Hilbert space of a fermionic TQFT in [5]; see [36, 37, 118–121] for related work. Here we summarize the main ingredients, leaving most details to chapter 5.

Given an arbitrary spin TQFT, one may sum over all spin structures in order to yield a bosonic TQFT. We refer to this theory as the *bosonic parent/shadow* of the original spin TQFT. This process of summing over spin structures corresponds to gauging the zero-form symmetry generated by fermion parity $\mathbb{Z}_2^F = \langle (-1)^F \rangle$. Given such bosonic theory, one may undo the gauging, i.e., we can recover the original spin TQFT by gauging a dual \mathbb{Z}_2 symmetry, this time a one-form symmetry [32]. This symmetry is generated by a certain fermionic line operator ψ , i.e., $\mathbb{Z}_2^{(1)} = \langle \psi \rangle$. Gauging this one-form symmetry is also known as *condensing* the anyon ψ .

With this in mind, the Hilbert space of the spin TQFT is easily obtained in terms of the Hilbert space of the bosonic parent, by means of the standard procedure of gauging a symmetry. The Hilbert space of the bosonic parent, being a standard TQFT, is well-understood. Specifically, the torus Hilbert space has a basis of states labelled by the anyons [22]:

$$\mathcal{H}(T^2) = \operatorname{Span}_{\beta \in \mathcal{A}} \left[\bigotimes^{\times} \beta \right]$$
(1.4.1)

Here \mathcal{A} denotes the set of all anyons in the bosonic parent – a finite set – and the loop denotes a Wilson line labeled by β wrapped along the b-cycle of the torus (see figure 1.1).



Figure 1.1: Schematic notation for an arbitrary configuration of anyons on the torus, in the presence of a puncture **g**. The green line represents the vertical (time) direction, orthogonal to the torus. We insert an anyon **g** along this direction, which from the point of view of the torus becomes a puncture (a marked point). The red line represents the **a**-cycle, and the blue one the **b**-cycle. We insert Wilson lines with anyons α, β along these cycles, respectively. The cross × represents the hole. The states in the Hilbert space \mathcal{H} are created by wrapping anyons around the **b**-cycle. The states in the *twisted* Hilbert space \mathcal{H}^{g} are created by wrapping anyons around the **b**-cycle, in presence of a vertical anyon **g**.

The set \mathcal{A} comes equipped with extra structure; for example, we have the modular matrix $S: \mathcal{A} \times \mathcal{A} \to \mathbb{C}$ that implements the large diffeomorphism $(a, b) \mapsto (b, -a)$. Similarly, we also have the modular matrix $T: \mathcal{A} \times \mathcal{A} \to \mathbb{C}$ that implements the large diffeomorphism $(a, b) \mapsto (a, a+b)$; these two transformations S, T generate the set of all large diffeomorphisms,

the modular group $SL_2(\mathbb{Z})$. By our choice of basis (1.4.1) where the states are wrapped around the b-cycle, the *T*-matrix is diagonal, with $T = \text{diag}(e^{2\pi(h-c/24)})$, where *c* is the central charge of the system and $h: \mathcal{A} \to \mathbb{Q}/\mathbb{Z}$ denotes the topological spin of the lines.

The Hilbert space of the fermionic theory, let us call it \mathcal{H} , depends on the spin structure of the torus. We denote these tori as T_{s^a,s^b}^2 , where $s^a, s^b = \pm 1$ refers to the boundary condition around the cycle **a**- and **b**-cycles, respectively. We also denote the s = -1 boundary condition by NS and the s = +1 boundary condition by R. We claim that the corresponding Hilbert spaces are spanned by the following bases:

$$\hat{\mathcal{H}}(T_{\text{NS-NS}}^2) = \operatorname{Span}_{a} \left[\begin{array}{c} \times \\ a \end{array} + \begin{array}{c} \times \\ a \times \psi \end{array} \right]$$

$$\hat{\mathcal{H}}(T_{\text{NS-R}}^2) = \operatorname{Span}_{a} \left[\begin{array}{c} \times \\ a \end{array} - \begin{array}{c} \times \\ a \times \psi \end{array} \right]$$

$$\hat{\mathcal{H}}(T_{\text{R-NS}}^2) = \operatorname{Span}_{x} \left[\begin{array}{c} \times \\ x \end{array} + \begin{array}{c} \times \\ x \times \psi \end{array} \right] \oplus \operatorname{Span}_{m} \left[\begin{array}{c} \times \\ m \end{array} \right]$$

$$\hat{\mathcal{H}}(T_{\text{R-R}}^2) = \operatorname{Span}_{x} \left[\begin{array}{c} \times \\ x \end{array} - \begin{array}{c} \times \\ x \times \psi \end{array} \right] \oplus \operatorname{Span}_{m} \left[\begin{array}{c} \times \\ m \end{array} \right]$$

$$(1.4.2)$$

Here, a takes values in the subset of lines of \mathcal{A} with the property that $h_{\alpha} \equiv h_{\alpha \times \psi} \mod 1$. On the other hand, both x and m denote the lines in \mathcal{A} such that $h_{\alpha} \equiv h_{\alpha \times \psi} + 1/2 \mod 1$; the difference between x lines and m lines is that the former satisfy $x \times \psi \neq x$ while the latter satisfy $m \times \psi = m$. Finally, the state with an open line denotes the anyon β around the spatial torus and the anyon ψ running along the time direction, cf. figure 1.1. (From the point of view of the spatial torus, the line operator ψ looks like a puncture, i.e., a local operator; it is essentially a constant spinor, a zero-mode, which explains why it only exists in the odd spin structure).

The rationale behind the structure above is the following. Given the bosonic theory, gauging the one-form symmetry generated by ψ means inserting this operator in all possible ways; summing over insertions along the spatial cycles projects the spectrum into the invariant states, and summing over insertions along the time circle introduces twisted sectors. One can check that the states in (1.4.2) are indeed invariant under insertion of ψ along any of the spatial cycles. The twisted sectors are precisely the states with a puncture, which do not live in \mathcal{H} but in the defect Hilbert space \mathcal{H}^{ψ} instead. The details of this construction are elaborated upon in section 5.1.

The explicit geometric structure of states in (1.4.2) also allows us find how the operators of the spin TQFT act on the different states. For example, the Wilson lines act by inserting an anyon around the cycle they are supported on, and therefore they act as

$$W^{(\mathbf{a})}(\alpha) \colon \bigotimes_{\beta} \mapsto \bigotimes_{\alpha} \bigotimes_{\beta} \equiv \frac{S_{\alpha,\beta}}{S_{\mathbf{1},\beta}} \bigotimes_{\beta}$$

$$W^{(\mathbf{b})}(\alpha) \colon \bigotimes_{\beta} \mapsto \bigotimes_{\beta} \equiv \bigotimes_{\alpha \times \beta}$$

$$(1.4.3)$$

on states without puncture, and as [122]

$$W^{(\mathbf{b})}(\alpha): \quad \psi \xrightarrow{\times}_{\beta} \mapsto \qquad \psi \xrightarrow{\times}_{\beta} \equiv F_{\beta,\alpha\times\beta} \begin{bmatrix} \psi & \alpha \times \beta \\ \beta & \alpha \end{bmatrix} \quad \psi \xrightarrow{\times}_{\alpha\times\beta}, \qquad (1.4.4)$$

on states with a puncture. Here $S: \mathcal{A} \times \mathcal{A}$ denotes the modular matrix of the bosonic parent, and F its F-symbols. From these expressions it is trivial to write down the Wilson operators as matrices acting on $\hat{\mathcal{H}}$ (with respect to the basis (1.4.2)).

Similarly, one can also write down how modular transformations map the different Hilbert spaces $\hat{\mathcal{H}}(T_{XY}^2)$ into each other. For example, an S-transformation acts as

$$S: \bigotimes_{\beta} \mapsto \sum_{\alpha \in \mathcal{A}} S_{\alpha,\beta} \bigotimes_{\alpha}$$
(1.4.5)

on states without puncture, and as

$$S: \quad \psi \xrightarrow{\times}_{\beta} \mapsto \sum_{\alpha \in \mathcal{A}} S_{\alpha,\beta}(\psi) \quad \psi \xrightarrow{\times}_{\alpha \times \beta}, \qquad (1.4.6)$$

on states with a puncture. Here $S(\psi)$ is the S-matrix of the bosonic parent in the oncepunctured torus (cf. (5.1.18)). Given these two expressions one can easily check that Stransformations reshuffle the different spin structures as expected, namely $(X, Y) \mapsto (Y, X)$. Identical considerations hold for T-transformations (these are actually simpler because they do not see the puncture, so the formula $T = \text{diag}(e^{2\pi(h-c/24)})$ is still valid for states in \mathcal{H}^{ψ}).

The final important remark concerning the fermionic Hilbert space $\hat{\mathcal{H}}$ is that it is a super-vector space, i.e., its states are either bosons or fermions. Given that $(-1)^F$ is, by definition, dual to the gauged one-form symmetry $\mathbb{Z}_2^{(1)}$, it is clear that the states charged under the former correspond to the states coming from the twisted sector, i.e., the fermions in $\hat{\mathcal{H}}$ are precisely those that include a ψ -puncture. In this sense, $(-1)^F$ is trivial in the even spin structure Hilbert spaces, and in the odd spin structure Hilbert space it equals +1 on x-lines and -1 on m-lines. In a fermionic theory the modular group is no longer $SL_2(\mathbb{Z})$, but rather a \mathbb{Z}_2^F extension, known as the metaplectic group $Mp_1(\mathbb{Z})$, defined by the relations

$$S^{2} = (ST)^{3}, \qquad S^{4} = (-1)^{F}.$$
 (1.4.7)

The torus Hilbert spaces of spin TQFTs realize a unitary representation of this group.

We next illustrate all these considerations by explicitly working out several specific examples of spin TQFTs. We show that time-reversal invariant theories with $\mathsf{T}^2 = (-1)^F$ with \mathbb{Z}_{16} anomalies $\nu = 2 \mod 4$ have time-reversal in the Hilbert space realized as

$$\mathsf{T}^{2} = (-1)^{F} \times (-1)^{\operatorname{Arf}(T^{2})}, \qquad (1.4.8)$$

thus exhibiting the anomalies associated with the Arf layer. We then show that spin TQFTs with ν odd have a time-reversal symmetry that anticommutes with $(-1)^F$ on \mathcal{H}_{R-R}

$$\{\mathsf{T}, (-1)^F\} = 0, \qquad (1.4.9)$$

thus exhibiting the anomalies associated with the ψ layer

1.4.1 $\nu = 2 \mod 4$: Arf layer

In this section we consider time-reversal invariant theories with $\nu = 2 \mod 4$ and show that they realize the expected behavior associated with the Arf layer. We work out in detail here the example of the semion-fermion theory which has $\nu = 2$, although the same behaviour is observed in any theory with $\nu = 2 \mod 4$. Other common examples of time-reversal invariant spin TQFTs are $\text{Sp}(n)_n$ and $\text{SO}(n)_n$, which have $\nu = 2n$ and $\nu = n$, respectively. One can check that e.g. $\text{Sp}(3)_3$ and $\text{SO}(6)_6$, which have $\nu = 6$, exhibit the same behaviour. We do not reproduce the explicit computation here because the matrices are very large and the details do not contain any new ingredients.

The semion-fermion theory is a fermionic TQFT with four anyons: the vacuum 1, a semion s, a transparent fermion ψ , and the composite $s \times \psi$. A Chern-Simons realization of this theory is $U(1)_2 \times \{1, \psi\}$, where $U(1)_2 = \{1, s\}$ is the TQFT of a single semion, and ψ denotes a transparent fermion. The invertible factor can be written as $\{1, \psi\} = SO(n)_1$ for any $n \in \mathbb{Z}$; the most convenient choice is n = -2, so that the theory has vanishing central charge (as required by time-reversal). In other words, we shall consider $U(1)_2 \times U(1)_{-1}$, whose Lagrangian reads

$$\mathcal{L} = \frac{1}{4\pi} (2a \,\mathrm{d}a - b \,\mathrm{d}b) \,, \tag{1.4.10}$$

where a, b are U(1) connections. The time-reversal symmetry of this Lagrangian acts as follows [4, 111]:

$$T(a) = a - b,$$
 $T(b) = 2a - b.$ (1.4.11)

One way of constructing the Hilbert space of this theory is to take $U(1)_2 \times U(1)_{-4}$, which is bosonic, and condense the line $\psi = (0, 2)$; this is a fermionic quotient, and so the result is a spin TQFT, where ψ becomes transparent. We begin by constructing the Hilbert space of the bosonic parent, $U(1)_2 \times U(1)_{-4}$. This is a $2 \times 4 = 8$ -dimensional space, whose basis can be taken to be

$$|\alpha,\beta\rangle = W^{(\mathbf{b})}(\alpha,\beta)|0\rangle, \qquad (\alpha,\beta) \in \mathbb{Z}_2 \times \mathbb{Z}_4, \qquad (1.4.12)$$

where $|0\rangle$ denotes the vacuum state – the empty torus – and W denotes a Wilson line:

$$W^{(c)}(\alpha,\beta) := \exp i \oint_{c} (\alpha a + \beta b), \qquad (1.4.13)$$

for any $c \in H_1(T^2, \mathbb{Z}) = \mathbb{Z}[a] \oplus \mathbb{Z}[b]$.

The Wilson lines along the b-cycle act on a generic state as follows:

$$W^{(\mathbf{b})}(\alpha,\beta)|\alpha',\beta'\rangle = |\alpha + \alpha' \mod 2, \ \beta + \beta' \mod 4\rangle.$$
(1.4.14)

The action of the Wilson lines associated to other cycles can be obtained using the modular operations. For example, on the **a**-cycle, one has

$$W^{(a)}(\alpha)|\alpha'\rangle = \frac{S_{\alpha\alpha'}}{S_{1\alpha'}}|\alpha'\rangle, \qquad (1.4.15)$$

where S denotes the S-matrix of the system. In the theory $U(1)_2 \times U(1)_{-4}$, this matrix reads $S_{(\alpha,\beta),(\alpha',\beta')} = e^{\pi i (\beta \beta'/2 - \alpha \alpha')}/2\sqrt{2}$.

We now condense the fermion $\psi = (0, 2)$. The braiding phase of a generic line (α, β) with respect to the fermion is $B((\alpha, \beta), \psi) = e^{-i\pi\beta}$, and so the NS- and R-lines are as follows:

NS:
$$\mathcal{A}_{NS} = \{(\alpha, \beta) : \beta = 0, 2\}$$

R: $\mathcal{A}_{R} = \{(\alpha, \beta) : \beta = 1, 3\}$

$$(1.4.16)$$

Furthermore, there are no fixed-points under fusion with ψ . Indeed, the lines are all in two-dimensional orbits, paired as follows:

NS:
$$(\alpha, 0) \stackrel{\times \psi}{\longleftrightarrow} (\alpha, 2)$$

R: $(\alpha, 1) \stackrel{\times \psi}{\longleftrightarrow} (\alpha, 3)$. (1.4.17)

The lack of fixed-points indicates that there are no Majorana lines in the theory, i.e., all states are bosonic. In the terminology of section 5.1.2, all lines (α, β) are *a*-type or *x*-type, depending on whether β is even or odd; and there are no *m*-lines.

Hilbert space. With these preliminaries in mind, we now construct the torus Hilbert space(s) of the fermionic theory. As in section 5.1.2, the states of the condensed phase are expressed as linear combinations of those of the bosonic parent, and the specific combinations are determined by the spin structure (cf. (5.1.33)):
• If we take **NS-NS** boundary conditions, the two states are

$$|0; \text{NS-NS}\rangle = \frac{1}{\sqrt{2}} (|0, 0\rangle + |0, 2\rangle)$$

|1; \text{NS-NS}\rangle = $\frac{1}{\sqrt{2}} (|1, 0\rangle + |1, 2\rangle).$ (1.4.18)

• If we take **NS-R** boundary conditions, the two states are

$$|0; \text{NS-R}\rangle = \frac{1}{\sqrt{2}} (|0, 0\rangle - |0, 2\rangle)$$

|1; \text{NS-R}\rangle = $\frac{1}{\sqrt{2}} (|1, 0\rangle - |1, 2\rangle).$ (1.4.19)

• If we take **R-NS** boundary conditions, the two states are

$$|0; \text{R-NS}\rangle = \frac{1}{\sqrt{2}} (|0, 1\rangle + |0, 3\rangle)$$

|1; \text{R-NS}\rangle = $\frac{1}{\sqrt{2}} (|1, 1\rangle + |1, 3\rangle).$ (1.4.20)

• If we take **R-R** boundary conditions, the two states are

$$|0; \text{R-R}\rangle = \frac{1}{\sqrt{2}} (|0, 1\rangle - |0, 3\rangle)$$

$$|1; \text{R-R}\rangle = \frac{1}{\sqrt{2}} (|1, 1\rangle - |1, 3\rangle).$$

(1.4.21)

Modularity. As a consistency check, we can study how modular transformations move us around these four Hilbert spaces. Take for example the S-transformation. In the bosonic parent, this operation acts as

$$S|\alpha,\beta\rangle = \sum_{\substack{\alpha'\in\mathbb{Z}_2\\\beta'\in\mathbb{Z}_4}} S_{(\alpha,\beta),(\alpha',\beta')}|\alpha',\beta'\rangle$$
(1.4.22)

with $S_{(\alpha,\beta),(\alpha',\beta')} = e^{i\pi(\beta\beta'/2-\alpha\alpha')}/2\sqrt{2}$. Using this we obtain the action of S on the fermionic theory. For example, it acts on the NS-NS states as follows:

$$S|0; \text{NS-NS}\rangle = \frac{1}{\sqrt{2}} \left[S|0,0\rangle + S|0,2\rangle \right]$$

= $\frac{1}{4} \left[|0,0\rangle + |0,1\rangle + |0,2\rangle + |0,3\rangle + |1,0\rangle + |1,1\rangle + |1,2\rangle + |1,3\rangle + |0,0\rangle - |0,1\rangle + |0,2\rangle - |0,3\rangle + |1,0\rangle - |1,1\rangle + |1,2\rangle - |1,3\rangle \right]$
= $\frac{1}{2} (|0,0\rangle + |0,2\rangle + |1,0\rangle + |1,2\rangle).$ (1.4.23)

We recognize this state as $\frac{1}{\sqrt{2}}(|0; \text{NS-NS}\rangle + |1; \text{NS-NS}\rangle)$. Through an identical computation one can show that S maps $|1; \text{NS-NS}\rangle$ into $\frac{1}{\sqrt{2}}(|0; \text{NS-NS}\rangle - |1; \text{NS-NS}\rangle)$. In both cases we see that an S-transformation maps states in $\hat{\mathcal{H}}_{\text{NS-NS}}$ into $\hat{\mathcal{H}}_{\text{NS-NS}}$, precisely as expected (cf. (5.1.26)); and, moreover, the specific matrix that realizes this transformation is

$$\hat{S}_{\text{NS-NS}\to\text{NS-NS}} = \frac{1}{\sqrt{2}} \begin{pmatrix} 1 & 1\\ 1 & -1 \end{pmatrix}$$
 (1.4.24)

By performing S-transformations on the other three Hilbert spaces we see that they are permuted exactly as they should, namely $S: \hat{\mathcal{H}}_{s^a,s^b} \to \hat{\mathcal{H}}_{s^b,s^a}$; and that they act as the following matrices:

$$\hat{S}_{\text{NS-R}\to\text{R-NS}} = \hat{S}_{\text{R-NS}\to\text{NS-R}} = -i\hat{S}_{\text{R-R}\to\text{R-R}} = \frac{1}{\sqrt{2}} \begin{pmatrix} 1 & 1\\ 1 & -1 \end{pmatrix}$$
. (1.4.25)

A T-transformation, on the other hand, acts in the bosonic parent as

$$T|\alpha,\beta\rangle = e^{i\pi(\alpha^2/2-\beta^2/4)}|\alpha,\beta\rangle, \qquad (1.4.26)$$

which induces the following transformation in the fermionic quotient: $T: \hat{\mathcal{H}}_{s^a,s^b} \to \hat{\mathcal{H}}_{s^a,s^as^b}$, with matrices

$$\hat{T}_{\text{NS-NS}\to\text{NS-R}} = \hat{T}_{\text{NS-R}\to\text{NS-NS}} = \hat{T}_{\text{R-NS}\to\text{R-R}} = \hat{T}_{\text{R-R}\to\text{NS-R}} = \begin{pmatrix} 1 & 0\\ 0 & i \end{pmatrix}.$$
(1.4.27)

(The semion-fermion theory is rather degenerate, not least due to the fact that it factorizes into a bosonic TQFT and a trivial spin TQFT; in the general case, the matrices \hat{S}_{s^a,s^b} , \hat{T}_{s^a,s^b} are all typically distinct.)

A final ingredient as regards modularity is the charge-conjugation operation, which acts in homology as $C: (a, b) \mapsto (-a, -b)$. Unlike the *S*- and *T*-operations, charge-conjugation fixes all spin structures: $C: \hat{\mathcal{H}}_{s^a,s^b} \to \hat{\mathcal{H}}_{s^a,s^b}$. This operation acts on the U(1) connection as $a \mapsto -a$ or, equivalently, on the anyons as $\alpha \mapsto \bar{\alpha} = -\alpha$. The semion and the fermion are both self-conjugate (cf. $-1 = 1 \mod 2$), which means that C acts trivially on all the anyons of the theory. That being said, this operator need not act trivially on the Hilbert space. Its action is easily computed given the expression of the fermionic Hilbert space in terms of the bosonic parent, namely

$$\mathsf{C}|\alpha,\beta\rangle = |-\alpha \mod 2, -\beta \mod 4\rangle. \tag{1.4.28}$$

For example, this action induces the following action on the quotient theory:

$$C|0; \text{NS-NS}\rangle = \frac{1}{\sqrt{2}} (C|0, 0\rangle + C|0, 2\rangle) = \frac{1}{\sqrt{2}} (|0, 0\rangle + |0, 2\rangle) = |0; \text{NS-NS}\rangle.$$
(1.4.29)

Repeating this operation for the rest of basis vectors, we arrive at

$$\hat{\mathsf{C}}_{\text{NS-NS}\to\text{NS-NS}} = \hat{\mathsf{C}}_{\text{NS-R}\to\text{NS-R}} = \hat{\mathsf{C}}_{\text{R-NS}\to\text{R-NS}} = -\hat{\mathsf{C}}_{\text{R-R}\to\text{R-R}} = \mathbf{1}_2 \tag{1.4.30}$$

or, more succinctly, $\hat{\mathsf{C}} = (-1)^{\operatorname{Arf}(s)}$.

Given the explicit expressions for the $\hat{S}, \hat{T}, \hat{C}$ matrices, we can check that they realize a unitary representation of the modular group. In this case, the lack of Majorana lines means that $(-1)^F$ is trivial, which means that the modular algebra is just that of the regular torus with no punctures. In other words, the modular transformations satisfy $S^2 = (ST)^3 = C$, with $C^2 = 1$. The matrices $\hat{S}, \hat{T}, \hat{C}$ calculated above indeed satisfy this algebra, as expected. In checking this one must keep in mind that \hat{S}, \hat{T} do not live in $\text{End}(\hat{\mathcal{H}}_s)$ (unlike in the bosonic case) but rather in $\text{Hom}(\hat{\mathcal{H}}_s, \hat{\mathcal{H}}_{s'})$ (cf. (5.1.41)).

Wilson lines. An arbitrary element $|v\rangle \in \hat{\mathcal{H}}_{s^{a},s^{b}} \cong \mathbb{C}^{2|0}$ can be written as $|v\rangle = c_{0}|0; s^{a}s^{b}\rangle + c_{1}|1; s^{a}s^{b}\rangle$ for some coefficients $c_{0}, c_{1} \in \mathbb{C}$. Furthermore, all operators $\mathcal{O} \in \text{End}(\mathcal{H}_{s^{a},s^{b}})$ can be represented as 2×2 complex matrices. The Wilson lines themselves, in particular, are represented as 2×2 matrices. Their explicit form can be inferred from the expression for our basis vectors in terms of those of the bosonic parent, and the fact that the Wilson lines in the bosonic theory act as in (1.4.14). For example,

$$W^{(a)}(1,0)|1; \text{NS-NS}\rangle = \frac{1}{\sqrt{2}}W^{(a)}(1,0)(|1,0\rangle + |1,2\rangle)$$

= $\frac{1}{\sqrt{2}}e^{-i\pi}(|1,0\rangle + |1,2\rangle)$
= $-|1; \text{NS-NS}\rangle$, (1.4.31)

and

$$W^{(\mathbf{b})}(1,0)|1; \text{NS-NS}\rangle = \frac{1}{\sqrt{2}}W^{(\mathbf{b})}(1,0)(|1,0\rangle + |1,2\rangle)$$

= $\frac{1}{\sqrt{2}}(|2,0\rangle + |2,2\rangle)$
= $+|0; \text{NS-NS}\rangle.$ (1.4.32)

The rest of matrix elements are computed using the same idea. Denoting the semion by $\varsigma = (1, 0)$, and the fermion by $\psi = (0, 2)$, the end result is the Pauli matrices

$$W_{s^{\mathbf{a}},s^{\mathbf{b}}}^{(\mathbf{a})}(\varsigma) = \sigma^{3}, \qquad W_{s^{\mathbf{a}},s^{\mathbf{b}}}^{(\mathbf{b})}(\varsigma) = \sigma^{1}, \qquad W_{s^{\mathbf{a}},s^{\mathbf{b}}}^{(\mathbf{c})}(\psi) = -s^{\mathbf{c}}\mathbf{1}_{2}.$$
 (1.4.33)

(As before, the fact that $W^{(c)}(\varsigma)$ is independent of s^a, s^b is rather particular to this simple system; in generic spin TQFTs these matrices depend non-trivially on the spin structure.)

One can easily check that the Wilson lines satisfy the fusion rules of the theory, namely $\varsigma^2 = \psi^2 = \mathbf{1}$.

Time-reversal. Finally, we implement time-reversal as an explicit operator in $\hat{\mathcal{H}}_{s^a,s^b}$. We write $T = \tau K$, where K denotes complex conjugation and $\tau \in \mathbb{C}^2 \times \mathbb{C}^2$; this factorisation is not canonical, in the sense that τ and K are separately convention-dependent – only their product is meaningful. We shall fix them by declaring that our basis is real, $K|\alpha; s^a s^b\rangle = |\alpha; s^a s^b\rangle$, where $\alpha = 0, 1$. In other words, K acts by complex-conjugating the coefficients:

$$K(c_0|0; s^{\mathbf{a}}s^{\mathbf{b}}\rangle + c_1|1; s^{\mathbf{a}}s^{\mathbf{b}}\rangle) \equiv c_0^*|0; s^{\mathbf{a}}s^{\mathbf{b}}\rangle + c_1^*|1; s^{\mathbf{a}}s^{\mathbf{b}}\rangle.$$
(1.4.34)

Naturally, this definition is not basis-independent. But T, which is the object we care about, is, so this is enough for our purposes.

We recall that T acts on the U(1) fields as T(a) = a - b, T(b) = 2a - b (cf. (1.4.11)). This induces the following transformation on the Wilson lines:

$$W^{(c)}(\varsigma) = \exp i \oint_{c} a \mapsto \exp i \oint_{c} (a-b) \equiv W^{(c)}(\varsigma) W^{(c)}(\psi)$$

$$W^{(c)}(\psi) = \exp i \oint_{c} b \mapsto \exp i \oint_{c} (2a-b) \equiv W^{(c)}(\psi)$$
(1.4.35)

where we have used $W(\varsigma)^2 = W(\varsigma^2) = \mathbf{1}_2$. More to the point, time-reversal acts on the anyons by fixing the vacuum and the fermion, and by exchanging $\varsigma \leftrightarrow \varsigma \times \psi$.

With this, time-reversal acts on the Hilbert space as follows:

$$\mathsf{T} W_{s^{\mathsf{a}},s^{\mathsf{b}}}^{(\mathsf{c})}(\alpha)\mathsf{T}^{-1} = W_{s^{\mathsf{a}},s^{\mathsf{b}}}^{(\mathsf{c})}(\mathsf{T}(\alpha)) \qquad \mathsf{c} \in \{\mathsf{a},\mathsf{b}\},\tag{1.4.36}$$

where $\alpha \in \{\mathbf{1}, \varsigma, \psi, \psi \times \varsigma\}$. As $W(\psi) \propto \mathbf{1}_2$, the only non-trivial equation corresponds to the semion, $W(\varsigma)$, which in our basis reads

$$\tau_{s^{a},s^{b}}(W_{s^{a},s^{b}}^{(c)}(\varsigma))^{*}\tau_{s^{a},s^{b}}^{-1} = -s^{c}W_{s^{a},s^{b}}^{(c)}(\varsigma) \qquad c \in \{a,b\}.$$
(1.4.37)

This is nothing but a set of linear equations in the components of τ , with solution

$$(\tau_{s^{\mathbf{a}},s^{\mathbf{b}}})_{\alpha,\beta} = (-s^{\mathbf{b}})^{\alpha} \delta_{\alpha+\beta+\frac{1}{2}(s^{\mathbf{a}}+1)}$$
(1.4.38)

up to an inconsequential global phase, and where $\alpha, \beta = 0, 1$ and $\delta_x = 1$ if x is even, and $\delta_x = 0$ if odd.

We next note that

$$(\tau\tau^*)_{\alpha,\beta} = \sum_{\gamma} \tau_{\alpha,\gamma} \tau^*_{\gamma,\beta}$$

=
$$\sum_{\gamma} (-s^{\mathbf{b}})^{\alpha+\gamma} \delta_{\alpha+\gamma+\frac{1}{2}(s^{\mathbf{a}}+1)} \delta_{\beta+\gamma+\frac{1}{2}(s^{\mathbf{a}}+1)}$$

=
$$(-s^{\mathbf{b}})^{\frac{1}{2}(s^{\mathbf{a}}+1)} \delta_{\alpha+\beta}.$$
 (1.4.39)

Finally, observe that the expression above can also be written as $\tau \tau^* = (-1)^{\operatorname{Arf}(s)}$, which means that time-reversal satisfies

$$\mathsf{T}^{2} = \tau \tau^{*} \equiv (-1)^{\operatorname{Arf}(s)} \,. \tag{1.4.40}$$

In this theory, fermion parity is trivial, $(-1)^F \equiv 1$. Therefore, the equation above means that the time-reversal algebra $\mathsf{T}^2 = (-1)^F$ is deformed when acting on the Hilbert space, in the form

$$\mathsf{T}^{2} = (-1)^{F} \times (-1)^{\operatorname{Arf}(s)} \tag{1.4.41}$$

which is precisely what we expected, given that the theory has $\nu = 2 \mod 4$.

1.4.2 $\nu = 1 \mod 2$: fermion layer

In this section we consider time-reversal invariant theories with ν odd and show that they realize the expected behavior associated with the fermion layer. We work out in detail here the example of SO(3)₃ Chern-Simons theory that has $\nu = 3$. One can repeat the exercise for other theories with ν odd, such as SO(5)₅, which has $\nu = 5$; the main conclusions are identical.

With this in mind, we next study the spin TQFT $SO(3)_3$. This is the smallest intrinsically fermionic topological theory, in the sense that it supports both bosonic and fermionic states, unlike the previous section, where all states were bosonic. The presence of fermionic states is directly related to the presence of Majorana lines, i.e., of fixed-points of the condensing fermion. These will introduce new ingredients into the picture.

The bosonic parent of the theory is $\text{Spin}(3)_3 = \text{SU}(2)_6$, which becomes $\text{SO}(3)_3$ upon gauging the \mathbb{Z}_2 center. So we first construct the bosonic theory. This theory has seven states, labelled by their isospin: $|j\rangle$, where $j = 0, \frac{1}{2}, 1, \ldots, 3$. Under modular transformations, these states transform as

$$S|j\rangle = \sum_{j'} S_{j,j'}|j'\rangle, \qquad S_{j,j'} = \frac{1}{2}\sin\frac{\pi}{8}(2j+1)(2j'+1)$$

$$T|j\rangle = e^{\pi i(2j(2j+2)-3)/16}|j\rangle.$$
(1.4.42)

The condensing fermion ψ corresponds to the line j = 3. The braiding phase of a generic line j with respect to ψ is $B(j, \psi) = (-1)^{2j}$, which means that the NS-lines are those with integral isospin, and the R-lines are those with half-integral isospin:

NS:
$$\mathcal{A}_{NS} = \{j = 0, 1, 2, 3\}$$

R: $\mathcal{A}_{R} = \{j = \frac{1}{2}, \frac{3}{2}, \frac{5}{2}\}.$ (1.4.43)

The NS-lines are all in two-dimensional orbits, paired up as follows:

$$0 \stackrel{\times\psi}{\longleftrightarrow} 3, \qquad 1 \stackrel{\times\psi}{\longleftrightarrow} 2. \tag{1.4.44}$$

On the other hand, in the R sector there is one two-dimensional orbit, and one fixed-point:

$$\frac{1}{2} \stackrel{\times\psi}{\longleftrightarrow} \frac{5}{2}, \qquad \frac{3}{2} \stackrel{\searrow}{\to} \stackrel{\times\psi}{} . \tag{1.4.45}$$

In other words, 0, 1, 2, 3 are all *a*-type; $\frac{1}{2}, \frac{5}{2}$ are both *x*-type; and $\frac{3}{2}$ is *m*-type.

Hilbert space. With the information above we have all we need in order to construct the Hilbert space of the fermionic theory. As usual, the states of the condensed phase $SO(3)_3$ are expressed in terms of those of its parent, the specific expression being determined by the choice of spin structure:

• If we take **NS-NS** boundary conditions, the two states are

$$|0; \text{NS-NS}\rangle = \frac{1}{\sqrt{2}}(|0\rangle + |3\rangle)$$

$$|1; \text{NS-NS}\rangle = \frac{1}{\sqrt{2}}(|1\rangle + |2\rangle).$$
(1.4.46)

• If we take **NS-R** boundary conditions, the two states are

$$|0; \text{NS-R}\rangle = \frac{1}{\sqrt{2}}(|0\rangle - |3\rangle)$$

$$|1; \text{NS-R}\rangle = \frac{1}{\sqrt{2}}(|1\rangle - |2\rangle).$$
(1.4.47)

• If we take **R-NS** boundary conditions, the two states are

$$|0; \text{R-NS}\rangle = \frac{1}{\sqrt{2}}(|1/2\rangle + |5/2\rangle)$$

 $|1; \text{R-NS}\rangle = |3/2\rangle.$ (1.4.48)

• If we take **R-R** boundary conditions, the two states are

$$|0; \text{R-R}\rangle = \frac{1}{\sqrt{2}}(|1/2\rangle - |5/2\rangle)$$

 $|1; \text{R-R}\rangle = |3/2; \psi\rangle,$ (1.4.49)

where, we remind the reader, $|\alpha; \beta\rangle$ denotes the anyon α in presence of a β puncture (cf. (5.1.6)).

Modularity. As a check of the formalism so far, let us construct the modular data associated to these states, and check that it behaves as expected from general considerations. The even-spin-structure Hilbert spaces do not contain punctures, which means their modular data is computed in exactly the same way as in the previous section. For example, performing an S-transformation on an NS-R state we expect to obtain an R-NS state, which is easily confirmed:

where in the second line we have denoted $\xi = \sin \pi/4 = 1/\sqrt{2}$.

Acting with S on the rest of even-spin-structure Hilbert spaces we see that they are indeed permuted as $S: \hat{\mathcal{H}}_{s^a,s^b} \to \hat{\mathcal{H}}_{s^b,s^a}$; and, moreover, this action is effected by the following matrices:

$$\hat{S}_{\text{NS-NS}\to\text{NS-NS}} = \frac{1}{2} \begin{pmatrix} +\sqrt{2} - \sqrt{2} & +\sqrt{2} + \sqrt{2} \\ +\sqrt{2} + \sqrt{2} & -\sqrt{2} - \sqrt{2} \end{pmatrix}$$

$$\hat{S}_{\text{NS-R}\to\text{R-NS}} = \hat{S}_{\text{R-NS}\to\text{NS-R}} = \frac{1}{\sqrt{2}} \begin{pmatrix} +1 & +1 \\ +1 & -1 \end{pmatrix}.$$
(1.4.51)

T-transformations work similarly. For example, they should map states in the NS-NS sector into the NS-R sector, which is indeed what happens:

$$T|0; \text{NS-NS} \rangle = \frac{1}{\sqrt{2}} (T|0\rangle + T|3\rangle)$$

= $e^{-3\pi i/16} \frac{1}{\sqrt{2}} (|0\rangle - |3\rangle)$
= $e^{-3\pi i/16} |0; \text{NS-R}\rangle.$ (1.4.52)

Acting with T on the rest of basis vectors, one confirms that T-transformations map

 $T: \hat{\mathcal{H}}_{s^{a},s^{b}} \to \hat{\mathcal{H}}_{s^{a},s^{a}s^{b}}$, through the following matrices:

$$\hat{T}_{\text{NS-NS}\to\text{NS-R}} = \hat{T}_{\text{NS-R}\to\text{NS-NS}} = e^{-3\pi i/16} \begin{pmatrix} 1 & 0\\ 0 & i \end{pmatrix}$$

$$\hat{T}_{\text{R-NS}\to\text{R-NS}} = \begin{pmatrix} 1 & 0\\ 0 & e^{3\pi i/4} \end{pmatrix}.$$
(1.4.53)

One can easily check that all the expected properties of the modular group (i.e., (5.1.41)) are satisfied, where $(-1)^F = +\mathbf{1}_2$, as all states are bosonic.

The odd-spin-structure Hilbert space $\hat{\mathcal{H}}_{R-R}$ is much more interesting. The state $|0; R-R\rangle \sim |1/2\rangle - |5/2\rangle$ contains no punctures, so it is a boson, whereas the state $|1; R-R\rangle = |3/2; \psi\rangle$ has a ψ -puncture, so it is a fermion. In other words, in the R-R sector the fermion parity operator is non-trivial, $(-1)^F = \sigma_z$. This makes the analysis of modular transformations more involved. In particular, these transformations should not mix these two states, and they should not take us outside $\hat{\mathcal{H}}_{R-R}$ (as the s = (+1, +1) spin structure is fixed by all of the mapping class group). These expectations are confirmed by direct computation. For example, acting on $|0; R-R\rangle$ with an S-transformation we get

$$S|0; \text{R-R}\rangle = \frac{1}{\sqrt{2}} (S|1/2\rangle - S|5/2\rangle) = \frac{1}{\sqrt{2}} (|1/2\rangle - |5/2\rangle) = |0; \text{R-R}\rangle$$
(1.4.54)

which is indeed in $\hat{\mathcal{H}}_{R-R}$ (and has not mixed with the fermion, as it never could: modular transformations do not mix configurations with different punctures).

The action of S on $|1; \text{R-R}\rangle = |3/2; \psi\rangle$ is more subtle, because the state contains a puncture, so we need the S-matrix in the once-punctured torus. This matrix is given by (5.1.18)

$$S_{3/2,3/2}(\psi) = \sum_{j=0}^{3} \frac{\theta_j}{\theta_{3/2}^2} S_{0,j} F_{3/2,3/2} \begin{bmatrix} \psi & 3/2 \\ 3/2 & j \end{bmatrix}$$

$$= \sum_{j=0}^{3} e^{\pi i (2j(j+1)-15)/8} \times \frac{1}{2} \sin \frac{\pi}{8} (2j+1) \times (-1)^j$$

$$= (-1)^{7/4},$$

(1.4.55)

which means that, altogether,

$$\hat{S}_{\text{R-R}\to\text{R-R}} = \begin{pmatrix} 1 & 0\\ 0 & (-1)^{7/4} \end{pmatrix} .$$
(1.4.56)

T-transformations, on the other hand, do not care about the puncture, so they are just given by the spin of the states:

$$\hat{T}_{\text{R-R}\to\text{R-R}} = \begin{pmatrix} 1 & 0\\ 0 & (-1)^{3/4} \end{pmatrix}.$$
(1.4.57)

These two matrices are also easily seen to satisfy the expected modular properties, namely they are unitary and obey the algebra of $Mp_1(\mathbb{Z})$, to wit, $\hat{S}^2 = (\hat{S}\hat{T})^3$, $\hat{S}^4 = (-1)^F$.

Wilson lines. Given the choice of basis for the different Hilbert spaces as above, one can express the operators of the theory – the Wilson lines – as 2×2 complex matrices. The even-spin-structure Hilbert spaces contain no punctures, so the basic idea is identical to the previous sections, namely we induce the action on the quotient theory from that of its bosonic parent. In the bosonic parent the Wilson lines act as

$$W^{(\mathbf{b})}(j)|j'\rangle = |j \times j'\rangle \equiv \sum_{j''=|j-j'|}^{\min(j+j',6-j-j')} |j''\rangle,$$
 (1.4.58)

which means that, for example,

$$W^{(\mathbf{b})}(1)|0; \text{NS-NS}\rangle = \frac{1}{\sqrt{2}} (W^{(\mathbf{b})}(1)|0\rangle + W^{(\mathbf{b})}(1)|3\rangle) = \frac{1}{\sqrt{2}} (|1\rangle + |2\rangle), \qquad (1.4.59)$$

which we identify as $|1; NS-NS\rangle$. Similarly, $W^{(b)}(1)|1; NS-NS\rangle = |0; NS-NS\rangle + 2|1; NS-NS\rangle$. The **a**-cycle computation is analogous: in the bosonic parent Wilson lines act as

$$W^{(a)}(j)|j'\rangle = \frac{S_{j,j'}}{S_{0,j'}}|j'\rangle,$$
 (1.4.60)

which implies that

$$W^{(a)}(1)|0; \text{NS-NS}\rangle = \frac{1}{\sqrt{2}} (W^{(a)}(1)|0\rangle + W^{(a)}(1)|3\rangle) = \frac{1}{\sqrt{2}} (1+\sqrt{2})(|0\rangle + |3\rangle),$$
(1.4.61)

which equals $(1 + \sqrt{2})|0; \text{NS-NS}\rangle$. Repeating this calculation on all even-spin-structure Hilbert spaces, we obtain the following collection of matrices:

$$W_{\rm NS-NS}^{(a)}(1) = \begin{pmatrix} 1+\sqrt{2} & 0\\ 0 & 1-\sqrt{2} \end{pmatrix} \qquad W_{\rm NS-NS}^{(b)}(1) = \begin{pmatrix} 0 & 1\\ 1 & 2 \end{pmatrix}$$
$$W_{\rm NS-R}^{(a)}(1) = \begin{pmatrix} 1+\sqrt{2} & 0\\ 0 & 1-\sqrt{2} \end{pmatrix} \qquad W_{\rm NS-R}^{(b)}(1) = \begin{pmatrix} 0 & 1\\ 1 & 0 \end{pmatrix} \qquad (1.4.62)$$
$$W_{\rm R-NS}^{(a)}(1) = \begin{pmatrix} 1 & 0\\ 0 & -1 \end{pmatrix} \qquad W_{\rm R-NS}^{(b)}(1) = \begin{pmatrix} 1 & \sqrt{2}\\ \sqrt{2} & 1 \end{pmatrix},$$

The same computation for the rest of NS-lines yields $W_{s^{a},s^{b}}^{(c)}(3) = -s^{c}\mathbf{1}_{2}$ for the transparent fermion, and $W_{s^{a},s^{b}}^{(c)}(2) = W_{s^{a},s^{b}}^{(c)}(1)W_{s^{a},s^{b}}^{(c)}(3)$ (which is the expected relation given the fusion rule $2 = 1 \times 3$, i.e., that the lines j = 1, 2 are paired up through fusion with j = 3). One can also check that these matrices satisfy the algebra required by the fusion rule $1 \times 1 = 0 + 1 + 2$, namely $W_{s^{a},s^{b}}^{(c)}(1)^{2} = \mathbf{1}_{2} + (1 - s^{c})W_{s^{a},s^{b}}^{(c)}(1)$. Finally, it is also checked that, under modular transformations, these matrices are permuted as they should, e.g. $S_{s^{a},s^{b}}W_{s^{a},s^{b}}^{(a)}(\alpha)(S_{s^{a},s^{b}})^{\dagger} = W_{s^{b},s^{a}}^{(b)}(\bar{\alpha})$.

We now move on to the odd-spin-structure sector, the R-R Hilbert space. The bosonic state $|0; \text{R-R}\rangle \sim |1/2\rangle - |5/2\rangle$ contains no punctures, so it behaves in the same manner as the states in the even-spin-structure sector, for example

$$W^{(\mathbf{b})}(1)|0; \mathbf{R} \cdot \mathbf{R} \rangle = \frac{1}{\sqrt{2}} (W^{(\mathbf{b})}(1)|1/2\rangle - W^{(\mathbf{b})}(1)|5/2\rangle)$$

= $\frac{1}{\sqrt{2}} (|1/2\rangle + |3/2\rangle - |3/2\rangle - |5/2\rangle)$ (1.4.63)
= $|0; \mathbf{R} \cdot \mathbf{R} \rangle$.

The fermionic state $|1; R-R\rangle = |3/2; \psi\rangle$, on the other hand, requires using the data of the once-punctured torus, cf. (5.1.11):

$$W^{(\mathbf{b})}(1)|1; \mathbf{R} \cdot \mathbf{R}\rangle = W^{(\mathbf{b})}(1)|3/2; \psi\rangle$$

= $F_{3/2,3/2} \begin{bmatrix} 3 & 3/2 \\ 3/2 & 1 \end{bmatrix} |3/2; \psi\rangle$ (1.4.64)

which evaluates to $-|3/2;\psi\rangle$.

The a-cycle does not see the puncture (cf. (5.1.10)), and so its evaluation is straightforward. All in all, the Wilson lines in the R-R sector read

$$W_{\text{R-R}}^{(a)}(1) = W_{\text{R-R}}^{(b)}(1) = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix},$$
 (1.4.65)

together with $W_{\text{R-R}}^{(c)}(3) = -\mathbf{1}_2$ for the transparent fermion, and $W_{\text{R-R}}^{(c)}(2) = -W_{\text{R-R}}^{(c)}(1)$ (as expected from the fusion rule $2 = 1 \times 3$). It is easily checked that these matrices behave properly under modular transformations, e.g. $S_{\text{R-R}}W_{\text{R-R}}^{(a)}(\alpha)(S_{\text{R-R}})^{\dagger} = W_{\text{R-R}}^{(b)}(\bar{\alpha})$.

Time-reversal. Finally, we discuss the behaviour of the theory under time-reversal. Recall that $SO(3)_3$ is time-reversal symmetric thanks to the level-rank duality [112]

$$SO(3)_3 \longleftrightarrow SO(3)_{-3} \times SO(9)_1$$
 (1.4.66)

A key aspect of this duality is that time-reversal is not really a symmetry of $SO(3)_3$, but rather a map $T: SO(3)_3 \mapsto SO(3)_3 \times SO(9)_{-1}$. The factor $SO(9)_1$ is invertible, so we should think of $SO(3)_3$ being time-reversal invariant only if we mod out by SPTs. In the strict sense, it is not.

In the $U(1)_k$ case this obstruction was easily circumvented: the duality $U(1)_k \leftrightarrow U(1)_{-k} \times$ SO(4)₁ could be rewritten as $U(1)_k \times U(1)_{-1} \leftrightarrow U(1)_{-k} \times U(1)_{+1}$, i.e., we could break up the invertible factor into two, and split them symmetrically into the two theories. In this situation, we would say time-reversal is not a symmetry of $U(1)_k$, but rather of $U(1)_k \times U(1)_{-1}$: this theory is identical to its conjugate, even taking into account SPTs.

In the SO(3)₃ case, no such solution is possible: the SPT SO(9)₁ cannot be split into two equal factors; such splitting would require fractional levels SO(9/2)₁², which is not well-defined as a 3d theory.

An equivalent way to phrase this discussion is by thinking of the central charge – indeed, this number is what classifies 3*d*-SPTs with no symmetry. Time-reversal always maps c into -c, which means a theory can only be time-reversal invariant, in the strict sense, if c = 0. If $c \neq 0$, we may be able to correct this by multiplying by a suitable SPT, but this is not always possible. Indeed, the SPT SO $(n)_1$ has c = n/2, which means we can only correct the central charge in multiples of 1/2. In other words, the minimal SPT has c = 1/2, corresponding to a single edge Majorana fermion. Any other SPT will consist of an integral number of copies of this system.

In the $U(1)_k$ case, the central charge takes value c = 1, so this obstruction is avoidable: we just have to tensor the theory with two copies of the Majorana fermion, i.e., $SO(1)_1^2 = U(1)_1$. This makes the central charge of the product theory, $U(1)_k \times U(1)_{-1}$, vanish, making it a valid candidate for a time-reversal invariant theory. In the $SO(3)_3$ case, c = 9/4, which is not a multiple of 1/2, which means no redefinition can correct the central charge. The theory is not, and cannot be made, time-reversal invariant in the strict sense. Only in the relative sense when we think of QFTs as absolute theories, modulo invertible ones.

More generally, if a given theory A is known to be time-reversal invariant in the relative sense, then necessarily $c \propto 1/4$. Indeed, time-reversal maps A into \overline{A} , modulo some SPT, and so $A \leftrightarrow \overline{A} \times SO(n)_1$ for some n. The central charge of A therefore satisfies c(A) = -c(A) + n/2, which means that c(A) = n/4, as claimed. If $c = 0 \mod 1/4$, i.e., if $c \propto 1/2$, then the theory can be made time-reversal invariant in the strict sense, by considering $A \times SO(2c(A))_{-1}$, whose central charge vanishes. If $c \neq 0 \mod 1/4$, this is not possible. In other words, $c \mod 1/4$ measures the most primitive obstruction to being time-reversal invariant as a pure 3d theory. This is the ν_1 layer.²⁹

The discussion above is set up in the framework of spin TQFTs, but an analogous situation happens in other families of theories. For example, the minimal *bosonic* SPT is $(E_8)_1$, which has c = 8, which means that bosonic time-reversal invariant theories always have $c \propto 4$, and that $c \mod 4$ is the first layer in the anomaly of time-reversal invariance.

Going back to our example of $SO(3)_3$, let us see what we can say about time-reversal,

 $^{^{29}}$ See [123] for a very similar recent discussion.

neglecting the fact that it is not an operator acting on $SO(3)_3$, but rather a map from this theory into $SO(3)_3 \times SO(9)_{-1}$. Time-reversal acts on the lines of $SO(3)_3$ as $1 \leftrightarrow 2 \equiv 1 \times \psi$. If we ignore the SPT, this descends to the Hilbert space action

$$\tau \left(W_{s^{\mathbf{a}},s^{\mathbf{b}}}^{(\mathbf{c})}(1) \right)^{*} \tau^{-1} = W_{s^{\mathbf{a}},s^{\mathbf{b}}}^{(\mathbf{c})}(2)$$

= $-s^{\mathbf{c}} W_{s^{\mathbf{a}},s^{\mathbf{b}}}^{(\mathbf{c})}(1)$ (1.4.67)

where $T = \tau K$. The solution to this matrix equation is

$$\tau_{\text{NS-NS}} \propto \mathbf{1}_2$$

$$\tau_{\text{NS-R}} \propto \sigma^z$$

$$\tau_{\text{R-NS}} \propto \sigma^x$$
(1.4.68)

for the even spin structures, and

$$\tau_{\text{R-R}} = \begin{pmatrix} 0 & z_1 \\ z_2 & 0 \end{pmatrix} \tag{1.4.69}$$

for the odd spin structure, where $z_{1,2} \in \mathbb{C}$ are some arbitrary coefficients.

Finally, recall that $(-1)^F = \mathbf{1}_2$ for even spin structure, and $(-1)^F = \sigma^z$ for the odd spin structure. It is clear from these expressions that τ commutes with $(-1)^F$ for even spin structures, and anti-commutes for the odd spin structure

$$\{\mathsf{T}, (-1)^F\} = 0, \qquad (1.4.70)$$

precisely as expected from a system with ν odd and associated with the ψ layer. We have also established that this behavior is present in other time-reversal invariant theories with ν odd.

Chapter 2 Infrared phases of 2d QCD.

Authorship. The content of this chapter is taken almost verbatim from the paper [2] written in collaboration with Jaume Gomis and Matthew Yu.

Abstract. We derive the necessary and sufficient conditions for a 2d gauge theory to develop a mass gap, by studying the QCD Hamiltonian. The conditions can be explicitly solved, and we provide the complete list of all 2d QCD theories that have a quantum mechanical gap in their spectrum. The list of gapped theories includes QCD models with quarks in vector-like as well as chiral representations. The gapped theories consist of several infinite families of classical gauge groups with quarks in rank 1 and 2 representations, plus a finite number of isolated cases. We also put forward and analyze the effective infrared description of QCD — TQFTs for gapped theories and CFTs for gapless theories — and exhibit several interesting features in the infrared, such as the existence of non-trivial global 't Hooft anomalies and emergent supersymmetry. We identify 2d QCD theories that flow in the infrared to celebrated CFTs such as minimal models, Wess-Zumino-Witten and Kazama-Suzuki models.

Acknowledgments. We would like to thank Joaquim Gomis, Kentaro Hori, Theo Johnson-Freyd, Zohar Komargodski, Adam Schwimmer, and Ryan Thorngren for useful discussions.

2.1	Introduction
2.2	$2d$ QCD theories $\ldots \ldots $ 81
2.3	Symmetries, 't Hooft Anomalies and Gaplessness
2.4	Mass spectrum and QCD Hamiltonians
2.5	Infrared dynamics of $2d$ QCD
2.6	Conventions and Background
2.7	Infrared coset CFTs
2.8	Abelian theories

2.1 Introduction

A central theme in physics is unraveling the low energy phenomena that emerges from a physical system described by a collection of microscopic degrees of freedom and interactions. The long distance behavior of the system crucially depends on whether the spectrum of the Hamiltonian is gapped or gapless, but determining which phase is realized is often a nonperturbative problem.

In broad terms, a gapped system is described at low energies by a topological quantum field theory (TQFT), while the asymptotic low energy dynamics of a gapless one is captured by a conformal field theory (CFT).³⁰ Ascertaining whether a system flows to a TQFT or a CFT, and to which one, can be out of reach because of large quantum fluctuations, which are responsible for a wealth of low energy phenomena.

QCD theories are an important class of strongly coupled systems in which it is nontrivial to postulate the infrared dynamics. Determining whether Yang-Mills theory with gauge group G coupled to massless quarks in a representation R of G in $d \leq 4$ spacetime dimensions is gapped or gapless remains an open problem. We henceforth refer to such theories of massless quarks and gluons as QCD theories. The following qualitative picture is expected:

- QCD theories without quarks, that is, pure Yang-Mills theory, are believed to be gapped. For simply connected gauge group G, the infrared is described by the trivial TQFT.³¹
- QCD theories with a large number of quarks more precisely, with a large Dynkin index³² are gapless. In 4d, this is by virtue of the beta function [124, 125] being positive for a sufficiently large number of quarks, which implies that the infrared is described by a CFT of free massless particles. In 3d, the fact that QCD theories flow to a weakly coupled CFT can be established in the limit of large Dynkin index [25, 126].

While gapped QCD theories in 4d have been known for some time [42, 43], it is only recently that examples of gapped QCD theories in 3d, together with their infrared TQFTs, have been put forward [30] (see also [3, 113, 127-130]). Little is otherwise known about whether

³⁰The CFT can be either a symmetry preserving nontrivial fixed point of the renormalization group, the extreme infrared limit of the nonlinear theory of Goldstone bosons when the vacuum spontaneously breaks a continuous symmetry, or free massless particles in a symmetric vacuum (e.g. infrared free gauge theories).

³¹In 4*d* the theory has a unique vacuum for $\theta \neq \pi$ while for $\theta = \pi$ the time-reversal symmetry is spontaneously broken and there are two trivially gapped vacua [40]. Yang-Mills theory in 3*d* can be enriched by a Chern-Simons term and then the theory in the infrared is gapped and described by a nontrivial TQFT.

 $^{{}^{32}}N_F$ fermions in a representation R of G has Dynkin index $N_F \times I(R)$, where $\operatorname{tr}(t_R^a t_R^b) = I(R)\delta^{ab}$.

		I(R)
R = 0	R small	R large
TQFT	??	CFT

Figure 2.1: Diagram describing infrared dynamics of QCD with massless quarks in a representation R of the gauge group G. The theory without quarks is expected to be gapped, and is gapless for large enough R. The intermediate regime where the representation R is small remains an open problem.

a given QCD theory is gapped or not, and which TQFT/CFT describes its infrared limit (see fig. 2.1).

In this paper we determine all the QCD theories in 2d that are gapped, and therefore those that are gapless. The full classification of gapped QCD theories is summarized in tables 2.1 and 2.2.

In 2*d* QCD, the quark content is specified by a pair of representations (R_{ℓ}, R_r) of the gauge group *G* acting on the left and right chiral quarks. We denote such a QCD theory by $(G; R_{\ell}, R_r)$.³³ We derive the necessary and sufficient conditions for a QCD theory $(G; R_{\ell}, R_r)$ to be gapped by analyzing the explicit lighcone and temporal Hamiltonians of QCD. Lightcone quantization, where x^+ and x^- are time, and the canonical Hamiltonian formalism, where x^0 is time, yield exactly the same conditions.

From our Hamiltonian analysis, the following criterion is derived: a QCD theory is gapless if and only if there exists a canonical, chiral, dimension 2 primary operator of the quark current algebra constructed from either the left chiral quarks or right chiral quarks. A QCD theory is gapped if and only if both these left and right chiral operators vanish identically. These operator equations, derived by studying the Hamiltonian(s) in the ultraviolet, can be completely solved, yielding the classification of gapped theories.³⁴ The exhaustive list of gauge groups G and quark contents (R_{ℓ}, R_r) of all the QCD theories that are gapped appears in tables 2.1 and 2.2, corresponding to vector-like and chiral QCD theories respectively. Any

³⁴The operators equations that are the necessary and sufficient conditions for a QCD theory to be gapped

$$T_{\mathfrak{so}(\dim R_\ell)_1} - T_{G_{I(R_\ell)}} = 0$$

$$\overline{T}_{\mathfrak{so}(\dim R_r)_1} - \overline{T}_{G_{I(R_r)}} = 0$$

correspond to all the conformal embeddings into the $\mathfrak{so}(\dim R_\ell)_1$ and $\mathfrak{so}(\dim R_r)_1$ current algebras, and are in one-to-one correspondence with Cartan's classification of symmetric spaces. $T_{\mathfrak{so}(\dim R_\ell)_1}/\overline{T}_{\mathfrak{so}(\dim R_r)_1}$ is the canonical energy-momentum tensor of the left/right chiral quarks in the ultraviolet, and $T_{G_{I(R_\ell)}}/\overline{T}_{G_{I(R_r)}}$ is the left/right moving Sugawara energy-momentum tensor of the current algebra $G_{I(R)}$ at level I(R). See section 2.4 for details.

³³See section 2.2 for the role of the topology of the gauge group G in defining topological sectors, discrete theta angles, gauge anomalies, etc.

other QCD theory not in the tables is gapless.

\mathfrak{g}	R	\mathfrak{g}	R
$\forall \mathfrak{g}$	adjoint	$\mathfrak{su}(2)$	5
$\mathfrak{so}(N)$		$\mathfrak{so}(9)$	16
$\mathfrak{u}(N)$	\Box_q	F_4	26
$\mathfrak{so}(N)$		$\mathfrak{sp}(4)$	42
$\mathfrak{sp}(N)$		$\mathfrak{su}(8)$	70
$\mathfrak{u}(N)$	\square_q	$\mathfrak{so}(16)$	128
$\mathfrak{u}(N)$	\square_q	$\mathfrak{so}(10) + \mathfrak{u}(1)$	16_q
$\mathfrak{su}(M) + \mathfrak{su}(N) + \mathfrak{u}(1)$	$(\Box,\Box)_q$	$E_6 + \mathfrak{u}(1)$	27_{q}
$\mathfrak{so}(M) + \mathfrak{so}(N)$	(\Box, \Box)	$\mathfrak{su}(2) + \mathfrak{su}(2)$	(2 , 4)
$\mathfrak{sp}(M) + \mathfrak{sp}(N)$	(\Box, \Box)	$\mathfrak{su}(2) + \mathfrak{sp}(3)$	(2 , 14)
		$\mathfrak{su}(2) + \mathfrak{su}(6)$	$({f 2},{f 20})$
		$\mathfrak{su}(2) + \mathfrak{so}(12)$	(2 , 32)
		$\mathfrak{su}(2) + E_7$	$({f 2},{f 56})$
$\bigoplus_i \mathfrak{g}_i$	$\bigoplus_i (1,,R_i,,1)_{\vec{q_i}}$		

Table 2.1: Classification of vector-like gapped QCD theories (G; R, R). \mathfrak{g} denotes the Lie algebra of the gauge group and R the representation of the quarks (given in terms of a Young diagram or dimension of the representation). The global form of the gauge group G is arbitrary, as long as it admits R as a representation. $q \in \mathbb{Z}$ is the charge under the $\mathfrak{u}(1)$ gauge group factor. The left columns include adjoint QCD for arbitrary gauge group and families of theories for the classical groups, while the right columns contain isolated theories. Gapped QCD theories with classical gauge groups must have quarks transforming in rank-one or rank-two representations, any other representation leading to a gapless theory. The bottom entry indicates an arbitrary tensor product of gapped theories (\mathfrak{g}_i, R_i) constructed from the entries in the table that have a U(1) gauge group factor. These theories are coupled together via the $\mathfrak{u}(1)$ matrix of charges $\{\vec{q}_i\}$, which must be non-singular (see section 2.4.3 for details).

Remarkably, there exist chiral QCD theories that are gapped. The complete classification of chiral gapped QCD theories is given in table 2.2. From the classification of vector-like gapped theories in table 2.1, we can construct chiral gapped theories in two ways:

• A chiral QCD theory $(G; R_{\ell}, R_r)$ is gapped if and only if $(R_{\ell}, R_r) = (\sigma_{\ell} \cdot R, \sigma_r \cdot R)$ and the vector-like theory (G; R, R) appears in table 2.1, where σ_{ℓ}, σ_r are outer automorphisms of \mathfrak{g} . Here $\sigma \cdot R$ denotes the action of σ on R.³⁵ An example of such a chiral gapped

 $^{^{35}}$ See table 2.3 for a list of the automorphisms of simple Lie algebras. If G contains a U(1) factor then σ

$$\begin{array}{c|c}
\mathfrak{g} & (R_{\ell}, R_{r}) \\
\\
\mathfrak{g} & (\sigma_{\ell} \cdot R, \sigma_{r} \cdot R) \\
\bigoplus_{i} \mathfrak{g}_{i} & \bigoplus_{i} (\mathbf{1}, \dots, (\sigma_{\ell,i} \cdot R_{i}, \sigma_{r,i} \cdot R_{i}), \dots, \mathbf{1})_{\vec{q}_{\ell,i}, \vec{q}_{r,i}}
\end{array}$$

Table 2.2: Classification of chiral gapped QCD theories. Here \mathfrak{g} and R label the vector-like gapped theories from table 2.1. $\sigma_{\ell,i}, \sigma_{r,i}$ denote outer automorphisms of \mathfrak{g}_i (such as complex conjugation for simply-laced groups, or triality for $\mathfrak{so}(8)$). $\vec{q}_{\ell,i}, \vec{q}_{r,i}$ are tuples of charges for $\mathfrak{u}(1)$ factors. These charge matrices must have trivial kernel and cancel gauge anomalies, but are otherwise arbitrary (see section 2.4.3 for details).

theory is

$$(\operatorname{Spin}(8); \mathbf{8}_v, \mathbf{8}_c), \qquad (2.1.1)$$

corresponding to the triality automorphism acting on the vector-like gapped theory $(\text{Spin}(8); \mathbf{8}_v, \mathbf{8}_v)$ that appears in table 2.1.

• A chiral gapped QCD theory can be constructed by taking arbitrary tensor products of the basic gapped theories with quarks in a complex representation (there are seven such entries in table 2.1). The theories are coupled via the integral matrices q_{ℓ} and q_r that specify the charges under the $\mathfrak{u}(1)$ gauge group factors for the left and right chiral quarks. In order for theory to be gapped these matrices must be non-singular. A concrete example of such a chiral gapped theory is

$$\prod_{i} \mathcal{U}(n_{i}) \text{ with quarks } R = \bigoplus_{i} (\mathbf{1}, \dots, \Box_{i}, \dots, \mathbf{1})_{\vec{q}_{\ell,i}, \vec{q}_{r,i}}, \qquad (2.1.2)$$

corresponding to the tensor product of the vector-like gapped theories $(U(n_i); \Box_q, \Box_q)$, coupled via their U(1) gauge subgroups.

Having established which QCD theories are gapped and which are gapless, our next goal is to put forward the explicit low energy description of all QCD theories. In the gapped case this means finding the specific topological degrees of freedom carried by the vacua, the infrared TQFT, and in the gapless case finding the specific massless degrees of freedom of the infrared CFT.

Determining the long distance description of a given QCD theory is nontrivial. Unlike the question of whether a QCD theory is gapped or gapless, which can be answered rigorously, the task of finding the specific infrared degrees of freedom requires some guesswork. The most natural and straightforward conjecture is that the infrared description of $(G; R_{\ell}, R_r)$ is given by the $g^2 \to \infty$ limit of the QCD Lagrangian, since g has mass dimension. This limit

can also be chosen to act on the U(1) charge by reversing its sign.

yields the gauged WZW description of a CFT with the following left and right chiral algebras

$$\frac{\mathrm{SO}(\dim(R_{\ell}))_{1}}{G_{I(R_{\ell})}} \times \frac{\mathrm{SO}(\dim(R_{r}))_{1}}{G_{I(R_{r})}}.$$
(2.1.3)

This coset can be shown to be a TQFT – and hence to correspond to a gapped QCD theory – if and only if $(G; R_{\ell}, R_r)$ is in table 2.1 or 2.2. In this sense, our results derived from the Hamiltonian analysis are perfectly consistent with the conjectured infrared description. One can study many interesting aspects of QCD beyond the existence of a gap using the infrared coset description (2.1.3). For example, many well-known CFTs, such as minimal models, emerge in the infrared of QCD, which allows us to map non-trivial questions about the dynamics of QCD theories into questions about these CFTs, which can then be answered explicitly.

If a QCD theory has continuous chiral symmetries, then the infrared CFT necessarily contains a Wess–Zumino–Witten (WZW) factor for these symmetries; this sector carries the corresponding perturbative 't Hooft anomalies for the continuous symmetries. Even in the absence of continuous chiral symmetries, the infrared CFT is nontrivial when the QCD theory is gapless, and is based on a chiral algebra without spin one currents.³⁶ Interestingly, 't Hooft anomaly matching predicts the existence of some hitherto unknown 't Hooft anomalies for discrete global symmetries in these CFTs, such as nonperturbative anomalies for time-reversal symmetry and discrete chiral symmetries.

As an example, the proposal implies that QCD with a classical gauge group and with fundamental quarks flows in the infrared to a WZW CFT:

$$\begin{array}{lll}
\operatorname{SU}(N) + N_F \square & \xrightarrow{\operatorname{infrared}} & \operatorname{U}(N_F)_N \operatorname{WZW} \\
\operatorname{SO}(N) + N_F \square & \xrightarrow{\operatorname{infrared}} & \operatorname{SO}(N_F)_N \operatorname{WZW} \\
\operatorname{Sp}(N) + N_F \square & \xrightarrow{\operatorname{infrared}} & \operatorname{Sp}(N_F)_N \operatorname{WZW}.
\end{array}$$
(2.1.4)

These theories indeed carry the 't Hooft anomalies for the continuous flavor symmetries. Moreover, using that the 't Hooft anomalies must match predicts that these WZW models are endowed with several non-trivial global anomalies, which can be exhibited by general arguments or by brute-force computation in specific examples. For instance, the renormalization group flow predicts that the $SO(N_F)_N$ WZW model has a global anomaly associated with time-reversal symmetry, with $T^2 = (-1)^F$, which takes the value $NN_F \mod 2$. Many other such examples can be constructed, leading to a wealth of CFTs in the infrared and 't Hooft anomalies thereof.

³⁶For example, the QCD theories with G = SU(2) with a single quark in a spin $j \in \mathbb{Z}$ representation, that is (SU(2), j, j), has an infrared chiral algebra given by the W-algebra $\mathcal{W}(2, 4, \ldots, 2j)$. For j = 1, 2 the theory is gapped and flows to a TQFT, while for j = 3 the spin 4 and 6 currents become null and the chiral algebra is the Virasoro algebra, and the theory flows to the fermionic tricritical Ising model.

The plan for the rest of the chapter is as follows. In section 2.2 we set up the stage by carefully analyzing the microscopic description of QCD, its free parameters, topological sectors, and gauge anomalies. In section 2.3 we begin our investigations of the mass gap problem; in particular, here we exploit the symmetries and 't Hooft anomalies of 2d theories to constrain as much as possible the theories that can potentially be gapped. We find several simple criteria that automatically force the theory to be gapless, thus reducing considerably the landscape of gapped theories. These criteria alone are not enough to actually prove that a given theory is gapped, so in section 2.4 we turn our attention to an explicit analysis of the Hamiltonian of QCD. We recover the necessary conditions laid out in the previous section, and also find sufficient conditions as well, culminating in a concrete list of gapped theories. In section 2.5 we reconsider our results, this time in light of the conjecture (2.1.3) which proposes a concrete description of QCD at low energies. We give further evidence for the correctness of this conjecture, and subsequently apply it to many explicit examples.

We also include several extra sections (which were initially sub-appendices, but this was not allowed by University formatting code). Section 2.6 can be used as a reference for our conventions, and it contains some technical computations that supplement the main text. Section 2.7 reviews some relevant facts about 2d CFTs and, in particular, cosets of the form (2.1.3) that conjecturally encapsulate the low-energy degrees of freedom of QCD. We also work out some examples in some detail. Finally, section 2.8 contains a separate discussion of QCD theories where the gauge group is abelian, i.e., where G consists of factors of U(1) only. While this type of theories is covered by our general discussion from other sections, when studied in isolation one can be more explicit in some of our claims. Also, they illustrate some general features of other non-abelian QCD theories that contain U(1) factors, such as the breaking of some U(1) flavor symmetries as the result of flavor-gauge mixed anomalies.

$2.2 \quad 2d \text{ QCD theories}$

The field content of a 2*d* QCD theory is specified by a choice of gauge group *G* and a pair of representations R_{ℓ} and R_r of *G* acting on left and right chiral quarks.³⁷ We label such a QCD theory by the triple $(G; R_{\ell}, R_r)$. *G* is an arbitrary compact, connected Lie group with Lie algebra $\mathfrak{g} = \bigoplus_I \mathfrak{g}_I \bigoplus_m \mathfrak{u}(1)_m$, a direct sum of simple Lie algebras \mathfrak{g}_I and abelian Lie algebras $\mathfrak{u}(1)_m$. The Lagrangian of a QCD theory with massless quarks is

$$\mathcal{L}_{\rm QCD} = -\frac{1}{2} \operatorname{tr}(g^{-2} F_{\mu\nu} F^{\mu\nu}) + i \psi_{\ell}^{\dagger} D_{-} \psi_{\ell} + i \psi_{r}^{\dagger} D_{+} \psi_{r} , \qquad (2.2.1)$$

where

$$D_{-}\psi_{\ell} = (\partial_{-} - iA_{-}^{a}t_{\ell}^{a})\psi_{\ell}, \qquad D_{+}\psi_{r} = (\partial_{+} - iA_{+}^{a}t_{r}^{a})\psi_{r}, \qquad (2.2.2)$$

³⁷In 2*d*, complex conjugation does not reverse the chirality of a fermion since the chirality matrix $\gamma_3 = \gamma^0 \gamma^1$ does not include an *i*, unlike in 4*d* where $\gamma_3 = i\gamma^0\gamma^1\gamma^2\gamma^3$ and conjugation does flip chirality. This implies that the most general 2*d* QCD theory cannot be written using just left chiral fermions, in contrast to 4*d*.

 $t_{\ell}^{a}(t_{r}^{a})$ are the generators of the Lie algebra \mathfrak{g} in representation $R_{\ell}(R_{r})$ and we have introduced lightcone coordinates $x^{\pm} = \frac{1}{\sqrt{2}}(x^{0} \pm x^{1})$, and $A_{\pm}^{a} = \frac{1}{2}(A_{0}^{a} \pm A_{1}^{a})$. Each gauge group factor has a gauge coupling, which is captured by g^{-2} inside the trace. See section 2.6 for details and conventions. See also table 2.3 for a summary of simple Lie groups and relevant properties.

$G_{\rm sc}$	$\mathrm{SU}(N)$	$\operatorname{Sp}(N)$	$\operatorname{Spin}(2N+1)$	$\operatorname{Spin}(4N)$	$\operatorname{Spin}(4N+2)$	E_6	E_7	E_8	F_4	G_2
h	N	N+1	2N - 1	4N-2	4N	12	18	30	9	4
$\operatorname{Out}(\mathfrak{g})$	\mathbb{Z}_2	•	•	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{Z}_2	•	•	•	•
$Z(G_{\rm sc})$	\mathbb{Z}_N	\mathbb{Z}_2	\mathbb{Z}_2	$\mathbb{Z}_2 imes \mathbb{Z}_2$	\mathbb{Z}_4	\mathbb{Z}_3	\mathbb{Z}_2	•	•	

Table 2.3: Lie data for the simply-connected simple Lie groups G_{sc} . Here *h* denotes the dual Coxeter number (defined as the Dynkin index of the adjoint representation). Out(\mathfrak{g}) is the group of outer automorphisms of \mathfrak{g} , which corresponds to charge conjugation symmetry of QCD. $Z(G_{sc})$ is the center of the gauge group, which contains the one-form center symmetry of QCD. For SU(2), Out(\mathfrak{g}) is trivial, and for Spin(8), it is enhanced to Out(\mathfrak{g}) = S₃ (triality).

Global issues, flux tubes and theta terms.

A QCD theory requires specifying a global choice of a gauge group G with Lie algebra \mathfrak{g} . We consider first QCD with the simply-connected form of the gauge group $G_{\rm sc}$, which we denote by $(G_{\rm sc}; R_{\ell}, R_r)$. Such a QCD theory may have a one-form symmetry Γ [32], where $\Gamma \subset Z(G_{\rm sc})$ is a subgroup of the center (cf. table 2.3).

2d QFTs with a one-form symmetry Γ have topological sectors labeled by a representation $\rho \in \Gamma^{\vee}$ of Γ , where Γ^{\vee} is the Pontryagin dual group. Physically, a topological sector labeled by $\rho \in \Gamma^{\vee}$ describes the theory in the presence of a flux tube created by a quark-antiquark pair of charge ρ at \pm -infinity [131, 132], a background that preserves Poincaré invariance in 2d.

We now consider the theory with gauge group $G = G_{\rm sc}/\Gamma$.³⁸ Since *G*-bundles are classified by $H^2(M, \pi_1(G)) \cong \Gamma$, the sum over gauge fields in the functional integral can be weighted by a discrete theta term labeled by $\rho \in \Gamma^{\vee}$, which takes the form of a generalized Stiefel-Whitney class

$$i \int_M w_\rho(G) \,. \tag{2.2.3}$$

We label such a QCD theory by $(G; R_{\ell}, R_r)_{\rho}$.

We proceed to prove that:

³⁸The discussion can be easily extended to the case $G_{\rm sc}/K$, where $K \subset \Gamma$.

• $(G_{\rm sc}; R_{\ell}, R_r)$ with a ρ -flux tube is the same as $(G; R_{\ell}, R_r)_{\rho}$

The one-form global symmetry Γ of $(G_{sc}; R_{\ell}, R_r)$ implies that there is a topological local operator U_g , with $g \in \Gamma$, which acts on line operators.³⁹ Diagonalizing the topological local operators U_g on the Hilbert space leads to the decomposition

$$\mathcal{H} = \bigoplus_{\rho \in \Gamma^{\vee}} \mathcal{H}_{\rho} \,, \tag{2.2.6}$$

where ρ is an irreducible representation of Γ and

$$|\psi\rangle \in \mathcal{H}_{\rho} \iff U_g |\psi\rangle = \chi_{\rho}(g) |\psi\rangle.$$
 (2.2.7)

The Hilbert space \mathcal{H}_{ρ} corresponds to $(G_{sc}; R_{\ell}, R_r)$ in the presence of a ρ -flux tube.

The QCD theory $(G; R_{\ell}, R_r)_{\rho}$ can be constructed by gauging the one-form symmetry of $(G_{sc}; R_{\ell}, R_r)$ tensored with an SPT phase for the one-form symmetry Γ ; such SPTs are labeled by an element ρ of the reduced cobordism group $\tilde{\Omega}^2_{\text{Spin}}(B^2\Gamma) \cong \Gamma^{\vee}$. A nontrivial SPT_{ρ} weights the sum over $g \in \Gamma$ that defines $(G; R_{\ell}, R_r)_{\rho}$ by gauging Γ with the phase

$$\chi^*_{\rho}(g),$$
 (2.2.8)

where $\rho \in \Gamma^{\vee}$ is a representation of Γ and $\chi_{\rho}(g)$ is a character of Γ in the representation ρ . This is an alternative way to think about the discrete theta term (2.2.3).

Consider a theory T in the presence of a fixed two-form gauge field B_2 for the one-form symmetry Γ , which takes values in $H^2(M, \Gamma) = \Gamma$. The partition function of the theory in such a background is given by

$$Z_T[g] = \sum_{\rho \in \Gamma^{\vee}} Z_T(\rho) \chi_{\rho}(g) , \qquad (2.2.9)$$

where the sum over ρ is due to the Hilbert space structure (2.2.6) and $g \in \Gamma$ labels the choice of background gauge field B_2 . $Z_T(\rho)$ is the partition function of the theory in the presence of a ρ -flux tube.

$$U_g \mathcal{L} U_g^{-1} = \chi_\rho(g) \mathcal{L} \,, \tag{2.2.4}$$

$$[\mathcal{L}] = \bigoplus_{\rho \in \Gamma^{\vee}} [\mathcal{L}]_{\rho}, \qquad (2.2.5)$$

where line operators in $[\mathcal{L}]_{\rho}$ carry charge ρ .

³⁹The charge of a line operator \mathcal{L} under Γ is measured by U_g as

where $\rho \in \Gamma^{\vee}$ is an irreducible representation of Γ and $\chi_{\rho}(g)$ is a character of Γ in the representation ρ . This means that the spectrum of line operators in the theory can be organized according to their charges under Γ as

The theory T/Γ obtained by gauging Γ has a dual (-1)-form symmetry Γ^{\vee} [133–135], and T/Γ can be coupled to a background zero-form gauge field for this symmetry, which corresponds to an element $\hat{\rho}$ of Γ^{\vee} . The partition of the gauged theory in the presence of this background gauge field is

$$Z_{T/\Gamma}[\hat{\rho}] = \frac{1}{|\Gamma|} \sum_{g \in \Gamma} Z_T[g] \chi_{\hat{\rho}}^*(g) , \qquad (2.2.10)$$

where $\chi_{\hat{\rho}}^*(\hat{g})$ encodes the coupling of the background two-form gauge field for Γ with the zero-form gauge field for Γ^{\vee} . This describes the theory T/Γ with a discrete theta term labeled by $\hat{\rho} \in \Gamma^{\vee}$. Using equation (2.2.9) we arrive at

$$Z_{T/\Gamma}[\hat{\rho}] = \frac{1}{|\Gamma|} \sum_{g \in \Gamma} \sum_{\rho \in \Gamma^{\vee}} Z_T(\rho) \chi_{\rho}(g) \chi_{\hat{\rho}}^*(g) = Z_T(\hat{\rho}) , \qquad (2.2.11)$$

where we have used that $\sum_{g\in\Gamma} \chi_{\rho}(g)\chi_{\hat{\rho}}^*(g) = |\Gamma|\delta_{\rho,\hat{\rho}}$. Therefore, the partition function of the theory T/Γ with a theta term $\rho \in \Gamma^{\vee}$ is the same as the partition function of the original theory T in the sector with a with a ρ -flux tube, thus completing the proof that $(G_{sc}; R_{\ell}, R_r)$ with a ρ -flux tube is the same as $(G; R_{\ell}, R_r)_{\rho}$. This implies that it is sufficient to study QCD theories with a simply connected gauge group, which we will do henceforth.

We now turn to the next result

• $(G_{\rm sc}; R_{\ell}, R_r)$ is gapless if and only if it is gapless in the $\rho = 0$ flux tube sector.

This implies that for the purposes of classifying gapped QCD theories it suffices to consider QCD theories with simply connected gauge group and in the trivial flux tube sector.

This conclusion is a consequence of the fact that the (massless) QCD theory $(G_{\rm sc}; R_{\ell}, R_r)$ admits topological line operators \mathcal{L} that carry any charge under the one-form symmetry Γ [136].⁴⁰ Since \mathcal{L} carries one-form symmetry charge ρ , it defines a map between the Hilbert spaces $\mathcal{H}_{\rho=0}$ and \mathcal{H}_{ρ} : acting with \mathcal{L} on $\mathcal{H}_{\rho=0}$ creates states in \mathcal{H}_{ρ} . Physically, acting with a topological line \mathcal{L} inserts static probe charges ρ at \pm -infinity. Such a topological line operator \mathcal{L} interpolates between the Hamiltonian of the theory in distinct flux tube sectors

$$\mathcal{L}H_{\rho=0} = H_{\rho}\mathcal{L}\,,\tag{2.2.12}$$

where $H_{\rho=0}$ and H_{ρ} are the Hamiltonians of the theory in the trivial and ρ -flux tube sector respectively. Note that in general \mathcal{L} is a non-invertible topological operator and therefore (2.2.12) cannot be written as a similarity transformation. Since \mathcal{L} is topological it carries vanishing energy density (zero tension). Therefore it cannot lower the energy and the sector with a ρ -flux tube is gapless if and only it is gapless in the sector with no string (that is with $\rho = 0$).⁴¹ We note that this conclusion relies on the existence of topological line

 $^{^{40}}$ The global symmetries of the QCD Lagrangian (2.2.1) together with these topological lines make it technically natural to study the theory without four-fermi terms, cf. [99, 136].

⁴¹If $|\Omega\rangle$ is the ground state of $H_{\rho=0}$, the ground state of H_{ρ} is $\mathcal{L}|\Omega\rangle$.

operators, and these are not present generically in the theory with massive quarks, where indeed a massless particle can appear in the theory with a flux tube (see e.g. [137]).

In summary, for the purposes of classifying all gapped QCD theories we can, without loss of generality, consider the theory with simply connected gauge group in the trivial topological sector, without a string. Henceforth, we will use G to refer to the simply connected gauge group or use instead the Lie algebra \mathfrak{g} .

Gauge anomaly cancellation.

In order to define a consistent QCD theory, the global symmetry G acting on the free fermions in the deep ultraviolet must have no obstructions to being gauged. Therefore all anomalies for G gauge transformations, perturbative and nonperturbative, must cancel. Perturbative anomalies, that is, anomalies associated to G gauge transformations connected to the identity, are classified by the first summand in the free part of the spin cobordism group

$$\operatorname{Free}\left(\Omega_{\operatorname{spin}}^{4}(BG)\right) = \mathbb{Z}^{|I| + \frac{1}{2}|m|(|m|+1)} \oplus \mathbb{Z}, \qquad (2.2.13)$$

where |I| and |m| is the number of simple and abelian factors in \mathfrak{g} respectively. These anomalies are determined by a one-loop diagram and encoded in the first line of the anomaly polynomial

$$\left(\operatorname{tr}_{R_{\ell}} e^{F/2\pi} - \operatorname{tr}_{R_{r}} e^{F/2\pi}\right) \hat{A}(\mathcal{R}) \Big|_{4} = \sum_{a,b} \left[\operatorname{tr}(t_{\ell}^{a} t_{\ell}^{b}) - \operatorname{tr}(t_{r}^{a} t_{r}^{b})\right] \frac{F^{a} \wedge F^{b}}{8\pi^{2}} - \frac{p_{1}(\mathcal{R})}{24} \left[\operatorname{dim}(R_{\ell}) - \operatorname{dim}(R_{r})\right],$$

$$(2.2.14)$$

where F^a is the two-form field strength and $p_1(\mathcal{R})$ the first Pontryagin class for the background metric. Gauge anomaly cancellation requires that the representations R_{ℓ} and R_r of the left and right chiral quarks obey

$$\operatorname{tr}(t_{\ell}^{a}t_{\ell}^{b}) = \operatorname{tr}(t_{r}^{a}t_{r}^{b}) \qquad \forall \ a, b.$$

$$(2.2.15)$$

The nontrivial anomaly constraints in (2.2.15) are:⁴²

1. $\mathfrak{g}_{I}-\mathfrak{g}_{I}$ anomaly: $t^{a}_{\ell,r}$ are generators of the simple Lie algebra \mathfrak{g}_{I} . The anomaly cancelation condition requires that

$$I(R_{\ell}) - I(R_{r}) = 0, \qquad (2.2.16)$$

⁴²This is to be contrasted with the 4*d* anomaly cancelation equation $\operatorname{tr}(t_{\ell}^{a}\{t_{\ell}^{b}, t_{\ell}^{c}\}) = 0$ when the theory is written using left chiral fermions, which is nontrivial for the $\mathfrak{g}_{I}-\mathfrak{g}_{I}-\mathfrak{g}_{I}$, $\mathfrak{g}_{I}-\mathfrak{g}_{I}-\mathfrak{u}(1)_{m}$ and $\mathfrak{u}(1)_{m}-\mathfrak{u}(1)_{p}-\mathfrak{u}(1)_{p}$ anomalies. In 2*d* a chiral fermion in any irreducible representation of any \mathfrak{g} contributes to the $\mathfrak{g}_{I}-\mathfrak{g}_{I}$ anomaly, while in 4*d* only chiral fermions transforming in a complex representation of SU(N) contribute to the pure $\mathfrak{g}_{I}-\mathfrak{g}_{I}$ anomaly, because the rest of the simple Lie algebras have no cubic Casimir. Since a chiral fermion cannot be given a mass in 2*d*, unlike for a 4*d* chiral fermion in a real representation, any 2*d* chiral fermion can potentially contribute to the anomaly and, indeed, it does.

where I(R) is the Dynkin index of the representation R, defined by $\operatorname{tr}(t^a t^b) = I(R)\delta^{ab}$. The index of a reducible representation follows from $I(R_1 \oplus R_2) = I(R_1) + I(R_2)$.

2. $\mathfrak{u}(1)_m \cdot \mathfrak{u}(1)_n$ anomaly: $t^a_{\ell,r}$ are generators of an abelian Lie algebra. The anomaly cancelation condition is

$$\sum_{\ell} Q_{\ell,m} Q_{\ell,n} - \sum_{r} Q_{r,m} Q_{r,n} = 0, \qquad (2.2.17)$$

where $Q_{\ell,m}$ and $Q_{r,m}$ are the left and right $U(1)_m$ charges of the quarks.

A global symmetry G may have a more subtle obstruction to being gauged associated to a background G gauge transformation not connected to the identity, like the celebrated SU(2) global anomaly in 4d [75]. If the symmetry group G is gauged, like in QCD, global anomalies for G must also cancel for the gauge theory to be consistent. Topologically nontrivial gauge transformations in (compactified) 2d flat spacetime are classified by $\pi_2(G)$, which vanishes for any continuous Lie group G, and 2d gauge theories do not have this type of global anomalies. From the cobordism point of view of anomalies, the vanishing of the anomalies is seen through the fact that $\Omega_{\text{spin}}^3(BG) = 0$ (see e.g. [72]).⁴³ Therefore the anomaly cancelation conditions (2.2.15) are necessary and sufficient for a QCD theory to be consistent.

Since gravity couples to QCD as a nondynamical background field, it can be afflicted by gravitational anomalies without rendering the theory inconsistent. These anomalies are captured by the second \mathbb{Z} summand in (2.2.13) and by the second line of the anomaly polynomial in (2.2.14).⁴⁴ We discuss in the next section the implications that 't Hooft anomalies, including gravitational anomalies, have for the infrared dynamics of QCD theories.

Of course, vector-like theories (G; R, R), with $R_{\ell} = R_r$, are manifestly free of gauge anomalies. But in 2d, gauge-anomaly-free chiral QCD theories are abundant. Most of these chiral theories, however, have gravitational anomalies. There are, nonetheless, chiral gauge theories with neither gauge nor gravitational anomalies, i.e., simultaneous solutions to⁴⁵

$$\operatorname{tr}(t_{\ell}^{a}t_{\ell}^{b}) = \operatorname{tr}(t_{r}^{a}t_{r}^{b}) \qquad \forall \ a, b$$
$$\operatorname{dim}(R_{\ell}) = \operatorname{dim}(R_{r}). \qquad (2.2.18)$$

Unlike in 4d, where the beta-function for the gauge coupling constrains the quark content of 4d QCD theories that are strongly coupled in the infrared, any 2d QCD flows to strong coupling at low energies. Our first goal is to determine which 2d QCD theories are gapped, and which are gapless.

⁴³Global anomalies for a discrete symmetry group can be nontrivial. For example $\Omega^3_{\text{spin}}(B\mathbb{Z}_2) = \mathbb{Z}_8$.

⁴⁴A mixed $\mathfrak{u}(1)$ -gravity anomaly governed by $\nabla_{\mu}J^{\mu} = \alpha R$ can be written down, where J^{μ} is the $\mathfrak{u}(1)$ current. But $\alpha = 0$ in a unitary theory. It can be nonvanishing in a nonunitary theory, like in the string theory *bc* ghost system. Thus there are no mixed gauge-gravity anomalies in 2*d* QCD.

 $^{^{45}}$ A simple example of a chiral theory with no gravitational anomalies is (Spin(5); **35**, **5** + **30**). As a matter of fact, this theory has no continuous flavor symmetries, so it does not have any perturbative 't Hooft anomalies whatsoever.

2.3 Symmetries, 't Hooft Anomalies and Gaplessness

In this section we use symmetry and 't Hooft anomaly considerations to derive necessary conditions for a 2d QFT theory to be gapped. We start with a discussion of symmetries and 't Hooft anomalies and then use them to constraint the phases of 2d QFTs.

Symmetries provide a powerful organizing principle parametrizing the most general solution of a QFT consistent with the symmetries. But without further input, either perturbative or nonperturbative, symmetries do not inform the actual dynamics of a physical system.

An 't Hooft anomaly for a global symmetry, diagnosed by violations of Ward identities in the presence of nondynamical background gauge fields for global symmetries, instead, does inform the dynamics of the system. Since 't Hooft anomalies are quantized, they are invariant under symmetric deformations, and define invariants in the space of symmetric QFTs. In particular they are invariant under renormalization group transformations. While 't Hooft anomalies alone cannot determine the dynamics of a system, they rule out any dynamical scenario that does not match the microscopic 't Hooft anomalies. As such, 't Hooft anomalies provide nonperturbative guidance about the dynamics of QFTs.

A system defined at short distances with an 't Hooft anomaly for a symmetry cannot flow in the deep infrared to a trivially gapped theory, as this has vanishing 't Hooft anomalies. A system with an 't Hooft anomaly can flow either to a symmetry-preserving gapless phase or a symmetry breaking phase, which is gapless if the broken symmetry is continuous⁴⁶ and a TQFT if the broken symmetry is discrete.⁴⁷ If the anomalous symmetry transformation is discrete, the system may also flow to a symmetry preserving gapped phase described by a TQFT with topological order, which can saturate anomalies that are torsion classes.

A system with an 't Hooft anomaly for a continuous symmetry transformation cannot flow to a TQFT because an 't Hooft anomaly for a continuous symmetry implies a nonvanishing correlation function for conserved currents at separated points, and a TQFT, being topological, does not have such correlation functions. This implies that a system with perturbative anomalies, corresponding to anomalies for continuous symmetry transformations connected to the identity, can only flow to a symmetry preserving gapless phase or a symmetry breaking gapless phase.

In 2d, the fate of a system with an 't Hooft anomaly is further constrained by important theorems. These theorems, once combined with the discussion above, leads to the following implications:

1. Coleman-Mermin-Wagner theorem [138, 139]: a continuous global symmetry cannot be spontaneously broken in 2d.

⁴⁶More precisely, the anomaly is not torsion.

 $^{^{47}\}mathrm{More}$ precisely, the anomaly is torsion.

A 2d system with an 't Hooft anomaly for a continuous symmetry must flow to a symmetry preserving gapless phase.

2. A 2d TQFT does not have intrinsic topological order [140]: in 2d a symmetry preserving gapped phase cannot saturate 't Hooft anomalies

A 2d system with an 't Hooft anomaly for a discrete symmetry must flow to a symmetry preserving gapless phase or a symmetry breaking gapped phase described by a TQFT.

We are now ready to state the following far-reaching result for the dynamics of 2d QFTs: **Proposition 2.3.1** A 2d QFT with a continuous chiral global symmetry is symmetry preserving and gapless.

Consider a QFT with a U(1) global symmetry. The one-form current for the U(1) global symmetry is $J = J_{\mu} dx^{\mu} \equiv J_{+} dx^{+} + J_{-} dx^{-}$, with $J_{\pm} = \frac{1}{\sqrt{2}} (J_{0} \pm J_{1})$. This obeys the conservation equation (see section 2.6 for conventions)

$$\partial_{-}J_{+} + \partial_{+}J_{-} = 0. (2.3.1)$$

This is an operator equation that holds inside any correlation function as long as the location of the current J does not coincide with any operator insertions (conservation may fail at coincident points). Poincaré's lemma implies that locally the current takes the following form

$$J_{\mu} = \epsilon_{\mu\nu} \partial^{\nu} \phi \Longleftrightarrow J_{\pm} = \pm \partial_{\pm} \phi(x^+, x^-), \qquad (2.3.2)$$

where $\phi(x^+, x^-)$ is a scalar operator in the theory.

Consider now a chiral symmetry. A right-moving $U(1)_r$ symmetry is implemented by a conserved current that is antiselfdual

$$U(1)_r: J = -\star J \Longleftrightarrow J_+ = 0 \Longleftrightarrow \partial_+ J_- = 0.$$
(2.3.3)

By virtue of (2.3.2), a theory with a $U(1)_r$ symmetry contains a scalar operator ϕ that is right-moving

$$J_{+} = 0 \Longrightarrow \phi = \phi(x^{-}). \qquad (2.3.4)$$

Likewise, a left-moving $U(1)_{\ell}$ symmetry is generated by a conserved current that is selfdual

$$U(1)_{\ell}: J = +\star J \iff J_{-} = 0 \iff \partial_{-}J_{+} = 0, \qquad (2.3.5)$$

and a $U(1)_{\ell}$ symmetry implies the existence of a left-moving scalar operator

$$J_{-} = 0 \Longrightarrow \phi = \phi(x^{+}). \qquad (2.3.6)$$

This implies that a QFT with either a left or a right moving U(1) symmetry is necessarily gapless: the theory has a chiral scalar operator that creates chiral massless states when acting on the vacuum.

This proposition may seem at odds with our discussion above since, typically, symmetries alone cannot determine the infrared phase of a system. The reason that it does in this case is that a chiral U(1) symmetry automatically leads to an 't Hooft anomaly for that symmetry, as we show below. And as we mentioned above, an 't Hooft anomaly for a continuous global symmetry in 2d necessarily leads to a symmetry preserving gapless phase.

Consider the renormalization group flow out of a CFT with a $U(1)_{\ell}$ symmetry that is triggered by a $U(1)_{\ell}$ -invariant relevant operator.⁴⁸ Since the flow preserves the $U(1)_{\ell}$ symmetry, the most general two-point function for the $U(1)_{\ell}$ current consistent with dimensional analysis and Poincaré invariance is

$$\langle J_+(x)J_+(0)\rangle = \frac{K_\ell(x^+x^-/\mu^2)}{x^+x^+},$$
(2.3.7)

where μ is a scale generated along the renormalization group flow. In a unitary theory $K_{\ell} \ge 0$, with $K_{\ell} = 0$ if and only if $J_{+} = 0$. Demanding conservation law of $U(1)_{\ell}$ symmetry current at separated points $\partial_{-}J_{+} = 0$ implies that K_{ℓ} is a renormalization group invariant

$$\frac{\partial}{\partial \rho^2} K_\ell = 0, \qquad (2.3.8)$$

where we have introduced Rindler coordinates $x^{\pm} = \rho e^{\pm \sigma}$, so that

$$\langle J_+(x)J_+(0)\rangle = \frac{k_\ell}{x^+x^+}.$$
 (2.3.9)

In the deep ultraviolet, $K_{\ell}^{\text{UV}} = k_{\ell} \in \mathbb{Z}$ is the level of the $U(1)_{\ell}$ current algebra of the ultraviolet CFT. Therefore, k_{ℓ} is the 't Hooft anomaly coefficient for the $U(1)_{\ell}$ symmetry.⁴⁹ Since $k_{\ell} \neq 0$ implies that correlators have support at separated points (2.3.9), and $U(1)_{\ell}$ cannot be spontaneously broken, the infrared of a system with a $U(1)_{\ell}$ symmetry must be symmetry preserving and gapless.

This argument admits an interesting generalization. Consider now the renormalization group flow of a U(1)-symmetric CFT triggered by a U(1)-invariant relevant operator. In a unitary CFT with a normalizable vacuum, the conservation law for the U(1) current $J = J_+ dx^+ + J_- dx^-$ implies a separate conservation law for chiral U(1)_{ℓ} and U(1)_r symmetries,

 $^{^{48}}$ In QCD, the CFT in the ultraviolet is the CFT of free fermions, and the renormalization group flow is triggered by the gauge coupling.

⁴⁹The contact term implied by (2.3.9) leads to a violation of the conservation equation $\partial_{-}j_{+} = \frac{k_{\ell}}{2\pi}\partial_{+}A_{-}$ upon coupling system to a background gauge field for U(1)_{ℓ} via $\int d^{2}x A_{-}J_{+}$. In QCD, $k_{\ell} = \sum_{i} q_{i,\ell}^{2}$, where $q_{i,\ell}$ are the U(1)_{ℓ} charges of chiral fermions.

generated by J_{-} and J_{+} respectively [141]. The most general current two-point functions consistent with dimensional analysis and Poincaré invariance are

$$\langle J_{+}(x)J_{+}(0)\rangle = \frac{K_{\ell}(x^{+}x^{-}/\mu^{2})}{x^{+}x^{+}},$$

$$\langle J_{-}(x)J_{-}(0)\rangle = \frac{K_{r}(x^{+}x^{-}/\mu^{2})}{x^{-}x^{-}},$$

$$\langle J_{+}(x)J_{-}(0)\rangle = \frac{G(x^{+}x^{-}/\mu^{2})}{x^{+}x^{-}}.$$

$$(2.3.10)$$

In a parity invariant system $K_{\ell} = K_r$. Conservation of the U(1) current $\partial_- J_+ + \partial_+ J_- = 0$ at separated points implies that

$$\rho^2 \frac{\partial}{\partial \rho^2} (K_\ell + G) = G, \qquad \rho^2 \frac{\partial}{\partial \rho^2} (K_r + G) = G. \qquad (2.3.11)$$

Therefore, the quantity $K_{\ell} - K_r$ is a renormalization group invariant

$$\rho^2 \frac{\partial}{\partial \rho^2} (K_\ell - K_r) = 0. \qquad (2.3.12)$$

In the deep ultraviolet, $K_{\ell}^{\text{UV}} = k_{\ell} \in \mathbb{Z}$ and $K_r^{\text{UV}} = k_r \in \mathbb{Z}$ are the levels of the U(1)_{ℓ} and U(1)_r current algebras of the ultraviolet CFT. $K_{\ell} - K_r = k_{\ell} - k_r$ is the 't Hooft anomaly coefficient for the U(1) symmetry and is constant everywhere in the flow.⁵⁰ This must be reproduced by be infrared phase, and it can only be realized by a symmetry preserving gapless phase.

Proposition 2.3.2 A 2d QFT with a gravitational anomaly is gapless.

An almost identical reasoning applies to the conservation law of the energy-momentum tensor $T_{\mu\nu}$ along a renormalization group flow out of a CFT, for which we have

$$\partial_{-}T_{+\pm} + \partial_{+}T_{-\pm} = 0. \qquad (2.3.13)$$

In a unitary CFT with a normalizable vacuum, $T_{+-} = 0$ and T_{++} and T_{--} are chiral, that is $\partial_{-}T_{++} = \partial_{+}T_{--} = 0$, so that in the ultraviolet CFT, the correlators with support at separated points are

UV CFT:
$$\langle T_{++}(x)T_{++}(0)\rangle = \frac{c_{\ell}^{\text{UV}}}{2(x^{+})^4}, \qquad \langle T_{--}(x)T_{--}(0)\rangle = \frac{c_{r}^{\text{UV}}}{2(x^{-})^4}, \qquad (2.3.14)$$

where c_{ℓ}^{UV} and c_{r}^{UV} are the central chargers of the left and right-moving Virasoro algebras. In a parity invariant theory $c_{\ell}^{\text{UV}} = c_{r}^{\text{UV}}$. The quantity $c_{\ell}^{\text{UV}} - c_{r}^{\text{UV}}$ detects a gravitational 't Hooft

⁵⁰In QCD $k_{\ell} - k_r = \text{tr}_{\text{fermions}}(\gamma^3 Q^t Q)$, where Q is the U(1) charge of the fermions.

anomaly, and must be matched by the infrared phase.⁵¹ While the *c*-theorem [142] says that c_{ℓ} and c_r decrease along a renormalization group flow, the difference $c_{\ell} - c_r$ must remain constant. In a theory with a gravitational 't Hooft anomaly the energy-momentum tensor is a nontrivial operator, with separated point correlation functions. Since such correlation functions cannot be realized by a TQFT, a 2*d* theory with a gravitational 't Hooft anomaly is necessarily gapless.

Comparing with the anomaly polynomial (2.2.14), we have that the gravitational 't Hooft anomaly of the QCD theory $(G; R_{\ell}, R_r)$ is

$$c_{\ell}^{\rm UV} - c_{r}^{\rm UV} = \frac{1}{2} \left(\dim(R_{\ell}) - \dim(R_{r}) \right) .$$
 (2.3.15)

 $(G; R_{\ell}, R_r)$ with a gravitational 't Hooft anomaly in the ultraviolet is gapless.

2.3.1 Towards Gapped QCD Theories

In the previous section we established that a 2d theory with a continuous chiral symmetry is automatically gapless. Therefore, if we wish to classify QCD theories that are gapped, the first step is to determine which QCD theories have no such symmetries. These can be expressed as conditions on the quark content of the QCD theory as follows:⁵²

A 2d QFT with a gravitational anomaly is gapless.

Proposition 2.3.3 A necessary condition for $(G; R_{\ell}, R_r)$ with semisimple G to be gapped is that the representations R_{ℓ} and R_r of G are the direct sum of distinct, real irreducible representations of G. A QCD theory with a quark content that is not of this type is necessarily gapless.

 $(G; R_{\ell}, R_r)$ is obtained by gauging a diagonal subgroup G of the global symmetry acting on the left and right chiral quarks, and giving the gauge fields a kinetic term. For the purpose of identifying the continuous global symmetries of a QCD theory it suffices to discuss the Lie algebra of symmetries. The continuous global symmetry algebra acting on the quarks in the ultraviolet is

$$\mathfrak{so}(\dim(R_\ell)) \oplus \mathfrak{so}(\dim(R_r)),$$
 (2.3.16)

where dim $(R_{\ell/r})$ is the real dimension of the representation $R_{\ell/r}$ of \mathfrak{g} . Our immediate task is to answer for what choices of R_{ℓ} and R_r does a QCD theory admit a continuous chiral global symmetry, and is therefore gapless (cf. proposition 2.3.1).

In order to answer this question it suffices to consider the left chiral fermions, as an identical discussion holds for the right chiral ones. Consider left chiral fermions transforming

⁵¹This can be derived by imposing energy-momentum conservation law (2.3.13) on the most general two-point functions of T_{++}, T_{--} and T_{+-} at separated points.

⁵²We discuss QCD with a reductive gauge group, that is with abelian gauge group factors, below.

in an irreducible representation R of a semisimple Lie algebra \mathfrak{g} . A QCD theory with this quark content has a left chiral flavor symmetry if and only if the embedding

$$\mathfrak{so}(\dim(R)) \supset \mathfrak{g}$$
 (2.3.17)

has a nontrivial commutant, that is, there exists an algebra $\mathfrak{h} \subset \mathfrak{so}(\dim(R))$ such that $[\mathfrak{g}, \mathfrak{h}] = 0$. This depends on the nature of the representation R, which for now we take to be irreducible:

• A chiral quark in a complex representation has a chiral U(1) global symmetry.

A complex representation R of a semisimple Lie algebra \mathfrak{g} is described by traceless, antihermitian $(\dim(R)/2) \times (\dim(R)/2)$ matrices t. Therefore the pair (\mathfrak{g}, R) defines the following Lie algebra embedding and branching

$$\mathfrak{su}(\dim(R)/2) \supset \mathfrak{g}$$

fundamental $\mapsto R$. (2.3.18)

Since R is irreducible, the commutant of \mathfrak{g} in $\mathfrak{su}(\dim(R)/2)$ is trivial by Schur's lemma.

Let us now determine whether there is a commutant of \mathfrak{g} in $\mathfrak{so}(\dim(R))$, the symmetry algebra acting on the quarks. The Lie algebra $\mathfrak{su}(\dim(R)/2)$ embeds into the $\mathfrak{so}(\dim(R))$ symmetry algebra of the quarks as

$$\hat{t} = \begin{pmatrix} \operatorname{re}(t) & \operatorname{im}(t) \\ -\operatorname{im}(t) & \operatorname{re}(t) \end{pmatrix} \subset \mathfrak{so}(\dim(R)), \qquad (2.3.19)$$

where $\operatorname{re}(t)^T = -\operatorname{re}(t)$, $\operatorname{im}(t)^T = \operatorname{im}(t)$, with T denoting the transpose, and $\operatorname{tr}(t) = 0$. Since

$$U = \begin{pmatrix} 0 & 1\\ -1 & 0 \end{pmatrix} \subset \mathfrak{so}(\dim(R))$$
(2.3.20)

commutes with \hat{t} and $U \not\subset \mathfrak{su}(\dim(R)/2)$, a chiral quark in a complex representation has a chiral U(1) global symmetry. This also follows from the following sequence of embeddings

$$\mathfrak{so}(\dim(R)) \supset \mathfrak{su}(\dim(R)/2) \oplus \mathfrak{u}(1) \supset \mathfrak{g} \oplus \mathfrak{u}(1).$$
 (2.3.21)

• A chiral quark in a pseudoreal representation has a chiral $Sp(1) \simeq SU(2)$ global symmetry.

A pseudoreal representation R of a Lie algebra \mathfrak{g} is described by traceless, antihermitian $(\dim(R)/2) \times (\dim(R)/2)$ matrices t obeying

$$-t^T = JtJ^{-1}, (2.3.22)$$

where J is the canonical antisymmetric matrix

$$J = \begin{pmatrix} 0 & 1\\ -1 & 0 \end{pmatrix} . \tag{2.3.23}$$

These are precisely the generators of the $\mathfrak{sp}(\dim(R)/4)$ Lie algebra in the fundamental representation. Therefore the pair (\mathfrak{g}, R) defines the following Lie algebra embedding and branching

$$\mathfrak{sp}(\dim(R)/4) \supset \mathfrak{g}$$
fundamental $\mapsto R$.
$$(2.3.24)$$

Since R is irreducible, the commutant of \mathfrak{g} in $\mathfrak{sp}(\dim(R)/4)$ is trivial by Schur's lemma.

Let us now determine whether there is a commutant of \mathfrak{g} in $\mathfrak{so}(\dim(R))$, the symmetry algebra acting on the quarks. Since $J = i\sigma_2 \otimes 1$, the $\mathfrak{sp}(\dim(R)/4)$ matrices t, which obey (2.3.22), can be written as

$$t = \sum_{M=1}^{4} t_M \otimes q_M \,, \tag{2.3.25}$$

where t_M are real matrices obeying $t_a^T = t_a$ for a = 1, 2, 3 and $t_4^T = -t_4$.⁵³ Here we denote $q_M = (i\vec{\sigma}, 1)$, with $\vec{\sigma}$ the Pauli matrices, a two-dimensional complex-valued representation of the quaternions.

The embedding of $\mathfrak{sp}(\dim(R)/4)$ into the $\mathfrak{so}(\dim(R))$ symmetry algebra of the quarks is

$$\hat{t} = \sum_{M=1}^{4} t_M \otimes \hat{\sigma}_M \subset \mathfrak{so}(\dim(R)), \qquad (2.3.26)$$

where $\hat{\sigma}_M = (\sigma_1 \otimes i\sigma_2, i\sigma_2 \otimes 1, \sigma_3 \otimes i\sigma_2, 1 \otimes 1)$ is a four-dimensional real-valued representation of the quaternions. Since the matrices $U_a \subset \mathfrak{so}(\dim(R))$

$$U_{1} = 1 \otimes i\sigma_{2} \otimes \sigma_{1}$$

$$U_{2} = 1 \otimes 1 \otimes i\sigma_{2}$$

$$U_{3} = 1 \otimes i\sigma_{2} \otimes \sigma_{3}$$

$$(2.3.27)$$

commute with \hat{t} , generate an $\mathfrak{sp}(1)$ algebra (namely, $[\tau_a, \tau_b] = i\epsilon_{abc}\tau_c$ with $\tau_a = \frac{1}{2i}U_a$) and $U_a \not\subset \mathfrak{sp}(\dim(R)/4)$, a chiral quark in a pseudoreal representation has a chiral $\mathfrak{sp}(1)$ global symmetry. This also follows from the following sequence of embeddings

$$\mathfrak{so}(\dim(R)) \supset \mathfrak{sp}(\dim(R)/4) \oplus \mathfrak{sp}(1) \supset \mathfrak{g} \oplus \mathfrak{sp}(1).$$
 (2.3.28)

• A chiral quark in a real representation has no continuous global symmetry.

A real irreducible representation R of a Lie algebra \mathfrak{g} is described by traceless, antihermitian $(\dim(R)) \times (\dim(R))$ matrices t obeying

$$-t^T = t. (2.3.29)$$

⁵³There are three symmetric matrices and one antisymmetric, each being $n \times n$ with $n = \dim(R)/4$. Thus, there are $\frac{3}{2}n(n+1) + \frac{1}{2}n(n-1) = n(2n+1)$ degrees of freedom, which is precisely the dimension of the algebra $\mathfrak{sp}(n) \equiv \mathfrak{sp}(\dim(R)/4)$.

These are precisely the generators of the $\mathfrak{so}(\dim(R))$ Lie algebra in the fundamental representation. Therefore the pair (\mathfrak{g}, R) defines the following Lie algebra embedding and branching

$$\mathfrak{so}(\dim(R)) \supset \mathfrak{g}$$

fundamental $\mapsto R$. (2.3.30)

Since R is irreducible, the commutant of \mathfrak{g} in $\mathfrak{so}(\dim(R))$ is trivial by Schur's lemma. Therefore, a chiral quark in a real representation has no continuous global symmetry.

Let us now consider the case where the representation R is reducible. Since we are seeking QCD theories that are gapped, which means that they cannot have any continuous flavor symmetries, we take R to be the direct sum of irreducible, real representations R_{α} with multiplicity M_{α}

$$R = \bigoplus_{\alpha} M_{\alpha} \cdot R_{\alpha}, \quad \text{with} \quad M_{\alpha} \in \{0, 1, 2, \dots\}.$$
(2.3.31)

If $M_{\alpha} > 1$, then there is an $\mathfrak{so}(M_{\alpha})$ chiral flavor symmetry acting on the quarks. Indeed, the representation matrix for the reducible representation $M_{\alpha} \cdot R_{\alpha}$ can be written as

$$t = 1 \otimes t_{\alpha} \,, \tag{2.3.32}$$

where t_{α} is a representation of R_{α} . The matrix $U \subset \mathfrak{so}(M_{\alpha} \cdot \dim(R_{\alpha}))$

$$U = O \otimes 1 \qquad \text{with} \qquad O^T = -O \tag{2.3.33}$$

is a representation of $\mathfrak{so}(M_{\alpha})$ and commutes with t. It therefore generates an $\mathfrak{so}(M_{\alpha})$ flavor symmetry. This is also a consequence of the sequence of embeddings

$$\mathfrak{so}(M_{\alpha} \cdot \dim(R_{\alpha})) \supset \mathfrak{so}(\dim(R_{\alpha})) \oplus \mathfrak{so}(M_{\alpha}) \supset \mathfrak{g} \oplus \mathfrak{so}(M_{\alpha}).$$
 (2.3.34)

Finally, by virtue of Schur's lemma, the direct sum of distinct irreducible real representations $\bigoplus_{\alpha} R_{\alpha}$ does not have a continuous flavor symmetry, as the commutant of $\mathfrak{so}(\sum_{\alpha} \dim(R_{\alpha})) \supset \mathfrak{g}$ is trivial.

Let us now consider QCD with a reductive gauge group $G = K \times U(1)^n$, where K is semisimple. It suffices to consider the left chiral fermions, as an identical discussion holds for the right chiral ones. When the gauge group has U(1) factors, a classical U(1)_F chiral flavor symmetry may be broken by the Adler-Bell-Jackiw (ABJ) anomaly, that is, by a mixed a U(1)-U(1)_F anomaly. A classical semisimple symmetry always remains unbroken. Therefore, for the purposes of classifying QCD theories with $K \times U(1)^n$ gauge group and no flavor symmetries, consider QCD with chiral quarks transforming under $K \times U(1)^n$ as

$$\bigoplus_{I=1}^{N_F} (R_I, \vec{q}_I) , \qquad (2.3.35)$$

where R_I is a complex irreducible representation of K, q_I is an integral *n*-component charge vector under $U(1)^n$ and all pairs $(R_I, \vec{q_I})$ are distinct (any other quark content leads to a flavor symmetry). The classical chiral flavor symmetry is

$$\mathfrak{h}_{\text{classical}} = \bigoplus_{I=1}^{N_F} \mathfrak{u}(1)_I \,. \tag{2.3.36}$$

We want to determine under what conditions this symmetry is completely broken by the ABJ anomaly. Upon defining the integral matrix Q whose columns are

$$Q = (\dim(R_1)\vec{q_1}, \dim(R_2)\vec{q_2}, \dots, \dim(R_{N_F})\vec{q_{N_F}})$$
(2.3.37)

we arrive at the following result

Proposition 2.3.4 A necessary condition for $(G; R_{\ell}, R_r)$ with $G = K \times U(1)^n$ to be gapped is that the representations R_{ℓ} and R_r of G are of the irreducible form (2.3.35) and the charge matrices Q_{ℓ}, Q_r have trivial kernel. A QCD theory with a quark content that is not of this type is necessarily gapless.

The proof is straightforward. An arbitrary chiral $\mathfrak{u}(1) \subseteq \mathfrak{h}_{\text{classical}}$ flavor symmetry is specified by an integer vector $\vec{n} = (n_1, n_2, \ldots, n_{N_F})$ such that, given an angle $\alpha \in \mathfrak{u}(1)$, the *I*-th quark is rotated by an angle αn_I . The mixed ABJ anomaly between this $\mathfrak{u}(1)$ flavor symmetry and the $U(1)^n$ gauge group is $Q\vec{n}$. Therefore, no flavor symmetries remain if and only if Q has empty kernel (so that there are no nontrivial solutions to $Q\vec{n} = 0$). See also section 2.8 for a more in-depth discussion of QCD theories with $U(1)^n$ gauge group.

Our findings thus far are summarized in the two propositions of this subsection. This is as far as one can get using 't Hooft anomaly considerations. Answering whether a QCD theory $(G; R_{\ell}, R_r)$ obeying the conditions in the propositions is gapped or gapless requires studying the dynamics. The analysis thus far does not say, for example, whether the vector-like QCD theory with G = SU(2) and quarks in the isospin $j \in \mathbb{Z}$ representation (which is real) is gapped or gapless. We will provide a complete answer to these questions in the rest of the paper.

Before closing this section let us make one final remark. In this section we have capitalized on the symmetries of 2*d* QFTs as much as we could. In a nutshell, we showed that, if \mathfrak{h} denotes the chiral symmetry algebra of a 2*d* system, then the system is automatically gapless as soon as \mathfrak{h} is non-trivial. There is a nice physical interpretation of this result. In a unitary CFT, a chiral symmetry \mathfrak{h} is always enhanced to \mathfrak{h}_k affine algebra, for a suitable level *k*. Therefore, if \mathfrak{h} is non-zero, the system contains \mathfrak{h}_k massless currents which automatically make the system gapless: the infrared contains, at the very least, an \mathfrak{h}_k WZW CFT subsector. Therefore, a necessary condition for being gapped is that the chiral symmetry is trivial, $\mathfrak{h} \equiv 0$. We will see the \mathfrak{h}_k currents reappear explicitly in the Hamiltonian of QCD in the following section, and we will study them in more detail when we look at the infrared of QCD in section 2.5.

2.4 Mass spectrum and QCD Hamiltonians

In this section we analyze the mass spectrum of QCD by studying the quantization of the QCD Hamiltonians. The main result is a derivation of the necessary and sufficient conditions for a QCD theory to be gapped. Along the way, we prove that a QCD theory with a continuous global symmetry has massless particles in the spectrum, reproducing the result derived in section 2.3 by symmetry and 't Hooft anomaly arguments. We analyze the lightcone and temporal Hamiltonians, both yielding the same conditions for QCD to be gapped.

2.4.1 Lightcone Hamiltonian

Our aim in this section is to study the mass spectrum of QCD by quantizing the theory in the lightcone frame [143–145]. We review here the most salient features and formulas (see [146, 147] for reviews and recent work). The basic idea is to use the a lightcone coordinate, say $x^+ = \frac{1}{\sqrt{2}}(x^0 + x^1)$, as the time variable. This quantization defines the Hilbert space and the Cauchy data of the theory on a constant x^+ surface, and the conjugate lightcone Hamiltonian P^- evolves states in x^+ . When we choose x^+ to play the role of time, the lightcone coordinate x^- plays the role of a spatial coordinate. The momentum P^+ conjugate to x^- commutes with the lightcone Hamiltonian P^- , i.e., $[P^+, P^-] = 0$. Therefore, the mass spectrum of QCD can be obtained by simultaneously diagonalizing the operators P^+ and P^- , since

$$M^2 = 2P^+P^-. (2.4.1)$$

Positive semidefiniteness of M^2 and of the lightcone Hamiltonian P^- implies that all states in the Hilbert space have $P^+ \ge 0$. This combined with the fact that interactions preserve P^+ implies that the vacuum state in lightcone quantization is trivial, that is, the vacuum has no particles in it, and the nonperturbative vacuum coincides with the Fock vacuum. This makes lightcone quantization well adapted to study the meson and hadron spectrum of QCD.

Since P^- evolves states along x^+ , left-moving massless particles are not visible in the P^- Hamiltonian. This implies that the spectrum of the Hamiltonian P^- correctly accounts for all massive and right-moving massless particles, but does not detect left-moving massless particles. This shortcoming can be overcome by quantizing QCD using instead $x^- = \frac{1}{\sqrt{2}}(x^0 - x^1)$ as the lightcone time. In this frame, the lightcone Hamiltonian is instead \hat{P}^+ and its spectrum contains all the massive and left-moving massless particles. Therefore, by diagonalizing the QCD lightcone Hamiltonians P^- and \hat{P}^+ in the quantizations where x^+ and x^- is time respectively, all the massless particles of QCD are accounted for.

Let us proceed with the lightcone quantization of QCD with left and right chiral quarks in representations R_{ℓ} and R_r of the gauge group G. We start with the QCD Lagrangian

$$\mathcal{L}_{\text{QCD}} = \frac{1}{g^2} F^a_{+-} F^a_{+-} + i\psi^{\dagger}_{\ell} (\partial_- - iA^a_- t^a_{\ell})\psi_{\ell} + i\psi^{\dagger}_r (\partial_+ - iA^a_+ t^a_r)\psi_r \,. \tag{2.4.2}$$

We fix the gauge $A_{-}^{a} = 0$, in which all states have positive norm and there are no ghosts. In the lightcone gauge with x^{+} being time, the left-chiral quarks ψ_{ℓ} and A_{+}^{a} are not dynamical. They can be integrated out to yield

$$\mathcal{L}_{\text{QCD}} = i\psi_r^{\dagger}\partial_+\psi_r - g^2 J_r^a \frac{1}{\partial_-^2} J_r^a , \qquad (2.4.3)$$

where

$$J_{r}^{a} = \frac{1}{2} : \psi_{r}^{\dagger} t_{r}^{a} \psi_{r} :$$
 (2.4.4)

generates the right-moving quark current for \mathfrak{g} , the Lie algebra of G, which obeys $\partial_{-}J_{r}^{a}=0$. The Noether charges for the lightcone Hamiltonian and momentum are

$$P^{-} = -g^{2} \int dx^{-} : J_{r}^{a} \frac{1}{\partial_{-}^{2}} J_{r}^{a}:$$

$$P^{+} = i \int dx^{-} : \psi_{r}^{\dagger} \partial_{-} \psi_{r}:.$$
(2.4.5)

Choosing x^- as lightcone time instead results in the associated lightcone Hamiltonian and momentum

$$\hat{P}^{+} = -g^{2} \int \mathrm{d}x^{+} : J_{\ell}^{a} \frac{1}{\partial_{+}^{2}} J_{\ell}^{a}:$$

$$\hat{P}^{-} = i \int \mathrm{d}x^{+} : \psi_{\ell}^{\dagger} \partial_{+} \psi_{\ell}:,$$
(2.4.6)

where now

$$J^a_\ell = \frac{1}{2} : \psi^\dagger_\ell t^a_\ell \psi_\ell : \tag{2.4.7}$$

generates the left-moving quark current for G, with $\partial_+ J^a_\ell = 0$.

We are interested in determining when a QCD theory is gapped, and when it is gapless. This requires determining when the lightcone Hamiltonians (2.4.5) and (2.4.6) have zero eigenvalues. We will answer this question by studying the eigenvalues of the operators

$$H = -g^2 \int dx : J^a \frac{1}{\partial_x^2} J^a:$$

$$\hat{P} = i \int dx : \psi^{\dagger} \partial_x \psi:,$$
(2.4.8)

where $x = x^-$ or $x = x^+$ and $J^a = J^a_r$ or $J^a = J^a_\ell$ depending on whether x^+ or x^- is the lightcone time. Canonical quantization yields the following equal-time commutation relations

$$\{\psi_i^{\dagger}(x), \psi^j(y)\} = \delta_i^j \delta(x - y), \qquad (2.4.9)$$

where the Latin indices are the representation labels for the representation R of \mathfrak{g} of the relevant chiral quarks. The quark field expansion in Fourier modes is

$$\psi^{i}(x) = \frac{1}{\sqrt{2\pi}} \int_{0}^{\infty} \mathrm{d}k \left(a^{i}(k)e^{-ikx} + b^{\dagger}_{i}(k)e^{ikx} \right) \,, \tag{2.4.10}$$

where $k = k_{-}$ or $k = k_{+}$ depending on whether x^{+} or x^{-} is taken as the lightcone time. Note that in the lightcone frame the Fourier modes carry nonnegative longitudinal momentum. Using (2.4.9) we get the anticommutation relations

$$\{a^{i}(k), a^{\dagger}_{j}(k')\} = \delta^{i}_{j}\delta(k-k'), \qquad \{b^{i}(k), b^{\dagger}_{j}(k')\} = \delta^{i}_{j}\delta(k-k'), \qquad (2.4.11)$$

with $a_i^{\dagger}(k) = b_i^{\dagger}(k)$ if the quark field is a Majorana fermion. These operators define the Fock vacuum, and in fact the nonperturbative vacuum $|0\rangle$ of lightcone QCD

$$a^{i}(k)|0\rangle = 0, \qquad b^{i}(k)|0\rangle = 0 \qquad \forall k.$$
 (2.4.12)

Normal ordering in (2.4.8) implies that the vacuum has zero lightcone energy and momentum

$$H|0\rangle = 0, \qquad P|0\rangle = 0.$$
 (2.4.13)

The goal is to diagonalize the lightcone Hamiltonian(s) H on the Hilbert space \mathcal{H} created by the quarks

$$|\Psi^{i_1 i_2 \dots i_L}\rangle \equiv a_{i_1}^{\dagger}(k_1) a_{i_2}^{\dagger}(k_2) \dots a_{i_L}^{\dagger}(k_L) |0\rangle .$$
(2.4.14)

The physical states of QCD must be gauge invariant, which implies that all physical states must be invariant under the action of \mathfrak{g}

$$\int dx \, J^a(x) |\Psi^{i_1 i_2 \dots i_L}\rangle = 0. \qquad (2.4.15)$$

While H mixes states in the Hilbert space \mathcal{H} , the longitudinal momentum operator P is diagonal, it is the sum of the longitudinal momentum of each parton.

QCD has massless particles if and only if there exist states $|\Psi^{i_1i_2...i_L}\rangle \in \mathcal{H}$, other than the vacuum state, that are gauge invariant and have zero lightcone energy,

$$\int \mathrm{d}x : J^a \frac{1}{\partial^2} J^a : |\Psi^{i_1 i_2 \dots i_L}\rangle = 0.$$
(2.4.16)

The currents $J^a = \frac{1}{2}: \psi^{\dagger} t^a \psi$: constructed out of the chiral quarks ψ transforming in a representation R of \mathfrak{g} generate an affine chiral current algebra $\mathfrak{g}_{I(R)}$

$$J^{a}(x)J^{b}(0) \sim \frac{I(R)\delta^{ab}}{x^{2}} + \frac{if^{ab}{}_{c}J^{c}(0)}{x}, \qquad (2.4.17)$$

where the level is the Dynkin index I(R) of the representation R. The OPE (2.4.17) implies, upon putting the longitudinal coordinate on the circle, that the Fourier modes obey

$$[J_n^a, J_m^b] = i f^{ab}{}_c J^c_{n+m} + I(R) \delta^{ab} n \delta_{m+n,0} \,.$$
(2.4.18)

The zero energy state condition (2.4.16) takes the form

$$\sum_{n=1}^{\infty} \frac{1}{n^2} J^a_{-n} J^a_n |\Psi^{i_1 i_2 \dots i_L}\rangle = 0.$$
(2.4.19)

Since the Hamiltonian H in (2.4.8) is a positive semidefinite operator, the necessary and sufficient conditions for a state $|\Psi^{i_1i_2...i_L}\rangle$ to have zero energy are:
- 1. $|\Psi^{i_1 i_2 \dots i_L}\rangle$ is a primary state of the current algebra $\mathfrak{g}_{I(R)}$, that is $J_n^a |\Psi^{i_1 i_2 \dots i_L}\rangle = 0$ $\forall n \geq 1$.
- 2. $|\Psi^{i_1 i_2 \dots i_L}\rangle$ transforms in the trivial representation of \mathfrak{g} .

We now proceed to study under what conditions these states exist.

The quark Hilbert space \mathcal{H} decomposes into modules of the $\mathfrak{so}(\dim(R))_1$ current algebra at level one [14] (see also [148]), labeled by representations of $\mathfrak{so}(\dim(R))$. These are labeled by $(\mathbf{0}, \mathbf{v}, \mathbf{s}, \mathbf{c})$ when $\dim(R)$ is even and by $(\mathbf{0}, \mathbf{v}, \mathbf{s})$ when it is odd, where $\mathbf{0}, \mathbf{v}, \mathbf{s}, \mathbf{c}$ are the trivial, the vector and spinor representation(s) of $\mathfrak{so}(\dim(R))$. The precise relation between the modules of $\mathfrak{so}(\dim(R))_1$ current algebra and fermion Hilbert space is (e.g. for $\dim(R)$ even)

$$\begin{aligned}
\mathcal{H}_{\rm NS} &= \mathcal{H}_{\mathbf{0}} \oplus \mathcal{H}_{\mathbf{v}} \\
\mathcal{H}_{\rm R} &= \mathcal{H}_{\mathbf{s}} \oplus \mathcal{H}_{\mathbf{c}},
\end{aligned}$$
(2.4.20)

where \mathcal{H}_X denotes the fermion Hilbert space with fermions obeying $X \in \{NS, R\}$ (Neveu-Schwarz and Ramond) boundary conditions on the circle.

Since QCD is obtained by gauging the subalgebra $\mathfrak{g} \subset \mathfrak{so}(\dim(R))$, the current algebra embeds as $\mathfrak{g}_{I(R)} \subset \mathfrak{so}(\dim(R))_1$ into the fermion current algebra. That means that any state in \mathcal{H} fits inside a module of the $\mathfrak{g}_{I(R)}$ current algebra. An elegant way to describe how states embed is through the branching functions $b_{\Lambda\lambda}$, which encode how the characters of the $\mathfrak{so}(\dim(R))_1$ current algebra decompose into $\mathfrak{g}_{I(R)}$ characters:

$$\chi_{\Lambda}(q) = \sum_{\lambda} b_{\Lambda\lambda}(q) \chi_{\lambda}(q) \,. \tag{2.4.21}$$

Here $\Lambda \in \{0, \mathbf{v}, \mathbf{s}, \mathbf{c}\}$ or $\{0, \mathbf{v}, \mathbf{s}\}$ and λ is a highest weight vector labeling the integrable representations of $\mathfrak{g}_{I(R)}$, which obeys

$$\sum_{i=1}^{\operatorname{rank}(\mathfrak{g})} a_i^{\vee} \lambda_i \le I(R) \,, \tag{2.4.22}$$

where a_i^{\vee} are the comarks of \mathfrak{g} . The function $b_{\Lambda\lambda}(q)$ counts how many primary states of $\mathfrak{g}_{I(R)}$ with highest weight λ appear in the decomposition of the module of $\mathfrak{so}(\dim(R))_1$ with highest weight Λ . They also capture at which level the primaries of $\mathfrak{g}_{I(R)}$ appear in the $\mathfrak{so}(\dim(R))_1$ modules.

The branching function $b_{\Lambda\lambda}(q)$ has a module interpretation. $b_{\Lambda\lambda}(q)$ is a character of the chiral algebra \mathcal{A} , where \mathcal{A} is the commutant chiral algebra of $\mathfrak{g}_{I(R)}$ inside $\mathfrak{so}(\dim(R))_1$. \mathcal{A} necessarily contains the Virasoro algebra with central charge the difference of the central charges of the two current algebras $c_{\mathcal{A}} = c(\mathfrak{so}(\dim R)_1) - c(\mathfrak{g}_{I(R)}) = \frac{1}{2} \dim R - \frac{I(R)\dim(\mathfrak{g})}{I(R)+h}$, where h is the dual Coxeter number of \mathfrak{g} (see table 2.3). Also, if \mathfrak{h} is the commutant of \mathfrak{g} inside

 $\mathfrak{so}(\dim R)$, then \mathcal{A} contains a current algebra \mathfrak{h}_k . This current algebra has the interpretation as the flavor symmetry current algebra in QCD.

We have established that a QCD theory is gapless if and only if the fermion Hilbert space contains a nontrivial primary state of $\mathfrak{g}_{I(R)}$ labeled by the trivial representation of \mathfrak{g} . We are therefore interested in the functions $b_{\Lambda\hat{0}}(q)$, where we denote the trivial representation of \mathfrak{g} by $\hat{\mathbf{0}}$. In order to determine whether a QCD theory has a massless state it will suffice to look at the branching function for the integrable representation of $\mathfrak{so}(\dim(R))_1$ labeled by the trivial representation of $\mathfrak{so}(\dim(R))$ into the trivial representation of $\mathfrak{g}_{I(R)}$. Whether the theory has massless states or not is encoded in the properties of the function $b_{0\hat{0}}(q)$.

Given that $b_{00}(q)$ is the vacuum character of the chiral algebra \mathcal{A} , this branching function takes the following general form⁵⁴

$$b_{\mathbf{0}\hat{\mathbf{0}}}(q) = q^{-c_{\mathcal{A}}/24} \left(1 + a_1 q + a_2 q^2 + \dots \right).$$
(2.4.23)

The theory has massless particles if $a_l \neq 0$ for any l.

We explain now the physical meaning of the coefficients a_l . The coefficient of the q^0 term being $a_0 = 1$ corresponds to the vacuum state $|0\rangle$, which is unique. The coefficient $a_1 = \dim(\mathfrak{h})$ is the dimension of the commutant of \mathfrak{g} in $\mathfrak{so}(\dim(R))$. If \mathfrak{h} is nontrivial, we can build the primary, singlet states of $\mathfrak{g}_{I(R)}$ at level one by acting on the vacuum with the flavor symmetry currents

$$\widetilde{J}^{\alpha} = \frac{1}{2} \psi^{\dagger} \widetilde{t}^{\alpha} \psi, \qquad \alpha = 1, \dots, \dim(\mathfrak{h}).$$
(2.4.24)

The currents \tilde{J}^{α} generate an \mathfrak{h}_k current algebra, the level k being determined by the embedding $\mathfrak{so}(\dim R) \supset \mathfrak{h} \oplus \mathfrak{g}$. Indeed, these states are annihilated by J^a since $[\tilde{J}^{\alpha}, J^a] = 0$, by virtue of \mathfrak{h} commuting with \mathfrak{g} . If such an operator \tilde{J}^{α} exists, then the theory is gapless. This reproduces the result we proved in section 2.3 stating that any theory with a continuous, chiral global symmetry — which means \mathfrak{h} is nontrivial — is necessarily gapless. Therefore, a necessary condition for the QCD theory to be gapped is that the theory has no continuous, chiral flavor symmetries and therefore that $a_1 = 0$.

We turn our attention to the physics of the coefficient a_2 . Recall that given a current algebra $\mathfrak{g}_{I(R)}$, one can construct the chiral energy momentum tensor [149, 150]

$$T_{\mathfrak{g}_{I(R)}} = \frac{1}{8(I(R)+h)} : J^a J^a : .$$
(2.4.25)

The operator $T_{\mathfrak{g}_{I(R)}}$ generates a Virasoro algebra of central charge $c(\mathfrak{g}_{I(R)}) = \frac{I(R) \dim(G)}{I(R)+h}$. The canonical level 2 state in the Hilbert space \mathcal{H}

$$(T_{\mathfrak{so}(\dim(R))_1} - T_{\mathfrak{g}_{I(R)}})|0\rangle \tag{2.4.26}$$

⁵⁴There are no $q^{n+1/2}$ terms in the expansion since $\mathcal{H}_{\mathbf{0}}$ contains states created with an even number of fermions.

is a singlet, primary state of the $\mathfrak{g}_{I(R)}$ current algebra, where $T_{\mathfrak{so}(\dim(R))_1}$ is the energy momentum tensor of the $\mathfrak{so}(\dim(R))_1$ current algebra generated by the quarks in the ultraviolet. This is a consequence of the OPEs

$$T_{\mathfrak{g}_{I}(R)}(x)J^{a}(0) \sim \frac{J^{a}(0)}{x^{2}} + \frac{\partial J^{a}(0)}{x}$$

$$T_{\mathfrak{so}(\dim(R))_{1}}(x)J^{a}(0) \sim \frac{J^{a}(0)}{x^{2}} + \frac{\partial J^{a}(0)}{x},$$

(2.4.27)

so that

$$[J^a, T_{\mathfrak{so}(\dim(R))_1} - T_{\mathfrak{g}_{I(R)}}] = 0.$$
(2.4.28)

This proves that the state (2.4.26) is a gauge invariant, primary state of $\mathfrak{g}_{I(R)}$. Therefore as long as $T_{\mathfrak{so}(\dim(R))_1} - T_{\mathfrak{g}_{I(R)}} \neq 0$, so that $a_2 \neq 0$, the theory has massless states and the spectrum is gapless.⁵⁵

This implies that a necessary (and as we will show also sufficient) condition for a QCD theory to be gapped is that the following operator equation holds

$$T_{\mathfrak{so}(\dim(R))_1} - T_{\mathfrak{g}_{I(R)}} = 0.$$
 (2.4.30)

This equation is very constraining, it implies that $a_l = 0$ for all l. Indeed, when (2.4.30) is obeyed then $b_{\Lambda\lambda}(q)$ is as a character of the Virasoro algebra with $c_A = 0$, which has a unique, trivial unitary representation. Therefore, when equation (2.4.30) holds then⁵⁶

$$b_{\Lambda \hat{\mathbf{0}}}(q) = \delta_{\Lambda \mathbf{0}} \,, \tag{2.4.31}$$

and the only singlet primary state of $\mathfrak{g}_{I(R)}$ is the vacuum state: the theory has no massless particles.

By demanding that there are no left-moving or right-moving massless particles, we arrive at the following proposition:

Proposition 2.4.1 *QCD theory* $(G; R_{\ell}, R_r)$ *is gapped if and only if both operator equations hold*

$$T_{\mathfrak{so}(\dim R_{\ell})_{1}} - T_{\mathfrak{g}_{I(R_{\ell})}} = 0,$$

$$\overline{T}_{\mathfrak{so}(\dim R_{r})_{1}} - \overline{T}_{\mathfrak{g}_{I(R_{r})}} = 0.$$
(2.4.32)

 55 In the presence of a global symmetry \mathfrak{h} there are additional level 2 primary, singlet states of $\mathfrak{g}_{I(R)}$

$$\tilde{J}^{\alpha}_{-2}|0\rangle, \qquad \tilde{J}^{\alpha}_{-1}\tilde{J}^{\beta}_{-1}|0\rangle.$$
(2.4.29)

Generically, there are $\dim(\mathfrak{h})(\dim(\mathfrak{h})+3)/2$ such states constructed using the flavor affine algebra \mathfrak{h}_k currents (for very small values of k one may need to subtract some null states).

⁵⁶This follows by comparing the scaling dimensions of operators.

This was derived by looking at the lightcone Hamiltonians with x^+ and x^- being time.

We will come back to these equations momentarily. Before we do that we shall also derive these equations in a different quantization scheme in order to gain more insight into the mechanism behind the gap.

2.4.2 Temporal gauge Hamiltonian

In this section we rederive the necessary and sufficient conditions (2.4.32) for a QCD theory $(G; R_{\ell}, R_r)$ to be gapped by studying the canonical Hamiltonian where time is the time-like coordinate $x^0 = t$. The lightcone coordinates are well-suited to algebraic considerations because the two chiralities are mostly decoupled. By contrast, a time-like coordinate requires more work but it also leads to a more transparent understanding of the spectrum, because the Hamiltonian takes the traditional form, which evolves states in physical time. For previous work on the temporal Hamiltonian of QCD see [19, 151, 152].

We follow the same conventions as in $\S2.2$, which we present here for convenience

$$x^{\pm} = \frac{1}{\sqrt{2}} (x^0 \pm x^1), \quad A^a_{\pm} = \frac{1}{\sqrt{2}} (A^a_0 \pm A^a_1).$$
 (2.4.33)

We start with the Lagrangian

$$\mathcal{L} = i\psi_{i\ell}^{\dagger}\partial_{-}\psi_{i\ell} + i\psi_{ir}^{\dagger}\partial_{+}\psi_{ir} + A_{-}^{a}J_{\ell}^{a} + A_{+}^{a}J_{r}^{a} - \frac{1}{2g^{2}}\operatorname{tr}(F_{\mu\nu}F^{\mu\nu}), \qquad (2.4.34)$$

where $J^a_{\ell,r} = \frac{1}{2} \psi^{\dagger}_{i\,\ell,r} t^a_{ij} \psi_{j\,\ell,r}$. Canonical quantization leads to the following equal-time commutation relations

$$\{\psi_{i\ell}^{\dagger}(x),\psi_{j\ell}(y)\} = \delta_{ij}\delta(x-y),$$

$$\{\psi_{ir}^{\dagger}(x),\psi_{jr}(y)\} = \delta_{ij}\delta(x-y),$$

$$[\Pi^{a}(x),A_{1}^{b}(y)] = i\delta^{ab}\delta(x-y),$$

(2.4.35)

where $\Pi^a = \frac{\delta \mathcal{L}}{\delta A_0^a} = \frac{1}{g^2} F_{10}^a = \frac{1}{g^2} E^a(x)$. Consequently, classically the currents obey the commutation relations

$$[J^{a}_{\ell,r}(x), J^{b}_{\ell,r}(y)] = i f^{abc} J^{c}_{\ell,r}(x) \delta(x-y) .$$
(2.4.36)

The Hamiltonian density is

$$\mathcal{H} = g^{2} E^{a}(x)^{2} + A_{0}^{a} G^{a} + \frac{1}{\sqrt{2}} A_{1}^{a}(x) (J_{\ell}^{a} - J_{r}^{a})(x) + \frac{1}{\sqrt{2}} i \psi_{i\ell}^{\dagger} \partial_{x} \psi_{i\ell}(x) - \frac{1}{\sqrt{2}} i \psi_{ir}^{\dagger} \partial_{x} \psi_{ir}(x) , \qquad (2.4.37)$$

and

$$G^{a}(x) = J^{a}_{\ell}(x) + J^{a}_{r}(x) - (\partial_{x}\Pi^{a} + if^{abc}A^{b}_{1}\Pi^{c})(x)$$
(2.4.38)

is the Gauss' law operator, which obeys the following commutation relation with the currents

$$[G^{a}(x), J^{b}_{\ell,r}(y)] = i f^{abc} J^{c}_{\ell,r}(x) \delta(x-y) .$$
(2.4.39)

The Gauss law operator commutes with the Hamiltonian, that is $[G^a(x), H] = 0$, where $H = \int dx \mathscr{H}$.

The phase space of QCD has a primary constraint, namely the momentum conjugate to A_0 vanishes

$$\Pi^a = \frac{\delta \mathcal{L}}{\delta \dot{A}_0^a} = 0. \qquad (2.4.40)$$

Demanding stability of this constraint leads to the secondary constraint

$$\frac{d\Pi^a}{dt} = [H, \Pi^a] = G^a = 0, \qquad (2.4.41)$$

Since $\frac{dG^a}{dt} = [H, G^a] = 0$ on the constraint surface, there are no further constraints. Hamilton's equations derived from (2.4.37) reproduce the equations of motion obtained by varying the Lagrangian (2.4.34). We work in the gauge with $A_0^a = 0$ so that $A_+^a = A_1^a \equiv A^a$.

In the quantum theory, fields are promoted to operators and composites need to be renormalized due to quantum fluctuations at arbitrary short distances. The quark currents in the quantum theory must be normal ordered $\hat{J}^a_{\ell,r} = \frac{1}{2}:\psi^{\dagger}_{i\,\ell,r}t^a_{ij}\psi_{j\,\ell,r}:$. In the quantum theory, the commutation relations become

$$\begin{aligned} \left[\widehat{J}^{a}_{\ell,r}(x), \widehat{J}^{b}_{\ell,r}(y)\right] &= i f^{abc} J^{c}_{\ell,r}(x) \delta(x-y) \pm i I(R) \delta^{ab} \partial_x \delta(x-y) ,\\ \left[\widehat{\Pi}^{a}(x), \widehat{A}^{b}(y)\right] &= i \delta^{ab} \delta(x-y) , \end{aligned}$$
(2.4.42)

where $I(R) \equiv I(R_{\ell}) = I(R_r)$ by virtue of gauge anomaly cancellation, and \pm corresponds to ℓ and r respectively. The operators $\hat{J}^a_{\ell,r}$ generate the current algebra $\mathfrak{g}_{I(R_{\ell,r})}$. Quantization leads to the Schwinger term in the current commutators (2.4.42) (cf. with (2.4.36)), which will have important implications.

In order to determine the conditions for the spectrum of Hamiltonian to be gapped, we first define the fermion operators in (2.4.37) in normal ordered form

$$\frac{1}{\sqrt{2}}i\psi_{i\ell}^{\dagger}\partial_{x}\psi_{i\ell}(x) = \frac{1}{\sqrt{2}}:i\psi_{i\ell}^{\dagger}\partial_{x}\psi_{i\ell}(x): -\frac{1}{\sqrt{2}}i\lim_{\epsilon\to 0}\langle\psi_{i\ell}^{\dagger}(x+\epsilon)\partial_{x}\psi_{i\ell}(x-\epsilon)\rangle
-\frac{1}{\sqrt{2}}i\psi_{ir}^{\dagger}\partial_{x}\psi_{ir}(x) = -\frac{1}{\sqrt{2}}:i\psi_{ir}^{\dagger}\partial_{x}\psi_{ir}(x): +\frac{1}{\sqrt{2}}i\lim_{\epsilon\to 0}\langle\psi_{ir}^{\dagger}(x+\epsilon)\partial_{x}\psi_{ir}(x-\epsilon)\rangle.$$
(2.4.43)

where $\langle \psi_{i\ell,r}^{\dagger}(x)\psi_{j\ell,r}(y)\rangle \sim \frac{i\delta_{ij}}{x-y}$. We then express $\pm \frac{1}{\sqrt{2}}:i\psi_{\ell,r}^{\dagger}\partial_x\psi_{\ell,r}$: in terms of the Sugawara tensor for $\mathfrak{so}(\dim R_{\ell,r})_1$ via

$$\frac{1}{2}T_{\mathfrak{so}(\dim R_{\ell})_{1}} = \frac{i}{8(I(R)+h)}:\widehat{J}^{a}_{\ell}\widehat{J}^{a}_{\ell}(x): = \frac{1}{\sqrt{2}}:i\psi^{\dagger}_{i\ell}\partial_{x}\psi_{i\ell}(x):$$

$$\frac{1}{2}\overline{T}_{\mathfrak{so}(\dim R_{r})_{1}} = \frac{i}{8(I(R)+h)}:\widehat{J}^{a}_{r}\widehat{J}^{a}_{r}(x): = -\frac{1}{\sqrt{2}}:i\psi^{\dagger}_{ir}\partial_{x}\psi_{ir}(x):.$$
(2.4.44)

The crucial insight (see also [19, 152]) is that we can split the energy momentum tensor into a piece that couples to the gauge fields and a piece that is decoupled

$$T_{\mathfrak{so}(\dim R_{\ell})_{1}} = T_{\mathfrak{g}_{I(R_{\ell})}} + \left(T_{\mathfrak{so}(\dim R_{\ell})_{1}} - T_{\mathfrak{g}_{I(R_{\ell})}}\right)$$

$$\overline{T}_{\mathfrak{so}(\dim R_{r})_{1}} = \overline{T}_{\mathfrak{g}_{I(R_{r})}} + \left(\overline{T}_{\mathfrak{so}(\dim R_{r})_{1}} - \overline{T}_{\mathfrak{g}_{I(R_{r})}}\right).$$

$$(2.4.45)$$

The quantized Hamiltonian must commute with the quantum Gauss' law operator

$$\widehat{G}^a(x) = \widehat{J}^a_\ell(x) + \widehat{J}^a_r(x) - (\partial_x \widehat{\Pi}^a + i f^{abc} \widehat{A}^b \,\widehat{\Pi}^c)(x) \,. \tag{2.4.46}$$

It is given by (see section 2.6.2 for details)

$$\widehat{\mathscr{H}} = \frac{1}{2} \left(T_{\mathfrak{g}_{I(R_{\ell})}} + T_{\mathfrak{g}_{I(R_{r})}} \right) + \frac{1}{\sqrt{2}} \widehat{A}^{a}(x) (\widehat{J}^{a}_{\ell} - \widehat{J}^{a}_{r})(x) + \frac{1}{\sqrt{2}} I(R) \widehat{A}^{a}(x)^{2} + g^{2} \widehat{E}^{a}(x)^{2} + \frac{1}{\sqrt{2}} i \lim_{\epsilon \to 0} \left(\langle \psi^{\dagger}_{i\ell}(x+\epsilon) \partial_{x} \psi_{i\ell}(x-\epsilon) \rangle - \langle \psi^{\dagger}_{ir}(x+\epsilon) \partial_{x} \psi_{ir}(x-\epsilon) \rangle \right) + \frac{1}{2} \left(T_{\mathfrak{so}(\dim R_{\ell})_{1}} - T_{\mathfrak{g}_{I(R_{\ell})}} \right) + \frac{1}{2} \left(\overline{T}_{\mathfrak{so}(\dim R_{r})_{1}} - \overline{T}_{\mathfrak{g}_{I(R_{r})}} \right).$$

$$(2.4.47)$$

Let us discuss some of the most salient features of this Hamiltonian. Expressing the Hamiltonian in terms of the energy momentum tensor and splitting it as in (2.4.45) shows that there is a decoupled CFT with energy momentum tensors $T_{\mathfrak{so}(\dim R_\ell)_1} - T_{\mathfrak{g}_{I(R_\ell)}}$ and $\overline{T}_{\mathfrak{so}(\dim R_r)_1} - \overline{T}_{\mathfrak{g}_{I(R_r)}}$. The last line in (2.4.47) describes a gapless sector. The term $I(R)\widehat{A}^a(x)^2$ in (2.4.47), which is present due to the Schwinger term in (2.4.42), gaps out the gauge fields, and strongly suggests that the first two lines in (2.4.47) describe a gapped Hamiltonian.⁵⁷ This analysis makes manifest that the massless degrees of freedom decouple in the ultraviolet and go along for the ride during the renormalization group flow (see [19]). In conclusion, the QCD theory $(G; R_\ell, R_r)$ is gapped if and only if the decoupled CFT is trivial, that is if

$$T_{\mathfrak{so}(\dim R_{\ell})_{1}} - T_{\mathfrak{g}_{I(R_{\ell})}} = 0$$

$$\overline{T}_{\mathfrak{so}(\dim R_{r})_{1}} - \overline{T}_{\mathfrak{g}_{I(R_{r})}} = 0$$
(2.4.48)

We have thus recovered the conditions (2.4.32) we had derived in the previous section using the lightcone Hamiltonians.

2.4.3 Classification of gapped theories

We now return to our main task: deducing whether a given QCD theory is gapped or not. We showed, both looking at lightcone quantization and standard canonical quantization, that a theory labelled by $(G; R_{\ell}, R_r)$ is gapped if and only if the two operator equations hold:

$$T_{\mathfrak{so}(\dim R_{\ell})_{1}} - T_{\mathfrak{g}_{I(R_{\ell})}} = 0,$$

$$\overline{T}_{\mathfrak{so}(\dim R_{r})_{1}} - \overline{T}_{\mathfrak{g}_{I(R_{r})}} = 0.$$
(2.4.49)

 $^{^{57}}$ It would be interesting to give a rigorous proof that it is gapped.

The key point is that the equations (2.4.49) can in fact be solved. It suffices to consider one chirality first, and then combine solutions that merge left and right chiral sectors. In [153, 154] it was shown that the energy-momentum tensor of the affine algebra $\mathfrak{g}_{I(R)}$ coincides with that of a free fermion theory $\mathfrak{so}(\dim(R))_1$ if and only if the matrices t_{ij}^a that generate the representation R satisfy the Jacobi-like identity (see section 2.6.1 for derivation)

$$t^{a}_{ij}t^{a}_{k\ell} + t^{a}_{ik}t^{a}_{\ell j} + t^{a}_{i\ell}t^{a}_{jk} = 0. aga{2.4.50}$$

In turn, this identity is satisfied if and only if there exists some Lie algebra $\hat{\mathfrak{g}}$ that contains \mathfrak{g} such that the homogeneous space⁵⁸ \hat{G}/G is symmetric, and the fermions transform with respect to \mathfrak{g} in the same way as the generators of \hat{G}/G . The condition $t_{ij}^a t_{k\ell}^a + t_{ik}^a t_{\ell j}^a + t_{i\ell}^a t_{jk}^a = 0$ is nothing but the Bianchi identity for the Riemann tensor of \hat{G}/G . The symmetric spaces have been fully classified [155], and we can read off the list of gapped QCD theories from this classification: for each symmetric space of the form \hat{G}/G where the symmetric space generators transform according to a representation R of G, there is a gapped QCD theory with gauge group G and quarks in the representation R, and vice versa.

A different perspective yields the same answer. The equality of the energy-momentum tensors of $\mathfrak{so}(\dim(R))_1$ and $\mathfrak{g}_{I(R)}$ implies, by definition, that the affine algebra $\mathfrak{g}_{I(R)}$ embeds conformally into the affine algebra $\mathfrak{so}(\dim(R))_1$. The conformal embeddings have been fully classified [156–158], and we can read off the list of gapped QCD theories from this classification: for each conformal embedding of an algebra \mathfrak{g}_k into $\mathfrak{so}(n)_1$ via a representation R of \mathfrak{g} , there is a gapped QCD theory with gauge group G and quarks in the representation R, and vice versa.

Either point of view yields table 2.4. This table contains the list of "minimal" gapped QCD theories. Naturally, one can also take the tensor product of two gapped theories to obtain another gapped theory. This operation corresponds to reducible symmetric spaces, or non-maximal conformal embeddings; the most general symmetric space is a product of irreducible ones, and the most general conformal embedding is a sequence of maximal ones.

In QCD, this stacking operation gives rise to theories with decoupled gapped sectors so they are also gapped in a trivial way – and so they are of little interest by themselves. The exception is when the different theories contain abelian $\mathfrak{u}(1)$ factors in their gauge group, in which case we can couple the minimal theories through these factors, which generates another gapped theory which is *not* just the product of decoupled theories. With this in mind, the most general gapped QCD theory is either a theory in table 2.4, or a product of such theories provided they contain a $\mathfrak{u}(1)$ gauge group, in which case the matrix of charges for these $\mathfrak{u}(1)$ factors must be non-singular. Any other gapped theory is a trivial product of these two options.

⁵⁸The global structure of \hat{G}, G is arbitrary, the condition $t^a_{ij}t^a_{k\ell} + t^a_{ik}t^a_{\ell j} + t^a_{i\ell}t^a_{jk} = 0$ is purely algebraic and insensitive to the choice of \hat{G}, G for a given $\hat{\mathfrak{g}}, \mathfrak{g}$. At the algebraic level, the condition can be recast as the existence of an algebra $\hat{\mathfrak{g}} = \mathfrak{g} + \mathfrak{p}$ such that $[p^i, p^j] \in \mathfrak{g}$ for $p^i, p^j \in \mathfrak{p}$, and $[p^i, g^a] = t^a_{ij}p^j$ for $g^a \in \mathfrak{g}$.

g	R	IR TQFT	$\hat{\mathfrak{g}}$
g	Adj	$\operatorname{SO}(\dim G)_1/G_h$	$\mathfrak{g}+\mathfrak{g}$
$\mathfrak{so}(N)$		$SO(\frac{1}{2}(N+2)(N-1))_1/Spin(N)_{N+2}$	$\mathfrak{su}(N)$
$\mathfrak{sp}(N)$	H	$SO((2N+1)(N-1))_1/Sp(N)_{N-1}$	$\mathfrak{su}(2N)$
$\mathfrak{s}(\mathfrak{u}(M) + \mathfrak{u}(N))$	$(\Box,\Box)_q$	$\mathrm{U}(MN)_1/S(\mathrm{U}(M)_N \times \mathrm{U}(N)_M)$	$\mathfrak{su}(M+N)$
$\mathfrak{so}(M) + \mathfrak{so}(N)$	(\Box,\Box)	$SO(MN)_1/(Spin(M)_N \times Spin(N)_M)$	$\mathfrak{so}(M+N)$
$\mathfrak{u}(N)$	\square_q	$U(\frac{1}{2}N(N-1))_1/U(N)_{N-2,2(N-1)q^2}$	$\mathfrak{so}(2N)$
$\mathfrak{u}(N)$	\square_q	$U(\frac{1}{2}N(N+1))_1/U(N)_{N+2,2(N+1)q^2}$	$\mathfrak{sp}(N)$
$\mathfrak{sp}(M) + \mathfrak{sp}(N)$	(\Box,\Box)	$\mathrm{SO}(4MN)_1/(\mathrm{Sp}(M)_N \times \mathrm{Sp}(N)_M)$	$\mathfrak{sp}(M+N)$
$\mathfrak{sp}(4)$	42	$SO(42)_1/Sp(4)_7$	E_6
$\mathfrak{su}(2) + \mathfrak{su}(6)$	$({f 2},{f 20})$	$SO(40)_1/(SU(2)_{10} \times SU(6)_6)$	E_6
$\mathfrak{so}(10) + \mathfrak{u}(1)$	16_{q}	$U(16)_1/(Spin(10)_4 \times U(1)_{16q^2})$	E_6
F_4	26	$SO(26)_1/F_{4,3}$	E_6
$\mathfrak{su}(8)$	70	$SO(70)_1/SU(8)_{10}$	E_7
$\mathfrak{su}(2) + \mathfrak{so}(12)$	(2 , 32)	$SO(64)_1/(SU(2)_{16} \times Spin(12)_8)$	E_7
$E_6 + \mathfrak{u}(1)$	27_{q}	$U(27)_1/(E_{6,6} \times U(1)_{27q^2})$	E_7
$\mathfrak{so}(16)$	128	$SO(128)_1/Spin(16)_{16}$	E_8
$\mathfrak{su}(2) + E_7$	(2 , 56)	$SO(112)_1/(SU(2)_{28} \times E_{7,12})$	E_8
$\mathfrak{su}(2) + \mathfrak{sp}(3)$	(2 , 14)	$\mathrm{SO}(28)_1/(\mathrm{SU}(2)_7 imes \mathrm{Sp}(3)_5)$	F_4
$\mathfrak{so}(9)$	16	$SO(16)_1/Spin(9)_2$	F_4
$\mathfrak{so}(4)$	(2 , 4)	$SO(8)_1/Spin(4)_{10,2}$	G_2

Table 2.4: List of irreducible gapped theories. The **first column** denotes the gauge algebra. Any global choice of G for a given \mathfrak{g} leads to a gapped theory. The **second column** denotes the representation of the quarks, either in the Young diagram notation or directly in terms of its dimension. \Box, \Box, \Box denote the fundamental, symmetric, and anti-symmetric representations, respectively (with appropriate reality conditions, e.g. Majorana if real, and with traces removed, if possible). q denotes the charge of the fermions under $\mathfrak{u}(1)$ factors, if any; this charge can be chosen arbitrarily. The **third column** denotes the TQFT that describes the space of vacua of these gapped theories (see section 2.5); here we choose the simply-connected form $G_{\rm sc}$ for concreteness. The **fourth column** denotes the Lie algebra $\hat{\mathfrak{g}}$ that makes \hat{G}/G a symmetric spaces.

For illustration purposes, consider the gapped theory $\mathfrak{u}(N) + \Box_q$, where \Box_q denotes the fundamental representation and $q \in \mathbb{Z}$ is an arbitrary integer that specifies the charge of the quark under the trace part $\mathfrak{u}(1) \subset \mathfrak{u}(N)$. Stacking a family of these gapped theories, and

coupling the abelian factors via an arbitrary matrix of charges, one obtains the gapped theory

$$\prod_{i=1}^{m} \mathfrak{u}(N_i), \qquad R = \bigoplus_{i=1}^{m} (\mathbf{1}, \dots, \mathbf{1}, \square, \mathbf{1}, \dots, \mathbf{1})_{\vec{q}_i}, \qquad (2.4.51)$$

where $\vec{q_i}$ is a vector of charges that specifies how the *i*-th fermion couples to $\mathfrak{u}(1)^m$. The special case where $N_i = 1$ for all *i* corresponds to abelian QCD, i.e., QED with *m* photons and *m* fermions, which we analyze in more detail in section 2.8.

The theories in table 2.4 are written in terms of non-chiral data. As a matter of fact, one can also modify these theories to obtain gapped theories that are chiral. The idea is that, if the theory with vector-like matter $(R_{\ell}, R_r) = (R, R)$ is gapped, then the theory with chiral matter $(R_{\ell}, R_r) = (\sigma_{\ell} \cdot R, \sigma_r \cdot R)$ is also gapped, where σ_{ℓ}, σ_r denote outer automorphisms of \mathfrak{g} (see table 2.3). For example, for simply-laced groups $\sigma \cdot R$ may denote the conjugate representation \overline{R} , while for $\mathfrak{g} = \mathfrak{so}(8), \sigma \cdot R$ may denote any representation related by triality. As $\dim(\sigma \cdot R) = \dim(R)$ and $I(\sigma \cdot R) = I(R)$, the chiral theory with $(R_{\ell}, R_r) = (\sigma_{\ell} \cdot R, \sigma_r \cdot R)$ is also gapped. In the case of theories that contain abelian gauge groups, the statement becomes that one can use different charges for the two chiralities, (q_{ℓ}, q_r) , provided they satisfy the gauge anomaly cancellation condition (2.2.17).

For example, given the gapped theory in (2.4.51), one can generate other gapped theories by replacing some of the fundamentals by anti-fundamentals (for one chirality only, or for both), and also by assigning generically different U(1) charges to the two chiralities $\vec{q_i} \mapsto (\vec{q_{\ell,i}}, \vec{q_{r,i}})$.

Table 2.4, together with the two operations we just described (stacking gapped theories and coupling them together via their abelian factors, and acting with outer automorphisms on the representations), give the extensive list of gapped QCD theories. See tables 2.1 and 2.2 for summary of gapped vector-like and chiral theories respectively. Any other theory is either a trivial product of gapped theories, or is gapless.

2.5 Infrared dynamics of 2d QCD

Having classified all QCD theories that are gapped, and consequently, those that are gapless, it remains an interesting open question to determine the effective field theory describing the low energy dynamics. The most natural proposal is that the low energy theory is a gauged WZW coset model with chiral algebra (see [19] and more recently [136, 159])

$$\frac{\operatorname{SO}(\dim R_{\ell})_{1}}{G_{I(R_{\ell})}} \times \frac{\operatorname{SO}(\dim R_{r})_{1}}{G_{I(R_{r})}} \,. \tag{2.5.1}$$

In order to simplify notation we focus on the chiral half, with the understanding that the full theory is constructed by putting together the left and right sectors. The idea behind (2.5.1) is that QCD can be thought of as $\dim(R)$ free fermions, which can be described as the fermionic WZW theory $\operatorname{SO}(\dim(R))_1$, where a symmetry $G \subset \operatorname{SO}(\dim(R))$ has been gauged, and one has added a kinetic term for the gluons. The coupling constant gis dimensionful so it grows in the infrared and it is self-consistent to assume that $g \to \infty$ as $E \to 0$, which means that we can drop the gluon kinetic term for very low energies. All in all, it is expected that the deep infrared of QCD theories is described by the CFT coset (2.5.1), namely an $\operatorname{SO}(\dim(R))_1$ WZW model with gauged $G_{I(R)}$ symmetry. The level of the gauge current algebra is determined by the Dynkin embedding index of $\operatorname{SO}(\dim(R)) \supset G$; this embedding is defined by the branching rule $\Box \mapsto R$, and hence the embedding index is I(R).

This proposal is also suggested by our canonical analysis of section 2.4, where we highlighted the presence of $\mathfrak{so}(\dim(R))_1, \mathfrak{g}_{I(R)}$ current algebras in the Hamiltonian of QCD, and the fact that the operator $T_{\mathfrak{so}(\dim(R))_1} - T_{\mathfrak{g}_{I(R)}}$ naturally appears in this Hamiltonian, playing the role of the energy-momentum of a low-energy CFT that is decoupled from massive modes, which disappear in the deep infrared.

Gapped spectrum. One nice aspect of (2.5.1) is that it is perfectly consistent with our classification from the previous section, because the coset (2.5.1) is a full-fledged CFT if its central charge is non-zero, but describes a TQFT when its central charge vanishes. In other words, the chiral energy-momentum tensor of the coset is $T_{\mathfrak{so}(\dim(R))_1/\mathfrak{g}_{I(R)}} \equiv T_{\mathfrak{so}(\dim(R))_1} - T_{\mathfrak{g}_{I(R)}}$, and this is a non-trivial operator if and only if the theory is gapless. In any case, it is important to stress that the criterion for masslessness $T_{\mathfrak{so}(\dim(R))_1/\mathfrak{g}_{I(R)}} \neq 0$ was obtained in previous sections independently of the conjecture (2.5.1), but the two are perfectly consistent with each other, a fact that gives more evidence for the latter.

Continuous symmetries. In 2*d*, continuous chiral symmetries cannot appear nor disappear along a symmetric renormalization group flow. Therefore, the effective low energy description of QCD must have the exact same such symmetries as the original ultraviolet theory. This is nicely reproduced by the coset, because the symmetries of both theories have the same definition: the flavor symmetry group is the commutant of G inside SO(dim(R)), i.e., the rotations of the chiral quarks that commute with gauge transformations.

't Hooft anomalies Another nice property of the conjecture (2.5.1) is that it automatically matches all the 't Hooft anomalies of the original QCD theory. Indeed, while the argument above does not strictly speaking prove that this coset is the low energy limit of the ultraviolet theory, it does prove that they are in the same deformation class. In other words, even though in principle the limits $E \to 0$ and $g^2 \to \infty$ need not be equivalent, it is still true that they are connected by a path in parameter space. Therefore, these two theories will carry the same 't Hooft anomalies for all the symmetries that are preserved along the path. This provides a strong consistency check on the proposal that the coset really is the low-energy limit of QCD.

The case of perturbative anomalies can be exhibited explicitly. The chiral flavor symmetry H in the ultraviolet is generated by the free fermion currents that commute with the gauge group, that is, commutant of G inside SO(dim(R)). If R is given by N_F copies of a given irreducible representation R_0 , that is $R = N_F \cdot R_0$ the flavor symmetry is $H(N_F)$, with H = O, Sp, U for real, pseudo-real, and complex representations, respectively (in the complex case, the symmetry may be either U(N_F) or SU(N_F), depending on whether the diagonal U(1) is broken by the ABJ anomaly or not, see section 2.3.1).

The 't Hooft anomaly for H is the Dynkin index of the representation under the flavor group (2.2.14), in this case the fundamental representation. This means that the flavor symmetry carries dim (R_0) units of anomaly. This is reproduced by the coset in a straightforward manner, because one can write

$$\frac{\text{SO}(\dim(R))_1}{G_{I(R)}} \equiv H_{\dim(R_0)} \times \frac{\text{SO}(\dim(R))_1}{G_{I(R)} \times H_{\dim(R_0)}}.$$
(2.5.2)

The factor $H_{\dim(R_0)}$ matches the ultraviolet 't Hooft anomaly, and the factor $\frac{\mathrm{SO}(\dim(R))_1}{G_{I(R)} \times H_{\dim(R_0)}}$ has no continuous global symmetries (no commutant). We point out that this latter coset is actually well-defined, which might not be entirely obvious. One way to see this is that one could imagine gauging the diagonal symmetry H in the ultraviolet (which is anomaly-free), to yield the gauge theory $G \times H + (R_0, \Box)$. The infrared coset for this theory is precisely $\frac{\mathrm{SO}(\dim(R))_1}{G_{I(R)} \times H_{\dim(R_0)}}$.

The case of nonperturbative global anomalies is more subtle, and requires a case-by-case analysis. That being said, the argument above proves that the coset CFT will automatically match all the anomalies, perturbative and global. This has a nice bonus consequence, namely that it predicts that many well-known CFTs actually carry nonperturbative anomalies, a fact that may not have been fully appreciated in the past. For example, below we will describe many gauge theories that flow in the infrared to common CFTs such as minimal models or WZW models. These theories necessarily carry the same nonperturbative anomalies of the ultraviolet theory, and the latter are often easy to determine (because one can flow to the deep ultraviolet, where the fermions and gluons are essentially free and semiclassical considerations often suffice). Among others, this predicts global 't Hooft anomalies for discrete symmetries such time-reversal, whose presence is seldom discussed in the CFT literature.

While there is not much one can say about global anomalies in full generality, there is one feature that is actually rather universal. There are several discrete symmetries, such as discrete chiral symmetries or antiunitary time-reversal symmetry, whose anomalies have the following effect on the Hilbert space: when the number fermions in ultraviolet is odd, the Ramond Hilbert space is automatically supersymmetric [1]. This is a nonperturbative statement that affects the whole spectrum of the theory and, in particular, the low-energy spectrum. Therefore, the effective infrared description must satisfy this property as well. This is indeed reproduced by the coset (2.5.1), because the states in the Ramond sector come from branchings from the spinor representation(s) of $SO(\dim(R))_1$; and, famously, when $\dim(R)$ is odd there is a single spinor whose Ramond-Ramond character is identically zero, a property that is inherited to the full coset. (Another diagnosis of this anomaly is that the twisted Hilbert space becomes ill-defined, which is also reproduced by the coset because the spinor character of $SO(\dim(R))_1$ carries a factor of $\sqrt{2}$ for odd $\dim(R)$, and hence the twisted partition function does not have an integral expansion; see (2.7.20) for the characters of $SO(n)_1$).

One-form symmetry. QCD theories can have one-form symmetry associated to a subgroup center of the gauge group (see table 2.3). This symmetry is discrete,⁵⁹ and hence by the generalized Coleman-Mermin-Wagner theorem [32], it cannot break spontaneously. Therefore, the infrared effective description must realize all the one-form symmetries of the ultraviolet theory.

In two dimensions, the effect of a one-form symmetry is to break up the theory into distinct sectors, or universes. The full Hilbert space of the theory is the direct sum of the Hilbert spaces of the different universes (see section 2.2). The total theory suffers from a mild violation of cluster decomposition, but the theory projected to a given universe is perfectly well-defined by itself, and satisfies decomposition.

Given a QFT with one-form symmetry, the emergent infrared CFT inherits it. Hence, in these CFTs the vacuum is not unique (the coefficient of the vacuum character in the torus partition function is an integer larger than 1). Instead, the infrared CFT is a direct sum (not a direct product) of "conventional" CFTs with a unique vacuum each.

In QCD, the one-form symmetry is the subgroup of the center that is not screened by the fermions, namely the kernel of the representation R under which the quarks transform, $\Gamma = \ker(R) \subseteq Z(G)$. This is a symmetry for all g^2 and in particular it remains a symmetry in the $g^2 \to \infty$ limit, and therefore the coset CFT also has a Γ one-form symmetry. As Γ does not act on the fermions, it does not embed into SO(dim(R)), and hence in the quotient SO(dim(R))₁/ $G_{I(R)}$ we are trying to gauge a group Γ that does not act on anything – this is an orbifold by a symmetry that does not act faithfully (cf. with [135, 160, 161]). This indeed leads to $|\Gamma|$ different universes, labelled by elements $\rho \in \Gamma^{\vee}$ (see section 2.2).

The CFT on a given universe labelled by ρ corresponds to the coset SO $(\dim(R))_1/(G/\Gamma)_{I(R)}$ with a theta term labelled by $\rho \in \Gamma^{\vee}$. The functional integral of the coset sums over G/Γ bundles, which are labeled by Γ . The sum over bundles is weighted by the theta term. It is interesting to compare this perspective with the algebraic approach to cosets in the literature

⁵⁹If the gauge group is reductive, then the one-form symmetry may include U(1) factors associated to the photons. These U(1) groups exist only when the photons are free; otherwise the screening by quarks explicitly breaks U(1) down to a discrete subgroup.

[150, 162, 163]. In the algebraic approach to cosets one organizes representations of the coset into long and short(er) orbits under the action of Γ , as Γ permutes the coset representations. When the action of Γ has only long orbits, the algebraic prescription is to divide the partition function by Γ , so that the vacuum character appears with multiplicity one, and only the trivial bundles contribute. This yields the partition function in one universe, which when there are no fixed points, is the same in all universes. When the model has shorter orbits, one has to deal with "fixed point resolution", and correct by a series of prescriptions and ansätze for the fact that characters enter with fractional multiplicity. These prescriptions have a rather clear interpretation from our perspective. When the coset has no fixed points, the CFT in each universe is the same and only trivial bundles contribute. Instead, when the coset has fixed points, the CFT in each universe is generically different. In order to identify the partition function in a given universe when there are fixed points, one must sum over nontrivial bundles, weighted by a discrete theta term, which gives a non-vanishing constant partition function [164]. These contributions combine with those of the long orbits to produce a partition function that is modular invariant in each universe. In a sense, the algebraic approach to cosets in the literature constructs the partition function in one universe, while from our perspective one can construct more modular invariant partition functions by weighing the sum over nontrivial bundles (which are constant) by distinct discrete theta terms.

Central charge. The central charge of the CFT in the deep ultraviolet is $\frac{1}{2} \dim(R_{\ell/r})$, and in the deep infrared is $\frac{1}{2} \dim(R_{\ell/r}) - c(G_{I(R_{\ell/r})})$. Note that c decreases and dynamics is compatible with the c-theorem. Note also that both c_{ℓ} and c_r decrease by the same amount (because $I(R_{\ell}) \equiv I(R_r)$, by gauge anomaly cancellation, cf. (2.2.16)), which is a consequence of the conservation of the gravitational anomaly $c_{\ell} - c_r$. It might be interesting to note that gapped theories "erase information maximally" in the sense that they decrease the c function as much as possible.

It should be pointed out that the infrared theory described by the coset (2.5.1) is not expected to be robust under deformations in the ultraviolet. If we add mass terms or fourfermi terms, in general one would find that the infrared theory is deformed as well, and the coset (2.5.1) flows to a different theory. This new theory has smaller (or equal) central charge. In the case of TQFTs, the central charge is already zero so deformations in the ultraviolet will map the infrared theory to a different TQFT, with generically fewer vacua. This is to be contrasted with the similar situation in 3d: here, infrared TQFTs are actually robust under small ultraviolet deformations. The reason is that 2d TQFTs have local operators, while 3dTQFTs do not; therefore, local deformations in the ultraviolet map to non-trivial infrared operators in 2d, but to the trivial operator in 3d. **Some simple examples.** While we will work out plenty of examples in the next few subsections, we can list a couple of simple examples here, which will hopefully illustrate some of the main features.

Take the QCD theory (SU(2); 7, 7). The infrared dynamics is conjecturally described by the coset $SO(7)_1/SU(2)_{28}$. This CFT has central charge c = 7/10, which agrees with the central charge of the tricritical Ising model. There are only two fermionic CFTs with this central charge: the tricritical Ising model itself (thought of as a fermionic CFT that does not in fact depend on the spin structure), or its fermionization. In other words, it is either a bosonic minimal model, promoted to fermionic in a trivial way, or it is a fermionic minimal model [165–167].

Here it is easy to determine which of these options is correct. In the deep ultraviolet there are 7 free fermions, so the system carries $-1 \mod 8$ units of 't Hooft anomaly under the chiral \mathbb{Z}_2 symmetry. A bosonic theory cannot match this, so the second option is correct: this QCD system flows in the infrared to the fermionized tricritical Ising model.⁶⁰ Note that this theory precisely matches the 't Hooft anomaly for the discrete chiral symmetry [168].

Once the correct low-energy degrees of freedom have been identified, one can ask several interesting questions. For example, one could try to determine the mapping between relevant operators in the ultraviolet to operators in the infrared. The spectrum of infrared operators, together with their quantum numbers, is well understood. The most relevant operator in the ultraviolet is the mass term, and the most relevant operator in the infrared is the (1/10, 1/10) operator, so it is a very natural guess that these operators are identified. Moreover, both are odd under the chiral \mathbb{Z}_2 symmetry. A similar analysis can be performed for the rest of operators. When the mapping is complete, one can study the deformed theory, where one adds suitable scalar operators to the Lagrangian; this gives us a window to the infrared of the massive QCD theory, by turning on the deformation (1/10, 1/10) to the infrared CFT.

Finally, this scenario predicts that the fermionic tricritical Ising model is invariant under time-reversal, with $\mathsf{T}^2 = (-1)^F$, and that this symmetry has a nonperturbative 't Hooft anomaly. This symmetry, and anomaly, are manifest in the ultraviolet, where it acts as $\psi(t) \mapsto \gamma^0 \psi(-t)$, with 't Hooft anomaly measured by the number of fermions mod 2, in this case $7 \equiv 1 \mod 2$. It would be interesting to understand how this symmetry acts on the infrared CFT, and to determine its anomaly directly.

A very similar story holds for the QCD theory $(\text{Spin}(7); \mathbf{8}, \mathbf{8})$. The infrared dynamics is conjecturally described by the coset $SO(8)_1/Spin(7)_1$. This CFT has central charge c = 1/2, so it is either the bosonic Ising model (promoted to a fermionic theory in a trivial way), or the fermionized Ising model, i.e., a free Majorana fermion. As before, it is easy to determine which of these options is actually realized: there are 8 fermions in the deep ultraviolet, so the \mathbb{Z}_2 chiral symmetry has no 't Hooft anomalies. This is only matched by the first option,

 $^{^{60}}$ We also show this directly in section 2.7.2.

namely the bosonic Ising model; hence, this is what QCD flows to in the infrared.⁶¹

Much like above, one can try to determine how the ultraviolet operators are mapped to the infrared ones, and what happens when we deform the theory by these operators.

In these two examples we extracted the physics of the coset directly from its central charge. This was possible thanks to the fact that they are both smaller than unity: c < 1. For generic QCD theories, the central charge is c > 1 and its knowledge alone does not uniquely determine the CFT. In this situation, the properties of the infrared are to be extracted from the CFT SO(n)/ G_k by the standard coset construction. We review this construction in section 2.7. Here we also revisit the c = 7/10 and c = 1/2 examples again, and confirm that they correspond to the fermionic tricritical Ising model and the bosonic Ising mode, respectively, by explicitly working out the branching functions of the coset.

From now on we will assume that the conjecture (2.5.1) is correct. We can use it to propose explicit descriptions of the strongly coupled infrared dynamics of interesting QCD theories.

2.5.1 Gapped theories

Let us make a few remarks about QCD theories on table 2.4; these theories are gapped, so their infrared involves a certain TQFT that describes their vacua.

Adjoint QCD. The first interesting example is adjoint QCD, namely the gauge theory with gauge group G and a fermion in the adjoint representation. This theory has received a lot of attention in the past, see [99, 136, 137, 169–173] for a sample of papers.

The vacua of these theories are described by the topological coset

$$\frac{\mathrm{SO}(\dim(\mathfrak{g}))_1}{G_h},\qquad(2.5.3)$$

where h is the dual Coxeter number of \mathfrak{g} (cf. table 2.3). The branching functions of this coset are well understood [174]:

$$d_{\text{NS-NS}} = \sum_{\lambda \in \mathcal{R}} \chi_{\lambda}$$

$$d_{\text{NS-R}} = \sum_{\lambda \in \mathcal{R}} (-1)^{h_{\lambda}} \chi_{\lambda}$$

$$d_{\text{R-NS}} = 2^{r/2} \chi_{\rho}$$

$$d_{\text{R-R}} = 0,$$
(2.5.4)

where $r = \operatorname{rank}(\mathfrak{g})$, ρ denotes the Weyl vector and

$$\mathcal{R} = \{ \lambda \mid \exists \hat{w} \in \hat{W} \text{ such that } \lambda = h\hat{\omega}_0 + (\hat{w} - 1)\hat{\rho} \}.$$
(2.5.5)

 $^{^{61}}$ We also show this directly in section 2.7.2.

From these equations it immediately follows that \mathcal{H}_{R} is always supersymmetric and has 2^{r} states. The space \mathcal{H}_{NS} also has these many states and it is purely bosonic. In other words,

infrared of adjoint QCD:
$$\begin{cases} \mathcal{H}_{\rm NS} &= \mathbb{C}^{2N|0} \\ \mathcal{H}_{\rm R} &= \mathbb{C}^{N|N} \end{cases}, \qquad N := 2^{r-1}. \tag{2.5.6}$$

Furthermore, by explicitly constructing \mathcal{R} for the different simple algebras, one observes that half the states in $\mathcal{H}_{\rm NS}$ are charged under $(-1)^{F_L}$, and the other half is not. In other words, half the representations in \mathcal{R} have integral spin $h_{\lambda} \in \mathbb{Z}$, and the other half have half-integral spin $h_{\lambda} \in \mathbb{Z} + \frac{1}{2}$. The only exception is $\mathrm{SU}(2n+1)$, which has $2^{n-1}(2^n+1)$ states with integral spin and $2^{n-1}(2^n-1)$ states with half-integral spin.

QCD with bifundamentals. The next few interesting examples correspond to theories with gauge group $G \times G$ and fermions in the bifundamental representation, namely

$$S(U(N) \times U(M)) + (\Box, \Box)$$

$$SO(N) \times SO(M) + (\Box, \Box)$$

$$Sp(N) \times Sp(M) + (\Box, \Box) ,$$

$$(2.5.7)$$

whose space of vacua are described by the following cosets:

$$\frac{\mathrm{U}(NM)_1}{S(\mathrm{U}(N)_M \times \mathrm{U}(M)_N)}, \qquad \frac{\mathrm{SO}(NM)_1}{\mathrm{SO}(N)_M \times \mathrm{SO}(M)_N}, \qquad \frac{\mathrm{SO}(4NM)_1}{\mathrm{Sp}(N)_M \times \mathrm{Sp}(M)_N}.$$
(2.5.8)

The branching rules of these cosets are well-known [175–178]: they describe the level-rank dualities $\mathfrak{g}(N)_k \leftrightarrow \mathfrak{g}(k)_N$. The decomposition of numerator characters $d_{\pm,\pm}$ into denominator characters χ takes the following general form:

$$d_{\text{NS-X}}(q, g_1, g_2) = \sum_{\lambda} (\pm 1)^{2(h_{\lambda} + h_{\lambda^t})} \chi_{\lambda}(q, g_1) \chi_{\lambda^t}(q, g_2) d_{\text{R-X}}(q, g_1, g_2) = \sum_{\lambda} (\pm 1)^{2(h_{\lambda} + h_{\lambda^t})} \chi_{\gamma \cdot \lambda}(q, g_1) \chi_{\lambda^t}(q, g_2) ,$$
(2.5.9)

where λ denotes a primary of $\mathfrak{g}(N)_k$, and λ^t the primary of $\mathfrak{g}(k)_N$ obtained from λ by transposing the Young diagram. Moreover, h_{λ} denotes the conformal dimension of λ , and $\gamma \in Z(G_{sc})$ a suitable simple current. Finally, g_1, g_2 denote flavor symmetry elements of $\mathfrak{g}(N), \mathfrak{g}(k)$, respectively.

These branching functions imply that there are as many vacua as primaries in \mathfrak{g} , i.e., the

number of vacua is (see e.g. [5])

$$S(U(N) \times U(M)): \binom{N+M-1}{M}$$

$$SO(N) \times SO(M): \begin{cases} \binom{n+m}{m} + \binom{n+m-1}{m-1} & (N,M) = (2n,2m+1) \\ \frac{1}{2}\binom{n+m}{m} + \binom{n+m-1}{m-1} + \frac{1}{2}\binom{n+m-2}{m-2} + \frac{3}{2}\binom{n+m-2}{m} & (N,M) = (2n,2m) \\ \frac{1}{2}\binom{n+m}{m} + \frac{1}{2}\binom{n+m-1}{m-1} + \frac{3}{2}\binom{n+m-1}{m} & (N,M) = (2n+1,2m) \\ \binom{n+m}{m} & (N,M) = (2n+1,2m+1) \end{cases}$$

$$Sp(N) \times Sp(M): \binom{N+M}{M}.$$

$$(2.5.10)$$

QCD with rank-2. Another interesting example is the theories with rank-2 quarks. Here we discuss the two theories

$$\frac{\operatorname{Spin}(N) + \Box}{\operatorname{Sp}(N) + \Box}$$
(2.5.11)

whose vacua are described by the cosets

$$\frac{\mathrm{SO}((N+2)(N-1)/2)_1}{\mathrm{Spin}(N)_{N+2}}, \qquad \frac{\mathrm{SO}((2N+1)(N-1))_1}{\mathrm{Sp}(N)_{N-1}}.$$
 (2.5.12)

We are not aware of an explicit discussion of the branchings of these cosets in the literature. That being said, the Ramond sector turns out to be particularly simple, and is reminiscent of the adjoint case (2.5.4):

$$d_{\text{R-R}} = 0,$$
Spin(N), N odd:
$$d_{\text{R-NS}} = 2^{(N-1)/4} \chi_{[3,1,1,\dots,1,1,3]} + 2^{(N-1)/4} \chi_{[1,3,1,1,\dots,1,1,3]}$$
Spin(N), N even:
$$d_{\text{R-NS}} = 2^{(N-2)/4} \chi_{[3,1,1,\dots,1,1,3]} + 2^{(N-2)/4} \chi_{[1,3,1,1,\dots,1,1,3]} + \text{c.c.}$$
Sp(N):
$$d_{\text{R-NS}} = 2^{(N-1)/2} \chi_{[0,1,\dots,1,0]}.$$
(2.5.13)

From this we automatically conclude that there are $2^{\lfloor N/2 \rfloor+1}$ and 2^{N-1} vacua, respectively. In the Ramond sector these vacua are split half-and-half into bosons and fermions, while in the Neveu-Schwartz sector they are all bosonic. In this latter sector, half the states are charged under $(-1)^{F_L}$ and the other half is neutral (i.e., half the primaries have integral spin and the other half-integral spin).

The branching rules for the other gapped theories with rank-2 quarks, namely U(N) plus a symmetric or anti-symmetric quark, are analyzed in [179].

Exceptionals. We close this section with an example involving an exceptional Lie group, to wit

$$F_4 + \mathbf{26}$$
. (2.5.14)

The vacua of this theory are described by the coset

$$\frac{\mathrm{SO}(26)_1}{F_{4,3}},\tag{2.5.15}$$

whose branching functions are

$$d_{\rm NS-X} = \chi_1 \pm \chi_{26} + \chi_{273} \pm \chi_{1274}$$

$$d_{\rm R-NS} = 2\chi_{4096}$$

$$d_{\rm R-R} = 0.$$

(2.5.16)

Hence, this theory has $\mathcal{H}_{NS} = \mathbb{C}^{4|0}$ and $\mathcal{H}_{R} = \mathbb{C}^{2|2}$.

2.5.2 QCD with fundamental matter

Here we describe the infrared dynamics of QCD with quarks in the fundamental representation. More precisely, we shall discuss the following theories

- $\operatorname{SU}(N) + N_F \square$.
- $\operatorname{SO}(N) + N_F \square$.
- $\operatorname{Sp}(N) + N_F \square$.

These describe the celebrated 't Hooft model [143].

The coset CFTs that describe the low energy limit of these theories are

$$\frac{\mathrm{U}(NN_F)_1}{\mathrm{SU}(N)_{N_F}}, \qquad \frac{\mathrm{SO}(NN_F)_1}{\mathrm{SO}(N)_{N_F}}, \qquad \frac{\mathrm{SO}(4NN_F)_1}{\mathrm{Sp}(N)_{N_F}}.$$
(2.5.17)

We now claim that these CFTs are in fact the well-known WZW theories

$$U(N_F)_N$$
, $SO(N_F)_N$, $Sp(N_F)_N$. (2.5.18)

Indeed, the characters of the coset $\mathrm{SO}(\cdots)_1/\mathfrak{g}(N)_{N_F}$ are given by the coefficients of the characters of $\mathfrak{g}(N)_{N_F}$ in the decomposition of $\mathrm{SO}(\cdots)_1$; but, as in (2.5.9), these coefficients are precisely the characters of $\mathfrak{g}(N_F)_N$. In other words, the equality $\mathrm{SO}(\cdots)_1/\mathfrak{g}(N)_{N_F} \equiv \mathfrak{g}(N_F)_N$ is tantamount to the level-rank duality $\mathfrak{g}(N)_{N_F} \leftrightarrow \mathfrak{g}(N_F)_N$.

Let us make a few remarks:

- Note that the infrared CFT is just the WZW model for the flavor symmetry. This CFT manifestly matches the perturbative 't Hooft anomalies for the flavor symmetry in the ultraviolet. So this is the simplest scenario for the infrared dynamics, and could have been guessed independently of the general conjecture (2.5.1). These WZW models also match the nonperturbative anomalies, although in a less obvious way (see below for an explicit example).
- The equality $SO(\dots)_1/\mathfrak{g}(N)_{N_F} \equiv \mathfrak{g}(N_F)_N$ can also be understood as the consequence of the triviality of the coset $SO(\dots)_1/(\mathfrak{g}(N)_{N_F} \times \mathfrak{g}(N_F)_N)$, i.e., of the fact that the gauge theory obtained from $G + N_F \square$ by gauging the flavor symmetry is gapped.
- Similar considerations hold for other gauge groups; for example, if we use Spin(N) instead of SO(N), the flavor symmetry is $O(N_F)$ instead of $\text{SO}(N_F)$, and the infrared CFT is a WZW model with target space $O(N_F)$. This is again a consequence of the level-rank duality $\text{Spin}(N)_{N_F} \leftrightarrow O(N_F)_N$ [113]. Similarly, one could use U(N) instead of SU(N), in which case the infrared CFT is $\text{SU}(N_F)_N$, again by level-rank duality.
- This predicts for example that $SO(N_F)_N$ has an 't Hooft anomaly for time-reversal, measured by $NN_F \mod 2$.

An interesting special case is $\mathrm{SU}(N) + \Box$, i.e., a single copy of the fundamental representation $N_F = 1$. The infrared coset in this case is $\mathrm{U}(1)_N$. For N = 3 this coset is actually a supersymmetric minimal model, $\mathrm{U}(1)_3 = \mathcal{M}_1^{\mathcal{N}=2}$, and therefore $\mathrm{SU}(3) + 3$ has emergent supersymmetry in the infrared. This is a consequence of the fact that $\wedge^3\Box$ contains a gauge singlet, if and only if N = 3 [153].

In the ultraviolet of $\mathrm{SU}(N) + \Box$ there is a manifest \mathbb{Z}_2 chiral symmetry that acts as $(-1)^{F_L} : \psi \mapsto \gamma^3 \psi$, and whose 't Hooft anomaly is the number of Majorana fermions, 2N mod 8. This anomaly must be reproduced by the infrared degrees of freedom, i.e., by $\mathrm{U}(1)_N$. We check this as follows.

The equality $U(N)_1/SU(N)_1 = U(1)_N$ is due to the character decomposition

$$d_{\text{NS-X}}(q, g, \theta) = \sum_{\ell=0}^{N-1} (\pm 1)^{\ell} \chi_{\ell}^{\text{NS-X}}(q, \theta) \chi_{\gamma^{\ell} \cdot \mathbf{0}}(q, g)$$

$$d_{\text{R-X}}(q, g, \theta) = \sum_{\ell=0}^{N-1} (\pm 1)^{\ell+1} \chi_{\ell+\lfloor N/2 \rfloor}^{\text{R-X}}(q, \theta) \chi_{\gamma^{\ell} \cdot \mathbf{0}}(q, g) ,$$
(2.5.19)

where γ is the generator of the $Z(SU(N)) = \mathbb{Z}_N$ center symmetry and **0** is the vacuum character. Also, $\chi_{\ell}(q, \theta)$ denote the regular (bosonic) characters of $U(1)_N$ if N is even, and the super-characters if N is odd; and θ denotes a U(1) flavor fugacity. Finally, $\chi_{\lambda}(q, g)$ denotes an $SU(N)_1$ character with $g \in SU(N)$ flavor fugacity. These branching relations imply that the characters of the coset CFT are

$$b_{\ell}^{\text{NS-NS}} = \chi_{\ell}^{\text{NS-NS}}$$

$$b_{\ell}^{\text{NS-R}} = (-1)^{\ell} \chi_{\ell}^{\text{NS-R}}$$

$$b_{\ell}^{\text{R-NS}} = \chi_{\ell+\lfloor N/2 \rfloor}^{\text{R-NS}}$$

$$b_{\ell}^{\text{R-R}} = (-1)^{\ell} \chi_{\ell+\lfloor N/2 \rfloor}^{\text{R-R}}.$$
(2.5.20)

Hence, the partition function twisted by the $(-1)^{F_L}$ symmetry is

$$\operatorname{tr}_{\mathcal{H}_{\mathrm{NS}}}\left((-1)^{F_{L}}q^{L_{0}-1/24}\bar{q}^{\bar{L}_{0}-1/24}\right) = \sum_{\ell=0}^{N-1} \bar{b}_{\ell}^{\mathrm{NS-R}} b_{\ell}^{\mathrm{NS-NS}}$$
$$= \sum_{\ell=0}^{N-1} (-1)^{\ell} \bar{\chi}_{\ell}^{\mathrm{NS-R}} \chi_{\ell}^{\mathrm{NS-NS}}.$$
$$(2.5.21)$$

The 't Hooft anomaly under $(-1)^{F_L}$ is measured by the phase acquired by this partition function under an ST^2S^{-1} modular transformation. The idea is that S moves the operator $(-1)^{F_L}$ from the spatial cycle into the temporal cycle, so it allows us to access the spin of the operators in the twisted sector. In a non-anomalous fermionic theory, the spin should be half-integral; hence, T^2 measures precisely the extent to which this condition fails. If we use the modular matrices of U(1)_N (see e.g. [1, 180]), we obtain $ST^2S^{-1} = e^{2\pi i \frac{2N}{8}}$, precisely as in the ultraviolet.

2.5.3 WZW models

We noticed in the previous section that the infrared CFT that describes $G + N_F \square$ is just the WZW model for the chiral flavor symmetry. An interesting question one could ask is how general this situation is, i.e., for which QCD theories are the infrared degrees of freedom just the affinization of the ultraviolet currents. As stated in (2.5.2)

$$\frac{\mathrm{SO}(\dim(R))_1}{G_{I(R)}} \equiv H_{\dim(R_0)} \times \frac{\mathrm{SO}(\dim(R))_1}{G_{I(R)} \times H_{\dim(R_0)}}, \qquad (2.5.22)$$

one can always factor out these currents from the infrared coset, and the question becomes for which theories is the remaining sector a trivial CFT. Note that this extra part is precisely the infrared coset of the QCD theory where one gauges the (diagonal) flavor symmetry, $G \times H + (R_0, \Box)$. From this perspective, the answer is straightforward: the theory G + Rflows in the infrared to $H_{\dim(R_0)}$ (plus possibly a trivial CFT, i.e., a TQFT) if and only if the theory $G \times H + (R_0, \Box)$ is gapped. But now we can utilize our classification of gapped theories (cf. table 2.4) to give the list we are after. This way one obtains table 2.5.

G	R	IR WZW
$G \in \text{table } 2.4$	R	Ø
$\mathrm{SU}(N)$	$N_F \square$	$\mathrm{U}(N_F)_N$
$\mathrm{U}(N)$	$N_F \square$	$\mathrm{SU}(N_F)_N$
$\mathrm{SO}(N)$	$N_F \square$	$\mathrm{SO}(N_F)_N$
$\operatorname{Sp}(N)$	$N_F \square$	$\operatorname{Sp}(N_F)_N$
$\mathrm{SU}(N)$	B	$\mathrm{U}(1)_{\frac{1}{2}N(N-1)}$
$\mathrm{SU}(N)$		$U(1)_{\frac{1}{2}N(N+1)}^{2}$
$\operatorname{Spin}(10)$	16	$U(1)_{16}$
E_6	27	$U(1)_{27}$
SU(2)	2	$\mathrm{SU}(2)_1$
SU(2)	4	$\mathrm{SU}(2)_2$
$\operatorname{Sp}(3)$	14	$SU(2)_7$
SU(6)	20	$SU(2)_{10}$
$\operatorname{Spin}(12)$	32	$\mathrm{SU}(2)_{16}$
E_7	56	$SU(2)_{28}$

Table 2.5: Classification of QCD theories that realize in the infrared a pure WZW model for the ultraviolet flavor symmetry (modulo a TQFT). Any theory not on this table will flow in the infrared to H current algebra plus a non-trivial CFT (which has no continuous flavor symmetry). The first line " $G \in$ table 2.4" refers to the fact that gapped theories themselves satisfy this criterion, in the sense that both their flavor symmetry and their infrared CFT are trivial.

2.5.4 Minimal models

It is interesting to note that the (regular and supersymmetric) minimal models [181–187], a celebrated family of 2*d* CFTs, also appear in the infrared of QCD gauge theories. We already noticed three instances of this phenomenon so far, where we found QCD theories that flow to the c = 1/2, c = 7/10 and c = 1 theories in the infrared. Here we describe some families of QCD theories that realize all the minimal models. The examples are by no means exhaustive: there are many other QCD theories that also flow to minimal models in the infrared. In order to simplify the discussion, in this and subsequent examples we shall make no distinction between the bosonized/fermionized versions of a given CFT, and we will not be careful with certain discrete quotients of the gauge group (so for example SO(4) = SU(2) × SU(2)).

Virasoro minimal models. Consider the following QCD theories:

$$SU(2)^{3} \times SO(k) + (2, 1, 2, 1) + (2, 2, 1, \Box)$$

$$SU(2)^{2} \times Sp(k) + (2, 2, 1) + (1, 2, \Box).$$
(2.5.23)

Their infrared is

$$\frac{\mathrm{SO}(4k+4)_1}{\mathrm{SU}(2)_{k+1} \times \mathrm{SU}(2)_k \times \mathrm{SU}(2)_1 \times \mathrm{SO}(k)_4}, \qquad \frac{\mathrm{SO}(4k+4)_1}{\mathrm{SU}(2)_1 \times \mathrm{SU}(2)_{k+1} \times \mathrm{Sp}(k)_1}, \qquad (2.5.24)$$

and we claim that these are both coset realizations of \mathcal{M}_k . Indeed, they can both be written as

$$\mathcal{M}_k \equiv \frac{\mathrm{SU}(2)_k \times \mathrm{SU}(2)_1}{\mathrm{SU}(2)_{k+1}} \tag{2.5.25}$$

thanks to the level-rank dualities $SO(nm)_1/SO(n)_m = SO(m)_n$ and $SO(4nm)_1/Sp(n)_m = Sp(m)_n$.

 $\mathcal{N} = 1$ minimal model. Consider the QCD theories

$$SU(2)^{2} \times Sp(2) \times SO(k) + (2, 1, 4, 1) + (2, 2, 1, \Box)$$

$$SU(2)^{2} \times SO(2) \times Sp(k) + (2, 1, 0, \Box) + (2, 2, 1, 1).$$
(2.5.26)

Their infrared is

$$\frac{\mathrm{SO}(4k+8)_1}{\mathrm{SU}(2)_{k+2} \times \mathrm{SU}(2)_k \times \mathrm{Sp}(2)_1 \times \mathrm{SO}(k)_4}, \quad \frac{\mathrm{SO}(4k+8)_1}{\mathrm{SU}(2)_{k+2} \times \mathrm{SU}(2)_2 \times \mathrm{SO}(2)_4 \times \mathrm{Sp}(k)_1} \quad (2.5.27)$$

and we claim that these are both coset realizations of $\mathcal{M}_k^{\mathcal{N}=1}$. Indeed, they can both be written as

$$\mathcal{M}_{k}^{\mathcal{N}=1} = \frac{\mathrm{SU}(2)_{k} \times \mathrm{SU}(2)_{2}}{\mathrm{SU}(2)_{k+2}}$$
(2.5.28)

for the same reason as for \mathcal{M}_k .

2.5.5 Diagonal coset

Here we discuss a class of QCD theories whose infrared leads to the so-called diagonal cosets $(\mathfrak{g}_k \times \mathfrak{g}_{k'})/\mathfrak{g}_{k+k'}$, whose structure is better understood than that of generic cosets [188, 189]. In particular, consider the following linear quivers:

$$S(U(N) \times U(M) \times U(L)) + (\Box, \Box, \mathbf{1}) + (\mathbf{1}, \Box, \Box)$$

$$SO(N) \times SO(M) \times SO(L) + (\Box, \Box, \mathbf{1}) + (\mathbf{1}, \Box, \Box)$$

$$Sp(N) \times Sp(M) \times Sp(L) + (\Box, \Box, \mathbf{1}) + (\mathbf{1}, \Box, \Box).$$

(2.5.29)

Their infrared theories are

$$\frac{\mathrm{U}(NM + LM)_{1}}{S(\mathrm{U}(N)_{M} \times \mathrm{U}(M)_{N+L} \times \mathrm{U}(L)_{M})},$$

$$\frac{\mathrm{SO}(NM + LM)_{1}}{\mathrm{SO}(N)_{M} \times \mathrm{SO}(M)_{N+L} \times \mathrm{SO}(L)_{M}},$$

$$\frac{\mathrm{SO}(4NM + 4LM)_{1}}{\mathrm{Sp}(N)_{M} \times \mathrm{Sp}(M)_{N+L} \times \mathrm{Sp}(L)_{M}},$$

$$(2.5.30)$$

which, thanks to level-rank duality, become the following diagonal cosets

$$\frac{\mathrm{SU}(M)_N \times \mathrm{SU}(M)_L}{\mathrm{SU}(M)_{N+L}}, \qquad \frac{\mathrm{SO}(M)_N \times \mathrm{SO}(M)_L}{\mathrm{SO}(M)_{N+L}}, \qquad \frac{\mathrm{Sp}(M)_N \times \mathrm{Sp}(M)_L}{\mathrm{Sp}(M)_{N+L}}.$$
(2.5.31)

2.5.6 Kazama-Suzuki

Here we describe QCD theories that acquire an emergent $\mathcal{N} = 2$ supersymmetry in the infrared. In particular, they become Kazama-Suzuki models [190].

Consider the QCD quiver associated to an arbitrary complete graph K_n , where each node represents a gauge group G_i , and each edge a bifundamental quark (see figure 2.2). We take the gauge groups to be any of

$$G = S(U(N_1) \times U(N_2) \times \dots \times U(N_n))$$

$$G = SO(N_1) \times SO(N_2) \times \dots \times SO(N_n)$$

$$G = Sp(N_1) \times Sp(N_2) \times \dots \times Sp(N_n).$$

(2.5.32)

Note that for n = 1, 2 these are pure Yang-Mills and $G \times G + (\Box, \Box)$, i.e., they are both gapped theories (the latter are entries on table 2.4). We claim that for $n \ge 3$, the theory has emergent $\mathcal{N} = 2$ supersymmetry in the infrared, they are Kazama-Suzuki models. Indeed, their infrared cosets are

$$\frac{\mathrm{U}\left(\sum_{i>j}N_{i}N_{j}\right)_{1}}{S\left(\prod_{i}\mathrm{U}(N_{i})_{\sum_{j\neq i}N_{j}}\right)}, \qquad \frac{\mathrm{SO}\left(\sum_{i>j}N_{i}N_{j}\right)_{1}}{\prod_{i}\mathrm{SO}(N_{i})_{\sum_{j\neq i}N_{j}}}, \qquad \frac{\mathrm{SO}\left(4\sum_{i>j}N_{i}N_{j}\right)_{1}}{\prod_{i}\mathrm{Sp}(N_{i})_{\sum_{j\neq i}N_{j}}}, \qquad (2.5.33)$$

which are the cosets that describe Kazama-Suzuki models associated to the embeddings

$$SU(\sum_{i \neq \star} N_i) \supset S(\prod_{i \neq \star} U(N_i)),$$

$$SO(\sum_{i \neq \star} N_i) \supset \prod_{i \neq \star} SO(N_i),$$

$$Sp(\sum_{i \neq \star} N_i) \supset \prod_{i \neq \star} Sp(N_i)$$

$$(2.5.34)$$

at level N_{\star} , where \star is an arbitrary node of K_n .

2.6 Conventions and Background

We work in 1 + 1 dimensional Minkowski spacetime with metric $\eta = \text{diag}(-1, +1)$ and 2 × 2 gamma matrices γ^{μ} , $\mu = 0, 1$. The Hodge dual of a one-form is $(\star j)_{\mu} = \epsilon_{\mu\nu} j^{\nu}$ where we take $\epsilon^{tx} = -\epsilon_{tx} = +1$. In particular, $\star dt = dx$, $\star dx = dt$. In null coordinates x^{\pm} the metric is $\eta_{++} = \eta_{--} = 0$, $\eta_{+-} = \eta_{-+} = 1$, and the star is $\star dx^{\pm} = \pm dx^{\pm}$.



Figure 2.2: The first few complete graphs K_1, K_2, \ldots, K_5 . If we associate to each node a gauge group $U(n_i)$, $SO(n_i)$, $Sp(n_i)$ (with the global U(1) modded out in the unitary case), and to each edge a bifundamental quark, then the quiver gauge theory associated to K_n has a Kazama-Suzuki model as its effective low energy description.

The minimal spinor is Majorana-Weyl, namely one can impose the simultaneous conditions $\gamma^3 \psi = \pm \psi$ and $\psi^* = C \psi$, where $\gamma^3 = \gamma^0 \gamma^1$ is the chirality matrix and C is the chargeconjugation matrix, defined by $(\gamma^{\mu})^* = C \gamma^{\mu} C^{-1}$. It is convenient to choose the Majorana basis $\gamma^{\mu} = (i\sigma^2, \sigma^1)$ where $\gamma^3 = \sigma^3$ and C = 1. In this basis, Majorana fermions are real $\psi^* = \psi$, and chiral fermions are either $\psi \propto \begin{pmatrix} 1 \\ 0 \end{pmatrix}$ or $\psi \propto \begin{pmatrix} 0 \\ 1 \end{pmatrix}$. We take $\psi = 2^{-1/4} \begin{pmatrix} \psi_\ell \\ \psi_r \end{pmatrix}$ and $x^{\pm} = \frac{1}{\sqrt{2}} (x^0 \pm x^1)$. The fermion kinetic term is

$$-i\bar{\psi}\gamma^{\mu}\partial_{\mu}\psi = i\psi_{\ell}^{*}\partial_{-}\psi_{\ell} + i\psi_{r}^{*}\partial_{+}\psi_{r}. \qquad (2.6.1)$$

For Grassmann odd a, b we use $(ab)^* = b^*a^*$.

Mass terms. For massless fermions the two chiralities are decoupled. These couple through mass terms

$$i\bar{\psi}\psi = \frac{i}{\sqrt{2}}(\psi_{\ell}^{*}\psi_{r} + \psi_{\ell}\psi_{r}^{*})$$

$$i\bar{\psi}\gamma^{3}\psi = \frac{-i}{\sqrt{2}}(\psi_{\ell}^{*}\psi_{r} - \psi_{\ell}\psi_{r}^{*}).$$
(2.6.2)

both of which are hermitian. Although less obvious, one can also use bilinears of the form

$$re(\bar{\psi}^{*}\psi) = \frac{1}{\sqrt{2}}(\psi_{\ell}\psi_{r} - \psi_{\ell}^{*}\psi_{r}^{*})$$

$$im(\bar{\psi}^{*}\psi) = \frac{i}{\sqrt{2}}(\psi_{\ell}\psi_{r} + \psi_{\ell}^{*}\psi_{r}^{*}).$$
(2.6.3)

both of which are hermitian and Lorentz scalars (recall that $\psi_{\ell} \mapsto e^{\eta/2} \psi_{\ell}$ and $\psi_r \mapsto e^{-\eta/2} \psi_r$ under a boost with rapidity $\eta \in \mathbb{R}$).

If the fermion is real, $\psi_{\ell,r}^* = \psi_{\ell,r}$, then the bilinears $i\bar{\psi}\gamma^3\psi$ and $\operatorname{re}(\bar{\psi}^*\psi)$ both become zero, due to fermi statistics, and the other two bilinears $i\bar{\psi}\psi$ and $\operatorname{im}(\bar{\psi}^*\psi)$ become identical.

Symmetries. Let us list some of the manifest symmetries of QCD theories. The center one-form symmetry is straightforward: it is given by the subgroup of the center that is not screened by the fermions, namely $\ker(R) \subseteq Z(G)$, the kernel of the representation R.

As a matter of principle, it is possible that there are other one-form symmetries that are not associated to the center of the gauge group, although exhibiting these is a much more complicated task.⁶²

Zero-form symmetries are abundant too. For example, if we have N_F massless chiral fermions in a given representation R, one has the following continuous flavor symmetries acting on them:

$$R$$
 complex: $U(N_F)$ R real: $O(N_F)$ (2.6.4) R pseudoreal: $Sp(N_F)$.

When G has no U(1) factors, so that the generators are traceless, the classical axial symmetries are unbroken (unlike in 3 + 1d). If there are abelian factors then U(1) flavor symmetries are generically broken into discrete subgroups. An important discrete chiral subgroup, which is very often present even if U(1) is broken, is the chiral fermion parity \mathbb{Z}_2^L , which negates left movers and fixes right movers, to wit, $\mathbb{Z}_2^L: \psi \mapsto \gamma^3 \psi$. This subgroup has a very well understood group of 't Hooft anomalies, valued in \mathbb{Z}_8 and responsible for many interesting properties of QCD theories.

In presence of mass terms, the continuous chiral symmetries (2.6.4) descend to their diagonal vector-like subgroups, or even smaller subgroups if the different flavors have different masses.

Finally, there are two other discrete symmetries that are quite useful: charge conjugation $\mathbb{Z}_2^{\mathsf{C}}$, which sends a representation R to its conjugate \bar{R} , and time-reversal $\mathbb{Z}_2^{\mathsf{T}}$, which is anti-linear and satisfies $\mathsf{T}^2 = (-1)^F$. Note that charge conjugation exists only if the group admits complex representations, for otherwise the operation is a gauge transformation and thus not a symmetry. On the other hand, time-reversal is a symmetry only if the theory is vector-like $R_\ell = R_r$, because γ^0 interchanges the two chiralities. The combination CT is a symmetry only if $R_\ell = \bar{R}_r$. Charge conjugation and time-reversal act on the gauge fields as

$$C: \begin{cases} \psi(x,t) \mapsto \psi^*(x,t) \\ A_{\mu}(x,t) \mapsto -A^t_{\mu}(x,t) \end{cases}$$

$$T: \begin{cases} \psi(x,t) \mapsto \gamma^0 \psi(x,-t) \\ A_0(x,t) \mapsto +A^t_0(x,-t) \\ A_1(x,t) \mapsto -A^t_1(x,-t) , \end{cases}$$

$$(2.6.5)$$

where t denotes transposition.

 $^{^{62}}$ For example it has recently been appreciated [191] that pure Yang-Mills has a very large space of non-invertible one-form symmetries, valued in a maximal torus of G. It is not clear which if these, if any, survive the introduction of matter, or whether there are other one-form symmetries – invertible or otherwise – beyond these, although at face value it seems unlikely.

The three discrete \mathbb{Z}_2 transformations, chiral fermion parity, charge conjugation, and time-reversal, act as follows on the fermion mass terms:

$$i\bar{\psi}\psi \ i\bar{\psi}\gamma^{3}\psi \ \operatorname{re}(\bar{\psi}^{*}\psi) \ \operatorname{im}(\bar{\psi}^{*}\psi)$$

$$\mathbb{Z}_{2}^{L} - - - - -$$

$$\mathbb{Z}_{2}^{C} + - - +$$

$$\mathbb{Z}_{2}^{T} - + - - -$$
(2.6.6)

Note that all the mass terms are odd under the chiral symmetry, which suggests that this symmetry might be anomalous. On the other hand, there is at least one mass term that is even under the other two symmetries, so they are non-anomalous. The exception is when the fermions are real, because in that case the mass terms $i\bar{\psi}\gamma^3\psi$ and $\operatorname{re}(\bar{\psi}^*\psi)$ vanish due to fermi statistics. In this situation, the remaining mass terms $i\bar{\psi}\psi$ and $\operatorname{im}(\bar{\psi}^*\psi)$ (which are in fact equal) are odd under time-reversal, which suggests that such symmetry might be anomalous too (and, if so, the anomaly will be at most a mod 2 effect; this is confirmed by $\Omega_3^{\text{pin}^+} = \mathbb{Z}_2$).

2.6.1 Matching Energy Momentum Tensors

We have the canonical energy momentum tensor built from the fermion fields which takes the form

$$T(z) = -\frac{1}{2} \sum_{i} :\psi^{i} \partial \psi^{i} : (z). \qquad (2.6.7)$$

The normal ordering is defined as the constant part of the OPE of AB. The notation above for two operators is defined by

$$:AB: (z) = \frac{1}{2\pi i} \oint_{z} \frac{dx}{x-z} A(x)B(z) , \qquad (2.6.8)$$

and extracts the constant part of the OPE of A and B via a contour integral. We can define the current $J^a(z) = \frac{1}{2} : \psi^i t^a_{ij} \psi^j : (z)$, for t^a_{ij} the generators of the \mathfrak{g} Lie algebra. We write the energy momentum tensor given by the Sugawara construction as

$$\mathcal{T}(z) = \gamma \sum_{a} ::: \psi^{i} t^{a}_{ij} \psi^{j} ::: \psi^{k} t^{a}_{kl} \psi^{l} ::: (z)$$

$$= \gamma \sum_{a} t^{a}_{ij} t^{a}_{kl} ::: \psi^{i} \psi^{j} ::: \psi^{k} \psi^{l} ::: (z) .$$
(2.6.9)

We first consider the term :: $\psi^i \psi^j$:: $\psi^k \psi^l$::. By the rearrangement lemma [150, Appendix 6.C] we have

$$::\psi^{i}\psi^{j}::\psi^{k}\psi^{l}::(z) = :\psi^{i}:\psi^{j}:\psi^{k}\psi^{l}::(z) + :[:\psi^{i}\psi^{j}:,:\psi^{k}\psi^{l}:]:(z) + ::[:\psi^{k}\psi^{l}:,\psi^{i}]:(z) + ::\psi^{i}:[:\psi^{k}\psi^{l}:,\psi^{j}]::(z),$$
(2.6.10)

When we substitute these results into (2.6.9) we get

$$\begin{split} \gamma \sum_{a} t^{a}_{ij} t^{a}_{kl} &:: \psi^{i} \psi^{j} ::: \psi^{k} \psi^{l} ::: (z) = \gamma \sum_{a} t^{a}_{ij} t^{a}_{kl} :: \psi^{i} :: \psi^{j} :: \psi^{k} \psi^{l} ::: (z) \\ &+ \gamma \sum_{a} t^{a}_{ij} t^{a}_{kl} [-\delta^{jk} :: \psi^{l} \partial_{z} \psi^{i} :+ \delta^{jl} :: \psi^{k} \partial_{z} \psi^{i} :- \delta^{il} :: \psi^{k} \partial_{z} \psi^{j} :+ \delta^{ik} :: \psi^{l} \partial_{z} \psi^{j} :] \\ &= \gamma \sum_{a} t^{a}_{ij} t^{a}_{kl} :: \psi^{i} :: \psi^{j} :: \psi^{k} \psi^{l} ::: (z) + \gamma \sum_{a} [-4t^{a}_{ij} t^{a}_{jl} :: \psi^{l} \partial_{z} \psi^{i} :] . \end{split}$$
(2.6.11)

The second term above has the form of (2.6.7), where we use the fact that

$$\sum_{a} t_{ij}^{a} t_{jl}^{a} = 2I(R) \frac{\dim(G)}{\dim(R)} \delta^{il}, \quad \gamma = \frac{1}{8(I(R) + h)}, \quad (2.6.12)$$

and take G and R from our list of gapped theories. A necessary condition for equality of T(z) and $\mathcal{T}(z)$ is if $\gamma \sum_{a} t^{a}_{ij} t^{a}_{kl} : \psi^{i} : \psi^{j} : \psi^{k} \psi^{l} ::: (z) = 0$. By definition, this term is

$$\frac{\gamma}{(2\pi i)^4} \sum_a \oint \frac{dw}{w-z} \frac{dx}{x-z} \frac{dy}{y-z} t^a_{ij} t^a_{kl} \psi^i(w) \psi^j(x) \psi^k(y) \psi^l(z), \qquad (2.6.13)$$

and since ψ^i is Grassmann, this vanishes if the total antisymmetrization of $\sum_a t_{ij}^a t_{kl}^a$ vanishes, and this gives us the condition

$$\sum_{a} t^{a}_{ij} t^{a}_{kl} + t^{a}_{ik} t^{a}_{lj} + t^{a}_{il} t^{a}_{jk} = 0.$$
(2.6.14)

If the group $G = \tilde{G} \times U(1)$ which has a U(1) factor, we can write t_{ij}^a as a decomposition under \tilde{G} and U(1), where it is charge q under the U(1) i.e. the current is $J(z) = \frac{1}{2} : \psi^i q \, \psi^i$. As an example we take U(N) with a fermion that is in the antisymmetric representation of SU(N) and charge q under U(1). Then the SU(N) part of the Sugawara tensor reads

$$\mathcal{T}_{SU(N)}(z) = \frac{1}{8(2N-2)} \left[-4 \frac{2(N^2 - N - 2)}{N} \right] :\psi^i \partial_z \psi^i :$$

= $-\frac{1}{2} :\psi^i \partial_z \psi^i : +\frac{1}{N(N-1)} :\psi^i \partial_z \psi^i :$ (2.6.15)

The U(1) part of the Sugawara tensor reads

$$\mathcal{T}_{\mathrm{U}(1)}(z) = \frac{1}{4N(N-1)} :: \psi^{i} \psi^{i} :: \psi^{j} \psi^{j} ::= \frac{-1}{N(N-1)} : \psi^{i} \partial_{z} \psi^{i} :, \qquad (2.6.16)$$

and by summing (2.6.15) and (2.6.16) we reproduce (2.6.7).

Now suppose we are working with an Abelian theory and we consider n complex fermions, with q_{Ia} charge matrix, i.e. $\mathrm{U}(1)^n_{q^tq}$. The canonical energy momentum is given by

$$T(z) = \frac{1}{2} (:\partial_z \psi^{I\dagger} \psi^I : -: \psi^{I\dagger} \partial_z \psi^I :)(z).$$
(2.6.17)

With the charge matrix we can define the current $J_a = \psi^{I\dagger} q_{Ia} \psi^{I}$, which satisfies the OPE

$$J_a(z)J_b(0) \sim \frac{k_{ab}}{z^2}.$$
 (2.6.18)

From this we build the Sugawara tensor

$$\mathcal{T}(z) = \frac{1}{2} \sum_{a,b} (k_{ab})^{-1} ::: \psi^{I\dagger} q_{Ia} \psi^{I} ::: \psi^{J\dagger} q_{Jb} \psi^{J} ::: (z) , \qquad (2.6.19)$$

where $k_{ab} = (q^t q)_{ab} = q_{Ia} q_{Ib}$, so $(k_{ab})^{-1} = (q_{Ia} q_{Ib})^{-1}$. and we can define the current as $J_a = :\psi^{I\dagger} q_{Ia} \psi^I$: Again by the rearrangement lemma we have

$$q_{Ia} q_{Jb} :: \psi^{I\dagger} \psi^{I} :: \psi^{J\dagger} \psi^{J} :: (z) = q_{Ia} q_{Jb} : \psi^{I\dagger} : \psi^{I} : \psi^{J\dagger} \psi^{J} ::: (z) + q_{Ia} q_{Jb} (-\delta^{IJ} \psi^{J} : \partial_{z} \psi^{I\dagger} : -\delta^{IJ} : \psi^{J\dagger} \partial_{z} \psi^{I} :)(z) = q_{Ia} q_{Ib} (-: \psi^{I} \partial_{z} \psi^{I\dagger} : -: \psi^{I\dagger} \partial_{z} \psi^{I} :)(z) .$$
(2.6.20)

After substituting into (2.6.19) we get

$$\begin{aligned} \mathcal{T}(z) &= \frac{1}{2} \sum_{a,b} (q_{Ib}^{-1} q_{Ia}^{-1}) [q_{Ia} q_{Jb} : \psi^{I\dagger} : \psi^{I} : \psi^{J\dagger} \psi^{J} ::: (z) \\ &+ q_{Ia} q_{Jb} (-\delta^{IJ} : \psi^{J} \partial_{z} \psi^{I\dagger} : -\delta^{IJ} : \psi^{J\dagger} \partial_{z} \psi^{I} :) (z)] \\ &= \frac{1}{2} [: \psi^{I\dagger} : \psi^{I} : \psi^{J\dagger} \psi^{J} ::: (z) + (-\delta^{IJ} : \psi^{J} \partial_{z} \psi^{I\dagger} : -\delta^{IJ} : \psi^{J\dagger} \partial_{z} \psi^{I} :) (z)] \\ &= \frac{1}{2} (: \partial_{z} \psi^{I\dagger} \psi^{I} : - : \psi^{I\dagger} \partial_{z} \psi^{I} :) (z) \\ &= T(z) , \end{aligned}$$
(2.6.21)

which is the canonical energy momentum tensor. We have used the fact that $\sum_{b} (q_{Ib})^{-1} q_{Jb} = \delta_{IJ}$, and the first term in the second equality vanishes by applying (2.6.13) and evaluating the contour integrals.

2.6.2 Temporal Gauge Hamiltonian commutation

The quantized Hamiltonian is given by integrating the Hamiltonian action in (2.4.47), where we take the left and right handed fermions $\psi_{\ell}(x)$ and $\psi_{r}(x)$ to be operators on a circle where $0 \le x \le 2\pi$:

$$\begin{split} \widehat{H} &= \int_{0}^{2\pi} \Big[\frac{i}{8(I(R)+h)} \left(:\widehat{J}_{\ell}^{a} \widehat{J}_{\ell}^{a}(x) :+ :\widehat{J}_{r}^{a} \widehat{J}_{r}^{a}(x) : \right) \\ &+ \frac{1}{\sqrt{2}} i \lim_{\epsilon \to 0} \left(\langle \psi_{i\,\ell}^{\dagger}(x+\epsilon) \partial_{x} \psi_{i\,\ell}(x-\epsilon) \rangle - \langle \psi_{i\,r}^{\dagger}(x+\epsilon) \partial_{x} \psi_{i\,r}(x-\epsilon) \rangle \right) \\ &+ \frac{1}{\sqrt{2}} \widehat{A}^{a}(x) (\widehat{J}_{\ell}^{a} - \widehat{J}_{r}^{a})(x) + \frac{1}{\sqrt{2}} I(R) \widehat{A}^{a}(x)^{2} \Big] dx \\ &+ g^{2} \int_{0}^{2\pi} (\widehat{E}^{a}(x))^{2} dx. \end{split}$$
(2.6.22)

We use the expression for $\widehat{J}^a_{\ell,r}$ as specifically given in section 2.4.2.

For the commutator $[\widehat{G}^a, \widehat{H}] = 0$, the pieces proportional to g^0 and g^2 vanish separately. The commutator of \widehat{G}^a with $:\widehat{J}^a_{\ell,r}\widehat{J}^a_{\ell,r}(x)$: vanishes [192], and we proceed to use the commutation relations in (2.4.42) to show that the other terms in (2.6.22) vanish when commuted with \widehat{G}^a . Consider first the terms:

$$\left[\widehat{J}^{a}_{\ell}(x), \int_{0}^{2\pi} \widehat{A}^{b} \widehat{J}^{b}_{\ell}(y) dy\right] = \int_{0}^{2\pi} \widehat{A}^{b}(y) \left(if^{abc} J^{c}_{\ell}(x)\delta(x-y) + iI(R)\delta^{ab}\partial_{x}\delta(x-y)\right) dy \quad (2.6.23)$$

$$\left[\widehat{J}^{a}_{\ell}(x), \int_{0}^{2\pi} \widehat{J}^{b}_{\ell}(y)dy\right] = \int_{0}^{2\pi} \widehat{J}^{b}(y) \left(if^{abc} J^{c}_{\ell}(x)\delta(x-y) + iI(R)\delta^{ab}\partial_{x}\delta(x-y)\right) dy \quad (2.6.23)$$

$$\left[\widehat{J}_r^a(x), \int_0^{\infty} \widehat{A}^b \widehat{J}_r^b(y) dy\right] = \int_0^{\infty} \widehat{A}^b(y) \left(if^{abc} J_r^c(x) \delta(x-y) - iI(R) \delta^{ab} \partial_x \delta(x-y)\right) dy. \quad (2.6.24)$$

We then look at the terms:

$$-[D_x\widehat{\Pi}^a(x), \int_0^{2\pi} \widehat{A}^d (\widehat{J}^d_\ell - \widehat{J}^d_r)(y)dy] = -\int_0^{2\pi} i\delta^{ab}\partial_x \delta(x-y) \left(\widehat{J}^b_\ell - \widehat{J}^b_r\right)(y)dy \qquad (2.6.25)$$
$$-\int_0^{2\pi} if^{abc}\widehat{A}^b(x)\delta^{cd} \left(\widehat{J}^d_\ell - \widehat{J}^d_r\right)(y)\delta(x-y)dy,$$
$$-I(R)[D_x\widehat{\Pi}^a(x), \int_0^{2\pi} \widehat{A}^d(y)^2dy] = -2iI(R)\int_0^{2\pi} \widehat{A}^d(y)\delta^{ad}\partial_x\delta(x-y)dy \qquad (2.6.26)$$
$$-2iI(R)\int_0^{2\pi} if^{abc}\widehat{A}^d(y)\widehat{A}^b(x)\delta^{cd}\delta(x-y)dy.$$

The second term in (2.6.23) and (2.6.24) cancel with the first term in (2.6.26). The first term in (2.6.23) and (2.6.24) cancel with the second term in (2.6.25); the last term in (2.6.26) vanishes by antisymmetry. We are thus left to negotiate the term

$$-\int_{0}^{2\pi} i\delta^{ab}\partial_x\delta(x-y)\left(\widehat{J}^b_\ell - \widehat{J}^b_r\right)(y)dy \qquad (2.6.27)$$

in (2.6.25). For this we consider

$$\left[\widehat{J}^{a}_{\ell,r}(x), \pm \int_{0}^{2\pi} i \lim_{\epsilon \to 0} \langle \psi^{\dagger}_{k\,\ell,r}(y+\epsilon) \partial_{y} \psi_{k\,\ell,r}(y-\epsilon) \rangle dy\right]$$
(2.6.28)

where we first compute the commutator by treating the second term as an operator, and then using the propagator while taking the $\epsilon \to 0$ limit. This is the same procedure used to prove that $[\widehat{J}^a_{\ell,r}(x), \widehat{J}^b_{\ell,r}(y)]$ contains a Schwinger term. Working with just the left handed part of 2.6.28 we get

$$\begin{split} \left[\hat{J}_{\ell}^{a}(x), \int_{0}^{2\pi} i \lim_{\epsilon \to 0} \langle \psi_{k\,\ell}^{\dagger}(y+\epsilon) \partial_{y}\psi_{k\,\ell}(y-\epsilon) \rangle dy\right] \\ &= \left[\frac{1}{2}\psi_{i\ell}^{\dagger}t_{ij}^{a}\psi_{j\,\ell}(x), \int_{0}^{2\pi} i \lim_{\epsilon \to 0} \langle \psi_{k\,\ell}^{\dagger}(y+\epsilon) \partial_{y}\psi_{k\,\ell}(y-\epsilon) \rangle dy\right] \\ &= \frac{i}{2} \lim_{\epsilon \to 0} \left\{ \int_{0}^{2\pi} \delta(x-y-\epsilon) \partial_{y}(\psi_{i\ell}^{\dagger}(y+\epsilon)t_{ij}^{a}\psi_{j\ell}(y-\epsilon)) dy \right. \\ &+ \int_{0}^{2\pi} \partial_{y}\delta(x-y+\epsilon)\psi_{j\,\ell}(y+\epsilon)t_{ij}^{a}\psi_{j\ell}(y-\epsilon) dy \right\} \\ &= \frac{i}{2} \lim_{\epsilon \to 0} \left\{ -\int_{0}^{2\pi} \partial_{y}\delta(x-y-\epsilon)\psi_{i\ell}^{\dagger}(y+\epsilon)t_{ij}^{a}\psi_{j\ell}(y-\epsilon) dy \right\} \\ &+ \int_{0}^{2\pi} \partial_{y}\delta(x-y+\epsilon)\psi_{j\ell}(y+\epsilon)t_{ij}^{a}\psi_{i\ell}(y-\epsilon) dy \right\} \\ &= i \left\{ -\int_{0}^{2\pi} \partial_{y}\delta(x-y) :\psi_{i\ell}^{\dagger}(y)t_{ij}^{a}\psi_{j\ell}(y) :dy \right\} \\ &= i \int_{0}^{2\pi} \partial_{x}\delta(x-y)\hat{J}^{a}(y) dy \,, \end{split}$$

which cancels the first term in (2.6.27), and we can do an analogous computation for \hat{J}_r^a . The first equality we use the fact that $\hat{J}^a(x) = \frac{1}{2} \psi_{i\ell}^{\dagger} t_{ij}^a \psi_{j\ell}(x) + \text{(singular term)}$, where the singular term vanishes in the commutator. To go from the third equal sign to the fourth equal sign we replace the fermions by the normal ordered version where [193]

$$\psi_{i\ell}^{\dagger}(y+\epsilon)t_{ij}^{a}\psi_{j\ell}(y-\epsilon) = :\psi_{i\ell}^{\dagger}(y+\epsilon)t_{ij}^{a}\psi_{j\ell}(y-\epsilon): +\lim_{\tilde{\epsilon}\to 0}\langle\psi_{i\ell}^{\dagger}(x+\epsilon+\tilde{\epsilon})t_{ij}^{a}\psi_{j\ell}(x-\epsilon-\tilde{\epsilon})\rangle, \ (2.6.30)$$

and the second term vanishes under the derivative. For the term proportional to g^2 we consider

$$[D_x\widehat{\Pi}^a(x), \int_0^{2\pi} \widehat{E}^d(y)^2 dy] = -2i \int_0^{2\pi} i f^{abc} \widehat{E}^b(x) \widehat{E}^c(y) \delta(x-y) dy, \qquad (2.6.31)$$

which vanishes due to the antisymmetry of f^{abc} .

2.7 Infrared coset CFTs

In this section we review the formalism of coset CFTs [150], our primary goal being to understand the CFTs that appear in the deep infrared of QCD (2.5.1).

Chiral characters. One of the most important concepts in RCFT is that of a chiral character. These consist of a finite family of functions $\{\chi_{\lambda}(q)\}$ of the complex structure

 $q = e^{2\pi i \tau}$, holomorphic in the upper half plane, and labelled by the primaries of the theory λ (the representations of the chiral algebra). Given these characters, the torus partition function of the theory takes the form

$$Z(q) = \sum_{\lambda,\bar{\lambda}} M_{\bar{\lambda},\lambda} \,\bar{\chi}_{\bar{\lambda}}(\bar{q}) \chi_{\lambda}(q) \equiv \bar{\chi}^{\dagger} M \chi \,. \tag{2.7.1}$$

Here M, the so-called mass matrix, specifies how the left-moving sectors are paired up with right-moving ones. The possible choices for M are constrained by the requirement of Z being a modular-invariant function of q. This is archived by a key property of the characters, to wit, their covariance under modular transformations. Under a generic such transformation, the characters mix with each other in a well-defined fashion, and the role of M is to ensure that the sesquilinear form $Z = \chi^{\dagger} M \chi$ is a scalar under these transformations. As a result, while Z(q) is a well-defined function on $\mathbb{H}/SL(2,\mathbb{Z})$, the tuple χ is best thought of as a non-trivial section thereon.

When $\chi_{\lambda}(q)$ admits a Hilbert space interpretation, it is defined as

$$\chi_{\lambda}(q) := \operatorname{tr}_{\mathcal{H}_{\lambda}}(q^{L_0 - c/24}), \qquad (2.7.2)$$

such that

$$Z(q) = \operatorname{tr}_{\mathcal{H}}(q^{L_0 - c/24} \bar{q}^{\bar{L}_0 - c/24}), \qquad \mathcal{H} = \bigoplus_{\lambda,\bar{\lambda}} M_{\bar{\lambda},\lambda} \mathcal{H}_{\bar{\lambda}} \otimes \mathcal{H}_{\lambda}.$$
(2.7.3)

Here \mathcal{H} is the full Hilbert space of the theory, and \mathcal{H}_{λ} is the representation space (module) for λ . The mass matrix M dictates how these chiral modules combine into \mathcal{H} . From now on, and in order to simplify the notation and presentation, we always have in mind the diagonal theory $M_{\bar{\lambda},\lambda} = \delta_{\bar{\lambda},\lambda}$. Non-diagonal theories can often be thought of as the diagonal theory of a larger algebra via (potentially non-abelian) anyon condensation.

Fermionic CFTs. A CFT is said to be *fermionic* if, on top of the dependence on the conformal structure of spacetime, it also depends on the choice of spin structure thereof. In other words, a fermionic CFT depends on the boundary conditions for fermionic fields. The CFTs that appear at RG fixed points of QCD theories are naturally fermionic, because the microscopic theory contains quarks. Hence our main interest is in fermionic CFTs.

In the case of the torus there are four spin structures, corresponding to either periodic or anti-periodic boundary conditions around the two non-trivial cycles. We also refer to these boundary conditions as Ramond and Neveu-Schwartz, respectively, and we use the notation + = R, - = NS interchangeably.

In a fermionic CFT, the characters acquire a dependence on the spin structure: they

become super-characters. Consequently, we denote them as $\chi_{\lambda}^{\pm,\pm}$ where

$$\chi_{\lambda}^{\text{NS-NS}}(q) := \operatorname{tr}_{\mathcal{H}_{\text{NS};\lambda}}(q^{L_0-c/24})$$

$$\chi_{\lambda}^{\text{NS-R}}(q) := \operatorname{tr}_{\mathcal{H}_{\text{NS};\lambda}}((-1)^{F_L}q^{L_0-c/24})$$

$$\chi_{\lambda}^{\text{R-NS}}(q) := \operatorname{tr}_{\mathcal{H}_{\text{R};\lambda}}(q^{L_0-c/24})$$

$$\chi_{\lambda}^{\text{R-R}}(q) := \operatorname{tr}_{\mathcal{H}_{\text{R};\lambda}}((-1)^{F_L}q^{L_0-c/24}).$$
(2.7.4)

Here, $\mathcal{H}_{\pm;\lambda}$ denotes the module of λ with fermion boundary conditions \pm , while $(-1)^{F_L}$ denotes the chiral fermion parity operator, which assigns +1 to bosonic left-movers and -1 to fermionic left-movers, while it acts trivially on the right-moving modes.

The partition function of a fermionic CFT is obtained by combining the two chiral halves in a modular covariant way:

$$Z_{\pm,\pm}(q) = \sum_{\bar{\lambda},\lambda} M_{\bar{\lambda},\lambda}^{\pm} \bar{\chi}_{\bar{\lambda}}^{\pm,\pm}(\bar{q}) \chi_{\lambda}^{\pm,\pm}(q) , \qquad (2.7.5)$$

which computes

$$Z_{\pm,\pm}(q) = \operatorname{tr}_{\mathcal{H}_{\pm}}((\mp 1)^{F} q^{L_{0}-c/24} \bar{q}^{\bar{L}_{0}-c/24}), \qquad \mathcal{H}_{\pm} = \bigoplus_{\lambda,\bar{\lambda}} M_{\bar{\lambda},\lambda}^{\pm} \mathcal{H}_{\pm;\bar{\lambda}} \otimes \mathcal{H}_{\pm;\lambda}, \qquad (2.7.6)$$

where $(-1)^F \equiv (-1)^{F_L} (-1)^{F_R}$ is the total fermion parity.

Here the mass matrices M^{\pm} , which dictate how the two chiral halves $\mathcal{H}_{\pm;\bar{\lambda}}$, $\mathcal{H}_{\pm;\lambda}$ combine into the full Hilbert space \mathcal{H}_{\pm} , are chosen so as to ensure that $Z_{\pm,\pm}$ transforms appropriately under modular transformations. Unlike in the case of bosonic CFTs, $Z_{\pm,\pm}$ is not in general invariant under $SL(2,\mathbb{Z})$. Indeed, modular transformations generically map the different spin structures into each other, which induces a reshuffling of the partition functions $Z_{\pm,\pm}$. Specifically, under the standard generators of $SL(2,\mathbb{Z}) = \langle S,T \rangle$, the partition functions transform as

$$S \bigcap Z_{\text{NS-NS}} \xleftarrow{T} Z_{\text{NS-R}} \xleftarrow{S} Z_{\text{R-NS}} \bigcap T$$

$$S \bigcap Z_{\text{R-R}} \bigcap T$$

$$(2.7.7)$$

The choices for the mass matrices M^{\pm} are constrained by the requirement of $Z_{\pm,\pm}$ being a modular-covariant function of q. As usual, we will always have in mind the diagonal theory $M_{\bar{\lambda},\lambda}^{\pm} = \delta_{\bar{\lambda},\lambda}$.

In order to simplify the notation, we shall frequently leave the dependence on the spin structure \pm, \pm implicit.

Flavor-twisted characters. If the chiral algebra has some flavor symmetry U, then it is often useful to introduce flavor-twisted characters (i.e., we turn on fugacities for the Cartan generators; these are roots of \mathfrak{u}). This allows us to organize the modules \mathcal{H}_{λ} into irreducible representations of U so as to have a more transparent understanding of the structure of states therein. To this end, we can define extended characters as

$$\chi_{\lambda}(q,g) := \operatorname{tr}_{\mathcal{H}_{\lambda}}(q^{L_0 - c/24}\rho(g)), \qquad (2.7.8)$$

where $g \in U$ is a symmetry group element and $\rho: g \to \mathcal{H}_{\lambda}$ is its representation on the Hilbert space. The character is a class function, so its dependence on g is only through its conjugacy class.

Regular characters $\chi_{\lambda}(q)$ are obtained from the extended ones $\chi_{\lambda}(q, g)$ by setting g = 1. The former only keep track of the conformal weights of the states in \mathcal{H}_{λ} , while the latter also keeps track of their quantum numbers under U.

Coset CFTs. Whenever the chiral algebra has a subalgebra, one can expand the characters of the former in terms of those of the latter,

$$\chi_{\lambda}(q) = \sum_{\Lambda} b_{\lambda}^{\Lambda}(q) \chi_{\Lambda}(q) , \qquad (2.7.9)$$

where χ_{λ} are the characters of the original chiral algebra, and χ_{Λ} those of the subalgebra.

The key point of this construction is that, if χ_{λ} and χ_{Λ} are both modular covariant, then so are the coefficients b_{λ}^{Λ} . This means that one can think of these coefficients as the characters of a new theory, which we call the *coset CFT*; this is the celebrated GKO coset construction [194]. If at least one of $\chi_{\lambda}, \chi_{\Lambda}$ is a super-character, then so is b_{λ}^{Λ} , and hence the coset is a fermionic CFT.

This new theory, the coset CFT, has characters $b^{\Lambda}_{\lambda}(q)$, and therefore its partition function takes the form

$$Z^{\text{coset}}(q) = \sum_{b} \bar{b}(\bar{q})b(q), \qquad (2.7.10)$$

where we restrict to diagonal partition functions for simplicity. If b is a super-character, the expression above defines the partition function of a fermionic CFT, while if it is a regular character, it defines the partition function of a bosonic CFT.

It should be remarked that, in order to actually calculate the coefficients b_{λ}^{Λ} , it is often unavoidable to turn on flavor fugacities, for otherwise the computation becomes impracticable. One is therefore forced to look at the extended characters $\chi_{\lambda}(q, g)$, $\chi_{\Lambda}(q, g')$, where g is a symmetry group element of the original algebra, and g' its restriction to the subalgebra. Specifically, if the original algebra has flavor symmetry U and its subalgebra has flavor symmetry U' $\subseteq U$, then the character decomposition can be extended to

$$\chi_{\lambda}(q,g) = \sum_{\Lambda} b_{\lambda}^{\Lambda}(q,g'')\chi_{\Lambda}(q,g'), \qquad (2.7.11)$$

where $g \in U$, $g' \in U'$, and g'' is a group element of the flavor symmetry of the coset, namely the commutant of U' inside U.

WZW CFTs. The discussion so far has been rather abstract and general. To be concrete, let us discuss Wess–Zumino–Witten (WZW) theories, which we review next. WZW theories are labelled by a compact Lie group G, which we take to be simple and connected, and a "level" k, an integer that specifies the central extension for the loop algebra of G. We denote the corresponding model by G_k . When G is simply-connected (which we henceforth assume, unless specified otherwise) the chiral algebra is a Kač-Moody algebra:

$$[J_n^a, J_m^b] = f^{ab}_{\ c} J^c_{n+m} + kn\delta_{ab}\delta_{n+m} , \qquad (2.7.12)$$

while if G is not simply-connected, then the chiral algebra is Kač-Moody extended by the simple currents that generate $\pi_1(G)$. WZW theories with $\pi_1(G) = 0$ are always bosonic, while those with $\pi_1(G) \neq 0$ are fermionic if any of the currents that generate $\pi_1(G)$ is a fermion (it has half-integral conformal weight).

The representations of the chiral algebra are required to be unitary with respect to $(J_n^a)^{\dagger} := J_{-n}^a$.

The enveloping algebra of (2.7.12) contains the Virasoro algebra via the Sugawara construction:

$$L_n = \frac{1}{2(k+h)} \sum_{a,m} : J_m^a J_{n-m}^a:, \qquad (2.7.13)$$

such that

$$[L_n, L_m] = (m-n)L_{n+m} + \frac{c(G_k)}{12}n(n^2 - 1)\delta_{n+m}, \qquad c(G_k) := \frac{k}{k+h}\dim(\mathfrak{g})$$

$$[L_n, J_m^a] = -mJ_{n+m}^a.$$
(2.7.14)

Note that this last expression indicates that J_{-n} carries n units of L_0 eigenvalue.

The primaries of the theory are labelled by the integrable representations of \mathfrak{g}_k , to wit, the highest-weight representations $\lambda \in \operatorname{Rep}(\mathfrak{g})$ that satisfy

$$(\theta, \lambda) \le k \,, \tag{2.7.15}$$

with θ the highest root of \mathfrak{g} (the highest weight of the adjoint representation) and (\cdot, \cdot) its Killing form, normalized to $(\theta, \theta) = 2$. The module \mathcal{H}_{λ} is constructed as follows: at lowest grade, one begins with the vacuum states $|\lambda\rangle$, which live in the finite dimensional representation λ generated by J_0^a . On top of these vacua one constructs the excited states. For example, at grade-one one has the states $J_{-1}^a |\lambda\rangle$, which live inside the representation $\theta \otimes \lambda$. At grade-two one has the states $J_{-2}^a |\lambda\rangle$ and $J_{-1}^a J_{-1}^b |\lambda\rangle$, so the states live inside $\theta \otimes \lambda + \theta_{sym}^2 \otimes \lambda$. Etc. Importantly, not all of these states are physical: we must project out the null states, i.e., the states whose norm vanishes. Continuing this way one obtains the character

$$\chi_{\lambda}(q,g) = q^{h_{\lambda}-c/24} (\chi_{\lambda}(g) + \chi_{R_1}(g)q + \chi_{R_2}(g)q^2 + \chi_{R_3}(g)q^3 + \cdots), \qquad (2.7.16)$$

where R_n is the space of physical states at grade n, and $\chi_R(g) := \operatorname{tr}_R(g)$ the finite character of R^{63} . A more convenient way to obtain the characters is the so-called Weyl-Kač formula [195],

$$\chi_{\lambda}(q,g) = \frac{\hat{\chi}_{\lambda+\rho}(q,g)}{\hat{\chi}_{\rho}(q,g)}, \qquad \hat{\chi}_{\lambda}(q,g) := \sum_{\substack{w \in W\\ \alpha^{\vee} \in Q^{\vee}}} \det(w) \, z^{k\alpha^{\vee}+w\lambda} q^{|k\alpha^{\vee}+w\lambda|^2/2k} \,, \tag{2.7.17}$$

where W denotes the Weyl group of \mathfrak{g} , Q^{\vee} its coroot lattice, ρ its Weyl vector, and z is the value of $g \in G$ when conjugated to any maximal torus.

Spin $(n)_1$ **CFT.** A particularly important family of WZW models is Spin $(n)_1$. For n odd this theory has three primaries $0, v, \sigma$ (the scalar, vector, and spinor representation, respectively), and for n even it has four, 0, v, s, c (where s, c denote the two spinors). The corresponding characters read

$$\chi_{0}(q,g) = q^{-n/48} \left[\bullet + \Box q + \left(\bullet + \Box + \Box + \Box \right) q^{2} + \left(\bullet + 3\Box + \Box + \Box + \Box + \Box \right) q^{3} + \cdots \right]$$

$$\chi_{v}(q,g) = q^{-n/48+1/2} \Box \left[\bullet + \bullet q + \left(\bullet + \Box \right) q^{2} + \left(\bullet + 2\Box + \Box \right) q^{3} + \cdots \right]$$

$$+ q^{-n/48+3/2} \left[\Box + \Box + \Box \right] q^{2} + \left(\bullet + \Box + \Box \right) q^{2} + \left(\bullet + \Box + \Box \right) q^{2} + \left(2\Box + \Box + \Box + \Box \right) q^{2} + \left(3\Box + 2\Box + 3\Box + \Box + \Box + \Box \right) q^{3} + \cdots \right]$$

$$\chi_{\sigma}(q,g) = q^{n/24} \sigma \left[\bullet + \Box q^{2} + \left(\bullet + \Box + \Box \right) q^{3} + \left(\bullet + \Box + \Box \right) q^{3} + \left(\bullet + \Box + \Box \right) q^{4} + \cdots \right]$$

$$+ q^{n/24+1} \overline{\sigma} \left[\Box + \Box q + \left(\Box + \Box \right) q^{2} + \left(2\Box + \Box + \Box + \Box \right) q^{3} + \cdots \right]$$

⁶³By definition, $\chi_R(g) := \sum_{\lambda \in \Omega(R)} z^{\lambda}$, where $\Omega(R)$ is the space of weights of the representation R, with multiplicities.

$$\equiv q^{n/24} \left[\sigma + (\sigma + \overleftarrow{\Box})q + (2\sigma + \dot{\Box} + 2\overrightarrow{\Box})q^2 + (4\sigma + \dot{\overline{\Box}} + 2\dot{\Box} + 4\overline{\dot{\Box}} + \underline{\Box})q^3 + \cdots \right],$$

where σ denotes the spinor of Spin(n) when n is odd, and any of the two spinors when n is even (in which case $\bar{\sigma}$ denotes the conjugate spinor, obtained by permuting the last two Dynkin labels). Here, a Young diagram stands for the finite character $\chi_R(g)$ associated to the representation R of $\mathfrak{so}(n)$, and a dot \dot{R} is a short-hand notation for the representation whose highest weight is given by $\lambda_{\dot{R}} := \lambda_R + \omega_{\lfloor n/2 \rfloor}$ (so for example $\dot{\Box} = (1, 0, \ldots, 0, 1)$).

These expressions make it manifest how the different states of $\text{Spin}(n)_1$ appear at each level. For example, in χ_0 at grade n = 2 the states live inside

$$J_{-2}^{a}|0\rangle + J_{-1}^{a}J_{-1}^{b}|0\rangle \subseteq \Box + (\Box \otimes \Box)_{\text{sym}} = \left(\bullet + \Box + \Box + \Box\right) + \Box, \qquad (2.7.19)$$

and one can check that the representations in parentheses are physical and the isolated one is null (its norm vanishes).

 $SO(n)_1$ CFT. A related – and also very important – WZW model is $SO(n)_1$. This theory is obtained from $Spin(n)_1$ through fermionization. There are two key properties of $SO(n)_1$ that make it special. First, it is a *fermionic* theory, meaning that its characters and partition functions depend on the choice of spin structure. Second, it is a *holomorphic* theory, meaning that its partition function factorizes as $Z_{\pm,\pm} = |d_{\pm,\pm}|^2$ (as opposed to non-holomorphic theories whose partition function is a sum of such terms). In other words, $SO(n)_1$ has a *unique* primary (for fixed spin structure). Following the convention of [150] we denote this unique character as $d_{\pm,\pm}$.

The characters of $SO(n)_1$ can be obtained from those of $Spin(n)_1$ (cf. equation (2.7.18)) via the standard bosonization/fermionization dictionary:

$$d_{\text{NS-X}}(q,g) = \chi_0(q,g) \pm \chi_v(q,g)$$

$$n \text{ odd:} \quad d_{\text{R-NS}}(q,g) = \sqrt{2}\chi_\sigma(q,g)$$

$$d_{\text{R-R}}(q,g) = 0$$

$$n \text{ even:} \quad d_{\text{R-X}}(q,g) = \chi_s(q,g) \pm \chi_c(q,g) .$$

$$(2.7.20)$$

 $U(1)_k$ CFT. So far we have described WZW models for simple groups. The case of U(1) requires a separate discussion. By $U(1)_k$, with k even, we mean a free compact boson at radius $R^2 = k$. This has a chiral U(1) flavor symmetry; if we turn on a fugacity $z \in U(1)$ for this symmetry, the characters of this CFT are

$$\chi_{\ell}(q,z) = \eta(q)^{-1} \sum_{u \in \mathbb{Z}} q^{\frac{1}{2}k(u+\ell/k)^2} z^{\sqrt{k}u+\ell/\sqrt{k}}, \qquad (2.7.21)$$
with $\ell = 0, 1, \dots, k - 1$, and where η is the Dedekind function.

When k is odd, by $U(1)_k$ we mean the theory $U(1)_{4k}$ extended by the vertex operator of weight k/2, i.e., by $\ell = 2k$. As this weight is half-integral, the operator is a fermion and the extension results in a fermionic CFT. Its super-characters can be obtained for example by following the rules of [1]:

$$\chi_{\text{NS-X};\ell}^{(k)}(q,z) = \chi_{2\ell}^{(4k)}(q,z) \pm \chi_{2\ell+2k}^{(4k)}(q,z)$$

$$\chi_{\text{R-X};\ell}^{(k)}(q,z) = \chi_{2\ell+1}^{(4k)}(q,z) \pm \chi_{2\ell+1+2k}^{(4k)}(q,z).$$
(2.7.22)

These characters have also been discussed in e.g. [180]. The special case k = 1 is equivalent to a free fermion theory, $U(1)_1 = SO(2)_1$, as can be checked by comparing the corresponding super-characters; this is nothing but the trivial statement that one complex fermion equals two real fermions. More generally, $U(n)_1$ denotes the CFT of *n* complex fermions, and we shall use the notation $U(n)_1 = SO(2n)_1$ interchangeably. (The former is more natural when the free fermions are associated to a complex representation of the gauge group *G*).

WZW coset models. We are now ready to discuss the class of models of interest, namely cosets of the form $SO(n)_1/G_k$, which appear at the deep infrared of QCD theories with gauge group G.

The CFT SO $(n)_1/G_k$ is obtained by embedding G_k into SO $(n)_1$. As above, this embedding gives rise to a character decomposition of the form

$$d_{\pm,\pm} = \sum_{\lambda} b_{\lambda}^{\pm,\pm} \chi_{\lambda} , \qquad (2.7.23)$$

where $d_{\pm,\pm}$ are the characters of $\mathrm{SO}(n)_1$, and χ_{λ} those of G_k . The denominator theory G_k is allowed to be fermionic, in which case it is understood that $\chi_{\lambda} = \chi_{\lambda}^{\pm,\pm}$ is a super-character at spin structure \pm, \pm . In any case, whether G_k is fermionic or not, the coset is a fermionic theory, because the numerator $\mathrm{SO}(n)_1$ is fermionic. Consequently, the coefficients $b_{\lambda}^{\pm,\pm}$ depend on the spin structure, as indicated by the superscript. These coefficients are the super-characters of the coset CFT $\mathrm{SO}(n)_1/G_k$, and they determine the dynamics of QCD in the infrared. In particular, at low energies the partition function of QCD becomes

$$Z_{\pm,\pm} = \sum_{\lambda} |b_{\lambda}^{\pm,\pm}|^2 \,. \tag{2.7.24}$$

In this sense, the whole problem of describing the strongly coupled dynamics of QCD has been reduced to the task of finding the coefficients b_{λ} in (2.7.23). While for generic cosets this is a computationally demanding task, for cosets of the form $SO(n)_1/G_k$ there is a substantial simplification: the numerator $SO(n)_1$ is in fact equivalent to *n* free Majorana fermions.

This free fermion representation can be exploited as follows. Consider the coset $SO(n)_1/G_k$, where $n = \dim(R)$ and k = I(R) with embedding $G \subseteq SO(n)$ via the representation R. In the Neveu-Schwartz sector the free fermions have half-integral modding $\psi_{r+1/2}$, while in the Ramond sector they have integral modding ψ_r . These modes are independent, so the SO $(\dim(R))_1$ partition function is just the product of the individual partition functions over all $r \in \mathbb{N}$. The fermions $\psi_{r+1/2}, \psi_r$ all generate *R*-modules except for the Ramond zero modes ψ_0 , which generate a spinor module. With this, the different partition functions of SO $(\dim(R))_1$ read

$$d_{\rm NS-X}(q, g; R) = q^{-\dim(R)/48} \prod_{r=0}^{\infty} \prod_{\lambda \in \Omega(R)} 1 \pm z^{\lambda} q^{r+1/2}$$

dim(R) odd : $d_{\rm R-NS}(q, g; R) = \sqrt{2} q^{\dim(R)/24} \chi_{\sigma}(g) \prod_{r=1}^{\infty} \prod_{\lambda \in \Omega(R)} 1 + z^{\lambda} q^{r}$
 $d_{\rm R-R}(q, g; R) = 0$
dim(R) even : $d_{\rm R-X}(q, g; R) = q^{\dim(R)/24} (\chi_{s}(g) \pm \chi_{c}(g)) \prod_{r=1}^{\infty} \prod_{\lambda \in \Omega(R)} 1 \pm z^{\lambda} q^{r}$, (2.7.25)

where $g \in G$ is the restriction of any flavor $SO(\dim(R))$ symmetry to the subgroup G, and z its value on any maximal torus.

One can use these expressions to compute the first few terms of the q-expansion of $d_{\pm,\pm}$. These terms are then reorganized into $G_{I(R)}$ characters, whose q-expansion can be obtained with e.g. the Weyl-Kač formula (2.7.17). The characters of the coset $SO(\dim(R))_1/G_{I(R)}$ are identified with the coefficients of this reorganized series. Computer software is often instrumental in these computations, for example the LieART Mathematica package [196]. The extensive tables of Lie algebras, representations, and branchings in [197] can also come in handy.

Topological cosets and conformal embeddings. A special role is played by cosets $SO(n)_1/G_k$ where G_k embeds into $SO(n)_1$ conformally, i.e., when the central charge of $SO(n)_1/G_k$ vanishes. We argued in the main text that this happens if and only if the QCD theory with group G is gapped. When this happens, the infrared theory becomes a trivial CFT. That being said, the coset is not an empty theory, even though it has no local degrees of freedom; in other words, it is a topological QFT. The low energy dynamics of gapped theories is entirely contained in the topological degrees of freedom carried by the topological coset $SO(n)_1/G_k$.

By topological invariance, all observables of such cosets become q-independent, so the branching functions b_{λ} are just numbers instead of functions of q. Note that topological invariance is just a special case of conformal invariance: TQFTs are invariant under all diffeomorphisms instead of just the conformal ones. This means, in particular, that the formula (2.7.24) is still valid for TQFTs. In this case, as $L_0 \equiv 0$, the partition function

actually computes the total number of states in the theory, which has a finite-dimensional Hilbert space:

$$Z_{\pm,-} = \operatorname{tr}_{\mathcal{H}_{\pm}}(1) \equiv \text{bosons plus fermions in } \mathcal{H}_{\pm}$$

$$Z_{\pm,+} = \operatorname{tr}_{\mathcal{H}_{\pm}}(-1)^{F} \equiv \text{bosons minus fermions in } \mathcal{H}_{\pm}$$
(2.7.26)

or, equivalently,

number of bosons in
$$\mathcal{H}_{\pm} = \frac{1}{2}(Z_{\pm,-} + Z_{\pm,+}) \equiv \sum_{\lambda} \frac{1}{2}(|b_{\lambda}^{\pm,-}|^2 + |b_{\lambda}^{\pm,+}|^2)$$

number of fermions in $\mathcal{H}_{\pm} = \frac{1}{2}(Z_{\pm,-} - Z_{\pm,+}) \equiv \sum_{\lambda} \frac{1}{2}(|b_{\lambda}^{\pm,-}|^2 - |b_{\lambda}^{\pm,+}|^2).$ (2.7.27)

In what follows we shall work out several examples in some detail in order to illustrate some of the previous considerations.

2.7.1 Examples of topological cosets

Here we demonstrate the coset construction for the conformal embedding $SO(8)_1 \supset SU(3)_3$. As $c = 8/2 - 3 \cdot 8/6 = 0$, the resulting theory is topological, i.e., it has a finite-dimensional Hilbert space. In order to find this Hilbert space we need to decompose the SO(8) characters into SU(3) characters. The former are given by (2.7.25):

$$\chi_{0}(q,g) = q^{-1/6} \bigg[\mathbf{1} + (\mathbf{8} + \mathbf{10} + \overline{\mathbf{10}}) q + (\mathbf{1} + 4 \mathbf{8} + \mathbf{10} + \overline{\mathbf{10}} + 3 \mathbf{27}) q^{2} + (2 \mathbf{1} + 10 \mathbf{8} + 6 \mathbf{10} + 6 \overline{\mathbf{10}} + 6 \mathbf{27} + 2 \mathbf{35} + 2 \overline{\mathbf{35}} + \mathbf{64}) q^{3} + \cdots \bigg]$$

$$\chi_{v}(q,g) = q^{1/3} \bigg[\mathbf{8} + (\mathbf{1} + 2 \mathbf{8} + \mathbf{10} + \overline{\mathbf{10}} + \mathbf{27}) q + (2 \mathbf{1} + 6 \mathbf{8} + 3 \mathbf{10} + 3 \overline{\mathbf{10}} + 4 \mathbf{27} + \mathbf{35} + \overline{\mathbf{35}}) q^{2} + \cdots \bigg],$$

(2.7.28)

and, by triality,

$$\chi_s(q,g) = \chi_c(q,g) = \chi_v(q,g).$$
(2.7.29)

We next reorganize these characters in terms of $SU(3)_3$ characters. The characters of $SU(3)_3$ are given by

$$\chi_{1}(q,g) = q^{-1/6} \bigg[\mathbf{1} + \mathbf{8}q + \big(\mathbf{1} + 2\,\mathbf{8} + \mathbf{27}\big)q^{2} \\ + \big(2\,\mathbf{1} + 4\,\mathbf{8} + 2\,\mathbf{10} + 2\,\overline{\mathbf{10}} + 2\,\mathbf{27} + \mathbf{64}\big)q^{3} + \cdots \bigg] \\\chi_{3}(q,g) = q^{1/18} \bigg[\mathbf{3} + \big(\mathbf{3} + \mathbf{6} + \mathbf{15}\big)q + \big(\mathbf{3}\,\mathbf{3} + 2\,\mathbf{6} + \mathbf{3}\,\mathbf{15} + \mathbf{24} + \mathbf{42}\big)q^{2} + \cdots \bigg]$$

$$\begin{split} \chi_{6}(q,g) &= q^{7/18} \bigg[6 + \big(3 + 6 + 15 + 24 \big) q \\ &+ \big(2 \, 3 + 4 \, 6 + 3 \, 15 + 15' + 3 \, 24 + 42 \big) q^{2} + \cdots \bigg] \\ \chi_{8}(q,g) &= q^{1/3} \bigg[8 + \big(1 + 2 \, 8 + 10 + \overline{10} + 27 \big) q \\ &+ \big(2 \, 1 + 6 \, 8 + 3 \, 10 + 3 \, \overline{10} + 4 \, 27 + 35 + \overline{35} \big) q^{2} + \cdots \bigg] \\ \chi_{10}(q,g) &= q^{5/6} \bigg[10 + \big(8 + 10 + 27 \big) q \\ &+ \big(3 \, 8 + 3 \, 10 + \overline{10} + 2 \, 27 + 35 + \overline{35} \big) q^{2} + \cdots \bigg] \\ \chi_{15}(q,g) &= q^{13/18} \bigg[15 + \big(3 + 6 + 2 \, 15 + 15' + 24 \big) q \\ &+ \big(3 \, 3 + 3 \, 6 + 6 \, 15 + 2 \, 15' + 21 + 3 \, 24 + 2 \, 42 \big) q^{2} + \cdots \bigg] \,. \end{split}$$

By comparing (2.7.28) to (2.7.30) it is easily checked that

$$d_{\text{NS-NS}} = \chi_{1} + \chi_{8} + \chi_{10} + \chi_{\overline{10}}$$

$$d_{\text{NS-R}} = \chi_{1} - \chi_{8} + \chi_{10} + \chi_{\overline{10}}$$

$$d_{\text{R-NS}} = 2\chi_{8}$$

$$d_{\text{R-R}} = 0,$$

$$(2.7.31)$$

which implies that the NS sector of $SO(8)_1/SU(3)_3$ has four bosons and no fermions, and the R sector has two and two, i.e., $\mathcal{H}_{NS} = \mathbb{C}^{4|0}$ and $\mathcal{H}_{R} = \mathbb{C}^{2|2}$.

It is interesting to note that, out of the four bosons in \mathcal{H}_{NS} , one of them (the one corresponding to $b_{\mathbf{8}}$) is charged under $(-1)^{F_R}$, while the other three are neutral. In the full non-chiral theory, this state is a boson because it comes from $\bar{b}_{\mathbf{8}}b_{\mathbf{8}}$, which is charged under both $(-1)^{F_L}$ and $(-1)^{F_R}$ (and is therefore neutral under $(-1)^F = (-1)^{F_L}(-1)^{F_R}$).

In the previous example we found that \mathcal{H}_{R} was supersymmetric (it contains the same number of bosons as fermions), which was a consequence of d_{R-R} vanishing. In order to show that this is not always the case, we next describe an example where \mathcal{H}_{R} is *not* supersymmetric.

Consider the coset $SO(16)_1/Spin(9)_2$. Using (2.7.25), the characters of the numerator are

$$\begin{split} \chi_0(q,g) &= q^{-1/3} \big(\mathbf{1} + (\mathbf{36} + \mathbf{84})q \\ &+ (\mathbf{9} + \mathbf{16} + \mathbf{36} + \mathbf{44} + \mathbf{84} + 2\,\mathbf{126} + \mathbf{231} + \mathbf{495} + \mathbf{924})q^2 \\ &+ (2\,\mathbf{9} + \mathbf{16} + 5\,\mathbf{36} + \mathbf{44} + 5\,\mathbf{84} + 3\,\mathbf{126} + 3\,\mathbf{231} + \mathbf{495} \\ &+ 3\,\mathbf{594} + \mathbf{910} + 3\,\mathbf{924} + \mathbf{1650} + \mathbf{2457} + 2\,\mathbf{2772})q^3 + \cdots \big) \\ \chi_v(q,g) &= q^{1/6} \big(\mathbf{16} + (\mathbf{16} + \mathbf{128} + \mathbf{432})q \end{split}$$

$$\begin{aligned} &+ \left(3\,\mathbf{16} + 3\,\mathbf{128} + 3\,\mathbf{432} + 576 + \mathbf{672} + 768 + \mathbf{2560}\right)q^2 \\ &+ \left(7\,\mathbf{16} + 9\,\mathbf{128} + 9\,\mathbf{432} + 4\,\mathbf{576} + 2\,\mathbf{672} + 5\,768 \\ &+ \mathbf{1920} + 4\,\mathbf{2560} + \mathbf{4608} + \mathbf{4928} + 2\,\mathbf{5040}\right)q^3 + \cdots \right) \\ \chi_s(q,g) &= q^{2/3}\left(\mathbf{44} + \mathbf{84} + \left(9 + \mathbf{36} + \mathbf{44} + \mathbf{84} + \mathbf{126} + \mathbf{231} + \mathbf{594} + \mathbf{924}\right)q \qquad (2.7.32) \\ &+ \left(\mathbf{1} + 2\,\mathbf{9} + 3\,\mathbf{36} + 3\,\mathbf{44} + 4\,\mathbf{84} + 3\,\mathbf{126} + 3\,\mathbf{231} + \mathbf{495} \\ &+ 3\,\mathbf{594} + \mathbf{910} + 4\,\mathbf{924} + \mathbf{1650} + \mathbf{1980} + 2\mathbf{457} + 2772\right)q^2 \\ &+ \left(2\,\mathbf{1} + 6\,\mathbf{9} + \mathbf{8}\,\mathbf{36} + 6\,\mathbf{44} + \mathbf{10}\,\mathbf{84} + \mathbf{10}\,\mathbf{126} + 2\,\mathbf{156} + 9\,\mathbf{231} \\ &+ 3\,\mathbf{495} + \mathbf{11}\,\mathbf{594} + 3\,\mathbf{910} + \mathbf{11}\,\mathbf{924} + 5\,\mathbf{1650} + 2\,\mathbf{1980} \\ &+ 3\,\mathbf{2457} + \mathbf{2574} + 5\,\mathbf{2772} + 3\,\mathbf{3900} + 2\,\mathbf{4158} + 9009 + \mathbf{15444}\right)q^3 + \cdots \right) \\ \chi_c(q,g) &= q^{2/3}\left(\mathbf{128} + \left(\mathbf{16} + 2\,\mathbf{128} + \mathbf{432} + 576 + 768\right)q \\ &+ \left(3\,\mathbf{16} + 6\,\mathbf{128} + 4\,\mathbf{432} + 3\,\mathbf{576} + 672 + 3\,768 + 2\,\mathbf{2560} + 50\mathbf{40}\right)q^2 \\ &+ \left(8\,\mathbf{16} + \mathbf{16}\,\mathbf{128} + \mathbf{13}\,\mathbf{432} + 9\,\mathbf{576} + 4\,\mathbf{672} + 9\,768 + \mathbf{1920} \\ &+ 8\,\mathbf{2560} + 2\,\mathbf{4608} + \mathbf{4928} + 5\,\mathbf{5040} + \mathbf{9504} + \mathbf{12672}\right)q^3 + \cdots \right). \end{aligned}$$

In order to express these in terms of the characters of the denominator, we need the ${\rm Spin}(9)_2$ characters:

$$\begin{split} \chi_1(q,g) &= q^{-1/3} \Big(1 + 36 \, q + (1 + 36 + 44 + 126 + 495) q^2 \\ &\quad + (1 + 436 + 44 + 84 + 126 + 495 + 2594 + 910 + 2772) q^3 + \cdots \Big) \\ \chi_{16}(q,g) &= q^{1/6} \Big(16 + (16 + 128 + 432) q \\ &\quad + (316 + 3128 + 3432 + 576 + 672 + 768 + 2560) q^2 \\ &\quad + (716 + 9128 + 9432 + 4576 + 2672 + 5768 \\ &\quad + 1920 + 42560 + 4608 + 4928 + 25040) q^3 + \cdots \Big) \\ \chi_{44}(q,g) &= q^{2/3} \Big(44 + (36 + 44 + 594) q \\ &\quad + (1 + 236 + 344 + 126 + 495 + 2594 + 910 + 924 + 1980) q^2 \\ &\quad + (1 + 636 + 544 + 84 + 3126 + 231 + 2495 + 7594 + 3910 \\ &\quad + 2924 + 1980 + 22772 + 3900 + 4158 + 9009) q^3 + \cdots \Big) \\ \chi_{84}(q,g) &= q^{2/3} \Big(84 + (9 + 84 + 126 + 231 + 924) q \\ &\quad + (29 + 36 + 484 + 2126 + 3231 + 594 + 3924 + 1650 + 2457 + 2772) q^2 \\ &\quad + (1 + 69 + 236 + 44 + 984 + 7126 + 2156 + 8231 + 495 + 4594 + 9924 \\ &\quad + 51650 + 1980 + 32457 + 2574 + 32772 + 23900 + 4158 + 15444) q^3 + \cdots \Big) \\ \chi_{128}(q,g) &= q^{2/3} \Big(128 + (16 + 2128 + 432 + 576 + 768) q \\ &\quad + (316 + 6128 + 13432 + 9576 + 4672 + 9768 + 1920 \\ &\quad + 82560 + 24608 + 4928 + 55040 + 9504 + 12672) q^3 + \cdots \Big) \,, \end{split}$$

in terms of which one can write

$$d_{\rm NS-NS}(q,g) = \chi_{1}(q,g) + \chi_{16}(q,g) + \chi_{84}(q,g)$$

$$d_{\rm NS-R}(q,g) = \chi_{1}(q,g) - \chi_{16}(q,g) + \chi_{84}(q,g)$$

$$d_{\rm R-NS}(q,g) = \chi_{44}(q,g) + \chi_{84}(q,g) + \chi_{128}(q,g)$$

$$d_{\rm R-R}(q,g) = \chi_{44}(q,g) + \chi_{84}(q,g) - \chi_{128}(q,g).$$
(2.7.34)

These affine branching rules imply that the Hilbert spaces of $SO(16)_1/Spin(9)_2$ are $\mathcal{H}_{NS} = \mathcal{H}_R = \mathbb{C}^{3|0}$. As promised, \mathcal{H}_R is not supersymmetric. While the unextended character $d_{R-R}(q) = 0$ vanishes, the extended one $d_{R-R}(q,g) = \chi_{84}(q,g) + \chi_{44}(q,g) - \chi_{128}(q,g)$ is non-zero, and hence the coset does not have the same number of bosons and fermions.

In section 2.5 we attributed the supersymmetry of the Ramond sector to certain 't Hooft anomalies. These anomalies are present when the number of quarks is odd. In the theory $SO(16)_1/Spin(9)_2$ the number of fermions is even, 16, so there is no reason to expect that the Ramond sector is supersymmetric – and indeed it is not.

2.7.2 Example of a non-topological coset

Here we study the coset $\text{Spin}(7)_1/\text{SU}(2)_{28}$, which has $c = 7/2 - 3 \times 28/30 \equiv 7/10$. This is non-zero so the coset is non-topological, i.e., it is a traditional CFT. We begin by writing the characters of the numerator $\text{Spin}(7)_1$:

$$\begin{split} \chi_0(q,g) &= q^{-7/48} \bigg[\mathbf{1} + (\mathbf{3} + \mathbf{7} + \mathbf{11})q + (2\mathbf{1} + \mathbf{3} + 2\mathbf{5} + 2\mathbf{7} + 2\mathbf{9} + \mathbf{11} + 2\mathbf{13})q^2 \\ &+ (2\mathbf{1} + 5\mathbf{3} + 4\mathbf{5} + 8\mathbf{7} + 5\mathbf{9} + 6\mathbf{11} + 4\mathbf{13} + 2\mathbf{15} + \mathbf{17} + \mathbf{19})q^3 + \cdots \bigg] \\ \chi_v(q,g) &= q^{17/48} \bigg[\mathbf{7} + (\mathbf{1} + \mathbf{5} + 2\mathbf{7} + \mathbf{9} + \mathbf{13})q \\ &+ (\mathbf{1} + 2\mathbf{3} + 3\mathbf{7} + \mathbf{11} + 3\mathbf{5} + 2\mathbf{7} + 3\mathbf{9} + 2\mathbf{11} + 2\mathbf{13} + \mathbf{15} + \mathbf{17})q^2 \\ &+ (4\mathbf{1} + 5\mathbf{3} + 9\mathbf{5} + \mathbf{12}\mathbf{7} + \mathbf{10}\mathbf{9} + 8\mathbf{11} + 7\mathbf{13} + 2\mathbf{15} + 3\mathbf{17} + \mathbf{19})q^3 + \cdots \bigg] \\ \chi_\sigma(q,g) &= q^{7/24} \bigg[\mathbf{1} + \mathbf{7} + (\mathbf{1} + \mathbf{3} + \mathbf{5} + 2\mathbf{7} + \mathbf{9} + \mathbf{11} + \mathbf{13})q \\ &+ (2\mathbf{1} + 3\mathbf{3} + 4\mathbf{5} + 6\mathbf{7} + 4\mathbf{9} + 4\mathbf{11} + 3\mathbf{13} + \mathbf{15} + \mathbf{17})q^2 + \\ &+ (2\mathbf{1} + 8\mathbf{3} + \mathbf{115} + \mathbf{167} + \mathbf{139} + \mathbf{1111} \\ &+ \mathbf{10}\mathbf{13} + 5\mathbf{15} + 3\mathbf{17} + 2\mathbf{19})q^3 + \cdots \bigg] \,. \end{split}$$

Similarly, the characters of the denominator $\mathrm{SU}(2)_{28}$ are

$$\begin{split} \chi_1(q,g) &= q^{-7/60} (1+3q+(1+3+5)q^2+(1+33+5+7)q^3+\cdots) \\ \chi_7(q,g) &= q^{17/60} (7+(5+7+9)q+(3+25+37+29+11)q^2 \\ &+ (1+23+55+67+59+211+13)q^3+\cdots) \\ \chi_{11}(q,g) &= q^{53/60} (11+(9+11+13)q+(7+29+311+213+15)q^2 \\ &+ (5+27+59+611+513+215+17)q^3+\cdots) \\ \chi_{13}(q,g) &= q^{77/60} (13+(11+13+15)q+(9+211+313+215+17)q^2 \\ &+ (7+29+511+613+515+217+19)q^3+\cdots) \\ \chi_{17}(q,g) &= q^{137/60} (17+(15+17+19)q+(13+215+317+219+21)q^2 \\ &+ (11+213+515+617+519+221+23)q^3+\cdots) \\ \chi_{19}(q,g) &= q^{173/60} (19+(17+19+21)q+(15+217+319+221+23)q^2 \\ &+ (13+215+517+619+521+223+25)q^3+\cdots) \\ \chi_{23}(q,g) &= q^{257/60} (23+(21+23+25)q+(19+221+323+225+27)q^2 \\ &+ (17+219+521+623+525+227+29)q^3+\cdots) \\ \chi_{29}(q,g) &= q^{413/60} (29+(27+29)q+(25+227+229+31)q^2 \\ &+ (23+225+427+429+231)q^3+\cdots) \\ \dots \end{split}$$

Given these expressions one can check that the ${\rm Spin}(7)_1$ characters decompose into ${\rm SU}(2)_{28}$ characters as

According to the coset prescription we are instructed to regard the q-dependent coefficients as the characters of a new theory, which we denote as

$$\chi_{1,1}(q) = q^{-7/240}(1+q^2+q^3+2q^4+2q^5+4q^6+4q^7+7q^8+\cdots)$$

$$\chi_{1,3}(q) = q^{137/240}(1+q+2q^2+2q^3+4q^4+5q^5+7q^6+9q^7+\cdots)$$

$$\chi_{1,4}(q) = q^{353/240}(1+q+2q^2+2q^3+3q^4+4q^5+6q^6+7q^7+\cdots)$$

$$\chi_{1,2}(q) = q^{17/240}(1+q+q^2+2q^3+3q^4+4q^5+6q^6+8q^7+\cdots)$$

$$\chi_{2,1}(q) = q^{49/120}(1+q+q^2+2q^3+3q^4+4q^5+6q^6+8q^7+\cdots)$$

$$\chi_{2,2}(q) = q^{1/120}(1+q+2q^2+3q^3+4q^4+6q^5+8q^6+11q^7+\cdots),$$

(2.7.38)

which we recognize as the Virasoro characters of the minimal model M(4,5) with central charge c = 7/10:

$$\chi_{r,s}(q) = k_s(q) - k_{-s}(q), \qquad k_s(q) := \eta(q)^{-1} \sum_{n \in \mathbb{Z}} q^{((2np(p+1)+r(p+1)-sp)^2)/4p(p+1)}.$$
(2.7.39)

Fermionizing, we get the super-characters

$$\chi^{\text{NS-X}}(q) = \chi_{1,1} \pm \chi_{1,4} \quad \& \quad \chi_{1,3} \pm \chi_{1,2} \chi^{\text{R-NS}}(q) = \chi_{2,1} \quad \& \quad \chi_{2,2} ,$$
(2.7.40)

which, nonsurprisingly, are the characters of the fermionic c = 7/10 minimal model. In other words, the infrared theory of SU(2) + 7 is the fermionic M(4, 5) minimal model, with coset realization $SO(7)_1/SU(2)_{28}$.

Another interesting example is $SO(8)_1/Spin(7)_1$. The numerator and denominator algebras are both of the type $\mathfrak{so}(n)_1$, so the characters are straightforward. By working out the decomposition one obtains

$$d_{\rm NS-X} = \chi_0^{(7)} \chi_0^{(1)} + \chi_v^{(7)} \chi_v^{(1)} \pm \chi_\sigma^{(7)} \chi_\sigma^{(1)} d_{\rm R-X} = \chi_0^{(7)} \chi_v^{(1)} + \chi_v^{(7)} \chi_0^{(1)} \pm \chi_\sigma^{(7)} \chi_\sigma^{(1)} ,$$
(2.7.41)

where $\chi^{(7)}$ are the characters of Spin(7)₁ and $\chi^{(1)}$ are the Ising characters. Therefore, the infrared chiral algebra of Spin(8) + **8** is the (bosonic) Ising CFT.

2.8 Abelian theories

Consider a QED theory with N_c photons and N_F non-chiral Dirac fermions:

$$\mathcal{L} = \frac{1}{2} \sum_{i,j=1}^{N_c} g_{ij}^{-2} \mathrm{d}a_i \wedge \star \mathrm{d}a_j + \frac{i}{2\pi} \sum_{i=1}^{N_c} \theta_i \mathrm{d}a_i + \sum_{I=1}^{N_F} \bar{\psi}_I \not{D}_I \psi^I , \qquad (2.8.1)$$

where the gauge fields are normalized to integral periods:

$$\int \frac{\mathrm{d}a_i}{2\pi} \in \mathbb{Z} \,, \tag{2.8.2}$$

and D denotes the covariant derivative

$$D_I = \partial + i \sum_{i=1}^{N_c} Q_{Ii} a_i ,$$
 (2.8.3)

with $Q_{Ii} \in \mathbb{Z}$ the charge of the field ψ_I under U(1)_i.

In order to simplify the notation we will often think of indexed objects g_{ij} , θ_i , a_i , Q_{Ii} as arrays of suitable shape, e.g., g is a square matrix of dimension $N_c \times N_c$, θ a row vector of dimension $1 \times N_c$, Q a rectangular matrix of dimension $N_F \times N_c$, etc.

In (2.8.1), g_{ij} , θ_i denote the coupling constants of the model. These are not all independent: for example, one can always perform linear changes of basis in photon space $a \mapsto Aa$, under which $g^{-2} \mapsto A^t g^{-2}A$, $\theta \mapsto \theta A$. Here $A \in GL(N_c, \mathbb{Z})$ is an integral unimodular matrix so as to preserve the quantization condition (2.8.2). Under this redefinition, the matrix of charges Q transforms as $Q \mapsto QA$. We will come back to this momentarily.

Classically, the model has flavor symmetry $U(1)^{N_F}$ corresponding to the axial rotations of the fermions $\psi_I \mapsto e^{i\alpha_I \gamma^3} \psi_I$. This flavor symmetry may be enhanced to a non-abelian group if some of the rows of Q are equal. These classical symmetries often have a mixed anomaly with the gauge group $U(1)^{N_c}$. Specifically, consider the flavor subgroup $U(1)_F \subset U(1)^{N_F}$ defined by a certain row vector of integers $n = (n_1, n_2, \ldots, n_{N_F})$ such that, under $\alpha \in U(1)_F$, the fermions rotate with angle $\alpha_I = \alpha n_I$. Under $U(1)_F$ the left-movers have charge $+n_I$, while the right-movers have charge $-n_I$. Under $U(1)_i$, both chiralities have charge Q_{Ii} . Therefore, the mixed flavor-gauge anomaly is

$$U(1)_F - U(1)_i: \qquad \sum_{I=1}^{N_F} 2n_I Q_{Ii} \equiv 2(nQ)_i. \qquad (2.8.4)$$

This mixed anomaly has two (dual) interpretations: first, it stems from the fact that, under $U(1)_F$ rotations, the theta terms in (2.8.1) shift, so these terms are rendered unphysical; and second, it corresponds to the fact that the current that generates $U(1)_F$ is not conserved, and hence $U(1)_F$ is not an actual symmetry. Let us analyze both these points in turn.

Theta terms. The mixed anomaly between $U(1)^{N_F}$ and $U(1)^{N_c}$ can be understood as the statement that, under $U(1)^{N_F}$ rotations, the theta terms in (2.8.1) shift. In particular, under the $U(1)_F$ subgroup specified by the integer vector n, one has

$$U(1)_F \colon \theta \mapsto \theta + 2\alpha n Q \,. \tag{2.8.5}$$

This means that θ are unphysical parameters, as they can generically be rotated away. More precisely, the linear combination θv , where v is a $N_c \times 1$ column vector, shifts as $\Delta(\theta v) = 2\alpha nQv$, and this is zero for all $\alpha \in U(1)_F$ if and only if $Qv \equiv 0$. In other words, the physical theta parameters of the system are in correspondence with the vectors that are annihilated by Q on the right:

• In the non-chiral QED system (2.8.1), the space of physical theta parameters is given by the kernel (right-null-space) of Q. The linear combination θv is physical if and only if $v \in \ker Q$. **Axial currents.** The mixed anomaly between $U(1)^{N_F}$ and $U(1)^{N_c}$ can also be understood as the statement that, in the quantum theory, some of the currents that generate $U(1)^{N_F}$ are not conserved. In particular, the subgroup $U(1)_F$ specified by the integer vector n is generated by the current

$$J_n := \sum_{I=1}^{N_F} n_I J_I \,, \tag{2.8.6}$$

where $J_I^{\mu} = \bar{\psi}_I \gamma^3 \gamma^{\mu} \psi_I$ is the current that generates axial rotations of the *I*-th fermion. The mixed U(1)_F-U(1)_i anomaly violates the conservation law for J_n if and only if 2nQ is non-zero. In other words, there are as many conserved axial currents as there are row vectors that are annihilated by Q on the left:

• In the non-chiral QED system (2.8.1), the algebra of axial flavor symmetries is given by the cokernel (left-null-space) of Q. The linear combination $n \cdot J$ is conserved if and only if $n \in \operatorname{coker} Q$.

Hermite Normal form. We noticed above that one can always perform changes of basis in photon space according to $a \mapsto Aa$, where A is a matrix in $GL(N_c, \mathbb{Z})$. This change of basis redefines the matrix of charges according to $Q \mapsto QA$. One can always use this freedom to put Q in (column) Hermite normal form, namely $Q = \tilde{Q}A$ where \tilde{Q} is lower triangular and columns of zeros, if any, are to the far right (see figure 2.3).



Figure 2.3: Hermite normal form of an $N_F \times N_c$ integral matrix Q. The gray region represents the non-zero entries. This decomposition is unique if we impose some further restrictions, such as positivity of pivots; this shall play no role in this work.

In this basis the matrix of charges has ker Q columns that are identically zero. This means that the corresponding photons are essentially decoupled. (They still couple topologically, via the kinetic term. This does not affect local properties like the existence of a mass gap). This explains why the physical theta terms come from ker Q: a theta term is physical if and only if it multiplies a free photon. In other words, the parameter $\theta_i v_i$ is physical if and only if the photon $a_i v_i$ is decoupled from the fermions.

Putting together the last two observations we learn that, as far as classifying gapped theories is concerned, we can assume without loss of generality that there is the same number of photons than fermions, $N_F \equiv N_c$. This follows from the rank-nullity theorem, $|\ker Q| - |\operatorname{coker} Q| \equiv N_c - N_F$, which implies that if $N_c \neq N_F$, then at least one of ker Q, coker Q will be non-empty. If ker Q is non-empty the system contains decoupled photons, which are gapped and hence do not affect the classification. Conversely, if coker Qis non-empty, the system contains continuous chiral symmetries and hence it is automatically gapless. The interesting question is, therefore, what happens if Q is square and non-singular.

If $N_F \equiv N_c$ and Q is full-rank, there are no rows nor columns that are zero, so there are no decoupled sectors and no continuous axial symmetries. There are no U(1)^{N_F} axial symmetries because of the mixed anomaly, and no non-abelian chiral symmetries because there are no repeated rows in Q (for otherwise the matrix would not be full-rank). We now claim that these conditions are not only necessary for being gapped, but also sufficient:

Proposition 2.8.1 The non-chiral QED system (2.8.1) defined by a square matrix of charges Q is gapped if and only if Q is full rank.

This claim follows from the analysis of section 2.4, and the fact that free fermions have chiral algebra $U(N_F)_1$ and the photons a chiral algebra $U(1)_{Q^tQ}$. The lattice generated by the compact scalars in $U(1)_{Q^tQ}$ is non-degenerate if and only if Q is full-rank. Equality of the energy-momentum tensors of $U(N_F)_1$ and $U(1)_{Q^tQ}$ is nothing but the standard boson-fermion correspondence in 2d.

One can reach the this conclusion by looking directly at the central charges. The central charge of the free fermions is N_F and, and that of the compact bosons is⁶⁴ sign $(Q^tQ) \equiv \operatorname{rank}(Q)$, and these match if and only if Q is full-rank, as required.

As an interesting remark, note that if $Q^t Q$ is not full rank, then in the gauge chiral algebra $U(1)_{Q^tQ}$ there is a factor of $U(1)_0$ for each zero eigenvalue of Q^tQ . This factor of $U(1)_0$ should be thought of as a free photon (a similar phenomenon was observed in [3] in a 3d QCD system). This is consistent with the discussion so far, in the sense that if the rank is not maximal there will be columns of zeros in Q, signaling decoupled photons.

⁶⁴Here sign(K) denotes the signature of the matrix K, defined as +1 for each positive eigenvalue, -1 for each negative eigenvalue, and 0 for each zero eigenvalue. As $K = Q^t Q$ is positive semi-definite, sign $(Q^t Q) \equiv \operatorname{rank}(Q^t Q)$.

Discrete symmetries. If rank $Q = N_c$, we saw earlier that Q defines a QED theory with no continuous chiral symmetries. That being said, the system in general enjoys several discrete symmetries. Let us look at purely left-handed transformations. If we consider a $U(1)_{\ell}$ transformation defined by an integer vector n, then the theta term shifts as

$$U(1)_{\ell} \colon \theta \mapsto \theta + \alpha n Q \,, \tag{2.8.7}$$

as per the flavor-gauge mixed anomaly. This shift means that $U(1)_{\ell}$ is not a true symmetry of the quantum system. On the other hand, if we choose α in such a way that θ stays invariant modulo 2π , then the corresponding transformation does constitute a true symmetry of the quantum theory. This is simplest to ensure in the Hermite basis 2.3. In this basis it becomes clear that there is a $\mathbb{Z}_{\tilde{q}}$ discrete symmetry for each diagonal component of \tilde{Q} , obtained by choosing $\alpha = 2\pi k/\tilde{q}$ with $k = 0, 1, \ldots, \tilde{q} - 1$. Note that there is another factor of $\mathbb{Z}_{\tilde{q}}$ that acts on the right-handed fermions alone, but the two factors of $\mathbb{Z}_{\tilde{q}}$ are not two distinct symmetries, inasmuch as their simultaneous action is nothing but a gauge transformation. Hence, all in all, the flavor symmetry group of QED is

$$\prod_{\tilde{q}\in \operatorname{diag}(\tilde{Q})} \mathbb{Z}_{\tilde{q}} \,. \tag{2.8.8}$$

Note that if rank $Q < N_c$, then some of the diagonal components of \tilde{Q} will be zero. If for $\tilde{q} = 0$ we agree to denote $\mathbb{Z}_0 \equiv U(1)$, then the group above also contains the case where the system has non-trivial continuous symmetries.

For future reference we mention the fact that the order of the symmetry group is $\prod \tilde{q} \equiv \det(Q)$.

Chiral theories. We finally make a few remarks concerning chiral theories. These are labelled by pairs of integral matrices Q_{ℓ}, Q_r which specify the charges of the left-movers and right-movers, respectively. Gauge anomaly cancellation requires (2.2.17)

$$Q_\ell^t Q_\ell \equiv Q_r^t Q_r \,. \tag{2.8.9}$$

The reader might find it useful to have at their disposal examples of chiral theories. A trivial class of examples is $Q_{\ell} = -Q_r$. A more interesting class of examples is provided by choosing any non-symmetric normal matrix Q, and taking $Q_{\ell} = Q$, $Q_r = Q^t$. More generally, it is easy to show that if Q_{ℓ}, Q_r satisfy the gauge anomaly cancellation condition (2.8.9), then there exists some orthogonal matrix \mathcal{O} such that $Q_{\ell} \equiv \mathcal{O}Q_r$. Therefore, we can generate other families of examples by fixing Q_r and looking for orthogonal matrices \mathcal{O} that make $\mathcal{O}Q_r$ integral.

In any case, many of the previous claims for non-chiral theories can be easily generalized to chiral theories. For example, if $N_c > N_F$, the extra photons are still decoupled. Indeed, if the matrices Q_{ℓ}, Q_r are fat (more columns than rows, see figure 2.3) then they necessarily have a non-trivial kernel. The anomaly cancellation condition says that they in fact *share* the kernel: the equality $Q_{\ell}^t Q_{\ell} \equiv Q_r^t Q_r$ implies that $||Q_{\ell}v||^2 = ||Q_rv||^2$ for any vector v, so v is either annihilated by both Q_{ℓ}, Q_r or by neither. If v is in their kernel, then the linear combination $v_i a_i$ is indeed a decoupled photon.

Similarly, if $N_c < N_F$, then there will necessarily be some anomalous continuous symmetry, because the matrices Q_ℓ, Q_r will have a non-trivial cokernel, so the associated currents will be conserved — the mixed anomaly with the gauged U(1) will vanish.

All in all, in classifying gapped theories we can assume without loss of generality that $N_c \equiv N_F$, and that Q_ℓ, Q_r are full rank. In this situation, the exact same argument from before proves that these are not only necessary conditions for being gapped, but also sufficient:

Proposition 2.8.2 A chiral QED system defined by a pair of square matrices of charges (Q_{ℓ}, Q_r) , subject to the gauge anomaly cancellation condition (2.8.9), is gapped if and only if (Q_{ℓ}, Q_r) are full rank. (Both matrices necessarily have the same rank, due to (2.8.9)).

As a consistency check, note that a gapped theory cannot have continuous chiral symmetries, and it is not entirely obvious from the discussion above that a model with full rank matrices (Q_{ℓ}, Q_r) has no such symmetries. It is clear that, being full rank, there are no purely left handed (nor purely right handed) symmetries; but there is no immediate reason that excludes symmetries where both chiralities transform at the same time. It is not hard to show that, as a matter of fact, no such symmetries exist either: any would-be flavor symmetry where both chiralities transform simultaneously is either broken by a mixed flavor-gauge anomaly, or a pure gauge transformation itself. Hence, chiral models with full rank (Q_{ℓ}, Q_r) have no continuous chiral symmetries, as required for a supposedly gapped theory.

The conjectural infrared TQFT has left chiral algebra $U(N_F)_1/U(1)_{Q_\ell^t Q_\ell}$, and right chiral algebra $U(N_F)_1/U(1)_{Q_r^t Q_r}$. These two are in fact identical, by the anomaly cancellation condition. The sectors of the two chiral halves are combined via the orthogonal matrix \mathcal{O} discussed above, namely $\mathcal{O} := Q_\ell Q_r^{-1}$.

2.8.1 U(1) with N charge-q Dirac fermions

Here we analyse the infrared dynamics of U(1) plus N copies of a charge-q Dirac fermion. The claim is that the low energy theory of this system corresponds to a copy of the $SU(N)_1$ WZW model on each of the q universes. (These universes are the result of the \mathbb{Z}_q one-form symmetry). This nicely reproduces the analysis of [198].

According to the general conjecture (2.5.1), the infrared dynamics of the model are described by the coset

$$\frac{\mathrm{U}(N)_1}{\mathrm{U}(1)_{Nq^2}}.$$
(2.8.10)

Note that if N > 1, the central charge is non-zero, so the coset describes a gapless theory.

In order to project the theory into a specific universe we gauge the one-form symmetry \mathbb{Z}_q , to wit

$$\left(\frac{\mathrm{U}(N)_1}{\mathrm{U}(1)_{Nq^2}}\right)/\mathbb{Z}_q \equiv \frac{\mathrm{U}(N)_1}{\mathrm{U}(1)_N} \equiv \mathrm{SU}(N)_1.$$
(2.8.11)

The second equality involves the character decomposition

$$d_{\text{NS-X}}(q, g, \theta) = \sum_{n=0}^{N-1} (-1)^n \chi_n(q, \theta) \chi_{\gamma^n \cdot \mathbf{0}}(q, g)$$

$$d_{\text{R-X}}(q, g, \theta) = \sum_{n=0}^{N-1} (-1)^{n+1} \chi_{n+\lfloor N/2 \rfloor}(q, \theta) \chi_{\gamma^n \cdot \mathbf{0}}(q, g) ,$$
(2.8.12)

where θ is a flavor U(1) parameter and g an SU(N) flavor parameter. The characters of U(1)_N are denoted by $\chi_n(q, \theta)$ and those of SU(N)₁ by $\chi_\lambda(q, g)$. When N is odd, χ_n denotes a super-character and when even, a regular character.

Note that when N = 1 the CFT SU(1)₁ becomes trivial, which means that the charge-qSchwinger model has a unique, trivial vacuum in each universe. In this case, the infrared coset U(1)₁/U(1)_{q²} describes a gapped theory, and the character decomposition is

$$d_{\rm NS-X}(q,\theta) = \sum_{\ell=0}^{q-1} (\pm 1)^{\ell} \chi_{\ell q}^{\rm NS-X}(q,\theta)$$

$$d_{\rm R-X}(q,\theta) = \sum_{\ell=0}^{q-1} (\pm 1)^{\ell} \chi_{\ell q+\lfloor q/2 \rfloor}^{\rm R-X}(q,\theta) ,$$
(2.8.13)

where χ_{ℓ} are the characters of $U(1)_{q^2}$; these are regular (bosonic) characters when q is even (cf. (2.7.21)), and super-characters when odd (cf. (2.7.22)). From this character decomposition we learn that the theory has q vacua, all bosonic, in both sectors $\mathcal{H}_{\rm NS} = \mathcal{H}_{\rm R} = \mathbb{C}^{q|0}$. These qvacua live in the q universes, one in each⁶⁵. Out of these, $\lceil q/2 \rceil$ are neutral under $(-1)^{F_L}$, and the rest $\lfloor q/2 \rfloor$ are charged.

2.8.2 Vacua of gapped theories

In the previous section we described the infrared dynamics of a gapless theory. Here we study the gapped case, namely those theories where the matrix of charges Q is square and full rank. Conjecturally, the vacua of such theories are described by the coset (2.5.1)

$$\frac{\mathrm{U}(n)_1}{\mathrm{U}(1)_{Q^tQ}}\,.\tag{2.8.14}$$

⁶⁵More precisely, the states that live in a specific universe are linear combination of these q states. $(-1)^{F_L}$ does not commute with the one-form symmetry, and therefore the states in a given universe do not have well-defined charge under $(-1)^{F_L}$.

We focus on the case where Q describes an even lattice, i.e., where all the diagonal components of Q^tQ are even. In this situation the CFT $U(1)_{Q^tQ}$ is bosonic, that is, its characters do not depend on the spin structure.

The characters of the numerator $U(n)_1$ are given by

$$d_{\rm NS-X}(q,\theta) = q^{-n/24} \prod_{r=0}^{\infty} \prod_{i=1}^{n} (1 \pm z_i q^{r+1/2}) (1 \pm z_i^{-1} q^{r+1/2}) d_{\rm R-X}(q,\theta) = q^{n/12} \left[\prod_{i=1}^{n} z_i^{1/2} \pm z_i^{-1/2} \right] \prod_{r=1}^{\infty} \prod_{i=1}^{n} (1 \pm z_i q^r) (1 \pm z_i^{-1} q^r) ,$$
(2.8.15)

where $\theta \in U(n)$ is a flavor parameter conjugate to $\theta \sim \operatorname{diag}(z_1, \ldots, z_n)$.

On the other hand, the characters of the denominator are

$$\chi_{\ell}(q,\theta) = \frac{1}{\eta(q)^n} \sum_{u \in \mathbb{Z}^n + K^{-1}\ell} q^{\frac{1}{2}u^t K u} z^{Qu}, \qquad K := Q^t Q, \qquad (2.8.16)$$

where η is the Dedekind eta function.

The vacua of the QED theory labelled by a matrix Q are determined by the branching functions of $U(n)_1$ into $U(1)_{Q^tQ}$, i.e., by the decomposition of $d_{\pm,\pm}$ characters into χ_ℓ characters. We propose that the branching functions of the coset $U(n)_1/U(1)_{Q^tQ}$ are given by the following:

$$d_{\text{NS-X}}(q,\theta) = \sum_{\ell \in \Gamma(Q)} (\pm 1)^{||\ell||^2} \chi_{Q^t \ell}(q,\theta)$$

$$d_{\text{R-X}}(q,\theta) = \sum_{\ell \in \Gamma(Q)} (\pm 1)^{||\ell||^2} \chi_{Q^t (\ell+1/2)}(q,\theta),$$

(2.8.17)

where $\Gamma(Q) \cong \mathbb{Z}^n / \sim$ is the set of all integral vectors modulo the identification through rows of Q: two vectors are declared to be equivalent if they differ by some integral linear combination of the rows of Q. As a consistency check, note that $2h_{Q^t\ell} = ||\ell||^2$ so the chiral fermion parity in the NS-sector corresponds precisely to the spin of the characters. Similarly, in the R-sector the spin is $2h_{Q^t(\ell+1/2)} = n/4 \mod 1$ so the spin is in $\mathbb{Z} + \frac{1}{8}n$, as expected from the zero-point energy of the fermions.

The branchings above predict that the QED theory with matrix of charges Q has $|\Gamma(Q)| \equiv |\det(Q)|$ vacuum states. As in the Schwinger model, these can be thought of as the result of the spontaneous symmetry breaking of the axial symmetry, which also has $|\det(Q)|$ elements. Note also that $|\det(Q)|$ is the order of the one-form symmetry, which suggests that each universe has a single vacuum state.

Chapter 3

Three-Dimensional Gauge Theories with Two-Index Matter Fields.

Authorship. The content of this chapter is reproduced almost verbatim from the paper [3] written in collaboration with Changha Choi, Jaume Gomis, and Zohar Komargodski.

Abstract. We study the nonperturbative dynamics of $3d \operatorname{SU}(N)$ gauge theories coupled to a fermion in a rank-two representation. We argue that when the Chern-Simons level is sufficiently small the theory develops a quantum phase with an emergent topological field theory. When the level vanishes, we further argue that a baryon condenses and hence baryon symmetry is spontaneously broken. The infrared theory then consists of a Nambu-Goldstone boson coupled to a TQFT. Our proposals lead to new fermion-fermion dualities involving fermions in two-index representations. We make contact with some recently discussed aspects of four-dimensional gauge theories and their (non supersymmetric) domain walls. Finally, we discuss some aspects of the time-reversal anomaly in theories with a one-form symmetry.

Acknowledgments. We would like to thank A. Armoni, C. Córdova, T. Johnson-Freyd, P.S. Hsin, D. Radičević, N. Seiberg, T. Senthil, and R. Thorngren for useful discussions.

3.1	Introduction
3.2	Phase Diagrams
3.3	Additional Consistency Checks
3.4	Domain Walls in Four Dimensional Gauge Theories
3.5	Comments on Future Directions
3.6	Review of QCD_3 dynamics
3.7	Time-reversal and flavor symmetries
3.8	Time-reversal anomaly

3.1 Introduction

Strongly coupled Quantum Field Theories (QFTs) can develop interesting infrared "quantum" (nonperturbative) phases, distinct from those that can be inferred by semiclassical considerations. Recently the study of the infrared dynamics of 2 + 1 dimensional gauge theories has resulted in the discovery of novel nonperturbative quantum phases [29, 30, 113, 127–129]. In addition, some related aspects have been already studied on the lattice [199, 200]. The subject involves several conceptual differences from the (perhaps) more familiar setting of 3 + 1 dimensional gauge theories coupled to matter:

- Since the gauge coupling has positive mass dimension, 2 + 1 dimensional gauge theories are always asymptotically free. In particular, these theories are interesting even when the gauge group is U(1).
- 2 + 1 dimensional gauge theories are labeled by a gauge group, matter fields, and the Chern-Simons couplings, which are quantized.
- Since in 2 + 1 dimensions there is no notion of spinor chirality, one can typically add a mass term for the matter fields preserving all the continuous symmetries of the massless theory. In some theories, however, the massless point is distinguished by the presence of (a discrete) antilinear time-reversal symmetry. The phases of these theories can be studied as a function of the continuous mass deformations for the matter fields.
- There are no continuous 't Hooft anomalies for the symmetries of 2 + 1 dimensional theories. However, there are many discrete anomalies and they have to be consistent with the infrared phases of these theories. Even though these anomalies are discrete they nevertheless provide highly nontrivial constraints on the infrared dynamics. In addition, such quantum phases are constrained by the matching of some counterterms. This matching of counterterms physically corresponds to consistency conditions on conductivity coefficients. We will encounter some examples in this paper.

The subject of 2 + 1 dimensional gauge theories connects in an obvious way to condensed matter physics (where the gauge symmetry is typically emergent) and in somewhat less obvious ways to particle physics. Many 3 + 1 dimensional gauge theories have degenerate trivial vacua. As a result, a domain wall connecting two vacua supports at low energies a 2 + 1 dimensional theory, and one can often make this connection very natural (as we will see in this paper). In addition, one can study 3 + 1 dimensional gauge theories compactified on a circle (e.g. in the context of finite temperature physics), a problem that similarly reduces to the study of 2 + 1 dimensional systems.

In spite of the fact that 2 + 1 dimensional gauge theories are asymptotically free, these theories admit regimes in parameter space where they are weakly coupled and the infrared dynamics can be inferred by a careful semiclassical analysis:

- When all the matter fields have a large mass (in units of the Yang-Mills coupling constant) they can be integrated out at energy scales above the strong coupling scale. This only leads to a shift in the infrared Chern-Simons level (this shift is one-loop exact for fermions and trivial for bosons) and hence the deep infrared theory is given by the infrared dynamics of the pure Yang-Mills-Chern-Simons gauge theory, which is quite well understood. This typically leads in the infrared to a Chern-Simons Topological Quantum Field Theory (TQFT).⁶⁶ Since the shift of the Chern-Simons level depends on the sign of the mass of the fermion [24, 201, 202], the large mass limit defines an asymptotically large-positive mass phase and an asymptotically large-negative mass phase, described by two distinct TQFTs.
- When the number of matter fields is very large⁶⁷ [25, 126] (i.e. many species or large representations) one can demonstrate that there exists a weakly coupled conformal field theory (CFT) which interpolates between the TQFTs describing the two asymptotically large mass phases. Such a theory does not develop interesting new quantum phases.
- Likewise, when the Chern-Simons level is very large [26] one can show that there exists a weakly coupled CFT interpolating between the TQFTs describing the two asymptotically large mass phases. For large k the theory does not develop interesting new quantum phases.

While no new quantum phases emerge for "large representations" or large Chern-Simons level, it is a wide-open nonperturbative problem to determine for which representations and which levels new quantum phases develop. For some recent work on such questions in the context of quiver gauge theories see [203, 204] and references therein.

It follows from our discussion above that the dynamics of 2 + 1 dimensional gauge theories is especially interesting when neither the Chern-Simons level, the dimension of the representation, nor the mass of the matter fields are too large. In this regime there is no semiclassical approximation to the dynamics of the theory. This is when we may expect

 $^{^{66}}$ An exception to this is when the Chern-Simons level vanishes upon integrating out massive matter fields. In that case, for simply connected groups, the TQFT is trivial. For non-simply connected groups the situation is more complicated, but we will not need it here except in the case of U(1), where the low energy theory is the gapless theory of a compact scalar. This fact will play an important role in this paper.

⁶⁷The relevant group theory factor is the index of the (possibly reducible) representation of the matter fields.

quantum effects to dominate the dynamics and new interesting phenomena may emerge, including new nonperturbative phases.

In this paper we study a class of 2 + 1 dimensional gauge theories for which we provide a large body of evidence that they indeed develop new nonperturbative phases along with new phase transitions, for which we propose novel dual descriptions. (We do not know, in general, if these phase transitions are 1st or 2nd order.) The theories we analyze are SU(N)Yang-Mills gauge theories coupled to a fermion in the rank-two symmetric or antisymmetric representation of SU(N) and a Chern-Simons term at level k. (The fermion is a Dirac fermion, i.e. a complex fermion with two components.)

For generic k, these models have a global baryon number symmetry, $U(1)_B$, acting on the fermion as $\psi \mapsto e^{i\alpha}\psi$. In addition, there is charge-conjugation symmetry. Both of these symmetries are unbroken by the mass term $im\psi\psi$. For k=0 (which is only allowed for even N) the model also admits a time-reversal symmetry. The mass term breaks the time-reversal symmetry. Finally, since there are no dynamical degrees of freedom in the fundamental representation, these theories have a one-form \mathbb{Z}_2 symmetry when N is even. (In the context of condensed matter physics, the one-form symmetry is expected to be accidental.)

Let us now summarize the main results:

1. These theories have a critical value of the level $k_{\rm crit}$ below which a new intermediate quantum phase appears between the semiclassically accessible asymptotic large-positive and large-negative mass phases. The critical value is^{68}

symmetric:
$$k_{\text{crit}} = \frac{N+2}{2}$$
, antisymmetric: $k_{\text{crit}} = \frac{N-2}{2}$. (3.1.1)

2. For $0 \neq k < k_{\rm crit}$ there is an intermediate quantum phase described by the following "emergent" TQFTs⁶⁹

symmetric:
$$U\left(\frac{N+2}{2}-k\right)_{\frac{N+2}{2}+k,2k}$$
, antisymmetric: $U\left(\frac{N-2}{2}-k\right)_{\frac{N-2}{2}+k,2k}$.
(3.1.2)

3. For k = 0 there is an intermediate quantum phase that includes a Nambu-Goldstone boson (NGB) corresponding to the spontaneous breaking of $U(1)_B$, along with a TQFT:

symmetric:
$$\frac{\mathrm{SU}\left(\frac{N+2}{2}\right)_{\frac{N+2}{2}} \times S^{1}}{\mathbb{Z}_{\frac{N+2}{2}}}, \quad \text{antisymmetric:} \quad \frac{\mathrm{SU}\left(\frac{N-2}{2}\right)_{\frac{N-2}{2}} \times S^{1}}{\mathbb{Z}_{\frac{N-2}{2}}}. \tag{3.1.3}$$

⁶⁸The quantized level k must be integer for N even and half-integer for N odd. See section 3.2.

⁶⁹We recall that $U(N)_{P,Q} := \frac{SU(N)_P \times U(1)_{NQ}}{\mathbb{Z}_N}$ with $P \equiv Q \mod N$. The quotient by \mathbb{Z}_N gauges an anomaly-free one-form symmetry [32, 205].

The notation S^1 stands for the linear sigma model of a compact real scalar field $\{\phi \sim \phi + 2\pi\}$, which is dual to pure U(1)₀ gauge theory in 2 + 1 dimensions.⁷⁰ The scalar couples to the TQFT SU $\left(\frac{N\pm 2}{2}\right)_{\frac{N\pm 2}{2}}$ by gauging a diagonal, anomaly-free $\mathbb{Z}_{\frac{N\pm 2}{2}}$ one-form symmetry.

The microscopic (ultraviolet) theory for k = m = 0 is time-reversal invariant. It may then seem odd that we are proposing that this model flows to a TQFT coupled to a Nambu-Goldstone boson, as TQFTs are typically non-time-reversal invariant. It is encouraging to observe that the $SU(n)_n/\mathbb{Z}_n$ Chern-Simons theory is in fact also a time-reversal invariant (spin) TQFT [112]. This is a nontrivial consistency check of our proposal.

4. For $k < k_{crit}$ these theories undergo two phase transitions as a function of the mass of the fermion. The phase transitions connect the intermediate quantum phase with the asymptotic large-positive mass phase and with the asymptotic large-negative mass phase, respectively. We propose that these transitions have a dual description in terms of another 2+1 dimensional gauge theory. This leads us to propose new (fermion-fermion) dualities in 2+1 dimensions.

Dualities for $SU(N)_k$ + symmetric ψ for $k < \frac{N+2}{2}$:

$$\begin{aligned} \mathrm{SU}(N)_k + & \text{symmetric } \psi \longleftrightarrow \mathrm{U}\left(\frac{N+2}{2}+k\right)_{\frac{1}{2}(-1+k-3N/2),k-N/2} + & \text{antisymmetric } \tilde{\psi} \\ \mathrm{SU}(N)_k + & \text{symmetric } \psi \longleftrightarrow \mathrm{U}\left(\frac{N+2}{2}-k\right)_{\frac{1}{2}(+1+k+3N/2),k+N/2} + & \text{antisymmetric } \hat{\psi} . \end{aligned}$$

$$(3.1.4)$$

Dualities for $SU(N)_k$ + antisymmetric ψ for $k < \frac{N-2}{2}$:

$$SU(N)_{k} + \text{ antisymmetric } \psi \longleftrightarrow U\left(\frac{N-2}{2}+k\right)_{\frac{1}{2}(+1+k-3N/2),k-N/2} + \text{ symmetric } \tilde{\psi}$$

$$SU(N)_{k} + \text{ antisymmetric } \psi \longleftrightarrow U\left(\frac{N-2}{2}-k\right)_{\frac{1}{2}(-1+k+3N/2),k+N/2} + \text{ symmetric } \hat{\psi}.$$

$$(3.1.5)$$

We note that the fermion in the dual gauge theory transforms in the other rank-two representation compared to the fermion in the original gauge theory.

5. For $k \ge k_{\text{crit}}$ the phase diagram has just two phases: the asymptotic large-positive mass and large-negative mass semiclassical phases, separated by a phase transition. For very

⁷⁰For k = 0, the U(1)₀ factor in (3.1.2) should not be interpreted as a TQFT, but rather as a gapless U(1)₀ gauge theory, which can be dualized to the compact scalar.

large k the phase transition is controlled by a weakly coupled CFT. The asymptotic large mass phases are the TQFTs

symmetric:
$$SU(N)_{k\pm\frac{N+2}{2}}$$
, antisymmetric: $SU(N)_{k\pm\frac{N-2}{2}}$, (3.1.6)

where the upper/lower sign is for the large positive/negative mass asymptotic phase. These phases are present in these theories for any k, but for $k < k_{\text{crit}}$ they are separated by the intermediate quantum phases discussed above while for $k \ge k_{\text{crit}}$ they are separated by a single transition (which for sufficiently large k must be given by a CFT).

The plan for the rest of the chapter is as follows. In section 3.2 we present our conjectures for the phases of SU(N) Yang-Mills gauge theory coupled to a fermion in the symmetric and antisymmetric representation and with a Chern-Simons term. In section 3.3 we present a list of nontrivial consistency checks, involving a comparison to known special cases, and the highly nontrivial matching of some contact terms. In section 3.4 we study some domain wall solutions in four-dimensional gauge theories and show that at least in one case one can explicitly demonstrate baryon symmetry breaking on the domain wall, in agreement with the predictions for the infrared behavior of the corresponding three-dimensional gauge theory. In section 3.5 we make a few forward-looking comments about the time-reversal anomaly and about baryons.

3.2 Phase Diagrams

Consider Yang-Mills theory with gauge group SU(N), a Chern-Simons term and a Dirac fermion ψ in the rank-two symmetric or antisymmetric representation R of SU(N):

$$\mathcal{L} = \operatorname{tr}\left(-\frac{1}{2g^2}F^2 + \frac{ik_{\text{bare}}}{4\pi}\left(A\mathrm{d}A + \frac{2}{3}A^3\right) + i\bar{\psi}\mathcal{D}\psi + im\bar{\psi}\psi\right).$$
(3.2.1)

In this section we make a proposal for the phase diagram of these theories as a function of the effective Chern-Simons level $k := k_{\text{bare}} - T(R)$ and of the mass $m \in \mathbb{R}$ of the fermion.⁷¹ T(R) is the Dynkin index of the SU(N) representation under which the Dirac fermion transforms. Since under time-reversal $k \to -k$ (along with reversing the sign of the mass), we restrict our discussion to $k \ge 0$. Since $k_{\text{bare}} \in \mathbb{Z}$, it follows from table 3.1 that $k \in \mathbb{Z}$ for N even and $k \in \mathbb{Z} + \frac{1}{2}$ for N odd.

⁷¹The shift of the Chern-Simons level in (3.2.1) by T(R) arises from the determinant of the massless fermion. It is convenient to use k_{bare} when writing Lagrangians since k_{bare} is always an integer. However, the infrared phases of the theory are more conveniently labeled by $k := k_{\text{bare}} - T(R)$ because time-reversal symmetry acts on k by simply reversing it, along with reversing the mass of the fermion.

	symmetric	antisymmetric	adjoint
T(R)	$\frac{1}{2}(N+2)$	$\frac{1}{2}(N-2)$	N

Table 3.1: Index for SU(N) rank-two and adjoint representations. (Since the adjoint representation is real, one could also take the corresponding fermion to be a Majorana fermion – that model was discussed in detail in [30].)

We now discuss the global symmetries of these theories. There is a $U(1)_B$ flavor symmetry and a $\mathbb{Z}_2^{\mathsf{C}}$ charge-conjugation symmetry C acting as⁷²

$$U(1)_B: \ \psi \mapsto e^{i\alpha}\psi, \qquad C: \begin{cases} A_\mu \mapsto -A_\mu^T \\ \psi \mapsto +\psi^*. \end{cases}$$
(3.2.2)

These transformations do not commute and generate the group $O(2)_B = U(1)_B \rtimes \mathbb{Z}_2^{\mathsf{C}}$. Since the center of the gauge group $\mathbb{Z}_N \subset \mathrm{SU}(N)$ acts as $\psi \mapsto e^{\frac{4\pi i}{N}}\psi$, the global symmetry group is $O(2)_B/\mathbb{Z}_{N/2}$ for N even and $O(2)_B/\mathbb{Z}_N$ for N odd. The operators charged under this symmetry are baryons, which will be discussed briefly at the end of this paper.

Since the gauge group is simply-connected, the magnetic symmetry group is trivial. For N even a $\mathbb{Z}_2 \subset \mathbb{Z}_N$ subgroup of the center acts trivially on ψ and the theory has a \mathbb{Z}_2 one-form global symmetry. For N odd the one-form symmetry is trivial. Finally, for k = m = 0 the theory is time-reversal invariant. Time-reversal symmetry acts on the fermion by

$$\mathsf{T}: \ \psi \mapsto \gamma_0 \psi^* \ . \tag{3.2.3}$$

Therefore, in these theories, $T^2 = (-1)^F$. It is easy to verify that the mass term is odd under time-reversal symmetry but it preserves all other symmetries.

We proceed now to analyzing the phase diagram. We start with the phases that can be established by a semiclassical analysis. When $|m| \gg g^2$ we can reliably integrate out the fermion before the interactions become strong. Integrating out a massive fermion shifts the Chern-Simons level to $k + \operatorname{sign}(m)T(R)$, and the resulting effective theory is pure SU(N) Yang-Mills with an integer-quantized Chern-Simons term at level $k + \operatorname{sign}(m)T(R)$. This theory, which now has no matter fields, flows at low energies to the topological SU(N) Chern-Simons theory at level $k + \operatorname{sign}(m)T(R)$, which we denote by SU(N)_{$k+\operatorname{sign}(m)T(R)$}. Therefore the infrared dynamics is captured by the TQFTs SU(N)_{$k\pm T(R)$} for large positive and negative mass respectively. These asymptotic large mass phases are present for all k and all N. In the above discussion of the asymptotic phases, k = T(R) is an exception since the infrared theory (after integrating the fermions out) is pure Yang-Mills theory (without a Chern-Simons term) with gauge group SU(N). In this case the infrared theory is trivial and gapped due to confinement and due to the fact that the gauge group is simply-connected.

 $^{^{72}}$ For N = 2 the action of C on the gauge field is a gauge transformation.

Figure 3.1: Phase diagram of SU(N) gauge theory with a symmetric fermion for $k \ge \frac{N+2}{2}$. The solid circle represents a phase transition between the asymptotic phases. For sufficiently large k we know for certain that the phase transition is associated with a CFT.

As long as k is sufficiently large, the above two topological phases $SU(N)_{k+sign(m)T(R)}$ are separated by a single transition. The question is below which value of k additional phases appear. Our proposal is that as long as $k \ge T(R)$ the above picture holds true, namely, the two semiclassically accessible phases are separated by a phase transition at some value of the mass. The phase diagrams for $k \ge T(R)$ are summarized in figures 3.1 and 3.2. Note that the boundary of the above region, k = T(R), is exactly where one of the phases becomes a trivial infrared theory, with no Chern-Simons TQFT.

$$\underbrace{\mathrm{SU}(N)_{k} + \Box \psi}_{k \ge \frac{N-2}{2}}$$

$$\xrightarrow{m \to +\infty}$$

$$\mathrm{SU}(N)_{k-\frac{1}{2}(N-2)}$$

$$\mathrm{SU}(N)_{k+\frac{1}{2}(N-2)}$$

m \cdot

Figure 3.2: Phase diagram of SU(N) with an antisymmetric fermion for $k \ge \frac{N-2}{2}$. The solid circle represents a phase transition between the asymptotic phases. For sufficiently large k we know for certain that the phase transition is associated with a CFT.

For $0 \le k < T(R)$ we propose that there is a new intermediate "quantum phase" in between the asymptotic large mass phases.⁷³ This phase is inherently quantum mechanical, and is not visible semiclassically. This new quantum phase connects to each of the asymptotic phases through a phase transition. The phase diagrams for 0 < k < T(R) are summarized in

⁷³In the case of symmetric fermion the new phase appears at $k = T(R) - 2 = \frac{N}{2} - 1$. See below.

figures 3.3 and 3.4. The reason that k = 0 is excluded from the figures is that it requires a separate discussion, as we shall see below.

$$\operatorname{SU}(N)_k + \Box \psi \qquad \qquad 0 < k < \frac{N+2}{2}$$

Figure 3.3: Phase diagram of SU(N) with a symmetric fermion for $0 < k < \frac{N+2}{2}$. The solid circles represent a phase transition between the asymptotic phases and the intermediate quantum phase. Each phase transition has a dual gauge theory description, which appears with an arrow pointing to the phase transition. The mass deformations are related by $m_{\psi} = -m_{\hat{\psi}}$ and $m_{\psi} = -m_{\tilde{\psi}}$.

The way we arrive at the phase diagrams in figures 3.3 and 3.4 is as follows. As mentioned above the asymptotic positive and negative mass phases are described by the TQFTs $SU(N)_{k\pm T(R)}$. These TQFTs admit a level/rank SU/U dual description [28, 206]⁷⁴

$$\operatorname{SU}(N)_{k\pm T(R)} \longleftrightarrow \operatorname{U}(T(R)\pm k)_{\mp N,\mp N}.$$
 (3.2.4)

We start with the level/rank dual description of the asymptotic positive mass phase and search for a dual description that would allow us to understand the quantum phase semiclassically in the dual variables. Similarly, we consider the asymptotic negative mass phase and search for an ultraviolet gauge theory that could describe the phase transition between the semiclassical asymptotic negative mass phase and the quantum phase. These steps involve some guesswork. The fact that this can be done at all is already a highly nontrivial consistency check. Indeed, the two required dual descriptions are mutually non-local⁷⁵ but

⁷⁴Level/rank dualities are generically valid only as spin TQFTs, and therefore, whenever the theory on one side of the duality is not spin (i.e. it does not have a transparent spin 1/2 line) we must tensor that theory with a trivial spin TQFT. $SU(N)_k$ is never spin and $U(N)_{k,k}$ is spin for k odd.

 $^{^{75}}$ By "mutually non-local" we mean that there exists no local map between the fields in the two dual

Figure 3.4: Phase diagram of SU(N) with an antisymmetric fermion for $0 < k < \frac{N-2}{2}$. The solid circles represent a phase transition between the asymptotic phases and the intermediate quantum phase. Each phase transition has a dual gauge theory description, which appears with an arrow pointing to the phase transition. The mass deformations are related by $m_{\psi} = -m_{\hat{\psi}}$ and $m_{\psi} = -m_{\tilde{\psi}}$.

there is only one quantum phase, which both of them have to describe simultaneously (in our case this happens thanks to the new level/rank duality in [28]). In the present context, luckily, we were able to find consistent dual descriptions describing the same quantum phase. Furthermore, this guess satisfies very nontrivial additional consistency checks, as we shall see. One of the dual theories is based on the gauge group U(T(R) - k) and the other on the gauge group U(T(R) + k) with appropriate Chern-Simons levels and matter representations.

We are thus led to propose the following new fermion-fermion dualities for $0 \le k < T(R)$:

Dualities for $SU(N)_k$ + symmetric ψ for $k < \frac{N+2}{2}$:

$$SU(N)_{k} + \text{ symmetric } \psi \longleftrightarrow U\left(\frac{N+2}{2}+k\right)_{\frac{1}{2}(-1+k-3N/2),k-N/2} + \text{ antisymmetric } \tilde{\psi}$$

$$SU(N)_{k} + \text{ symmetric } \psi \longleftrightarrow U\left(\frac{N+2}{2}-k\right)_{\frac{1}{2}(+1+k+3N/2),k+N/2} + \text{ antisymmetric } \hat{\psi}.$$

$$(3.2.5)$$

descriptions, yet, they have some region of overlap in the physics they describe. This is reminiscent of the Seiberg-Witten solution, which has two mutually non-local theories describing two different massless theories, with a region of overlap.

Dualities for $SU(N)_k$ + antisymmetric ψ for $k < \frac{N-2}{2}$:

$$\begin{split} &\mathrm{SU}(N)_k + \text{ antisymmetric } \psi \longleftrightarrow \mathrm{U}\bigg(\frac{N-2}{2}+k\bigg)_{\frac{1}{2}(+1+k-3N/2),k-N/2} + \text{ symmetric } \tilde{\psi} \\ &\mathrm{SU}(N)_k + \text{ antisymmetric } \psi \longleftrightarrow \mathrm{U}\bigg(\frac{N-2}{2}-k\bigg)_{\frac{1}{2}(-1+k+3N/2),k+N/2} + \text{ symmetric } \hat{\psi} \,. \end{split}$$

For k = T(R) - 1 = N/2 in the theory of the fermion in the symmetric representation the intermediate phase coincides with the asymptotic large negative mass phase and the first phase transition is therefore unnecessary. Indeed, the associated duality in the second line of (3.2.5) trivializes since the antisymmetric representation of U(1) is trivial.

The case of k = 0 is particularly interesting and requires a separate discussion. The quantum phase that has appeared in the figure 3.3 and 3.4 would seem to make sense also for k = 0. However, while for k > 0 it is a pure TQFT, for k = 0 it is not. Indeed, after integrating the fermion in the dual theory with gauge group U(T(R)), we are left with pure $U(T(R))_{T(R),0}$ Yang-Mills-Chern-Simons theory, with $T(R) = \frac{N}{2} \pm 1$ in the symmetric/antisymmetric case. The crucial point is that the $U(1)_0$ factor is not topological. This latter theory can be dualized to the theory of a compact, real scalar field ϕ

$$\mathcal{L}_{\mathrm{U}(1)_{0}} = \frac{f_{\pi}^{2}}{2} (\partial \phi)^{2} \,. \tag{3.2.7}$$

(3.2.6)

with $\phi \sim \phi + 2\pi$ and f_{π}^2 the "decay constant".

This theory is combined with the Chern-Simons theory in the following way: the NGB theory (3.2.7) has a non-anomalous U(1) one-form symmetry, corresponding to the conserved two-form current $\epsilon_{\mu\nu\rho}\partial^{\rho}\phi$. Likewise, $\mathrm{SU}(T(R))_{T(R)}$ Chern-Simons theory has a non-anomalous $\mathbb{Z}_{T(R)}$ one-form symmetry (it is non-anomalous when one views $\mathrm{SU}(T(R))_{T(R)}$ as a spin TQFT). We gauge the diagonal $\mathbb{Z}_{T(R)}$ symmetry, and denote this by

$$\frac{\operatorname{SU}(T(R))_{T(R)} \times S^1}{\mathbb{Z}_{T(R)}}, \qquad (3.2.8)$$

where $\mathbb{Z}_{T(R)}$ is the diagonal one-form symmetry. The phase diagrams for k = 0 are summarized in figures 3.5 and 3.6.

An immediate consistency check of this scenario – which we have already discussed in the introduction – is that the ultraviolet theory is time-reversal symmetric, so it is reassuring to realize that $SU(\frac{N}{2} \pm 1)_{\frac{N}{2} \pm 1} / \mathbb{Z}_{\frac{N}{2} \pm 1}$ Chern-Simons theory and the NGB theory are time-reversal invariant. The time-reversal invariance of the quotient $SU(\frac{N}{2} \pm 1)_{\frac{N}{2} \pm 1} / \mathbb{Z}_{\frac{N}{2} \pm 1}$ can be shown from level/rank duality as in [112].

The main feature of the k = 0 model is, of course, the nonperturbative spontaneous breaking of the U(1)_B baryon number symmetry. The U(1)_B symmetry breaking occurs due

$SU(N)_0 + \Box \psi$



Figure 3.5: Phase diagram of SU(N) gauge theory with a symmetric fermion for k = 0. The circle S^1 represents the corresponding sigma model. Each phase transition has a dual gauge theory description, which appears with an arrow pointing to the phase transition. The mass deformations are related by $m_{\psi} = -m_{\hat{\psi}}$ and $m_{\psi} = -m_{\tilde{\psi}}$.

to the condensation of a baryon. We will discuss the baryon operators in this theory very briefly in the last section.

This spontaneous symmetry breaking is not in contradiction with the Vafa-Witten type theorems [207]. In essence, if a symmetry cannot be preserved by a time-reversal invariant mass term then there is no obstruction for the spontaneous breaking of that symmetry.

Indeed, in our class of theories, it is not possible to deform by a mass term while preserving both $U(1)_B$ baryon number symmetry and time-reversal symmetry. This is obvious for N > 2since there is no mass term whatsoever that preserves time-reversal symmetry. However, the case of N = 2 requires special attention. In the case of N = 2 with an antisymmetric fermion the theory is always in the large k two-phase regime, and there is no quantum phase and no spontaneous breaking occurs, of course, since the fermion is completely decoupled. For N = 2 with a symmetric fermion, the situation is more interesting. A Dirac fermion in the rank-two symmetric representation is equivalent to two Majorana fermions in the adjoint representation. Let us denote the two Majorana fermions by Ψ_1, Ψ_2 , such that $\psi = \Psi_1 + i\Psi_2$. The $U(1)_B = SO(2)_B$ baryon symmetry simply rotates these two real fermions

$$\begin{pmatrix} \Psi_1 \\ \Psi_2 \end{pmatrix} \mapsto \begin{pmatrix} \cos\theta & \sin\theta \\ -\sin\theta & \cos\theta \end{pmatrix} \begin{pmatrix} \Psi_1 \\ \Psi_2 \end{pmatrix} .$$
(3.2.9)

Time-reversal symmetry can be taken to act as $\Psi_1 \mapsto \gamma_0 \Psi_1$, and $\Psi_2 \mapsto \gamma_0 \Psi_2$. Finally, we

$\mathrm{SU}(N)_0 + \Box \psi$



Figure 3.6: Phase diagram of SU(N) gauge theory with a antisymmetric fermion for k = 0. The circle S^1 represents the gapless sigma model with S^1 target space. Each phase transition has a dual gauge theory description, which appears with an arrow pointing to the phase transition. The mass deformations are related by $m_{\psi} = -m_{\hat{\psi}}$ and $m_{\psi} = -m_{\tilde{\psi}}$.

have a charge-conjugation symmetry C that acts as $\Psi_2 \mapsto -\Psi_2$ while keeping Ψ_1 intact. The minimal scalar baryon operator transforms with charge 2 under $U(1)_B$. The hermitian combinations $i(\bar{\Psi}_1\Psi_1 - \bar{\Psi}_2\Psi_2)$, and $i\bar{\Psi}_1\Psi_2$ are the components of this baryon operator.⁷⁶ The hermitian combination $i(\bar{\Psi}_1\Psi_1 + \bar{\Psi}_2\Psi_2)$ is instead invariant under baryon symmetry, but obviously breaks time-reversal symmetry when added to the Lagrangian.

Therefore, clearly, if we want to preserve a time-reversal symmetry, we must use the hermitian baryon operators above. Indeed, for example, adding $i\bar{\Psi}_1\Psi_2$ to the Lagrangian would preserve CT. However, there is no way to add a time-reversal invariant mass term that also preserves the U(1)_B baryon symmetry. Therefore, there is no obstruction for baryon symmetry to be spontaneously broken. Interestingly, in the case of N = 2, the TQFT trivializes (see figure 3.5) because SU(2)₂/ $\mathbb{Z}_2 =$ SO(3)₁, which is a trivial spin TQFT. The fact that in the particular case N = 2 the NGB is not accompanied by a TQFT will be crucial later, when we make contact with 3 + 1 dimensional physics. Note also that in the case of N = 2 it is quite clear that the operator which condenses and leads to the NGB is (without loss of generality) $i(\bar{\Psi}_1\Psi_1 - \bar{\Psi}_2\Psi_2)$.

⁷⁶They are the real and imaginary parts of the baryon constructed with a Dirac spinor in the symmetric representation $\epsilon_{\alpha_1\alpha_2}\epsilon_{\beta_1\beta_2}\psi^{\alpha_1\beta_1}\psi^{\alpha_2\beta_2}$.

3.3 Additional Consistency Checks

3.3.1 Special Cases

Here we discuss special values of N where we can compare our proposed dynamics with previously conjectured phase diagrams for other families of theories. We also embed SU(2) with a rank-two symmetric fermion in a renormalization group flow of $\mathcal{N} = 2$ supersymmetric SU(2) pure gauge theory, and contrast our proposed phase diagram with the expected infrared dynamics of the supersymmetric theory.

• $SU(4)_k$ with antisymmetric fermion

A consistency check on our proposed dynamics follows from the isomorphism $SU(4) \simeq$ Spin(6) and the fact that the antisymmetric representation of SU(4) is the six-dimensional, vector, real representation of Spin(6). Since ψ is a Dirac fermion we have the following equivalence of theories

$$SU(4)_k + \text{antisymmetric } \psi \equiv Spin(6)_k + (N_f = 2) \Psi,$$
 (3.3.1)

where by $N_f = 2$ we mean two Majorana fermions in the vector representation of Spin(6).

For $k \ge 1$ the phase diagram of $SU(4)_k$ + antisymmetric has two asymptotic phases (see figure 3.2). The phase diagram of $Spin(6)_k + N_f \Psi$ was derived in [113]. For $N_f = 2$ and $k \ge 1$ the phase diagrams agree trivially by virtue of the identity of the TQFTs $SU(4)_n = Spin(6)_n$.

We now proceed to the nontrivial matching for k = 0 where both theories have an intermediate phase, which we want to compare. Plugging N = 4 in figure 3.4 we find that the intermediate phase of $SU(4)_k$ + antisymmetric ψ is described by

$$U(1)_0$$
, (3.3.2)

which, as we explained in detail, is simply a free compact scalar ϕ with periodicity 2π . By contrast, the intermediate phase of $\text{Spin}(6)_k + N_f \Psi$ is described by the following coset [113]

$$\frac{\mathrm{SO}(N_F)}{S\left(\mathrm{O}\left(\frac{N_F}{2}\right) \times \mathrm{O}\left(\frac{N_F}{2}\right)\right)} + 6\Gamma_{WZ}, \qquad (3.3.3)$$

where Γ_{WZ} is a Wess-Zumino term. The coset in (3.3.3) can be described more explicitly by the equivalence relation of SO(N_F) matrices O

$$O \sim P \cdot O \,, \tag{3.3.4}$$

where P is a block-diagonal matrix with two $\frac{N_F}{2} \times \frac{N_F}{2}$ blocks A and B

$$P = \begin{pmatrix} A & 0\\ 0 & B \end{pmatrix} \tag{3.3.5}$$

such that $\det(A \cdot B) = 1$.

For $N_f = 2$ the Wess-Zumino term vanishes (because $\pi_3(S^1) = 0$) and we are left with a sigma model on the space of SO(2) matrices $O = \begin{pmatrix} \cos \theta & \sin \theta \\ -\sin \theta & \cos \theta \end{pmatrix}$ subject to the equivalence relation

$$\begin{pmatrix} \cos\theta & \sin\theta \\ -\sin\theta & \cos\theta \end{pmatrix} \sim \begin{pmatrix} -1 & 0 \\ 0 & -1 \end{pmatrix} \cdot \begin{pmatrix} \cos\theta & \sin\theta \\ -\sin\theta & \cos\theta \end{pmatrix},$$
(3.3.6)

with implies that $\theta \sim \theta + \pi$. Therefore, after the quotient the space is still isomorphic to a circle (albeit with a radius smaller by a factor of 2). Therefore this precisely coincides with the result that we obtained for SU(4) with an antisymmetric fermion.

In summary, our phase diagram for $SU(N)_k$ with a fermion in the antisymmetric representation for N = 4 precisely matches the proposed phase diagram of Spin(M) with N_f Majorana fermions in the vector representation for M = 6 and $N_f = 2$. This supports the validity of both phase diagrams.

• $SU(3)_k$ with an antisymmetric fermion

A somewhat more trivial consistency check can be made for $SU(3)_k$ with an antisymmetric fermion by noting that the rank-two antisymmetric representation of SU(3) is the same as the complex conjugate of the fundamental, three dimensional representation of SU(3). Thus we have the equivalence of theories

$$SU(3)_k$$
 + antisymmetric $\psi \equiv SU(3)_k$ + $(N_f = 1) \Upsilon$, (3.3.7)

where Υ is a Dirac fermion. These theories, which always have two phases (i.e. there is no intermediate phase regime), can be seen to have the same phase diagram by comparing figure 3.4 for N = 3 with the phase diagram of $SU(N)_k$ with N_f fermions in the fundamental representation for N = 3 and $N_f = 1$ in [29].

• $SU(2)_k$ with an antisymmetric fermion

A very degenerate special case is $SU(2)_k$ with a fermion in the antisymmetric representation. In this case the fermion is decoupled from the gauge dynamics and there is no intermediate phase. The infrared is captured by $SU(2)_k$ Chern-Simons theory except at one point in the phase diagram, which coincides with the phase transition in figure 3.2. The phase transition simply corresponds to a neutral fermion becoming massless.

• $SU(2)_k$ with a symmetric fermion

The rank-two symmetric representation of SU(2) is the adjoint representation. Therefore, figures 3.1 and 3.3 for N = 2 describe the infrared dynamics of SU(2) QCD with $N_f = 2$ adjoint Majorana fermions.⁷⁷ This extends the phase diagram of SU/SO/Sp adjoint QCD with $N_f = 1$ put forward in [30].

⁷⁷We recall that our theory is based on a Dirac fermion and therefore $N_f = 2$ Majorana adjoint fermions of SU(2).

The particular case N = 2 with a symmetric fermion admits an embedding in the $\mathcal{N} = 2$ supersymmetric theory of one SU(2) vector multiplet. In addition to the fields we have in our theory, this model also has a real scalar field φ in the adjoint representation, with couples to the fermions via Yukawa terms. What we called U(1)_B is naturally referred to as the R-symmetry in the supersymmetric context. We can flow from the supersymmetric theory to our theory by simply adding a (supersymmetry-breaking) mass term for the real scalar field φ .⁷⁸ Below we analyze what happens if that mass term is very small compared to the scale set by the gauge coupling.

The infrared of the supersymmetric model consists of [208] a complex scalar field Y, whose imaginary part transforms inhomogeneously under the U(1) R-symmetry, thus signaling spontaneous symmetry breaking of this symmetry. The kinetic term for Y is approximately canonical for large re(Y) and the potential is the runaway potential $V \sim e^{-\frac{1}{g} \operatorname{re}(Y)}$. Adding a small (supersymmetry-breaking) mass term $m^2 \varphi^2$ in the ultraviolet translates to adding a small (supersymmetry-breaking) mass term $m^2 \operatorname{re}(Y)^2$ in the infrared. (The map between the deformations in the UV and IR is rather simple for large re(Y) because the theory is weakly coupled there.) For small enough mass of φ the minimum of

$$V = e^{-\frac{1}{g}\operatorname{re}(Y)} + m^2(\operatorname{re}(Y))^2$$
(3.3.8)

is therefore at large re(Y) and we can analyze the physics semiclassically. The fermions are all lifted due to the Yukawa couplings and re(Y) is likewise massive at the minimum of the potential. The deep infrared theory therefore consists of just the (compact) Nambu-Goldstone boson im(Y) without an additional TQFT, exactly as in the scenario we proposed above for SU(2) with a Dirac fermion in the rank-two symmetric representation.

We shall return to this theory later when we discuss domain walls in 3 + 1 dimensional SU(2) Yang-Mills with $N_f = 2$ adjoint Majorana fermions.

3.3.2 Gravitational Counterterm Matching

Another nontrivial check of our proposed phase diagrams in figures 3.3 and 3.4 can be devised by coupling the theories to background gravity. A well-defined (scheme independent) observable is the difference in the gravitational counterterm $2\Delta c \operatorname{CS}_{\text{grav}}$ between the asymptotic negative and asymptotic positive mass phases. This difference is closely related to the difference in the thermal conductance in the two phases. This is an interesting quantity to study because it can be easily computed in the original "electric variables" where it is one loop exact. But it can also be computed in the dual variables, followed by traversing the quantum phase, and using the dual variables again. Therefore, one can devise a concrete nontrivial consistency check. Such computations were done in the context of supersymmetric

⁷⁸A closely related $\mathcal{N} = 1$ preserving mass deformation was analyzed in [127].

dualities in [209, 210], where the connection to the physically observable thermal conductance (or the analogous charge conductance which we will study soon) is explained.

The jump in the gravitational counterterm in the electric variables is given by the number of fermions in the ultraviolet gauge theory (this is related to the parity anomaly [24, 201, 202]). Therefore, in our theories

$$\Delta c = -\dim(R) , \qquad (3.3.9)$$

where R is either the rank-two symmetric or antisymmetric representation, respectively.

Our phase diagrams for k < T(R) in figures 3.3 and 3.4 provide us with another way to compute this difference. The two computations must agree for consistency. We start with the TQFT in the asymptotic negative mass phase $SU(N)_{k-T(R)}$ and work our way towards the asymptotic positive mass phase $SU(N)_{k+T(R)}$. This requires tracking the jump of the gravitational counterterm across level/rank dualities, where a gravitational counterterm is generated, and across the phase transitions:

• SU/U TQFT level/rank duality in the asymptotic negative phase [28]:

$$\Delta c_1 = N(k - T(R))$$
(3.3.10)

• Jump induced from positive to negative mass of the leftmost dual gauge theory:

$$\Delta c_2 = \dim(\hat{R}) \tag{3.3.11}$$

• U/U TQFT level/rank duality in the intermediate phase [28]:

$$\Delta c_3 = (T(R) - k)(T(R) + k) - 1 \tag{3.3.12}$$

• Jump induced from positive to negative mass of the rightmost dual gauge theory:

$$\Delta c_4 = \dim(\tilde{R}) \tag{3.3.13}$$

• U/SU TQFT level/rank duality in the asymptotic positive phase [28]:

$$\Delta c_5 = -(k + T(R))N \tag{3.3.14}$$

Here \hat{R} and \tilde{R} denote the representation of the fermion in the leftmost and rightmost dual descriptions in figures 3.3 and 3.4.

Adding up the contributions to the jump following this path we find that it precisely matches that in (3.3.9)

$$\sum_{I=1}^{5} \Delta c_I \equiv \Delta c \,, \tag{3.3.15}$$

both for the theory with a symmetric and antisymmetric fermion!

3.3.3 Baryon Counterterm Matching

An entirely analogous exercise to (3.3.15) is to match the difference in the baryon number conductivity coefficient. This coefficient is simply the difference between the two asymptotic phases in the Chern-Simons term for the baryon background gauge field B, i.e., we are after the difference

$$\frac{\Delta\kappa}{4\pi}\int B\wedge \mathrm{d}B\,.\tag{3.3.16}$$

Here we think about B as an ordinary U(1) connection and the space-time is assumed to be a spin manifold.⁷⁹

We need to carefully normalize the baryon charge of the fermion. The most convenient choice is to imagine that the fermion is in the (anti-)symmetric representation of U(N) rather than SU(N) and the diagonal of U(N) is the baryon number. This would lead to the fermion carrying charge 2. However, the corresponding baryon gauge field would then have possible fractional fluxes and in order to fix that we need to take the charge to be 2/N.

We can therefore compute $\Delta \kappa$ straightforwardly in the electric variables as

$$\Delta \kappa = \left(\frac{2}{N}\right)^2 \dim(R) = \frac{4}{N^2} \frac{1}{2} N(N \pm 1) = \frac{2}{N} (N \pm 1).$$
(3.3.17)

As before, we can also compute $\Delta \kappa$ using the dual description:

• First, in the phase with large negative mass we need to perform level/rank duality between $SU(N)_{k-T(R)}$ and $U(T(R) - k)_N$, which leads to a jump⁸⁰ [28]

$$\Delta \kappa_1 = -\frac{k - T(R)}{N} = -\frac{k - \frac{1}{2}(N \pm 2)}{N}.$$
(3.3.20)

• Next, there is a crucial difference with our computation of the thermal conductivity. Since the baryon symmetry maps to the magnetic symmetry in the dual variables, and

$$\mathcal{L}_{SU(N)_{K}} = \frac{K}{4\pi} \operatorname{tr} \left[b \mathrm{d}b - \frac{2}{3} i b^{3} \right] + \frac{\epsilon_{K}}{4\pi} (\operatorname{tr} b) \mathrm{d}(\operatorname{tr} b) + \frac{1}{2\pi} c \mathrm{d}(\operatorname{tr} b + B)$$
(3.3.18)

where b us a $\mathfrak{u}(N)$ gauge field. If we integrate out c and remove the trace $b := \tilde{b} - \frac{1}{N}B$ (with tr $\tilde{b} \equiv 0$), we get

$$\mathcal{L}_{SU(N)_K} \to \frac{K}{4\pi} \operatorname{tr} \left[\tilde{b} \mathrm{d} \tilde{b} - \frac{2}{3} i \tilde{b}^3 \right] + \frac{K}{4\pi N} B \mathrm{d} B + \frac{\epsilon_K}{4\pi} B \mathrm{d} B \,. \tag{3.3.19}$$

The level/rank dual $U(K)_{-N}$ also has a term $\frac{\epsilon_K}{4\pi}BdB$, so the relative shift by the contact term is only given by the term $\frac{K/N}{4\pi}BdB$.

⁷⁹As all our theories are fermionic and the baryon number clearly satisfies a spin-charge relation, we can in principle also study our theories on spin_c manifolds. This leads to some additional nontrivial consistency checks which we do not present here.

⁸⁰Let us explain briefly how to derive this shift from [28]. Using the notation of [28], the Lagrangian for $SU(N)_K$ is

since the dual fermions are not charged under the magnetic symmetry, we get that $\Delta \kappa_2 \equiv \Delta \kappa_4 \equiv 0$. In other words, the phase transitions do not lead to a jump in the conductivity.

• Next we need to address $\Delta \kappa_3$, which arises from the U/U level/rank duality in the quantum phase. We find again from [28] that

$$\Delta \kappa_3 = 1. \tag{3.3.21}$$

• Finally, the positive-mass SU/U level/rank duality leads to

$$\Delta \kappa_5 = \frac{k + \frac{1}{2}(N \pm 2)}{N} \,. \tag{3.3.22}$$

We see that if we add all the partial jumps $\Delta \kappa_I$ we get precisely the same shift in the baryon counterterm (3.3.17) computed in the electric theory:

$$\sum_{I=1}^{5} \Delta \kappa_I \equiv \Delta \kappa \tag{3.3.23}$$

for both the symmetric and antisymmetric representation!

The matching of the gravitational contact term guarantees that the phase diagram remains consistent in curved space and the thermal conductivities are single-valued, as they should be in physical theories. The matching of the baryon contact term further guarantees that we can consistently gauge the ultraviolet baryon symmetry in all the phases. Therefore, one can derive from our phase diagrams and dualities also the phase diagrams and corresponding dualities for U(N) gauge theories coupled to two-index matter fields.

3.3.4 Self-consistency check for the dualities

The fact that the conductivity coefficient κ is continuous across the phase diagram means that $U(1)_B$ can be consistently gauged. Moreover, given our choice of normalization for the baryon charge of the fermion, one can also gauge the diagonal \mathbb{Z}_N one-form symmetry, and this process conserves the phase diagram as well [29]. The end result is the phase diagram of $U(N)_{k,k'} + \psi$, where ψ is a fermion in the symmetric or antisymmetric representation; the details of these phase diagrams will be discussed elsewhere.

For our purposes, it suffices to note that $SU(N)_k + \psi$ and $U(N)_{k,k'} + \psi$ have a qualitatively identical phase diagram and, in particular, they have the same number of phases. More specifically, either they are both in the two-phase regime, or they both develop an intermediate nonperturbative phase, depending on whether k is larger or smaller than T(R). This implies yet another consistency check of our phase diagram, because the dualities (3.2.5) and (3.2.6)

$$\mathrm{SU}(N)_k + \text{ symmetric } \psi \longleftrightarrow \mathrm{U}\left(\frac{N+2}{2}+k\right)_{\frac{1}{2}(-1+k-3N/2),k-N/2} + \text{ antisymmetric } \tilde{\psi}$$

$$\mathrm{SU}(N)_k + \text{ symmetric } \psi \longleftrightarrow \mathrm{U}\left(\frac{N+2}{2}-k\right)_{\frac{1}{2}(+1+k+3N/2),k+N/2} + \text{ antisymmetric } \hat{\psi}$$

$$SU(N)_{k} + \text{ antisymmetric } \psi \longleftrightarrow U\left(\frac{N-2}{2}+k\right)_{\frac{1}{2}(+1+k-3N/2),k-N/2} + \text{ symmetric } \tilde{\psi}$$

$$SU(N)_{k} + \text{ antisymmetric } \psi \longleftrightarrow U\left(\frac{N-2}{2}-k\right)_{\frac{1}{2}(-1+k+3N/2),k+N/2} + \text{ symmetric } \hat{\psi}$$

$$(3.3.24)$$

are only consistent if one side is in the two-phase regime and the other one is in the three-phase regime, and vice versa⁸¹. It is straightforward to check that this is indeed the case: assuming that 0 < k < T(R), one has

$$-\frac{1}{2}(-1+k-3N/2) > \frac{1}{2}\left[\frac{N+2}{2}+k-2\right] + \frac{1}{2}(+1+k+3N/2) > \frac{1}{2}\left[\frac{N+2}{2}-k-2\right]$$
(3.3.25)

for the antisymmetric case, and

$$-\frac{1}{2}(+1+k-3N/2) > \frac{1}{2}\left[\frac{N-2}{2}+k+2\right] +\frac{1}{2}(-1+k+3N/2) > \frac{1}{2}\left[\frac{N-2}{2}-k+2\right]$$
(3.3.26)

for the symmetric case, as required.

Notice the interplay between the symmetric and antisymmetric representations in the original UV description and in the dual description, which are manifested non-trivially in the above checks. This makes the criteria for quantum phases in (3.1.2) natural.

3.4 Domain Walls in Four Dimensional Gauge Theories

Let us consider the four-dimensional theory of a Dirac fermion in the symmetric/antisymmetric representation coupled to SU(N) gauge fields. We can equivalently think about it as SU(N) gauge theory coupled to a Weyl fermion in the symmetric/antisymmetric representation and another Weyl fermion in the conjugate representation. Let us begin with the massless theory.

⁸¹We would like to thank A. Baumgartner for an interesting discussion regarding this point.

This theory has a $(\mathbb{Z}_{2(N-2)})$ $\mathbb{Z}_{2(N+2)}$ discrete chiral symmetry that acts by re-phasing the two Weyl fermions together,

$$\mathbb{Z}_{2(N\pm2)}: \ \psi, \tilde{\psi} \mapsto e^{i\alpha}\psi, e^{i\alpha}\tilde{\psi} \ , \qquad \alpha = \frac{2\pi k}{2(N\pm2)} \ , \ k \in \mathbb{Z} \ . \tag{3.4.1}$$

In addition the theory enjoys baryon number symmetry $U(1)_B$ which acts by re-phasing the two fermions in an opposite fashion

$$U(1)_B: \ \psi, \tilde{\psi} \mapsto e^{i\beta}\psi, e^{-i\beta}\tilde{\psi} \ . \tag{3.4.2}$$

The special case of N = 2 with a symmetric representation Dirac fermion is equivalent to SU(2) gauge theory with two Weyl fermions in the adjoint representation. In this case the $U(1)_B$ symmetry is in fact enhanced to $SU(2)_F$ flavor symmetry (and in addition, there is the discrete \mathbb{Z}_8 axial symmetry, where the order-two generator in \mathbb{Z}_8 is identified with the center of $SU(2)_F$ flavor symmetry. This order-two generator coincides with the fermion number symmetry, and it is hence unbreakable as long as the vacuum is Poincaré invariant.)

These theories admit a mass deformation, $M\psi\bar{\psi}$, and we can take M to be non-negative without loss of generality, at the expense of having to keep track of the θ parameter of the gauge theory. The mass perturbation breaks the $\mathbb{Z}_{2(N\pm 2)}$ symmetry down to \mathbb{Z}_2 . However, the mass term preserves $U(1)_B$. For $\theta = 0, \pi$ also the time-reversal symmetry is preserved.

In the special case of N = 2, the mass perturbation breaks $SU(2)_F$ symmetry, but it preserves baryon number, which can be identified with the Cartan subgroup of $SU(2)_F$. The Vafa-Witten-like theorems would imply that the massless theory cannot break $U(1)_B$. We will assume that $SU(2)_F$ is broken to $U(1)_B$.

It is reasonable to assume that the massless theory breaks the chiral symmetry $\mathbb{Z}_{2(N\pm 2)}$ as⁸²

$$\mathbb{Z}_{2(N\pm 2)} \to \mathbb{Z}_2 \,. \tag{3.4.3}$$

According to these assumptions, the vacuum structure of the theory is therefore:

- N > 2: The massless theory has $N \pm 2$ vacua, each of which is trivial and gapped. The order parameter distinguishing these vacua is the fermion bilinear $\langle \psi \tilde{\psi} \rangle$, which is charged under $\mathbb{Z}_{2(N\pm 2)}/\mathbb{Z}_2$.
- N = 2: Here $SU(2)_F$ breaks spontaneously to $U(1)_B$ but also the axial symmetry is spontaneously broken. The fermion bilinear is in the adjoint representation and it is

⁸²This can be proven in the planar limit [47], where the theory in the meson sector is equivalent to $\mathcal{N} = 1$ Supersymmetric Yang Mills theory, which is known to develop a condensate. Therefore our statement about the symmetry breaking pattern certainly holds for large enough finite N.
the order parameter leading to this symmetry breaking pattern.⁸³ Let us parameterize it without loss of generality by $\langle \psi \psi \rangle = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}$. This leads to the breaking pattern $SU(2)_F \to U(1)_B$, and the corresponding coset manifold is S^2 . Acting with the generator of \mathbb{Z}_8 we get a new vacuum, and therefore we have at least two copies of the coset S^2 . However, acting with the square of the generator of \mathbb{Z}_8 we get the matrix $\begin{pmatrix} -1 & 0 \\ 0 & 1 \end{pmatrix}$, but this is in fact on the same coset as the original condensate (more precisely, the Weyl group of $SU(2)_F$ relates these two configurations). Hence we have exactly two copies of S^2 , and the broken axial symmetry allows us to move from one copy to the other copy.⁸⁴

Let us now turn on a small positive mass M. This corresponds to a small potential on the above space of vacua which is a function of M, θ . For N > 2, for any $\theta \neq 0$ this lifts the degeneracy and picks up one of the $N \pm 2$ vacua. At $\theta = \pi$ there is a first order transition and two (adjacent) vacua are exactly degenerate. While this analysis is done at small M, it is natural to assume that this is true for any positive M. In particular, at asymptotically large M this agrees with the expectations from pure Yang-Mills theory, which is supposed to have a trivial ground state for any $\theta \neq \pi$ and two degenerate vacua at $\theta = \pi$.

For N = 2 we can again turn on some mass M and fix the theta angle. But now the mass M is an adjoint SU(2) matrix. Without loss of generality we take this matrix to be in the Cartan and hence the eigenvalues are real (and we keep track of θ). Therefore we choose $M = M_0 \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}$ with positive M_0 . The physics depends only on the combination

$$M_0 e^{i\theta/4} \tag{3.4.4}$$

and, furthermore, all the physical observables must be periodic under $\theta \to \theta + 2\pi$.

We can parameterize the vacuum configurations by the adjoint $SU(2)_F$ matrix of fermion condensates

$$\mathcal{M}_{\text{VAC}} = \left\{ U \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} U^{-1} \right\} \cup \left\{ V \begin{pmatrix} i & 0 \\ 0 & -i \end{pmatrix} V^{-1} \right\}$$
(3.4.5)

where U, V are SU(2) matrices. This is just the union of two S^{2} 's, as we explained above. The potential (up to an unimportant proportionality factor) induced by the deformation by

⁸³Indeed, since the order parameter must be a scalar in space-time the Lorentz indices are contracted antisymmetrically, and since it must be gauge invariant, the gauge indices are contracted symmetrically and hence the flavor indices must be contracted symmetrically as well, leading to the symmetric product of the fundamental representation of $SU(2)_F$ with itself, namely, the adjoint representation.

⁸⁴This scenario of the SU(2) gauge theory with two adjoint fermions flowing in the infrared to two copies of S^2 has been recently connected to the Seiberg-Witten solution of the $\mathcal{N} = 2$ vector multiplet theory [211]. Other possibilities for the infrared dynamics were recently discussed also in [212, 213].

 M, θ on the space \mathcal{M}_{VAC} is

$$E_{1} = M_{0}e^{i\theta/4} \operatorname{tr} \left(\begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} U \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} U^{-1} \right) + c.c.$$
(3.4.6)

on the first S^2 and

$$E_2 = iM_0 e^{i\theta/4} \operatorname{tr} \left(\begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} V \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} V^{-1} \right) + c.c.$$
(3.4.7)

on the second S^2 . We need to minimize over U and V and then find the global minimum by comparing these two sectors. First we simplify the expressions for $E_{1,2}$ and write (again, up to unimportant proportionality factors)

$$E_{1} = M_{0} \cos(\theta/4) \operatorname{tr} \left(\begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} U \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} U^{-1} \right),$$

$$E_{2} = M_{0} \sin(\theta/4) \operatorname{tr} \left(\begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} V \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} V^{-1} \right).$$
(3.4.8)

Using the Cauchy-Schwarz inequality we see that the expression in the trace is minimized when the gaugino condensate is $\langle \psi \psi \rangle = -\sigma_z$ and it is maximized when the gaugino condensate is $\langle \psi \psi \rangle = \sigma_z$. We refer to these two points on S^2 as the south and north pole of S^2 . We therefore have the following phases for small enough M_0 , as a function of θ :

- $-\pi < \theta < \pi$: The true vacuum is at the south pole of the first S^2 .
- $\pi < \theta < 3\pi$: The true vacuum is at the south pole of the second S^2
- $3\pi < \theta < 5\pi$: The true vacuum is at the north pole of the first S^2
- $5\pi < \theta < 7\pi$: The true vacuum is at the north pole of the second S^2
- $\theta = \pi, 3\pi, 5\pi, 7\pi$: The two vacua on the two sides of the transition are exactly degenerate.

While the periodicity of the above list of vacua is 8π , of course the physical observables are 2π periodic. In addition, as before, while our analysis is reliable for small M_0 , it is consistent to assume that the bulk vacua behave as above for any M_0 .

We now turn to analyzing domain walls in this theory. We can start from the massless case, which is the most difficult (and the richest) case. We consider first N > 2. Since we have $N \pm 2$ vacua, we can study the domain wall between any pair of vacua. However, using the spontaneously broken axial symmetry we see that the result only depends on the difference of the phases of the gaugino condensates on the two sides. Let us label the vacua by $J = 1, \ldots, N \pm 2$ according to the phase of the gaugino condensate $\langle \psi \tilde{\psi} \rangle = e^{\frac{2\pi i J}{N \pm 2}}$. A

natural conjecture for the domain wall theory is to identify it with the quantum phase in the corresponding 2 + 1 dimensional gauge theory of SU(N) with a symmetric (antisymmetric) fermion. This leads to the proposal that the theory on the domain wall connecting the vacuum I and the vacuum J. We can choose the orientation such that without loss of generality, say, I > J and J is on the right hand side of the wall. The other cases are obtained by simply reversing the orientation. We thus identify the rank of the gauge group of the quantum phase in (3.1.2) with the jump I - J, that is, $\frac{N\pm 2}{2} - k \leftrightarrow I - J$. Equivalently, the 2 + 1 level maps to $k \leftrightarrow \frac{N\pm 2}{2} - (I - J)$, whence the domain wall theory is

$$U(I-J)_{N\pm 2-I+J,N\pm 2-2I+2J} \quad . \tag{3.4.9}$$

Of course, this TQFT has to be accompanied by the decoupled center of mass degree of freedom (which is described to leading order by the Nambu-Goto action).

The proposal (3.4.9) is a generalization of the Acharya-Vafa theory for SYM [46]. Note a few interesting facts that follow from (3.4.9).

- If we take I = J + 1 we obtain the TQFT $U(1)_N$ in the case of the symmetric fermion, and $U(1)_{N-4}$ in the case of the antisymmetric fermion. Note that for finite (positive) M, and $\theta = \pi$ precisely these two adjacent vacua are degenerate and hence this domain wall continues to exist also at finite M. At very large M we can integrate out the fermion and remain with the pure Yang-Mills theory, where the domain wall is given by $U(1)_N$ (more precisely, the domain wall of pure Yang-Mills theory is level/rank dual to the $U(1)_N$ TQFT). We see that in the symmetric fermion case no phase transition occurs. This is similar to the theory with the adjoint fermion. But in the case of the antisymmetric fermion we see that a phase transition does occur as we crank up the mass. If the phase transition is second order, it would be natural to assume that it is in the same universality class as the corresponding 3d phase transition, namely, it is given by $U(1)_{N-2}$ plus a charge-2 fermion. This discussion (beautifully) makes sense also for the degenerate case N = 2, where the corresponding theory with the antisymmetric tensor is equivalent to the pure Yang-Mills theory with a neutral Dirac fermion. The domain wall theory is always $U(1)_2$ (here we use that $U(1)_2 \simeq U(1)_{-2}$) and the $U(1)_0$ plus a charge-2 fermion leads to a massless fermion on the wall [214–217], which can be thought of as arising precisely from the massless Dirac fermion in the bulk!
- In general, combining the four-dimensional spontaneously broken axial symmetry and time-reversal symmetry, we can derive that the domain wall theory connecting I and J should be isomorphic to the one connecting $N \pm 2 I + J$ and 0. Indeed, this is merely the statement that

$$U(I-J)_{N\pm 2-I+J,N\pm 2-2I+2J} \simeq U(N\pm 2-I+J)_{J-I,N\pm 2-2I+2J}$$
(3.4.10)

which is nothing but level/rank duality in three dimensions.

• If N is even then there exists a time-reversal invariant domain wall, given by $I - J = \frac{N \pm 2}{2}$. The corresponding theory on the wall is $U\left(\frac{N \pm 2}{2}\right)_{\frac{N \pm 2}{2},0}$. We discussed in detail that this theory should be interpreted as a massless NGB coupled to some TQFT. While baryon symmetry is not broken in the bulk, we see that it is broken on the wall! In principle, this domain wall theory is a nonperturbative object and it is hard to understand where this massless NGB comes from and why (recall that in the case of the Acharya-Vafa domain walls, there is no such massless field). We can think of this domain wall theory as a bound state of $\frac{N \pm 2}{2}$ elementary domain walls (the elementary domain walls are the ones described in the first bullet point). There is however one case, N = 2 with a symmetric fermion, where this massless NGB can be seen explicitly. We describe this mechanism below.

Now let us discuss in detail the case of N = 2 with a Dirac fermion in the adjoint representation. Recall that, as explained above, instead of having N + 2 = 4 isolated trivial vacua we have two copies of S^2 . The "elementary" domain wall corresponds to connecting the south pole of the first S^2 and the south pole of the second S^2 (or any isomorphic configuration thereof). This wall is hard to understand since it passes in regions of field space which are not within our effective theory. However, our prediction above for the physics of this domain wall is the U(1)₂ TQFT, which makes a lot of sense (in the limit of softly deformed pure $\mathcal{N} = 2$ SYM theory, this result can be derived directly from the Seiberg-Witten solution along the lines of [218]).

The bound state of two such elementary walls corresponds to a jump of θ by 4π and what it does is to connect the north pole and the south pole of the same given S^2 . Now, the wall can be analyzed entirely within classical field theory, since the wall is merely a geodesic running from the north to the south pole of the O(3) NLSM. The massless bosonic mode that our 3d model predicts is simply the azimuthal degree of freedom of that geodesic trajectory, see figure 3.7. Therefore, the domain wall has indeed a compact NGB which corresponds to the spontaneous breaking of baryon symmetry. In figure 3.8 we depict the four vacua at the north and south poles of the two S^2 's as the vertices of the square and we draw the two S^2 's that stretch along the diagonals, parameterizing the vacua of the theory.

For this story to hold up it is essential that the TQFT that accompanies the NGB is trivial. Indeed, it is $PSU(2)_2 = SO(3)_1$ which is a trivial theory. In the four dimensional theory there is nowhere for the domain wall to obtain a Chern-Simons term from since the sphere has the standard non-singular round metric and the domain wall can be understood entirely within effective field theory. Therefore the NGB in the infrared is not accompanied by a TQFT, in accordance with our prediction from the 3d analysis.

Note that the emergence of the NGB on the wall here is quite analogous to the way the symmetry breaking phases of QCD_3 emerge from the corresponding four-dimensional construction [41]. For similar constructions see also [40, 219–225]



Figure 3.7: These figures represent the NG boson that arises on the domain wall connecting the vacuum at the north pole with the vacuum at the south pole. The first figure is the NG boson on the domain wall and the second and third figures represent the bulk vacua.



Figure 3.8: Moduli space of SU(2) plus a symmetric fermion, consisting of two copies of S^2 (the S^2 's in the figure are stretched only for the convenience of the picture).

3.5 Comments on Future Directions

Our theories with m = k = 0 all have a time-reversal symmetry, satisfying

$$\mathsf{T}^2 = (-1)^F. \tag{3.5.1}$$

It is therefore possible to study these theories on pin_+ manifolds. It is well known that there is a purely gravitational \mathbb{Z}_{16} time-reversal anomaly in such cases [56, 87–89, 108]. We denote

the anomaly by $\nu \in \mathbb{Z}_{16}$. Here we would like to briefly mention some new aspects of this anomaly, which will be explained in more detail elsewhere.

A naive approach to computing the anomaly ν in the ultraviolet is to disregard the gauge interactions and compute the total (weighted) number of Majorana fermions. This seems physically justified because the gauge interactions are arbitrarily weak in the deep ultraviolet. However, in [30] it was noted that this procedure is often ambiguous; it is not invariant under gauge transformations, in the sense that if we compose T with a gauge transformation then ν may change by an integer which is not a multiple of 16.

In [34] it was pointed out that in theories where the gauge group is not simply connected, the time-reversal symmetry algebra could be deformed by the magnetic symmetry and as a result ν would not be well defined. Here we will demonstrate that $\nu \in \mathbb{Z}_{16}$ cannot be canonically defined (in general) in theories with a one-form symmetry. This even happens in theories without a magnetic symmetry (i.e. based on simply connected gauge groups). The lack of a canonical choice of ν has to do with the fact that the full symmetry group of the theory has more generators than just T, and there is no canonical way to set to zero the background fields for the one-form symmetry. See below.

Let us consider a simple example, which was already studied in [30]. Consider SU(2) gauge theory coupled to a single Majorana adjoint fermion (and k = m = 0). Let the time-reversal symmetry act in the standard way, which leads to $\nu = 3$, simply because there are three Majorana fermions in total. However, let us now compose T with the gauge transformation $U = i\sigma_y$. The (traceless) Hermitian matrix of Majorana fermions transforms under this gauge transformation as

$$\begin{pmatrix} \Psi & \chi \\ \chi^{\dagger} & -\Psi \end{pmatrix} \mapsto \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} \Psi & \chi \\ \chi^{\dagger} & -\Psi \end{pmatrix} \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix} = \begin{pmatrix} -\Psi & -\chi^{\dagger} \\ -\chi & \Psi \end{pmatrix} .$$
(3.5.2)

If our original time-reversal symmetry acts as $\mathsf{T}\Psi = \gamma^0 \Psi$ and $\mathsf{T}\chi = \gamma^0 \chi^{\dagger}$ as usual, then combining this with the gauge transformation we see that $\mathsf{T} \circ U\Psi = -\gamma^0 \Psi$ and $\mathsf{T} \circ U\chi = -\gamma^0 \chi$. Remembering that $\mathsf{T} \circ U$ is antilinear (squaring to $(-1)^F$) we thus see that now it would appear that the time-reversal anomaly is $\nu = -1$. We therefore clearly see that in theories with a one-form symmetry the $\nu \in \mathbb{Z}_{16}$ anomaly is not uniquely defined in spite of the fact that there is no magnetic symmetry. The same phenomenon takes place in many of the examples studied in [30, 34]. We will now give a physical as well as a mathematical interpretation of this phenomenon.

The SU(2) gauge theory coupled to the adjoint Majorana fermion was claimed to flow in the infrared to a free massless Majorana fermion alongside with the U(1)₂ TQFT [30]. The time-reversal anomaly of the U(1)₂ TQFT could be either $\nu = +2$ or $\nu = -2$ – this depends on the time-reversal transformation of the semion in the TQFT. The contribution of the decoupled Majorana fermion is always $\nu_{\text{Majorana}} = +1$ in our conventions (i.e. with our choice of orientation). Therefore the total time-reversal anomaly of the infrared theory is either -1 or 3, which is in exact agreement with the values found above in the ultraviolet gauge theory. (Analogous matching of the multiple possible values in the ultraviolet of ν can be carried out in the examples appearing in [30, 34].) Indeed, the ultraviolet theory has an unscreened Wilson line in the fundamental representation and the properties of the particle defining the worldline are not part of the definition of the ultraviolet theory.

A closely related point of view [212] is that we can break the one-form symmetry by adding heavy particles that transform faithfully under the center of the gauge group. We need to assign time-reversal transformations to these new particles. The fundamental Wilson line is now screened by these particles. In particular, $\nu \in \mathbb{Z}_{16}$ is well defined and the transformation properties of the infrared semions are completely determined in the presence of these heavy particles.⁸⁵

The mathematical interpretation⁸⁶ is that one can show that in the presence of a \mathbb{Z}_2 one-form symmetry the value of ν can be shifted by a change of variables involving the two-form gauge field and w_1^2 . This can lead to $\Delta \nu = 4$ which is precisely what we have found above. Ideally in such theories we should compute the full anomaly polynomial involving the two-form gauge field B and the time-reversal gauge field w_1 , and this should be compared across renormalization group flows and dualities. We leave this for the future.

An additional subtlety that arises in our theories (but not in the theories with an adjoint Majorana) at m = k = 0 is the presence of a massless scalar field. In the presence of such a gapless mode, it is nontrivial to evaluate the infrared contribution to ν (and to various other discrete anomalies).

To avoid the complications of having the Nambu-Goldstone boson when we study the theory on pin_+ manifolds, we could try to add a baryon to the Lagrangian, in such a way that some time-reversal symmetry remains and at the same time the Nambu-Goldstone boson would be lifted due to the explicit breaking of baryon symmetry.

This brings us back to the discussion of which baryon, in fact, condenses (see e.g. [226] for a discussion of some baryons in such theories). It turns out that the baryons in these theories are not as simple as one may initially expect. For instance, the naive baryons for the rank-two representation that were discussed in the literature

$$\epsilon_{i_1\cdots i_N}\epsilon_{j_1\cdots j_N}\psi^{i_1j_1}\cdots\psi^{i_Nj_N} \tag{3.5.3}$$

vanish identically because of Fermi statistics:

$$\epsilon_{i_1\dots i_N}\epsilon_{j_1\dots j_N}\psi^{i_1j_1}\psi^{i_2j_2}\psi^{i_3j_3}\dots \equiv 0 \tag{3.5.4}$$

for arbitrary values of the spinor indices and regardless of which other insertions are used. To construct a baryon with minimal baryon charge one therefore has to add various derivatives, insertions of the field strength or mesons. It would be nice to return to this in the future.

 $^{^{85}\}mathrm{We}$ thank T. Senthil for discussions on this.

⁸⁶We thank R. Thorngren for collaborating with us on this result.

3.6 Review of QCD_3 dynamics.

In this section we collect some known phase diagrams of QCD in d = 2 + 1 dimensions. We mostly focus on [3, 29, 30, 113]; we make no attempt at reviewing the subject comprehensively since the existing literature is huge. For concreteness we restrict ourselves to either N_F fundamental quarks, or $N_F = 1$ quarks in rank-2 representations; and to simple groups. In the literature one can also find $N_F = 2$ rank-2 quarks [130], as well as mixtures of quarks and scalars [227–231], including theories with supersymmetry [127–129, 232, 233]; and also quiver gauge theories [203, 234–236] and exceptional Lie groups [237]. The general picture is qualitatively similar although the finer points require a more specific analysis.

The following basic level-rank dualities of Chern-Simons theories play an important role (we take k > 0) [28, 112, 113]:

$$\begin{array}{rcl}
\operatorname{SU}(N)_k &\longleftrightarrow & \operatorname{U}(k)_{-N,-N} \\
\operatorname{SO}(N)_k &\longleftrightarrow & \operatorname{SO}(k)_{-N} \\
\operatorname{Spin}(N)_k &\longleftrightarrow & \operatorname{O}(k)_{-N,-N} \\
\operatorname{Sp}(N)_k &\longleftrightarrow & \operatorname{Sp}(k)_{-N}
\end{array}$$
(3.6.1)

We schematically write these dualities as $G(N)_k \leftrightarrow \hat{G}(k)_{-N}$, where \hat{G} is the "dual" group. Clearly, $\hat{G} = G$. Note that $\hat{G}(N_F)$ can be thought of as the flavor symmetry of the gauge theory with gauge group G(N).

Other important dualities are

Note that, under level-rank duality $G(N)_k \leftrightarrow \hat{G}(k)_{-N}$, integrable representations are exchanged as $R \leftrightarrow R^t \times f^r$, where R^t the representation whose Young diagram is the transpose of that of R, f is the transparent fermion, and r is the rank of R (i.e., the number of boxes in its Young diagram). Therefore, at a very heuristic level, it is reasonable to expect that if we deform G(N) by a field in the representation R, then the duality remains true if we also deform $\hat{G}(k)$ by a field in the representation R^t , with the same fermion parity as R if r is even, and with the opposite parity if odd. For example, adding fundamental fermions to G is equivalent to adding fundamental bosons to \hat{G} ; and adding symmetric fermions is equivalent to adding ant-symmetric fermions to the dual.

With this in mind, consider now QCD. We denote the gauge group as G(N), the renormalized Chern-Simons level as k, and the quark representation as R. The level is defined as transforming as $k \to -k$ under time-reversal. The gauge theory is denoted as $G(N)_k + \psi_R$. The Dynkin index of the representation is denoted as T(R), and it is scaled up by a factor of 2 when the representation is complex (so, for example, the fundamental representation always has $T(\Box) = 1$, whether the group is SU(N) or SO(N)). The general structure of the phase diagram is the following. When the level is large enough, presumably $k \ge T(R)/2$, the theory is in the two-phase regime, namely the infrared phases are the two semiclassical phases that follow from integrating out massive fermions:

$$\begin{array}{ccc}
G(N)_{k} + \psi_{R} & k \ge T(R)/2 \\
 & & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\$$

In the case of fundamental quarks $R = N_F \Box$, the transition point admits a weakly coupled dual theory, which takes the form [238]

$$\hat{G}(k + N_F/2)_{-N} + N_F \phi_{\Box}$$
 (3.6.4)

where \hat{G} is the dual group (cf. (3.6.1)). The masses are mapped as $\operatorname{sign}(m) = \operatorname{sign}(m'^2)$. Integrating out the fields, the two dual theories reproduce the same phase thanks to level-rank duality. For higher-rank fermions, no weakly coupled dual theory is known.

When the level is small, presumably k < T(R)/2, an intrinsically quantum phase opens up, and the transition points admit weakly coupled dual theories. The dual theory has matter fields in the representation R^t , and these are bosons if R is a rank-1 representation, and fermions if a rank-2 representation.

The diagram in the quantum regime is as follows. When R is the fundamental representation, the dual theory contains scalars:

$$G(N)_k + N_F \psi_{\Box} \qquad \qquad k \le N_F/2$$



Here the intermediate phase represents a sigma model over the displayed manifold, and it is the consequence of the spontaneous breaking of the flavor symmetry by a fermion bilinear condensate; this breaking is manifest in the dual variables. The sigma model also includes a Wess-Zumino term at level N. The mass deformations are mapped as $\operatorname{sign}(m) = -\operatorname{sign}(m'^2)$ for the left transition, and as $\operatorname{sign}(m) = +\operatorname{sign}(m'^2)$ for the right transition.

When R is a rank-2 representation, the dual theory contains fermions:

$$G(N)_k + \psi_R \qquad \qquad k \le T(R)/2$$



Here the intermediate phase is a topological phase; it is a non-perturbative phase from the point of view of the original fermions, but it is semi-classical with respect to the dual fermions. The mass deformations are mapped as $\operatorname{sign}(m) = -\operatorname{sign}(m')$. The two descriptions of the intermediate phase are apparently inequivalent, but they are in fact identical thanks to the level-rank dualities above.

Disconnected and non-simple groups have more than one level. The phase diagrams above require the following additions:

- In the case of fundamental quarks, the dual to $\operatorname{Spin}(N)$ is O(k), and the \mathbb{Z}_2 level is zero throughout, i.e., $\hat{G} \to O(\cdots)^0$.
- In the case of rank-2 quarks, the dual to $\operatorname{Spin}(N)$ is O(k), and the \mathbb{Z}_2 level is $\pm 1/2$ on the dual gauge theories, and 1 in the intermediate phase, i.e., the theories are $O(\cdots)^{\pm 1/2} + \psi_{R^t}$ and $O(\cdots)^1$ respectively.
- In the case of rank-2 quarks, the dual to SU(N) is U(k), and the U(1) level is $\mp N \pm \frac{q}{T(R)\pm k} \dim(R^t)$ on the dual theories, and $\mp N \pm \frac{2q}{T(R)\pm k} \dim(R^t)$ in the intermediate phase, where q is the U(1) charge of the fermions, i.e., q = 0 for adjoint quarks and q = 2 for symmetric and anti-symmetric quarks.
- It should also be noted that at small m, the theory with adjoint fermions is supersymmetric, and this symmetry is spontaneously broken, hence the infrared contains a

gapless point where, on top of the TQFT above, there is a decoupled Majorana fermion (i.e., a Goldstino).

Note that the picture above is only consistent if the dual theories are in the two-phase regime when the original theory is in the three-phase regime – indeed, the intermediate phase was derived by simply integrating out the fields in the dual theory. An immediate consistency check for the rank-2 phase diagrams comes from noticing that the dual theories are rank-2 gauge theories too, and therefore $k \leq T(R)/2$ must hold precisely when $N > T(R^t)$. Using the actual values of T(R) and $T(R^t)$ this is indeed seen to hold.

A simple test in the case of fundamental matter comes from looking at the k = 0 case, which is time-reversal invariant. The Vafa-Witten analysis [239] strongly suggests that here, the flavor symmetry is broken as $\hat{G}(N_F) \rightarrow \hat{G}(N_F/2) \times \hat{G}(N_F/2)$, consistently with the general arguments above. This, if correct, then implies the case of general k. Indeed, if the picture is correct for a given (N_F, k) , then it is also correct for $(N_F - 1, k \pm 1/2)$, as can be seen by turning on a mass for a single flavor in the conjectured dualities. This mass gaps out that single fermion, which decreases $N_F \rightarrow N_F - 1$ and it also shifts $k \rightarrow k \pm 1/2$, depending on the sign of the deformation.

3.7 Time-reversal and flavor symmetries.

A non-trivial consistency check comes from looking at how time-reversal acts on the phase diagram. Generically, this is not a symmetry, as it is explicitly broken by m and k. We can formally restore this symmetry by letting it act on these parameters, i.e., we take $(m,k) \rightarrow (-m,-k)$. In a loose sense, we think of m, k as some sort of spurions. In practice, this means that, if we reverse the orientation of spacetime (i.e., reverse the signs of all Chern-Simons levels), and also manually reverse the sign of m and k, the diagram should stay invariant. This is manifestly true for the diagram as written,⁸⁷ but becomes a much more stringent condition once we turn on background fields.

We check this as follows. First, we turn on background fields for all symmetries – in our case, gravity and the baryon symmetry – and adjust their Chern-Simons coefficients in the ultraviolet to their time-reversal invariant point (i.e., the renormalized levels for background fields are chosen to vanish). Then, we track how this coefficient changes as we move around the diagram. The coefficient found in the infrared quantum phase must be the same whether we compute it via the left part of the diagram (the negative mass phase) or the right part of the diagram (the positive mass phase). Furthermore, this coefficient must be an odd function

⁸⁷This explains why the mass deformation was chosen as sign(m) = -sign(m') for the fermion-fermion dualities (see below (3.6.5)), but as $sign(m) = \pm sign(m'^2)$ for the fermion-boson dualities (see below (3.6.6)): the mass of a scalar is even under time-reversal, while the mass of the fermion is odd. Therefore, the two transitions must have the same sign in the case of fermion-fermion, but opposite signs for fermion-boson.

of k; this last condition is what extends the discussion of sections 3.3.2 and 3.3.3 (there we only checked that the coefficient matched across the diagram, but not whether it was odd under $k \to -k$).

We next perform this test for the phase diagrams above. In all cases we checked, the test works out correctly and thus the phase diagrams are consistent as far as time-reversal and background symmetries is concerned. For a given Lie algebra, it is enough to only check one form of the gauge group (e.g., for $\mathfrak{so}(N)$ we only do SO(N) instead of also doing Spin(N)). The computation is insensitive to the global form of the group.

The Chern-Simons term for the gravitational term is denoted as CS[g] and that for the baryon symmetry as CS[B]. In the adjoint case, the infrared includes a Goldstino, which we shall denote by χ .

3.7.1 Unitary group.

Consider $SU(N)_k + \psi_R$. The asymptotic phases are

$$SU(N)_{k\pm T(R)/2} \qquad U(T(R)/2\pm k)_{\mp N} \\
\pm \frac{1}{2}\dim(R)CS[g] \qquad \longleftrightarrow \qquad \pm \left(\frac{1}{2}\dim(R) - N(T(R)\pm 2k)\right)CS[g] \qquad (3.7.1) \\
\pm \frac{1}{2}\frac{Q-T(R)}{N}CS[B] \qquad \qquad \pm \frac{1}{2}\frac{Q-T(R)}{N}CS[B]$$

Here $Q := \frac{1}{2}q^2 \dim(R)/N$ is the charge under U(1), with q the number of boxes of R. In particular, $Q(\Box) = N_F$, Q(adj) = 0, $Q(\Box) = 2(N-1)$, and $Q(\Box\Box) = 2(N+1)$.

The dual theories are

- $R = N_F \square$: $U(N_F/2 \pm k)_{\mp N} + N_F \phi_{\square} - 2Nk \operatorname{CS}[g] \pm 0 \operatorname{CS}[B]$ (3.7.2)
- R = adj:

$$U(N/2 \pm k)_{\mp \frac{3}{4}N + \frac{1}{2}k, \mp N} + \psi_{\text{adj}}$$

$$\mp \left(\frac{3}{8}N^2 - \frac{3}{2}Nk \pm \frac{1}{2}k^2 \mp 1\right) \text{CS}[g] \mp \frac{1}{2} \text{CS}[B]$$
 (3.7.3)

• $R = \square$

$$U((N-2)/2 \pm k)_{\mp \frac{3}{4}N + \frac{1}{2}k \pm \frac{1}{2}, k \mp \frac{1}{2}N} + \psi_{\Box} = \\ \mp \left(\frac{3}{8}N^2 - \frac{3}{2}Nk \pm \frac{5}{4}N \pm \frac{1}{2}k^2 - \frac{1}{2}k\right) CS[g] \pm \frac{1}{2} CS[B]$$
(3.7.4)

• $R = \square$:

$$U((N+2)/2 \pm k)_{\mp \frac{3}{4}N + \frac{1}{2}k \mp \frac{1}{2}, k \mp \frac{1}{2}N} + \psi_{\Box}$$

$$\mp \left(\frac{3}{8}N^2 - \frac{3}{2}Nk \mp \frac{5}{4}N \pm \frac{1}{2}k^2 + \frac{1}{2}k\right) CS[g] \pm \frac{1}{2} CS[B]$$
(3.7.5)

Integrating out the fermions in the dual theories, the intermediate phase is found to be

• $R = N_F \square$:

$$\frac{\mathrm{U}(N_F)}{\mathrm{U}(N_F/2+k) \times \mathrm{U}(N_F/2-k)} + N\Gamma - 2Nk \operatorname{CS}[g] + 0 \operatorname{CS}[B]$$
(3.7.6)

• R = adj:

$$U(N/2 \pm k)_{\mp \frac{1}{2}N + k, \mp N} + \chi \mp \left(\frac{1}{4}N^2 - Nk \pm k^2 \mp \frac{3}{2}\right) CS[g] \mp \frac{1}{2} CS[B]$$
(3.7.7)

- $R = \square$ $U((N-2)/2 \pm k)_{\pm \frac{1}{2}(N-2)+k,2k} \pm (\frac{1}{4}N^2 - Nk \pm N \pm k^2 - k)CS[g] \pm \frac{1}{2}CS[B]$ (3.7.8)
- $R = \square$:

$$U((N+2)/2 \pm k)_{\mp\frac{1}{2}(N+2)+k,2k} \mp \left(\frac{1}{4}N^2 - Nk \mp N \pm k^2 + k\right) CS[g] \pm \frac{1}{2} CS[B] \quad (3.7.9)$$

The central charge of the infrared phase is

- $R = N_F \square$: c = Nk (3.7.10)
- R = adj:

$$c = k \left[\frac{k^2 - 1}{N} + \frac{1}{4}N \right]$$
(3.7.11)

- $R = \square$ $c = k \left[\frac{k^2 - 1}{N - 2} + \frac{1}{4}N + 1 \right] + \operatorname{sign}(k)$ (3.7.12)
- $R = \square$: $c = k \left[\frac{k^2 - 1}{N + 2} + \frac{1}{4}N - 1 \right] + \operatorname{sign}(k)$ (3.7.13)

In all cases we see that c is an odd function of k, as expected.

Furthermore, the infrared coefficient of CS[B] matches when coming from both directions, as it should; indeed, recall that level-rank duality reads

$$U(N)_{+N,+(N\pm k)}[B] \quad \longleftrightarrow \quad U(k)_{-N,-(N\pm k)}[B] \mp \operatorname{CS}[B]$$
(3.7.14)

Not only does the counterterm match, but its value is precisely the one required by timereversal.

3.7.2 Orthogonal group.

Consider $SO(N)_k + \psi_R$ with R a rank-1 or rank-2 representation. The asymptotic phases are

$$\begin{array}{cccc}
\mathrm{SO}(N)_{k\pm T(R)/2} & \longleftrightarrow & \mathrm{SO}(T(R)/2\pm k)_{\mp N} \\
\pm \frac{1}{2}\dim(R)\mathrm{CS}[g] & \longleftrightarrow & \pm \left(\frac{1}{2}\dim(R) - N(\frac{1}{2}T(R)\pm k)\right)\mathrm{CS}[g]
\end{array}$$
(3.7.15)

The dual theories are

- $R = N_F \square$: $\operatorname{SO}(N_F/2 \pm k)_{\mp N} + N_F \phi_{\Box} - Nk \operatorname{CS}[g]$ (3.7.16)
- R = adj:

$$SO((N-2)/2 \pm k)_{\mp \frac{3}{4}N + \frac{1}{2}k \pm \frac{1}{2}} + \psi_{\Box\Box} \mp \left(\frac{3}{16}N^2 - \frac{3}{4}Nk \pm \frac{5}{8}N \pm \frac{1}{4}k^2 - \frac{1}{4}k \mp \frac{1}{2}\right) CS[g] (3.7.17)$$

• $R = \square$:

$$SO((N+2)/2\pm k)_{\mp\frac{3}{4}N+\frac{1}{2}k\pm\frac{1}{2}} + \psi_{adj} \mp \left(\frac{3}{16}N^2 - \frac{3}{4}Nk \mp \frac{5}{8}N \pm \frac{1}{4}k^2 + \frac{1}{4}k \mp \frac{1}{2}\right) CS[g] (3.7.18)$$

Integrating out the fermions in the dual theories, the intermediate phase is found to be

- $R = N_F \square$: $\frac{\mathrm{SO}(N_F)}{\mathrm{SO}(N_F/2 + k) \times \mathrm{SO}(N_F/2 - k)} + N\Gamma - Nk \operatorname{CS}[g] \qquad (3.7.19)$
- R = adj:

$$SO((N-2)/2 \pm k)_{\pm \frac{1}{2}(N-2)+k} + \chi \mp \left(\frac{1}{8}N^2 - \frac{1}{2}Nk \pm \frac{1}{2}N \pm \frac{1}{2}k^2 - \frac{1}{2}k \mp 1\right)CS[g] (3.7.20)$$

• $R = \square$:

$$SO((N+2)/2 \pm k)_{\pm \frac{1}{2}(N+2)+k} \mp \left(\frac{1}{8}N^2 - \frac{1}{2}Nk \mp \frac{1}{2}N \pm \frac{1}{2}k^2 + \frac{1}{2}k \mp \frac{1}{2}\right) CS[g] \quad (3.7.21)$$

The central charge of the infrared phase is

•
$$R = N_F \square$$
:
 $c = \frac{1}{2}Nk$ (3.7.22)

- $R = \operatorname{adj:}$ $c = \frac{k}{2(N-4)} \left[\frac{1}{4} N(N-2) + k^2 - 3 \right]$ (3.7.23)
- $R = \square$: $c = \frac{k}{2N} \left[\frac{1}{2} N(N-6) + k^2 - 1 \right]$ (3.7.24)

In all cases we see that c is an odd function of k, as expected.

3.7.3 Symplectic group.

Consider $Sp(N)_k + \psi_R$ with R a rank-1 or rank-2 representation. The asymptotic phases are

$$\begin{array}{cccc}
\operatorname{Sp}(N)_{k\pm T(R)/2} & \longleftrightarrow & \operatorname{Sp}(T(R)/2\pm k)_{\mp N} \\
\pm \frac{1}{2}\operatorname{dim}(R)\operatorname{CS}[g] & \longleftrightarrow & \pm \left(\frac{1}{2}\operatorname{dim}(R) - 2N(T(R)\pm 2k)\right)\operatorname{CS}[g] \\
\end{array} (3.7.25)$$

The dual theories are

• $R = N_F \square$:

$$\operatorname{Sp}(N_F/2 \pm k)_{\mp N} + N_F \phi_{\Box} - 4Nk \operatorname{CS}[g]$$
 (3.7.26)

• R = adj:

$$\operatorname{Sp}((N+1)/2 \pm k)_{\mp \frac{3}{4}N + \frac{1}{2}k \mp \frac{1}{4}} + \psi_{\Box} \mp \left(\frac{3}{4}N^2 - 3Nk \mp \frac{5}{4}N \pm k^2 + \frac{1}{2}k \mp \frac{1}{2}\right) \operatorname{CS}[g] (3.7.27)$$

• $R = \square$: $Sp((N-1)/2 \pm k)_{\mp \frac{3}{4}N + \frac{1}{2}k \pm \frac{1}{4}} + \psi_{adj} \mp \left(\frac{3}{4}N^2 - 3Nk \pm \frac{5}{4}N \pm k^2 - \frac{1}{2}k \mp \frac{1}{2}\right) CS[g] (3.7.28)$

Integrating out the fermions in the dual theories, the intermediate phase is found to be

•
$$R = N_F \square$$
:

$$\frac{\operatorname{Sp}(N_F)}{\operatorname{Sp}(N_F/2 + k) \times \operatorname{Sp}(N_F/2 - k)} + N\Gamma - 4Nk \operatorname{CS}[g] \qquad (3.7.29)$$

• R = adj:

$$\operatorname{Sp}((N+1)/2 \pm k)_{\mp \frac{1}{2}(N+1)+k} + \chi \mp \left(\frac{1}{2}N^2 - 2Nk \mp N \pm 2k^2 + k \mp 1\right) \operatorname{CS}[g] \quad (3.7.30)$$

• $R = \square$:

$$\operatorname{Sp}((N-1)/2 \pm k)_{\mp \frac{1}{2}(N-1)+k} \mp \left(\frac{1}{2}N^2 - 2Nk \pm N \pm 2k^2 - k \mp \frac{1}{2}\right) \operatorname{CS}[g] \quad (3.7.31)$$

The central charge of the infrared phase is

• $R = N_F \square$:

$$c = 2Nk \tag{3.7.32}$$

• R = adj:

$$c = \frac{k}{2(N+2)} \left[N(N+1) + 4k^2 - 3 \right]$$
(3.7.33)

• $R = \square$:

$$c = \frac{k}{2N} \left[N(N+3) + 4k^2 - 1 \right]$$
(3.7.34)

In all cases we see that c is an odd function of k, as expected.

3.8 Time-reversal anomaly.

Another non-trivial consistency check comes from looking at the pure time-reversal anomaly. In d = 2 + 1, an anti-unitary symmetry T with algebra $T^2 = (-1)^F$ has a \mathbb{Z}_{16} valued 't Hooft anomaly, often denoted as ν . For systems continuously connected to a free fermion phase, this anomaly is easily computed as follows.

Consider a system of free Majorana fermions (see A.2 for a quick review). The most general time-reversal operation is $\psi \mapsto V\gamma^0 \psi$, where V is a real matrix (that does not act in spinor space). This operation satisfies $\mathsf{T}^2 = (-1)^F$ if $V^2 = 1$. Note that such a matrix is diagonalizable, and has eigenvalues ± 1 ; the anomaly associated to this transformation is the number of positive eigenvalues, minus the number of negative ones, modulo 16 [56]. In other words,

$$\nu = \operatorname{tr} V \mod 16 \tag{3.8.1}$$

As this is invariant under smooth deformation, this formula remains true for interacting theories, as long as they can be continuously connected to the free phase. Such is the case, for example, in QCD, where $g \to 0$ turns off the interactions and leaves us with free fermions (and free gluons, which do not contribute to ν). The value of ν is also invariant under RG flow, and therefore it must be reproduced by whatever phase emerges at large distances, in particular, by the conjectured quantum phase in the phase diagrams reviewed above.

In QCD, the matrix V can be chosen to act on both the color indices and the flavor indices. For simplicity we shall restrict ourselves to color indices only, but the idea for the general case is the same. In this situation, for any matrix V in color space that squares to 1, we will have a value of ν , which must be matched by the infrared phase. If we denote the gauge group by G, and the quark representation as R, then the matrix V takes the form V = R(g) for any g such that $R(g^2) = 1$. In other words, for any $g \in G$ such that $g^2 \in \ker R$, the value of $\nu \equiv \chi_R(g) := \operatorname{tr}_R g$ modulo 16 is an invariant of the system, which is constant along any T-preserving deformation. Here χ_R denotes the group character of R, multiplied by 2 if R is complex (as we count Majorana fermions). Note that ν depends only on the conjugacy class of g, so it is enough to look at a maximal torus of G. Even simpler, note that for non-trivial R one has ker $R \subseteq Z(G)$.

We will next compute the value of ν for all the phase diagrams reviewed above, and check for 't Hooft anomaly matching whenever it is possible. Note that $G_k + \psi_R$ is time-reversal invariant only when k = 0. The bare level k - T(R)/2 has to be integral, which means that k = 0 only exists when T(R) is even.

3.8.1 Unitary group.

The special unitary group SU(N) is defined as the set of $N \times N$ complex matrices such that $gg^{\dagger} = \mathbf{1}_N$ and $\det(g) = 1$. It corresponds to the compact form of the Cartan series A_{N+1} .

Fundamental quarks. Consider SU(N) plus N_F fundamental quarks. The index is $T(R) = N_F$ and therefore ν is defined only when N_F is even. The most general matrix with $R(g^2) = 1$ is $g = -\mathbf{1}_{2p} \oplus \mathbf{1}_{N-2p}$ for some integer p. Evaluating the character,

$$\nu = 2N_F \operatorname{tr}(g) = 2N_F (N - 4p) \tag{3.8.2}$$

Note that N_F is even and therefore $\nu \equiv 2N_F N \mod 16$, the *p*-dependence drops out.

The low energy effective description is postulated to be a non-linear sigma model with a Wess-Zumino term [29]; the 't Hooft anomaly of such a system is not known, and therefore we cannot check for matching.

Adjoint quarks. Consider SU(N) plus one adjoint quark. The index is T(R) = N and therefore ν is defined only when N is even. The most general matrix with $R(g^2) = 1$ is $g = (-\mathbf{1}_p \oplus \mathbf{1}_{N-p})\theta$ where p is an integer and θ a phase subject to $\theta^N = (-1)^a$ (such that $\det(g) = 1$). Evaluating the character (cf. (B.2.3)),

$$\nu = \operatorname{tr}(g)\operatorname{tr}(g^{-1}) - 1 \equiv (N - 2p)^2 - 1 \tag{3.8.3}$$

Reducing modulo 16, the anomaly is $\nu = -1$ and $\nu = 3$.

The low energy effective description is postulated to be the TQFT $U(N/2)_{N/2,N}$ plus a decoupled Majorana fermion [30]. The former has anomaly $\nu = \pm 2$ while the latter has anomaly $\nu = +1$. Therefore, the infrared phase has anomaly $\nu = -1$ and $\nu = 3$, precisely as in the ultraviolet.

Symmetric quarks. Consider SU(N) plus one symmetric quark. The index is T(R) = N+2and therefore ν is defined only when N is even. The most general matrix with $R(g^2) = 1$ is either $g = -\mathbf{1}_{2p} \oplus \mathbf{1}_{N-2p}$ where p is an integer, or $g = (-\mathbf{1}_p \oplus \mathbf{1}_{N-p})i$ with $p \equiv N/2 \mod 2$ (such that det(g) = 1). Evaluating the character (cf. (B.2.2)),

$$\nu = \operatorname{tr}(g^2) + \operatorname{tr}(g)^2 = \begin{cases} N + (N - 4p)^2 \\ -N - (N - 2p)^2 \end{cases}$$
(3.8.4)

for the two families of matrices above. Note that N is even and therefore the first option is in fact independent of p, and equal to $N(N+1) \mod 16$.

Simplifying the expression above, we get the following anomalies:

$$\nu = \begin{cases} 0 & N \equiv 0 \mod 16 & 8 & N \equiv 8 \mod 16 \\ -2, 6 & N \equiv 2 \mod 16 & -2, 6 & N \equiv 10 \mod 16 \\ \pm 4 & N \equiv 4 \mod 16 & \pm 4 & N \equiv 12 \mod 16 \\ -6 & N \equiv 6 \mod 16 & 2 & N \equiv 14 \mod 16 \end{cases}$$
(3.8.5)

The low energy effective description is postulated to be the TQFT $PSU((N+2)/2)_{(N+2)/2}$ plus a decoupled compact scalar [3]; the 't Hooft anomaly of such a system is not known, and therefore we cannot check for matching. Anti-symmetric quarks. Consider SU(N) plus one anti-symmetric quark. The index is T(R) = N - 2 and therefore ν is defined only when N is even. The most general matrix with $R(g^2) = 1$ is either $g = -\mathbf{1}_{2p} \oplus \mathbf{1}_{N-2p}$ where p is an integer, or $g = (-\mathbf{1}_p \oplus \mathbf{1}_{N-p})i$ with $p \equiv N/2 \mod 2$ (such that $\det(g) = 1$). Evaluating the character (cf. (B.2.2)),

$$\nu = -\operatorname{tr}(g^2) + \operatorname{tr}(g)^2 = \begin{cases} -N + (N - 4p)^2 \\ N - (N - 2p)^2 \end{cases}$$
(3.8.6)

for the two families of matrices above. Note that N is even and therefore the first option is in fact independent of p, and equal to $N(N-1) \mod 16$.

Simplifying the expression above, we get the following anomalies:

$$\nu = \begin{cases} 0 & N \equiv 0 \mod 16 & 8 & N \equiv 8 \mod 16 \\ 2 & N \equiv 2 \mod 16 & -6 & N \equiv 10 \mod 16 \\ \pm 4 & N \equiv 4 \mod 16 & \pm 4 & N \equiv 12 \mod 16 \\ -2, 6 & N \equiv 6 \mod 16 & -2, 6 & N \equiv 14 \mod 16 \end{cases}$$
(3.8.7)

The low energy effective description is postulated to be the TQFT $PSU((N-2)/2)_{(N-2)/2}$ plus a decoupled compact scalar [3]; the 't Hooft anomaly of such a system is not known, and therefore we cannot check for matching.

3.8.2 Orthogonal group.

The special orthogonal group SO(N) is defined as the set of $N \times N$ real matrices such that $gg^t = \mathbf{1}_N$ and $\det(g) = 1$. It corresponds to the compact form of the Cartan series $B_{(N-1)/2}, D_{N/2}$ after quotienting out a central \mathbb{Z}_2 .

Fundamental quarks. Consider SO(N) plus N_F fundamental quarks. The index is $T(R) = N_F$ and therefore ν is defined only when N_F is even. The most general matrix with $R(g^2) = 1$ is $g = -\mathbf{1}_{2p} \oplus \mathbf{1}_{N-2p}$ for some integer p. Evaluating the character,

$$\nu = N_F \operatorname{tr}(g) = N_F (N - 4p) \tag{3.8.8}$$

When either $N_F/2$ or p is even, then $\nu = NN_F \mod 16$; on the other hand, if they are both odd, then $\nu = NN_F + 8 \mod 16$. Note that ν is only defined when $\mathsf{T}^2 = (-1)^F$, i.e., when the time-reversal algebra is not modified by the magnetic symmetry [34]; this happens if and only if $N_F/2$ is even, whence $\nu \equiv NN_F$, the *p*-dependence drops out.

The low energy effective description is postulated to be a non-linear sigma model with a Wess-Zumino term [29]; the 't Hooft anomaly of such a system is not known, and therefore we cannot check for matching.

Adjoint quarks. Consider SO(N) plus one adjoint quark. The index is T(R) = N - 2 and therefore ν is defined only when N is even. The most general matrix with $R(g^2) = 1$ is either $g = -\mathbf{1}_{2p} \oplus \mathbf{1}_{N-2p}$ for some integer p, or $g = \text{diag}(i\sigma_y, \cdots, i\sigma_y)$. Evaluating the character (cf. (B.2.7)),

$$\nu = \frac{1}{2}(-\operatorname{tr}(g^2) + \operatorname{tr}(g)^2) \equiv \begin{cases} \frac{1}{2}(-N + (N - 4p)^2) \\ \frac{1}{2}N \end{cases}$$
(3.8.9)

for the two families of matrices above.

Simplifying the expressions above, we get the following anomalies:

$$\nu = \begin{cases} 0,8 & N \equiv 0 \mod 32 & 0,8 & N \equiv 16 \mod 32 \\ 1 & N \equiv 2 \mod 32 & -7 & N \equiv 18 \mod 32 \\ \pm 2,6 & N \equiv 4 \mod 32 & -2, \pm 6 & N \equiv 20 \mod 32 \\ -1,3 & N \equiv 6 \mod 32 & -5,7 & N \equiv 22 \mod 32 \\ \pm 4 & N \equiv 8 \mod 32 & \pm 4 & N \equiv 24 \mod 32 \\ -3,5 & N \equiv 10 \mod 32 & -3,5 & N \equiv 26 \mod 32 \\ 2, \pm 6 & N \equiv 12 \mod 32 & \pm 2, -6 & N \equiv 28 \mod 32 \\ -5,7 & N \equiv 14 \mod 32 & -1,3 & N \equiv 30 \mod 32 \end{cases}$$
(3.8.10)

The low energy effective description is postulated to be the TQFT $SO((N-2)/2)_{(N-2)/2}$ plus a decoupled Majorana fermion [30]. The former has anomaly $\nu = \pm (N-2)/2$ while the latter has anomaly $\nu = +1$. One can see that this matches the computation in the ultraviolet if and only if N/2 is odd. For N/2 even, the time-reversal algebra is modified by the monopole symmetry $\pi_1(SO(N)) = \mathbb{Z}_2$ [34], and hence the 't Hooft anomaly is no longer given by \mathbb{Z}_{16} , i.e., there is no reason to expect matching (as we are computing the anomaly for the wrong symmetry group).

A way around this is to consider, instead, the cover of SO(N), which has trivial fundamental group and hence no magnetic symmetry. Thus, we look at Spin(N) plus an adjoint quark. The computation of ν in the ultraviolet is the same, but now the infrared TQFT is O($(N - 2)/2)^{1}_{(N-2)/2,N/2+2}$ [113]. For N/2 even this can also be written as SO($(N-2)/2)_{(N-2)/2} \times (\mathbb{Z}_2)_N$; the second factor has anomaly $\nu = 0, 8$ for N = 0 and $\nu = 0, \pm 4$ for N = 4 (with mod 8 periodicity in N). Adding the contribution of $(\mathbb{Z}_2)_N$ to the anomaly of SO($(N-2)/2)_{(N-2)/2}$ nicely matches the ultraviolet computation for N/2 even.

Note that another way around the monopole symmetry is to conjugate by a flavor symmetry. Reference [34] shows that time-reversal, composed with a reflection in color space, has no monopole deformation when N/2 is even. Following the same analysis as before, but adding the contribution of the flavor symmetry, the ultraviolet anomaly can be shown to be $\nu = \frac{1}{2}(N^2 - 8Np - 5N + 4)$ for some integer p. Again, this nicely matches the infrared anomaly $\nu = \pm (N - 2)/2 + 1$ for N/2 even.

Symmetric quarks. Consider SO(N) plus one symmetric quark. The index is T(R) = N+2and therefore ν is defined only when N is even. The most general matrix with $R(g^2) = 1$ is either $g = -\mathbf{1}_{2p} \oplus \mathbf{1}_{N-2p}$ for some integer p, or $g = \text{diag}(i\sigma_y, \cdots, i\sigma_y)$. Evaluating the character (cf. (B.2.7)),

$$\nu = \frac{1}{2} (\operatorname{tr}(g^2) + \operatorname{tr}(g)^2) - 1 \equiv \begin{cases} \frac{1}{2} (N + (N - 4p)^2) - 1 \\ -\frac{1}{2}N - 1 \end{cases}$$
(3.8.11)

for the two families of matrices above.

Simplifying the expressions above, we get the following anomalies:

$$\nu = \begin{cases} -1,7 & N \equiv 0 \mod 32 & -1,7 & N \equiv 16 \mod 32 \\ \pm 2 & N \equiv 2 \mod 32 & \pm 6 & N \equiv 18 \mod 32 \\ 1,-3,-7 & N \equiv 4 \mod 32 & 1,5,-7 & N \equiv 20 \mod 32 \\ \pm 4 & N \equiv 6 \mod 32 & \pm 4 & N \equiv 22 \mod 32 \\ 3,-5 & N \equiv 8 \mod 32 & 3,-5 & N \equiv 24 \mod 32 \\ \pm 6 & N \equiv 10 \mod 32 & \pm 2 & N \equiv 26 \mod 32 \\ -3,5,-7 & N \equiv 12 \mod 32 & 1,-3,5 & N \equiv 28 \mod 32 \\ 8 & N \equiv 14 \mod 32 & 0 & N \equiv 30 \mod 32 \end{cases}$$
(3.8.12)

The low energy effective description is postulated to be the TQFT SO $((N+2)/2)_{(N+2)/2}$ [30]. This has anomaly $\nu = \pm (N+2)/2$. One can see that this matches the computation in the ultraviolet if and only if N/2 is odd. For N/2 even, the time-reversal algebra is modified by the monopole symmetry $\pi_1(SO(N)) = \mathbb{Z}_2$ [34], and hence the 't Hooft anomaly is no longer given by \mathbb{Z}_{16} , i.e., there is no reason to expect matching (as we are computing the anomaly for the wrong symmetry group).

A way around this is to consider, instead, the cover of SO(N), which has trivial fundamental group and hence no magnetic symmetry. Thus, we look at Spin(N) plus a symmetric quark. The computation of ν in the ultraviolet is the same, but now the infrared TQFT is O((N + 2)/2)¹_{(N+2)/2,N/2} [113]. For N/2 even this can also be written as SO((N+2)/2)_{(N+2)/2} × (\mathbb{Z}_2)_N; the second factor has anomaly $\nu = 0, 8$ for N = 0 and $\nu = 0, \pm 4$ for N = 4 (with mod 8 periodicity in N). Adding the contribution of (\mathbb{Z}_2)_N to the anomaly of SO((N+2)/2)_{(N+2)/2} nicely matches the ultraviolet computation for N/2 even.

Note that another way around the monopole symmetry is to conjugate by a flavor symmetry. Reference [34] shows that time-reversal, composed with a reflection in color space, has no monopole deformation when N/2 is even. Following the same analysis as before, but adding the contribution of the flavor symmetry, the ultraviolet anomaly can be shown to be $\nu = \frac{1}{2}(N^2 - 8Np - 3N + 2)$ for some integer p. Again, this nicely matches the infrared anomaly $\nu = \pm (N + 2)/2$ for N/2 even.

3.8.3 Symplectic group.

The symplectic group $\operatorname{Sp}(N)$ is defined as the set of $\operatorname{SU}(2N)$ matrices such that $g\Omega g^t = \Omega$, where Ω is the canonical anti-symmetric matrix $\Omega = \operatorname{diag}(i\sigma_y, \cdots, i\sigma_y)$. It corresponds to the compact form of the Cartan series C_N . The alternative notation $\operatorname{USp}(2N)$ is also sometimes used for this group.

Fundamental quarks. Consider Sp(N) plus N_F fundamental quarks. The index is $T(R) = N_F$ and therefore ν is defined only when N_F is even. The most general matrix with $R(g^2) = 1$ is $g = -\mathbf{1}_{2p} \oplus \mathbf{1}_{2N-2p}$ for some integer p. Evaluating the character,

$$\nu = 2N_F \operatorname{tr}(g) = 4N_F (N - 2p) \tag{3.8.13}$$

Note that N_F is even and therefore $\nu \equiv 4N_F N \mod 16$, the *p*-dependence drops out.

The low energy effective description is postulated to be a non-linear sigma model with a Wess-Zumino term [29]; the 't Hooft anomaly of such a system is not known, and therefore we cannot check for matching.

Adjoint quarks. Consider Sp(N) plus one adjoint quark. The index is T(R) = N + 1and therefore ν is defined only when N is odd. The most general matrix with $R(g^2) = 1$ is either $g = -\mathbf{1}_{2p} \oplus \mathbf{1}_{2N-2p}$ for some integer p, or $g = (-\mathbf{1}_N \oplus \mathbf{1}_N)i$. Evaluating the character (cf. (B.2.8)),

$$\nu = \frac{1}{2} (\operatorname{tr}(g^2) + \operatorname{tr}(g)^2) \equiv \begin{cases} \frac{1}{2} (2N + (2N - 4p)^2) \\ -N \end{cases}$$
(3.8.14)

for the two families of matrices above.

Simplifying the expressions above, we get the following anomalies:

$$\nu = \begin{cases} -1,3 \quad N \equiv 1 \mod 16 & -5,7 \quad N \equiv 9 \mod 16 \\ -3,5 \quad N \equiv 3 \mod 16 & -3,5 \quad N \equiv 11 \mod 16 \\ -5,7 \quad N \equiv 5 \mod 16 & -1,3 \quad N \equiv 13 \mod 16 \\ -7 \quad N \equiv 7 \mod 16 & 1 \quad N \equiv 15 \mod 16 \end{cases}$$
(3.8.15)

The low energy effective description is postulated to be the TQFT $\text{Sp}((N+1)/2)_{(N+1)/2}$ plus a decoupled Majorana fermion [30]. The former has anomaly $\nu = \pm (N+1)$ while the latter has anomaly $\nu = +1$. Adding up these contributions, the infrared phase has anomaly $\nu = \pm (N+1) + 1$, which nicely matches the ultraviolet computation above.

Anti-symmetric quarks. Consider Sp(N) plus one anti-symmetric quark. The index is T(R) = N - 1 and therefore ν is defined only when N is odd. The most general matrix with

 $R(g^2) = 1$ is either $g = -\mathbf{1}_{2p} \oplus \mathbf{1}_{2N-2p}$ for some integer p, or $g = (-\mathbf{1}_N \oplus \mathbf{1}_N)i$. Evaluating the character (cf. (B.2.8)),

$$\nu = \frac{1}{2}(-\operatorname{tr}(g^2) + \operatorname{tr}(g)^2) - 1 \equiv \begin{cases} \frac{1}{2}(-2N + (2N - 4p)^2) - 1\\ N - 1 \end{cases}$$
(3.8.16)

for the two families of matrices above.

Simplifying the expressions above, we get the following anomalies:

$$\nu = \begin{cases} 0 & N \equiv 1 \mod 16 & 8 & N \equiv 9 \mod 16 \\ \pm 2 & N \equiv 3 \mod 16 & \pm 6 & N \equiv 11 \mod 16 \\ \pm 4 & N \equiv 5 \mod 16 & \pm 4 & N \equiv 13 \mod 16 \\ \pm 6 & N \equiv 7 \mod 16 & \pm 2 & N \equiv 15 \mod 16 \end{cases}$$
(3.8.17)

The low energy effective description is postulated to be the TQFT $\text{Sp}((N-1)/2)_{(N-1)/2}$ [30]. This theory has anomaly $\nu = \pm (N-1)$, which nicely matches the ultraviolet computation above.

Chapter 4

Symmetries of Abelian Chern-Simons Theories and Arithmetic.

Authorship. The content of this chapter is reproduced almost verbatim from the paper [4] written in collaboration with Jaume Gomis.

Abstract. We determine the unitary and anti-unitary symmetries of arbitrary abelian Chern-Simons theories. The symmetries depend sensitively on the arithmetic properties (e.g. prime factorization) of the matrix of Chern-Simons levels, revealing interesting connections with number theory. We give a complete characterization of the symmetries of abelian topological field theories and along the way find many theories that are non-trivially time-reversal invariant, including U(1)_k Chern-Simons theory and $(\mathbb{Z}_k)_{\ell}$ gauge theories. For example, we prove that U(1)_k Chern-Simons theory is time-reversal invariant if and only if -1 is a quadratic residue modulo k, which happens if and only if all the prime factors of k are Pythagorean (i.e., of the form 4n + 1), or Pythagorean with a single additional factor of 2. Many distinct non-abelian finite symmetry groups are found.

Acknowledgments. We would like to thank Clay Córdova, Dan Freed, Davide Gaiotto, Po-Shen Hsin, Theo Johnson-Freyd, Anton Kapustin, Zohar Komargodski, Nathan Seiberg, Ryan Thorngren, Senthil Todadri, Chong Wang and Jon Yard for useful discussions.

4.1	Introduction and Summary
4.2	TQFTs and Symmetry
4.3	$U(1)_k$ Chern-Simons
4.4	$U(1)^n$ Chern-Simons theory $\ldots \ldots \ldots$
4.5	Examples
4.6	Notation and definitions
4.7	Further results

4.1 Introduction and Summary

Symmetries play a pivotal role in our description of nature. In classical physics symmetries generate solutions of the equations of motion and in quantum mechanics symmetries imply selection rules and constrain physical observables. 't Hooft anomalies for global symmetries, being renormalization-group invariant, provide powerful nonperturbative constraints on the dynamics. By a classic result of Wigner, symmetries in quantum mechanics are implemented in the Hilbert space either by unitary or anti-unitary operators, and the corresponding transformations are linear and anti-linear, respectively.

Invariance of the classical action under a transformation g imposes nontrivial constraints on the correlation functions of the theory. These are encapsulated in Ward identities. Invariance of the action under a transformation g is a sufficient condition for g to be a symmetry. However, this is not necessary. A transformation g that does not leave the action S invariant

$$g \cdot S \neq S \tag{4.1.1}$$

is nevertheless a symmetry of the quantum theory if it obeys the Ward identities

$$\langle g \cdot \mathcal{O}_1 \cdots g \cdot \mathcal{O}_m \rangle = \begin{cases} \langle \mathcal{O}_1 \cdots \mathcal{O}_m \rangle & g \text{ unitary} \\ \langle \mathcal{O}_1 \cdots \mathcal{O}_m \rangle^* & g \text{ anti-unitary} \end{cases},$$
(4.1.2)

where * implements complex conjugation. We shall refer to such non-Lagrangian symmetries as *quantum symmetries*. Naturally, determining whether a theory has a quantum symmetry is nontrivial. In this work we characterize all the symmetries, quantum or otherwise, of abelian Chern-Simons theories.

Chern-Simons theories are ubiquitous in physics and mathematics. They arise as the emergent infrared description of gapped, quantum phases of matter such as the integer and fractional quantum Hall effect, quantum spin liquids and analogs of topological insulators and superconductors (see e.g [240, 241]). Chern-Simons theories capture the nonperturbative infrared dynamics of 2 + 1 dimensional gauge theories with massless fermions [3, 30, 113, 127–129, 237], and describe the low-energy dynamics of domain walls connecting vacua of 3 + 1 dimensional gauge theories [3, 40, 41, 46]. Chern-Simons theory, a topological quantum field theory (TQFT), has also found beautiful and profound applications in mathematics, starting with Witten's work [22] on the topological invariants of knots and three-manifolds.

In this paper we give a complete description of all the unitary and anti-unitary symmetries of abelian Chern-Simons theories, the simplest incarnation being $U(1)_k$ Chern-Simons theory, described by the Lagrangian

$$\mathcal{L} = \frac{k}{4\pi} a \,\mathrm{d}a\,,\tag{4.1.3}$$

where a is a U(1) gauge field and the coupling constant is quantized, $k \in \mathbb{Z}$. More generally, an arbitrary abelian TQFT can be described by a collection of such fields coupled via an integral symmetric matrix K with Lagrangian

$$\mathcal{L} = \frac{1}{4\pi} a^t K \mathrm{d}a \,, \tag{4.1.4}$$

where $a^t = (a_1, \ldots, a_n)$. These theories have been studied intensely and enjoy a myriad of applications. In spite of this, we unearth a rich structure of symmetries in these theories, which depends on the arithmetic properties of the Chern-Simons levels K, revealing interesting connections with number theory.

Symmetries in topological phases of matter have been at the forefront of recent developments at the intersection of condensed matter, particle physics, and mathematics. These gapped phases are encoded by emergent TQFTs. Gapped phases with no topological order (no nontrivial anyons) and protected by symmetries describe SPT phases (see e.g. [52, 242–245]) while phases with topological order (with nontrivial anyons) and enriched by symmetries give rise to the so-called SET phases (see e.g. [246–250]). Symmetries and 't Hooft anomalies of TQFTs have recently played a key role in understanding the nonperturbative infrared dynamics of gauge theories [3, 30, 113, 127–129, 237]. Despite a lot of work, little is concretely known about the symmetries of TQFTs. Here we tackle this problem for abelian TQFTs.

For the reader's convenience we summarize here a sample of our main results:

• $U(1)_k$ is a time-reversal invariant spin TQFT,⁸⁸ that is, it admits an anti-unitary symmetry, if and only if -1 is a quadratic residue modulo k (cf. proposition 4.3.2). Equivalently:

$$U(1)_{+k} \longleftrightarrow U(1)_{-k} \iff q^2 = -1 \mod k \text{ for some } q \in \mathbb{Z}.$$
 (4.1.5)

Therefore, $U(1)_k$ Chern-Simons theory is time-reversal invariant if and only if

$$k \in \mathbb{T} := \{k \in \mathbb{Z} \mid kp - q^2 = 1 \quad \text{for some } p, q \in \mathbb{Z}\}.$$

$$(4.1.6)$$

This result can also be stated as $U(1)_k$ being dual to $U(1)_{-k}$ when $k \in \mathbb{T}$, which we denote by $U(1)_{+k} \longleftrightarrow U(1)_{-k}$. The integer k is in \mathbb{T} if and only if all its prime factors are Pythagorean (i.e., congruent to 1 modulo 4), or Pythagorean with a single factor of 2. Any time-reversal symmetry is of order 4, except for k = 1, 2, when it is of order 2 (cf. proposition 4.3.3).

⁸⁸If k is odd, $U(1)_k$ is a spin TQFT. For k even it is bosonic but can be turned into a spin TQFT by tensoring with a transparent fermion $\{1, \psi\}$. See section 4.2 for details.

The set of time-reversal invariant $U(1)_k$ Chern-Simons theories includes the subset $k \in \mathbb{P} := \{k \in \mathbb{Z} \mid kp^2 - q^2 = 1 \text{ for some } p, q \in \mathbb{Z}\} \subset \mathbb{T}$. The set \mathbb{P} corresponds to those values of the level for which the (negative) Pell equation is solvable, which was shown by Witten [251, 252] to lead to time-reversal invariance.

We prove that the time-reversal symmetry is a quantum symmetry if and only if $k \in \mathbb{T} \setminus \mathbb{P}$ (cf. proposition 4.3.6). By studying the time-reversal invariance of $U(1)_k \times U(1)_{k'}$ we obtain an interesting number-theoretic conjecture, to wit, $k \in \mathbb{T}$ if and only if there exist some $k' \in \mathbb{P}$ such that $kk' \in \mathbb{P}$. We argue that this conjecture follows from a well-known conjecture by Hardy-Littlewood (cf. conjecture 4.7.1).

• All the unitary symmetries of $U(1)_k$ are of order 2, and the number of such symmetries depends on the number of distinct prime factors of k, usually denoted by $\omega(k)$. More precisely, the group of unitary symmetries of $U(1)_k$ is (cf. proposition 4.3.10)

$$(\mathbb{Z}_2)^{\varpi(k)}, \qquad \varpi(k) := \begin{cases} \omega(k) & k \text{ odd} \\ \omega(k/2) & k \text{ even.} \end{cases}$$
(4.1.7)

When $U(1)_k$ with k even is upgraded to an spin TQFT by considering $U(1)_k \times \{\mathbf{1}, \psi\}$, an additional factor of \mathbb{Z}_2 appears when k is a multiple of 8. All but one factor of \mathbb{Z}_2 in (4.1.7), which corresponds to charge conjugation, are quantum symmetries. When $k \in \mathbb{T}$, the total group of symmetries is the central product of its unitary subgroup and \mathbb{Z}_4 .

 The unitary and anti-unitary symmetries of U(1)ⁿ Chern-Simons theory with matrix of levels K correspond to the integral-valued matrices Q, invertible modulo K, that solve (cf. proposition 4.4.4)

unitary:
$$Q^{t}K^{-1}Q - K^{-1} = P$$

anti-unitary: $Q^{t}K^{-1}Q + K^{-1} = P$ (4.1.8)

for some integral-valued matrix P. While the first equation always admits solutions, the second one need not, and only when there is a solution is the theory time-reversal invariant. The group of symmetries is finite and generically non-abelian. A given symmetry is quantum if and only if $P \neq 0$ for all the Q's that implement it.

• Two abelian Chern-Simons theories described by matrices K_1, K_2 (not necessarily of the same dimension) are dual if and only if there exist suitable matrices Q, P such that

$$Q^t K_1^{-1} Q - K_2^{-1} = P (4.1.9)$$

(see section 4.4.2 for the precise formulation and the conditions on Q, P). In this sense, the unitary symmetries of K correspond to the self-dualities $K \leftrightarrow K$, and the anti-unitary symmetries to dualities $K \leftrightarrow -K$.

• The twisted gauge theory $(\mathbb{Z}_{k_1})_{k_2}$ (also known as \mathbb{Z}_{k_1} Dijkgraaf-Witten theory [35] when k_2 is even, and which can be realized by the U(1)² Chern-Simons theory with $K = \begin{pmatrix} 0 & k_1 \\ k_1 & k_2 \end{pmatrix}$ with $k_2 \in [0, 2k_1)$) is conjectured to be time-reversal invariant if and only if k_2 is proportional to $\mu(k_1)$ (cf. conjecture 4.4.2)

$$k_2 \propto \mu(k_1) \tag{4.1.10}$$

where $\mu(n)$ equals *n* divided by all its Pythagorean prime factors (e.g. $\mu(10) = \frac{2 \times 5}{5} = 2$). The conjecture has been verified for $k_1 \in [0, 200]$ and all k_2 . We compute the explicit group of unitary and anti-unitary symmetries of $(\mathbb{Z}_{k_1})_{k_2}$ for small values of the levels; see table 4.1 for a sample. The time-reversal symmetry of $(\mathbb{Z}_{k_1})_{k_2}$ implies in particular a duality between abelian TQFTs

$$(\mathbb{Z}_{k_1})_{+k_2} \longleftrightarrow (\mathbb{Z}_{k_1})_{-k_2} \iff k_2 \propto \mu(k_1).$$
 (4.1.11)

The theory $(\mathbb{Z}_k)_0$ has conjecturally $2^{\omega(k)}\phi(k)$ unitary transformations and as many anti-unitary ones (where $\phi(k)$ is the Euler totient function, which counts the number of integers $q \in [1, k)$ relatively prime to k). Among these symmetries, there is a unitary \mathbb{Z}_2 subgroup which is Lagrangian, and four anti-unitary Lagrangian symmetries (except for k = 2, which only has two). For k > 2 the group of symmetries is non-abelian (see 4.4.5 for the explicit conjecture), while for k = 2, the group of symmetries is \mathbb{Z}_2^2 , with a \mathbb{Z}_2 unitary subgroup.

• The so-called "minimal abelian TQFT" $\mathcal{A}^{N,t}$ is proven to be time-reversal invariant invariant if and only if t is proportional to $\mu(N)$ (cf. subsection 4.3.2)

$$t \propto \mu(N) \,. \tag{4.1.12}$$

These minimal theories have N anyons with a \mathbb{Z}_N fusion algebra, and their spin depends on the integer t.

TQFTs can also have a one-form symmetry group [32, 205] on top of the usual (zero-form) symmetry group that we study in this paper. The Wilson lines describing the worldline of anyons transform in representations of this group. The one-form symmetries of abelian Chern-Simons theories are well understood (see e.g. [253]). Given an abelian TQFT with an abelian Chern-Simons representation, the one-form symmetry group is $\mathbb{Z}_{k_1} \times \mathbb{Z}_{k_2} \dots \times \mathbb{Z}_{k_n}$, where $\{k_i\}$ are the Smith invariants of K (cf. section 4.4). Interestingly, given a QFT with a zero-form symmetry group and a one-form symmetry group, these can combine into a nontrivial extension known as a 2-group (see e.g. [252, 254]). When a theory has a 2-group symmetry, the zero-form and one-form symmetries do not factorize; rather, they are mixed non-trivially. However, it is known that abelian TQFTs have a trivial 2-group of

k	$\operatorname{Aut}((\mathbb{Z}_k)_0)$	$\operatorname{Aut}_U((\mathbb{Z}_k)_0)$	$\operatorname{Aut}((\mathbb{Z}_k)_{\mu(k)})$	$\operatorname{Aut}_U((\mathbb{Z}_k)_{\mu(k)})$
2	\mathbb{Z}_2^2	\mathbb{Z}_2	\mathbb{Z}_2	0
3	D_8	\mathbb{Z}_2^2	D_8	\mathbb{Z}_2^2
4	D_8	\mathbb{Z}_2^2	D_8	\mathbb{Z}_2^2
5	$\mathbb{Z}_4 \circ D_8$	D_8	\mathbb{Z}_4	\mathbb{Z}_2
6	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_2^3	D_8	\mathbb{Z}_2^2
7	$\mathbb{Z}_3 \rtimes D_8$	D_{12}	$\mathbb{Z}_3 \rtimes D_8$	D_{12}
8	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_2^3	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_2^3
9	$\mathbb{Z}_3 \rtimes D_8$	D_{12}	$\mathbb{Z}_3 \rtimes D_8$	D_{12}
10	$\mathbb{Z}_2 \times \mathbb{Z}_4 \circ D_8$	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_4	\mathbb{Z}_2
11	$\mathbb{Z}_5 \rtimes D_8$	D_{20}	$\mathbb{Z}_5 \rtimes D_8$	D_{20}
12	$\mathbb{Z}_2^2\wr\mathbb{Z}_2$	\mathbb{Z}_2^4	$\mathbb{Z}_2^2\wr\mathbb{Z}_2$	\mathbb{Z}_2^4

Table 4.1: The group of symmetries of $(\mathbb{Z}_{k_1})_{k_2}$, denoted by Aut(\star), and its unitary subgroup Aut_U(\star), for $k_1 \in [0, 12]$ and $k_2 = 0, \mu(k_1)$. For $k_2 \not\propto \mu(k_1)$ there are no anti-unitary symmetries. See table 4.2 for the group of symmetries up to $k_1 = 27$. (See section 4.6 for basic definitions).

symmetries [252, 255, 256]: the zero-form and one-form symmetries factorize, and since the one-form symmetries are completely understood, what remains are the zero-form symmetries, which is the problem we address in this paper. Furthermore, since the 2-group in an abelian TQFT is trivial, the zero-form and one-form 't Hooft anomalies are well defined and can be classified using cohomology and cobordism groups [55, 58, 59, 69, 71, 257], and "anomaly indicators" detecting the 't Hooft anomalies (see e.g. [63, 78]) can be investigated. These anomaly indicators – which are the partition function evaluated on the generators of the corresponding cobordism groups, and expressed in terms of the modular data of the TQFT (see below) – are only known for a handful of symmetry groups.

The plan for the remainder of the chapter is as follows. In section 4.2 we describe the general paradigm of symmetries in topological quantum field theories, and the simplifications that occur for abelian TQFTs. In section 4.3 we completely describe all the symmetries for the most characteristic abelian system: $U(1)_k$ Chern-Simons theory. In section 4.4 we generalize the analysis to arbitrary abelian TQFTs, by realizing them as $U(1)^n$ Chern-Simons theories. We prove several results, and make a number of conjectures. In section 4.5 we work out a couple dozen examples in some detail, so as to illustrate the general formalism. Finally, we summarize definitions and notations in section 4.6 and leave some proofs and further results to section 4.7.

4.2 TQFTs and Symmetry

Before delving into the study of the symmetries of abelian Chern-Simons theories we describe how symmetries are realized in a TQFT in 2 + 1 dimensions. We informally review the data defining a TQFT and how, in an abelian TQFT, it is completely fixed in terms of most elementary data, to wit, the anyon fusion algebra and the anyon spins. We then proceed with the physical and mathematical characterization of a symmetry in a TQFT. More details and mathematical elaborations can be found in the literature [38, 249, 258–261].

A TQFT can be understood as a finite collection of anyons – particles with fractional statistics – belonging to an anyon set \mathcal{A} endowed with the following additional data:

• Fusion: A commutative, associative product $\times : \mathcal{A} \times \mathcal{A} \to \mathcal{A}$ describing the fusion of anyons $a, b \in \mathcal{A}$ (see figure 4.1):

$$a \times b = \sum_{c \in \mathcal{A}} N_{ab}{}^c c , \qquad (4.2.1)$$

where $N_{ab}{}^c \in \mathbb{Z}_{\geq 0}$ are the so-called *fusion coefficients*. We denote the trivial anyon by **1**.



Figure 4.1: Fusion of anyons: two unbraided lines with labels a, b can be replaced by one with label $\sum_{c} N_{ab}{}^{c} c$.

- Topological spin: A map $\theta: \mathcal{A} \to U(1)$. The topological spin determines the anyonic character of an anyon. One usually writes $\theta(a) =: \exp(2\pi i h_a)$, where $h_a: \mathcal{A} \to \mathbb{Q}/\mathbb{Z}$ is the spin of a. The topological spin controls the framing anomaly of a knot (the dependence of observables on the choice of the homotopy class of a normal vector field, see figure 4.2).
- S- and T-matrices: A representation of the modular group. The S-matrix determines the braiding phase $B: \mathcal{A} \times \mathcal{A} \to U(1)$ between anyons (see figure 4.3)

$$B(a,b) = \frac{S_{ab}}{S_{1b}},$$
(4.2.2)



Figure 4.2: Topological spin: anyons are to be thought of as ribbons rather than knots. Observables depend on the twisting thereof, through their spin.

while $T_{ab} = \theta_a e^{-2\pi i c/24} \delta_{ab}$, where c is the chiral central charge of the TQFT, which controls the framing anomaly (the dependence of observables on the 2-framing of the manifold).

• **F- and R-symbols:** The associator and braiding isomorphism, encoding the fusion of multiple anyons and their half-braiding. This data is defined modulo local, redundant isomorphisms (gauge transformations) U defined on fusion vector spaces. The gauge-transformed data, which we denote by UF and UR, is physically equivalent to F and R, and define the same TQFT.



Figure 4.3: Braiding of anyons: if at least one of a, b is abelian, then the two lines may be unbraided, a process that generates a phase $B(a, b) \in U(1)$.

This data is subject to nontrivial consistency conditions, known as the Moore-Seiberg relations, which include the hexagon and pentagon relations involving the F- and R-symbols. These relations imply that some of the data above is actually redundant; for example, the topological spin θ is a gauge invariant combination of the F- and R-symbols. The TQFT data defines a modular tensor category. This data can be used to compute an arbitrary correlation function of the TQFT (cf. (4.1.2)).

An anyon a is said to be *abelian* if the fusion of a with an arbitrary anyon b contains a single anyon c = c(a, b), i.e.

$$a \times b = c \qquad \forall b \in \mathcal{A}.$$
 (4.2.3)

In terms of the fusion coefficients (4.2.1), a is abelian if for any b the sum $\sum_{c \in \mathcal{A}} N_{ab}{}^c$ equals 1. An abelian anyon $a \in \mathcal{A}$ has a unique inverse $\bar{a} \in \mathcal{A}$ such that $a \times \bar{a} = 1$, and therefore abelian anyons form a finite abelian group, the one-form symmetry group of the TQFT [32].

An abelian TQFT is a TQFT in which all anyons in \mathcal{A} are abelian. Therefore, in an abelian TQFT the anyon fusion algebra defines a finite abelian group, which we also denote by \mathcal{A} . Remarkably, an abelian TQFT is completely determined by the group \mathcal{A} encoding the fusion of anyons, and by the topological spin $\theta: \mathcal{A} \to U(1)$ of the anyons, which is a quadratic, homogeneous function on \mathcal{A} [256, 262–264].⁸⁹ The entire TQFT data can be reconstructed from \mathcal{A} and such a θ .⁹⁰ The braiding phase of the abelian TQFT with fusion \mathcal{A} and spin θ takes the form

$$B(a,b) = \frac{\theta(a \times b)}{\theta(a)\theta(b)}, \qquad a,b \in \mathcal{A}, \qquad (4.2.4)$$

while the corresponding S-matrix is

$$S(a,b) = \frac{B(a,b)}{\sqrt{|\mathcal{A}|}}.$$
(4.2.5)

Importantly, given (\mathcal{A}, θ) there is a unique equivalence class of F and R symbols, and therefore a unique TQFT with that (\mathcal{A}, θ) . Summarizing, in an abelian TQFT the entire theory is completely fixed in terms of (\mathcal{A}, θ) . This statement is *not* true in a generic non-abelian TQFT, which is what makes the abelian case more tractable.

The discussion above applies as stated for a bosonic TQFT, a theory that does not require specifying a spin structure on the three-manifold where it is defined. Many interesting TQFTs, including abelian Chern-Simons theories, do require a choice of a spin structure to be defined. Such TQFTs are known as *spin* TQFTs. In a spin TQFT there is a distinguished abelian anyon ψ with topological spin $\theta(\psi) = -1$ and trivial braiding with all other anyons. This implies that ψ squares to the trivial anyon, i.e. $\psi \times \psi = \mathbf{1}$, and that $\theta(a \times \psi) = -\theta(a)$ for all $a \in \mathcal{A}$. In other words, a spin TQFT has a local (spin 1/2) fermion, which endows the data above with a \mathbb{Z}_2 -grading (i.e., anyons come in pairs $\{a, a \times \psi\}$).

⁸⁹ θ is a quadratic function if the symmetric form in (4.2.4) is bilinear, i.e. $B(a \times b, c) = B(a, c)B(b, c)$. Homogeneity means that $\theta(a^n) = \theta(a)^{n^2}$ for any $n \in \mathbb{Z}$, which implies that $\theta(\mathbf{1}) = 1$.

⁹⁰The central charge is determined by (\mathcal{A}, θ) only modulo 8. This indeterminacy can be understood as coming from the fact that one may always tensor by an even unimodular lattice, which has no lines, but may add central charge; the minimal such lattice is E_8 , which has signature 8. Some more refined observables (see e.g. [262, 264]) are sensitive to the actual value of c, and not only to it modulo 8. If we are interested in such observables, the TQFT data should be taken as (\mathcal{A}, θ, c) rather than just (\mathcal{A}, θ) . This will not play a major role in this work.

Any abelian TQFT, bosonic or spin, admits a representation as an abelian Chern-Simons theory [33, 256, 262, 264, 265], and is completely determined by (\mathcal{A}, θ) . Therefore, in spite that a complete and universally accepted axiomatization of a spin TQFT from a categorical point of view is lacking, the abelian Chern-Simons realization of the TQFT and its datum (\mathcal{A}, θ) suffice to determine the symmetries of spin abelian TQFTs (we also provide path integral arguments to exhibit the symmetries of abelian Chern-Simons theories that do not rely on the precise categorical characterization of spin TQFTs).

The symmetries of a TQFT are, by definition, the automorphisms of its data [249]. An automorphism g of a TQFT is a permutation of the anyons $g: \mathcal{A} \to \mathcal{A}$

$$a \mapsto g(a) \tag{4.2.6}$$

that preserves the fusion algebra \mathcal{A}

$$g(a \times b) = g(a) \times g(b) \qquad \Longleftrightarrow \qquad N_{g(a)g(b)}{}^{g(c)} = N_{ab}{}^c. \tag{4.2.7}$$

If the symmetry of the TQFT is unitary it must preserve the data modulo gauge transformations

$$\theta(g(a)) = \theta(a), \qquad S_{g(a)g(b)} = S_{ab}, \qquad g \cdot F = UF \qquad g \cdot R = UR, \qquad (4.2.8)$$

while if the symmetry is anti-unitary it preserves the data modulo gauge transformations, up to complex conjugation

$$\theta(g(a)) = \theta(a)^*$$
 $S_{g(a)g(b)} = S_{ab}^*$ $g \cdot F = UF^*, \quad g \cdot R = UR^*.$ (4.2.9)

Despite this explicit characterization, little is known about the actual symmetries of TQFTs. By contrast, the one-form symmetries of a TQFT are completely understood; they are determined by the abelian anyons and their fusion. Henceforth, when we discuss symmetries we refer to usual (zero-form) symmetries.

As reviewed above, in an abelian TQFT the entire data is completely determined by the abelian group \mathcal{A} encoding the fusion algebra and the topological spin θ . A necessary condition for the transformation g to a symmetry of an abelian TQFT is that $g: \mathcal{A} \to \mathcal{A}$ is an automorphism of the finite group \mathcal{A}

$$g(a \times b) = g(a) \times g(b). \qquad (4.2.10)$$

The set of automorphisms of \mathcal{A} , denoted by Aut(\mathcal{A}), is a finite, generically nonabelian group. An automorphism g of \mathcal{A} lifts to a unitary symmetry of the abelian TQFT if and only if

$$\theta(g(a)) = \theta(a), \qquad (4.2.11)$$

and to an anti-unitary symmetry if and only if

$$\theta(g(a)) = \theta(a)^*. \tag{4.2.12}$$

If such an automorphism g exists, it is guaranteed that the entire data of the abelian TQFT is preserved and g is a symmetry. In other words, the group of symmetries of an abelian TQFT is the subgroup of Aut(\mathcal{A}) that preserves the topological spins (up to complex conjugation for anti-unitary symmetries). We introduce the following notation for this group:

Definition 4.2.1 Given an abelian TQFT, we let $\operatorname{Aut}(\mathcal{A}, \theta) \subseteq \operatorname{Aut}(\mathcal{A})$ denote the group of all symmetries, and $\operatorname{Aut}_U(\mathcal{A}, \theta) \subseteq \operatorname{Aut}(\mathcal{A}, \theta)$ the subgroup of unitary symmetries.

The main goal of this work is to study the object $\operatorname{Aut}(\mathcal{A}, \theta)$. We determine it explicitly in the case of $U(1)_k$, and give a complete characterization thereof for arbitrary abelian theories. We will also work out a few illustrative examples in some detail.

4.3 $U(1)_k$ Chern-Simons

We begin by reviewing Chern-Simons theory with gauge group U(1). The generalization to the gauge group $U(1)^n$ is the content of section 4.4.

The Lagrangian of $U(1)_k$ Chern-Simons theory is

$$\mathcal{L} = \frac{k}{4\pi} a \,\mathrm{d}a\,,\tag{4.3.1}$$

where a is a U(1) gauge gauge field and the coupling $k \in \mathbb{Z}$ is quantized. Being topological, the theory can be defined on an arbitrary (oriented, framed) three-manifold, perhaps with a choice of spin structure depending on the parity of k. The equations of motion are

$$\mathrm{d}a = 0 \tag{4.3.2}$$

and the classical field configurations are flat connections.

The gauge invariant operators in this theory are the Wilson lines

$$W_{\alpha}(\gamma) := \exp\left[i\alpha \int_{\gamma} a\right], \qquad \alpha \in \mathbb{Z}.$$
 (4.3.3)

Physically, W_{α} describes the worldline of an anyon α with topological spin

$$\theta(\alpha) = e^{2\pi i h_{\alpha}}, \qquad h_{\alpha} = \frac{\alpha^2}{2k}.$$
(4.3.4)

The spin of an anyon h_{α} is only well-defined modulo an integer, because it cannot be distinguished from an anyon enriched with a soft *a*-photon, which has spin h = 1. If we introduce a background electromagnetic field, the anyon α is seen to carry a fractional charge given by α/k , as follows from the coupling $\frac{1}{2\pi}A \, da$.

The anyon fusion algebra is determined by the OPE of the corresponding Wilson lines: $\alpha \times \beta = \alpha + \beta$. The braiding phase acquired by an anyon α circumnavigating around an anyon β is

$$B(\alpha,\beta) = \frac{\theta(\alpha \times \beta)}{\theta(\alpha)\theta(\beta)} = e^{2\pi i \frac{\alpha\beta}{k}}.$$
(4.3.5)

It follows from (4.3.5) and (4.3.4) that the anyon $\alpha = k$ has trivial braiding with respect to all other anyons, and has spin $h = 0 \mod 1$ for k even and spin $h = 1/2 \mod 1$ for kodd. Therefore U(1)_k is a spin TQFT for odd k, and a bosonic TQFT for even k. The former describes, for example, the fractional quantum Hall fluid at filling fraction $\nu = 1/k$, where the anyon $\alpha = k$ represents the microscopic electron.

Since the anyons α and $\alpha + k$ have indistinguishable braiding properties, and identical spins for k even, and spins that differ by 1/2 for k odd, the lines of U(1)_k are subject to an equivalence relation: anyons related by a transparent bosonic anyon are to be identified. A bosonic theory can be made into a spin theory by tensoring with the trivial spin TQFT of a transparent fermion $\{1, \psi\}$. We will often follow the convention of leaving this factor implicit when discussing spin TQFTs.

Summarizing, the anyon set and the fusion algebra of $U(1)_k$ is:

• U(1)_k, k even: the theory has k anyons labeled by $\alpha \in \{0, 1, ..., k-1\}$ and a $\mathcal{A} \cong \mathbb{Z}_k$ fusion algebra

$$\alpha \times \beta = \alpha + \beta \mod k \,. \tag{4.3.6}$$

The theory is bosonic and can be defined on an arbitrary three-manifold.

• U(1)_k, k odd: the theory has 2k anyons labeled by $\alpha \in \{0, 1, \dots, 2k-1\}$ and a $\mathcal{A} \cong \mathbb{Z}_{2k}$ fusion algebra

$$\alpha \times \beta = \alpha + \beta \mod 2k \,. \tag{4.3.7}$$

It is a spin TQFT, as signalled by the presence of the transparent fermion $\alpha = k$.

• $U(1)_k \times \{\mathbf{1}, \psi\}, k$ even: the theory has 2k anyons labeled by the pair (α, β) , where $\alpha \in \{0, 1, \dots, k-1\}$ and $\beta \in \{0, 1\}$, and the fusion algebra is $\mathcal{A} \cong \mathbb{Z}_k \times \mathbb{Z}_2$

$$(\alpha,\beta) \times (\alpha',\beta') = (\alpha + \alpha' \mod k, \ \beta + \beta' \mod 2). \tag{4.3.8}$$

It is a spin TQFT by virtue of the tensoring with $\{1, \psi\}$, where ψ is represented by the Wilson line with charges (0, 1).

We now proceed to determine the full set of symmetries of $U(1)_k$ Chern-Simons theory.

4.3.1 Symmetries of $U(1)_k$

We start with the manifest Lagrangian symmetries. $U(1)_k$ with k > 2 has a \mathbb{Z}_2 unitary Lagrangian symmetry $C: a \mapsto -a$, charge conjugation, under which $\mathcal{L} \mapsto \mathcal{L}$, and that acts on the anyons as

$$\mathsf{C} \colon \alpha \mapsto -\alpha \,. \tag{4.3.9}$$

The operation C is not a symmetry of $U(1)_1$ and $U(1)_2$ because charge conjugation acts trivially on all the lines, since $1 = -1 \mod 2$.

Time-reversal is an anti-unitary transformation

$$\mathsf{T}: \begin{cases} a_0(x^0) \mapsto +a_0(-x^0) \\ a_i(x^0) \mapsto -a_i(-x^0) \end{cases}$$
(4.3.10)

which acts on the Wilson lines as $T: W_{\alpha}(\gamma) \mapsto W_{\alpha}(\mathsf{T}\gamma)$, where $\mathsf{T}\gamma$ denotes the time-reflected image of the curve γ . While T is a symmetry of the equations of motion (4.3.2), it does not leave the action invariant, i.e. $\mathcal{L} \mapsto -\mathcal{L}$. This transformation is not a quantum symmetry either since it does not obey the corresponding Ward identity (4.1.2). Therefore, if T is to be a symmetry of $\mathrm{U}(1)_k$, it must act non-trivially on the anyon labels:

$$\mathsf{T} \colon W_{\alpha}(\gamma) \mapsto W_{\mathsf{T}(\alpha)}(\mathsf{T}\gamma) \tag{4.3.11}$$

for some $T: \mathcal{A} \to \mathcal{A}$.

In order to study the quantum symmetries of $U(1)_k$ Chern-Simons theory we first need to understand the automorphisms of its fusion algebra \mathcal{A} . Indeed, as explained in section 4.2, a transformation g is a symmetry of a TQFT if it is an automorphism of its data (\mathcal{A}, θ) which requires, first and foremost, that $g \in Aut(\mathcal{A})$. As usual, any element of $Aut(\mathcal{A})$ is completely determined by its action on the generators of \mathcal{A} . With this in mind, the automorphisms of the fusion algebra \mathcal{A} of $U(1)_k$ Chern-Simons theory are as follows:

• $U(1)_k$, k even. The most general endomorphism of $\mathcal{A} \cong \mathbb{Z}_k$ acts as $g : \alpha \mapsto q\alpha \mod k$, where $q := g(1) \in \mathcal{A}$ and $\alpha \in \{0, 1, \dots, k-1\}$. This lifts to an automorphism of \mathbb{Z}_k if and only if g maps a generator of \mathbb{Z}_k into a generator of \mathbb{Z}_k . This requires q to be relatively prime to k, i.e. gcd(q, k) = 1:

$$g: \alpha \mapsto q\alpha \mod k$$
, $\gcd(q, k) = 1$. (4.3.12)

The number of automorphisms (and of generators) of \mathbb{Z}_k is the number of totatives of k: the number of integers $1 \leq q \leq k$ such that gcd(q, k) = 1. This number is counted by the Euler totient function $\phi(k)$. The automorphism group $Aut(\mathbb{Z}_k)$ is the multiplicative group of integers modulo k, an abelian group often denoted as \mathbb{Z}_k^{\times} . • U(1)_k, k odd. The most general endomorphism of $\mathcal{A} \cong \mathbb{Z}_{2k}$ acts as $g: \alpha \mapsto q\alpha \mod 2k$, where $q := g(1) \in \mathcal{A}$ and $\alpha \in \{0, 1, \dots, 2k - 1\}$. It is an automorphism if and only if q is coprime to 2k:

 $g: \alpha \mapsto q\alpha \mod 2k$, $\gcd(q, 2k) = 1$. (4.3.13)

The automorphisms automatically preserve the transparent fermion $(\alpha = k)$ since $qk = k \mod 2k$ for q odd. The number of automorphisms of \mathbb{Z}_{2k} is the Euler totient function $\phi(2k) = \phi(k)$, the last equality by virtue of k being odd. The automorphism group is $\operatorname{Aut}(\mathbb{Z}_{2k}) = \mathbb{Z}_{2k}^{\times}$.

• $U(1)_k \times \{\mathbf{1}, \psi\}, k$ even. The most general endomorphism of $\mathcal{A} \cong \mathbb{Z}_k \times \mathbb{Z}_2$ acts as

$$g: \begin{pmatrix} \alpha \\ \beta \end{pmatrix} \mapsto \begin{pmatrix} a & b \\ c & d \end{pmatrix} \begin{pmatrix} \alpha \\ \beta \end{pmatrix} \mod k \qquad a, b \in \mathbb{Z}_k \\ \mod 2, \qquad c, d \in \mathbb{Z}_2$$
(4.3.14)

where $\alpha \in \{0, 1, \dots, k-1\}$ and $\beta \in \{0, 1\}$. Such a map is an automorphism if and only if it is invertible (mod k, mod 2). The automorphism group of $\mathbb{Z}_k \times \mathbb{Z}_2$ does not admit as straightforward a description as in the previous cases, but its order is known: $4\phi(k)$ if k/2 is even, and $6\phi(k)$ if k/2 is odd [266, 267]. The automorphism group is generically non-abelian.

Locality of the TQFT requires that the automorphism g preserves the transparent fermion, $g(\psi) = \psi$, that is, it fixes the anyon (0, 1). This implies that the candidate symmetries of $U(1)_k \times \{\mathbf{1}, \psi\}$ with k even are the automorphisms of $\mathbb{Z}_k \times \mathbb{Z}_2$ with b = 0and d = 1. In order for the transformation to be invertible, one must have gcd(a, k) = 1or, if k/2 is odd, gcd(a, k/2) = 1. The number of such transformations is $2\phi(k)$ and $3\phi(k)$ for k/2 even and odd, respectively.

This immediately shows that $U(1)_1$ and $U(1)_2$ have no symmetries since $Aut(\mathbb{Z}_2)$ is trivial, and indeed charge conjugation C acts trivially in these theories.

We have thus characterised all the automorphisms of \mathcal{A} . These are the candidate transformations to be a symmetry of the TQFT. They uplift to symmetries if they respect the topological spin of the lines (up to complex conjugation for anti-unitary symmetries). We turn to this question next.

Anti-unitary Symmetries.

We start by studying the anti-unitary symmetries of $U(1)_k$ Chern-Simons theory. We already established that the canonical time-reversal transformation (4.3.10) is not a symmetry of $U(1)_k$. Since the TQFT data of $U(1)_k$ Chern-Simons theory is determined by the fusion algebra \mathcal{A} and the topological spin θ , an automorphism $\mathsf{T} \in \operatorname{Aut}(\mathcal{A})$ will lead to an anti-unitary symmetry if and only if

$$\theta(\mathsf{T}(\alpha)) = \theta(\mathsf{T}(\alpha))^* \iff h_{\mathsf{T}(\alpha)} = -h_{\alpha} \mod 1.$$
 (4.3.15)
This condition is not satisfied by every automorphism of \mathcal{A} . More importantly, depending on the value of k, there will be cases where there are no automorphisms at all that satisfy (4.3.15). This is precisely what happens for even k, when we regard U(1)_k as a bosonic theory⁹¹:

Proposition 4.3.1 The bosonic theory $U(1)_k$, with k even, is never time-reversal invariant.

Proof. Consider the permutation $\mathsf{T} \colon \alpha \mapsto q\alpha$ for some $q \in [0, k)$. This operation satisfies $h_{\mathsf{T}(\alpha)} = -h_{\alpha} \mod 1$ if and only if

$$\frac{q^2\alpha^2}{2k} + \frac{\alpha^2}{2k} \in \mathbb{Z}.$$
(4.3.16)

If we take, for example, the fundamental line $\alpha = 1$, this requires $\frac{1+q^2}{2k}$ to be an integer. But q must odd for T to be an automorphism, and so $1 + q^2 = 2 \mod 4$, which means that $\frac{1+q^2}{2k}$ cannot be an integer.

We therefore see that the theory $U(1)_k$ can only possibly be time-reversal invariant if we regard it as a spin TQFT. And even if we do so, there will still be some values of k for which $U(1)_k$ admits no time-reversal permutation at all. To see this, define the following:

Definition 4.3.1 We let $\mathbb{T} \subset \mathbb{Z}$ be the set of integers k such that -1 is a quadratic residue modulo k, i.e. $q^2 = -1 \mod k$ for some $q \in \mathbb{Z}$. In other words,

$$\mathbb{T} := \left\{ k \in \mathbb{Z} \mid kp - q^2 = 1 \quad \text{for some } p, q \in \mathbb{Z} \right\}.$$
(4.3.17)

With this, we prove that

Proposition 4.3.2 The spin theory $U(1)_k$ is time-reversal invariant if and only if $k \in \mathbb{T}$.

Proof. We begin with the case of odd k, that is, $U(1)_k$, where $\mathcal{A} \cong \mathbb{Z}_{2k}$. We shall look for the most general automorphism $\mathsf{T} \in \operatorname{Aut}(\mathcal{A})$ that satisfies (4.3.15). Any such operation is of the form

$$\mathsf{T}(\alpha) = q\alpha, \qquad q := \mathsf{T}(1) \in [0, 2k).$$
 (4.3.18)

If we impose that $h_{\mathsf{T}(1)} = -h_1 \mod 1$, we get $1 + q^2 = 2pk$ for some integer p. It is easy to show that this equation is solvable if and only if $k \in \mathbb{T}$. One direction is obvious; for the opposite direction, assume that $1 + \tilde{q}^2 = \tilde{p}k$. If \tilde{p} is even, we are done; if it is odd, then we can set

$$q := \tilde{q} + k, \qquad p := \tilde{q} + \frac{\tilde{p} + k}{2}$$
 (4.3.19)

which satisfy $1 + q^2 = 2pk$, as required (note that $\tilde{p} + k$ is even, and so $p \in \mathbb{Z}$).

⁹¹This result also follows from the fact that the central charge of $U(1)_k$ is not proportional to 4.

Once we ensure the spin of the generator transforms properly under T, it is easy to show that so do the rest of lines. Indeed,

$$h_{\mathsf{T}(\alpha)} = \frac{q^2 \alpha^2}{2k} = \frac{(2pk-1)\alpha^2}{2k} = -\frac{\alpha^2}{2k} \mod 1, \qquad (4.3.20)$$

where we have used that $1 + q^2 = 2pk$.

Finally, it is also easy to show that any integer q that solves $1 + q^2 = 2pk$ will be a time-reversal operation. Indeed, $1 + q^2 = 2pk$ implies that any common factor to k and q must divide 1, and so gcd(q, k) = 1, which means that $\alpha \mapsto q\alpha$ is invertible.

We now move on to the even k case, that is, $U(1)_k \times \{\mathbf{1}, \psi\}$, where $\mathcal{A} \cong \mathbb{Z}_k \times \mathbb{Z}_2$, where the first factor is generated by the fundamental line (1, 0), and the second one by the transparent fermion $\psi = (0, 1)$.

Any fusion endomorphism is fixed once we choose its action on the generators. In fact, the transparent fermion is the only spin h = 1/2 line that braids trivially to all other lines (because $U(1)_k$ is bosonic), and thus the action of time-reversal on it is fixed to $\mathsf{T}(\psi) \equiv \psi$. Therefore, we only have freedom to choose how time-reversal acts on (1,0). We write $\mathsf{T}(1,0) := (q_1,q_2)$ for a pair of integers q_1, q_2 , where $q_1 \in \{0, 1, \ldots, k-1\}$ and $q_2 \in \{0, 1\}$.

Proposition 4.3.1 implies that $q_2 = 0$ is not possible. Therefore, the candidate antiunitary transformation is $T(1,0) \equiv (q,1)$ for some integer $q \in [0,k)$, and so the most general endomorphism is of the form

$$\mathsf{T}(\alpha,\beta) = (q\alpha,\alpha+\beta). \tag{4.3.21}$$

We now insist that the spin of (1,0) is mapped into its negative under time-reversal. Imposing that $h_1 = -h_{T(1)\otimes\psi} \mod 1$ we get $1 + q^2 = (2p - 1)k$ for some integer p. Once again, it is easy to show that this equation is solvable if and only if $k \in \mathbb{T}$. One direction is obvious; for the opposite direction, assume that $1 + \tilde{q}^2 = \tilde{p}k$. Then, upon reducing the equation modulo 4, it becomes clear that \tilde{p} has to be odd, and so we can write $\tilde{p} = 2p - 1$, as we wanted to show.

Once we ensure the spin of the generator transforms properly under T , it is easy to show that so do the rest of lines. Indeed,

$$h_{\mathsf{T}(\alpha,\beta)} = \frac{q^2 \alpha^2}{2k} + \frac{1}{2} (\alpha + \beta)^2 = \frac{(-1 + k(2p - 1))\alpha^2}{2k} + \frac{1}{2} (\alpha + \beta)^2, \qquad (4.3.22)$$

where we have used $q^2 = -1 + k(2p - 1)$. This is clearly equal to

$$h_{\mathsf{T}(\alpha,\beta)} = -\frac{\alpha^2}{2k} + \frac{1}{2}\beta^2 = -h_{(\alpha,\beta)} \mod 1$$
(4.3.23)

as required.

Finally, it is also easy to show that any integer q that solves $1 + q^2 = (2p - 1)k$ will be a time-reversal operation. Indeed, and as before, this equation can only be satisfied if gcd(q,k) = gcd(q,k/2) = 1, and so $(\alpha,\beta) \mapsto (q\alpha,\alpha+\beta)$ is invertible (i.e. an automorphism of $\mathbb{Z}_k \times \mathbb{Z}_2$).

As we can see, the set $\mathbb{T} \subset \mathbb{Z}$ plays a key role in the study of the time-reversal properties of $\mathrm{U}(1)_k$ (and, as we shall see, of $\mathrm{U}(1)^n$). We therefore make a few remarks about this set:

- The first few solutions are $k = 1, 2, 5, 10, 13, 17, 25, 26, 29, 34, 37, 41, 50, 53, 58, 61, 65, 73, 74, 82, 85, 89, 97, \dots$
- A given k is in T if and only if it can be written as $k = a^2 + b^2$ for relatively prime $a, b \in \mathbb{Z}$ (see e.g. [268], theorem 3.21).
- Given the prime decomposition of k

$$k = 2^e \left[\prod_{\pi=1 \mod 4} \pi^{\alpha} \right] \left[\prod_{\pi=3 \mod 4} \pi^{\beta} \right], \qquad (4.3.24)$$

 $k \in \mathbb{T}$ if and only if $e \in \{0, 1\}$ and $\beta_i \equiv 0$ (see e.g. [268], theorem 3.20). In other words, $k \in \mathbb{T}$ if and only if all its prime factors are Pythagorean, or Pythagorean with a single factor of 2. This implies, for example, that $\mathbb{T}^{\text{even}} = 2\mathbb{T}^{\text{odd}}$.

• The set \mathbb{T} contains a special subset \mathbb{P} , defined as those integers k for which the (negative) Pell equation is solvable:

$$\mathbb{P} := \left\{ k \in \mathbb{Z} \mid kp^2 - q^2 = 1 \quad \text{for some } p, q \in \mathbb{Z} \right\}.$$

$$(4.3.25)$$

Unlike \mathbb{T} , the set \mathbb{P} has no simple characterization in terms of the prime decomposition of k. See section 4.7 for some mode details about Pell numbers.

• The density of \mathbb{T} is $\#\{k \in \mathbb{T} \mid k \leq x\} \sim x/\sqrt{\log x}$. It is conjectured that around 57% of the numbers in \mathbb{T} are in \mathbb{P} [269, 270].

If $k \in \mathbb{T}$, there exists an integer $q \in [0, k)$ such that $q^2 = -1 \mod k$. We explain in the section 4.7 how to construct q explicitly.

We now go back to the theory $U(1)_k$. We have the following:

Proposition 4.3.3 The time-reversal symmetry of $U(1)_k$ is an order-four operation (except for k = 1, 2, where it is of order two).

Proof. We shall prove that $T^2 = C$, where $C: \alpha \mapsto -\alpha$ is the unitary \mathbb{Z}_2 charge conjugation symmetry (4.3.9). From this it follows that $T^4 = 1$, and therefore T is an order-four operation (except for k = 1, 2, where C is trivial).⁹²

⁹²The symmetry algebra $\mathsf{T}^2 = \mathsf{C}$ can in principle be extended by fermion parity $(-1)^F$, which does not act on the Wilson lines. The full symmetry algebra is, therefore, either $\mathbb{Z}_4 \times \mathbb{Z}_2$ (corresponding to $\mathsf{T}^4 = 1$) or \mathbb{Z}_8 (corresponding to $\mathsf{T}^4 = (-1)^F$). Figuring out which of these options is realized requires determining how $(-1)^F$ acts on these theories, a subject that is beyond the scope of this chapter.

Showing that $T^2 = C$ is straightforward. If k is odd, then

$$-\mathsf{T}^{2}(\alpha) = -q^{2}\alpha = (1 - 2pk)\alpha = \alpha \mod 2k.$$
(4.3.26)

Similarly, if k is even, then

$$-\mathsf{T}^{2}(\alpha,\beta) = (-q^{2}\alpha, (q+1)\alpha + \beta)$$

= $((1 - (2p - 1)k)\alpha, (q+1)\alpha + \beta)$
= $(\alpha,\beta) \mod (k,2)$ (4.3.27)

where we have used that q is odd.

We see that if $k \in \mathbb{T}$, then there exists some anti-unitary operation T which satisfies a \mathbb{Z}_4 algebra. That being said, there will be, in general, more than one such permutations, and therefore the time-reversal transformation is not unique. We have the following result:

Proposition 4.3.4 If $U(1)_k$ is time-reversal invariant, there are $2^{\varpi(k)}$ different antiunitary permutations, where $\varpi(k)$ denotes the number of distinct prime factors of k for k odd and of k/2 for k even (cf. (4.1.7)).

Proof. Indeed, there are as many permutations as there are solutions to $q^2 = -1 + (2p-1)k$ with $q \in [0, k)$ for k even, and to $q^2 = -1 + 2pk$ with $q \in [0, 2k)$ for k odd. We shall first show that this problem is equivalent to counting the solutions to $\tilde{q}^2 = -1 \mod k$:

- Consider the case with k even. Then any solution to $\tilde{q}^2 = -1 + \tilde{p}k$ must necessarily have \tilde{p} odd (for otherwise we reach a contradiction upon reducing the equation modulo 4), and so we can write $(\tilde{q}, \tilde{p}) = (q, 2p 1)$, which yields $q^2 = -1 + (2p 1)k$, as required.
- We now consider the case with k odd. We claim that the solutions to $q^2 = -1 + 2pk$ with $q \in [0, 2k)$ can be put in a bijection with solutions to $\tilde{q}^2 = -1 + \tilde{p}k$ with $\tilde{q} \in [0, k)$. First, assume we are given the set $\{\tilde{q} \in [0, k)\}$; we construct the set $\{q \in [0, 2k)\}$ as follows: if \tilde{q} is odd, then \tilde{p} must be even, and so $(q, 2p) = (\tilde{q}, \tilde{p})$; on the other hand, if \tilde{q} is even, then \tilde{p} must be odd, and so $(q, 2p) = (\tilde{q}+k, \tilde{p}+2\tilde{q}+k)$. Conversely, if we are given the set $\{q \in [0, 2k)\}$, we write $(\tilde{q}, \tilde{p}) = (q, p)$ if $q \in [0, k)$, and $(\tilde{q}, \tilde{p}) = (q - k, p - 2q + k)$ if $\tilde{q} \in [k, 2k)$.

We thus see that we may reduce our problem to counting solutions to $q^2 = -1 \mod k$, both for k even and odd. It is a well-known result that the number of solutions is precisely $2^{\varpi(k)}$, see for example theorem 6.3 in [271] (together with remark 6.2 therein). The intuition behind this result (and which can be generalised to any polynomial congruence) is the following. Any solution to $q^2 = -1 \mod k$ can be reconstructed uniquely from the solutions to $q_i^2 = -1 \mod \pi_i$, where π_i are the prime factors of k. Each congruence $q_i^2 = -1 \mod \pi_i$ is solvable (because π_i is Pythagorean), and it has two solutions $\pm q_i$ (and only two, as per

Lagrange's theorem, except for $\pi = 2$, where only solution is $q_i = 1$, inasmuch as $1 = -1 \mod 2$). As there are $\varpi(k)$ congruences, each having two solutions, the total number of solutions is $2^{\varpi(k)}$, as claimed.

For completeness, we mention that one can prove that $k \in \mathbb{T}$ is sufficient for time-reversal invariance using a path integral argument, which is quite similar to one in [251, 252] where it was used to show time-reversal invariance for $k \in \mathbb{P} \subset \mathbb{T}$. The argument is straightforward but it does not prove that the condition $k \in \mathbb{T}$ is also necessary.

Proposition 4.3.5 It follows from a path integral argument that when $k \in \mathbb{T}$ the theory $U(1)_k$ is time-reversal invariant as a spin TQFT.

Proof. Take two arbitrary integers m, n with m is odd and n even, and such that

$$mn - q^2 = 1 \tag{4.3.28}$$

for some integer q (which can easily seen to be odd). We shall prove that $U(1)_m$ and $U(1)_n \times \{\mathbf{1}, \psi\}$ are both time-reversal invariant.

Take the Lagrangian of $U(1)_m \times U(1)_{-n}$

$$4\pi \mathcal{L} = m \, a \, \mathrm{d}a - n \, b \, \mathrm{d}b \tag{4.3.29}$$

whose Wilson lines are of the form

$$\exp\left[i\alpha\int a+i\beta\int b\right],\qquad (\alpha,\beta)\in\mathbb{Z}_{2m}\times\mathbb{Z}_n.$$
(4.3.30)

Under the $\operatorname{GL}_2(\mathbb{Z})$ transformation

$$\mathsf{T}: \begin{pmatrix} a \\ b \end{pmatrix} \mapsto \begin{pmatrix} q & -n \\ m & -q \end{pmatrix} \begin{pmatrix} a \\ b \end{pmatrix}, \qquad (4.3.31)$$

the Lagrangian becomes

$$\mathsf{T} \colon 4\pi\mathcal{L} \mapsto -m\,a\,\mathrm{d}a + n\,b\,\mathrm{d}b \equiv -4\pi\mathcal{L} \tag{4.3.32}$$

and the lines map according to

$$\mathsf{T}: (\alpha, \beta) \mapsto (q\alpha + m\beta, -n\alpha - q\beta). \tag{4.3.33}$$

We therefore see that $U(1)_m \times U(1)_{-n} \leftrightarrow U(1)_{-m} \times U(1)_n$, i.e., the product is time-reversal invariant. The explicit duality map is given by (4.3.33).

We now prove that $U(1)_m$ is time-reversal invariant. To this end, we note that the theory above contains a sub-group of lines of the form $(\alpha, 0)$, which is isomorphic to $U(1)_m$, with isomorphism $\alpha \leftrightarrow (\alpha, 0)$. Time-reversal restricts to a well-defined action on $U(1)_m$, because

$$\mathsf{T}: (\alpha, 0) \mapsto (q\alpha, -n\alpha) \sim (q\alpha, 0) \tag{4.3.34}$$

where we have used the fact that n is even.

We next prove that $U(1)_n \times \{\mathbf{1}, \psi\}$ is time-reversal invariant. To this end, we note that the theory above contains a sub-group of lines of the form $(0, \beta)$ and (m, β) , which is isomorphic to $U(1)_n \times \{\mathbf{1}, \psi\}$, with isomorphism $\beta \otimes \mathbf{1} \leftrightarrow (0, \beta)$ and $\beta \otimes \psi \leftrightarrow (m, \beta)$. Time-reversal restricts to a well-defined action on $U(1)_n \times \{\mathbf{1}, \psi\}$, because

$$T: (0, \beta) \mapsto (m\beta, -q\beta)$$

$$T: (m, \beta) \mapsto (qm + m\beta, -nm - q\beta) \sim (m(1 + \beta), -q\beta)$$
(4.3.35)

where we have used the fact that n is even and q is odd. This completes the proof.

As a consistency check, we note that the action of time-reversal on the lines of $U(1)_m$ is $T(\alpha) = q\alpha$, and that on $U(1)_n \times \{\mathbf{1}, \psi\}$ is $T(\beta \otimes \psi^{\gamma}) = q\beta \otimes \psi^{\beta+\gamma}$, with $\gamma = 0, 1$. This is precisely the same map we found in proposition 4.3.2.

One can couple the theory $U(1)_k$ to electromagnetism by turning on a background $U(1)_B$ connection. If $k \in \mathbb{T}$, then time-reversal remains a symmetry in the presence of this background field, but at the cost of introducing a Chern-Simons counterterm for the electromagnetic field, with fractional coefficient. This means that there is a mixed $T - U(1)_B$ 't Hooft anomaly,⁹³ and so the system can only be defined on the boundary of a 3+1 manifold. Using the Lagrangian argument above, and following the same reasoning as in [111, 252], it is easy to prove that the anomaly is given by a 3+1 dimensional topological term $\theta = 2\pi/k$ for $U(1)_B$.

Remark 4.3.1 It is common that in theories that are symmetric under both time-reversal and charge conjugation, the operators T and CT constitute two separate \mathbb{Z}_2 symmetries, both of which represent suitable time-reversal operations. These two symmetries are independent: they have different anomalies, they may be affected by magnetic symmetries (if any), and may be interchanged under duality (see e.g. [34]). In our case, these two symmetries in fact combine into a single \mathbb{Z}_4 algebra, $T^3 = CT$, and so they do not correspond to independent symmetries.

Remark 4.3.2 It is interesting to note that we obtained $k \in \mathbb{T}$ as a *necessary* condition just by insisting that the fundamental line has a partner with opposite spin. In turns, this condition was also seen to be *sufficient*, so one may wonder if a similar phenomenon may occur in other topological systems. In other words, given an arbitrary TQFT, does the matching of the spin of a single line guarantee that the theory is time-reversal invariant? Generically speaking, the answer is *no*, as there are many examples where a specific pair of lines match but others do not. A much stronger test is the matching of *all* the lines, that is, the condition that $\{h\} = \{-h\} \mod 1$ (with equality as multisets, that is, taking into account multiplicities). For example, one may we check that the set of spins matches for the

⁹³We thank N. Seiberg for this comment.

theory $SU(N)_N$, for N = 1, 5, 13, 17, ... (both as a bosonic and a spin TQFTs), all of which happen to be Pythagorean primes. As suggestive as this may seem, the pairs of lines that have opposite spin do not in general have the same quantum dimension, so these theories are not time-reversal invariant. $(SU(N)_N/\mathbb{Z}_N)$ is, however, time-reversal invariant for all N [112])

Upon turning on a background metric, the duality $U(1)_k \leftrightarrow U(1)_{-k}$ no longer holds as written, because the two theories have a different framing anomaly, and so they couple to the background gravitational field differently. This can be interpreted as a mixed anomaly between time-reversal and gravity. To maintain the duality one must adjust gravitational Chern-Simons counterterms on both sides so that their central charges agree. In particular, one may use $U(1)_{\pm 1}$ to add/subtract one unit of central charge, without otherwise changing the topological content of the theory. With this in mind, the precise duality reads

$$U(1)_{+k} \times U(1)_{-1} \longleftrightarrow U(1)_{-k} \times U(1)_{+1}.$$

$$(4.3.36)$$

These theories can be represented by the matrices $K = \text{diag}(\pm k, \mp 1)$. In the bosonic case, we already included a factor $\{\mathbf{1}, \psi\}$ to make the theory into a spin theory; here we see that this factor also fixes the central charge, provided we identify $\{\mathbf{1}, \psi\} \equiv U(1)_{-\text{sign}(k)}$. In the spin case, this factor also fixes the central charge, but leaves the spectrum of lines unaffected.

It is clear that without the factor of $U(1)_{\pm 1}$, time-reversal cannot possibly be a Lagrangian symmetry of the $U(1)_k$ theory, because the only $\operatorname{GL}_1(\mathbb{Z})$ transformations are $a \to \pm a$, neither of which maps $k \to -k$. More generally, the signature of the K-matrix is invariant under congruence $(\operatorname{sign}(K) \equiv \operatorname{sign}(G^t K G)$ for any $G \in \operatorname{GL}_n(\mathbb{Z})$, as per the Sylvester law of inertia) and so time-reversal can only be a Lagrangian symmetry if the signature vanishes (inasmuch as the chiral central charge is odd under time-reversal). Once we fix the central charge, time-reversal may (but need not) become a Lagrangian symmetry. It is interesting to note that, in the case at hand, this happens only for a subset of \mathbb{T} : only for a specific set of values of k is the Lagrangian time-reversal invariant. One can show that this is so if and only if $k \in \mathbb{P}$:

Proposition 4.3.6 The Lagrangian of the theory $U(1)_k \times U(1)_{-1}$ is time-reversal invariant if and only if k satisfies the negative Pell equation.

Proof. The fact that this condition is necessary can be obtained by looking at the bottomright component of the equation $K = -G^t K G$, where K = diag(k, -1). That this is also sufficient was originally shown in [251, 252], and follows from the explicit change of variables

. .

$$\mathsf{T}: \begin{pmatrix} a \\ b \end{pmatrix} \mapsto \begin{pmatrix} q & p \\ -kp & -q \end{pmatrix} \begin{pmatrix} a \\ b \end{pmatrix}, \qquad kp^2 - q^2 = 1.$$
(4.3.37)

Proposition 4.7.4 in section 4.7 generalizes the construction to K = diag(k, k').

This means that if $k \in \mathbb{T}$ but it is not in \mathbb{P} , then $U(1)_k$ will be time-reversal invariant, but the invariance will not be a symmetry of the Lagrangian, not even if we include the factor of $U(1)_{\pm 1}$. It is a quantum symmetry of $U(1)_k \times U(1)_{-1}$. However, it is possible that in a different abelian Chern-Simons realization of the same TQFT data that the symmetry becomes Lagrangian.

Remark 4.3.3 As a physical application of proposition 4.3.2, note that given an integer ℓ such that both k and $k + \ell^2$ are in \mathbb{T} , the theory

$$U(1)_{k+\ell^2/2} + \Psi \tag{4.3.38}$$

with Ψ a Dirac fermions of charge ℓ is infrared time-reversal invariant for $m \neq 0$. Indeed, integrating the fermions out we get

$$U(1)_k \longleftrightarrow U(1)_{-k} \tag{4.3.39}$$

for $m \to -\infty$, and

$$U(1)_{k+\ell^2} \longleftrightarrow U(1)_{-(k+\ell^2)} \tag{4.3.40}$$

for $m \to +\infty$. This suggests that the CFT at the massless point m = 0 may be time-reversal invariant as well. These gauge theories, in spite of not being time-reversal invariant in the ultraviolet, have an emergent time-reversal symmetry across the entire infrared phase diagram. The first few solutions of $(k, k + \ell^2) \in \mathbb{T} \times \mathbb{T}$ are

$$\begin{split} \ell &= 0: & k = 1, 2, 5, 10, 13, 17, 26, 25, 29, \dots \\ \ell &= 1: & k = 1, 25, 73, 145, 169, 193, 289, \dots \\ \ell &= 2: & k = 1, 13, 37, 61, 85, 97, 109, 181, \dots \\ \ell &= 3: & k = 1, 17, 25, 41, 65, 73, 97, 113, \dots \\ \text{etc.} \end{split}$$

(For $\ell = 0$ one gets an infrared emergent time-reversal symmetry in Maxwell-Chern-Simons theories). A similar phenomenon occurs in non-abelian theories. For example, using the Chern-Simons dualities $U(1)_k \leftrightarrow SU(k)_{-1} \leftrightarrow SO(k)_{-2}$ we observe that the theories $SU(k)_0$ and $SO(k)_0$ with two fundamental Dirac fermions, and $k \in \mathbb{T}$, are time-reversal invariant in their massive phases (necessarily also in their massless phase, because the UV theory is time-reversal invariant).

It is an interesting number-theoretic problem whether there exists, for a given $\ell \in \mathbb{Z}$, an infinite number of pairs with $(k, k + \ell^2) \in \mathbb{T}^2$. This is similar in spirit to the so-called Polignac conjecture, which states that there exists an infinite number of pairs of primes of the form $(\pi, \pi + n)$ for any $n \in 2\mathbb{N}$ (recall that primes $\pi > 2$ are in \mathbb{T} iff they are Pythagorean). Assuming this conjecture with $\ell^2 = n$ (which requires ℓ to be even), and noting that π and $\pi + \ell^2$ are either both Pythagorean or neither is, suggests that indeed there exists an infinite number of pairs $(k, k + \ell^2) \in \mathbb{T}^2$, at least for ℓ even.

Unitary Symmetries.

We now move on to the unitary symmetries of $U(1)_k$. The principle is identical to the anti-unitary case, the only difference being a sign flip. By definition, an automorphism $U \in Aut(\mathcal{A})$ is a unitary symmetry of (\mathcal{A}, θ) if and only if

$$\theta(\mathsf{U}(\alpha)) = \theta(\alpha) \iff h_{\mathsf{U}(\alpha)} = h_{\alpha} \mod 1.$$
 (4.3.42)

As in the anti-unitary case, any permutation is fixed once we choose how the generators transform. The corresponding permutation will be a symmetry if it satisfies (4.3.42). But, unlike the case of anti-unitary symmetries, here the equation $h_{U(\alpha)} = h_{\alpha} \mod 1$ always admits solutions: at least, the trivial permutation and charge conjugation C exist. These are transformations that leave the action of the theory invariant. We thus solve a more refined problem: the interesting automorphisms will be those that are neither trivial nor C. Another difference with the anti-unitary case is that, in general, we will find non-trivial symmetries also in the bosonic case.

We begin with the following observation:

Proposition 4.3.7 All the unitary symmetries of $U(1)_k$ (as a bosonic TQFT if k is even) are transformations of the form

$$\mathsf{U}\colon \alpha \mapsto q\alpha \tag{4.3.43}$$

for some integer q that satisfies

$$q^2 = 1 + 2pk \,. \tag{4.3.44}$$

Similarly, the unitary symmetries of $U(1)_k \times \{\mathbf{1}, \psi\}$ for k even are of the form

$$\mathsf{U}\colon(\alpha,\beta)\mapsto(q\alpha,p\alpha+\beta)\tag{4.3.45}$$

for some integer q that satisfies

$$q^2 = 1 + pk \,. \tag{4.3.46}$$

The solutions $q = \pm 1$ (with p = 0) always exist and corresponds to the trivial permutation, and charge conjugation C (4.3.9), respectively. All other solutions correspond to quantum symmetries.

Proof. The case of $U(1)_k$ (as a bosonic TQFT if k is even) is essentially identical to the anti-unitary case. Let us therefore consider $U(1)_k \times \{\mathbf{1}, \psi\}$ with k even. Any fusion endomorphism that fixes the transparent fermion is of the form

$$\mathsf{U}\colon(\alpha,\beta)\mapsto(q\alpha,c\alpha+\beta)\tag{4.3.47}$$

for a pair of integers c, q. If c is even, U does not mix the lines of $U(1)_k$ with the transparent fermion, and so this is a symmetry that was also present in the bosonic case. If c is odd, the

permutation does mix the lines, and so it is only a symmetry of the fermionic theory. In any case, requiring that the spin of the fundamental line is equal to the spin of its image under U, we get

$$q^2 = 1 - (c^2 + 2\tilde{p})k \tag{4.3.48}$$

for some integer \tilde{p} . Letting $-p := c^2 + 2\tilde{p}$ we get the expression in the proposition (note that p and c have the same parity, and therefore we can replace the latter by the former in the transformation U). It is straightforward to check that if the spin of the fundamental line is invariant under U, so is the spin of the rest of lines. Finally, it is easy to show that *any* solution of (4.3.48) corresponds to a permutation (i.e. q automatically has the appropriate coprimality with k to define an automorphism).

As in the anti-unitary case, all the unitary permutations have the same order:

Proposition 4.3.8 All the unitary symmetries of $U(1)_k$ (either as a bosonic or as an spin TQFT) are of order-two.

Proof. For $U(1)_k$ we have

$$\mathsf{U}^2 \colon \alpha \mapsto q^2 \alpha = \alpha + 2pk\alpha \tag{4.3.49}$$

which indeed equals α . In the case of $U(1)_k \times \{1, \psi\}$, the argument is identical:

$$\mathsf{U}^2\colon (\alpha,\beta)\mapsto (q^2\alpha,p\alpha(q+1)+\beta) = (\alpha+pk\alpha,p\alpha(q+1)+\beta) \tag{4.3.50}$$

which, using the fact that q is odd, yields (α, β) , as claimed.

Take the theory $U(1)_k$, without the factor of $\{1, \psi\}$ for k even. A slight modification of the argument in proposition 4.3.4 proves that the number of solutions in the range $q \in [0, 2k)$ for k odd, and in the range $q \in [0, k)$ for k even, is $2^{\varpi(k)}$, as in the anti-unitary case. Therefore, in order to have solutions other than $U \in \{1, C\}$, the level k must not be a prime power or twice a prime power. Such non-trivial solutions will not be a symmetry of the classical Lagrangian, because $p \neq 0$. They correspond to quantum symmetries.

For k even, one may also study the unitary symmetries of the theory as a spin TQFT, that is, of $U(1)_k \times \{1, \psi\}$. The symmetries of the bosonic theory are inherited in the fermionic theory, but new symmetries may appear – those under which the transparent fermion mixes non-trivially. The automorphisms are given by the integers q that satisfy $q^2 = 1 + pk$, and whether the transparent fermion mixes is controlled by the parity of p. It is easy to show that the number of solutions is $2^{\varpi(k)}$ for $k = 2 \mod 4$, and $2^{\varpi(k/2)+1}$ for $k = 0 \mod 4$. Therefore, there is an enhancement of symmetry when going from the bosonic theory to the spin theory if and only if k is a multiple of 8: only in that case may the fermion mix. The additional transformation that appears when the theory is uplifted from bosonic to spin is generated by q = k/2 - 1 (with p = k/4 - 1). We summarise these claims as follows: **Proposition 4.3.9** All the unitary symmetries of $U(1)_k$ (both as a spin theory and as a bosonic theory in the case of k even) are \mathbb{Z}_2 -valued. There are $2^{\varpi(k)}$ permutations if k is not a multiple of 8. If $k = 0 \mod 8$, then there are $2^{\varpi(k)}$ permutations in the bosonic theory, and twice as many in the spin theory.

Needless to say, one may compose any non-trivial unitary symmetry with a given T to yield a different notion of time-reversal. Similarly, composing any two time-reversal operations results in a unitary symmetry, and composing two unitary symmetries leads to another unitary symmetry. In fact, a stronger result is true. Let $\{T_i\}$ be all time-reversal symmetries, and $\{U_i\}$ be unitary ones. Let $U_0 := 1$, pick some element of $\{T_i\}$, and denote it by T_0 . Then any T_i can be obtained by acting with some U_i on T_0 . Indeed, it is easy to see that the sets

$$\{\mathsf{T}_i\} \quad \text{and} \quad \{\mathsf{T}_0\mathsf{U}_i\} \tag{4.3.51}$$

contain the same number of elements (because T_0 is invertible, so $T_0U_i \neq T_0U_j$ for $i \neq j$), and so they must be identical. Thus, perhaps after relabelling its elements, we have

$$\mathsf{T}_i \equiv \mathsf{U}_i \mathsf{T}_0 \tag{4.3.52}$$

and so one time-reversal permutation suffices to generate them all.

Recalling definition 4.2.1, all these considerations can be put together to obtain the following:

Proposition 4.3.10 The group of symmetries of $U(1)_k$ as a spin TQFT is

$$\operatorname{Aut}(\operatorname{U}(1)_{k}^{\operatorname{spin}}) = \mathbb{Z}_{4} \times (\mathbb{Z}_{2})^{\varpi(k)-1}$$

$$\operatorname{Aut}_{U}(\operatorname{U}(1)_{k}^{\operatorname{spin}}) = (\mathbb{Z}_{2})^{\varpi(k)}$$

$$(4.3.53)$$

if $k \in \mathbb{T}$, and

$$\operatorname{Aut}(\operatorname{U}(1)_{k}^{\operatorname{spin}}) = \operatorname{Aut}_{U}(\operatorname{U}(1)_{k}^{\operatorname{spin}}) = \begin{cases} (\mathbb{Z}_{2})^{\varpi(k)+1} & k = 0 \mod 8\\ (\mathbb{Z}_{2})^{\varpi(k)} & \text{otherwise} \end{cases}$$
(4.3.54)

otherwise. On the other hand, as a bosonic theory (with k even), the group reads

$$\operatorname{Aut}(\operatorname{U}(1)_k^{\operatorname{bosonic}}) = \operatorname{Aut}_U(\operatorname{U}(1)_k^{\operatorname{bosonic}}) = (\mathbb{Z}_2)^{\varpi(k)}.$$
(4.3.55)

4.3.2 Minimal abelian TQFT

An important abelian theory that appears in the study of the one-form symmetries of threedimensional TQFTs is the so-called "minimal abelian TQFT" [38, 249, 253, 272]. This theory is denoted by $\mathcal{A}^{N,t}$ (also by $\mathbb{Z}_N^{(t)}$), with N, t two integers, which must be coprime if we require the theory to be modular. The number of lines is N, which can be labelled as s = 1, 2, ..., N. Fusion corresponds to addition modulo N, $s \times s' = (s + s' \mod N)$, i.e. the fusion algebra is \mathbb{Z}_N . The spin of the line s is $h_s = t \frac{s^2}{2N}$. For example, if k is even, then $U(1)_k = \mathcal{A}^{k,1}$; if k is odd, then $U(1)_k = \mathcal{A}^{2k,2}$ (which, indeed, is not modular, because the braiding matrix has a non-trivial kernel). All these theories admit an abelian Chern-Simons representation (e.g. for t = N - 1 the K-matrix is the Cartan matrix of SU(N)).

The analysis of the symmetries of $\mathcal{A}^{N,t}$ is essentially identical to that of $U(1)_k$ because the fusion algebra is also cyclic. For example, following the same reasoning as in the 1×1 case, this theory is seen to be time-reversal invariant if and only if

$$2Np = t(1+q^2) \tag{4.3.56}$$

is solvable for some integers p, q. It is easy to prove that this equation is solvable if and only if

$$t \in \mu(N)\mathbb{Z} \,. \tag{4.3.57}$$

Indeed, by reducing (4.3.56) modulo $\mu(N)$ we get $t(1+q^2) = 0 \mod \mu(N)$; but $(1+q^2)$ is never divisible by a prime of the form 4n + 3, and so t itself mush vanish modulo $\mu(N)$, showing that $t \propto \mu(N)$ is necessary. Conversely, noting that $N/\mu(N)$ is always in \mathbb{T}^{odd} , we know that there exists a pair of integers \tilde{p}, q such that $2\tilde{p}N = \mu(N)(1+q^2)$; multiplying this equation by $t/\mu(N)$ and letting $p = \tilde{p}t/\mu(N)$ we find that $t \propto \mu(N)$ is also sufficient.

Alternatively, one may rewrite (4.3.57) as a condition on N instead of t, as follows:

$$N \in \bigcup_{d|t} d \,\mathbb{T}^{\text{odd}} \,. \tag{4.3.58}$$

Indeed, if $N \in d\mathbb{T}^{\text{odd}}$ for some d|t, then there exists some \tilde{p}, q such that $2(N/d)\tilde{p} = 1 + q^2$; multiplying this equation by t/d and letting $p = \tilde{p}t/d$ shows that (4.3.56) is solvable. Conversely, if $N \notin d\mathbb{T}^{\text{odd}}$ for any d|t then, in particular, $N \notin t\mathbb{T}^{\text{odd}}$ (and, if $t \in 2\mathbb{Z}$, then $N \notin (t/2)\mathbb{T}^{\text{odd}}$ either), and so equation (4.3.56) is not solvable (note that if t is odd then N must be odd as well).

If we further assume that (N, t) = 1, the expression (4.3.58) can be simplified into

$$N \in \mathbb{T}^{\text{odd}} \setminus \left(\bigcup_{\pi \mid p} \pi \mathbb{Z}\right) \tag{4.3.59}$$

where π are the Pythagorean prime factors of p.

As $\mathcal{A}^{N,t}$ has a single generator, its group of symmetries is abelian, and can be studied along the same lines as in the U(1)_k case.

4.4 $U(1)^n$ Chern-Simons theory

We now move on to Chern-Simons theories that contain an arbitrary number of factors of U(1). As a Lagrangian theory, the system is described by

$$\mathcal{L} = \frac{1}{4\pi} a^t K \mathrm{d}a \tag{4.4.1}$$

for a $U(1)^n$ gauge field $a^t = (a_1, a_2, \ldots, a_n)$. The Lagrangian is metric independent and, although not manifestly so, gauge invariant provided the coefficient matrix $K \in \mathbb{Z}^{n \times n}$ is symmetric and integral-valued. Generically speaking, the theory depends on the orientation of spacetime and, if at least diagonal component of K is odd, on the spin structure. The theory has central charge $c = \operatorname{sign}(K)$ (the signature of K), which controls the coupling to the Chern-Simons form for the background metric, via the framing anomaly. To keep matters simple, we shall often turn off this metric, and any other background field one may ultimately want to couple a to.

The observables of the theory are the Wilson lines, modulo local bosonic operators. These lines are of the form

$$W_{\vec{\alpha}}(\gamma) := \exp\left[i\vec{\alpha}^t \int_{\gamma} a\right], \qquad (4.4.2)$$

where $\vec{\alpha} \in \mathbb{Z}^n$ is the representation $U(1)^n \ni \theta \mapsto e^{i\vec{\alpha}\cdot\theta}$. We shall call $\vec{\alpha}$ the *charge* of $W_{\vec{\alpha}}$, and we will often denote the line itself by $\vec{\alpha}$.

These lines can be thought of as the worldlines of anyons, i.e., particles with fractional statistics. In particular, they have spin and may braid non-trivially. If a line $\vec{\alpha}$ braids around a line $\vec{\beta}$, their product picks up a phase $B(\vec{\alpha}, \vec{\beta}) \in U(1)$, where

$$B(\vec{\alpha}, \vec{\beta}) := \exp\left[2\pi i \,\vec{\alpha}^t K^{-1} \vec{\beta}\right] \,. \tag{4.4.3}$$

Similarly, the topological spin of the line corresponds to half self-braiding,

$$\theta(\vec{\alpha}) := \exp\left[2\pi i \, h_{\vec{\alpha}}\right], \qquad h_{\vec{\alpha}} := \frac{1}{2}\vec{\alpha}^t K^{-1}\vec{\alpha}.$$
(4.4.4)

The function θ is said to be a *quadratic refinement* of the bilinear form B, because one has

$$B(\vec{\alpha}, \vec{\beta}) \equiv \frac{\theta(\vec{\alpha} + \vec{\beta})}{\theta(\vec{\alpha})\theta(\vec{\beta})}.$$
(4.4.5)

This implies that the spin of the lines determines their braiding unambiguously; one need not keep track of the latter.

An operator is said to be *local* if it braids trivially with any other line. In particular, any line with $\vec{\alpha}$ proportional to a column of K satisfies $B(\vec{\alpha}, \vec{\beta}) \equiv 1$ for any $\vec{\beta}$, and so it will be local. If, furthermore, the corresponding column has even diagonal element, then $h_{\vec{\alpha}} = 0$ mod 1, and so the local line will be bosonic. As before, lines differing by such a local operator are identified, and so the degrees of freedom of the theory are in fact finite. More explicitly, we have the following:

- If all the diagonal components of K are even, then all the local operators are bosonic, and we need not specify a spin structure to define the theory. It is a bosonic TQFT. Any two lines that are congruent modulo some linear combination (with integer coefficients) of the columns of K are identified, which means that the lines live in the lattice Zⁿ/KZⁿ. There are |det K| independent lines, which can be taken to be all the lattice points in the *n*-dimensional parallelepiped spanned by the columns of K.
- If at least one diagonal component of K is odd, the theory contains a local fermionic operator, which requires a choice of spin structure. The theory is a spin TQFT. Any two lines that are congruent modulo some linear combination (with integer coefficients) of the columns of K are identified, except if they differ by a local fermion. This means that the lines live in the lattice $(\mathbb{Z}^n/K\mathbb{Z}^n) \times \mathbb{Z}_2$. There are $2|\det K|$ independent lines, which can be taken to be all the lattice points in the *n*-dimensional parallelepiped spanned by the columns of K, together with a \mathbb{Z}_2 label that specifies if the line carries a local fermion or not. Alternatively, a basis of lines can be taken to be all the lattice points in the *n*-dimensional parallelepiped spanned by the columns of \tilde{K} , where \tilde{K} is the matrix given by doubling any one column of K with odd diagonal component.

The spectrum of lines is given by the set $\mathcal{A} := \mathbb{Z}^n / \sim$, where

$$\vec{\alpha} \sim \vec{\beta} \iff \vec{\alpha} = \vec{\beta} + K\vec{\gamma},$$
 (4.4.6)

where $\vec{\gamma}$ is any tuple of integers with

$$\sum_{K_{ii} \text{ odd}} \gamma_i = \text{even} \,. \tag{4.4.7}$$

Reducing \mathbb{Z}^n modulo K, instead of modulo \sim , would be tantamount to identifying the local fermion, if any, with the vacuum. In other words, we would forget about the information carried by such a line. This would not be correct: we need the \mathbb{Z}_2 label to signal the presence of ψ . This extra piece of information resolves the ambiguity in lifting the symmetric form B into the quadratic form θ . We shall nevertheless often refer to the equivalence \sim as "reduction modulo K", in order to keep the notation as simple as possible.

Due to the abelian nature of the gauge fields, any pair of unbraided lines $\vec{\alpha}, \vec{\beta}$ can be brought together to form a line of charge $\vec{\alpha} + \vec{\beta}$. In other words, the fusion rules of the theory are

$$\vec{\alpha} \times \vec{\beta} := (\vec{\alpha} + \vec{\beta} \mod K). \tag{4.4.8}$$

The theory described by a given matrix K may have several symmetries. The main focus of this paper is to study the zero-form symmetries, but for completeness we mention that the one-form symmetry group can be obtained by bringing K into its Smith normal form $K \to \text{diag}(k_1, k_2, \ldots, k_n)$, where k_i is the greatest common divisor of all $i \times i$ minors of K. Given this canonical decomposition, the one-form symmetry group is

$$\bigoplus_{i=1}^{n} \mathbb{Z}_{k_i} \,. \tag{4.4.9}$$

We now move on to the zero-form symmetries of the system. These are, by definition, the permutations of the lines that respect their topological properties. A unitary zero-form symmetry of the corresponding system is an automorphism $U: \mathcal{A} \to \mathcal{A}$ that satisfies

$$U(a \times b) = U(a) \times U(b)$$

$$\theta(U(a)) = \theta(a)$$

$$B(U(a), U(b)) = B(a, b)$$

(4.4.10)

for all $a, b \in \mathcal{A}$. Similarly, an anti-unitary zero-form symmetry is an automorphism $\mathsf{T} \colon \mathcal{A} \to \mathcal{A}$ that satisfies

$$I(a \times b) = I(a) \times I(b)$$

$$\theta(\mathsf{T}(a)) = \theta(a)^*$$

$$B(\mathsf{T}(a), \mathsf{T}(b)) = B(a, b)^*.$$

(4.4.11)

Thanks to (4.4.5), the braiding is determined by the spin, and so the third condition is automatically guaranteed to hold if the first two do; we nevertheless find it convenient to keep track of the braiding matrix explicitly.

We have denoted the anti-unitary symmetries by T because we will think of them as a time-reversal operation (or a reflection in the Euclidean setting). These symmetries do not always exist: only for some special matrices K is the system independent of the orientation of spacetime. In particular, as the Lagrangian is odd under the reversal of orientation, we require K and -K to describe equivalent theories: the theories with matrices K and -K must be dual.

A sufficient condition for the theories described by two matrices K_1, K_2 to be equivalent is that they are congruent, i.e., $\operatorname{GL}_n(\mathbb{Z})$ -equivalent: that there exists a unimodular matrix G such that $K_1 \equiv G^t K_2 G$, as follows from the redefinition $a_2 := Ga_1$. The matrix G is required to be unimodular because the change of variables has to be invertible and respect the normalisation of the gauge fields. We shall refer to these equivalences of theories as *Lagrangian* (or classical) symmetries, because they are manifest symmetries of the Lagrangian. As we shall show, one may have matrices K_1, K_2 that are not $\operatorname{GL}_n(\mathbb{Z})$ -equivalent, and yet the theories described by them are nevertheless equivalent. This latter notion of equivalence we refer to as a quantum symmetry, or as a duality. Dualities of TQFTs are often valid only when the theory is regarded as a spin TQFT. In order to turn a bosonic theory into a spin TQFT, it suffices to tensor the theory by the trivial spin TQFT $U(1)_{\pm 1} = \{\mathbf{1}, \psi\}$, where **1** is a local boson and ψ a local fermion. Tensoring a theory that is already spin by this trivial factor leaves the TQFT unaffected, inasmuch as we identify local fermions anyway (because they differ by a local boson: $\psi_1 = (\psi_1 \psi_2) \psi_2$).

If we turn on some background field that couples to a given TQFT, then one may need to adjust appropriate counterterms for it on both sides of the duality. The canonical example is the coupling to background gravity, which is controlled by the central charge of the theory (through the framing anomaly). In particular, the central charge – being the signature of the *K*-matrix – is odd under time-reversal, which means that a theory can only be time-reversal invariant in the presence of gravity if the central charge vanishes. In this sense, a theory being invariant in flat spacetime may require a gravitational counterterm to remain invariant when the metric is nontrivial. Noting that $U(1)_{\pm 1}$ is essentially trivial (it is an SPT) but has central charge ± 1 , one may add as many factors of this theory as necessary so that the theory under consideration has vanishing central charge, as required to maintain the time-reversal symmetry when turning on a background metric. If the theory is already spin, tensoring by $U(1)_{\pm 1} = \{1, \psi\}$ has no effect other than changing the central charge; but for a bosonic system, this factor turns the theory into a spin TQFT.

4.4.1 Symmetries of $U(1)^n$

The analysis of the symmetries of a system described by a matrix K is essentially identical to that of $U(1)_k$: the symmetries are those automorphisms of the fusion algebra that respect the spin of the lines. The most general endomorphism of $\mathcal{A} \cong \mathbb{Z}^n / \sim$ is

$$g \colon \vec{\alpha} \mapsto Q\vec{\alpha} \tag{4.4.12}$$

for some matrix Q, its *i*-th column being $g(\vec{e}_i)$, with \vec{e}_i the *i*-th unit vector. This map is an automorphism if the action of Q is invertible modulo \sim , i.e., if it is a permutation of \mathcal{A} . Finally, this permutation shall be a symmetry if it conserves the spin of all the lines, up to complex conjugation in the anti-unitary case. We discuss this in some more detail below.

Anti-unitary symmetries.

A natural generalisation of proposition 4.3.2 reads

Proposition 4.4.1 A necessary condition for the Chern-Simons theory with matrix K to admit an anti-unitary symmetry is that there exists a pair of matrices $(Q, P) \in \mathbb{Z}^{n \times n} \times \mathbb{Z}^{n \times n}$ where P has even diagonal elements, and such that

$$PK - Q^t K^{-1} QK = 1_n \,. \tag{4.4.13}$$

Proof. We shall look for the most general permutation that satisfies the conditions (4.4.11).

As in the case of a single U(1) factor, any putative time-reversal operation is fixed once we know how the generators transform. The most general fusion endomorphism reads

$$\Gamma(\vec{\alpha}) = Q\vec{\alpha} \tag{4.4.14}$$

for some matrix Q, the *i*-th column of which represents the action of T on the unit vector in the *i*-th direction $\vec{e_i}$.

Imposing that the spin of \vec{e}_i is the opposite of that of $\mathsf{T}(\vec{e}_i)$, we get

$$\frac{1}{2}\vec{e_i}^t K^{-1}\vec{e_i} = -\frac{1}{2}\vec{e_i}^t Q^t K^{-1} Q\vec{e_i} + P_{ii}$$
(4.4.15)

for some integer P_{ii} . Similarly, imposing that T commutes with braiding, $B(\vec{e}_i, \vec{e}_j) = B(\mathsf{T}(\vec{e}_i), \mathsf{T}(\vec{e}_j))^*$, we get

$$\vec{e}_i^{\ t} K^{-1} \vec{e}_j = -\vec{e}_i^{\ t} Q^t K^{-1} Q \vec{e}_j + P_{ij} \tag{4.4.16}$$

for some integer P_{ij} . These two equations, in matrix form, take the form quoted in the proposition, as claimed. Note that if this equation is satisfied, then the spin of all the lines behaves as expected, and not only that of the generators:

$$h_{\mathsf{T}(\vec{\alpha})} = \frac{1}{2} \vec{\alpha}^t \left(Q^t K^{-1} Q \right) \vec{\alpha} = \frac{1}{2} \vec{\alpha}^t \left(-K^{-1} + P \right) \vec{\alpha},$$
(4.4.17)

which indeed equals $-h_{\vec{\alpha}}$ modulo 1.

We stress that, unlike in the case of a single U(1) factor, the argument in proposition 4.4.1 does not prove that any map $\vec{\alpha} \mapsto Q\vec{\alpha}$ with $PK - Q^tK^{-1}QK = 1_n$ represents a time-reversal operation, even though the conditions (4.4.11) are satisfied. One must also require Q to be a permutation, that is, invertible modulo K over the integers. This is a non-trivial condition that is not satisfied for every solution of $PK - Q^tK^{-1}QK = 1_n$. (In the 1 × 1 case, the equation $pk - q^2 = 1$ implies that gcd(k, q) = 1, and so any solution is invertible; this is no longer necessarily true in the $n \times n$ case: some solutions may fail to be invertible).

As in proposition 4.3.5, one can also examine the time-reversal invariance of $U(1)^n$ through a Lagrangian argument:

Proposition 4.4.2 A sufficient condition for the Chern-Simons theory described by the matrix K to admit an anti-unitary symmetry is that there exists a pair of matrices $(Q, P) \in \mathbb{Z}^{n \times n} \times \mathbb{Z}^{n \times n}$ where P has even diagonal elements, and such that

$$PK - Q^t K^{-1} QK = 1_n (4.4.18)$$

subject to the conditions

$$[K^{-1}QK, Q^t] = 0, \qquad [KP, Q] = 0.$$
(4.4.19)

(Note that if Q is normal and commutes with K, then these equations are automatically satisfied).

Proof. By solving for P in (4.4.18), and taking the transpose, it becomes clear that P is symmetric, and so it defines a (bosonic) abelian Chern-Simons theory. Take the Lagrangian with matrix $K \oplus -P$

$$4\pi \mathcal{L} = a^t K \mathrm{d}a - b^t P \mathrm{d}b \tag{4.4.20}$$

and perform the $\operatorname{GL}_{2n}(\mathbb{Z})$ transformation

$$\mathsf{T}: \begin{pmatrix} a \\ b \end{pmatrix} \mapsto \begin{pmatrix} Q^t & -P \\ K & -Q \end{pmatrix} \begin{pmatrix} a \\ b \end{pmatrix}$$
(4.4.21)

under which

$$\begin{pmatrix} K & 0 \\ 0 & -P \end{pmatrix} \mapsto \begin{pmatrix} -KPK + QKQ^t & [KP,Q] \\ [Q^t,PK] & PKP - Q^tPQ \end{pmatrix}$$

$$\equiv \begin{pmatrix} -K & 0 \\ 0 & P \end{pmatrix}.$$

$$(4.4.22)$$

The off-diagonal entries vanish by virtue of [KP, Q] = 0 and P being symmetric, and the equality for the diagonal entries follows from the assumptions in the proposition.

This proves that the theory is time-reversal invariant. The mapping of lines reads

$$(\vec{\alpha}, \vec{\beta}) \mapsto (Q\vec{\alpha} + K\vec{\beta}, -P\vec{\alpha} - Q^t\vec{\beta}).$$
 (4.4.23)

Finally, and thanks to the evenness of P, the action of T descends to a well-defined operation on the lines of $U(1)_k$:

$$(\vec{\alpha}, 0) \mapsto (Q\vec{\alpha}, -P\vec{\alpha}) \sim (Q\vec{\alpha}, 0) \tag{4.4.24}$$

as required.

Remark 4.4.1 It is easy to argue that the conditions in proposition 4.4.2 are $\operatorname{GL}_n(\mathbb{Z})$ invariant. Indeed, if we redefine our gauge fields according to a := Ga' for some $G \in \operatorname{GL}_n(\mathbb{Z})$, then the lines transform as $\vec{\alpha} = (G^{-1})^t \vec{\alpha}'$, and

$$K = (G^{-1})^{t} K' G^{-1}$$

$$Q = (G^{-1})^{t} Q' G^{t}$$

$$P = GP' G^{t}$$
(4.4.25)

which leaves the equations (4.4.18), (4.4.19) invariant. This was to be expected, inasmuch as a Chern-Simons theory depends on K modulo congruences. (Two K-matrices in the same congruence class have the same determinant; however, the converse is not true: there can multiple congruence classes with a given determinant. The number of congruence classes depends nontrivially on the value of the determinant.) Deciding whether the equation $PK - Q^t K^{-1}QK = 1_n$ is solvable for a given K is a rather non-trivial problem, unlike in the case of $U(1)_k$ (where it suffices to scan $q \in [0, k)$ for solutions; moreover, and thanks to proposition 4.7.1, deciding whether $k \in \mathbb{T}$ requires at most $\omega(k) \leq \frac{2\log k}{\log \log k}$ operations if given the prime divisors of k). We shall make no attempt at finding an efficient characterisation of the set of K-matrices that solve this equation. We will content ourselves with focusing specifically to the case where K is a 2×2 matrix. In particular, we will consider the following two families of K-matrices:

- Diagonal U(1)_{k1} × U(1)_{k2}, with matrix $K = \begin{pmatrix} k_1 & 0 \\ 0 & k_2 \end{pmatrix}$, and
- \mathbb{Z}_{k_1} twisted gauge theory at level k_2 , denoted by $(\mathbb{Z}_{k_1})_{k_2}$, with matrix $K = \begin{pmatrix} 0 & k_1 \\ k_1 & k_2 \end{pmatrix}$.

Remark 4.4.2 The theory $(\mathbb{Z}_{k_1})_{k_2}$ is also known as Dijkgraaf-Witten theory when k_2 is even [35]. It admits a Chern-Simons gauge theory realization [205, 273]. One can show that any 2×2 matrix K with $\det(K) = -n^2$ for some integer n can be brought into this form by a $\operatorname{GL}_2(\mathbb{Z})$ congruence transformation $G^t K G$ (see e.g. [274]). Furthermore, it is easy to show that $(\mathbb{Z}_{k_1})_{k_2} \sim (\mathbb{Z}_{k_1})_{k_2+2k_1}$, because the corresponding matrices are congruent⁹⁴.

We conjecture the following:

Conjecture 4.4.1 The diagonal theory $U(1)_{k_1} \times U(1)_{k_2}$:

- If $k_1k_2 > 0$,
 - Never time-reversal invariant if $k_1k_2 = 0 \mod 4$,
 - If $k_1k_2 = 2 \mod 4$, say, $k_1 = 2\tilde{k}_1$, then the theory is T-invariant if and only if $k_2 \in \mu(\tilde{k}_1)\mathbb{T}$, i.e., if $\mu(\tilde{k}_1) = \mu(k_2)$,
 - If k_1k_2 is odd, then the theory is T-invariant if and only if $k_2 \in \mu(k_1)\mathbb{T}$, i.e., if $\mu(k_1) = \mu(k_2)$

⁹⁴Given $(\mathbb{Z}_a)_b$, not all the theories in $b \in [0, 2a)$ are independent. For example, if a is odd, one has the duality of spin TQFTs $(\mathbb{Z}_a)_b \longleftrightarrow (\mathbb{Z}_a)_{b+a}$. This follows from the more general duality

$$(\mathbb{Z}_a)_b \times \mathrm{U}(1)_k \longleftrightarrow (\mathbb{Z}_a)_{b+a} \times \mathrm{U}(1)_k \tag{4.4.26}$$

which holds if and only if $a = 2^{\alpha}(2m+1)$ and $k = 2^{\alpha}(2n+1)$ for some integers α, m, n . The explicit change of variables is $G^t K_{a,b} G \equiv K_{a,a+b}$, where

$$G := \begin{pmatrix} -1 & m+n+2mn & 2n+1 \\ 0 & -1 & 0 \\ 0 & 2m+1 & 1 \end{pmatrix}, \quad K_{a,b} := \begin{pmatrix} 0 & a & 0 \\ a & b & 0 \\ 0 & 0 & k \end{pmatrix}.$$
(4.4.27)

- If $k_1k_2 < 0$,
 - If k_1 is odd, the theory is T -invariant if and only if $k_2 \in \mu(k_1)\mathbb{T}$, - If $k_1 = 2\tilde{k}_1$ is even, the theory is T -invariant if and only if $k_2 \in \mu(\tilde{k}_1)(\mathbb{T} \cup 2\mathbb{T})$.

Conjecture 4.4.2 The theory $(\mathbb{Z}_{k_1})_{k_2}$ is time-reversal invariant if and only if $k_2 \in \mu(k_1)\mathbb{Z}$.

Some of these claims are easy to prove. For example, if k_1 and k_2 are both even and positive, then the theory $U(1)_{k_1} \times U(1)_{k_2}$ is bosonic and has central charge +2, and so it cannot be time-reversal invariant. More generally, the conditions above can be seen to be necessary just by insisting that the generating lines $\vec{e_1}, \vec{e_2}$ have a line with opposite spin. Proving that they are also sufficient requires more work, but in principle does not seem out of reach: an approach similar to the one-dimensional case $U(1)_k$ should work. In any case, we checked that the conjecture is correct up to $|k_i| \leq 200$ in the diagonal case, and $|k_1| \leq 200$ and $k_2 \in [0, 2k_1)$ in the $(\mathbb{Z}_{k_1})_{k_2}$ gauge theory case. We stress that the diagonal theory can be be time-reversal invariant even when neither of the factors by itself is; naturally, this also holds for more general theories: a product may have more symmetries than its individual factors.

Note that if the conjecture above is true, then any odd non-Pythagorean prime factor of det(K) must appear an even number of times. In fact, it seems that this is true for any 2×2 matrix, whether it is of the forms above or not:

Conjecture 4.4.3 A necessary condition for the matrix $K \in \mathbb{Z}^{2 \times 2}$ to describe a time-reversal invariant theory is that $\lambda(\det(K)) \in \mathbb{T}$, where $\lambda(n)$ denotes the squarefree part of n.

We recall that a number is said to be squarefree if its prime decomposition contains no repeated factors. We have checked that this conjecture is true for all matrices with $|\det(K)| \leq 500$. (For completeness, we remark that $\lambda(n) \in \mathbb{T}$ if and only if n can be expressed as the sum of two perfect squares, not necessarily coprime).

It also appears that all primitive matrices with det(K) > 0, if time-reversal invariant, have $T^2 = C$, as in the 1 × 1 case:

Conjecture 4.4.4 If $K \in \mathbb{Z}^{2 \times 2}$ is positive definite and primitive (i.e. with $gcd(K_{ij}) = 1$ for all i, j), then $T^2 = C$.

We checked that this is true for all matrices with $det(K) \leq 400$.

Unitary Symmetries.

An essentially identical philosophy allows us to study unitary symmetries rather than antiunitary ones. Following an argument equivalent to that of proposition 4.4.1 it is easy to prove that **Proposition 4.4.3** Given some $K \in \mathbb{Z}^{n \times n}$, the most general unitary symmetry (i.e., a permutation subject to (4.4.10)) is of the form

$$\mathsf{U}\colon\vec{\alpha}\mapsto Q\vec{\alpha}\tag{4.4.28}$$

for some $Q \in \mathbb{Z}^{n \times n}$, invertible over \mathcal{A} , the *i*-th column of which represents $U(\vec{e_i})$, the action of the unitary symmetry on the unit vector in the *i*-th direction. Invariance of spin and braiding requires

$$PK + Q^t K^{-1} QK = 1_n (4.4.29)$$

for some integral matrix P with even diagonal components.

There is always the trivial solution $Q = 1_n$, which leaves all the lines invariant, and its negative $Q = -1_n$, which corresponds to charge-conjugation $\mathsf{C} \colon \vec{\alpha} \mapsto -\vec{\alpha}$. Any other solution Q (invertible modulo K) will correspond to some non-trivial unitary zero-form symmetry of the system.

We can finally write down the general expression for the group of symmetries of a given theory:

Proposition 4.4.4 Given an arbitrary abelian TQFT realized as a $U(1)^n$ Chern-Simons theory with matrix of levels K, the group of (unitary and anti-unitary) zero-form symmetries can be expressed as

$$\operatorname{Aut}(K) \cong \{ Q \in \mathbb{Z}^{n \times n} \mid PK \pm Q^t K^{-1} QK = 1_n \text{ for some } P \in \mathbb{Z}^{n \times n} \} / \sim \qquad (4.4.30)$$

where P is required to have even diagonal components, Q is required to be invertible modulo K, and \sim denotes the equivalence

$$Q \sim Q' \quad \iff \quad Q\vec{e}_i \sim Q'\vec{e}_i, \quad i = 1, 2, \dots, n \quad (4.4.31)$$

where the last \sim denotes equivalence in \mathcal{A} (cf. (4.4.6)). The subgroup of unitary symmetries is given by

$$\operatorname{Aut}_{U}(K) \cong \{ Q \in \mathbb{Z}^{n \times n} \,|\, PK + Q^{t}K^{-1}QK = 1_{n} \text{ for some } P \in \mathbb{Z}^{n \times n} \} / \sim \qquad (4.4.32)$$

with the same restrictions as before. A given symmetry [Q] is quantum if and only if $P \neq 0$ for all $Q \in [Q]$.

Remark 4.4.3 Here we are making a slight abuse of notation in order to simplify the presentation: strictly speaking, if a given matrix Q satisfies both $PK + Q^t K^{-1}QK = 1_n$ and $PK - Q^t K^{-1}QK = 1_n$ (possibly with different P's), they are *different* symmetries, and so distinct elements of Aut(K). The same permutation on the anyons constitutes both a unitary,

and an anti-unitary symmetry of the system. In other words, the group of symmetries is the *disjoint union* of the set of anti-unitary symmetries, and the set of unitary symmetries. In order to implement this, one should think of $\operatorname{Aut}(K)$ as pairs (Q, σ) , where $\sigma = \pm 1$ keeps track of whether a given permutation is unitary or anti-unitary, and one must add the condition $\sigma(Q) = \sigma(Q')$ to the equivalence relation \sim .

We propose the following conjecture:

Conjecture 4.4.5 The group of unitary symmetries of $(\mathbb{Z}_k)_0$ is multiplicative in k:

$$\operatorname{Aut}_U((\mathbb{Z}_{ab})_0) = \operatorname{Aut}_U((\mathbb{Z}_a)_0) \times \operatorname{Aut}_U((\mathbb{Z}_b)_0), \qquad \operatorname{gcd}(a,b) = 1.$$
(4.4.33)

Furthermore, for prime powers, it is given by

$$\operatorname{Aut}_{U}((\mathbb{Z}_{\pi^{n}})_{0}) = D_{2\phi(\pi^{n})}, \qquad \operatorname{Aut}_{U}((\mathbb{Z}_{2^{n}})_{0}) = \mathbb{Z}_{2} \times D_{\phi(2^{n})}, \qquad (4.4.34)$$

where D_{2n} denotes the dihedral group of order 2n. The full group of symmetries, including anti-unitary transformations, is a \mathbb{Z}_2 extension of the unitary sub-group:

$$\operatorname{Aut}((\mathbb{Z}_k)_0) = \mathbb{Z}_2 \ltimes \operatorname{Aut}_U((\mathbb{Z}_k)_0).$$
(4.4.35)

Remark 4.4.4 Note the similarity of this group and $\mathbb{Z}_k^{\times} := \operatorname{Aut}(\mathbb{Z}_k)$, the multiplicative group of integers modulo k. As per a classic result of Gauss, this latter group is also multiplicative, and given by $\operatorname{Aut}(\mathbb{Z}_{\pi^n}) = \mathbb{Z}_{\phi(\pi^n)}$ and $\operatorname{Aut}(\mathbb{Z}_{2^n}) = \mathbb{Z}_2 \times \mathbb{Z}_{\phi(2^n)/2}$. For $k = \pi$ a prime, the group $\operatorname{Aut}_U((\mathbb{Z}_{\pi})_0) = D_{2(\pi-1)}$ has been computed in [275].

We next illustrate how to compute $Aut(\star)$ step by step, through a couple of examples. More examples are worked out, to a lesser degree of detail, in section 4.5.

Consider the theory $(\mathbb{Z}_{k_1})_{k_2}$, whose matrix is

$$K = \begin{pmatrix} 0 & k_1 \\ k_1 & k_2 \end{pmatrix} \tag{4.4.36}$$

where we can take without loss of generality $k_1 > 0$ and $0 \le k_2 < 2k_1$. The theory is bosonic if k_2 is even, and spin otherwise. In the first case, the lines are of the form $(\alpha, \beta) \in \mathbb{Z}_{k_1} \times \mathbb{Z}_{k_1}$, and in the second case $(\alpha, \beta) \in \mathbb{Z}_{2k_1} \times \mathbb{Z}_{k_1}$. The spin of an arbitrary line is

$$h_{\alpha,\beta} = \frac{\alpha\beta}{k_1} - \frac{k_2\alpha^2}{2k_1^2}$$
(4.4.37)

A common notation for the lines of $(\mathbb{Z}_{k_1})_{k_2}$ is $\mathbf{e}_i = (i, 0)$, called the *electric* lines, and $\mathbf{m}_j = (0, j)$, called the *magnetic* lines. Their product is $\mathbf{e}_i \mathbf{m}_j = (i, j)$. There are $i \in [0, k_1)$ electric lines if k_2 is even, and $i \in [0, 2k_1)$ lines of odd; and $j \in [0, k_1)$ magnetic lines. (The

electric line $\mathbf{e}_i = (i, 0)$ should not be confused with the unit vector $\vec{e}_i = (0, 0, \dots, 0, 1, 0, \dots, 0)$ in the *i*-th direction). The line $\mathbf{e}_{k_1} \equiv \psi$ is the transparent fermion, and so $\mathbf{e}_{i+k_1} \equiv \mathbf{e}_i \times \psi$.

Take for example $(\mathbb{Z}_3)_0$. A basis of lines is

$$\mathcal{A} = \{ (0,0), (1,0), (2,0), (0,1), (1,1), (2,1), (0,2), (1,2), (2,2) \}$$

= $\{ \mathbf{1}, \mathbf{e}_1, \mathbf{e}_2, \mathbf{m}_1, \mathbf{e}_1\mathbf{m}_1, \mathbf{e}_2\mathbf{m}_1, \mathbf{m}_2, \mathbf{e}_1\mathbf{m}_2, \mathbf{e}_2\mathbf{m}_2 \}$ (4.4.38)

with spins

$$h = \{0, 0, 0, 0, 1/3, 2/3, 0, 2/3, 1/3\}.$$
(4.4.39)

Any endomorphism of the fusion algebra is of the form

$$g: \begin{pmatrix} \alpha \\ \beta \end{pmatrix} \mapsto Q \begin{pmatrix} \alpha \\ \beta \end{pmatrix}, \qquad Q = (g(\vec{e_1}) \ g(\vec{e_2})).$$
 (4.4.40)

As \vec{e}_i both have vanishing spin, the condition $g: h \mapsto \pm h$ requires

$$g(\vec{e}_i) \in \{(1,0), (2,0), (0,1), (0,2)\}$$
(4.4.41)

and so there are $4^2 - 4 = 12$ candidates for the matrices Q:

By explicit computation, one may check that the only endomorphisms that are actually automorphisms (i.e., the only matrices Q that are invertible modulo K) are

$$\begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix}, \begin{pmatrix} 2 & 0 \\ 0 & 2 \end{pmatrix}, \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}, \begin{pmatrix} 0 & 2 \\ 2 & 0 \end{pmatrix}$$

$$\begin{pmatrix} 1 & 0 \\ 0 & 2 \end{pmatrix}, \begin{pmatrix} 2 & 0 \\ 0 & 1 \end{pmatrix}, \begin{pmatrix} 0 & 2 \\ 1 & 0 \end{pmatrix}, \begin{pmatrix} 0 & 1 \\ 2 & 0 \end{pmatrix}$$

$$(4.4.43)$$

and that the first line satisfies $K^{-1} - Q^t K^{-1}Q = P$, and the second one $K^{-1} + Q^t K^{-1}Q = P$, for some integral-valued matrix P. Therefore, the former generate unitary symmetries, and the latter anti-unitary symmetries.

One may check that the two matrices

$$\mathsf{T}: \begin{pmatrix} 0 & 2 \\ 1 & 0 \end{pmatrix}, \qquad \mathsf{U}: \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} \tag{4.4.44}$$

generate the whole group of symmetries, and they satisfy

$$\mathsf{T}^4 = \mathsf{U}^2 = (\mathsf{T}\mathsf{U})^2 = 1 \tag{4.4.45}$$

and so the group of symmetries is dihedral:

$$\operatorname{Aut}((\mathbb{Z}_3)_0) = D_8 = \langle \mathsf{T}, \mathsf{U} \rangle.$$
(4.4.46)

Similarly, the pair of matrices $\mathsf{C}:=\mathsf{T}^2$ and U generate the subgroup of unitary symmetries, and they satisfy

$$\mathsf{C}^2 = \mathsf{U}^2 = 1 \tag{4.4.47}$$

and so the latter is cyclic:

$$\operatorname{Aut}_{U}((\mathbb{Z}_{3})_{0}) = \mathbb{Z}_{2}^{2} = \langle \mathsf{C}, \mathsf{U} \rangle.$$

$$(4.4.48)$$

Consider now what happens when we turn on a non-trivial twisting, say, $(\mathbb{Z}_3)_2$. The spin of the lines is modified into

$$h = \{0, 8/9, 5/9, 0, 2/9, 2/9, 0, 5/9, 8/9\}.$$
(4.4.49)

As we can see, there is no line with spin $-8/9 = 1/9 \mod 1$, and so $\vec{e_1}$ has no partner under time-reversal: the theory does not admit anti-unitary symmetries. Therefore the symmetries, if any, must be unitary, and so they must fix the spin; thus, the condition $h \mapsto +h$ requires

$$U(\vec{e}_1) \in \{(1,0), (2,2)\} U(\vec{e}_2) \in \{(0,1), (0,2)\}$$
(4.4.50)

from where it follows that all the candidates for Q are

One may check that all these matrices are invertible, but the only two that satisfy $K^{-1} - Q^t K^{-1}Q = P$ for some integral-valued matrix P are

$$\begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix}, \begin{pmatrix} 2 & 0 \\ 2 & 2 \end{pmatrix}. \tag{4.4.52}$$

Finally, the second matrix is easily seen to implement charge conjugation C, and so it squares to the identity. In other words, the group of symmetries of the system is

$$\operatorname{Aut}((\mathbb{Z}_3)_2) = \operatorname{Aut}_U((\mathbb{Z}_3)_2) = \mathbb{Z}_2 = \langle \mathsf{C} \rangle.$$
(4.4.53)

By an identical argument one may calculate the group of symmetries of an arbitrary abelian theory. In table 4.2 we include the group of symmetries of $(\mathbb{Z}_{k_1})_{k_2}$ for small values of the levels.

k	$\operatorname{Aut}((\mathbb{Z}_k)_0)$	$\operatorname{Aut}_U((\mathbb{Z}_k)_0)$	$\operatorname{Aut}((\mathbb{Z}_k)_{\mu(k)})$	$\operatorname{Aut}_U((\mathbb{Z}_k)_{\mu(k)})$
2	\mathbb{Z}_2^2	\mathbb{Z}_2	\mathbb{Z}_2	0
3	D_8	\mathbb{Z}_2^2	D_8	\mathbb{Z}_2^2
4	D_8	\mathbb{Z}_2^2	D_8	\mathbb{Z}_2^2
5	$\mathbb{Z}_4 \circ D_8$	D_8	\mathbb{Z}_4	\mathbb{Z}_2
6	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_2^3	D_8	\mathbb{Z}_2^2
7	$\mathbb{Z}_3 \rtimes D_8$	D_{12}	$\mathbb{Z}_3 \rtimes D_8$	D_{12}
8	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_2^3	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_2^3
9	$\mathbb{Z}_3 \rtimes D_8$	D_{12}	$\mathbb{Z}_3 \rtimes D_8$	D_{12}
10	$\mathbb{Z}_2 \times \mathbb{Z}_4 \circ D_8$	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_4	\mathbb{Z}_2
11	$\mathbb{Z}_5 \rtimes D_8$	D_{20}	$\mathbb{Z}_5 \rtimes D_8$	D_{20}
12	$\mathbb{Z}_2^2\wr\mathbb{Z}_2$	\mathbb{Z}_2^4	$\mathbb{Z}_2^2\wr\mathbb{Z}_2$	\mathbb{Z}_2^4
13	$\mathbb{Z}_4 \circ D_{24}$	D_{24}	\mathbb{Z}_4	\mathbb{Z}_2
14	$\mathbb{Z}_2 \times \mathbb{Z}_3 \rtimes D_8$	$\mathbb{Z}_2^2 \times S_3$	$\mathbb{Z}_3 \rtimes D_8$	D_{12}
15	$D_8 \rtimes_5 D_8$	$\mathbb{Z}_2^2 \times D_8$	$\mathbb{Z}_2^2 \rtimes \mathbb{Z}_4$	\mathbb{Z}_2^3
16	$\mathbb{Z}_4 \rtimes D_8$	$\mathbb{Z}_2 \times D_8$	$\mathbb{Z}_4 \rtimes D_8$	$\mathbb{Z}_2 \times D_8$
17	$\mathbb{Z}_4 \circ D_{32}$	D_{32}	\mathbb{Z}_4	\mathbb{Z}_2
18	$\mathbb{Z}_2 \times \mathbb{Z}_3 \rtimes D_8$	$\mathbb{Z}_2^2 \times S_3$	$\mathbb{Z}_3 \rtimes D_8$	D_{12}
19	$\mathbb{Z}_9 \rtimes D_8$	D_{36}	$\mathbb{Z}_9 \rtimes D_8$	D_{36}
20	$D_8 \rtimes_5 D_8$	$\mathbb{Z}_2^2 \times D_8$	$\mathbb{Z}_2^2 \rtimes \mathbb{Z}_4$	\mathbb{Z}_2^3
21	$\mathbb{Z}_2^3 \rtimes_2 D_{12}$	$\mathbb{Z}_2^3 \times S_3$	$\mathbb{Z}_2^3 \rtimes_2 D_{12}$	$\mathbb{Z}_2^3 \times S_3$
22	$\mathbb{Z}_2 \times \mathbb{Z}_5 \rtimes D_8$	$\mathbb{Z}_2^2 \times D_{10}$	$\mathbb{Z}_5 \rtimes D_8$	D_{20}
23	$\mathbb{Z}_{11} \rtimes D_8$	D_{44}	$\mathbb{Z}_{11} \rtimes D_8$	D_{44}
24	$\mathbb{Z}_2 imes \mathbb{Z}_2^2 \wr \mathbb{Z}_2$	\mathbb{Z}_2^5	$\mathbb{Z}_2 imes \mathbb{Z}_2^2 \wr \mathbb{Z}_2$	\mathbb{Z}_2^5
25	$\mathbb{Z}_4 \circ D_{40}$	D_{40}	\mathbb{Z}_4	\mathbb{Z}_2
26	$\mathbb{Z}_2 \times \mathbb{Z}_4 \circ D_{24}$	$\mathbb{Z}_2 \times D_{24}$	\mathbb{Z}_4	\mathbb{Z}_2
27	$\mathbb{Z}_9 \rtimes D_8$	D_{36}	$\mathbb{Z}_9 \rtimes D_8$	D_{36}

Table 4.2: The group of symmetries of $(\mathbb{Z}_{k_1})_{k_2}$, denoted by Aut(\star), and its unitary subgroup Aut_U(\star), for $k_1 \in [0, 27]$ and $k_2 = 0, \mu(k_1)$. For $k_2 \not\propto \mu(k_1)$ there are no anti-unitary symmetries. (See section 4.6 for basic definitions).

Similarly, in tables 4.3 and 4.4 we include the group of symmetries of the diagonal theory $U(1)_{k_1} \times U(1)_{k_2}$.

	2	3	4	5	6	7	8	9	10	11	12	13
2	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{Z}_4	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{Z}_2^2	\mathbb{Z}_2	\mathbb{Z}_2^2	\mathbb{Z}_4
3	• ,	SD_{10}	$_6\mathbb{Z}_2^2$	\mathbb{Z}_2^2	D_8	\mathbb{Z}_2^2	\mathbb{Z}_2^3	D_{12}	\mathbb{Z}_2^2	\mathbb{Z}_2^2	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_2^2
4	•	•	D_8	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^3	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^3	\mathbb{Z}_2^2
5	•		• 7	$\mathbb{Z}_4 \circ D$	$_8 \mathbb{Z}_2^2$	\mathbb{Z}_2^2	\mathbb{Z}_2^3	\mathbb{Z}_2^2	$\mathbb{Z}_4 \times S_3$	\mathbb{Z}_2^2	\mathbb{Z}_2^3	$\mathbb{Z}_4 \times \mathbb{Z}_2$
6	•	•	•	•	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_2^2	\mathbb{Z}_2^2	D_{12}	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^3	\mathbb{Z}_2^2
7	•	•	•	•	•	SD_{32}	$_2$ \mathbb{Z}_2^3	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^3	\mathbb{Z}_2^2
8	•		•	•			$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_2^3	\mathbb{Z}_2^2	\mathbb{Z}_2^3	\mathbb{Z}_2^4	\mathbb{Z}_2^3
9	•	•	•	•	•	•	· 7	$\mathbb{Z}_{24} \rtimes \mathbb{Z}$	\mathbb{Z}_2^2	\mathbb{Z}_2^2	$\mathbb{Z}_2^2 \times S_3$	\mathbb{Z}_2^2
10	•	•	•	•				•	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_2^2	\mathbb{Z}_2^3	$\mathbb{Z}_4 \times \mathbb{Z}_2$
11	•	•	•	•				•	• 2	$\mathbb{Z}_{24} \rtimes \mathbb{Z}_2$	\mathbb{Z}_2^3	\mathbb{Z}_2^2
12	•	•	•	•				•		•	$D_8 \times D_8$	\mathbb{Z}_2^3
13	•	•	•	•		•		•		•	•	$\mathbb{Z}_4 \circ D_{24}$

Table 4.3: The group of symmetries of the diagonal theory $K = \text{diag}(k_1, k_2)$, to wit, Aut $(U(1)_{k_1} \times U(1)_{k_2})$, for small values of k_i .

	2	3	4	5	6	7	8	9	10	11	12	13
2	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{Z}_2^2	\mathbb{Z}_2	\mathbb{Z}_2^2	\mathbb{Z}_2
3		D_8	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^3	D_{12}	\mathbb{Z}_2^2	\mathbb{Z}_2^2	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_2^2
4			D_8	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^3	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^3	\mathbb{Z}_2^2
5			•	D_8	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^3	\mathbb{Z}_2^2	D_{12}	\mathbb{Z}_2^2	\mathbb{Z}_2^3	\mathbb{Z}_2^2
6	•	•	•	•	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_2^2	\mathbb{Z}_2^2	D_{12}	\mathbb{Z}_2^2	\mathbb{Z}_2^2	\mathbb{Z}_2^3	\mathbb{Z}_2^2
7			•	•	•	D_{16}	\mathbb{Z}_2^3	\mathbb{Z}_2^2	\mathbb{Z}_2^2	$\mathbb{Z}_2 \times \mathbb{Z}_2$	\mathbb{Z}_2^3	\mathbb{Z}_2^2
8	•	•	•	•	•	•	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_2^3	\mathbb{Z}_2^2	\mathbb{Z}_2^3	\mathbb{Z}_2^4	\mathbb{Z}_2^3
9	•	•	•	•	•	•	•	D_{24}	\mathbb{Z}_2^2	\mathbb{Z}_2^2	$\mathbb{Z}_2^2 \times S_3$	\mathbb{Z}_2^2
10		•	•	•	•	•	•	•	$\mathbb{Z}_2 \times D_8$	\mathbb{Z}_2^2	\mathbb{Z}_2^3	\mathbb{Z}_2^2
11		•	•	•	•	•	•	•	•	D_{24}	\mathbb{Z}_2^3	\mathbb{Z}_2^2
12	•		•	•	•	•	•	•		•	$D_8 \times D_8$	\mathbb{Z}_2^3
13			•	•	•	•	•	•		•		D_{24}

Table 4.4: The group of unitary symmetries of the diagonal theory $K = \text{diag}(k_1, k_2)$, to wit, $\text{Aut}_U(\text{U}(1)_{k_1} \times \text{U}(1)_{k_2})$, for small values of k_i .

4.4.2 Dualities of $U(1)^n$

A straightforward extension of the formalism so far can be used to diagnose dualities between different abelian Chern-Simons theories. Given two systems described by matrices K_1, K_2 , $n_1 \times n_1$ and $n_2 \times n_2$, respectively, the theories shall describe the same TQFT if they give rise to isomorphic anyon data \mathcal{A}_i, θ_i . This corresponds to a bijection $f: \mathcal{A}_1 \leftrightarrow \mathcal{A}_2$ that preserves fusion and spin. If we write the anyons \mathcal{A}_i as n_i -tuples of integral charges $\vec{\alpha}_i$, then preservation of fusion requires f to act linearly, say, $\vec{\alpha}_1 = Q\vec{\alpha}_2$, with Q an $n_1 \times n_2$ integral matrix that has a left inverse modulo K_2 (equivalently, $\vec{\alpha}_2 = \tilde{Q}\vec{\alpha}_1$ with \tilde{Q} an $n_2 \times n_1$ integral matrix with left inverse modulo K_1). Preservation of spin requires the existence of an $n_2 \times n_2$ integral matrix P, with even diagonal components, such that

$$Q^{t}K_{1}^{-1}Q - K_{2}^{-1} = P. (4.4.54)$$

The theories described by K_1, K_2 are dual if and only if such matrices Q, P exist:

$$K_1 \quad \longleftrightarrow \quad K_2 \,. \tag{4.4.55}$$

In light of this discussion, we can summarize the content of the previous sections as follows: a theory with matrix K has a unitary symmetry if and only if there is a self-duality $K \leftrightarrow K$, and an anti-unitary symmetry if and only if there is a duality $K \leftrightarrow -K$.

For example, it is a well-known fact that $U(1)_{-8}$ is level-rank dual (as bosonic TQFTs) to $SU(8)_1$. This latter theory can be represented as an abelian Chern-Simons theory with *K*-matrix equal to the Cartan matrix of SU(8). In our terminology, this duality is implemented, for example, via the matrix Q = (0, 0, 1, 0, 0, 0, 0), which indeed satisfies (4.4.54) with $P \equiv (2)$.

Other interesting examples of dual abelian theories can be found in twisted gauge theories $(\mathbb{Z}_a)_b$. Recall that if a is odd, then $(\mathbb{Z}_a)_b \leftrightarrow (\mathbb{Z}_a)_{a+b}$, which is already clear at the Lagrangian level (see footnote 94). There are extra dualities that go beyond this trivial one, for example $(\mathbb{Z}_7)_2 \leftrightarrow (\mathbb{Z}_7)_4$ and $(\mathbb{Z}_7)_3 \leftrightarrow (\mathbb{Z}_7)_5$. One can easily check these dualities by finding a suitable Q, P in (4.4.54).

Many more examples of (trivial and non-trivial) dualities between abelian theories can be exhibited. In contrast to the non-abelian case, abelian TQFTs enjoy infinitely-many dualities. For a given finite abelian group \mathcal{A} and a quadratic form θ on it, there are infinitely many integral matrices, of varying dimension, that generate the pair (\mathcal{A}, θ) , so all these matrices are dual. Trivially dual theories can be obtained by looking directly at the Lagrangian: matrices related by GL(\mathbb{Z})-conjugation give rise to the same dynamics, $GK_1G^t = K_2$. Non-trivially dual theories, which are not dual at the Lagrangian level, require the generalized condition $Q^t K_1^{-1}Q - K_2^{-1} = P$, which allows for matrices of different dimension. In any case, fixing K_1 , one can find infinitely-many matrices K_2 that are dual to it, just by varying G or P, Qin these equations.

4.5 Examples

Finally, we discuss some illustrative examples. To avoid repetition, we typically include a theory only if it incorporates a new feature that was not present in the previous examples. We begin by the case of a single abelian factor, $U(1)_k$.

Example 4.5.1 (k = 2) We have $\varpi(2) = 0$, and so the system has no unitary symmetries. As the system is bosonic, there are no anti-unitary symmetries either.

One may regard the system as a spin TQFT, in which case it is usually known as the semion-fermion theory [108, 111]. The system now admits one anti-unitary symmetry, which can be found by solving $2p - q^2 = 1$, whose only solution in the range $q \in [0, 2)$ is q = 1. This means that the permutation is $s \leftrightarrow s \times \psi$, as is well-known.

We thus have

$$\operatorname{Aut}(\mathrm{U}(1)_2) = \operatorname{Aut}_U(\mathrm{U}(1)_2) = 0$$
$$\operatorname{Aut}(\mathrm{U}(1)_2 \times \{\mathbf{1}, \psi\}) = \mathbb{Z}_2 = \langle \mathsf{T} \rangle$$
$$\operatorname{Aut}_U(\mathrm{U}(1)_2 \times \{\mathbf{1}, \psi\}) = 0.$$
(4.5.1)

The integer k = 2 is Pell, and so the time-reversal permutation above is a symmetry of the Lagrangian (provided by $\{\mathbf{1}, \psi\}$ we mean U(1)₋₁ rather than U(1)₊₁).

Example 4.5.2 (k = 3) We have $\varpi(3) = 1$, and so the system only has one unitary symmetry: charge conjugation. This is a Lagrangian symmetry.

Similarly, $3 \neq 1 \mod 4$, and so the system is not time-reversal invariant. We thus have

$$\operatorname{Aut}(\operatorname{U}(1)_3) = \operatorname{Aut}_U(\operatorname{U}(1)_3) = \mathbb{Z}_2 = \langle \mathsf{C} \rangle.$$
(4.5.2)

Example 4.5.3 (k = 5) We have $\varpi(5) = 1$, and so the system only has one unitary symmetry: charge conjugation. This is a Lagrangian symmetry.

The level satisfies $5 = 1 \mod 4$, and so the system is time-reversal invariant. The permutation can be found using equation (4.7.2): $q = \frac{5-1}{2}! + 5 = 7$ (there is a second solution, which differs by a sign: $q = -7 = 3 \mod 10$). The explicit map of lines is

We thus have

$$\operatorname{Aut}(\mathrm{U}(1)_5) = \mathbb{Z}_4 = \langle \mathsf{T} \rangle$$
$$\operatorname{Aut}_U(\mathrm{U}(1)_5) = \mathbb{Z}_2 = \langle \mathsf{C} \rangle.$$
(4.5.4)

The integer k = 5 is Pell, and so the time-reversal permutation above is a symmetry of the Lagrangian once we include the gravitational counterterm (but not without it).

Example 4.5.4 (k = 8) We have $\varpi(8) = 1$, and so the system only has one unitary symmetry: charge conjugation. This is a Lagrangian symmetry.

The system is bosonic, and so it is not time-reversal invariant. One may regard the system as a spin TQFT, but $8 = 0 \mod 4$, and so it is not time-reversal invariant either.

As a spin TQFT, one has $\varpi(4) + 1 = 2$, and so the system has three unitary symmetries: charge-conjugation and multiplication by ± 3 . The latter are not Lagrangian symmetries.

We thus have

$$\operatorname{Aut}(\mathrm{U}(1)_8) = \operatorname{Aut}_U(\mathrm{U}(1)_8) = \mathbb{Z}_2 = \langle \mathsf{C} \rangle$$

$$\operatorname{Aut}(\mathrm{U}(1)_8 \times \{\mathbf{1}, \psi\}) = \operatorname{Aut}_U(\mathrm{U}(1)_8 \times \{\mathbf{1}, \psi\}) = \mathbb{Z}_2 \times \mathbb{Z}_2 = \langle \mathsf{C}, \mathsf{U} \rangle.$$
(4.5.5)

where U is either of $(\alpha, \beta) \mapsto (\pm 3\alpha, \alpha + \beta)$ (the other sign being CU).

Example 4.5.5 (k = 12) We have $\varpi(12) = 2$, and so the system has three unitary symmetries: charge conjugation and multiplication by ± 5 . The latter are not Lagrangian symmetries.

The system is bosonic, and so it is not time-reversal invariant. One may regard the system as a spin TQFT, but $12 = 0 \mod 4$, and so it is not time-reversal invariant either.

As a spin TQFT, one has $\varpi(6) + 1 = 2$, and so the unitary symmetries are the same as in the bosonic case. They are not Lagrangian symmetries either.

We thus have

$$\operatorname{Aut}(\operatorname{U}(1)_{12}) = \operatorname{Aut}_U(\operatorname{U}(1)_{12}) = \mathbb{Z}_2 \times \mathbb{Z}_2 = \langle \mathsf{C}, \mathsf{U} \rangle$$

$$\operatorname{Aut}(\operatorname{U}(1)_{12} \times \{\mathbf{1}, \psi\}) = \operatorname{Aut}_U(\operatorname{U}(1)_{12} \times \{\mathbf{1}, \psi\}) = \mathbb{Z}_2 \times \mathbb{Z}_2 = \langle \mathsf{C}, \mathsf{U} \rangle.$$
(4.5.6)

where U denotes multiplication by either of ± 5 (the other sign being CU), while fixing the local fermion, if any.

Example 4.5.6 (k = 15) We have $\varpi(15) = 2$, and so the system has three unitary symmetries: charge conjugation and multiplication by ± 11 . The latter are not Lagrangian symmetries.

The level can be factored as $15 = 3 \cdot 5$, and $3 \neq 1 \mod 4$, and so the system is not time-reversal invariant.

We thus have

$$\operatorname{Aut}(\operatorname{U}(1)_{15}) = \operatorname{Aut}_U(\operatorname{U}(1)_{15}) = \mathbb{Z}_2 \times \mathbb{Z}_2 = \langle \mathsf{C}, \mathsf{U} \rangle.$$

$$(4.5.7)$$

where U denotes multiplication by either of ± 11 (the other sign being CU).

Example 4.5.7 (k = 24) We have $\varpi(24) = 2$, and so the system has three unitary symmetries: charge conjugation and multiplication by ± 7 . The latter are not Lagrangian symmetries.

The system is bosonic, and so it is not time-reversal invariant. One may regard the system as a spin TQFT, but $24 = 0 \mod 4$, and so it is not time-reversal invariant either.

As a spin TQFT, one has $\varpi(12) + 1 = 3$, and so the number of unitary symmetries is doubled. The new symmetries, those that mix the bosonic lines with the transparent fermions, are generated by multiplication by 13.

We thus have

$$\operatorname{Aut}(\operatorname{U}(1)_{24}) = \operatorname{Aut}_U(\operatorname{U}(1)_{24}) = \mathbb{Z}_2 \times \mathbb{Z}_2 = \langle \mathsf{C}, \mathsf{U} \rangle$$

$$\operatorname{Aut}(\operatorname{U}(1)_{24} \times \{\mathbf{1}, \psi\}) = \operatorname{Aut}_U(\operatorname{U}(1)_{24} \times \{\mathbf{1}, \psi\}) = \mathbb{Z}_2^3 = \langle \mathsf{C}, \mathsf{U}, \mathsf{U}' \rangle.$$
(4.5.8)

where U denotes multiplication by either of ± 7 (the other sign being CU) while fixing the local fermion, if any, and U' denotes multiplication by 13, while mixing the local fermion.

Example 4.5.8 (k = 25) We have $\varpi(25) = 1$, and so the system only has one unitary symmetry: charge conjugation. This is a Lagrangian symmetry.

The level can be factored as $25 = 5^2$, and $5 = 1 \mod 4$, which means that the system is time-reversal invariant. In order to find the solution to $q^2 = -1 \mod 25$ one may use Hensel lifting (4.7.4): the solutions modulo 5 are ± 3 , and so the solutions modulo 5^2 are $\pm 3 \mp (3^2 + 1) = \pm 7$. The explicit map of lines is

We thus have

$$\operatorname{Aut}(\mathrm{U}(1)_{25}) = \mathbb{Z}_4 = \langle \mathsf{T} \rangle$$

$$\operatorname{Aut}_U(\mathrm{U}(1)_{25}) = \mathbb{Z}_2 = \langle \mathsf{C} \rangle.$$
(4.5.10)

The integer k = 25 is not Pell, and so the time-reversal permutation above is not a symmetry of the Lagrangian, not even if we include the gravitational counterterm.

Example 4.5.9 (k = 65) We have $\varpi(65) = 2$, and so the system has three unitary symmetries: charge conjugation and multiplication by ± 51 . The latter are not Lagrangian symmetries.

The level can be factored as $65 = 5 \cdot 13$, and $5 = 13 = 1 \mod 4$, which means that the system is time-reversal invariant. In order to find the solution to $q^2 = -1 \mod 65$ one may use the Chinese Remainder Theorem (cf. the discussion below (4.7.4)). The solutions modulo 5 are $q = \pm 3$, and the solutions modulo 13 are $q = \pm 5$. Take for example the solution with $q = 3 \mod 5$ and $q = 5 \mod 13$; then, using the Euclidean algorithm, we find $5 \cdot 8 + 13 \cdot (-3) = 1$, which means that $q = 3 \cdot 13 \cdot (-3) + 5 \cdot 5 \cdot 8 = 83 \mod 65$. Similarly, taking the solution with $q = -3 \mod 5$ and $q = 5 \mod 13$ leads to $q = 57 \mod 65$. All in all, the solutions of $q^2 = -1 \mod 65$ are $q = \pm 47, \pm 57$. The explicit map of lines is

We thus have

$$\operatorname{Aut}(\mathrm{U}(1)_{65}) = \mathbb{Z}_4 \times \mathbb{Z}_2 = \langle \mathsf{T}, \mathsf{U} \rangle$$
$$\operatorname{Aut}_U(\mathrm{U}(1)_{65}) = \mathbb{Z}_2 \times \mathbb{Z}_2 = \langle \mathsf{C}, \mathsf{U} \rangle.$$
(4.5.12)

where T denotes either of T_1 , T_2 (the other one being UT), and U denotes multiplication by either of ± 51 (the other sign being T^2U).

The integer k = 65 is Pell, and so the permutation above is a symmetry of the Lagrangian once we include the gravitational counterterm (but not without it).

We now move on to 2×2 matrices. We denote by [a, b, c] the equivalence class (with respect to congruence) of all matrices of which $\begin{pmatrix} a & b \\ b & c \end{pmatrix}$ is a representative. We begin with positive-definite K, and order them by det(K). (We recall that there can be more than one congruence class with a given value of det(K)).

Example 4.5.10 (det(K) = 2) The first non-trivial positive-definite time-reversal invariant theory is K = [2, 0, 1], where the permutation is $Q = \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}$, which is of order 2. The system has no unitary symmetries. Therefore,

Aut([2,0,1]) =
$$\mathbb{Z}_2 = \langle \mathsf{T} \rangle$$

Aut_U([2,0,1]) = 0. (4.5.13)

The transformation T is not a symmetry of the Lagrangian (because the central charge is 2), but it becomes one once we subtract two units of central charge (i.e., we consider the theory diag(K, -1, -1), which is dual to U $(1)_2 \times U(1)_{-1}$).

There are no other 2×2 congruence classes with determinant equal to 2.

Example 4.5.11 (det(K) = 5) The next positive-definite time-reversal invariant theories are K = [5, 0, 1] and K = [2, 1, 3], where the permutations are

$$\pm Q = \begin{pmatrix} 3 & 0\\ 0 & 1 \end{pmatrix} \tag{4.5.14}$$

and

$$\pm Q = \begin{pmatrix} 1 & 2\\ 2 & 3 \end{pmatrix} \tag{4.5.15}$$

respectively. They all satisfy $T^2 = C$ (and thus are of order 4). The system has no non-trivial unitary symmetries. Therefore,

$$\operatorname{Aut}([5,0,1]) = \operatorname{Aut}([2,1,3]) = \mathbb{Z}_4 = \langle \mathsf{T} \rangle$$

$$\operatorname{Aut}_U([5,0,1]) = \operatorname{Aut}_U([2,1,3]) = \mathbb{Z}_2 = \langle \mathsf{C} \rangle.$$
 (4.5.16)

These are the only 2×2 congruence classes with determinant equal to 5.

Example 4.5.12 (det(K) = 9) The next positive-definite time-reversal invariant theory is K = [3, 0, 3], where the permutations are

$$\pm Q = \begin{pmatrix} 1 & 4\\ 2 & 1 \end{pmatrix}, \begin{pmatrix} 4 & 2\\ 1 & 1 \end{pmatrix}$$
(4.5.17)

all of which satisfy $T^4 = C$ (and are thus of order 8), and

$$\pm Q = \begin{pmatrix} 2 & 4 \\ 1 & 1 \end{pmatrix}, \begin{pmatrix} 4 & 1 \\ 1 & 2 \end{pmatrix}$$
(4.5.18)

all of which satisfy $T^2 = C$ (and are thus of order 4).

Similarly, the non-trivial unitary symmetries are

$$\pm Q = \begin{pmatrix} 0 & 5\\ 1 & 0 \end{pmatrix}, \begin{pmatrix} 0 & 1\\ 1 & 0 \end{pmatrix}, \begin{pmatrix} 5 & 0\\ 0 & 1 \end{pmatrix}$$
(4.5.19)

the first one of which squares to C (and is thus of order 4), and the other two are of order 2.

As it turns out, all these symmetries can be generated from just the two matrices

$$\mathsf{T}: \begin{pmatrix} 1 & 4 \\ 2 & 1 \end{pmatrix}, \quad \mathsf{U}_1: \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} \tag{4.5.20}$$

which satisfy $\mathsf{T}^8 = \mathsf{U}_1^2 = 1$ and $\mathsf{U}_1\mathsf{T} = \mathsf{T}^3\mathsf{U}_1$, and so the group of symmetries is semidihedral, $SD_{16} \cong \mathbb{Z}_8 \rtimes \mathbb{Z}_2$. Similarly, the matrices U_1 and $\mathsf{U}_2 := \mathsf{T}^2$ satisfy $\mathsf{U}_2^4 = \mathsf{U}_1^2 = 1$ and $\mathsf{U}_2\mathsf{U}_1\mathsf{U}_2 = \mathsf{U}_1$, and generate the whole group of unitary symmetries, and so the latter is dihedral, $D_8 \cong \mathbb{Z}_4 \rtimes \mathbb{Z}_2$. All in all, the group of symmetries is

Aut([3,0,3]) =
$$SD_{16} = \langle \mathsf{T}, \mathsf{U}_1 \rangle$$

Aut_U([3,0,3]) = $D_8 = \langle \mathsf{U}_1, \mathsf{U}_2 \rangle$. (4.5.21)

The rest of binary forms with det(K) = 9 are K = [1, 0, 9] and K = [2, 1, 5], neither of which is time-reversal invariant. They have no non-trivial unitary symmetries either.

Example 4.5.13 (det(K) = 12**)** The matrix K = [4, 2, 4] has a unitary symmetry with $U^6 = 1$ (and no time-reversal).

The non-trivial permutations are

$$\pm Q = \begin{pmatrix} 1 & 3 \\ 1 & 2 \end{pmatrix}, \begin{pmatrix} 2 & 1 \\ 3 & 1 \end{pmatrix}, \begin{pmatrix} 1 & 2 \\ 1 & 3 \end{pmatrix}, \begin{pmatrix} 2 & 3 \\ 3 & 2 \end{pmatrix}, \begin{pmatrix} 3 & 1 \\ 2 & 1 \end{pmatrix}$$
(4.5.22)

which are of order 6, 6, 2, 2, 2, respectively.

The whole group can be generated from one of the order 6 permutations, and one of the order 2 ones. They satisfy $U_1^6 = U_2^2 = 1$, together with $U_1U_2U_1 = U_2$, and so the group structure is dihedral:

The rest of binary forms of the same determinant are K = [1, 0, 12], K = [2, 0, 6], and K = [3, 0, 4], none of which is time-reversal invariant. One has

$$Aut([1, 0, 12]) = Aut_U([1, 0, 12]) = \mathbb{Z}_2 \times \mathbb{Z}_2$$

$$Aut([2, 0, 6]) = Aut_U([2, 0, 6]) = \mathbb{Z}_2$$

$$Aut([3, 0, 4]) = Aut_U([3, 0, 4]) = \mathbb{Z}_2 \times \mathbb{Z}_2.$$

(4.5.24)

Example 4.5.14 (det(K) = 18) The first positive-definite time-reversal invariant theory with a T such that $T^2 \neq C$ and det(K) not a perfect square is K = [3, 0, 6], where the permutations are

$$\pm Q = \begin{pmatrix} 3 & 1\\ 2 & 3 \end{pmatrix} \tag{4.5.25}$$

both of which satisfy $T^2 = C$ (and thus are of order 4), and

$$\pm Q = \begin{pmatrix} 3 & 1\\ 4 & 2 \end{pmatrix} \tag{4.5.26}$$

both of which are of order 2, i.e. $T^2 = 1$. If one chooses $T^2 = (-1)^F$, the latter admit a well-defined \mathbb{Z}_{16} anomaly, which is easily evaluated to be ± 2 .

The only non-trivial unitary symmetry is

$$\pm Q = \begin{pmatrix} 1 & 0\\ 0 & 5 \end{pmatrix} \tag{4.5.27}$$

which is of order 2.

If we denote by T one of the order 4 time-reversal symmetries, and by U one of the unitary ones, then one may check that these two operations generate the whole group of symmetries. One has $T^4 = U^2 = 1$ and TUT = U, and so

$$\operatorname{Aut}([3,0,6]) = D_8 = \langle \mathsf{T}, \mathsf{U} \rangle$$
$$\operatorname{Aut}_U([3,0,6]) = \mathbb{Z}_2 \times \mathbb{Z}_2 = \langle \mathsf{C}, \mathsf{U} \rangle .$$
(4.5.28)

The rest of binary forms with det(K) = 18 are K = [1, 0, 18] and K = [2, 0, 9], neither of which is time-reversal invariant. They have no non-trivial unitary symmetries either.

Example 4.5.15 (det(K) = 49) The first example of a time-reversal invariant theory where the order of the symmetry is greater than 8 is K = [7, 0, 7], where the permutations are

$$\pm Q = \begin{pmatrix} 2 & 11 \\ 3 & 2 \end{pmatrix}, \begin{pmatrix} 3 & 12 \\ 2 & 3 \end{pmatrix}, \begin{pmatrix} 9 & 3 \\ 4 & 2 \end{pmatrix}, \begin{pmatrix} 10 & 2 \\ 5 & 3 \end{pmatrix},$$
(4.5.29)

all of which satisfy $T^8 = C$ (and thus are of order 16), and

$$\pm Q = \begin{pmatrix} 2 & 10 \\ 3 & 5 \end{pmatrix}, \begin{pmatrix} 3 & 9 \\ 2 & 4 \end{pmatrix}, \begin{pmatrix} 9 & 4 \\ 4 & 5 \end{pmatrix}, \begin{pmatrix} 10 & 5 \\ 5 & 4 \end{pmatrix},$$
(4.5.30)

all of which satisfy $T^2 = C$ (and thus are of order 4).

The non-trivial unitary symmetries are

$$\pm Q = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}, \begin{pmatrix} 13 & 0 \\ 0 & 1 \end{pmatrix}, \begin{pmatrix} 5 & 9 \\ 2 & 2 \end{pmatrix}, \begin{pmatrix} 9 & 2 \\ 2 & 5 \end{pmatrix}, \begin{pmatrix} 0 & 13 \\ 1 & 0 \end{pmatrix}, \begin{pmatrix} 9 & 5 \\ 2 & 2 \end{pmatrix}, \begin{pmatrix} 2 & 9 \\ 5 & 2 \end{pmatrix}$$
(4.5.31)

which are of order 2, 2, 2, 2, 4, 8, 8, respectively.

If we let T denote one of the order 16 time-reversal permutations, and U_1 one of the order 2 unitary permutations, then one may check that these two operations generate the whole group. Furthermore, one has $T^{16} = U_1^2 = 1$ and $U_1TU_1 = T^7$, and so the group is the semidihedral group of order 32. On the other hand, if we let U_2 be one of the order 8 unitary symmetries, then one may check that these two operations generate the whole unitary group. One has $U_2^8 = U_1^2 = 1$ and $U_2U_1U_2 = U_1$, which is the dihedral group of order 16. All in all, the group of symmetries is

$$Aut([7,0,7]) = SD_{32} = \langle \mathsf{T}, \mathsf{U}_1 \rangle$$

$$Aut_U([7,0,7]) = D_{16} = \langle \mathsf{U}_1, \mathsf{U}_2 \rangle.$$
(4.5.32)

The rest of binary forms with det(K) = 49 are K = [1, 0, 49], K = [2, 1, 25], and $K = [5, \pm 1, 10]$, neither of which is time-reversal invariant. They have no non-trivial unitary symmetries either.

Example 4.5.16 (det(K) = 50) Take for example K = [5, 0, 10]. The anti-unitary symmetries are

$$\pm Q = \begin{pmatrix} 1 & 3 \\ 4 & 1 \end{pmatrix}, \begin{pmatrix} 1 & 7 \\ 6 & 1 \end{pmatrix}, \begin{pmatrix} 1 & 7 \\ 4 & 9 \end{pmatrix}, \begin{pmatrix} 1 & 3 \\ 6 & 9 \end{pmatrix}, \begin{pmatrix} 3 & 5 \\ 0 & 3 \end{pmatrix}, \begin{pmatrix} 3 & 5 \\ 0 & 7 \end{pmatrix}$$
(4.5.33)

the first two of which satisfy $T^6 = C$ (and are thus of order 12), and the rest of which satisfy $T^2 = C$ (and are thus of order 4).

The non-trivial unitary symmetries are

$$\pm Q = \begin{pmatrix} 3 & 4 \\ 2 & 3 \end{pmatrix}, \begin{pmatrix} 3 & 6 \\ 8 & 3 \end{pmatrix}, \begin{pmatrix} 3 & 6 \\ 2 & 7 \end{pmatrix}, \begin{pmatrix} 3 & 4 \\ 8 & 7 \end{pmatrix}, \begin{pmatrix} 1 & 0 \\ 0 & 9 \end{pmatrix}$$
(4.5.34)

the first two of which satisfy $U^3 = C$ (and are thus of order 6), and the rest of which are of order 2.

One may check that the three matrices

$$\mathsf{T}: \begin{pmatrix} 3 & 5 \\ 0 & 3 \end{pmatrix}, \quad \mathsf{U}_1: \begin{pmatrix} 7 & 4 \\ 2 & 7 \end{pmatrix}, \quad \mathsf{U}_2: \begin{pmatrix} 1 & 0 \\ 0 & 9 \end{pmatrix}$$
(4.5.35)

generate the whole group, and satisfy $\mathsf{T}^4 = \mathsf{U}_1^3 = \mathsf{U}_2^2 = [\mathsf{T}, \mathsf{U}_i] = (\mathsf{U}_1\mathsf{U}_2)^2 = 1$, and therefore

$$Aut([5,0,10]) = \mathbb{Z}_4 \times D_6 = \langle \mathsf{T}, \mathsf{U}_1, \mathsf{U}_2 \rangle$$

$$Aut_U([5,0,10]) = D_{12} = \langle \mathsf{T}^2 \mathsf{U}_1, \mathsf{U}_2 \rangle.$$
(4.5.36)

The rest of binary forms with det(K) = 50 are [6, ±2, 9], [3, ±1, 17], [1, 0, 50], and [2, 0, 25], and they are all time-reversal invariant with symmetry group Aut(\star) = $\mathbb{Z}_4 = \langle \mathsf{T} \rangle$ and Aut_U(\star) = $\mathbb{Z}_2 = \langle \mathsf{C} \rangle$.

We now move on to 2×2 indefinite matrices.

Example 4.5.17 (det(K) = -2) The only binary form with det(K) = -2 is [1, 1, -1], which contains four lines. The theory has no non-trivial unitary permutations, and one anti-unitary one, effected by

$$Q = \begin{pmatrix} 2 & 1\\ 1 & 0 \end{pmatrix} \tag{4.5.37}$$

which squares to the identity. Therefore,

Aut
$$([1, 1, -1]) = \mathbb{Z}_2 = \langle \mathsf{T} \rangle$$

Aut $_U([1, 1, -1]) = 0.$ (4.5.38)

When $\mathsf{T}^2 = (-1)^F$, this symmetry admits a well-defined \mathbb{Z}_{16} anomaly, which is easily evaluated to be $\nu = \pm 2$.

Example 4.5.18 (det(K) = -3) The two binary forms are K = [1, 1, -2] and K = [2, 1, -1], neither of which admits an anti-unitary permutation. The unitary permutations are the trivial one, i.e.,

$$\operatorname{Aut}([1, 1, -2]) = \operatorname{Aut}_U([1, 1, -2]) = \mathbb{Z}_2 = \langle \mathsf{C} \rangle$$

$$\operatorname{Aut}([2, 1, -1]) = \operatorname{Aut}_U([2, 1, -1]) = \mathbb{Z}_2 = \langle \mathsf{C} \rangle.$$
(4.5.39)

Example 4.5.19 (det(K) = -4) All the matrices are of the twisted gauge theory type, K = [0, 2, k], with k = 0, 1, 2, 3. For k odd there are no anti-unitary symmetries, while the unitary ones are trivial:

Aut
$$([0,2,k]) = \operatorname{Aut}_U([0,2,k]) = \mathbb{Z}_2 = \langle \mathsf{C} \rangle, \qquad k = 1,3.$$
 (4.5.40)

For k even, there are anti-unitary symmetries. In particular, for k = 0 we have the trivial permutation and the electric-magnetic duality $\mathbf{e} \leftrightarrow \mathbf{m}$, as is well known. There is also the unitary symmetry $\mathbf{e} \leftrightarrow \mathbf{m}$, which can be obtained from composing the two anti-unitary symmetries. Similarly, for k = 2, the anti-unitary permutation is $\mathbf{m} \leftrightarrow \mathbf{em}$, and there are no unitary symmetries. In short,

$$\operatorname{Aut}([0, 2, 0]) = \mathbb{Z}_{2} \times \mathbb{Z}_{2} = \langle \mathsf{T}, \mathsf{T}' \rangle$$

$$\operatorname{Aut}_{U}([0, 2, 0]) = \mathbb{Z}_{2} = \langle \mathsf{T}\mathsf{T}' \rangle$$

$$\operatorname{Aut}([0, 2, 2]) = \mathbb{Z}_{2} = \langle \mathsf{T} \rangle$$

$$\operatorname{Aut}_{U}([0, 2, 2]) = 0.$$

(4.5.41)

Example 4.5.20 (det(K) = -5) The representatives are K = [2, 1, -2] and K = [1, 2, -1]. They both have a $T^2 = C$ permutation, and no non-trivial unitary symmetries. In other words,

$$\operatorname{Aut}([2, 1, -2]) = \operatorname{Aut}([1, 2, -1]) = \mathbb{Z}_4 = \langle \mathsf{T} \rangle$$

$$\operatorname{Aut}_U([2, 1, -2]) = \operatorname{Aut}_U([1, 2, -1]) = \mathbb{Z}_2 = \langle \mathsf{C} \rangle.$$

(4.5.42)

Example 4.5.21 (det(K) = -9) All the matrices are of the twisted gauge theory type, K = [0, 3, k], with k = 0, ..., 5. There are anti-unitary symmetries only for k = 0, 3:

$$Aut([0,3,k]) = D_8$$

$$Aut_U([0,3,k]) = \mathbb{Z}_2 \times \mathbb{Z}_2, \qquad k = 0,3$$
(4.5.43)

while for the rest of levels the only symmetry is charge conjugation:

Aut
$$([0,3,k]) = \operatorname{Aut}_U([0,3,k]) = \mathbb{Z}_2, \qquad k = 1, 2, 4, 5.$$
 (4.5.44)

Example 4.5.22 (det(K) = -18) The first example with time-reversal with order greater than 4 is K = [3, 3, -3], whose anti-unitary permutations read

$$\pm Q = \begin{pmatrix} 2 & 5 \\ -1 & 0 \end{pmatrix}, \begin{pmatrix} 6 & 7 \\ 1 & 2 \end{pmatrix}, \begin{pmatrix} 2 & 5 \\ -1 & 4 \end{pmatrix}, \begin{pmatrix} 6 & 5 \\ 1 & 0 \end{pmatrix}$$
(4.5.45)

(which are of order 8, 8, 4, 4), and whose non-trivial unitary permutations read

$$\pm Q = \begin{pmatrix} 5 & 4 \\ 4 & 1 \end{pmatrix}, \begin{pmatrix} 1 & 2 \\ 0 & -1 \end{pmatrix}, \begin{pmatrix} 5 & 6 \\ 4 & 1 \end{pmatrix}$$
(4.5.46)

(which are of order 4, 2, 2). It is a simple exercise to check that

$$\operatorname{Aut}([3,3,-3]) = SD_{16}$$

$$\operatorname{Aut}_U([3,3,-3]) = D_8.$$
(4.5.47)

The rest of the binary forms with the same determinant are [1, 4, -2] and [2, 4, -1], which have $\operatorname{Aut}(\star) = \operatorname{Aut}_U(\star) = \mathbb{Z}_2 = \langle \mathsf{C} \rangle$.
Example 4.5.23 (det(K) = -20**)** The next interesting example is K = [4, 2, -4], which has

$$\operatorname{Aut}([4, 2, -4]) = \mathbb{Z}_4 \times D_6$$

$$\operatorname{Aut}_U([4, 2, -4]) = D_{12}.$$
(4.5.48)

The rest of binary forms with the same determinant are [2, 4, -2], which has $\operatorname{Aut}(\star) = \mathbb{Z}_4$ and $\operatorname{Aut}_U(\star) = \mathbb{Z}_2$, and [1, 4, -4] and [4, 4, -1], which have $\operatorname{Aut}(\star) = \operatorname{Aut}_U(\star) = \mathbb{Z}_2 \times \mathbb{Z}_2$.

Example 4.5.24 (det(K) = -27) Another interesting example is the pair K = [3, 3, -6], K = [6, 3, -3], which has

$$Aut(\star) = Aut_U(\star) = D_{12}.$$
 (4.5.49)

The rest of binary forms with the same determinant are [1, 5, -2] and [2, 5, -1], which have $\operatorname{Aut}(\star) = \operatorname{Aut}_U(\star) = \mathbb{Z}_2$.

Example 4.5.25 (det(K) = -49) As 49 is a perfect square, these matrices are of the twisted gauge theory type. One has

$$Aut([0,7,k]) = \mathbb{Z}_3 \rtimes D_8$$

$$Aut_U([0,7,k]) = D_{12}$$
(4.5.50)

if $k \propto 7$, and

$$\operatorname{Aut}([0,7,k]) = \operatorname{Aut}_U([0,7,k]) = \mathbb{Z}_2$$
 (4.5.51)

otherwise.

Example 4.5.26 (det(K) = -121) The next interesting example is, again, of the twisted gauge theory type. One has

$$\operatorname{Aut}([0, 11, k]) = \mathbb{Z}_5 \rtimes D_8$$

$$\operatorname{Aut}_U([0, 11, k]) = D_{20}$$

(4.5.52)

if $k \propto 11$, and

$$\operatorname{Aut}([0, 11, k]) = \operatorname{Aut}_U([0, 11, k]) = \mathbb{Z}_2$$
 (4.5.53)

otherwise.

Finally, we consider a few higher-dimensional examples, chosen at random:

Example 4.5.27 (det(K) = 16) The theory with matrix

$$K = \begin{pmatrix} 3 & -1 & -1 \\ -1 & 3 & -1 \\ -1 & -1 & 3 \end{pmatrix}$$
(4.5.54)

has

$$\operatorname{Aut}(\star) = A_4 \rtimes D_8$$
$$\operatorname{Aut}_U(\star) = \mathbb{Z}_2 \times S_4.$$
(4.5.55)

Example 4.5.28 (det(K) = 36) The theory with matrix

$$K = \begin{pmatrix} 3 & 0 & 0 \\ 0 & 4 & 2 \\ 0 & 2 & 4 \end{pmatrix}$$
(4.5.56)

has

$$\operatorname{Aut}(\star) = S_3 \times SD_{32}$$

$$\operatorname{Aut}_U(\star) = S_3 \times D_8.$$

(4.5.57)

Example 4.5.29 (det(K) = 48) The theory with matrix

$$K = \begin{pmatrix} 1 & 0 & 0 \\ 0 & 8 & 4 \\ 0 & 4 & 8 \end{pmatrix}$$
(4.5.58)

has

$$\operatorname{Aut}(\star) = \operatorname{Aut}_U(\star) = \mathbb{Z}_2^2 \times S_4.$$
(4.5.59)

4.6 Notation and definitions.

For the convenience of the reader, we gather here some common definitions we use throughout the text.

We denote by $\mathbb{Z} := \{0, \pm 1, \pm 2, ...\}$ the set of all integers, and by \mathbb{T}, \mathbb{P} the two subsets

$$\mathbb{T} := \{k \in \mathbb{Z} \mid kp - q^2 = 1 \text{ for some } p, q \in \mathbb{Z}\}$$

$$\mathbb{P} := \{k \in \mathbb{Z} \mid kp^2 - q^2 = 1 \text{ for some } p, q \in \mathbb{Z}\}.$$
(4.6.1)

One has $\mathbb{P} \subset \mathbb{T} \subset \mathbb{Z}$.

All primes greater than 2 are odd, and so they can be written as $4n \pm 1$ for some integer n. Those of the form 4n + 1 are called *Pythagorean* (because they can be written as the sum of two squares, unlike those of the form 4n - 1, as per Fermat's theorem).

The function $\phi: \mathbb{Z} \to \mathbb{Z}$ denotes the Euler totient function: $\phi(k)$ is the number of integers q such that 0 < q < k and gcd(q, k) = 1, where gcd denotes the greatest common divisor. In other words, there are $\phi(k)$ integers smaller than k that are coprime to it. This function is multiplicative, $\phi(ab) = \phi(a)\phi(b)$ for any $a, b \in \mathbb{Z}$ with gcd(a, b) = 1, and is given by $\phi(\pi^n) = \pi^{n-1}(\pi - 1)$ for prime π and integer n.

The function $\omega \colon \mathbb{Z} \to \mathbb{Z}$ counts the number of distinct prime factors, i.e. the prime decomposition of a given $k \in \mathbb{Z}$ reads

$$k \equiv \prod_{i=1}^{\omega(k)} \pi_i^{n_i} \,. \tag{4.6.2}$$

We also denote $\varpi(k) := \omega(k)$ if k is odd, and $\varpi(k) := \omega(k/2)$ if even. For example,

$$\omega(1) = 0, \quad \omega(2) = \omega(3) = \omega(4) = \omega(5) = 1, \quad \omega(6) = 2, \dots$$

$$\varpi(1) = \varpi(2) = 0, \quad \varpi(3) = \varpi(4) = \dots = \varpi(11) = 1, \quad \varpi(12) = 2, \dots$$

(4.6.3)

The function $\mu: \mathbb{Z} \to \mathbb{Z}$ denotes the operation of removing the Pythagorean prime factors:

$$\mu(1) = 1, \quad \mu(2) = 2, \quad \mu(3) = 3, \quad \mu(4) = 4, \quad \mu(5) = 1, \quad \mu(6) = 6, \dots$$
 (4.6.4)

One has $k \in \mathbb{T}$ if and only if $\mu(k) = 1$ or $\mu(k) = 2$. The function $\lambda \colon \mathbb{Z} \to \mathbb{Z}$ denotes the squarefree part (i.e., $\lambda(k)$ is the smallest divisor of k such that $k/\lambda(k)$ is a perfect square):

$$\lambda(1) = 1, \quad \lambda(2) = 2, \quad \lambda(3) = 3, \quad \lambda(4) = 1, \quad \lambda(5) = 5, \dots, \lambda(8) = 2, \dots$$
 (4.6.5)

We denote by $\mathbb{Z}^{n \times n}$ the set of all integral $n \times n$ matrices, and by $\operatorname{GL}_n(\mathbb{Z}) \subset \mathbb{Z}^{n \times n}$ the subset of invertible matrices over \mathbb{Z} . A given matrix is invertible over \mathbb{Z} if and only if its determinant is ± 1 , and so the elements of $\operatorname{GL}_n(\mathbb{Z})$ are known as unimodular matrices.

Given some set A with some extra structure σ , we denote by $\operatorname{Aut}(A, \sigma) \subseteq S_A$ the set of all permutations of A that "respect" the structure σ , and whose group operation is that inherited from S_A (i.e., composition). For example, if $\times : A \times A \to A$ is a binary product such that (A, \times) is a group, then $\operatorname{Aut}(A, \times)$ is the set of permutations that are group homomorphisms. Similarly, if A is a group and $\theta : A \to U(1)$ is a quadratic form on it, $\operatorname{Aut}(A, \theta)$ denotes the set of automorphisms of A that leave θ invariant, perhaps up to complex conjugation: $\theta(\pi(a)) = \theta(a)^{\pm 1}$ for all $a \in A$ and $\pi \in \operatorname{Aut}(A)$. If the data (A, θ) comes from a Chern-Simons theory with matrix K, we also use the notation $\operatorname{Aut}(K) \equiv \operatorname{Aut}(A, \theta)$, or even $\operatorname{Aut}(\operatorname{U}(1)_k)$ in the 1×1 case.

Given some unital ring A, we denote by A^{\times} the group of units of A – the set of its invertible elements. For example, one has $\operatorname{GL}_n(\mathbb{Z}) \equiv (\mathbb{Z}^{n \times n})^{\times}$.

The group \mathbb{Z}_k denotes the cyclic group of order k, which consists of the set $\{0, 1, \ldots, k-1\}$, where the product operation is just addition, followed by reduction modulo k. One can also endow \mathbb{Z}_k with integer product, which makes it into a ring (integer product is not usually invertible); the group of units is denoted by \mathbb{Z}_k^{\times} , and its order is $\phi(k)$.

We also recall some basic definitions from group theory, following [276].

Definition 2.1.3 Let N and G be groups. Then an *action* of G on N is a homomorphism $\theta: G \to \operatorname{Aut}(N)$. This is described by saying that G *acts* on N or that N is a G-group.

Definition 2.1.4 Let G and N be groups such that G acts on N with action given by θ . Then the *semi-direct product* $N \rtimes_{\theta} G$ of N by G with this action is defined as follows. The underlying set of $N \rtimes_{\theta} G$ is $G \times N$ and the multiplication is defined by $(g_1, n_2)(g_2, n_2) = (g_1g_2, (n_1^{g_2\theta})n_2)$. **Definition 2.2.6** [...] A group G is an external central product $H \circ K$ of two groups H and K if there exists an isomorphism $\theta: Z(H) \to Z(G)$ such that G is $(H \times K)/N$ where $N = \{(h, h^{-1}\theta) \mid h \in Z(H)\}.$

Definition 2.3.1 Let G be a group and Ω a non-empty finite set. Then G acts on Ω if, to each $\omega \in \Omega$ and $g \in G$, there corresponds a unique element $\omega^g \in \Omega$ such that, if g_1 and $g_2 \in G$ then $(\omega^{g_1})^{g_2} = \omega^{g_1g_2}$; and $\omega^1 = \omega$. If G acts on Ω then the *permutation representation* of G corresponding to the action is the homomorphism $\rho: G \to \Sigma_{\Omega}$, the symmetric group on Ω , defined by $\omega(g\rho) = \omega^g$ for all $\omega \in \Omega$ and all $g \in G$.

Definition 2.3.2 Let H be a group and Ω a non-empty finite set. Then H^{Ω} denotes the set of all maps from Ω to H. For $f_1, f_2 \in H$, define $f_1f_2 \in H^{\Omega}$ by $\omega(f_1f_2) = (\omega f_1)(\omega f_2)$ for all $\omega \in \Omega$.

Definition 2.3.3 Let H be a group, and G be a finite group acting on a non-empty finite set Ω . Then an action of G on the group H^{Ω} is defined as follows. For each $g \in G$ and $f \in H^{\Omega}$, define $f^g \in H^{\Omega}$ by $\omega f^g = \omega^{g^{-1}} f$ for all $\omega \in \Omega$. The *(permutational) wreath product* $H \wr G$ of H with G corresponding to this action of G on Ω is the split extension $H^{\Omega} \rtimes G$ with this action of G on H^{Ω} .

Finally, we define a few important finite groups (see e.g. Definition 2.1.11 in [276]):

• The *dihedral group* D_{2n} of order 2n is defined by

$$D_{2n} = \langle x, y \mid y^n = x^2 = (xy)^2 = 1 \rangle \cong \mathbb{Z}_n \rtimes \mathbb{Z}_2$$

$$(4.6.6)$$

• The semidihedral group $SD_{2^{n+1}}$ of order 2^{n+1} is defined by

$$SD_{2^{n+1}} = \langle x, y \mid y^{2^n} = x^2 = (xy)^2 y^{2^{n-1}} = 1 \rangle$$
 (4.6.7)

• The symmetric group S_n of order n!, corresponding to all the permutations of n objects, and its commutator subgroup A_n , of order n!/2, known as the alternating group and given by the even permutations of S_n . One has $S_n = A_n \rtimes \mathbb{Z}_2$ for $n \ge 5$.

4.7 Further results.

In this section we collect some further results concerning the theory $U(1)_k$ which may prove useful in subsequent studies of this system. We begin by making some remarks concerning the set \mathbb{T} , defined as those integers k such that -1 is a quadratic residue modulo k, i.e., those integers for which the equation $q^2 = -1 + pk$ is solvable for some integers p, q. It is straightforward to show that any solution (p,q) is such that q is congruent to q_0 modulo k, where (p_0, q_0) is a solution with $q_0 \in [0, k)$. More precisely, if (p_0, q_0) is a solution, then so is (P(n), Q(n)) for any $n \in \mathbb{Z}$, where

$$P(n) := p_0 + 2q_0 n + kn^2$$

$$Q(n) := q_0 + kn$$
(4.7.1)

as is easily checked. This is not particular to our problem; the solutions to congruences of the form $f(q) = 0 \mod k$, for some polynomial $f \colon \mathbb{Z} \to \mathbb{Z}$, are always defined modulo k.

Generically speaking, this type of congruences are solved by first solving them modulo the prime divisors of k. Indeed, if k is to divide f(q), then so must its divisors. This means that the prime divisors of k are essential in deciding whether $q^2 + 1 = 0 \mod k$ is solvable or not. To be precise, one of the key results concerning the set T is the following:

Proposition 4.7.1 A given k is in \mathbb{T} if and only if all its prime factors are Pythagorean (that is, congruent to 1 modulo 4), perhaps up to a single factor of 2.

Proof. By reducing $kp = 1 + q^2$ modulo 4, and considering the odd q and even q cases separately, it becomes clear that k cannot be a multiple of 4. Similarly, by Gaussian reciprocity, -1 is a quadratic residue modulo a prime π if and only if π is Pythagorean, and so k cannot be a multiple of a non-Pythagorean prime either. This proves that the conditions above are necessary; proving that they are also sufficient can be done by explicitly constructing a solution q. We now sketch how this can be done.

First off, if k is a Pythagorean prime, we can use Wilson's theorem to obtain an explicit expression for q. Indeed,

$$q = \left(\frac{k-1}{2}\right)! \tag{4.7.2}$$

satisfies $q^2 = -1 \mod k$. One can also take

$$q = (k - a)!! \tag{4.7.3}$$

where a is any of $\{\pm 1, 2, 3\}$.

Lifting the solution to a prime power $k = \pi^n$ can be done using the Hensel lemma. If we let q_1 be the solution for n = 1, then the general solution can be obtained via the quadratic map

$$q_n = q_{n-1} - a(q_{n-1}^2 + 1) \tag{4.7.4}$$

where a is a solution to $2q_1a = 1 \mod \pi$ (e.g., $a = (2q_1)^{\pi-2}$, as per Fermat's little theorem).

Finally, finding a solution for arbitrary k requires the use of the Chinese Remainder Theorem. For example, let $k = a_1a_2$ with a_1, a_2 two prime powers. Then $q^2 = -1 \mod k$ requires $q^2 = -1 \mod a_i$, which by the previous paragraph has a solution q_i . With this, the solution of $q^2 = -1 \mod k$ is $q = q_1\alpha_1a_2 + q_2\alpha_2a_1 \mod k$, where α_1, α_2 are the Bézout coefficients for a_1, a_2 (i.e., a pair of integers such that $a_1\alpha_1 + a_2\alpha_2 = 1$, which can be computed using the Euclidean algorithm). By iteration we can easily find the solutions for an arbitrary integer $k = a_1a_2...a_n$, and so the conditions in proposition 4.7.1 are also sufficient.

The integers q that solve $q^2 = -1 \mod k$ implement the time-reversal permutations on the anyons of $U(1)_k$. The lines $a \in \mathcal{A}$ that are fixed under time-reversal (modulo local operators) play a special role in analysing the time-reversal symmetry of a system (and its anomalies), see e.g. [64, 77]. We have the following:

Proposition 4.7.2 The only lines that satisfy $T(a) \equiv a$ are the identity and the transparent fermion. If k is odd, no line satisfies $T(a) = a \times \psi$, while if k is even, the only lines satisfying $T(a) = a \times \psi$ are $a = k/2 \times 1$ and $a = k/2 \times \psi$.

Proof. Any line fixed by T (perhaps up to ψ) has $a = \mathsf{T}^2(a) = \mathsf{C}(a)$. Let k be odd; then lines fixed by C satisfy $2\alpha = 0 \mod 2k$, that is, $\alpha \propto k$. Both lines $\alpha = 0, k$ have $\mathsf{T}(\alpha) = \alpha$, and so there are no lines with $\mathsf{T}(a) = a \times \psi$.

Now let k be even; then lines fixed by C satisfy $2\alpha = 0 \mod k$, that is, $\alpha \propto k/2$. One may check that $a = (0, \beta)$ satisfies $\mathsf{T}(a) = a$, and $a = (k/2, \beta)$ satisfies $\mathsf{T}(a) = a \times \psi$. \Box

We thus see that the property $\mathsf{T}^2 = \mathsf{C}$ implies that the set of lines that are fixed by time-reversal is very small. More generally, it is possible to argue that, due to $\theta(\mathsf{T}(a)) = \theta(a)^*$, an anyon can only be fixed by T (perhaps up to ψ) if its spin is either $\theta(a) = \pm 1$ or $\theta(a) = \pm i$, i.e., if $h \in \{0, \frac{1}{4}, \frac{1}{2}, \frac{3}{4}\}$. These are the bosons, fermions, semions, and anti-semions of the theory. For some purposes, it may be useful to know how many of these lines the theory supports. We have the following:

Proposition 4.7.3 Let $k \in \mathbb{Z}$, and denote by N_h the number of lines of spin h in $U(1)_k$ (as a spin TQFT), and by $\lambda(k)$ the squarefree part of k. Then we have $N_0 = N_{1/2} = \sqrt{k/\lambda(k)}$. Furthermore, if we write $k = 2^e \tilde{k}$, with \tilde{k} odd, then $N_{1/4} = N_{3/4} = 0$ if e is even, and $N_{1/4} = N_{3/4} = \sqrt{k/\lambda(k)}$ if odd.

Proof. We shall need the following trivial fact: given some integer $k \in \mathbb{Z}$, all solutions to the equation

$$\alpha^2 = k\beta, \qquad \alpha, \beta \in \mathbb{Z} \tag{4.7.5}$$

are of the form $(\alpha_n, \beta_n) = (n\sqrt{k\lambda(k)}, n^2\lambda(k))$ for some integer *n*. Indeed, if $k\beta$ is to be a perfect square, then β must be proportional to $\lambda(k)$; and the constant of proportionality must itself be a perfect square.

We next count the bosons and fermions of $U(1)_k$.

We begin with the k odd case. An anyon $\alpha \in [0, 2k)$ has vanishing spin iff $\alpha^2 = 2k\beta$ for some integer β . All the solutions to this equation are of the form $\alpha = n\sqrt{2k\lambda(2k)}$ for some integer $n = 0, 1, \dots, \lfloor \frac{2k-1}{\sqrt{2k\lambda(2k)}} \rfloor$. Therefore, there are

$$\left\lfloor \frac{2k-1}{\sqrt{2k\lambda(2k)}} \right\rfloor + 1 \equiv \sqrt{\frac{k}{\lambda(k)}}$$
(4.7.6)

bosons. Similarly, the fermions are given by the solutions to $\alpha^2 = k(2\beta + 1)$, that is, $\alpha = n\sqrt{k\lambda(k)}$ with $n = 1, 3, \ldots, \lfloor \frac{2k-1}{\sqrt{k\lambda(k)}} \rfloor$. Therefore, there are

$$\frac{1}{2} \left(\left\lfloor \frac{2k-1}{\sqrt{k\lambda(k)}} \right\rfloor + 1 \right) \equiv \sqrt{\frac{k}{\lambda(k)}}$$
(4.7.7)

fermions.

We now move on the the k even case. The bosons in the spin theory come from the bosons and fermions in the non-spin theory. The former solve $\alpha^2 = 2k\beta$ and the latter solve $\alpha^2 = k(2\beta + 1)$. Together, they solve $\alpha^2 = k\beta$, that is, $\alpha = n\sqrt{k\lambda(k)}$, with $n = 0, 1, \ldots, \lfloor \frac{k-1}{\sqrt{k\lambda(k)}} \rfloor$. Therefore, there are

$$\left\lfloor \frac{k-1}{\sqrt{k\lambda(k)}} \right\rfloor + 1 \equiv \sqrt{\frac{k}{\lambda(k)}}$$
(4.7.8)

bosons. The counting of the fermions is identical.

A very similar argument proves the claim for the semions. For k odd, the counting is straightforward. For k even, one is to count the spin 1/4 and 3/4 lines in the bosonic theory, which solve $2\alpha^2 = k(2p+1)$. Writing $k = 2^e \tilde{k}$, with \tilde{k} odd, it is clear that no solutions exist for e even (because $\sqrt{2}$ is not integral). For e odd, the solution is $\alpha = 2^{(e-1)/2} n \sqrt{\tilde{k}\lambda(\tilde{k})}$, with $n = 1, 3, \ldots, \lfloor \frac{2^e \tilde{k} - 1}{2^{(e-1)/2} \sqrt{\tilde{k}\lambda(\tilde{k})}} \rfloor$. Thus, there are

$$\frac{1}{2} \left(\left\lfloor \frac{2^e \tilde{k} - 1}{2^{(e-1)/2} \sqrt{\tilde{k} \lambda(\tilde{k})}} \right\rfloor + 1 \right) \equiv \sqrt{\frac{k}{\lambda(k)}}$$
(4.7.9)

spin h = 1/4 lines in the spin theory, and as many spin 3/4 lines.

A similar technique can be applied to counting other lines N_h .

We now move on to the so-called Pell numbers:

Definition 4.7.1 An integer k is said to be *Pell* if there exists a pair of integers p, q such that $kp^2 - q^2 = 1$. The set of Pell numbers is denoted by \mathbb{P} .

We include here some known facts about Pell numbers, the first few of which are $k = 1, 2, 5, 10, 13, 17, 26, 29, \ldots$:

- No perfect square other than 1 is ever Pell. (Indeed, $n^2 m^2 > 2m$ for n > m > 0, and so this expression cannot equal 1).
- All Pell numbers are in \mathbb{T} (but the converse is not true; the first few exceptions are $\mathbb{T} \setminus \mathbb{P} = \{25, 34, 146, 169, 178, 194, \dots\}$).
- A squarefree integer k is Pell iff the fundamental unit σ of $\mathbb{Q}(\sqrt{k})$ has norm -1. The rest of units are of the form $\pm \sigma^n$ for some integer n (see e.g. [277], theorem 11.4.1).
- k is Pell iff the convergents of \sqrt{k} have odd period. If (p_0, q_0) denotes the fundamental solution, then the rest of solutions are $q_n + p_n\sqrt{k} = (q_0 + p_0\sqrt{k})^{2n+1}$ (see e.g. [271], theorems 5.15 and 5.16). Equivalently,

$$\begin{pmatrix} p_n \\ q_n \end{pmatrix} = \begin{pmatrix} q_0 & p_0 \\ kp_0 & q_0 \end{pmatrix}^{2n} \begin{pmatrix} p_0 \\ q_0 \end{pmatrix}$$
(4.7.10)

(Note that the determinant of this matrix is -1, and so its odd powers generate *positive* norm units).

- k is Pell iff it can be written as $k = a^2 + b^2$ for relatively prime $a, b \in \mathbb{Z}$, with b odd, and such that the Gauss-type Diophantine equation $b(V^2 W^2) 2aVW = 1$ is solvable with $V, W \in \mathbb{Z}$ [278].
- Let π denote a prime not congruent to 3 mod 4. Then any integer of the form $k = \pi$, or $k = \pi_1 \pi_2$ with $(\pi_1, \pi_2) = -1$, is Pell (where (\cdot, \cdot) is the Legendre symbol; see e.g. [277], theorem 11.5.7). Furthermore, any odd integer of the form $k = \pi_1 \pi_2 \cdots \pi_{2n+1}$ such that there is no triplet (a, b, c) with $(\pi_a, \pi_b) = (\pi_b, \pi_c) = +1$, is Pell [279].

Pell numbers appear naturally in the study of the time-reversal properties of $U(1)_k$. For example, one has the following:

Proposition 4.7.4 If kk' satisfies the Pell equation the theory $U(1)_k \times U(1)_{-k'}$ is timereversal invariant.

Proof. Assume that

$$kk'p^2 - q^2 = 1, \qquad p, q \in \mathbb{Z}$$
 (4.7.11)

Let

$$4\pi \mathcal{L} = ka \,\mathrm{d}a - k'b \,\mathrm{d}b \tag{4.7.12}$$

and introduce the $\operatorname{GL}_2(\mathbb{Z})$ transformation

$$\mathsf{T}\colon \begin{pmatrix} a\\b \end{pmatrix} \mapsto \begin{pmatrix} q & k'p\\kp & q \end{pmatrix} \begin{pmatrix} a\\b \end{pmatrix} \tag{4.7.13}$$

The Lagrangian becomes

$$\mathsf{T} \colon 4\pi\mathcal{L} \mapsto -k\,a\,\mathrm{d}a + k'\,b\,\mathrm{d}b \tag{4.7.14}$$

as required.

Taking k' = 1 leads to the invariance of $U(1)_k \times U(1)_{-1}$ (cf. proposition 4.3.6). Moreover, this result, together with conjecture 4.4.1, leads to the following interesting purely number-theoretic conjecture:

Conjecture 4.7.1 An integer k satisfies $q^2 = -1 \mod k$ for some $q \in \mathbb{Z}$ if and only if there exists some Pell integer k' such that kk' is also Pell.

Recall that any solution of $q^2 = -1 + pk$ is of the form $p = p_0 + 2q_0n + kn^2$ (cf. (4.7.1)). If p is Pell for some n, then it suffices to take k' = p, from where the conjecture would follow (because $kp = q^2 + 1$ is automatically Pell). Noting that whenever this polynomial is prime, it is also Pell, our conjecture actually follows from the so-called Hardy-Littlewood "conjecture F" [280], which states that $ax^2 + bx + c$ is prime infinitely often unless $b^2 - 4ac$ is a perfect square or a + b and c are both even (neither condition being satisfied by our polynomials). It is widely believed that the Hardy-Littlewood conjecture is true, which implies that our conjecture – being much weaker – should be true as well.

There is a more specific result due to Lemke Oliver and Iwaniec [281, 282] that states that a polynomial of the type above represent primes or semiprimes infinitely often. But any prime, or any semiprime $\pi_1\pi_2$ with $(\pi_1, \pi_2) = -1$ is Pell. Having no reason to expect otherwise, one is lead to conjecture that both options $(\pi_1, \pi_2) = \pm 1$ appear with the same probability – which is confirmed by numerical analysis – from where it would follow that $p_0 + 2q_0n + kn^2$ generates infinitely many Pell numbers. In fact, the only possibility for a failure of our conjecture is that this polynomial never represents a prime (disproving the Hardy-Littlewood conjecture), and that all the semiprimes it represents have $(\pi_1, \pi_2) = +1$. This is extremely unlikely, but we have no proof that it cannot happen.

In any event, we checked that the conjecture is true for k up to 10^9 . For now it remains an interesting open question.

If the conjecture is true, we can in fact invert the logic and use the time-reversal invariance of $U(1)_k \times U(1)_{-k'}$ to argue that of $U(1)_k$, for any $k \in \mathbb{T}$, by mimicking the argument of proposition 4.3.5.

Added note: An unconditional proof of conjecture 4.7.1 has been discussed in MathOver-flow.

Chapter 5 Fermionic TQFTs in three dimensions.

Authorship. The first two sections of this chapter are taken almost verbatim from the paper [1], written in collaboration with Davide Gaiotto and Jaume Gomis. We also append here a few extra sections that were developed around the same time but not included in the original version of the paper.

Abstract. We systematically study the construction of three-dimensional topological theories that depend on the spin structure of spacetime. We explain how the Hilbert space is obtained, and how large diffeomorphisms act on it, which in turn is enough to calculate the partition function on arbitrary three-manifolds via surgery. The action of line operators (Wilson lines, a.k.a. anyons) on the fermionic Hilbert space is described. We also discuss the fermionic version of the Verlinde formula on arbitrary genus and with arbitrary punctures. The general picture involves some novel ingredients, for example, the use of some F-symbols of the bosonic parent of the TQFT.

5.1	Spin TQFTs and anyon condensation
5.2	Examples of anyon condensation
5.3	Fermionic surgery
5.4	Higher genus. Fermionic Verlinde formula

5.1 Spin TQFTs and anyon condensation

In this section we outline the construction of TQFTs that depend on the spin structure of the underlying manifold. The strategy we will pursue is the following. Given one such theory, one may sum over all spin structures to yield a bosonic TQFT. This corresponds to gauging the zero-form symmetry generated by fermion parity $\mathbb{Z}_2 = \langle (-1)^F \rangle$. This gauging generates a dual \mathbb{Z}_2 (d-2)-form symmetry, whose gauging takes us back to the original spin TQFT. Therefore, any spin TQFT can be constructed by gauging a certain (d-2)-form symmetry in a bosonic TQFT,⁹⁵ and we reduce the problem of constructing spin TQFTs to the more familiar problem of gauging a higher-form symmetry in regular (bosonic) TQFTs. We shall follow this strategy in d = 3 spacetime dimensions, where one can be quite explicit, thanks to the powerful formalism of modular tensor categories and two-dimensional chiral algebras.

With this in mind, we begin by reviewing known facts about 3d TQFTs, and the gauging of one-form symmetries. From the 2d point of view this corresponds to extending the chiral algebra by a simple current, and in the condensed-matter language to (abelian) anyon condensation.

Consider a 3*d* bosonic TQFT. The most basic observable of the theory is the partition function Z(M), where *M* is a compact 3-manifold. For example, if the manifold takes the form $M = \mathbb{S}^1 \times \Sigma$, with \mathbb{S}^1 a circle representing the time direction, and Σ a compact surface, then the partition function computes the dimension of the Hilbert space assigned, by canonical quantization, to the spatial slice:

$$Z(\mathbb{S}^1 \times \Sigma) = \dim(\mathcal{H}(\Sigma)).$$
(5.1.1)

The observables of the TQFT depend only on the topology of M, and therefore diffeomorphisms of Σ must act unitarily in $\mathcal{H}(\Sigma)$. Transformations that are continuously connected to the identity act trivially, so effectively we get a unitary representation of the mapping class group, the group of (equivalence classes of) large diffeomorphisms. If one understands the Hilbert space $\mathcal{H}(\Sigma)$, and the action of the MCG on it, one can compute – via surgery – the partition function on an arbitrary 3-manifold M.

With this in mind, our main task is to understand the Hilbert space assigned by a TQFT to a compact Riemann surface Σ , and how Dehn twists act on it. The basic data of the TQFT that determines this information is the following:

• The set of anyons \mathcal{A} , a finite set. This set contains a distinguished anyon, the vacuum **1**.

⁹⁵For example, in d = 2 this corresponds to a zero-form symmetry. This has been studied recently, see e.g. [117, 166, 167]. In d = 4 one gauges a two-form symmetry, cf. e.g. [283]. One should keep in mind that, potentially, an anomaly could make these gaugings ill-defined, e.g. if summing over spin structures leads to an identically vanishing partition function. This subtlety shall play no role in this work.

- The modular matrix $S: \mathcal{A} \times \mathcal{A} \to \mathbb{C}$.
- The topological spin $\theta = e^{2\pi i h} \colon \mathcal{A} \to \mathrm{U}(1).$

The full data of the TQFT involves other, more subtle objects, the so-called F- and R-symbols. These will play a role later on; for now, the S-matrix is enough. By a key result of Verlinde, the dimension of $\mathcal{H}(\Sigma)$ is determined by S as follows [38, 39]:

$$\dim(\mathcal{H}(\Sigma)) = \sum_{\alpha \in \mathcal{A}} S_{\mathbf{1},\alpha}^{\chi(\Sigma)}, \qquad (5.1.2)$$

where χ denotes the Euler characteristic $(\chi(\Sigma_g) = 2 - 2g$ for a genus g surface Σ_g). In particular, the torus has $\chi(\Sigma_1) = 0$, which means that $\mathcal{H}(\Sigma_1) \cong \mathbb{C}[\mathcal{A}]$, i.e., a basis of states of the torus Hilbert space is labelled by the anyons of the TQFT. The MCG of the torus, $SL_2(\mathbb{Z}) = \langle S, T \rangle$, is generated by S and $T := e^{-2\pi i c/24} \operatorname{diag}(\theta)$.

The theory also admits line defect operators, also labelled by \mathcal{A} . Namely, we can wrap an anyon $\alpha \in \mathcal{A}$ around the time circle \mathbb{S}^1 , which produces a defect Hilbert space $\mathcal{H}(\Sigma^{\alpha})$. From the point of view of the spatial surface, the anyon α looks like a marked point. Given a family of such punctures $\alpha_1, \ldots, \alpha_n$, the generalization of the Verlinde formula is [38]

$$\dim(\mathcal{H}(\Sigma^{\alpha_1\cdots\alpha_n})) = \sum_{\alpha\in\mathcal{A}} S_{\mathbf{1},\alpha}^{\chi(\Sigma)} \prod_{i=1}^n S_{\alpha_i,\alpha}, \qquad (5.1.3)$$

where $\chi(\Sigma_g^{\alpha_1\cdots\alpha_n}) = 2 - 2g - n$ for a surface with g handles and n boundary components. The most fundamental surface is the so-called *trinion*, i.e., a sphere with three punctures. This surface defines the *fusion coefficients*:

$$\mathcal{N}_{\alpha_1,\alpha_2,\alpha_3} := \dim(\mathcal{H}(\Sigma_0^{\alpha_1\alpha_2\alpha_3})) \equiv \sum_{\beta \in \mathcal{A}} \frac{S_{\alpha_1,\beta}S_{\alpha_2,\beta}S_{\alpha_3,\beta}}{S_{\mathbf{1},\beta}}, \qquad (5.1.4)$$

which endows \mathcal{A} with a product structure, leading to the *fusion ring* of the TQFT. The partition function on an arbitrary surface can be computed by gluing trinions (cf. the "pants decomposition"). Using unitarity of S one can recover the general case (5.1.3) from the trinion (5.1.4).

An explicit basis of states on $\mathcal{H}(\Sigma_g^{\alpha_1\cdots\alpha_n})$ can be written as follows:

where each cross \times represents a handle of Σ_g . Each segment carries an orientation and an anyon label (which we omit to simplify the notation). Each trivalent vertex with incoming

anyons α, β, γ carries an internal vector index, taking values in $1, 2, \ldots, \mathcal{N}_{\alpha,\beta,\gamma}$, which we always leave implicit.

In short, the states of $\mathcal{H}(\Sigma_g^{\alpha_1\cdots\alpha_n})$ can be represented as labelled oriented graphs with g cycles; leaves labelled by the punctures $\alpha_1, \ldots, \alpha_n$; edges labelled by anyons $\alpha \in \mathcal{A}$; and trivalent vertices labelled by internal vector indices taking $\mathcal{N}_{\alpha,\beta,\gamma}$ values, as determined by the incident edges α, β, γ .

Different bases of the Hilbert space are related to the one above by the F- and R-moves, effected by the aforementioned F- and R-symbols. These are in correspondence with the different pants decompositions of the surface.

Of particular relevance is the case of the torus with a single puncture, whose states we label as $|\beta; \alpha\rangle \in \mathcal{H}(\Sigma_1^{\alpha})$, corresponding to the configuration

$$|\beta;\alpha\rangle = \alpha - \underbrace{\times}_{\beta} \cdot \tag{5.1.6}$$

There is one such state for each possible vertex, i.e., the degeneracy of $|\beta; \alpha\rangle$ is given by the fusion coefficient $\mathcal{N}_{\alpha,\beta,\bar{\beta}}$. In particular, the diagram is allowed only if $\alpha \times \beta \propto \beta + \cdots$, i.e., if β may "absorb" the puncture α . The case of no punctures corresponds to the vacuum anyon $\alpha = \mathbf{1}$, so that all $\beta \in \mathcal{A}$ are allowed, and they all carry degeneracy $\mathcal{N}_{\mathbf{1},\beta,\bar{\beta}} = 1$. For non-trivial α , some $\beta \in \mathcal{A}$ may not contribute to $\mathcal{H}(\Sigma_1^{\alpha})$, and some other $\beta \in \mathcal{A}$ may contribute more than one state.

Given a basis of states for $\mathcal{H}(\Sigma_1)$ one can write down the operators acting on this space as matrices. In particular, the Wilson loop operators admit such a representation. Let $W^{(c)}(\alpha)$ denote the Wilson loop labelled by the anyon $\alpha \in \mathcal{A}$ running through the cycle $\mathbf{c} \in H_1(\Sigma_1, \mathbb{Z}) = \mathbb{Z}[\mathbf{a}] \oplus \mathbb{Z}[\mathbf{b}]$, where \mathbf{a}, \mathbf{b} are the standard homology cycles. These operators act by inserting α along the given cycle, e.g.

$$W^{(\mathbf{a})}(\alpha)|\beta\rangle = \bigotimes_{\alpha} \bigotimes_{\beta} = \frac{S_{\alpha,\beta}}{S_{\mathbf{1},\beta}} \bigotimes_{\beta}$$

$$W^{(\mathbf{b})}(\alpha)|\beta\rangle = \bigotimes_{\beta} = \bigotimes_{\alpha \times \beta}$$
(5.1.7)

whence

$$\langle \beta' | W^{(\mathbf{a})}(\alpha) | \beta \rangle = \delta_{\beta\beta'} \frac{S_{\alpha\beta}}{S_{\mathbf{1},\beta}}$$

$$\langle \beta' | W^{(\mathbf{b})}(\alpha) | \beta \rangle = \mathcal{N}^{\beta'}{}_{\alpha\beta}$$

$$(5.1.8)$$

Naturally, given that S interchanges the cycles **a** and **b** (up to a sign), one has

$$W^{(\mathbf{a})}(\alpha) = SW^{(\mathbf{b})}(\alpha)S^{\dagger}, \qquad W^{(\mathbf{b})}(\alpha) = SW^{(\mathbf{a})}(\bar{\alpha})S^{\dagger}$$
(5.1.9)

which is just the statement that the characters $S_{\alpha\beta}/S_{1,\beta}$ diagonalize the fusion rules.

The higher-genus case is analogous. Given a basis of $\mathcal{H}(\Sigma_g^{\alpha_1...\alpha_n})$ one can express the Wilson lines as matrices. As above, a Wilson loop $W^{(c)}(\alpha)$ inserts the anyon α along the cycle $\mathbf{c} \in H_1(\Sigma_g^{\alpha_1...\alpha_n}, \mathbb{Z})$. For example, the **a**-cycles are identical to the torus, inasmuch as wrapping an anyon in the orthogonal cycle is a local operation: one can shrink it to a point. The value of $W^{(\mathbf{a}^i)}$ acting on a given state only depends on the line running through the segment orthogonal to \mathbf{a}^i , irrespective of what the rest of the state is doing:

$$\alpha \bigotimes_{\beta} = \frac{S_{\alpha,\beta}}{S_{\mathbf{1},\beta}} \qquad (5.1.10)$$

The b-cycles, on the other hand, cannot be shrunk, and so depend on the entire configuration around such cycle: the lines running therein, and the punctures going in and out. For example, the once-punctured torus has [122]

$$W^{(\mathbf{b})}(\alpha)|\beta;\gamma\rangle = \gamma - \underbrace{\begin{pmatrix} \mathbf{x} \\ \alpha \\ \beta \end{pmatrix}}_{\beta} = F_{\beta,\alpha\times\beta} \begin{bmatrix} \gamma & \alpha\times\beta \\ \beta & \alpha \end{bmatrix} \gamma - \underbrace{\begin{pmatrix} \mathbf{x} \\ \mathbf{x} \\ \alpha\times\beta \end{pmatrix}}_{\alpha\times\beta}.$$
 (5.1.11)

Configurations with more punctures carry more factors of F. The matrix elements of arbitrary configurations of Wilson lines, on surfaces with arbitrary genus and arbitrary punctures, is entirely determined in terms of the TQFT data of the theory. Generalizing (5.1.9), the lines around the different cycles are unitarily related through the MCG of the surface.

The invertible defects – the abelian punctures – correspond to group symmetries of the theory. These are line operators, so the symmetry is a higher-form symmetry [32], in this case a one-form symmetry. Gauging the symmetry corresponds to summing over all possible insertions of the defect. This produces a new TQFT, whose set of anyons \hat{A} and modular data \hat{S} are fixed in terms of the data of the ungauged theory. Making this procedure precise is the goal of the rest of this section.

The one-form symmetry group is always a finite abelian group, i.e., a product of cyclic groups. Each abelian anyon in \mathcal{A} generates a cyclic subgroup; condensing this anyon means gauging this subgroup. If we are interested in gauging a product of cyclic groups, we can always condense a single generator at a time, iteratively. We can therefore assume without loss of generality that the group to be gauged is cyclic, say, $\mathbb{Z}_n^{(1)} = \langle \mathbf{g} \rangle$, with $\mathbf{g} \in \mathcal{A}$ a certain abelian anyon. The $\mathbb{Z}_n^{(1)}$ symmetry partitions the spectrum \mathcal{A} into n equivalence classes, according to their braiding with respect to \mathbf{g} :

$$\mathcal{A} = \bigsqcup_{q=0}^{n-1} \mathcal{A}_q, \qquad \mathcal{A}_q := \{ \alpha \in \mathcal{A} \mid B(\mathbf{g}, \alpha) = e^{2\pi i q/n} \}, \qquad (5.1.12)$$

where $B(\mathbf{g}, \alpha) := S_{\mathbf{g},\alpha}/S_{\mathbf{1},\alpha} \equiv \theta(\mathbf{g} \times \alpha)/\theta(\mathbf{g})\theta(\alpha)$ is the braiding phase with respect to \mathbf{g} . The modular data of the theory behaves nicely with respect to this grading, e.g. [284]

$$S_{\mathbf{g}^{i} \times \alpha, \mathbf{g}^{j} \times \beta} = B(\mathbf{g}, \mathbf{g})^{ij} B(\mathbf{g}, \alpha)^{j} B(\mathbf{g}, \beta)^{i} S_{\alpha, \beta}$$
(5.1.13)

The 't Hooft anomaly of $\mathbb{Z}_n^{(1)}$ is given by $B(\mathbf{g}, \mathbf{g})$, which must equal 1 if the symmetry is to be gauged. In this situation, one can prove that $\theta(\mathbf{g}) = \pm 1$, i.e., the generator is either a boson or a fermion. In the former case, the gauging yields another bosonic TQFT, while in the latter case the theory acquires a dependence on the spin structure, i.e., it becomes a spin TQFT. For now, we assume that \mathbf{g} is a boson, and return to the more interesting case of fermionic quotients later on.

5.1.1 Boson anyon condensation

We begin with some bosonic TQFT with anyons \mathcal{A} and modular matrix S, and wish to condense some boson $\mathbf{g} \in \mathcal{A}$, to produce some other bosonic TQFT, with anyons $\hat{\mathcal{A}}$ and new modular matrix \hat{S} . The standard lore of this procedure is as follows. First, in order to construct $\hat{\mathcal{A}}$, one performs the following three steps [285]:

- 1. Select the set of neutral lines, \mathcal{A}_0 (cf. (5.1.12)), i.e., those with trivial braiding with respect to g.
- 2. Identify any two lines that are in the same $\mathbb{Z}_n^{(1)}$ -orbit, i.e., if they differ by the action of \mathbf{g}^j for some $j \in \mathbb{Z}_n$.
- 3. If a given $\mathbb{Z}_n^{(1)}$ -orbit has less than *n* elements, it splits into several different anyons in $\hat{\mathcal{A}}$. Specifically, if the length is ℓ , then the orbit descends to n/ℓ copies in the condensed theory.

In what follows we shall describe the geometric interpretation of these rules, which will allow us to compute the modular data of the condensed theory from first principles, with no need to introduce any ansätze. It also admits a natural extension to spin TQFTs which shines a new light – and goes beyond – what is currently understood about such theories.

The main idea to obtain \mathcal{A} is to find the torus Hilbert space of the condensed theory, from which one can read off the set of anyons by writing down a basis of vectors (recall that $\mathcal{H}(\Sigma_1) \cong \mathbb{C}[\mathcal{A}]$). The condensed theory is obtained by gauging the \mathbb{Z}_n one-form symmetry, which means we are to insert the generator \mathbf{g} in all possible ways. Summing over all insertions \mathbf{g}^j , $j = 0, 1, \ldots, n-1$, along the spatial cycles project the Hilbert space into the invariant states. Insertions along the temporal cycles introduce twisted sectors. We will next see that insertions along the three cycles in $M = \mathbb{S}^1 \times \Sigma_1$ indeed reproduce the three steps above.

Let us begin with the time circle. Inserting g^{j} along the time direction means taking the torus with a puncture labelled by g^{j} . Therefore, the states of the condensed theory are generically of the form $|\alpha; \mathbf{g}^j\rangle$ for some $\alpha \in \mathcal{A}$. In other words, the Hilbert space of the condensed theory must be a subspace of the Hilbert space of the original theory, in the presence of an arbitrary puncture:

$$\hat{\mathcal{H}}(\Sigma_1) \subseteq \bigoplus_{j=0}^{n-1} \mathcal{H}(\Sigma_1^{\mathbf{g}^j}).$$
(5.1.14)

The states of $\mathcal{H}(\Sigma_1^{\mathbf{g}^j})$ are labelled by anyons $\alpha \in \mathcal{A}$ with the property $\mathbf{g}^j \times \alpha = \alpha$. In particular, j must be proportional to the length of the $\mathbb{Z}_n^{(1)}$ -orbit of α . We shall denote this orbit by $[\alpha]$, and its length $\ell_{\alpha} := |[\alpha]|$ equals the minimal integer such that $\mathbf{g}^{\ell_{\alpha}} \times \alpha = \alpha$. This integer divides n, and any other integer j with $\mathbf{g}^j \times \alpha = \alpha$ is of the form $j = \ell_{\alpha} k$, for some integer $k = 0, 1, \ldots, n/\ell_{\alpha} - 1$. This reproduces the third condition above, namely if a given orbit is shorter than $\ell_{\alpha} = n$, it descends to n/ℓ_{α} copies in the condensed theory. The copies just label the number of insertions of the symmetry defect we use to create the state.

Let us now move on to the spatial circles; insertions of the symmetry elements along these circles shall project into the invariant subspace. In this case, the meaning of *invariant* depends on which cycle we insert the symmetry element on; a symmetry along the a-cycle acts via braiding, and along the b-cycle via fusion. In the end, we must have states that are invariant under g, both with respect to braiding and to fusion. Let us discuss these two cases in turn.

• Take first the a-cycle, which is the circle that is orthogonal to the one we use to create states. Given a state created by a line $\alpha \in \mathcal{A}$ running along the b-cycle, the configuration we obtain by inserting **g** is $B(\alpha, \mathbf{g})|\alpha; \mathbf{g}^{\ell_{\alpha}k}\rangle$ (cf. (5.1.10)). The phase $B(\alpha, \mathbf{g})$ equals $e^{2\pi i q/n}$ for $\alpha \in \mathcal{A}_q$ (cf. (5.1.12)). Summing over all insertions \mathbf{g}^j produces the phase

$$\sum_{j=0}^{n-1} B(\alpha, \mathbf{g}^j) = \sum_{j=1}^{n-1} e^{2\pi i q j/n}$$

= $n \delta_{q,0}$ (5.1.15)

which indeed projects to the states with q = 0, i.e., to $\alpha \in \mathcal{A}_0$. We thus reproduce the first condition, namely the states in the condensed theory must be neutral under $\mathbb{Z}_n^{(1)}$, i.e., taken from \mathcal{A}_0 .

• Finally, if we now consider the second spatial circle, the b-cycle, and insert \mathbf{g}^{j} , we obtain $|\mathbf{g}^{j} \times \alpha; \mathbf{g}^{\ell_{\alpha}k}\rangle$. Summing over all j (and normalizing to have unit norm) leads to

$$|[\alpha], k\rangle := \frac{1}{\sqrt{\ell_{\alpha}}} \sum_{j=0}^{\ell_{\alpha}-1} |g^{j} \times \alpha; g^{k\ell_{\alpha}}\rangle, \qquad [\alpha] \in \mathcal{A}_{0}/\sim, \ k \in \mathbb{Z}_{n/\ell_{\alpha}}.$$
(5.1.16)

which is indeed invariant under \mathbf{g} , where now this symmetry acts via fusion (i.e., $\alpha \mapsto \mathbf{g} \times \alpha$). We thus reproduce the second condition, namely the fact that the anyons of the condensed theory $\hat{\mathcal{A}}$ are labelled by $\mathbb{Z}_n^{(1)}$ orbits.

We thus see that, as expected, the insertions along the three circles indeed reproduce the three conditions we are used to. All in all, a basis of states is labelled by a pair of indices: $[\alpha]$, denoting a $\mathbb{Z}_n^{(1)}$ orbit of neutral lines $\alpha \in \mathcal{A}_0$, plus a degeneracy label taking values in $k = 0, 1, \ldots, n/\ell_{\alpha} - 1$. This degeneracy label, arguably the most subtle ingredient so far, is in fact quite natural from the point of view of gauging $\mathbb{Z}_n^{(1)}$: $k\ell_{\alpha}$ just denotes how many copies of the **g**-puncture we insert in order to create the state, i.e., from which twisted Hilbert space the state comes from.

The presentation of $\hat{\mathcal{H}}(\Sigma_1)$ above also gives us a natural way to compute the modular data of the condensed theory, in particular, the modular matrix \hat{S} . Specifically, a modular transformation acting on a state $|[\alpha]; k\rangle$ is nothing but the S-matrix of the uncondensed theory, in the presence of a puncture $g^{k\ell_{\alpha}}$:

$$\langle [\alpha]; k | \hat{S} | [\alpha']; k' \rangle = \delta_{k\ell_{\alpha}, k'\ell_{\alpha'}} \sqrt{\ell_{\alpha}\ell_{\alpha'}} S_{\alpha, \alpha'}(\mathbf{g}^{k\ell_{\alpha}}) \,.$$
(5.1.17)

The modular matrix in the once-punctured torus can be expressed in terms of the F-symbols of the parent theory, namely [122]

$$S_{\alpha,\alpha'}(\mathbf{g}^j) = \sum_{\beta \in \alpha \times \alpha'} \frac{\theta(\beta)}{\theta(\alpha)\theta(\alpha')} S_{\mathbf{1},\beta} F_{\alpha,\alpha'} \begin{bmatrix} \mathbf{g}^j & \alpha' \\ \alpha & \beta \end{bmatrix} .$$
(5.1.18)

The basis (5.1.16) of $\hat{\mathcal{H}}(\Sigma_1)$ is natural because it makes \hat{S} block-diagonal, but it does not correspond to the anyon basis. The most obvious way to see this is that the would-be quantum dimension $d_{[\alpha];k} = \hat{S}_{[\alpha];k,[1];0}/\hat{S}_{[1];0,[1];0}$ vanishes for $k \neq 0$.

In order to identify the anyon basis we can look at the dual $\mathbb{Z}_n^{(0)}$ symmetry. The charged states are those with the puncture. More precisely, in the diagonal basis the states transform as $\mathbb{Z}_n^{(0)}$: $|[\alpha]; k\rangle \mapsto e^{2\pi i k/\ell_{\alpha}} |[\alpha]; k\rangle$. In the anyon basis, this symmetry should act as a cyclic permutation of the anyons, that is, as $\mathbb{Z}_n^{(0)}$: $|[\alpha]; \hat{k}\rangle \mapsto |[\alpha]; \hat{k} + 1\rangle$ for some label \hat{k} . We conclude that the anyon basis is in fact nothing but the Fourier transform (Pontryagin dual) of the diagonal basis (5.1.16):

$$|[\alpha], \hat{k}\rangle := \frac{1}{\sqrt{n/\ell_{\alpha}}} \sum_{k=0}^{n/\ell_{\alpha}-1} e^{2\pi i \hat{k} k \ell_{\alpha}/n} |[\alpha]; k\rangle$$
$$= \frac{1}{\sqrt{n}} \sum_{k=0}^{n/\ell_{\alpha}-1} \sum_{j=0}^{\ell_{\alpha}-1} e^{2\pi i \hat{k} k \ell_{\alpha}/n} \quad \underbrace{\mathsf{g}}_{\mathbf{g}^{j} \times \alpha}^{\mathbf{g}^{k\ell_{\alpha}}}, \tag{5.1.19}$$

where $\hat{k} \in \mathbb{Z}_{n/\ell_{\alpha}}^{*}$. In this basis, the quantum dimension takes the expected value $d_{[\alpha];\hat{k}} = (\ell_{\alpha}/n)S_{\alpha,\mathbf{1}}/S_{\mathbf{1},\mathbf{1}} = (\ell_{\alpha}/n)d_{\alpha}$. The anyons of the quotient $\hat{\mathcal{A}}$ create the states $|[\alpha], \hat{k}\rangle$ by acting on the vacuum.

Given the matrix \hat{S} in the fusion basis, one can use the Verlinde formula to compute the dimension of the Hilbert space of the condensed theory, for an arbitrary Riemann surface, with an arbitrary number of punctures. In the particular case of no external punctures, the formula only involves matrix elements with the vacuum, in which case the matrix with punctures $S(\mathbf{g}^j)$ does not contribute except for the vacuum insertion, that is, the regular S matrix of the uncondensed theory. In other words, the dimension of the Hilbert space of the condensed theory, in the case of no punctures, can conveniently be computed using only the S matrix of the parent theory, without the need to know the F-symbols:

$$\dim(\hat{\mathcal{H}}(\Sigma_g)) = \sum_{\alpha \in \hat{\mathcal{A}}} \hat{S}_{\mathbf{1},\alpha}^{\chi(\Sigma)}$$
$$\equiv \sum_{\alpha \in \mathcal{A}_0} \frac{n}{\ell_{\alpha}^{2g}} S_{\mathbf{1},\alpha}^{\chi(\Sigma)}.$$
(5.1.20)

It is possible to generalize the expressions above to the case where there is a non-trivial background flux for the $\mathbb{Z}_n^{(0)}$ magnetic symmetry dual to the gauged $\mathbb{Z}_n^{(1)}$ symmetry. For example, if the flux of such background is $q \in \mathbb{Z}_n$, then the states are created from the subset \mathcal{A}_q instead of \mathcal{A}_0 . Summing over all such backgrounds, i.e., gauging the $\mathbb{Z}_n^{(0)}$ symmetry, takes us back to the original ungauged theory \mathcal{A} . We shall not need this generalization here.

5.1.2 Fermion anyon condensation

We now move on to the more interesting case of fermion condensation: we have some bosonic TQFT, and we wish to condense a certain abelian fermion, which we denote by $\psi \in \mathcal{A}$. We can assume without loss of generality that this line generates a $\mathbb{Z}_2^{(1)}$ symmetry, i.e. $\psi^2 = \mathbf{1}$, for otherwise we can first condense the boson $\mathbf{g} = \psi^2$ (as in the previous section), and then condense the resulting fermion, which will satisfy $\psi^2 = \mathbf{1}$.

The philosophy underlying fermion condensation is essentially the same as in boson condensation: we construct the Hilbert space of the condensed theory from the states in the parent theory, perhaps in presence of ψ -punctures. Roughly speaking, the configurations with non-trivial background flux can be thought of as the different spin structures on Σ .

Before actually constructing the spin TQFT by condensing a fermion in a bosonic TQFT, let us discuss what we are to expect from this condensation in the first place. A spin TQFT should assign to manifolds of the form $\mathbb{S}^1 \times \Sigma$ a super-vector space $\hat{\mathcal{H}}(\Sigma)$, which depends only on the topology of Σ , together with its spin structure *s*. (We use a hat to denote the Hilbert space of the condensed theory, and reserve the notation $\mathcal{H}(\Sigma)$ for that of the bosonic parent). Depending on the spin structure on the time circle \mathbb{S}^1 , the partition function computes either the regular trace over $\hat{\mathcal{H}}(\Sigma)$, or the super-trace (i.e., the trace weighted by fermion parity). Specifically, the spin generalization of (5.1.1) is

$$Z(\mathbb{S}^{1}_{\mathrm{NS}} \times \Sigma) = \mathrm{tr}_{\hat{\mathcal{H}}(\Sigma)}(\mathrm{id})$$

$$Z(\mathbb{S}^{1}_{\mathrm{R}} \times \Sigma) = \mathrm{tr}_{\hat{\mathcal{H}}(\Sigma)}(-1)^{F}$$
(5.1.21)

where \mathbb{S}_{NS}^1 denotes the circle with anti-periodic boundary conditions, and \mathbb{S}_{R}^1 the circle with periodic boundary conditions. Therefore, if the super-vector space $\hat{\mathcal{H}}(\Sigma)$ is $\mathbb{C}^{b|f}$, then Neveu-Schwarz boundary conditions compute b + f, and Ramond boundary conditions compute b - f.

We shall denote a compact surface with genus g and spin structure s by $\Sigma_{g;s}$. As in the bosonic case, large diffeomorphisms act unitarily in $\hat{\mathcal{H}}(\Sigma_{g;s})$. The MCG as a spin surface is a subgroup of the MCG as a surface, $MCG(\Sigma_{g;s}) \subseteq MCG(\Sigma_g)$. The reason for this is that some diffeomorphisms that leave Σ invariant as a topological space, actually change the spin structure $s \mapsto s'$, and so do not constitute elements of the MCG as a spin surface. The canonical example is the *T*-transformation on the torus, which performs a Dehn twist around the a-cycle. As such, it maps $(s^a, s^b) \mapsto (s^a, s^a s^b)$. This is an element of the spin MCG if $s^a = +1$, but it is not if $s^a = -1$. On the other hand, T^2 is in the spin MCG for any spin structure.

Elements of the spin MCG act unitarily in the Hilbert space, namely,

$$\operatorname{MCG}(\Sigma_{g;s}) \colon \hat{\mathcal{H}}(\Sigma_{g;s}) \to \hat{\mathcal{H}}(\Sigma_{g;s}).$$
 (5.1.22)

On the other hand, elements of the regular MCG induce isomorphisms of (generically distinct) super-vector spaces,

$$\operatorname{MCG}(\Sigma_g) \colon \hat{\mathcal{H}}(\Sigma_{g;s}) \to \hat{\mathcal{H}}(\Sigma_{g;s'}).$$
 (5.1.23)

This means, for example, that the partition function $Z(\mathbb{S}^1 \times \Sigma_{g;s})$ is invariant under $MCG(\Sigma_g)$; and, more generally, observables only depend on the equivalence class of s under the regular MCG. It is known that there are only two equivalence classes of spin structures modulo MCG, the so-called *even* and *odd* spin structures. These are distinguished by the *Arf invariant* [286]. If two spin structures have the same Arf parity, then there exists some MCG element that maps one into the other. If they have different Arf parity, no such MCG element exists. In conclusion, observables of spin TQFTs depend on s only through Arf(s).

For fixed spin structure, $MCG(\Sigma_{g;s})$ is represented by a unitary operator in $\hat{\mathcal{H}}(\Sigma_{g;s})$. That being said, due to the \mathbb{Z}_2 grading of this vector space, this action typically gets extended. Namely, the Hilbert space of spin TQFTs realize a unitary representation of a certain nontrivial \mathbb{Z}_2 extension of the spin MCG. To be explicit, (modding out by Torelli, i.e., working in homology) the MCG of Σ_g is the integral symplectic group $Sp_g(\mathbb{Z})$, and the spin MCG is some subgroup thereof. The Hilbert space of the theory realizes a unitary representation of the so-called *metaplectic group* $Mp_g(\mathbb{Z})$, which is defined as the (essentially unique) \mathbb{Z}_2 extension of the symplectic group

$$\mathbb{Z}_2 \hookrightarrow Mp_g(\mathbb{Z}) \twoheadrightarrow Sp_g(\mathbb{Z}). \tag{5.1.24}$$

This extension corresponds to the fact that a 2π rotation is represented by the trivial element in $MCG(\Sigma_{g;s})$, while it lifts to $(-1)^F$ in $\hat{\mathcal{H}}(\Sigma_{g;s})$.

In order to illustrate these ideas it proves useful to focus on the torus, Σ_1 . There are four spin tori, depending on the boundary conditions on the two spatial circles $\Sigma_{1;s^a,s^b} = \mathbb{S}^1_{s^a} \times \mathbb{S}^1_{s^b}$:

$$(-1)^{\operatorname{Arf}(s^{\mathfrak{s}},s^{\mathfrak{b}})} = +1: \qquad \begin{cases} \mathbb{S}_{\operatorname{NS}}^{1} \times \mathbb{S}_{\operatorname{NS}}^{1} \\ \mathbb{S}_{\operatorname{NS}}^{1} \times \mathbb{S}_{\operatorname{R}}^{1} \\ \mathbb{S}_{\operatorname{R}}^{1} \times \mathbb{S}_{\operatorname{NS}}^{1} \\ \end{array}$$

$$(-1)^{\operatorname{Arf}(s^{\mathfrak{s}},s^{\mathfrak{b}})} = -1: \qquad \mathbb{S}_{\operatorname{R}}^{1} \times \mathbb{S}_{\operatorname{R}}^{1}$$

$$(5.1.25)$$

The MCG of the torus is the modular group $SL_2(\mathbb{Z}) = \langle \hat{S}, \hat{T} \rangle$, acting as

$$\begin{array}{l}
S \colon \Sigma_{1;s^{a},s^{b}} \to \Sigma_{1;s^{b},s^{a}} \\
\hat{T} \colon \Sigma_{1;s^{a},s^{b}} \to \Sigma_{1;s^{a},s^{a}s^{b}}
\end{array}$$
(5.1.26)

Therefore, the subgroup that fixes each spin structure is

$$MCG(\Sigma_{1;--}) = \langle \hat{S}, \hat{T}^2 \rangle$$

$$MCG(\Sigma_{1;-+}) = \langle \hat{S}\hat{T}\hat{S}, \hat{T}^2 \rangle$$

$$MCG(\Sigma_{1;+-}) = \langle \hat{S}\hat{T}^2\hat{S}, \hat{T} \rangle$$

$$MCG(\Sigma_{1;++}) = \langle \hat{S}, \hat{T} \rangle$$

$$(5.1.27)$$

Needless to say, the first three groups are all isomorphic, as they are related through $SL_2(\mathbb{Z})$ conjugation:

$$\langle \hat{S}\hat{T}\hat{S}, \hat{T}^2 \rangle = \hat{T}\langle \hat{S}, \hat{T}^2 \rangle \hat{T}^{-1}, \qquad \langle \hat{S}\hat{T}^2\hat{S}, \hat{T} \rangle = (\hat{S}\hat{T})\langle \hat{S}, \hat{T}^2 \rangle (\hat{S}\hat{T})^{-1}.$$
(5.1.28)

This group is a congruence subgroup of $SL_2(\mathbb{Z})$ of index 3, usually denoted by $\Gamma_0(2)$. The fourth group, on the other hand, is $SL_2(\mathbb{Z})$ itself.

The diffeomorphism \hat{S}^4 corresponds to a 2π rotation and, as such, acts trivially in a bosonic theory and so is represented by the identity element in $SL_2(\mathbb{Z})$; conversely, in a fermionic theory it is represented by $(-1)^F$. Thus, modular transformations in spin theories satisfy

$$\hat{S}^2 = (\hat{S}\hat{T})^3, \qquad \hat{S}^4 = (-1)^F, \qquad (5.1.29)$$

with $(-1)^F$ an order 2 central element. These relations define the group $Mp_1(\mathbb{Z})$.⁹⁶

⁹⁶One should keep in mind that $Mp_1(\mathbb{Z})$ does not act faithfully in $\hat{\mathcal{H}}(\Sigma_{1;s})$ (in fact, the metaplectic group is not a matrix group; it does not admit faithful finite-dimensional representations). This fact is most drastic when the theory, for whatever reason, has no fermionic states at all: in such cases, the actions of $Sp_1(\mathbb{Z})$ and $Mp_1(\mathbb{Z})$ are indistinguishable, inasmuch as $(-1)^F$ is trivial. For example, the theory lacks fermionic states if s is an even spin structure, or if the theory is secretly bosonic (through some non-trivial duality). In such cases, one can think of the modular group as being $SL_2(\mathbb{Z})$ instead of $Mp_1(\mathbb{Z})$: their difference is invisible in the Hilbert space anyway. Similar considerations hold in higher genus.

The Hilbert spaces $\mathcal{H}(\Sigma_{g;s})$ are best understood by giving an explicit basis for them. As in the previous section, the Hilbert space of the fermionic theory can be constructed by condensing a fermion in the bosonic parent. Indeed, the Hilbert space of the spin theory is a subspace of the Hilbert space of the bosonic parent, together with the space with a ψ -puncture,

$$\hat{\mathcal{H}}(\Sigma_{g;s}) \subseteq \mathcal{H}(\Sigma_g) \oplus \mathcal{H}(\Sigma_g^{\psi}).$$
(5.1.30)

Let us write down a basis for $\hat{\mathcal{H}}(\Sigma_{1;s})$ in terms of the states of the bosonic parent. Recall that the states on the torus in the bosonic theory are labelled by the anyons \mathcal{A} . The $\mathbb{Z}_2^{(1)}$ symmetry generated by ψ partitions the spectrum \mathcal{A} into two equivalence classes, distinguished by the braiding phase $B(\psi, \cdot) = \pm 1$. In the case of boson condensation we denoted these two equivalence classes by \mathcal{A}_0 and \mathcal{A}_1 ; in the present context it is more natural to denote them by \mathcal{A}_{NS} and \mathcal{A}_{R} :

$$\mathcal{A} = \mathcal{A}_{\rm NS} \sqcup \mathcal{A}_{\rm R}, \qquad \begin{cases} \mathcal{A}_{\rm NS} := \{ \alpha \in \mathcal{A} \mid B(\alpha, \psi) = +1 \} \\ \mathcal{A}_{\rm R} := \{ \alpha \in \mathcal{A} \mid B(\alpha, \psi) = -1 \} \end{cases}$$
(5.1.31)

The two equivalence classes are further partitioned according to the length of the orbits. For a generic $\mathbb{Z}_n^{(1)}$ symmetry, the orbits come in lengths that divide n; for n = 2, we have two-dimensional orbits and one-dimensional ones. We refer to the latter as *Majorana lines*. It is easy to convince oneself that these can only appear in \mathcal{A}_R . We shall use the label α to denote generic lines of \mathcal{A} ; on the other hand, lines of \mathcal{A}_{NS} will be denoted by the more specific label a, while two-dimensional orbits of \mathcal{A}_R by x and one-dimensional ones by m. We will say that α is a-type, x-type, or m-type, according to this classification:

$$a \in \mathcal{A}_{\rm NS}$$

$$x \in \mathcal{A}_{\rm R} \quad \& \quad |x| = 2 \qquad (5.1.32)$$

$$m \in \mathcal{A}_{\rm R} \quad \& \quad |m| = 1.$$

In other words, x-lines satisfy $x \times \psi \neq x$, while m-lines satisfy $m \times \psi = m$.

As in the previous section, we take the bosonic theory, and condense a fermion ψ . In the condensed theory, the Wilson line of ψ becomes almost trivial: it should be represented by the identity operator, up to a sign, depending on the spin structure around the cycle it is supported on. In other words, the anyon ψ represents the wordline of a local fermion. This determines how the states in the condensed phase are obtained in terms of those of the uncondensed one. We claim that the basis of $\mathcal{H}(\Sigma_{1,s})$ can be taken as

$$\hat{\mathcal{H}}(\Sigma_{1;--}) = \operatorname{Span}_{a \in \mathcal{A}_{NS}} \left[\frac{1}{\sqrt{2}} \left(\left| \bigcirc_{a} \right\rangle + \left| \bigotimes_{a \times \psi} \right\rangle \right) \right] \\
\hat{\mathcal{H}}(\Sigma_{1;-+}) = \operatorname{Span}_{a \in \mathcal{A}_{NS}} \left[\frac{1}{\sqrt{2}} \left(\left| \bigcirc_{a} \right\rangle - \left| \bigotimes_{a \times \psi} \right\rangle \right) \right] \\
\hat{\mathcal{H}}(\Sigma_{1;+-}) = \operatorname{Span}_{x \in \mathcal{A}_{R}} \left[\frac{1}{\sqrt{2}} \left(\left| \bigcirc_{x} \right\rangle + \left| \bigotimes_{x \times \psi} \right\rangle \right) \right] \oplus \operatorname{Span}_{m \in \mathcal{A}_{R}} \left[\left| \bigcirc_{m} \right\rangle \right] \\
\hat{\mathcal{H}}(\Sigma_{1;++}) = \operatorname{Span}_{x \in \mathcal{A}_{R}} \left[\frac{1}{\sqrt{2}} \left(\left| \bigcirc_{x} \right\rangle - \left| \bigotimes_{x \times \psi} \right\rangle \right) \right] \oplus \operatorname{Span}_{m \in \mathcal{A}_{R}} \left[\left| \psi \bigoplus_{m} \right\rangle \right]$$
(5.1.33)

The reasoning behind the construction of this basis is the same as in the case of boson condensation. Namely, the gauged theory is obtained by inserting ψ^j in all possible ways. Here ψ is of order two, so there are only two possible blocks: j = 0 or j = 1, i.e., no insertion, or a single ψ -insertion. Furthermore, the specific linear combination of states is decided by the spin structure. For example, inserting ψ along the **a**-cycle inserts the phase $|\alpha; \psi^j\rangle \mapsto B(\alpha, \psi) |\alpha; \psi^j\rangle$. This should reproduce the sign s^a , which means that $\mathcal{A}_{\rm NS}$ lines create states in $s^a = -1$ boundary conditions, and $\mathcal{A}_{\rm R}$ lines create states in $s^a = +1$ boundary conditions. This explains why the basis is constructed using *a*-type lines in $\hat{\mathcal{H}}(\Sigma_{1;-,\bullet})$, and *x*and *m*-type lines in $\hat{\mathcal{H}}(\Sigma_{1;+,\bullet})$.

Similarly, inserting ψ along the b-cycle fuses the state into $|\alpha; \psi^j \rangle \mapsto (-1)^j |\psi \times \alpha; \psi^j \rangle$. If this is to reproduce the boundary condition $s^{\mathbf{b}}$, we are required to consider the linear combination $|\alpha\rangle - s^{\mathbf{b}} |\psi \times \alpha\rangle$ for two-dimensional orbits; and, for Majorana lines, the puncture should be present if and only if $s^{\mathbf{b}} = +1$.

Note that, unlike in the case of bosonic condensation, here the multiple copies associated to the short orbits live in different spaces. Moreover, there are no short orbits in the NS sector, so the observables of the theories (the Wilson lines associated to the NS anyons) do not require fixed-point resolution. In this sense, the fusion rules of the condensed theory are inherited from those of the parent in a straightforward manner, without the need of knowing the once-punctured S-matrix. In the bosonic case, the fusion rules of the short orbits do require this extra structure.

As a consistency check for the basis above, we can easily show that modular transformations map the different Hilbert spaces as expected. For example, take $\hat{\mathcal{H}}(\Sigma_{1;--})$, and apply an *S*-transformation:

$$\frac{1}{\sqrt{2}}(|a\rangle + |\psi \times a\rangle) \xrightarrow{S} \sum_{\alpha' \in \mathcal{A}} \frac{1}{\sqrt{2}} (S_{a,\alpha'} + S_{\psi \times a,\alpha'}) |\alpha'\rangle$$
(5.1.34)

As in equation (5.1.13), we have $S_{\psi \times a,\alpha'} = B(\psi, \alpha')S_{a,\alpha'}$, which means that we may restrict the sum over α' to *a*-type lines, for the Ramond ones do not contribute – they cancel out pairwise. With this,

$$\frac{1}{\sqrt{2}}(|a\rangle + |\psi \times a\rangle) \xrightarrow{S} \sum_{a' \in \mathcal{A}_{\rm NS}/\sim} 2S_{a,a'} \frac{1}{\sqrt{2}}(|a'\rangle + |\psi \times a'\rangle)$$
(5.1.35)

which shows that S maps $\hat{\mathcal{H}}(\Sigma_{1;--})$ to itself, as expected (cf. (5.1.26)). Similarly, T-transformations map

$$\frac{1}{\sqrt{2}}(|a\rangle + |\psi \times a\rangle) \stackrel{T}{\mapsto} e^{-2\pi i c/24} \frac{1}{\sqrt{2}}(\theta(a)|a\rangle + \theta(\psi \times a)|\psi \times a\rangle)$$
(5.1.36)

Noting that $\theta(\psi \times a) = -\theta(a)$, this becomes

$$\frac{1}{\sqrt{2}}(|a\rangle + |\psi \times a\rangle) \xrightarrow{T} e^{-2\pi i c/24} \theta(a) \frac{1}{\sqrt{2}}(|a\rangle - |\psi \times a\rangle)$$
(5.1.37)

which shows that T maps $\hat{\mathcal{H}}(\Sigma_{1;--})$ into $\hat{\mathcal{H}}(\Sigma_{1;-+})$, again as expected (cf. (5.1.26)).

The other three Hilbert spaces can also be seen to transform into each other in the expected manner. Not only that, but the exercise gives us the explicit expression for the \hat{S} and \hat{T} matrices of the condensed theory:

$$\hat{S}: \hat{\mathcal{H}}(\Sigma_{1;--}) \to \hat{\mathcal{H}}(\Sigma_{1;--}) \implies \left\{ \hat{S}_{a,a'} = 2S_{a,a'} \\ \hat{S}: \hat{\mathcal{H}}(\Sigma_{1;-+}) \to \hat{\mathcal{H}}(\Sigma_{1;+-}) \implies \left\{ \hat{S}_{a,x} = 2S_{a,x} \\ \hat{S}_{a,m} = \sqrt{2}S_{a,m} \\ \hat{S}: \hat{\mathcal{H}}(\Sigma_{1;+-}) \to \hat{\mathcal{H}}(\Sigma_{1;-+}) \implies \left\{ \hat{S}_{x,a} = 2S_{x,a} \\ \hat{S}_{m,a} = \sqrt{2}S_{m,a} \\ \hat{S}_{m,a} = \sqrt{2}S_{m,a} \\ \hat{S}_{x,m'} = 2S_{x,x'} \\ \hat{S}_{x,m} = 0 \\ \hat{S}_{m,m'} = S_{m,m'}(\psi) \\ \end{array} \right.$$
(5.1.38)

where $S_{m,m'}(\psi)$ denotes the S-matrix of the bosonic parent, in the once-punctured torus

(cf. (5.1.18)). The \hat{T} -matrix is given by a similar expression:

$$\hat{T}: \hat{\mathcal{H}}(\Sigma_{1;--}) \rightarrow \hat{\mathcal{H}}(\Sigma_{1;-+}) \implies \left\{ \hat{T}_{a,a'} = e^{-2\pi i c/24} \theta(a) (\delta_{a,a'} - \delta_{a,\psi \times a'}) \\
\hat{T}: \hat{\mathcal{H}}(\Sigma_{1;+-}) \rightarrow \hat{\mathcal{H}}(\Sigma_{1;--}) \implies \left\{ \hat{T}_{a,a'} = e^{-2\pi i c/24} \theta(a) (\delta_{a,a'} + \delta_{a,\psi \times a'}) \\
\hat{T}: \hat{\mathcal{H}}(\Sigma_{1;+-}) \rightarrow \hat{\mathcal{H}}(\Sigma_{1;+-}) \implies \left\{ \hat{T}_{x,x'} = e^{-2\pi i c/24} \theta(a) (\delta_{x,x'} + \delta_{x,\psi \times x'}) \\
\hat{T}_{x,m} = 0 \\
\hat{T}_{m,m'} = e^{-2\pi i c/24} \theta(m) \delta_{m,m'} \\
\hat{T}_{x,m} = 0 \\
\hat{T}_{m,m'} = e^{-2\pi i c/24} \theta(m) \delta_{m,m'}$$
(5.1.39)

Finally, we discuss the third generator of the spin modular group, fermion parity. This zero-form symmetry is the dual symmetry to the gauged $\mathbb{Z}_2^{(1)}$, which means that the states with odd fermion parity are those that carry the puncture. This means that $(-1)^F = 1$ in all even spin structures, while in the R-R sector one has

$$((-1)^F)_{x,x'} = +\delta_{x,x'} ((-1)^F)_{m,m'} = -\delta_{m,m'}$$
(5.1.40)

These matrices are unitary, symmetric, and satisfy $\hat{S}^4 = (-1)^F$ and $\hat{S}^2 = (\hat{S}\hat{T})^3$. It is important to remark that these properties are understood in the \mathbb{Z}_2 -graded sense, i.e., taking into account (5.1.26). In other words, the precise relations are

$$\hat{S}^{\dagger}_{s^{a},s^{b}} = \hat{S}^{-1}_{s^{a},s^{b}}
\hat{T}^{\dagger}_{s^{a},s^{b}} = \hat{T}^{-1}_{s^{a},s^{b}}
\hat{S}^{t}_{s^{b},s^{a}} = \hat{S}_{s^{a},s^{b}}
(\hat{S}_{s^{b},s^{a}}\hat{S}_{s^{a},s^{b}})^{2} = (-1)^{F}_{s^{a},s^{b}}
\hat{S}_{s^{b},s^{a}}\hat{S}_{s^{a},s^{b}} = \hat{S}_{s^{b},s^{a}}\hat{T}_{s^{b},s^{a}s^{b}}\hat{S}_{s^{a}s^{b},s^{b}}\hat{T}_{s^{a}s^{b},s^{a}}\hat{S}_{s^{a},s^{a}s^{b}}\hat{T}_{s^{a},s^{b}},$$
(5.1.41)

where $\mathcal{O}_{s^{a},s^{b}}$ denotes the operator $\mathcal{O} \in \{\hat{S}, \hat{T}, (-1)^{F}\}$ when acting on $\hat{\mathcal{H}}(\Sigma_{1;s^{a},s^{b}})$.

In section 5.2 we construct several examples of quotient TQFTs, namely $SO(N)_k$ with k = 1, 2, 3. Some of these illustrate bosonic anyon condensation, and some others fermionic anyon condensation.

One final comment is in order: it is important to stress that the line operators in the fermionic theory – the anyons – are the NS-lines. The R-lines, on the other hand, are not genuine line operators: they live at the end of the topological surface that implements the $(-1)^F$ symmetry. Indeed, moving a local fermion around a Ramond line generates a minus

sign:



This is analogous to the situation in two-dimensional fermionic CFTs, where Ramond operators are not genuine point operators, but rather exist at the end of the $(-1)^{F}$ -line (see e.g. [287]).

5.2 Examples of anyon condensation

Here we collect some extra examples of boson and fermion condensation, using theories of the form $SO(n)_k$ for small values of k. In particular, k = 1 and k = 3 exemplify fermion condensation, and k = 2 boson condensation. The case k = 1, i.e., $SO(n)_1$, is the generator of fermionic SPTs with no symmetry, and so is a key theory in the study of fermionic TQFTs. The case k = 2 will be related to U(1) theories, through the level-rank duality $SO(n)_2 \leftrightarrow SO(2)_{-n} \equiv U(1)_{-n}$. Finally, the case k = 3 will be constructed through the SU(2) theory, thanks to the level-rank duality $SO(n)_3 \leftrightarrow SO(3)_{-n} \equiv SU(2)_{-2n}/\mathbb{Z}_2$. We also include the case of U(1)_k separately, this time focusing on its time-reversal invariance.

5.2.1 SO $(n)_1$

This is the minimal spin TQFT, and it has central charge n/2, so corresponds to n boundary Majorana fermions. A single fermion, $SO(1)_1$, is the generator of the group of fermionic SPTs with no extra symmetries, $\Omega_{\text{spin}}^4 = \mathbb{Z}$. In other words, any invertible fermionic phase is equivalent to $SO(n)_1$ for some n. The theory can also be written as n copies of (the inverse of) the gravitational Chern-Simons theory.

The bosonic parent of this theory is $\text{Spin}(n)_1$. The details of this theory depend on the parity of n.

n = 2m + 1. One can construct SO(n)₁ by condensing the fermion in the Ising category. The modular data for the parent theory is that of $\text{Ising}_m = \text{Spin}(2m + 1)_1$, which has three anyons:

$$\mathbf{1} = [1, 0, 0, \dots, 0, 0]$$

$$\sigma = [0, 0, 0, \dots, 0, 1]$$

$$\psi = [0, 1, 0, \dots, 0, 0]$$

(5.2.1)

where $[\lambda_0, \lambda_1, \ldots, \lambda_m]$ denote the extended Dynkin labels of the representation. These lines fuse according to $\psi \times \sigma = \sigma$ and $\sigma^2 = \mathbf{1} + \psi$, and transform under modular transformations as follows:

$$S|\mathbf{1}\rangle = \frac{1}{2}|\mathbf{1}\rangle + \frac{1}{\sqrt{2}}|\sigma\rangle + \frac{1}{2}|\psi\rangle \qquad T|\mathbf{1}\rangle = e^{2\pi i \left(0 - \frac{n}{48}\right)}|\mathbf{1}\rangle$$

$$S|\sigma\rangle = \frac{1}{\sqrt{2}}|\mathbf{1}\rangle - \frac{1}{\sqrt{2}}|\psi\rangle \qquad T|\sigma\rangle = e^{2\pi i \left(\frac{n}{16} - \frac{n}{48}\right)}|\sigma\rangle \qquad (5.2.2)$$

$$S|\psi\rangle = \frac{1}{2}|\mathbf{1}\rangle - \frac{1}{\sqrt{2}}|\sigma\rangle + \frac{1}{2}|\psi\rangle \qquad T|\psi\rangle = e^{2\pi i \left(\frac{1}{2} - \frac{n}{48}\right)}|\psi\rangle$$

The lines are partitioned according to their braiding with respect to ψ as

NS:
$$\mathcal{A}_{NS} = \{\mathbf{1}, \psi\}$$

R: $\mathcal{A}_{R} = \{\sigma\}$
(5.2.3)

and they are paired-up under fusion as

$$\mathbf{1} \stackrel{\times \psi}{\longleftrightarrow} \psi, \qquad \sigma \stackrel{\checkmark}{\searrow} \times \psi \quad . \tag{5.2.4}$$

Therefore, the four Hilbert spaces of the theory are

• If we take **NS-NS** boundary conditions, the state is

$$|0; \text{NS-NS}\rangle = \frac{1}{\sqrt{2}} (|\mathbf{1}\rangle + |\psi\rangle).$$
 (5.2.5)

• If we take **NS-R** boundary conditions, the state is

$$|0; \text{NS-R}\rangle = \frac{1}{\sqrt{2}} (|\mathbf{1}\rangle - |\psi\rangle). \qquad (5.2.6)$$

• If we take **R-NS** boundary conditions, the state is

$$|0; \text{R-NS}\rangle = |\sigma\rangle. \tag{5.2.7}$$

• If we take **R-R** boundary conditions, the state is

$$|0; \mathrm{R-R}\rangle = |\sigma; \psi\rangle, \qquad (5.2.8)$$

where, we remind the reader, $|\alpha; \beta\rangle$ denotes the anyon α in presence of a β puncture (cf. (5.1.6)).

We see that these spaces are all one-dimensional, as expected from an invertible theory. Furthermore, all states are bosonic, except for the one with a puncture, $|\sigma;\psi\rangle$, which means that $(-1)^F = (-1)^{\text{Arf}(s)}$.

The modular data of the quotient can be computed in a straightforward manner. The only non-trivial case is the S matrix in the R-R sector, which has a puncture. We can compute this matrix element using the general formula (5.1.18), namely

$$S_{\sigma,\sigma}(\psi) = \sum_{\alpha=\mathbf{1},\psi} \frac{\theta(\alpha)}{\theta(\sigma)^2} S_{\mathbf{1},\alpha} F_{\sigma,\sigma} \begin{bmatrix} \psi & \sigma \\ \sigma & \alpha \end{bmatrix}$$
(5.2.9)

The *F*-symbols of the Ising category are well-known, cf. $F(\alpha = 1) \equiv +1$ and $F(\alpha = \psi) \equiv -1$. With this,

$$\hat{S}_{\text{R-R}} = \frac{1}{2} (-1)^{3/4} (-i)^m (F(\psi) - F(\mathbf{1})) \equiv e^{i\pi (6m+7)/4}$$
(5.2.10)

This result, together with $\hat{T}_{\text{R-R}} = e^{\pi i (2m+1)/12}$, confirms that the theory satisfies the expected modularity relations, $S^2 = (ST)^3$ and $S^4 = (-1)^F$.

A different perspective yields the same answer. The CFT SO(n)₁ is identical to n free Majorana fermions, and the once-puctured conformal block $|\sigma;\psi\rangle$ is nothing but the torus one-point function of ψ . The insertion of ψ removes the Ramond zero-mode, and hence this one-point function is $\langle\psi\rangle = q^{1/24} \prod_{r=1}^{\infty} (1-q^r) \equiv \eta(\tau)$, the Dedekind eta function. The punctured S-matrix is nothing but the phase acquired by $\langle\psi\rangle$ under an S-transformation, namely $\eta(-1/\tau) = \sqrt{-i\tau}\eta(\tau)$. The factor of $\tau^{1/2}$ is the weight associated to a primary of spin h = 1/2, while the factor of $\sqrt{-i}$ is the sought-after S-matrix. For a system of n = 2m + 1fermions, the S-matrix is $(\sqrt{-i})^{2m+1} \equiv e^{i\pi(6m+7)/4}$, in agreement with (5.2.10).

n = 2m. Here the bosonic parent $\text{Spin}(2m)_1$ has four lines, the trivial representation, the vector representation, and the two spinor representations. The quotient is obtained by condensing the vector. The Lie algebra is simply-laced, which automatically implies that the fusion rules are abelian, and so there are no fixed-points under fusion. Therefore, all states are bosonic. The four lines are split into two NS-lines (the trivial and the vector) and two R-lines (the two spinors), and each pair belongs to a two-dimensional orbit. This means that each Hilbert space is one-dimensional, as expected from an invertible theory, and moreover all states have $(-1)^F = +1$. The modular data is trivially computed, given that there are no short orbits.

5.2.2 SO $(n)_2$

Here we illustrate the construction of the bosonic theory $SO(n)_2$, by condensing an abelian boson in $Spin(n)_2$. We focus in particular on the odd-*n* case, where all the modular data – especially the *F*-symbols – is fully known [288]. We follow the notation therein. Consider the algebra $\mathfrak{so}_{2n+1} = B_n$. Its comarks are $a^{\vee} = 1, 2, 2, \ldots, 2, 1$, which means that the theory $\operatorname{Spin}(2n+1)_2$ has n+4 lines. We denote them as $\mathbf{1}, \epsilon, \phi_i, \psi_{\pm}$, with $i = 1, 2, \ldots, n$. The corresponding affine Dynkin labels are as follows:

$$\mathbf{1} = [2, 0, 0, \dots, 0]
\epsilon = [0, 2, 0, \dots, 0]
\phi_1 = [1, 1, 0, \dots, 0]
\phi_i = [0, \dots, 0, 1, 0, \dots, 0]$$
at position $i + 1$ (5.2.11)
 $\phi_n = [0, 0, \dots, 0, 2]
\psi_+ = [1, 0, 0, \dots, 0, 1]
\psi_- = [0, 1, 0, \dots, 0, 1]$

The S-matrix reads

$$S_{1,1} = S_{1,\epsilon} = S_{\epsilon,\epsilon} = \frac{1}{2\sqrt{2n+1}}$$

$$S_{1,\psi_{\pm}} = +\frac{1}{2}$$

$$S_{\epsilon,\psi_{\pm}} = -\frac{1}{2}$$

$$S_{\psi_{s},\psi_{s'}} = \frac{1}{2}ss'$$

$$S_{1,\phi_{i}} = S_{\epsilon,\phi_{i}} = \frac{1}{\sqrt{2n+1}}$$

$$S_{\psi_{\pm},\phi_{i}} = 0$$

$$S_{\phi_{i},\phi_{j}} = \frac{2}{\sqrt{2n+1}}\cos\frac{2\pi i j}{2n+1}$$
(5.2.12)

and the spins are

$$h_{1} = 0$$

$$h_{\epsilon} = 1$$

$$h_{\phi_{i}} = \frac{1}{2} \frac{i(2n+1-i)}{2n+1}$$

$$h_{\psi_{+}} = \frac{1}{8}n$$

$$h_{\psi_{-}} = \frac{1}{8}n + \frac{1}{2}$$
(5.2.13)

From this one derives the quantum dimensions

$$d_{1} = 1$$

$$d_{\epsilon} = 1$$

$$d_{\phi_{i}} = 2$$

$$d_{\psi_{\pm}} = \sqrt{2n+1}$$
(5.2.14)

and fusion rules,

$$\epsilon \times \epsilon = \mathbf{1}, \quad \psi_{\pm} \times \psi_{\pm} = \mathbf{1} + \sum_{j=1}^{n} \phi_{j}, \quad \psi_{\pm} \times \psi_{\mp} = \epsilon + \sum_{j=1}^{n} \phi_{j}$$

$$\epsilon \times \phi_{i} = \phi_{i}, \quad \epsilon \times \psi_{\pm} = \psi_{\mp}, \quad \phi_{i} \times \psi_{\pm} = \psi_{\pm} + \psi_{\mp}$$

$$\phi_{i} \times \phi_{i} = \mathbf{1} + \epsilon + \phi_{g(2i)}, \quad \phi_{i} \times \phi_{j} = \phi_{g(i-j)} \times \phi_{g(i+j)}, \quad i > j$$
(5.2.15)

where g(i) = i if $1 \le i \le n$ and g(i) = 2n + 1 - i otherwise.

We see that there are no multiplicities, and all anyons are self-conjugate. We also note that ϵ is condensable, which leads to the bosonic theory $SO(2n + 1)_2$. Let us analyse the quotient explicitly.

By looking at the braiding phase $B(\alpha, \epsilon)$ one learns that the unscreened anyons are $\mathbf{1}, \epsilon, \phi_i$, while the screened anyons are ψ_{\pm} . Moreover, $\mathbf{1}$ and ϵ are in the same orbit, while all the ϕ_i are fixed points. Thus, a basis for the condensed Hilbert space is as follows:

$$|0\rangle = \frac{1}{\sqrt{2}} (|1\rangle + |\epsilon\rangle)$$

$$|i\rangle = |\phi_i\rangle$$

$$|n+i\rangle = |\phi_i; \epsilon\rangle$$

(5.2.16)

for i = 1, 2, ..., n, and where $|\cdot; \epsilon\rangle$ denotes a state in the once-punctured torus. The condensed S-matrix is

$$\hat{S}_{0,0} = 2S_{1,1} = \frac{1}{\sqrt{2n+1}}$$

$$\hat{S}_{0,i} = \sqrt{2}S_{1,\phi_i} = \sqrt{\frac{2}{2n+1}}$$

$$\hat{S}_{0,n+i} = 0$$

$$\hat{S}_{i,j} = S_{\phi_i,\phi_j} = \frac{2}{\sqrt{2n+1}}\cos\frac{2\pi i j}{2n+1}$$

$$\hat{S}_{i,j+n} = 0$$

$$\hat{S}_{i+n,j+n} = S_{\phi_i,\phi_j}(\epsilon)$$
(5.2.17)

where $S_{\phi_i,\phi_j}(\epsilon)$ is the S-matrix of $\text{Spin}(2n+1)_2$ in the presence of a puncture. This matrix

element can be obtain as in (5.1.18). For example, we compute

$$S_{\phi_{i},\phi_{i}}(\epsilon) = \sum_{\beta \in \phi_{i} \times \phi_{i}} \frac{\theta(\beta)}{\theta(\phi_{i})^{2}} S_{\mathbf{1},\beta} F_{\phi_{i},\phi_{i}} \begin{bmatrix} \epsilon & \phi_{i} \\ \phi_{i} & \beta \end{bmatrix}$$

$$= e^{-2\pi i \frac{i(2n+1-i)}{2n+1}} \frac{1}{\sqrt{2n+1}} \times$$

$$\left(\frac{1}{2} \underbrace{F_{\phi_{i},\phi_{i}}}_{+1} \begin{bmatrix} \epsilon & \phi_{i} \\ \phi_{i} & \mathbf{1} \end{bmatrix} + \frac{1}{2} \underbrace{F_{\phi_{i},\phi_{i}}}_{+1} \underbrace{\begin{bmatrix} \epsilon & \phi_{i} \\ \phi_{i} & \epsilon \end{bmatrix}}_{+1} + e^{2\pi i (\frac{1}{2} \frac{2i(2n+1-2i)}{2n+1})} \underbrace{F_{\phi_{i},\phi_{i}}}_{-1} \begin{bmatrix} \epsilon & \phi_{i} \\ \phi_{i} & \phi_{g(2i)} \end{bmatrix}}_{-1}\right) \quad (5.2.18)$$

$$= \frac{2i}{\sqrt{2n+1}} \sin \frac{2\pi i^{2}}{2n+1}$$

while for $i \neq j$,

$$S_{\phi_{i},\phi_{j}}(\epsilon) = \sum_{\beta \in \phi_{i} \times \phi_{j}} \frac{\theta(\beta)}{\theta(\phi_{i})\theta(\phi_{j})} S_{\mathbf{1},\beta} F_{\phi_{i},\phi_{j}} \begin{bmatrix} \epsilon & \phi_{j} \\ \phi_{i} & \beta \end{bmatrix}$$
$$= \frac{1}{\sqrt{2n+1}} \left(e^{\frac{2i\pi i j}{2n+1}} \underbrace{F_{\phi_{i},\phi_{j}} \begin{bmatrix} \epsilon & \phi_{j} \\ \phi_{i} & \phi_{g(i-j)} \end{bmatrix}}_{+1} + e^{-\frac{2i\pi i j}{2n+1}} \underbrace{F_{\phi_{i},\phi_{j}} \begin{bmatrix} \epsilon & \phi_{j} \\ \phi_{i} & \phi_{g(i+j)} \end{bmatrix}}_{-1} \right)$$
(5.2.19)
$$= \frac{2i}{\sqrt{2n+1}} \sin \frac{2\pi i j}{2n+1}$$

All in all, the S-matrix of the quotient takes the form

$$\hat{S} = \frac{1}{\sqrt{2n+1}} \begin{pmatrix} 1 & \sqrt{2} & 0 \\ \hline \sqrt{2} & 2\cos\frac{2\pi i j}{2n+1} & 0 \\ \hline 0 & 0 & 2i\sin\frac{2\pi i j}{2n+1} \end{pmatrix} \leftarrow |i\rangle \qquad (5.2.20)$$

One can easily check that this matrix is unitary, and satisfies the algebra of the (bosonic) modular group, $S^2 = (ST)^3$, $S^4 = 1$.

In order to write down the fusion rules of the quotient we have to switch into the fusion basis, namely

$$|\phi_i, \pm\rangle = \frac{1}{\sqrt{2}} (|\phi_i\rangle \pm |\phi_i; \epsilon\rangle) . \qquad (5.2.21)$$

In this basis, the S-matrix becomes $\hat{S}_{ij} \sim \frac{1}{\sqrt{2n+1}} e^{2\pi i \frac{ij}{2n+1}}$. This is in agreement with the level-rank duality $\mathrm{SO}(n)_2 \sim \mathrm{SO}(2)_{-n} = \mathrm{U}(1)_{-n}$, where the S-matrix of $\mathrm{U}(1)_k$ is $e^{-2\pi i \alpha \beta/k}/\sqrt{k}$. (Here ~ denotes duality modulo $\{\mathbf{1}, \psi\}$, since $\mathrm{U}(1)_k$ is spin for odd k.)

5.2.3 $SO(n)_3$

We construct the theory using level-rank duality $SO(n)_3 = SO(3)_{-n} = SU(2)_{-2n}/\mathbb{Z}_2$. So we consider $SU(2)_k$ first.

There are k + 1 lines, which we label as $j = 0, \frac{1}{2}, 1, \dots, \frac{1}{2}k$. The S-matrix reads

$$S_{ij} = \sqrt{\frac{2}{k+2}} \sin \frac{\pi (2i+1)(2j+1)}{k+2}$$
(5.2.22)

and the spins are $h_j = \frac{j(j+1)}{k+2}$. The fusion rules read

$$j_1 \times j_2 = \sum_{j=|j_1-j_2|}^{\min(J,k-J)} j, \qquad J = j_1 + j_2$$
 (5.2.23)

The quantum dimensions are $d_j = \frac{\sin \frac{\pi(2j+1)}{k+2}}{\sin \frac{\pi}{k+2}}$, and so j = k/2 is abelian. The corresponding \mathbb{Z}_2 symmetry acts as $j \mapsto \frac{1}{2}k - j$. The spin of this line is k/4, and so the symmetry is condensable if and only if k is even, k = 2n. The quotient theory is $PSU(2)_{2n} = SO(3)_n$; it is spin if n is odd.

The only fixed-point is j = n/2, whose S-matrix element is given by (5.1.18)

$$S_{n/2,n/2}(n) = \sum_{\beta \in n/2 \times n/2} \frac{\theta(\beta)}{\theta(n/2)^2} S_{1,\beta} F_{n/2,n/2} \begin{bmatrix} n & n/2 \\ n/2 & \beta \end{bmatrix}$$

$$= \frac{\theta(n/2)^{-2}}{\sqrt{n+1}} \sum_{j=0}^n \theta(j) \sin \frac{\pi}{2} \frac{2j+1}{n+1} F_{n/2,n/2} \begin{bmatrix} n & n/2 \\ n/2 & j \end{bmatrix}$$
(5.2.24)

which, using $F = (-1)^j$, becomes $S_{n/2,n/2}(n) = e^{-3i\pi n/4}$. One may check that modularity is satisfied, $S^2 = (ST)^3$ and $S^4 = (-1)^F \equiv (-1)^n$. This is indeed consistent with the quotient being bosonic if n is even, and fermionic if odd.

Let us consider the case of even n, where the quotient is bosonic. The unscreened lines are those with integer j, and the only fixed point is j = n/2. Thus, a basis for the Hilbert space is

$$|j\rangle = \frac{1}{\sqrt{2}}(|j\rangle + |n-j\rangle), \qquad j = 0, 1, \dots, n/2 - 1$$

$$|a_1\rangle = |n/2\rangle$$

$$|a_2\rangle = |n/2; n\rangle$$
(5.2.25)

where $|\cdot; \alpha\rangle$ denotes the corresponding state with an α -puncture.

The S-matrix of the quotient is

$$\hat{S}_{i,j} = 2S_{i,j} = 2\sqrt{\frac{1}{n+1}} \sin \frac{\pi(2i+1)(2j+1)}{2n+2}$$

$$\hat{S}_{i,a_1} = \sqrt{2}S_{i,n/2} = (-1)^i \sqrt{\frac{2}{n+1}}$$

$$\hat{S}_{a_1,a_1} = S_{n/2,n/2} = (-1)^{n/2} \sqrt{\frac{1}{n+1}}$$

$$\hat{S}_{a_1,a_2} = 0$$

$$\hat{S}_{a_2,a_2} = S_{n/2,n/2}(n) = i^{n/2}.$$
(5.2.26)

The fusion basis is defined by

$$|a_{\pm}\rangle = \frac{1}{\sqrt{2}} (|a_1\rangle \pm |a_2\rangle).$$
 (5.2.27)

One can easily compute the S-matrix in this basis, from where one can compute, for example, the fusion rules of the theory.

Consider now the case of odd n, where the quotient is spin. The NS lines are those with integral isospin, and the R lines with half-integral isospin. A basis of the quotient Hilbert space is

• If we take **NS-NS** boundary conditions, the states are

$$|j; \text{NS-NS}\rangle = \frac{1}{\sqrt{2}}(|j\rangle + |2n - j\rangle), \quad j = 0, 1, \dots, n.$$
 (5.2.28)

• If we take **NS-R** boundary conditions, the states are

$$|j; \text{NS-R}\rangle = \frac{1}{\sqrt{2}}(|j\rangle - |2n - j\rangle), \quad j = 0, 1, \dots, n.$$
 (5.2.29)

• If we take **R-NS** boundary conditions, the states are

$$|j; \text{R-NS}\rangle = \frac{1}{\sqrt{2}}(|j\rangle + |2n - j\rangle), \quad j = \frac{1}{2}, \frac{3}{2}, \dots, \frac{n}{2}, \dots, n$$

 $|n/2; \text{R-NS}\rangle = |n/2\rangle.$ (5.2.30)

• If we take **R-R** boundary conditions, the states are

$$|j; \text{R-R}\rangle = \frac{1}{\sqrt{2}} (|j\rangle - |2n - j\rangle), \quad j = \frac{1}{2}, \frac{3}{2}, \dots, \frac{n}{2}, \dots, n$$
$$|n/2; \text{R-R}\rangle = |n/2; n\rangle.$$
(5.2.31)

The modular data easily follows from this decomposition, and the once-punctured torus matrix element (5.2.24).

5.2.4 $U(1)_k$

In this section we construct the Hilbert space of the spin TQFT $U(1)_k$. This generalizes the construction of the semion-fermion theory of section 1.4.1. For some special values of k this theory is time-reversal invariant. The semion-fermion theory is recovered by taking k = 2. For k > 2, the time-reversal symmetry (when present) satisfies a more exotic algebra [4], namely $T^2 = C$, where C denotes an order-2 unitary symmetry (charge conjugation).

The construction of $U(1)_k$ is slightly different depending on whether k is even or odd. Indeed, for k odd the theory is naturally spin; but, for k even, it is bosonic, and so it has to be multiplied by the trivial factor $\{1, \psi\}$ if we are interested in its spin version. The latter case is rather similar to the semion-fermion theory, so here we will focus on the k odd case here, and sketch the main differences for k even at the end.

Consider the theory $U(1)_k$ with k odd. Its bosonic parent is $U(1)_{4k}$, whose anyons are labelled as $\alpha \in \mathbb{Z}_{4k}$. The spin theory is obtained by condensing the fermion $\psi = 2k$. The braiding phase of an arbitrary line α with respect to the fermion is $B(\alpha, \psi) = e^{\pi i \alpha}$, which means that the anyons are split as

NS:
$$2\alpha$$
, $\alpha = 0, 1, \dots, 2k - 1$
R: $2\alpha + 1$, $\alpha = 0, 1, \dots, 2k - 1$. (5.2.32)

These are all in two-dimensional orbits, paired up as

$$\alpha \stackrel{\times \psi}{\longleftrightarrow} \alpha + 2k \,. \tag{5.2.33}$$

As there are no fixed-points, all states are bosonic.

Hilbert space and modularity. Given the knowledge of the Hilbert space of the bosonic parent, and the action of the modular group on it, we easily construct the same objects in the quotient theory. In particular, the Hilbert space is $\mathcal{H} \cong \mathbb{C}^{4k}$, with states $|\alpha\rangle$, and modular transformations act as

$$S|\alpha\rangle = \sum_{\alpha' \in \mathbb{Z}_{4k}} S_{\alpha,\alpha'} |\alpha'\rangle$$

$$T|\alpha\rangle = e^{2\pi i (\alpha^2/4k - 1/24)} |\alpha\rangle$$

$$C|\alpha\rangle = |-\alpha \mod 4k\rangle,$$
(5.2.34)

where $S_{\alpha,\alpha'} = e^{-2\pi i \alpha \alpha'/4k}/2\sqrt{k}$, and the term -1/24 in the *T*-transformation refers to the central charge of the theory.

The quotient space is as follows:

• If we take **NS-NS** boundary conditions, the states are

$$|\alpha; \text{NS-NS}\rangle = \frac{1}{\sqrt{2}} [|2\alpha\rangle + |2\alpha + 2k\rangle], \quad \alpha = 0, \dots, k-1$$
 (5.2.35)

and one has

$$\hat{S}|\alpha; \text{NS-NS}\rangle = \frac{1}{\sqrt{k}} \sum_{\alpha'=0}^{k-1} e^{2\pi i \alpha \alpha'/k} |\alpha'; \text{NS-NS}\rangle$$

$$\hat{T}|\alpha; \text{NS-NS}\rangle = e^{2\pi i \left(\frac{\alpha^2}{2k} - \frac{1}{24}\right)} |\alpha; \text{NS-R}\rangle$$

$$\hat{C}|\alpha; \text{NS-NS}\rangle = \sum_{\alpha'=0}^{k-1} (\delta_{\alpha+\alpha'} + \delta_{\alpha+\alpha'-k}) |\alpha'; \text{NS-NS}\rangle$$
(5.2.36)

where $\delta_x = 1$ if $x \equiv 0 \mod 2k$, and $\delta_x = 0$ otherwise.

 $\bullet\,$ If we take ${\bf NS-R}$ boundary conditions, the states are

$$|\alpha; \text{NS-R}\rangle = \frac{1}{\sqrt{2}} \left[|2\alpha\rangle - |2\alpha + 2k\rangle \right], \quad \alpha = 0, \dots, k-1$$
 (5.2.37)

and one has

$$\hat{S}|\alpha; \text{NS-R}\rangle = \frac{1}{\sqrt{k}} \sum_{\alpha'=0}^{k-1} e^{2\pi i (2\alpha+1)\alpha'/2k} |\alpha'; \text{R-NS}\rangle$$

$$\hat{T}|\alpha; \text{NS-R}\rangle = e^{2\pi i \left(\frac{\alpha^2}{2k} - \frac{1}{24}\right)} |\alpha; \text{NS-NS}\rangle$$

$$\hat{C}|\alpha; \text{NS-R}\rangle = \sum_{\alpha'=0}^{k-1} (\delta_{\alpha+\alpha'} - \delta_{\alpha+\alpha'-k}) |\alpha'; \text{NS-R}\rangle$$
(5.2.38)

 $\bullet\,$ If we take ${\bf R-NS}$ boundary conditions, the states are

$$|\alpha; \text{R-NS}\rangle = \frac{1}{\sqrt{2}} [|2\alpha + 1\rangle + |2\alpha + 1 + 2k\rangle], \quad \alpha = 0, \dots, k - 1$$
 (5.2.39)

and one has

$$\hat{S}|\alpha; \text{R-NS}\rangle = \frac{1}{\sqrt{k}} \sum_{\alpha'=0}^{k-1} e^{2\pi i \alpha (2\alpha'+1)/2k} |\alpha'; \text{NS-R}\rangle$$

$$\hat{T}|\alpha; \text{R-NS}\rangle = e^{2\pi i \left(\frac{(2\alpha+1)^2}{8k} - \frac{1}{24}\right)} |\alpha; \text{R-NS}\rangle$$

$$\hat{C}|\alpha; \text{R-NS}\rangle = \sum_{\alpha'=0}^{k-1} (\delta_{\alpha+\alpha'+1} + \delta_{\alpha+\alpha'+1-k}) |\alpha'; \text{R-NS}\rangle$$
(5.2.40)

 $\bullet\,$ If we take ${\bf R}\text{-}{\bf R}$ boundary conditions, the states are

$$|\alpha; \text{R-R}\rangle = \frac{1}{\sqrt{2}} [|2\alpha + 1\rangle - |2\alpha + 1 + 2k\rangle], \quad \alpha = 0, \dots, k - 1$$
 (5.2.41)

and one has

$$\hat{S}|\alpha; \text{R-R}\rangle = \frac{1}{\sqrt{k}} \sum_{\alpha'=0}^{k-1} e^{2\pi i (2\alpha+1)(2\alpha'+1)/4k} |\alpha'; \text{R-R}\rangle$$

$$\hat{T}|\alpha; \text{R-R}\rangle = e^{2\pi i \left(\frac{(2\alpha+1)^2}{8k} - \frac{1}{24}\right)} |\alpha; \text{R-R}\rangle$$

$$\hat{C}|\alpha; \text{R-R}\rangle = \sum_{\alpha'=0}^{k-1} (\delta_{\alpha+\alpha'+1} - \delta_{\alpha+\alpha'+1-k}) |\alpha'; \text{R-R}\rangle$$
(5.2.42)

It is reassuring to see that these modular transformations map the different Hilbert spaces precisely as they should (cf. (5.1.26)). Moreover, these matrices are unitary, \hat{S} is symmetric $(\hat{S}_{s^{a},s^{b}}^{t} = \hat{S}_{s^{b},s^{a}})$, and they satisfy the modular algebra $(\hat{S}\hat{T})^{3} = \hat{S}^{2} = \hat{C}$ with $\hat{C}^{2} = 1$.

Wilson lines. The Wilson lines are given by

$$W^{(\mathbf{a})}(\alpha)|\gamma; s^{\mathbf{a}}s^{\mathbf{b}}\rangle = e^{-2\pi i\alpha(\gamma + (1+s^{\mathbf{a}})/4)/k}|\gamma; s^{\mathbf{a}}s^{\mathbf{b}}\rangle$$

$$W^{(\mathbf{b})}(\alpha)|\gamma; s^{\mathbf{a}}s^{\mathbf{b}}\rangle = |\alpha + \gamma; s^{\mathbf{a}}s^{\mathbf{b}}\rangle - s^{\mathbf{b}}|\alpha + \gamma + k; s^{\mathbf{a}}s^{\mathbf{b}}\rangle$$
(5.2.43)

where $\alpha \in \mathbb{Z}_{2k}$. They satisfy the expected properties, e.g.,

$$W_{s^{a},s^{b}}^{(c)}(\psi) = W_{s^{a},s^{b}}^{(c)}(k) = -s^{c}\mathbf{1}_{k}$$

$$W_{s^{a},s^{b}}^{(c)}(\alpha \times \alpha') = W_{s^{a},s^{b}}^{(c)}(\alpha)W_{s^{a},s^{b}}^{(c)}(\alpha')$$

$$S_{s^{a},s^{b}}W_{s^{a},s^{b}}^{(a)}(\alpha)(S_{s^{a},s^{b}})^{\dagger} = W_{s^{b},s^{a}}^{(b)}(\bar{\alpha})$$

$$S_{s^{a},s^{b}}W_{s^{a},s^{b}}^{(b)}(\alpha)(S_{s^{a},s^{b}})^{\dagger} = W_{s^{b},s^{a}}^{(a)}(\alpha)$$
(5.2.44)

Time-reversal. We now implement time-reversal invariance. Recall that $U(1)_k$ is time-reversal invariant if and only if $q^2 = -1 \mod k$ is solvable for some $q \in \mathbb{Z}$, in which case time-reversal acts as $\alpha \mapsto q\alpha$ [4]. This means that, given $\mathsf{T} = \tau K$, we require

$$\tau(W_2^{(c)})^* \tau^{-1} = W_{2q}^{(c)}$$

= $(W_2^{(c)})^q$ (5.2.45)

with solution

$$\tau_{\alpha,\beta} = (-s^{\mathsf{b}})^{\alpha+\beta} \delta_{2\alpha q+2\beta+\frac{1}{2}(s^{\mathsf{a}}+1)(q+1)}$$
(5.2.46)

up to a global phase. One can check that

$$\tau \tau^* = (-1)^{\operatorname{Arf}(s)} \left(\delta_{\frac{1}{2}(s^{\mathfrak{a}}+1)+\alpha+\beta} - s^{\mathfrak{b}} \delta_{\frac{1}{2}(s^{\mathfrak{a}}+1)+\alpha+\beta-k} \right)$$
(5.2.47)

and so $\mathsf{T}^2 = (-1)^{\operatorname{Arf}(s)} \hat{\mathsf{C}}$.

We see that the time-reversal algebra is deformed by Arf, signaling an anomaly. In this case, the source of the anomaly is clear: the theory has non-vanishing central charge, c = 1,

so it is not time-reversal invariant in the strict sense. We have to multiply by a suitable SPT in order to subtract off the central charge. In this case, $U(1)_{-1}$ does the trick, as this SPT has c = -1.

The Hilbert space of $U(1)_{-1}$ is straightforward: it suffices to take k = 1 in the discussion above. Looking at the action of \hat{C} on the (one-dimensional) Hilbert space of $U(1)_{-1}$ we learn that $\hat{C} = (-1)^{\operatorname{Arf}(s)}$. Therefore, multiplying a given theory by $U(1)_{\pm 1}$ has the effect of redefining $\hat{C} \to \hat{C} \times (-1)^{\operatorname{Arf}(s)}$, which means that the theory $U(1)_k \times U(1)_{-1}$ has undeformed algebra, namely $T^2 = \hat{C}$. The Arf deformation in the case of $U(1)_k$ was just signaling that we had not corrected the central charge down to zero; after doing so, the deformation disappears from the time-reversal algebra.

Finally, we make a few remarks concerning the k even case. Now the theory is naturally bosonic, and can be made spin by tensoring with an invertible spin TQFT. If we are interested in time-reversal invariance, the natural choice is $U(1)_k \times U(1)_{-1}$, so as to have vanishing central charge. As the theory is a tensor product, one factor being bosonic, the total Hilbert space is straightforward:

$$\hat{\mathcal{H}}_s(\mathrm{U}(1)_k \times \mathrm{U}(1)_{-1}) = \mathcal{H}(\mathrm{U}(1)_k) \otimes \hat{\mathcal{H}}_s(\mathrm{U}(1)_{-1})$$
(5.2.48)

where $\mathcal{H}(\mathrm{U}(1)_k)$ is the space of the bosonic theory $\mathrm{U}(1)_k$, and $\hat{\mathcal{H}}_s(\mathrm{U}(1)_{-1})$ is the space of the fermionic theory $\mathrm{U}(1)_{-1}$. The Hilbert space of $\mathrm{U}(1)_{-1}$ was discussed above, and that of $\mathrm{U}(1)_k$ is well-known, being bosonic. In this sense, no new computation is required in the case of $\mathrm{U}(1)_k$ with k even. One can easily check through straightforward computation that the main conclusions are identical to those of the k odd case, in particular, time-reversal satisfies $\mathsf{T}^2 = \hat{\mathsf{C}}$, with no deformation. (If we reintroduce a non-zero value of the central charge, by multiplying by an extra factor of $\mathrm{U}(1)_{\pm 1}$, the deformation reappears, and we get $\mathsf{T}^2 = (-1)^{\mathrm{Arf}(s)}\hat{\mathsf{C}}$, again signalling the anomaly due to c).

5.3 Fermionic surgery.

The partition function of a topological theory on an arbitrary three-manifold can be computed explicitly via surgery [22] (see also [289] for a nice review). The idea is that any manifold M can be written as $M_1 \cup M_2$, where M_1, M_2 are two handlebodies glued via some homeomorphism D acting on their shared boundary Σ . The path integral of the theory, perhaps in the presence of operator insertions, can be expressed as suitable matrix elements of D, i.e., $\langle v_1 | D | v_2 \rangle$, where v_1, v_2 are the states prepared by M_1, M_2 . In this sense, if we understand the action of homeomorphisms on Σ , we can compute any observable on any manifold M. The novelty in the spin case is that the topological spaces depend on a choice of spin structure, and this dependence is reflected in the choice of splitting and the action of D.

The simplest case is that of M = L(a, b), a lens space, since for these spaces (and these only) the surface Σ has genus g = 1, i.e., it is a torus. Here we will content ourselves with
illustrating the general idea through a few simple examples, mainly following [180].

A lens space L(a, b) is defined as the quotient of a sphere S^3 by a \mathbb{Z}_a group, whose action on S^3 is specified by b, with a, b two coprime integers. This space can be expressed as the gluing of two solid tori via the action of some $D \in SL_2(\mathbb{Z})$ matrix that depends on a, b; for example, one can take

$$D = \begin{pmatrix} b & a' \\ a & b' \end{pmatrix} \in SL_2(\mathbb{Z})$$
(5.3.1)

where a', b' are any two integers that make D unimodular, i.e., such that bb' - aa' = 1 (these exist because a, b are coprime). We shall mostly be interested in the case where a is even, where L(a, b) has two spin structures, while for a odd it has a unique spin structure.

For a even, the spin structures of L(a, b) are described as follows. There are two such structures, which we denote as \pm . Given the surgical presentation of the lens space as the sum of two tori, these two structures are induced from the spin structures of the tori. The **a**-cycle of these tori are filled-up so their spin structures take the form (NS, s_i^b), where i = 1, 2 refers to the sign around the **b**-cycle of these two tori. We can choose one of the signs s_i^b at will, but then the other will be determined too. For example, if we choose $s_1^b = \pm 1$, then (5.3.1) fixes $s_2^b = (-1)^{a'}(\pm 1)^{b'}$. In other words, the 3d spin structures \pm of L(a, b) correspond to tori with 2d spin structures $(-1, \pm 1)$ and $(-1, (-1)^{a'}(\pm 1)^{b'})$, respectively. Therefore, the spin-dependent partition function of the lens space is

$$Z[L(a,b);\pm] = \langle 0; \text{NS-}\pm |D|0; \text{NS-}(-1)^{a'}(\pm 1)^{b'} \rangle$$
(5.3.2)

where $|0; \pm \pm \pm\rangle$ is the vacuum state in the torus Hilbert space with spin structure \pm, \pm .

Let us do a few simple examples. If we take b = 1, then we can take a' = 0, b' = 1, and so the modular transformation reads

$$D = \begin{pmatrix} 1 & 0\\ a & 1 \end{pmatrix} = ST^{-a}S \tag{5.3.3}$$

where $S = \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}$ and $T = \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}$. The spin-dependent partition functions of the lens space are, then,

$$Z[L(a,1);\pm] = \langle 0; \mathrm{NS}-\pm | ST^{-a}S | 0; \mathrm{NS}-\pm \rangle$$
(5.3.4)

or, more explicitly,

$$Z[L(a, 1); -1] = (ST^{-a}S)_{\text{NS-NS} \to \text{NS-NS}}$$

= $S_{\text{NS-NS} \to \text{NS-NS}} (T_{\text{NS-R} \to \text{NS-NS}} T_{\text{NS-NS} \to \text{NS-NS}})^{-a/2} S_{\text{NS-NS} \to \text{NS-NS}}$
$$Z[L(a, 1); +1] = (ST^{-a}S)_{\text{NS-R} \to \text{NS-R}}$$

= $S_{\text{R-NS} \to \text{NS-R}} T_{\text{R-NS} \to \text{R-NS}}^{-a} S_{\text{NS-R} \to \text{R-NS}}$ (5.3.5)

respectively. Using the known expressions for the fermionic modular data in terms of the data of the bosonic parent (cf. (5.1.38), (5.1.39)), this is easily evaluated as

$$Z[L(a,1);\pm] \equiv \frac{2e^{2\pi i a c/24}}{\mathscr{D}^2} \sum_{\alpha \in \mathcal{A}_{\pm}} \theta(\alpha)^{-a} d_{\alpha}^2$$
(5.3.6)

where: c denotes the chiral central charge; $\mathscr{D}^2 \equiv \sum_{\alpha \in \mathcal{A}} d_{\alpha}^2$ the total quantum dimension of the bosonic parent; $\theta(\alpha) = e^{2\pi i h_{\alpha}}$ the topological spin; and $d_{\alpha} := S_{\alpha,0}/S_{0,0}$ the quantum dimension of $\alpha \in \mathcal{A}$.

For example, the bosonic parent of $U(1)_k$ (with k odd) is given by $U(1)_{4k}$. Being abelian, one has $d_{\alpha} = 1$ and $\mathscr{D}^2 = 4k$. Also, the central charge is c = 1 and the NS lines are the even lines of $\mathcal{A} \cong \mathbb{Z}_{4k}$ and the R-lines are the odd lines. Plugging these facts into the general formula (5.3.6) one finds

$$Z[L(a,1);-1] = \frac{1}{k} \sum_{\alpha=0}^{k-1} e^{-2\pi i a \left(\frac{\alpha^2}{2k} - \frac{1}{24}\right)}$$

$$Z[L(a,1);+1] = \frac{1}{k} \sum_{\alpha=0}^{k-1} e^{-2\pi i a \left(\frac{(2\alpha+1)^2}{8k} - \frac{1}{24}\right)}$$
(5.3.7)

which agree with the expressions in [180]. Of course, one also obtains the same result by plugging the fermionic modular matrices of $U(1)_k$ (cf. (5.2.36), (5.2.38), (5.2.40)) directly into (5.3.5).

One can also look at other theories, for example $SO(n)_1$, which is invertible and for which the partition function should be just a phase. Say, for n odd, one has three lines with quantum dimensions $d_0 = d_{\psi} = 1$ and $d_{\sigma} = \sqrt{2}$ and spins $\theta(0) = 1, \theta(\psi) = -1, \theta(\sigma) = e^{2\pi i n/16}$. Then, using $\mathscr{D}^2 = 4$ and c = n/2 in (5.3.6), one gets

$$Z[L(a,1);-1] = e^{2\pi i na/48}$$

$$Z[L(a,1);+1] = e^{-2\pi i na/24}$$
(5.3.8)

As before, one can also derive this by plugging the fermionic data (cf. (5.2.2)) directly into (5.3.5). Using the modular data of SO $(n)_1$ for n even, it is not hard to show that (5.3.8) is also correct for n even, i.e., these formulas are valid for any n.

As a simple consistency check of (5.3.7) and (5.3.8), note that

$$\frac{Z[L(a,1);-1]}{Z[L(a,1);+1]}\Big|_{U(1)_k} \equiv \frac{Z[L(a,1);-1]}{Z[L(a,1);+1]}\Big|_{SO(2k)_1}$$
(5.3.9)

which is the expected relation, given the level-rank duality $U(1)_k \leftrightarrow SU(k)_{-1} \times SO(2k)_1$, and the fact that $SU(k)_{-1}$ is bosonic (i.e., independent of the spin structure). One can use similar techniques to obtain the partition function of other lens spaces. For example, for b = 3 and $a = 3n \pm 1$, one can use $a' = \mp 1$ and $b' = \mp n$, whence

$$D = \begin{pmatrix} 3 & \mp 1 \\ a & (1 \mp a)/3 \end{pmatrix} = \mp S T^{-n} S T^{\pm 3} S$$
(5.3.10)

Using the known values of S and T one can compute $Z[L(a,3);\pm]$ for any spin structure. The general case is analogous: for a given a, b one writes down the corresponding $SL_2(\mathbb{Z})$ gluing matrix as a word in S and T, and evaluates the partition function using the values of these matrices as computed in the previous sections, e.g. cf. (5.1.38), (5.1.39).

5.4 Higher genus. Fermionic Verlinde formula.

We saw in the previous section how the knowledge of the modular data on the torus allowed us to compute the partition function of fermionic theories on more general spin 3-manifolds. The torus data was only good for lens spaces, while more general 3-manifolds usually split into higher genus Riemann surfaces. The purpose of this section is to explore fermionic theories on such surfaces. We let Σ denote the Riemann surface such that the manifold is $M = \mathbb{S}^1 \times \Sigma$, with the circle denoting the time direction. We choose to display the surface like so:



where $\mathbf{a}^i, \mathbf{b}^i$ is a basis of homology. Punctures, if any, are taken to be placed to the very left of the surface.

Spin structures. In order to define fermionic theories on Σ we must specify a spin structure, i.e., a consistent choice of signs a spinor picks up as we move it around. A contractible cycle is necessarily anti-periodic, but non-contractible ones are allowed to be periodic. A basis of non-contractible cycles is $\{\mathbf{a}^i, \mathbf{b}^i\}_{i=1,2,\ldots,g}$, so a spin structure is specified by 2^{2g} signs, which we denote as s^1, s^2, \ldots, s^g , where each $s^i = (s^{\mathbf{a}^i}, s^{\mathbf{b}^i})$ is a pair of signs. We denote by $\Sigma_{g;s}$ the surface Σ_g with choice (s^1, \ldots, s^g) . We say a choice signs is *even* if $(-1)^{\operatorname{Arf}(s)} = +1$, and *odd* if $(-1)^{\operatorname{Arf}(s)} = -1$, where

$$\operatorname{Arf}(s_1, \dots, s_g) := \sum_{i=1}^g (s^{\mathbf{a}^i} + 1)(s^{\mathbf{b}^i} + 1)/4 \mod 2$$
(5.4.1)

By construction, $\operatorname{Arf}(s^1 \cup s^2) = \operatorname{Arf}(s^1) + \operatorname{Arf}(s^2) \mod 2$, and so $s^1 \cup s^2$ is even if both s^1, s^2 are even, or both odd, and odd if only one of them is odd. Thus, the number # of even and odd spin structures satisfies

$$\#_{\text{even}}(g_1 + g_2) = \#_{\text{even}}(g_1) \#_{\text{even}}(g_2) + \#_{\text{odd}}(g_1) \#_{\text{odd}}(g_2) \\
\#_{\text{odd}}(g_1 + g_2) = \#_{\text{even}}(g_1) \#_{\text{odd}}(g_2) + \#_{\text{odd}}(g_1) \#_{\text{even}}(g_2)$$
(5.4.2)

Solving this recurrence with the obvious initial conditions $\#_{\text{even}}(1) = 3$ and $\#_{\text{odd}}(1) = 1$ we find the well-known results $\#_{\text{even}}(g) = 2^{g-1}(2^g + 1)$, $\#_{\text{odd}}(g) = 2^{g-1}(2^g - 1)$. Naturally, $\#_{\text{even}}(g) + \#_{\text{odd}}(g) = 2^{2g}$, the total number of spin structures.

Hilbert space. Recall that the Hilbert space of the fermionic theory can be obtained by gauging a suitable one-form symmetry in the bosonic parent theory. On a general surface Σ , the states of the bosonic theory are labelled as in (5.1.5), namely they consist of anyon configurations of the form

where α_i denote external punctures (i.e., Wilson lines running in the time direction), and the internal lines denote anyons running along the interior of Σ (with the **a**-cycles being contractible), the crosses × representing the holes. Gauging the one-form symmetry means inserting the condensing line ψ in all possible ways, i.e., once in the time-direction (which introduces an extra ψ -puncture), and then also in a homology basis of Σ (which projects into the invariant states). For example, wrapping ψ around a given \mathbf{a}^i -cycle must reproduce the spin structure around the same, $s^{\mathbf{a}^i}$, which means that the anyon around the orthogonal cycle \mathbf{b}^i must be in $\mathcal{A}_{s^{\mathbf{a}^i}}$, i.e., an NS line if $s^{\mathbf{a}^i} = -1$ and a R-line if $s^{\mathbf{a}^i} = +1$. Similarly, wrapping ψ around a \mathbf{b}^i -cycle must reproduce the spin structure around the same, $s^{\mathbf{b}^i}$, which instructs us to take suitable linear combinations such that fusion with ψ produces the correct sign.

Let us do a few explicit examples. We begin with the bosonic theory $Spin(n)_1$ (with n

odd); all its genus-2 untwisted states are



while the twisted states are



Condensing ψ yields the fermione theory SO(n)₁, on where these 10 + 6 states descend to a unique state on each of the 10 + 6 spin structures of Σ_2 . For example, say we are interested in the Hilbert space with spin structure $s^{a^1} = +1$ and $s^{a^2} = -1$. Then the states in the

condensed theory are linear combinations of the following bosonic states:



The reason is that wrapping ψ around the left anyon should yield $s^{a^1} = +1$, and hence the left anyon must be Ramond type (in this example, $\mathcal{A}_{R} = \{\sigma\}$), while wrapping ψ around the right anyon should yield $s^{a^2} = -1$, and hence the right anyon must be Neveu-Schwarz type (in this example, $\mathcal{A}_{NS} = \{1, \psi\}$).

The specific linear combinations of states in (5.4.6) is decided by the boundary conditions around the **b**-cycles. For example, given that

we learn that the first two lines in (5.4.6) must correspond to $s^{b^1} = -1$ boundary conditions, while the second two lines must correspond to $s^{b^1} = +1$ boundary conditions. Finally, given that

$$\cdots \xrightarrow{1} \underbrace{\begin{pmatrix} x \\ \psi \\ 1 \end{pmatrix}}_{1} = \cdots \xrightarrow{1} \underbrace{\begin{pmatrix} x \\ \psi \\ \psi \end{pmatrix}}_{\psi}$$
(5.4.8)
$$\cdots \xrightarrow{1} \underbrace{\begin{pmatrix} x \\ \psi \\ \psi \end{pmatrix}}_{\psi} = \cdots \xrightarrow{1} \underbrace{\begin{pmatrix} x \\ \chi \end{pmatrix}}_{1}$$

we learn that we must sum the two columns in (5.4.6) if $s^{b^2} = -1$, and subtract them if $s^{b^2} = +1$. In other words, the final result is the following four Hilbert spaces:

$$\hat{\mathcal{H}}(\Sigma_{2;+-,-\pm}) = \frac{1}{\sqrt{2}} \left(\begin{array}{ccc} & & 1 & \\ & & & \\ \sigma & & & 1 \end{array} \right) \\ \hat{\mathcal{H}}(\Sigma_{2;++,-\pm}) = \frac{1}{\sqrt{2}} \left(\psi \underbrace{\times}_{\sigma} \underbrace{1}_{\psi} \underbrace{\times}_{\sigma} \underbrace{1}_{\psi} \right) \\ & & \\ \end{array}$$
(5.4.9)

The other twelve Hilbert spaces of the fermionic phase are analogous. They are all one-dimensional, as expected from an invertible theory $SO(n)_1$. Furthermore, the fermionic punctures only appear for the odd spin structures of Σ_2 , i.e., the even-spin-structure Hilbert spaces have a single bosonic vacuum while the odd-spin-structure Hilbert spaces have a single fermionic vacuum. In general, the partition function of $SO(n)_1$ on an arbitrary surface Σ is $(-1)^{n \operatorname{Arf}(s)}$, where s is the spin structure of Σ , as we shall show below.

Let us now do an example that includes external punctures (on top of the condensing line). We can look at, for example, the torus Hilbert space of $U(1)_1$ in the presence of a ψ puncture. This means that we must look at the states of the bosonic parent $U(1)_4$, both in the presence of one ψ puncture and two ψ punctures. Being abelian, it is clear that there are, in fact, no states with a single ψ puncture, so all states in this example actually come from the twisted Hilbert space (the untwisted one is empty):



where $0, 1, 2, 3 \in \mathbb{Z}_4$ denote the four lines of $U(1)_4$, and $\psi = 2$ is the condensing line. Following the same reasoning as before, the four twisted Hilbert spaces are

$$\hat{\mathcal{H}}(\Sigma_{1;-\pm}^{\psi}) = \frac{1}{\sqrt{2}} \left(\begin{array}{c} \psi & \stackrel{0}{\longrightarrow} \psi \\ 2 \end{array} \psi \\ \downarrow & \stackrel{0}{\longleftarrow} \psi \end{array} \right)$$

$$\hat{\mathcal{H}}(\Sigma_{1;+\pm}^{\psi}) = \frac{1}{\sqrt{2}} \left(\begin{array}{c} \psi & \stackrel{1}{\longrightarrow} \psi \\ 3 \end{array} \psi \\ \mp \psi & \stackrel{1}{\longrightarrow} \psi \end{array} \right)$$

$$(5.4.11)$$

Note that all of the states come from the twisted Hilbert space of the bosonic parent; this means that they are all fermionic states. In particular, the path integral with Ramond boundary conditions on the time-circle, which computes $\operatorname{tr}_{\hat{\mathcal{H}}(\Sigma_{1;s}^{\psi})}(-1)^{F}$, yields -1 for all spin structures s, while the same path integral with Neveu-Schwarz boundary conditions, which computes $\operatorname{tr}_{\hat{\mathcal{H}}(\Sigma_{1;s}^{\psi})}(\operatorname{id})$, yields +1. This reproduces the geometric computation in appendix A of [216].

The construction of the Hilbert space for an arbitrary surface $\Sigma_{g,s}^{\alpha}$ (with arbitrary genus g, punctures α , and spin structure s), for an arbitrary fermionic TQFT, is a straightforward generalization. We begin by writing down all the states in the bosonic parent, both untwisted

and twisted. The former correspond to the states in the presence of external α punctures, while the latter we include α punctures and an extra ψ puncture. Then, the choice of spin structure *s* decides how these bosonic states are distributed among the different Hilbert spaces $\hat{\mathcal{H}}(\Sigma_{g;s}^{\alpha})$, namely, the boundary condition $s^{a^{i}}$ decide the anyon type on the \mathbf{b}^{i} cycle, while the boundary condition $s^{\mathbf{b}^{i}}$ decides which specific linear combinations of states to take. The states coming from the untwisted Hilbert space descend to bosons in the condensed phase, while those coming from the twisted Hilbert space descend to fermions.

Modular data. Given the explicit basis of states, the modular data of the fermionic theory is directly read off the data of the bosonic theory: as the fermione states are written as linear combinations of bosonic states, the action of a given Dehn twist on the former can be written in terms of its action on the latter, which is assumed known if the bosonic parent is sufficiently understood. Needless to say, this last part can be quite tricky since this requires the F-symbols of the bosonic parent, which are calculable in principle but time-consuming in practice. In any case, as a matter of principle, the problem is solved: all the modular data of the fermionic theory is uniquely specified as a function of the modular data of the bosonic parent. Given this, and a suitable surgery presentation of an arbitrary three-manifold M, one can compute any path integral on any fermionic TQFT, which completely solves the theory, at least formally.

Verlinde formula. Writing down an explicit basis of states of $\Sigma_{g;s}^{\alpha}$ is useful if we intend to calculate the modular data, as a means to compute partition functions on general 3-manifolds. If one is just interested in the path integral over $\mathbb{S}^1 \times \Sigma_{g;s}^{\alpha}$, then it is enough to know the number of states in $\hat{\mathcal{H}}(\Sigma_{g;s}^{\alpha})$, and not the form of an explicit basis. To this end, the Verlinde formula [38, 39] offers a significant simplification, since it gives us directly the number of states.

The fermionic generalization of the Verlinde formula is straightforward: if we denote $\hat{\mathcal{H}}(\Sigma_{a:s}^{\alpha_1\cdots\alpha_n}) := \mathbb{C}^{b|f}$, with b the number of bosons and f the number of fermions, then

$$b = \sum_{\alpha \in \mathcal{A}} c_{\alpha} S_{\mathbf{1},\alpha}^{\chi(\Sigma)} \prod_{i=1}^{n} S_{\alpha_{i},\alpha}$$

$$f = \sum_{\alpha \in \mathcal{A}} B(\psi, \alpha) c_{\alpha} S_{\mathbf{1},\alpha}^{\chi(\Sigma)} \prod_{i=1}^{n} S_{\alpha_{i},\alpha}$$
(5.4.12)

where: \mathcal{A} denotes the set of anyons in the bosonic parent; c_{α} is defined as $c_{\alpha} = 2^{-2g}$ if α is non-Majorana (i.e., either *a*-type or *x*-type) and as $c_{\alpha} = 2^{-g}(-1)^{\operatorname{Arf}(s)}$ if α is Majorana (i.e., *m*type); $S_{\alpha,\beta}$ is the *S*-matrix of the bosonic parent; and $B(\psi, \alpha) := S_{\psi,\alpha}/S_{1,\alpha} \equiv -\theta(\psi \times \alpha)/\theta(\alpha)$ is the braiding phase with respect to ψ (such that $B(\psi, \alpha) = +1$ if α is an NS line and $B(\psi, \alpha) = -1$ if α is an R line). This is just the regular Verlinde formula, but the fermionic states are weighted by the braiding phase $B(\psi, \alpha)$ (because these are the twisted sectors with respect to the one-form gauging), and there is a normalization factor c_{α} for Majorana lines (since non-Majorana lines are paired up as $\alpha, \alpha \times \psi$ while Majorana lines are not, as they are their own fermionic partner $m \times \psi = m$, cf. e.g. (5.1.33)).

As a note, we mention that one can isolate the contribution of Majorana lines and write the formula in the equivalent form

$$b = 2^{-2g} \dim(\Sigma_g^{\alpha_1 \cdots \alpha_n}) + 2^{-g} ((-1)^{\operatorname{Arf}(s)} - 2^{-g}) \left(\sum_m S_{1,m}^{\chi(\Sigma)} \prod_{i=1}^n S_{\alpha_i,m}\right)$$

$$f = 2^{-2g} \dim(\Sigma_g^{\alpha_1 \cdots \alpha_n \psi}) - 2^{-g} ((-1)^{\operatorname{Arf}(s)} - 2^{-g}) \left(\sum_m S_{1,m}^{\chi(\Sigma)} \prod_{i=1}^n S_{\alpha_i,m}\right)$$
(5.4.13)

where dim $(\Sigma_g^{\alpha_1\cdots\alpha_n})$ is the dimension of the bosonic Hilbert space (5.1.3). If we write $d_{\alpha} = S_{\alpha,1}/S_{1,1}$ for the quantum dimension of α , we can also write the formula as

$$b = \left(\sum_{\alpha \in \mathcal{A}} d_{\alpha}^{2}\right)^{g-1} \left(\sum_{\alpha \in \mathcal{A}} c_{\alpha} d_{\alpha}^{2-2g} \prod_{i=1}^{n} \frac{S_{\alpha_{i},\alpha}}{S_{\mathbf{1},\alpha}}\right)$$

$$f = \left(\sum_{\alpha \in \mathcal{A}} d_{\alpha}^{2}\right)^{g-1} \left(\sum_{\alpha \in \mathcal{A}} B(\psi, \alpha) c_{\alpha} d_{\alpha}^{2-2g} \prod_{i=1}^{n} \frac{S_{\alpha_{i},\alpha}}{S_{\mathbf{1},\alpha}}\right)$$
(5.4.14)

Examples. In order to illustrate the formula we can look at a few examples. The first one we shall look at is an arbitrary fermionic abelian order (with integral central charge), i.e., one where all lines have unit quantum dimension $d_{\alpha} = 1$. If we let 2k denote the number of lines in the theory (where k is the absolute value of the determinant of the K-matrix), then the bosonic parent is another abelian theory, this time bosonic, and which has $|\mathcal{A}| = 4k$ lines, half of which are NS lines (and descend to the 2k lines in the fermionic theory) and the other half R lines. As the bosonic parent is abelian, the theory has no Majorana lines, since $m \times \psi = m$ is incompatible with abelian fusion rules (which are group-like). Using this information, the Verlinde formula predicts that the number of fermions is f = 0, since half the lines have $B(\psi, \alpha) = +1$ and the other half $B(\psi, \alpha) = -1$, which cancel out pairwise. On the other hand, the number of bosons is $b = 2^{-2g} |\mathcal{A}|^g \equiv k^g$. So, to summarize, the fermionic Verlinde formula predicts no fermionic states, and k^g bosonic ones, which is what one expects in abelian theories.

We can do another simple example, namely $SO(n)_1$. For *n* odd, the bosonic parent $Spin(n)_1$ has three lines $\mathcal{A} = \{\mathbf{1}, \psi, \sigma\}$, the first two being NS lines and the last one a Majorana line. The quantum dimensions are $d_1 = d_{\psi} = 1$ and $d_{\sigma} = \sqrt{2}$. Plugging this information into the Verlinde formula one finds $b = (1 + (-1)^{Arf(s)})/2$ and $f = (1 - (-1)^{Arf(s)})/2$. In other words, there is a single state for any *s*, and it is a boson for even spin structures, and a fermion for odd spin structures, which we already saw in an explicit example above (cf. the discussion below (5.4.9)). For n even the theory is abelian, and therefore follows the analysis in the previous paragraph, namely b = 1 and f = 0. All in all, the Ramond partition function of $SO(n)_1$ is $(-1)^{n \operatorname{Arf}(s)}$.

As a final example, consider $SO(3)_k$ for k odd, in which case the theory is fermionic with bosonic parent is $SU(2)_{2k}$. This theory was discussed in section 5.2.3, where we showed that the NS lines are those with integer isospin j, while the R lines are those with half-integer isospin j; and, furthermore, there is a single Majorana line with j = k/2. Finally, the quantum dimension of a generic line is $d_j = \frac{\sin \frac{\pi(2j+1)}{2k+2}}{\sin \frac{\pi}{2k+2}}$. Plugging this information into the Verlinde formula, we find that

$$b = \left(\sum_{j=0}^{k} \frac{\sin\left[\frac{\pi(2j+1)}{2k+2}\right]^{2}}{\sin\left[\frac{\pi}{2k+2}\right]^{2}}\right)^{g-1} \left(\sum_{j=0}^{k} c_{j} \frac{\sin\left[\frac{\pi(2j+1)}{2k+2}\right]^{2-2g}}{\sin\left[\frac{\pi}{2k+2}\right]^{2-2g}}\right)$$

$$= \left(\frac{k+1}{2}\right)^{g-1} \left(\frac{1}{2}(-1)^{\operatorname{Arf}(s)} + 2^{-g} \sum_{\lambda=1}^{k} \sin\left[\frac{\pi\lambda}{2k+2}\right]^{2-2g}\right)$$

$$f = \left(\sum_{j=0}^{k} \frac{\sin\left[\frac{\pi(2j+1)}{2k+2}\right]^{2}}{\sin\left[\frac{\pi}{2k+2}\right]^{2}}\right)^{g-1} \left(\sum_{j=0}^{k} (-1)^{2j} c_{j} \frac{\sin\left[\frac{\pi(2j+1)}{2k+2}\right]^{2-2g}}{\sin\left[\frac{\pi}{2k+2}\right]^{2-2g}}\right)$$

$$= \left(\frac{k+1}{2}\right)^{g-1} \left(-\frac{1}{2}(-1)^{\operatorname{Arf}(s)} + 2^{-g} \sum_{\lambda=1}^{k} (-1)^{\lambda+1} \sin\left[\frac{\pi\lambda}{2k+2}\right]^{2-2g}\right)$$
(5.4.15)

where we relabelled $\lambda = 2j + 1$. For k = 1 we recover $b = (1 + (-1)^{\operatorname{Arf}(s)})/2$ and $f = (1 - (-1)^{\operatorname{Arf}(s)})/2$, as expected. These formulas agree with older derivations in the literature, see e.g. [290].

Generalities. After these simple examples, and to close this section, we shall next make a few interesting general remarks concerning the spin Verlinde formula. First off, let us note that, by definition, $S_{\alpha \times \psi,\beta} = B(\psi,\beta)S_{\alpha,\beta}$ and $S_{\alpha,\beta \times \psi} = B(\psi,\alpha)S_{\alpha,\beta}$. This means that, if we redefine one of the external punctures as $\alpha_i \to \alpha_i \times \psi$, this just inserts a factor of $B(\psi,\alpha)$ into the sums above, which has the effect of interchanging b and f. In formulas,

$$\hat{\mathcal{H}}(\Sigma_{g;s}^{\alpha_1\cdots\alpha_i\times\psi\cdots\alpha_n}) = \mathbb{C}^{0|1} \otimes \hat{\mathcal{H}}(\Sigma_{g;s}^{\alpha_1\cdots\alpha_i\cdots\alpha_n})$$
(5.4.16)

In other words, adding a fermion ψ along the time direction has the sole effect of reversing the fermion parity of all states in the Hilbert space.

It also follows from the previous equation that, if one of the punctures α_i is Majorana, then the Hilbert space satisfies $\hat{\mathcal{H}} = \mathbb{C}^{0|1} \otimes \hat{\mathcal{H}}$, i.e., it has the same number of bosons and fermions, b = f. This means that the trace $\operatorname{tr}_{\hat{\mathcal{H}}}(-1)^F = b - f$ vanishes or, in other words, the path integral with Ramond boundary conditions around the time circle is zero. This is as expected since the Majorana line carries a zero-mode which makes the path integral vanish. Another simple fact that follows from Verlinde is that if we have an odd number of Ramond lines, the path integral is identically zero. Indeed, in the sum over α the terms with α and $\alpha \times \psi$ carry the opposite sign and thus cancel out pairwise. The terms with α Majorana do not appear pairwise but they do not contribute anyway, since *m*-lines have no *S*-matrix elements with Ramond lines, $S_{m,\alpha} \equiv 0$ for $\alpha \in \mathcal{A}_R$. This vanishing of the path integral is precisely what one would expect, since a surface with an odd number of Ramond lines is inconsistent (the boundary condition around the punctures is +1 if contracted towards the punctures, but -1 if contracted away from the punctures).

Furthermore, it is easy to see that the Hilbert space $\hat{\mathcal{H}}(\Sigma_{g;s}^{\alpha})$ depends on s only through $\operatorname{Arf}(s)$, which is as expected since large diffeomorphisms induce isomorphisms $\hat{\mathcal{H}}(\Sigma_{g;s}^{\alpha}) \to \hat{\mathcal{H}}(\Sigma_{g;s'}^{\alpha})$ if and only if $\operatorname{Arf}(s) \equiv \operatorname{Arf}(s')$. In other words, Hilbert spaces with the same Arf parity are isomorphic. Along the same lines, it is easily checked that the total dimension b + f is in fact independent of s, as the Arf-dependent terms cancel out (since these only appear with *m*-lines, which have $B(\psi, \alpha) = -1$). This is a generic feature of spin TQFTs, where the distribution of bosons and fermions depends on the spin structure but the total number of states does not.

Moreover, if we sum over all spin structures, we recover the Hilbert spaces of the bosonic parent, namely

$$\bigoplus_{\in H^1(\Sigma_g, \mathbb{Z}_2)} \hat{\mathcal{H}}(\Sigma_{g;s}^{\alpha}) = \mathcal{H}(\Sigma_g^{\alpha}) \oplus \mathcal{H}(\Sigma_g^{\alpha, \psi})$$
(5.4.17)

as one would expect. This can be shown by noting that the number of even spin structures is $\#_{\text{even}} = 2^g (2^{g-1} + 1)$ and the number of odd spin structures is $\#_{\text{odd}} = 2^g (2^{g-1} - 1)$, and hence

se

$$\#_{\text{even}}b_{\text{even}} + \#_{\text{odd}}b_{\text{odd}} = \left(\sum_{\alpha \in \mathcal{A}} d_{\alpha}^{2}\right)^{g-1} \sum_{\alpha \in \mathcal{A}} d_{\alpha}^{2-2g} \prod_{i=1}^{n} \frac{S_{\alpha_{i},\alpha}}{S_{\mathbf{1},\alpha}}$$

$$\#_{\text{even}}f_{\text{even}} + \#_{\text{odd}}f_{\text{odd}} = \left(\sum_{\alpha \in \mathcal{A}} d_{\alpha}^{2}\right)^{g-1} \sum_{\alpha \in \mathcal{A}} B(\psi,\alpha) d_{\alpha}^{2-2g} \prod_{i=1}^{n} \frac{S_{\alpha_{i},\alpha}}{S_{\mathbf{1},\alpha}}$$
(5.4.18)

which are precisely the dimension of $\mathcal{H}(\Sigma_g^{\alpha})$ and $\mathcal{H}(\Sigma_g^{\alpha,\psi})$, respectively, as computed by the bosonic Verlinde formula (cf. (5.1.3)).

Finally, we mention that the factorization property of TQFTs holds, in the appropriate (fermionic) sense. Namely, if we take two surfaces $\Sigma_{g_1;s_1}^{\alpha_1\cdots\alpha_m\beta}$ and $\Sigma_{g_2;s_2}^{\alpha_m+1\cdots\alpha_n\beta'}$, then we can glue them together to yield a surface $\Sigma_{g_1+g_2;s_1\cup s_2}^{\beta_1\beta_1} \sqcup \Sigma_0^{\beta_1\beta_1'}$, where Σ_0 is a two-sphere used to glue the Riemann surfaces along the boundaries β , β' . Noting that $\hat{\mathcal{H}}(\Sigma_0^{\beta,\beta'}) = \delta_{\beta,\bar{\beta}'} \mathbb{C}^{1|0} \oplus \delta_{\beta,\bar{\beta}'\times\psi} \mathbb{C}^{0|1}$, the factorization property of spin TQFTs requires that, for non-Majorana β , the fermionic

Hilbert spaces must satisfy

$$\mathbb{C}^{1|0} \otimes \hat{\mathcal{H}}(\Sigma_{g;s}^{\alpha_{1}\cdots\alpha_{n}}) \cong \bigoplus_{\beta \in \mathcal{A}/\sim} \hat{\mathcal{H}}(\Sigma_{g_{1};s_{1}}^{\alpha_{1}\cdots\alpha_{m}\beta}) \otimes \hat{\mathcal{H}}(\Sigma_{g_{2};s_{2}}^{\bar{\beta}\alpha_{m+1}\cdots\alpha_{n}}) \\
\mathbb{C}^{0|1} \otimes \hat{\mathcal{H}}(\Sigma_{g;s}^{\alpha_{1}\cdots\alpha_{n}}) \cong \bigoplus_{\beta \in \mathcal{A}/\sim} \hat{\mathcal{H}}(\Sigma_{g_{1};s_{1}}^{\alpha_{1}\cdots\alpha_{m}\beta}) \otimes \hat{\mathcal{H}}(\Sigma_{g_{2};s_{2}}^{(\psi \times \bar{\beta})\alpha_{m+1}\cdots\alpha_{n}})$$
(5.4.19)

Here, $\beta \in \mathcal{A}/\sim$ means that we only consider one representative for each two-dimensional orbit $\beta \leftrightarrow \psi \times \beta$ (it does not matter which one we choose, due to (5.4.16)). By a straightforward algebraic manipulation, it is not hard to show that the spin Verlinde formula indeed satisfies this property, as required.

Chapter 6 Domain Walls in $4d \mathcal{N} = 1$ SYM.

Authorship. The content of this chapter is reproduced almost verbatim from the paper [5] written in collaboration with Jaume Gomis.

Abstract. $4d \ \mathcal{N} = 1$ super Yang-Mills (SYM) with simply connected gauge group G has h gapped vacua arising from the spontaneously broken discrete R-symmetry, where h is the dual Coxeter number of G. Therefore, the theory admits stable domain walls interpolating between any two vacua, but it is a nonperturbative problem to determine the low energy theory on the domain wall. We put forward an explicit answer to this question for all the domain walls for G = SU(N), Sp(N), Spin(N) and G_2 , and for the minimal domain wall connecting neighboring vacua for arbitrary G. We propose that the domain wall theories support specific nontrivial topological quantum field theories (TQFTs), which include the Chern-Simons theory proposed long ago by Acharya-Vafa for SU(N). We provide nontrivial evidence for our proposals by exactly matching renormalization group invariant partition functions twisted by global symmetries of SYM computed in the ultraviolet with those computed in our proposed infrared TQFTs.

Acknowledgments. We would like to thank Davide Gaiotto, Zohar Komargodski, Bruno Le Floch and Nathan Seiberg for useful discussions.

6.1	Domain Walls in $4d \mathcal{N} = 1$ SYM $\dots \dots 292$
6.2	Twisted Witten Indices
6.3	Hilbert Space of Spin TQFTs
6.4	Domain Wall TQFT Partition Functions
6.5	Concluding Remarks and Open Questions
6.6	Chern-Simons with Unitary and Orthogonal groups
6.7	The Exceptional Groups

6.1 Domain Walls in $4d \mathcal{N} = 1$ SYM

 $4d \ \mathcal{N} = 1$ super Yang-Mills (SYM) – Yang-Mills theory with a massless adjoint fermion – is believed to share with QCD nonperturbative phenomena such as confinement, existence of a mass gap, and chiral symmetry breaking. $4d \ \mathcal{N} = 1$ SYM with simple and simplyconnected gauge group G has h trivial vacua arising from the spontaneously broken \mathbb{Z}_{2h} chiral R-symmetry down to \mathbb{Z}_2 , where h is the dual Coxeter number of G (see table 6.1). The vacua are distinguished by the value of the gluino condensate [43–45]

$$\langle \operatorname{tr} \lambda \lambda \rangle = \Lambda^3 e^{2\pi i a/h}, \qquad a = 0, 1, \dots, h - 1.$$
 (6.1.1)

A supersymmetric domain wall that interpolates between two arbitrary vacua a and b at $x_3 \to \pm \infty$ can be defined. The \mathbb{Z}_{2h} symmetry implies that the domain wall theory depends only on the difference between vacua, on $n \equiv a - b \mod h$. We denote the resulting 3d low energy theory on the wall by W_n (see figure 6.1).



Figure 6.1: Vacua of $4d \ \mathcal{N} = 1$ SYM realized as the *h* roots of unity in the $\langle \operatorname{tr} \lambda \lambda \rangle$ -plane, where \mathbb{Z}_{2h} acts by a $2\pi/h$ rotation. W_n denotes the domain wall interpolating between vacua separated by *n* steps counterclockwise, and $\overline{W_n}$ the domain wall connecting the same vacua but with its orientation reversed. Clearly, $\overline{W_n} = W_{h-n}$.

While the *n*-wall tension is fixed by the supersymmetry algebra [291], it is a nonperturbative problem to determine the low energy (i.e. $E \ll \Lambda$) effective theory on the domain

wall. A supersymmetric domain wall preserves $3d \mathcal{N} = 1$ supersymmetry and therefore a universal $3d \mathcal{N} = 1$ Goldstone multiplet describes the spontaneously broken translation and supersymmetry. The nontrivial dynamical question is whether anything else remains in the infrared, a topological quantum field theory (TQFT) or gapless modes and, if so, which one(s).

In this paper we put forward a detailed answer to this question for all the *n*-domain walls for G = SU(N), Sp(N), Spin(N) and G_2 , and for n = 1 for arbitrary gauge group G.⁹⁷ The proposal for G = SU(N) was put forward long ago by Acharya-Vafa [46] motivated by brane constructions.⁹⁸ We provide nontrivial new evidence for the SU(N) proposal and for all the new proposals in this paper. The case G = Spin(N) is particularly subtle and rich.

We conjecture that the infrared of the *n*-domain wall theory in $4d \mathcal{N} = 1$ SYM with gauge group *G* is the infrared phase of $3d \mathcal{N} = 1$ SYM with gauge group *G* and Chern-Simons level

$$k = \frac{1}{2}h - n. (6.1.2)$$

In other words, we propose the infrared description

$$W_n \text{ in } 4d \ \mathcal{N} = 1 \text{ SYM with } G \iff 3d \ \mathcal{N} = 1 \text{ SYM with } G_{h/2-n}.$$
 (6.1.3)

Since the *n* and h - n domain walls are related by time-reversal (see figure 6.1), consistency of this proposal requires that the corresponding infrared phases must also be related by time-reversal, that is, by sending $k \to -k$ in the 3*d* theory. This requirement is indeed fulfilled by the identification between *n* and *k* in (6.1.2).

Determining the infrared phase of $3d \ \mathcal{N} = 1$ SYM is also a nonperturbative problem. In [30] it was proposed that this theory flows in the infrared to a nontrivial TQFT. The domain wall theories, we conjecture, are the "quantum phases" put forward in [30, 113, 237]. This predicts the following domain wall theories:⁹⁹

- G = SU(N). The *n*-domain wall theory is $W_n = U(n)_{N-n,N}$ Chern-Simons theory. This reproduces the proposal in [46].
- $G = \operatorname{Sp}(N)$. The *n*-domain wall theory is $W_n = \operatorname{Sp}(n)_{N+1-n}$ Chern-Simons theory.
- G = Spin(N). The *n*-domain wall theory is $W_n = O(n)_{N-2-n,N-n+1}^1$ Chern-Simons theory. We review the construction of this TQFT in section 6.4.3.

⁹⁷Vacua of $4d \mathcal{N} = 1$ SYM when G is not simply connected can be nontrivial [292, 293]. The presence of a 4d TQFT means that there is no purely 3d wall theory.

⁹⁸The Sp(N) case was mentioned in [294, 295]. For a partial list of references on domain walls in 4d gauge theories see e.g. [3, 40, 41, 223, 253, 296–302].

⁹⁹The notation G_k for Chern-Simons theories refers to Chern-Simons theory with gauge group G at level $k \in \mathbb{Z}$. The Chern-Simons theory $U(n)_{k,k'} \equiv \frac{SU(n)_k \times U(1)_{nk'}}{\mathbb{Z}_n}$ has two levels, and the theory based on O(n) has three levels (see section 6.4.3).

• $G = G_2$. The theory has h = 4 vacua and two independent walls: n = 1, 2. The 2-domain wall theory is $W_2 = SO(3)_3 \times S^1$, with $SO(3)_3$ Chern-Simons theory and S^1 the 3d sigma model on the circle. For the n = 1 wall see below.

The expectation is that the *n* and (h - n)-domain walls are related by time-reversal, that is $W_{h-n} = \overline{W_n}$ (see figure 6.1). This is realized by virtue of the level-rank dualities of Chern-Simons theories [28, 112, 113] and time-reversal flipping the sign of the Chern-Simons levels:

$$U(N-n)_{n,N} \longleftrightarrow U(n)_{-(N-n),-N}$$

$$Sp(N+1-n)_n \longleftrightarrow Sp(n)_{-(N+1-n)}$$

$$O(N-2-n)_{n,n+3}^1 \longleftrightarrow O(n)_{-(N-2-n),-(N-n+1)}^1.$$
(6.1.4)

The domain walls with n = h/2 are nontrivially time-reversal invariant. These TQFTs emerge in the infrared of $3d \mathcal{N} = 1$ SYM G_0 , with vanishing Chern-Simons level, which is time-reversal invariant.

We also conjecture that:

• Arbitrary group G. The n = 1 domain wall theory connecting neighboring vacua is $W_1 = G_{-1}$ Chern-Simons theory. This is consistent with the proposals put forward above due to the level-rank dualities $U(1)_N \leftrightarrow SU(N)_{-1}$, $Sp(1)_N \leftrightarrow Sp(N)_{-1}$ and $O(1)_{N-3,N}^1 = (\mathbb{Z}_2)_N \leftrightarrow Spin(N)_{-1}$.

We subject the conjecture (6.1.3) to a number of nontrivial quantitative tests. We exactly match renormalization-group invariant partition functions computed in the $4d \mathcal{N} = 1$ domain walls with the corresponding partition functions computed in the proposed infrared 3d TQFTs. This lends nontrivial support for our domain wall proposals in $4d \mathcal{N} = 1$ SYM. We stress that one computation is performed using the 4d degrees of freedom, and the other using the proposed 3d TQFT degrees of freedom.

The most basic partition function of the *n*-domain wall is the Witten index [42, 303]

$$I_n = \operatorname{tr}_{W_n}(-1)^F,$$
 (6.1.5)

where tr_{W_n} denotes the trace over the torus Hilbert space of W_n with periodic boundary conditions, and $(-1)^F$ fermion parity. This partition function was first computed by Acharya and Vafa in [46] using the 4d $\mathcal{N} = 1$ SYM fields.

We introduce and compute additional partition functions on the domain wall theory where the Witten index is twisted by a global symmetry of SYM. $4d \mathcal{N} = 1$ SYM with gauge group G can have charge conjugation zero-form symmetry C and one-form symmetry Γ [32].¹⁰⁰ Γ is the center of G, since the fermion in $4d \mathcal{N} = 1$ SYM is in the adjoint representation of the

¹⁰⁰C is the outer automorphism group of the Dynkin diagram \mathfrak{g} of G while S is the outer automorphism group of the extended Dynkin diagram $\mathfrak{g}^{(1)}$ of the affine Lie algebra associated to G. The group Γ is defined as the quotient S/C, i.e., the symmetries of $\mathfrak{g}^{(1)}$ that are *not* symmetries of \mathfrak{g} .

gauge group. The symmetries C and Γ do not commute when acting on Wilson lines, and combine into $S = \Gamma \rtimes C$ (see table 6.1). C acts on local operators and Wilson lines, and Γ on the Wilson lines of the theory. These symmetries are unbroken in each of the *h* vacua of $4d \ \mathcal{N} = 1$ SYM. S is the unbroken symmetry at each vacuum, while \mathbb{Z}_{2h} is spontaneously broken to \mathbb{Z}_2 . This allows us to define the following twisted Witten indices on the *n*-domain wall theory¹⁰¹

$$I_n^{\mathsf{c}} = \operatorname{tr}_{\mathbf{W}_n}(-1)^F \mathsf{c}\,,\tag{6.1.6}$$

where $c \in C$, and

$$I_n^{\mathsf{g}} = \operatorname{tr}_{\mathbf{W}_n}(-1)^F \mathsf{g}\,,\tag{6.1.7}$$

where $\mathbf{g} \in \Gamma$. Consistency of our conjecture requires that these partition functions, computed on either side of (6.1.3), match. We compute the Witten indices in terms of the 4*d* degrees of freedom in section 6.2, and in the 3*d* TQFTs in section 6.4.

Computing the domain wall Witten indices on the 3d side of the proposal requires understanding the Hilbert space of spin TQFTs, and not merely counting the number of states on the torus, as has been often stated in the literature. We delve into the details of constructing the Hilbert space of spin TQFTs and determining the fermionic parity of the states in section 6.3. The (twisted) Witten indices (6.1.5), (6.1.6), (6.1.7) map to twisted partition functions in the infrared spin TQFT. Importantly, the dimension of the Hilbert space and the index differ in general, as we shall see. In particular, the index sometimes vanishes in theories of interest. While the index can vanish, the twisted indices are non-vanishing, and supersymmetry on the domain wall is unbroken.

G	SU(N)	$\operatorname{Sp}(N)$	$\operatorname{Spin}(2N+1)$	$\operatorname{Spin}(4N)$	$\operatorname{Spin}(4N+2)$	E_6	E_7	E_8	F_4	G_2
h	N	N+1	2N - 1	4N - 2	4N	12	18	30	9	4
С	\mathbb{Z}_2	•	•	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{Z}_2	•	•	•	•
Γ	\mathbb{Z}_N	\mathbb{Z}_2	\mathbb{Z}_2	$\mathbb{Z}_2 \times \mathbb{Z}_2$	\mathbb{Z}_4	\mathbb{Z}_3	\mathbb{Z}_2	•	•	•
S	\mathbb{D}_N	\mathbb{Z}_2	\mathbb{Z}_2	\mathbb{D}_4	\mathbb{D}_4	\mathbb{S}_3	\mathbb{Z}_2	•	•	•

Table 6.1: Lie data for the simple Lie groups G. Here h denotes the dual Coxeter number (defined as $\operatorname{tr}(t_{\operatorname{adj}}t'_{\operatorname{adj}}) \equiv 2h(t,t')$, where (\cdot, \cdot) denotes the Killing form on \mathfrak{g} , normalized so that the highest root has $(\theta, \theta) = 2$). C, Γ are the zero-form and one-form symmetry groups of $4d \ \mathcal{N} = 1$ SYM with gauge group G, and $S = \Gamma \rtimes C$. \mathbb{D}_N denotes the dihedral group with 2N elements, and \mathbb{S}_N the symmetric group with N! elements. For SU(2) the zero-form symmetry group is trivial, and for Spin(8) the zero-form symmetry group is enhanced to \mathbb{S}_3 and the total symmetry group to \mathbb{S}_4 . The \mathbb{D}_N symmetry of pure SU(N) YM was considered in [304].

¹⁰¹One could also twist by any element $cg \in S$.

We summarize here the results of our computations, performed both in terms of the 4d fields and the conjectured 3d topological degrees of freedom, for which we find perfect agreement. We find it convenient to organize the results into master partition functions, which are defined as the generating functions for the twisted Witten indices. In other words, we sum the (twisted) partition functions over all *n*-walls:

$$Z^{s}(q) := \sum_{n=0}^{h} I_{n}^{s} q^{n} , \qquad (6.1.8)$$

where q is a fugacity parameter, and where $s \in S$ is an element of the unbroken symmetry group. These partition functions have an elegant interpretation as twisted partition functions of a collection of free fermions in 0 + 1 dimensions with energies determined by the Lie data of G (see section 6.2). Interestingly, the twisted partition function can be expressed as the untwisted partition function of an associated affine Lie algebra, whose extended Dynkin diagram is obtained by the "folding procedure" introduced in [305].

The master partition functions take a rather simple form:

• SU(N):

$$Z(q) = (1 - q)^{N}$$
(6.1.9)

$$Z^{\mathsf{c}}(q) = \begin{cases} (1-q)(1-q^2)^{(N-1)/2} & N \text{ odd,} \\ (1-q)^2(1-q^2)^{(N-2)/2} & N \text{ even,} \end{cases}$$
(6.1.10)

$$Z^{\mathsf{g}}(q) = \prod_{i=0}^{N-1} (1 - \mathsf{g}^{i}q), \qquad (6.1.11)$$

where **c** denotes the non-trivial element of $C = \mathbb{Z}_2$, and **g** is any element of $\Gamma = \mathbb{Z}_N$, thought of as an N-th root of unity.

• $\operatorname{Sp}(N)$:

$$Z(q) = (1-q)^{N+1}$$
(6.1.12)

$$Z^{\mathbf{g}}(q) = \begin{cases} (1-q)(1-q^2)^{N/2} & N \text{ even,} \\ (1-q^2)^{(N+1)/2} & N \text{ odd,} \end{cases}$$
(6.1.13)

where **g** denotes the non-trivial element of $\Gamma = \mathbb{Z}_2$.

• $\operatorname{Spin}(N)$, N odd:

$$Z(q) = (1-q)^3 (1-q^2)^{(N-1)/2-2}$$
(6.1.14)

$$Z^{\mathsf{g}}(q) = (1+q)(1-q)^2(1-q^2)^{(N-1)/2-2}, \qquad (6.1.15)$$

where **g** denotes the non-trivial element of $\Gamma = \mathbb{Z}_2$.

• $\operatorname{Spin}(N)$, N even:

$$Z(q) = (1-q)^4 (1-q^2)^{N/2-3}$$
(6.1.16)

$$Z^{\mathsf{c}}(q) = (1-q)^2 (1-q^2)^{N/2-2}, \qquad (6.1.17)$$

where c denotes the non-trivial element of $C = \mathbb{Z}_2$.

$$- N = 0 \mod 4: \ \Gamma = \mathbb{Z}_2 \times \mathbb{Z}_2 = \{\mathbf{1}, \mathbf{g}_1, \mathbf{g}_2, \mathbf{g}_1 \mathbf{g}_2\}$$
$$Z^{\mathbf{g}_1}(q) = (1 - q^2)^{N/2 - 1}$$
$$Z^{\mathbf{g}_2}(q) = Z^{\mathbf{g}_1 \mathbf{g}_2}(q) = (1 - q^2)^3 (1 - q^4)^{N/4 - 2}.$$
(6.1.18)

$$- N = 2 \mod 4$$
: $\Gamma = \mathbb{Z}_4 = \{\mathbf{1}, \mathbf{g}, \mathbf{g}^2, \mathbf{g}^3\}$

$$Z^{\mathbf{g}}(q) = Z^{\mathbf{g}^{3}}(q) = (1 - q^{4})^{(N/2 - 1)/2}$$

$$Z^{\mathbf{g}^{2}}(q) = (1 - q^{2})^{N/2 - 1}.$$
(6.1.19)

• *G*₂:

$$Z(q) = (1-q)^2 (1-q^2).$$
(6.1.20)

Expanding these formulas in a series in q yields I_n^s (see section 6.2). See also section 6.2.6 for the n = 1 domain wall twisted Witten indices for arbitrary simply-connected G.

The plan of the rest of the chapter is as follows. In section 6.2 we review the calculation of the untwisted Witten index for general domain walls in $4d \mathcal{N} = 1$ SYM, develop the necessary tools to study the twisted indices, and present a detailed calculation thereof, for all the classical Lie groups. In section 6.3 we explain how the Hilbert space of a spin Chern-Simons theory is constructed and, in particular, how to determine the fermion parity $(-1)^F$ of the different states. In section 6.4 we use this refined understanding of spin Chern-Simons theories to compute the twisted partition functions of the 3d TQFTs that, conjecturally, describe the infrared dynamics of the domain walls, and show exact agreement. We end with some forward-looking comments in section 6.5. We delegate to extra sections some technical details that are needed in the computation of the twisted partition functions in section 6.4 and some additional material.

6.2 Twisted Witten Indices

In this section we study the twisted Witten indices on the $3d \mathcal{N} = 1$ domain walls, as computed in terms of the ultraviolet 4d degrees of freedom, namely the gluons and gluinos. This requires considering $4d \mathcal{N} = 1$ SYM on a two-torus and quantizing the space of zero energy states. This leads to a $2d \mathcal{N} = (2, 2)$ sigma model on the moduli space of flat *G*-connections on a two-torus, which is the weighted projective space $\mathbf{WCP}_{a_0^{\vee},a_1^{\vee},...,a_r^{\vee}}^r$, where a_i^{\vee} is the comark for the *i*-th node in the extended Dynkin diagram $\mathfrak{g}^{(1)}$ of the affine Lie algebra associated to *G* and $r = \operatorname{rank}(G)$ [306, 307]. Just as $4d \mathcal{N} = 1$ SYM, this 2*d* theory also has *h* quantum vacua. A supersymmetric domain wall in $4d \mathcal{N} = 1$ SYM corresponds to a supersymmetric soliton in the $2d \mathcal{N} = (2, 2)$ sigma model [46].

Using the $2d \mathcal{N} = (2, 2)$ sigma model, Acharya and Vafa argued that the Witten index of the domain wall is encoded in the Hilbert space of r + 1 free fermions in 0 + 1 dimensions. Each fermion ψ_i is associated to the *i*-th node of the extended Dynkin diagram $\mathfrak{g}^{(1)}$ of G and the energy of each fermion is a_i^{\vee} . The fermion Hilbert space is graded by the energy of the states

$$\mathcal{H}_{\mathrm{F}} = \bigoplus_{n=0}^{h} \mathcal{H}_{\mathrm{F}}^{n}, \qquad (6.2.1)$$

where the maximal energy is h since $h = \sum_{i=0}^{r} a_i^{\vee}$. $\mathcal{H}_{\mathrm{F}}^n$ denotes the subspace of energy n, that is, the configurations such that

$$\sum_{i=0}^{\prime} \lambda_i a_i^{\vee} = n \,, \tag{6.2.2}$$

where $\lambda_i \in \{0, 1\}$ is the occupation number of the *i*-th fermion. The Witten index for the *n*-domain wall (6.1.5), with the Goldstino multiplet contribution removed, is the trace over the fermion Hilbert space $\mathcal{H}_{\mathrm{F}}^n$ [46]

$$I_n = \operatorname{tr}_{W_n}(-1)^F \equiv \operatorname{tr}_{\mathcal{H}_{\mathrm{F}}^n}(-1)^F.$$
(6.2.3)

The Witten index of all *n*-domain walls is encoded in the partition function of the fermions with periodic boundary conditions on a circle, corresponding to a sum over all states weighted by the energy:

$$Z(q) = \operatorname{tr}_{\mathcal{H}_{\mathrm{F}}}(-1)^{F} q^{H} = \sum_{n=0}^{h} I_{n} q^{n} \,.$$
(6.2.4)

This partition function is readily evaluated

$$Z(q) = \prod_{i=0}^{r} (1 - q^{a_i^{\vee}}), \qquad (6.2.5)$$

which implies, in particular, that the Witten index for the n and h - n wall are the same $(I_n = (-1)^{r+1}I_{h-n})$ since the fermionic Hilbert space for the n and h - n walls are related by particle-hole symmetry. This beautifully reproduces the expectation that the n domain wall and the h - n domain wall (cf. figure 6.1) are related to each other by time-reversal!

A symmetry $s \in S$ of $4d \mathcal{N} = 1$ SYM acts in a simple way on the Wilson lines of the gauge theory. A Wilson line is labeled by a representation of G with highest weight $\lambda \equiv \lambda_1 \omega_1 + \lambda_2 \omega_2 + \cdots + \lambda_r \omega_r$, where ω_i is the fundamental weight associated to the *i*-th node of the Dynkin diagram \mathfrak{g} . The Wilson line W_i labeled by the fundamental weight ω_i transforms under $\mathbf{c} \in \mathbf{C}$ as

$$\mathsf{c} \colon W_i \mapsto W_{\mathsf{c}(i)} \,, \tag{6.2.6}$$

where $\omega_{c(i)}$ is the fundamental weight which is charge conjugate to ω_i . An element $\mathbf{g} \in \Gamma$ acts by

$$\mathbf{g} \colon W_i \mapsto \alpha_{\mathbf{g}}(\omega_i) \, W_i \,, \tag{6.2.7}$$

where $\alpha_{\mathbf{g}}(\omega_i) \in \Gamma^*$ is the charge of ω_i under the center Γ of G. The action of a symmetry on the fundamental Wilson lines W_i induces an action on the fermions ψ_i , which are labeled by a node in the extended Dynkin diagram $\mathbf{g}^{(1)}$. We recall that C acts as an outer automorphism of \mathfrak{g} , S acts as an outer automorphism of $\mathbf{g}^{(1)}$ and $\Gamma = \mathsf{S}/\mathsf{C}$.

We now proceed to compute the Witten index on the domain wall twisted by the symmetries of the system, S. This group is identified with the group of symmetries of the extended Dynkin diagram $\mathfrak{g}^{(1)}$, i.e., a given $\mathbf{s} \in \mathbf{S}$ can be thought of as a permutation of the nodes $i \mapsto \mathbf{s}(i)$ that leaves the diagram $\mathfrak{g}^{(1)}$ invariant. The induced action on the effective 0+1 system of fermions is

$$\mathbf{s} \colon \psi_i \mapsto \psi_{\mathbf{s}(i)} \,, \tag{6.2.8}$$

where ψ_i is the fermion associated to the *i*-th node of $\mathfrak{g}^{(1)}$. This means that the symmetry **s** lifts to a map $\mathcal{H}_{\mathrm{F}} \to \mathcal{H}_{\mathrm{F}}$ which, by definition, commutes with the Hamiltonian,

$$[H, \mathbf{s}] = 0, \qquad (6.2.9)$$

inasmuch as $a_i^{\vee} \equiv a_{\mathsf{s}(i)}^{\vee}$. Thus, **s** restricts to a well-defined action on $\mathcal{H}_{\mathsf{F}}^n$, i.e., it preserves the grading (6.2.1). The twisted Witten index is

$$I_n^{\mathbf{s}} = \operatorname{tr}_{\mathbf{W}_n}(-1)^F \mathbf{s} \equiv \operatorname{tr}_{\mathcal{H}_{\mathbf{F}}^n}(-1)^F \mathbf{s}.$$
(6.2.10)

Similarly, the twisted partition function computes the generating function of twisted indices:

$$Z^{s}(q) = \operatorname{tr}_{\mathcal{H}_{F}}(-1)^{F} \mathsf{s} \, q^{H} = \sum_{n=0}^{h} I_{n}^{s} q^{n} \,.$$
(6.2.11)

An efficient way to compute this partition function is as follows. Take the *i*-th fermion, and consider its orbit under s:

$$\psi_i \mapsto \psi_{\mathsf{s}(i)} \mapsto \psi_{\mathsf{s}^2(i)} \mapsto \dots \mapsto \psi_{\mathsf{s}^{N_i}(i)} \equiv \psi_i \,, \tag{6.2.12}$$

where N_i denotes the length of the orbit of the *i*-th node under the symmetry **s**, i.e., the minimal integer such that $\mathbf{s}^{N_i}(i) \equiv i$. In the trace (6.2.10), the only configurations that contribute are those where the occupation number λ_i in (6.2.2) is constant along the orbit:

$$\lambda_i = \lambda_{\mathsf{s}(i)} = \lambda_{\mathsf{s}^2(i)} = \dots = \lambda_{\mathsf{s}^{N_i - 1}(i)} \,. \tag{6.2.13}$$

This means that we may restrict the sum over $\mathcal{H}_{\mathrm{F}}^{n}$ in (6.2.2) to those configurations where this identity is satisfied. We enforce this by dropping all but one of these labels, and multiplying its energy by N_{i} , i.e., we replace (6.2.2) by

$$\sum_{i=0}^{r'} \lambda_i a_i'^{\vee} = n \,, \tag{6.2.14}$$

where the sum is over one representative for each orbit, r' is the number of orbits of \mathbf{s} , and $a_i'^{\vee} = N_i a_i^{\vee}$ is the combined energy of all the elements of the orbit of λ_i . With this, the twisted Witten index (6.1.6) on the domain wall can be computed as the untwisted partition function of r' + 1 free fermions with energies $a_i'^{\vee}$:

$$Z^{\mathsf{s}}(q) = \prod_{i=0}^{r'} (1 - q^{a_i'^{\vee}}).$$
(6.2.15)

Since $h = \sum_{i=0}^{r'} a_i^{\vee}$ we see that $I_n^{\mathfrak{s}} = (-1)^{r'+1} I_{h-n}^{\mathfrak{s}}$, as required by time-reversal.

Diagrammatically, twisting by a symmetry folds the Dynkin diagram $\mathfrak{g}^{(1)}$ according to the action of \mathfrak{s} on the nodes [305]. This yields a new affine Dynkin diagram, which has r' + 1 < r + 1 nodes, and comarks $a'_i = N_i a_i^{\vee}$. The twisted Witten index is identical to the untwisted Witten index of the folded diagram.

A quick remark is in order. Let λ'_i be a node in the folded diagram, and let N_i be the number of nodes in the original diagram that folded into λ'_i . The node λ'_i is therefore a bound state of N_i fermions, and thus has fermion parity $(-1)^F = (-1)^{N_i}$. Moreover, the symmetry **s** permutes these fermions, which generates an extra sign corresponding to the signature of the permutation. When the permutation is cyclic, which is the case relevant to this paper, the signature is just $N_i - 1$. All in all, the contribution of λ'_i to the twisted trace is $(-1)^F \mathbf{s} = (-1)^{(N_i+N_i-1)} = -1$. Therefore, in the folded diagram the node behaves as a regular fermion, just with more energy, and so (6.2.15) is correct as written: the fermionic signs are all taken care of automatically by the folding.

The twisted Witten index can also be computed by diagonalizing the action of s (6.2.12) by a direct sum of unitary transformations, one for each orbit, which is a symmetry of the collection of fermions. In this basis, s acts with eigenvalue s_i on the *i*-th fermion, where s_i is an N_i -th root of unity. The twisted partition function can, therefore, also be expressed as

$$Z^{\mathsf{s}}(q) = \prod_{i=0}^{r} (1 - \mathsf{s}_{i} \, q^{a_{i}^{\vee}}) \,. \tag{6.2.16}$$

This also makes the action of time-reversal symmetry on the domain walls manifest, cf. $I_n^{\mathbf{s}} = (-1)^{r+1} \det(\mathbf{s}) I_{h-n}^{\mathbf{s}}$, where $\det(\mathbf{s}) = \mathbf{s}_0 \mathbf{s}_1 \cdots \mathbf{s}_r = (-1)^{r+r'}$ is the parity of the permutation induced by \mathbf{s} . We now discuss zero-form symmetries and one-form symmetries in turn.

Zero-form symmetries. $4d \mathcal{N} = 1$ SYM has charge conjugation symmetry C if and only if the (unextended) Dynkin diagram \mathfrak{g} of the Lie algebra of G has a symmetry. This corresponds to an outer automorphism of the Lie algebra of G (see table 6.1). Such symmetry is present for the A_r, D_r, E_6 algebras, where C acts as a transposition (order-two permutation) on the nodes of the Dynkin diagram (with low-rank exceptions A_1, D_4). In this case, folding the diagram by C gives rise to what is usually called the twisted affine Dynkin diagram $\mathfrak{g}^{(2)}$ [308, 309], which is constructed by identifying the nodes of \mathfrak{g} that are permuted by C (and adding the extending node).

As $C = \mathbb{Z}_2$, the eigenvalues in (6.2.16) are trivial to determine: if a node *i* is fixed by $c \in C$, then its eigenvalue is $c_i = +1$. On the other hand, if the pair of nodes *i*, *j* are swapped, then the eigenvalues are ± 1 , which can be assigned as $c_i = +1$ and $c_j = -1$ (or vice-versa).

One-form symmetries. As discussed in section 6.1, $4d \ \mathcal{N} = 1$ has a one-form symmetry group Γ given by the center of the gauge group G. Here, the eigenvalues \mathbf{s}_i in (6.2.16) have a very natural interpretation. An element $\mathbf{g} \in \Gamma$ acts as an outer automorphism of $\mathbf{g}^{(1)}$, a permutation of the r + 1 nodes. Diagonalizing this permutation results on an eigenvalue \mathbf{g}_i on the fermion ψ_i associated to the *i*-th node, which is the charge of the element of the center $\mathbf{g} \in \Gamma$ on the *i*-th fundamental weight of \mathbf{g} [15, 284, 310–312], that is

$$\mathbf{g}_i = \alpha_{\mathbf{g}}(\omega_i) \,. \tag{6.2.17}$$

Note that this is precisely how $\mathbf{g} \in \Gamma$ acts on the ultraviolet Wilson loops W_i (cf. (6.2.7)).

We now proceed to compute the twisted Witten indices for G = SU(N), Sp(N), Sp(N), and G_2 respectively.

6.2.1 G = SU(N)

Consider the algebra $A_{N-1} = \mathfrak{su}_N$. The symmetries of this algebra are as follows:

- The group SU(N) has a \mathbb{Z}_2 zero-form symmetry, which corresponds to complex conjugation. It acts by interchanging the *i*-th node with the (N i)-th node in \mathfrak{g} . The associated diagonal action can be chosen as follows: take $c_0 = +1$ for the extended node, and $c_i = +1$ for the first half of the unextended nodes, and $c_i = -1$ for the second half.
- The group SU(N) has a \mathbb{Z}_N one-form symmetry, whose associated charge is the *N*-ality (the number of boxes in the Young diagram modulo *N*). If **g** denotes a primitive root of unity, then a generic element $\mathbf{g}^t \in \mathbb{Z}_N$ acts on the extended diagram $\mathbf{g}^{(1)}$ as a cyclic permutation by *t* units, $\lambda_i \mapsto \lambda_{i+t \mod N}$. The center acts on a representation with highest weight λ as follows:

$$\alpha_{\mathbf{g}}(\lambda) \equiv \prod_{i=0}^{N-1} \mathbf{g}^{i\lambda_i} , \qquad (6.2.18)$$

which means that the eigenvalues in (6.2.16) are $g_i = g^i$.

Let us begin by computing the untwisted partition function. The comarks are all $a_i^{\vee} = 1$. Plugging this into equation (6.2.5) we obtain the untwisted partition function

$$Z(q) = (1-q)^N, (6.2.19)$$

and, expanding, the Witten index

$$I_n = (-1)^n \binom{N}{n}.$$
 (6.2.20)

This result was also obtained, by an entirely different method, in [313].

We now move on to the twisted indices. Charge conjugation acts on the extended Dynkin diagram $A_{N-1}^{(1)}$ as follows:



where the blue node denotes the affine root, and the integers denote the comarks a_i^{\vee} . The automorphism folds the diagram in half.¹⁰² The result is

$$A_{2m-1}^{(1)} \mapsto A_{2m-1}^{(2)}: \qquad \stackrel{1}{\longrightarrow} \stackrel{2}{\longrightarrow} \stackrel{2}{\longrightarrow} \stackrel{2}{\longrightarrow} \cdots \stackrel{2}{\longrightarrow} \stackrel{2}{\rightarrow} \stackrel{2}{\rightarrow}$$

where the folded diagrams both have $m + 1 = \lfloor N/2 \rfloor + 1$ nodes. From this we conclude that the folded diagram has r' + 1 = (N + 1)/2 and r' = (N + 2)/2 nodes for N odd and N even, respectively. In the first case, one node has comark equal to 1, and the rest equal to 2; while in the second case, there are two nodes with comark 1, and the rest equal to 2. Using (6.2.15), the c-twisted partition function is¹⁰³

$$Z^{\mathsf{c}}(q) = \begin{cases} (1-q)(1-q^2)^{(N-1)/2} & N \text{ odd,} \\ (1-q)^2(1-q^2)^{(N-2)/2} & N \text{ even,} \end{cases}$$
(6.2.23)

¹⁰²For N odd, the $(N \pm 1)/2$ -th nodes would naively fold into a loop, which does not yield a valid Dynkin diagram. The correct folding is given by the theory of twisted Kač-Moody algebras [308, 309]. We henceforth fold the diagrams following [308, 309]. The only information we need from the diagram are the comarks.

¹⁰³In SU(2) the action of **c** is a gauge transformation and **c** is not a symmetry; indeed, $Z^{c}(q) = Z(q)$.

and, expanding, the c-twisted Witten indices are

$$I_{n}^{c} = \begin{cases} (-1)^{n/2} \binom{(N-1)/2}{n/2} & N \text{ odd, } n \text{ even,} \\ (-1)^{(n+1)/2} \binom{(N-1)/2}{(n-1)/2} & N \text{ odd, } n \text{ odd,} \\ (-1)^{n/2} \left[\binom{(N-2)/2}{n/2} - \binom{(N-2)/2}{n/2-1} \right] & N \text{ even, } n \text{ even,} \\ 2(-1)^{(n+1)/2} \binom{(N-2)/2}{(n-1)/2} & N \text{ even, } n \text{ odd.} \end{cases}$$
(6.2.24)

One can also compute the partition function in the diagonal basis, where $c_i = +1$ for the first half of the nodes, and $c_i = -1$ for the second half. Plugging this into (6.2.16) yields the same expression for the twisted partition function.

Let us now consider the partition function twisted by the $\Gamma = \mathbb{Z}_N$ one-form symmetry. If **g** denotes a primitive *N*-th root of unity, then a generic element $\mathbf{g}^t \in \mathbb{Z}_N$ acts on the extended diagram as follows:



where the folded diagram has gcd(N,t) nodes, each with energy N/gcd(N,t). In other words, g^t folds the diagram into the affine diagram of SU(gcd(N,t)), with comarks N/gcd(N,t). This immediately yields the twisted partition function as (6.2.15)

$$Z^{\mathbf{g}^{t}}(q) = (1 - q^{N/\gcd(N,t)})^{\gcd(N,t)}.$$
(6.2.26)

The twisted index reads

$$I_n^{\mathbf{g}^t} = \begin{cases} (-1)^{n \operatorname{gcd}(N,t)/N} {\operatorname{gcd}(N,t) \choose n \operatorname{gcd}(N,t)/N} & N | n \operatorname{gcd}(N,t), \\ 0 & \text{otherwise.} \end{cases}$$
(6.2.27)

Naturally, for t = 0 this reduces to the untwisted result.

Alternatively, we may compute the same partition function in the diagonal basis. Using equations (6.2.16) and (6.2.18), the twisted partition function is given by

$$Z^{\mathbf{g}}(q) = \prod_{i=0}^{N-1} (1 - \mathbf{g}^{i}q) \equiv (q; \mathbf{g})_{N}, \qquad (6.2.28)$$

the so-called q-Pochhammer symbol, essentially defined by this product. One may prove that this is in fact identical to (6.2.26). Expanding the product, the twisted index becomes

$$I_{n}^{\mathsf{g}} = (-1)^{n} \mathsf{g}^{\frac{1}{2}n(n-1)} \binom{N}{n}_{\mathsf{g}}, \qquad \binom{N}{n}_{\mathsf{g}} := \frac{(\mathsf{g}; \mathsf{g})_{N}}{(\mathsf{g}; \mathsf{g})_{n}(\mathsf{g}; \mathsf{g})_{N-n}}, \qquad (6.2.29)$$

where the term in parentheses denotes the so-called q-binomial coefficient. This is again identical to (6.2.27).

6.2.2 G = Sp(N)

Consider the algebra $C_N = \mathfrak{sp}_N$. The symmetries of this algebra are as follows:

- The group Sp(N) has no zero-form symmetry.
- The group $\operatorname{Sp}(N)$ has a \mathbb{Z}_2 one-form symmetry, whose charged representations are the pseudo-real ones. The non-trivial element $\mathbf{g} \in \mathbb{Z}_2$ acts on the extended diagram by reversing the nodes $\lambda_i \mapsto \lambda_{N-i}$. The center acts on a representation λ as follows:

$$\alpha_{\mathbf{g}}(\lambda) \equiv (-1)^{\sum_{i=0}^{\lfloor (N-1)/2 \rfloor} \lambda_{2i+1}}, \qquad (6.2.30)$$

which means that the eigenvalues in (6.2.16) are $\mathbf{g}_i = (-1)^i$.

Let us begin by computing the untwisted partition function. The comarks for Sp(N) are all equal to one, i.e. $a_i^{\vee} = 1$ for i = 0, 1, ..., N. Plugging this into equation (6.2.5) we obtain the untwisted partition function

$$Z(q) = (1-q)^{N+1}, (6.2.31)$$

and, expanding, the Witten index

$$I_n = (-1)^n \binom{N+1}{n}.$$
 (6.2.32)

 $4d \operatorname{Sp}(N) \mathcal{N} = 1$ SYM has no charge conjugation symmetry. We can consider instead the index twisted by the $\Gamma = \mathbb{Z}_2$ one-form center symmetry, which acts on the extended Dynkin

diagram as follows:

$$(6.2.33)$$

where the blue node denotes the affine root, and the integers denote the comarks a_i^{\vee} . The automorphism folds the diagram in half (see footnote 102). The result is

$$C_{2m}^{(1)} \mapsto A_{2m}^{(2)} : \overset{2}{\Box} \overset{2}{\Box$$

where the folded diagrams have $m + 1 = \lfloor N/2 \rfloor + 1$ nodes.

From this we learn that the folded diagram has r' + 1 = (N+2)/2 and r' + 1 = (N+1)/2nodes, for N even and N odd, respectively. In the first case, one of these nodes has energy equal to 1, and the rest equal to 2; while in the second case, they are all of energy 2. Plugging this into (6.2.15) the one-form twisted partition function is

$$Z^{\mathsf{g}}(q) = \begin{cases} (1-q)(1-q^2)^{N/2} & N \text{ even,} \\ (1-q^2)^{(N+1)/2} & N \text{ odd,} \end{cases}$$
(6.2.35)

and, expanding, the twisted Witten index

$$I_{n}^{g} = \begin{cases} (-1)^{(n+1)/2} \binom{N/2}{(n-1)/2} & N \text{ even, } n \text{ odd,} \\ (-1)^{n/2} \binom{N/2}{n/2} & N \text{ even, } n \text{ even,} \\ 0 & N \text{ odd, } n \text{ odd,} \\ (-1)^{n/2} \binom{(N+1)/2}{n/2} & N \text{ odd, } n \text{ even.} \end{cases}$$
(6.2.36)

One can also compute the partition function in the diagonal basis, where $\mathbf{g}_i = +1$ for the even nodes, and $\mathbf{g}_i = -1$ for the odd ones. Plugging this into (6.2.16) yields the same expression for the twisted partition function.

6.2.3 G = Spin(2N+1)

Consider the algebra $B_N = \mathfrak{so}_{2N+1}$. The symmetries of this algebra are as follows:

- The group Spin(2N+1) has no zero-form symmetry.
- The group $\operatorname{Spin}(2N+1)$ has a \mathbb{Z}_2 one-form symmetry, whose charged representations are the spinors. The non-trivial element $\mathbf{g} \in \mathbb{Z}_2$ acts on the extended diagram by permuting the zeroth and first nodes, $\lambda_0 \leftrightarrow \lambda_1$. The center acts on a representation λ as follows:

$$\alpha_{\mathbf{g}}(\lambda) \equiv (-1)^{\lambda_N} \tag{6.2.37}$$

which means that the eigenvalues in (6.2.16) are $\mathbf{g}_i = (-1)^{\delta_{i,N}}$.

Let us begin by computing the untwisted partition function. The comarks for Spin(2N+1) are $a_i^{\vee} = 1$ for i = 0, 1, N, and $a_i^{\vee} = 2$ for $i = 2, 3, \ldots, N-1$. Plugging this into equation (6.2.5) we obtain the untwisted partition function

$$Z(q) = (1-q)^3 (1-q^2)^{N-2}, \qquad (6.2.38)$$

and, expanding, the Witten index

$$I_n = \begin{cases} (-1)^{n/2} \left[\binom{N-2}{n/2} - 3\binom{N-2}{n/2-1} \right] & n \text{ even,} \\ (-1)^{(n-1)/2} \left[\binom{N-2}{(n-1)/2-1} - 3\binom{N-2}{(n-1)/2} \right] & n \text{ odd.} \end{cases}$$
(6.2.39)

Note that the index vanishes for $N = 1 \mod 4$ and n = (N - 1)/2 and by time-reversal for n' = h - n = (3N - 1)/2. This clearly illustrates the crucial difference between the dimension of the Hilbert space and the index.

 $4d \operatorname{Spin}(2N+1) \mathcal{N} = 1$ SYM has no charge conjugation symmetry. We can consider instead the index twisted by the $\Gamma = \mathbb{Z}_2$ one-form center symmetry. The non-trivial element $\mathbf{g} \in \mathbb{Z}_2$ acts on the extended Dynkin diagram as follows:

$$B_N^{(1)}: \left(\begin{array}{c} 1 & 0 & 2 \\ 1 & 0 & - \end{array} \right)^2 \xrightarrow{2}{-} 2 & 2 & 2 \\ 1 & 0 & - \end{array} \right) \mapsto A_{2N}^{(2)}: \begin{array}{c} 2 & 2 & 2 \\ - & 0 & - \end{array} \xrightarrow{2}{-} 2 & - \\ - & 0 & - \end{array} \right) (6.2.40)$$

where the blue node denotes the affine root, and the integers denote the comarks a_i^{\vee} .

From this we learn that the folded diagram has r' + 1 = N nodes, one of which has energy equal to 1, and the rest all energy equal to 2. Plugging this into (6.2.15) the one-form twisted partition function is

$$Z^{\mathsf{g}}(q) = (1-q)(1-q^2)^{N-1}, \qquad (6.2.41)$$

and, expanding, the Witten index

$$I_n^{\mathbf{g}} = \begin{cases} (-1)^{n/2} \binom{N-1}{n/2} & n \text{ even,} \\ \\ (-1)^{(n+1)/2} \binom{N-1}{(n-1)/2} & n \text{ odd.} \end{cases}$$
(6.2.42)

One can also compute the partition function in the diagonal basis, where $g_i = +1$ for all the nodes except for the last one, which has $g_N = -1$. Plugging this into (6.2.16) yields the same expression for the twisted partition function.

$6.2.4 \quad G = \mathrm{Spin}(2N)$

Consider the algebra $D_N = \mathfrak{so}_{2N}$. The symmetries of this algebra are as follows:

- The group Spin(2N) has a \mathbb{Z}_2 zero-form symmetry. The corresponding charge is the chirality of the representation. This symmetry acts by permuting the last two nodes in the unextended Dynkin diagram. The associated diagonal action can be chosen as follows: take $c_i = +1$ for all but the last two nodes, and $c_{N-1} = +1$ and $c_N = -1$.
- The group Spin(2N) has a $\mathbb{Z}_2 \times \mathbb{Z}_2$ one-form symmetry if N is even, and \mathbb{Z}_4 if odd. They act on the extended Dynkin diagram as follows: one of the \mathbb{Z}_2 's for N even, and the \mathbb{Z}_2 subgroup of \mathbb{Z}_4 for N odd, acts as the permutation $\lambda_0 \leftrightarrow \lambda_1$ and $\lambda_{N-1} \leftrightarrow \lambda_N$, while fixing the rest of Dynkin labels in the extended diagram. The other \mathbb{Z}_2 factor reverses the order of the extended Dynkin labels, while \mathbb{Z}_4 acts as $\lambda_0 \mapsto \lambda_N \mapsto \lambda_1 \mapsto \lambda_{N-1} \mapsto \lambda_0$, and it reverses the order of the rest of Dynkin labels.

For N even, let $(\mathbf{g}_1, \mathbf{g}_2) \in \mathbb{Z}_2 \times \mathbb{Z}_2$; and, for N odd, let $\mathbf{g} \in \mathbb{Z}_4$; all thought of as roots of unity. The center acts on a representation λ as follows:

$$\alpha_{\mathbf{g}_{1},\mathbf{g}_{2}}(\lambda) \equiv \mathbf{g}_{1}^{\lambda_{N-1}+\lambda_{N}} \mathbf{g}_{2}^{(N/2-1)\lambda_{N-1}+(N/2)\lambda_{N}+\sum_{i=0}^{N/2-2}\lambda_{2i+1}} \alpha_{\mathbf{g}}(\lambda) \equiv \mathbf{g}^{-(N-2)\lambda_{N-1}-N\lambda_{N}+2\sum_{i=0}^{(N-3)/2}\lambda_{2i+1}}.$$
(6.2.43)

Therefore, the eigenvalues in (6.2.16) are

$$(\mathbf{g}_{1}, \mathbf{g}_{2})_{2i} = 1, \qquad i \in [0, N/2 - 1)$$

$$(\mathbf{g}_{1}, \mathbf{g}_{2})_{2i+1} = \mathbf{g}_{2}, \qquad i \in [0, N/2 - 1)$$

$$(\mathbf{g}_{1}, \mathbf{g}_{2})_{N-1} = \mathbf{g}_{1}\mathbf{g}_{2}^{N/2 - 1}$$

$$(\mathbf{g}_{1}, \mathbf{g}_{2})_{N} = \mathbf{g}_{1}\mathbf{g}_{2}^{N/2}$$

$$\mathbf{g}_{2i+1} = \mathbf{g}^{2}, \qquad i \in [0, (N-1)/2)$$

$$\mathbf{g}_{2i} = 1, \qquad i \in [1, (N-1)/2)$$

$$\mathbf{g}_{N-1} = \mathbf{g}^{N-2}$$

$$\mathbf{g}_{N} = \mathbf{g}^{N}.$$

$$(6.2.44)$$

Let us begin by computing the untwisted partition function. The comarks of Spin(2N) are $a_i^{\vee} = 1$ for i = 0, 1, N - 1, N, and $a_i^{\vee} = 2$ for $i = 2, 3, \ldots, N - 2$. Plugging this into equation (6.2.5) we obtain the untwisted partition function

$$Z(q) = (1-q)^4 (1-q^2)^{N-3}, \qquad (6.2.45)$$

and, expanding, the Witten index

$$I_n = \begin{cases} (-1)^{n/2} \left[\binom{N-3}{n/2} - 6\binom{N-3}{n/2-1} + \binom{N-3}{n/2-2} \right] & n \text{ even,} \\ 4(-1)^{(n-1)/2} \left[\binom{N-3}{(n-1)/2-1} - \binom{N-3}{(n-1)/2} \right] & n \text{ odd.} \end{cases}$$
(6.2.46)

Note that the index vanishes when N is even and n corresponds to the time-reversal symmetric wall n = h/2 = N - 1. It also vanishes for the exceptional pairs (N, n) such that $2 + 4n + 2n^2 - 3N - 4nN + N^2 = 0$.

Let us now consider the index twisted by charge conjugation. Its action on the extended Dynkin diagram, and the resulting folded diagram, are as follows:

where the folded diagram has N nodes. From this we learn that the folded diagram has r' + 1 = N nodes, two of which have energy equal to 1, and the rest all energy equal to 2. Plugging this into (6.2.15) the zero-form twisted partition function is

$$Z^{\mathsf{c}}(q) = (1-q)^2 (1-q^2)^{N-2} \tag{6.2.48}$$

and, expanding, the twisted Witten index

$$I_n^{\mathsf{c}} = \begin{cases} (-1)^{n/2} \left[\binom{N-2}{n/2} - \binom{N-2}{n/2-1} \right] & n \text{ even,} \\ \\ 2(-1)^{(n+1)/2} \binom{N-2}{(n-1)/2} & n \text{ odd.} \end{cases}$$
(6.2.49)

One can also compute the partition function in the diagonal basis, where $c_i = +1$ for all the nodes except for the last one, which has $c_N = -1$. Plugging this into (6.2.16) yields the same expression for the twisted partition function.

Let us now consider the one-form-twisted partition functions. The symmetry depends on whether N is even or odd, which we consider in turn.

N even. Here the symmetry is $\mathbb{Z}_2 \times \mathbb{Z}_2$. g_1 and g_2 act as follows:



The folded diagrams are

$$C_{N-2}^{(1)}: \qquad \overset{2}{\longrightarrow} \overset{2$$

which have N - 1 and N/2 + 1 nodes, respectively. The folding by g_1g_2 is in fact identical to that of g_2 , i.e., the second diagram.

The twisted partition functions read

$$Z^{g_1}(q) = (1 - q^2)^{N-1}$$

$$Z^{g_2}(q) = (1 - q^2)^3 (1 - q^4)^{N/2 - 2},$$
(6.2.52)

and, expanding, the twisted Witten indices

$$I_n^{g_1} = \begin{cases} (-1)^{n/2} \binom{N-1}{n/2} & n \text{ even,} \\ 0 & n \text{ odd.} \end{cases}$$

$$I_n^{g_2} = \begin{cases} (-1)^{n/4} \left[\binom{N/2-2}{n/4} - 3\binom{N/2-2}{n/4-1} \right] & n \equiv 0 \mod 4, \qquad (6.2.53) \\ (-1)^{(n-2)/4} \left[\binom{N/2-2}{(n-2)/4-1} - 3\binom{N/2-2}{(n-2)/4} \right] & n \equiv 2 \mod 4, \\ 0 & n \text{ odd,} \end{cases}$$

while $I_n^{\mathbf{g}_1\mathbf{g}_2} = I_n^{\mathbf{g}_2}$.

N odd. Here the one-form symmetry is \mathbb{Z}_4 , whose action on the extended Dynkin diagram, and the corresponding folded diagram, are as follows:



where the comarks are all 4 if we fold by a generator of \mathbb{Z}_4 , and all 2 if we fold by a generator squared. The number of nodes is (N-1)/2 in the first case, and N-1 in the second case. The folded diagram corresponds to $C_{(N-1)/2}^{(1)}$ and $C_{N-1}^{(1)}$, respectively.

If we let g denote a generator of \mathbb{Z}_4 , the twisted partition function is

$$Z^{g}(q) = Z^{g^{3}}(q) = (1 - q^{4})^{(N-1)/2},$$

$$Z^{g^{2}}(q) = (1 - q^{2})^{N-1},$$
(6.2.55)

and, expanding, the twisted Witten indices

$$I_{n}^{g} = I_{n}^{g^{3}} = \begin{cases} (-1)^{n/4} \binom{(N-1)/2}{n/4} & n \equiv 0 \mod 4, \\ 0 & \text{otherwise,} \end{cases}$$

$$I_{n}^{g^{2}} = \begin{cases} (-1)^{n/2} \binom{N-1}{n/2} & n \text{ even,} \\ 0 & n \text{ odd.} \end{cases}$$
(6.2.56)

As usual, one may also compute these partition functions in the diagonal basis. Using the phases (6.2.44) in (6.2.16) yields the same expressions for the twisted partition functions, as expected.

6.2.5 $G = G_2$

 G_2 has no zero-form or one-form symmetry. The comarks for G_2 are $a_0^{\vee} = a_2^{\vee} = 1$ and $a_1^{\vee} = 2$. Plugging this into equation (6.2.5) we obtain the untwisted partition function

$$Z(q) = (1-q)^2 (1-q^2), \qquad (6.2.57)$$

and, expanding, the Witten indices

$$I_1 = -2 I_2 = 0 I_3 = 2.$$
(6.2.58)

Note that $I_3 = -I_1$, as expected from the action of time-reversal on domain walls.

6.2.6 Minimal Wall for Arbitrary Gauge Group

The domain wall theory for n = 1 admits a uniform description for all simply-connected groups, including the exceptional ones. Indeed, the only fermion configurations with total energy equal to 1, that is the solutions to (6.2.2)

$$\sum_{i=0}^{r} \lambda_i a_i^{\vee} = 1 , \qquad (6.2.59)$$

are clearly of the form $\lambda_i = 1$ for one *i* such that $a_i^{\vee} = 1$, and $\lambda_j = 0$ for all $j \neq i$. In other words, in each configuration there is only one excited fermion, which moreover necessarily has energy $a_i^{\vee} = 1$. All these configurations have the same fermion number, namely $(-1)^F = -1$, which means that the index is

$$I_1 \equiv -m_1 \tag{6.2.60}$$

where m_1 denotes the number of nodes in the extended Dynkin diagram $\mathfrak{g}^{(1)}$ with comark equal to 1. The values of m_1 are given in the following table:

G	SU(N)	$\operatorname{Sp}(N)$	$\operatorname{Spin}(2N+1)$	$\operatorname{Spin}(2N)$	E_6	E_7	E_8	F_4	G_2	(6 2 61)
m_1	N	N+1	3	4	3	2	1	2	2	(0.2.01)

Note that, for simply-laced G, m_1 is the order of Γ .

The index twisted by a symmetry $s \in S$ is

$$I_1^{\mathsf{s}} \equiv -m_1^{\mathsf{s}} \,, \tag{6.2.62}$$

where m_1^{s} denotes the number of nodes in the extended Dynkin diagram $\mathfrak{g}^{(1)}$ with comark equal to 1 that are fixed by \mathfrak{s} . m_1^{s} has already been computed for the classical groups $\mathrm{SU}(N), \mathrm{Sp}(N), \mathrm{Spin}(2N+1)$ and $\mathrm{Spin}(2N)$. For the exceptional groups, only E_6 and E_7 have non-trivial symmetry group S (see table 6.1). In E_6 , the zero-form charge-conjugation symmetry leaves invariant the extended node, which has comark 1, and permutes the other two nodes with comark 1. In E_6 and E_7 , the one-form center symmetry permutes all the nodes with comark 1. Therefore, letting c denote the non-trivial element of C , and g any non-trivial element of Γ , the indices are

$$E_6: I_1^{\mathbf{c}} = -1, \quad I_1^{\mathbf{g}} = 0 \qquad \mathbf{g} \in \Gamma = \mathbb{Z}_3 \\
 E_7: I_1^{\mathbf{g}} = 0 \qquad \mathbf{g} \in \Gamma = \mathbb{Z}_2.$$
(6.2.63)

The (twisted) indices for the exceptional groups E_6, E_7, E_8, F_4 and arbitrary *n* have been included in section 6.7 for completeness.

This concludes our discussion of the twisted Witten indices of the domain walls, as computed in terms of the the 4d ultraviolet degrees of freedom. A rather nontrivial consistency test of our proposal is that the Witten indices on the domain walls we just computed are reproduced by the corresponding partition functions of our conjectured 3d TQFTs. Computing the image of the (twisted) Witten indices in these TQFTs is nontrivial and requires understanding in detail the Hilbert space of spin TQFTs and the action of $(-1)^F$ on it, a subject to which we now turn.

6.3 Hilbert Space of Spin TQFTs

The domain wall theory preserves $3d \mathcal{N} = 1$ supersymmetry and observables depend on the choice of a spin structure. Therefore, the TQFT that emerges in the deep infrared of the domain wall must also depend on a choice of a spin structure, that is, it must be a spin TQFT [35].

The data of a TQFT in 3*d* includes the set of anyons \mathcal{A} (or Wilson lines) and the braiding matrix $B: \mathcal{A} \times \mathcal{A} \to U(1)$ encoding their braiding. A spin TQFT is a TQFT that has

an abelian¹⁰⁴ line ψ that braids trivially with all lines in \mathcal{A} and has half-integral spin.¹⁰⁵ Transparency of ψ implies that it fuses with itself into the vacuum, that is $\psi \times \psi = \mathbf{1}$. Since ψ is transparent and has half-integral spin, the observables of a spin TQFT depend on the choice of a spin structure.

A spin TQFT can be constructed from a parent bosonic TQFT which has an abelian, non-transparent fermion ψ with $\psi \times \psi = \mathbf{1}$, that is, a bosonic TQFT that has a \mathbb{Z}_2^{ψ} one-form symmetry generated by a fermion [36, 37, 120, 314, 315]. The bosonic parent theory defines a spin TQFT upon gauging its \mathbb{Z}_2^{ψ} one-form symmetry generated by ψ

spin TQFT =
$$\frac{\text{bosonic TQFT}}{\mathbb{Z}_2^{\psi}}$$
. (6.3.1)

This procedure is an extension of the notion of bosonic "anyon condensation" [285, 316, 317].¹⁰⁶ Upon gauging, the fermion ψ in the parent bosonic theory becomes the transparent fermion ψ in the spin TQFT. The gauged one-form symmetry \mathbb{Z}_2^{ψ} of the parent bosonic theory gives rise to an emergent zero-form symmetry \mathbb{Z}_2 in the spin TQFT that is generated by the fermion parity operator $(-1)^F$, and which acts on the "twisted sector". We will discuss the action of $(-1)^F$ on the Hilbert space of spin TQFTs shortly.

The lines of the parent bosonic theory \mathcal{A} can be arranged as the disjoint union of two sets $\mathcal{A} = \mathcal{A}_{NS} \cup \mathcal{A}_{R}$ according to their braiding with ψ . Lines in \mathcal{A}_{NS} , by definition, braid trivially with ψ while lines in \mathcal{A}_{R} have braiding -1 with ψ . This partitions the lines of the bosonic TQFT according to their \mathbb{Z}_{2}^{ψ} quantum number. The lines in each set can be organized into orbits of \mathbb{Z}_{2}^{ψ} , generated by fusion with ψ . The orbits can be either two- or one-dimensional. The lines in one-dimensional orbits are referred to as "Majorana lines" in that they can freely absorb the fermion ψ :

$$\psi \times m = m \,. \tag{6.3.2}$$

The Majorana lines, if any, are necessarily in $\mathcal{A}_{\rm R}$.¹⁰⁷ The lines of the bosonic parent theory thus split as

$$\mathcal{A}_{\rm NS} = \{\{a, a \times \psi\} \mid B(\psi, a) = +1\} \mathcal{A}_{\rm R} = \{\{x, x \times \psi\}, \ \{m\} \mid B(\psi, x) = B(\psi, m) = -1\}.$$
(6.3.4)

$$B(\psi,m) = e^{2\pi i (h_{\psi} + h_m - h_{\psi \times m})} = e^{2\pi h_{\psi}} = -1, \quad \Longrightarrow \quad m \in \mathcal{A}_{\mathbf{R}},$$
(6.3.3)

where h denotes spin of lines and in the second equality we have used the defining relation for a Majorana line (6.3.2). See also [120].

 $^{^{104}}$ An abelian line is one that yields a single line in its fusion with any line in \mathcal{A} .

 $^{^{105}}$ A TQFT that has a line with half-integral spin which braids nontrivially with at least one line in the theory is not spin. There is no unambiguous way to assign a sign to the fermion as we move it around a circle since the phase it acquires depends on which lines link with the circle, unlike when the fermion is transparent. 106 More precisely, the parent bosonic TQFT must be attached to a suitable 4d SPT phase so that the

combined system is non-anomalous, and the symmetry can be gauged.

 $^{^{107}}$ To prove this we compute the braiding of ψ with a Majorana line m and show that it necessarily has braiding -1 with ψ

The first set, referred to as the Neveu-Schwarz (NS) lines, is what is usually regarded as the set of Wilson line operators in the spin TQFT. The second set, the Ramond (R) lines, change the spin structure background. This decomposition will be useful shortly in the construction of the Hilbert space of the spin TQFT.

The Hilbert space of the spin TQFT on the spatial torus depends on the choice of spin structure. There are two equivalence classes of spin structures on the torus (or, more generally, on any Riemann surface): even and odd spin structures. Consider the even and odd spin structure Hilbert spaces \mathcal{H}_{NS-NS} and \mathcal{H}_{R-R} . \mathcal{H}_{NS-NS} correspond to choosing antiperiodic boundary conditions on the two circles while \mathcal{H}_{R-R} corresponds to periodic boundary conditions. The other two even spin-structure Hilbert spaces \mathcal{H}_{NS-R} and \mathcal{H}_{R-NS} can be obtained from \mathcal{H}_{NS-NS} by the action of the mapping class group. This group is a non-trivial extension of the modular group $SL_2(\mathbb{Z})$ by the \mathbb{Z}_2 fermion parity symmetry. It is known as the *metaplectic group* $Mp_1(\mathbb{Z})$. It does not preserve the individual spin structures but it does preserve their equivalence class. The Hilbert spaces of spin TQFTs realize a unitary representation of this group.

The states in the Hilbert space \mathcal{H}_B of a bosonic TQFT are constructed from the path integral on a solid torus by inserting lines $M \in \mathcal{A}$ along the non-contractible cycle [22]. This defines conformal blocks on the torus. We represent this pictorially by

$$|M\rangle = \Big| \bigotimes_{M} \Big\rangle \in \mathcal{H}_B \,. \tag{6.3.5}$$

The Hilbert space of the spin TQFT can be constructed from its definition as a quotient of the bosonic parent TQFT (6.3.1).¹⁰⁸ The states in $\mathcal{H}_{\text{NS-NS}}$ are labeled by $a \in \mathcal{A}_{\text{NS}}$, and are represented as

$$|a\rangle_{\rm spin} = \frac{1}{\sqrt{2}} \left(\left| \bigotimes_{a} \right\rangle + \left| \bigotimes_{a \times \psi} \right\rangle \right) \in \mathcal{H}_{\rm NS-NS}.$$
 (6.3.6)

The states in $\mathcal{H}_{\text{R-R}}$ are constructed from conformal blocks of the bosonic parent TQFT on the torus and, in the presence of Majorana lines (6.3.2), from the once-punctured torus conformal blocks of the bosonic parent TQFT. By virtue of m being a Majorana line obeying the fusion rule $\psi \times m = m$, the one-point conformal block on the torus with m along the cycle and ψ at the puncture is nontrivial, as it is allowed by the fusion rules. The states in

¹⁰⁸More details of the explicit construction of the Hilbert space of spin TQFTs will appear elsewhere [1].

 $\mathcal{H}_{\text{R-R}}$ are labeled by $x, m \in \mathcal{A}_{\text{R}}$, and are represented as¹⁰⁹

$$|x\rangle_{\rm spin} = \frac{1}{\sqrt{2}} \left(\left| \bigotimes_{x} \right\rangle - \left| \bigotimes_{x \times \psi} \right\rangle \right) \in \mathcal{H}_{\rm R-R},$$

$$|m\rangle_{\rm spin} = \left| \bigotimes_{m}^{\psi} \right\rangle \in \mathcal{H}_{\rm R-R}.$$

$$(6.3.7)$$

Modular transformations preserve the odd spin structure, i.e., they map \mathcal{H}_{R-R} into itself. The negative sign in (6.3.7) guarantees that under under modular transformations the states in \mathcal{H}_{R-R} are mapped into themselves.

Note that the pair of lines a and $a \times \psi$ in the bosonic parent descend to a pair of lines in the spin TQFT, because these are distinct anyons, being distinguishable by their spin. On the other hand, the pair of states $|a\rangle$ and $|a \times \psi\rangle$ descend to a *single* state in the spin TQFT. Thus, while in a bosonic TQFT the number of states is the same as the number of lines, in a spin TQFT there are twice as many lines as there are states.

Our next task is to compute the action of fermion parity, i.e. $(-1)^F$, on the Hilbert space of the spin TQFT. The \mathbb{Z}_2 symmetry generated by $(-1)^F$ is the emergent zero-form symmetry that appears upon quotienting the parent bosonic theory by \mathbb{Z}_2^{ψ} in (6.3.1). The charged states are therefore those constructed from the once-punctured torus in the bosonic theory

$$(-1)^{F} |m\rangle_{\rm spin} = -|m\rangle_{\rm spin},$$

$$(-1)^{F} |a\rangle_{\rm spin} = +|a\rangle_{\rm spin},$$

$$(-1)^{F} |x\rangle_{\rm spin} = +|x\rangle_{\rm spin}.$$

(6.3.8)

 $(-1)^F$ acts nontrivially on the ψ puncture in the once-punctured torus.

Depending on the choice of spin structure on the "time" circle we can define the following $2^3 = 8$ partition functions for spin TQFTs:¹¹⁰

$$\begin{aligned} \operatorname{tr}_{-,-}(\mathcal{O}) &\equiv \operatorname{tr}_{\mathcal{H}_{\mathrm{NS-NS}}}(\mathcal{O}) , \\ \operatorname{tr}_{-,+}(\mathcal{O}) &\equiv \operatorname{tr}_{\mathcal{H}_{\mathrm{NS-R}}}(\mathcal{O}) , \\ \operatorname{tr}_{+,-}(\mathcal{O}) &\equiv \operatorname{tr}_{\mathcal{H}_{\mathrm{R-NS}}}(\mathcal{O}) , \\ \operatorname{tr}_{+,+}(\mathcal{O}) &\equiv \operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}}(\mathcal{O}) , \end{aligned}$$

$$\end{aligned}$$

$$(6.3.9)$$

¹⁰⁹Unlike in bosonic anyon condensation, where a fixed line in the parent theory yields multiples states in the quotient theory, a Majorana line is in an irreducible representation of Cliff(1|1) and yields a unique state in the quotient (spin) TQFT.

 $^{^{110}-}$ represents antiperiodic boundary condition while + period boundary conditions.
and

$$\operatorname{tr}_{-,-}((-1)^{F}\mathcal{O}) \equiv \operatorname{tr}_{\mathcal{H}_{\mathrm{NS-NS}}}((-1)^{F}\mathcal{O}) ,$$

$$\operatorname{tr}_{-,+}((-1)^{F}\mathcal{O}) \equiv \operatorname{tr}_{\mathcal{H}_{\mathrm{NS-R}}}((-1)^{F}\mathcal{O}) ,$$

$$\operatorname{tr}_{+,-}((-1)^{F}\mathcal{O}) \equiv \operatorname{tr}_{\mathcal{H}_{\mathrm{R-NS}}}((-1)^{F}\mathcal{O}) ,$$

$$\operatorname{tr}_{+,+}((-1)^{F}\mathcal{O}) \equiv \operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}}((-1)^{F}\mathcal{O}) ,$$

$$(6.3.10)$$

where \mathcal{O} is an operator in the theory. We will be interested in the case when \mathcal{O} is a symmetry of the TQFT. We note that $(-1)^F$ is only non-trivial in the R-R sector, because this is the only Hilbert space that may contain Majorana states. This is the most subtle and rich sector, and the one of interest as far as the twisted Witten indices is concerned.

6.3.1 Partition Function of Spin TQFTs

The twisted Witten index of the domain wall theory is computed by considering the odd spin structure on the spatial torus and periodic boundary condition on the time circle. This implies that the twisted Witten indices, computed via the 4d ultraviolet fields, must be reproduced by appropriate odd-spin-structure partition functions of our conjectured infrared 3d spin TQFTs. In other words, a nontrivial check that our proposed infrared spin TQFTs describe the *n*-domain wall theories is proving that

$$I_n^{\mathbf{s}} \equiv \operatorname{tr}_{\mathcal{H}_{\mathbf{R}-\mathbf{R}}}(-1)^F \mathbf{s}$$
(6.3.11)

for symmetries $s \in S$. This requires, in particular, identifying the image of the symmetries $s \in S$ in the infrared TQFT.

Let us begin by considering the untwisted partition function. Given the construction of the Hilbert space $\mathcal{H}_{\text{R-R}}$ in (6.3.7) and the action of $(-1)^F$ in (6.3.8) we can compute the desired partition function as follows

$$\operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}}(-1)^F = N_x - N_m \,.$$
 (6.3.12)

This requires determining in the bosonic parent theory the number N_x of two-dimensional orbits and the number N_m of one-dimensional orbits (the number of Majorana lines) in $\mathcal{A}_{\rm R}$ (see (6.3.4)) under fusion with ψ .

Let us illustrate this in a simple example. The simplest spin TQFT is $SO(N)_1$ Chern-Simons theory, which is a trivial, invertible spin TQFT with lines $\{1, \psi\}$. The bosonic parent theory is $Spin(N)_1$ Chern-Simons theory:

• For N odd, $\operatorname{Spin}(N)_1$ is the Ising category, which has three lines $\{\mathbf{1}, \sigma, \psi\}$: the vacuum $\mathbf{1}$, the spin operator σ , and the energy operator ψ . These primaries have spins $h = 0, \frac{N}{16}, \frac{1}{2}$, and fusion rules $\sigma^2 = \mathbf{1} + \psi, \psi^2 = \mathbf{1}, \psi \times \sigma = \sigma$.

• For N even, $\operatorname{Spin}(N)_1$ has four lines, with spins $h = 0, \frac{N}{16}, \frac{N}{16}, \frac{1}{2}$, and which we denote by $\{\mathbf{1}, \mathbf{e}, \mathbf{m}, \psi\}$, which correspond to the trivial representation, the two fundamental spinor representations, and the vector representation, respectively. The theory for $N \equiv 0$ mod 4 has \mathbb{Z}_2^2 fusion rules, with $\mathbf{e}^2 = \mathbf{m}^2 = \psi^2 = \mathbf{1}$ and $\mathbf{e} \times \mathbf{m} = \psi$. For $N \equiv 2 \mod 4$ the fusion ring is \mathbb{Z}_4 , with $\mathbf{e}^2 = \mathbf{m}^2 = \psi$, and $\psi^2 = \mathbf{e} \times \mathbf{m} = \mathbf{1}$.

The theory $SO(N)_1$ is obtained by condensing ψ , that is $SO(N)_1 = Spin(N)_1/\mathbb{Z}_2^{\psi}$. Using the fusion rules and the spins we see that $\mathbf{1}, \psi$ are neutral under \mathbb{Z}_2^{ψ} and \mathbf{e}, \mathbf{m} and σ are charged. In other words, the Neveu-Schwarz sector is

$$\mathcal{A}_{\rm NS} = \{\mathbf{1}, \psi\}, \qquad (6.3.13)$$

while the Ramond sector is

$$\mathcal{A}_{\mathrm{R}} = \begin{cases} \{\sigma\} & N \text{ odd,} \\ \{\mathsf{e},\mathsf{m}\} & N \text{ even.} \end{cases}$$
(6.3.14)

This implies that $N_a = 1$ in the NS sector. On the other hand, in the R sector, $(N_x, N_m) = (0, 1)$ for odd N and $(N_x, N_m) = (1, 0)$ for even N. Thus there is a unique state in each spin structure, and all states are bosonic except in $\mathcal{H}_{\text{R-R}}$, where $(-1)^F = (-1)^N$, since there is a Majorana line for odd N. The $(-1)^F$ odd state is created by the well-known once-punctured torus conformal block in the Ising category with the insertion of σ , and ψ at the puncture.

For a second example let us now consider the spin TQFT $SO(3)_3$ Chern-Simons theory, which is the simplest non-trivial spin TQFT. The bosonic parent theory is $SU(2)_6$ since

$$SO(3)_3 = \frac{SU(2)_6}{\mathbb{Z}_2^{\psi}},$$
 (6.3.15)

where the abelian line ψ is the line in SU(2)₆ with j = 3 and spin h = 3/2. The lines in $\mathcal{A} = \{j = 0, \frac{1}{2}, 1, \ldots, 3\}$ which have braiding -1 with ψ are those with half-integral isospin: $\mathcal{A}_{\mathrm{R}} = \{j = \frac{1}{2}, \frac{3}{2}, \frac{5}{2}\}$. Under fusion with ψ we have the following \mathbb{Z}_{2}^{ψ} orbits

$$3 \times \frac{1}{2} = \frac{5}{2}
3 \times \frac{3}{2} = \frac{3}{2}.$$
(6.3.16)

Therefore, in the R-R sector of SO(3)₃ there is a length-2 orbit with $(j = \frac{1}{2}, \frac{5}{2})$ and a Majorana line with $j = \frac{3}{2}$. Thus, $N_x = N_m = 1$. There are $N_x + N_m = 2$ states, but one of them is a boson and the other is a fermion, which means that the partition function with periodic boundary conditions vanishes

$$\operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}}(-1)^F = N_x - N_m \equiv 0.$$
 (6.3.17)

The vanishing of this trace will be important when discussing the 2-domain wall theory in 4d $\mathcal{N} = 1$ SYM with gauge group G_2 (cf. section 6.4.4). This example clearly illustrates the importance of looking at the appropriate partition function and not merely at the dimension of the Hilbert space.

We now discuss a different way to compute the partition function that does not rely on computing N_x and N_m directly. The basic idea is to gauge the emergent zero-form \mathbb{Z}_2 symmetry generated by $(-1)^F$ in the spin TQFT to obtain the bosonic parent theory back [11, 36, 37, 66, 318]

$$\frac{\text{spin TQFT}}{\mathbb{Z}_2} = \text{bosonic TQFT}.$$
(6.3.18)

Gauging this \mathbb{Z}_2 amounts to summing the spin TQFT over all spin structures of the threemanifold M. Taking M to the three-torus, and summing over the $2^3 = 8$ spin structures, corresponding to either periodic or antiperiodic boundary conditions around each of the three circles, we find that

$$\frac{1}{2} \sum_{\pm,\pm} (\operatorname{tr}_{\pm,\pm}(\mathbf{1}) + \operatorname{tr}_{\pm,\pm}(-1)^F) = \operatorname{tr}_{\mathcal{H}_B}(\mathbf{1}).$$
(6.3.19)

 $tr_{\pm,\pm}$ (see (6.3.9)–(6.3.10)) denotes the trace over the Hilbert space on the spatial torus with boundary conditions \pm,\pm , and $tr_{\mathcal{H}_B}$ the trace over the torus Hilbert space of the bosonic parent theory.

Using the fact that the dimension of the torus Hilbert space is the same in all spin structures and that $(-1)^F$ acts nontrivially only in \mathcal{H}_{R-R} (see (6.3.8)), we find the formula

$$\operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}}(-1)^F = 2\dim(\mathcal{H}_B) - 7\dim(\mathcal{H}_F), \qquad (6.3.20)$$

where dim(\mathcal{H}_B) is the dimension of the torus Hilbert space of the bosonic parent TQFT and dim(\mathcal{H}_F) the dimension of the torus Hilbert space in any one spin structure of the spin TQFT. This formula offers a significant advantage in that it requires computing the total number of states dim(\mathcal{H}_F) = $N_x + N_m$, and not separately N_x and N_m , as in formula (6.3.12). Even simpler, one may compute dim(\mathcal{H}_F) = N_a in the NS-NS sector directly, where all orbits are of length-2: the number of states is just half the number of lines of the spin TQFT.¹¹¹

As a consistency check, consider the case where G_F is the product of a bosonic theory \tilde{G} times a trivial/invertible spin TQFT

$$G_F = \hat{G} \times \mathrm{SO}(N)_1, \qquad (6.3.21)$$

whose bosonic parent is $G_B = \tilde{G} \times \text{Spin}(N)_1$. As $\text{SO}(N)_1$ is a trivial spin TQFT, we get $\dim(\mathcal{H}_F) = \dim(\mathcal{H}_{\tilde{G}})$. Similarly, using that $\text{Spin}(N)_1$ has a four-dimensional Hilbert space if N is even, and a three dimensional Hilbert space if N is odd, we get $\dim(\mathcal{H}_B) = \frac{1}{2}(7 + (-1)^N) \dim(\mathcal{H}_{\tilde{G}})$. Plugging this into (6.3.20), we get

$$\operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}^{G_F}}(-1)^F = (-1)^N \dim(\mathcal{H}_{\tilde{G}}),$$
 (6.3.22)

¹¹¹In the SO(3)₃ example dim(\mathcal{H}_B) = 7 and dim(\mathcal{H}_F) = 2, and using (6.3.20) the partition function indeed vanishes.

which is precisely what one would expect, given the tensor product structure of G_F and the fact that the trace over $SO(N)_1$ is $(-1)^N$. Put differently, in the Hilbert space of $G_F = \tilde{G} \times SO(N)_1$ we have $N_a = N_x$ and $N_m = 0$ for N even, and $N_a = N_m$ and $N_x = 0$ for N odd. That is, in the R-R sector, either no states are Majorana or all are, depending on the parity of N. This implies that

$$\operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}^{G_F}}(-1)^F = \begin{cases} +N_a & N \text{ even} \\ -N_a & N \text{ odd} , \end{cases}$$
(6.3.23)

which indeed equals (6.3.22).

There are spin TQFTs which factorize in a nontrivial fashion into the product of a bosonic TQFT and a trivial spin TQFT by virtue of a level-rank duality, as for example $U(1)_k \leftrightarrow SU(k)_{-1} \times \{\mathbf{1}, \psi\}$. In these theories $\operatorname{tr}_{\mathcal{H}_{R-R}}(-1)^F$ also just measures the dimension of the Hilbert space (up to possibly a global sign).

More generally, stacking a TQFT, spin or bosonic, with a trivial spin TQFT defines

$$\operatorname{TQFT} \times \operatorname{SO}(N)_1$$
. (6.3.24)

This theory has the same number of states as the original TQFT but $SO(N)_1$ can change the global sign of the action of $(-1)^F$ on all states in \mathcal{H}_{R-R} . The partition functions are the same up to possibly a sign

$$\operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}^{\mathrm{TQFT}\times\mathrm{SO}(N)_{1}}}(-1)^{F} = (-1)^{N} \operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}^{\mathrm{TQFT}}}(-1)^{F}.$$
(6.3.25)

Indeed, the single state of $SO(N)_1$ in \mathcal{H}_{R-R} is a Majorana state and thus has odd fermion parity for N odd only. Therefore, when comparing the spin TQFT partition function with the Witten index of the domain wall, we will match their absolute values, as if those match, the signs can be also be matched by stacking a suitable trivial spin TQFT, which can be thought of as a purely gravitational counterterm [111].¹¹²

The generalization to twisted indices is straightforward. Given a symmetry $s \in S_{TQFT}$ of the TQFT, which acts $s: \mathcal{H}_{R-R} \to \mathcal{H}_{R-R}$, the partition function

$$\operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}}((-1)^F \mathbf{s}) \tag{6.3.26}$$

counts the number of bosons fixed by s, minus the number of fermions fixed by s. That being said, there are some subtleties that must be kept in mind. A state fixed by s does not necessarily contribute with s = +1 to the trace – it might contribute with s = -1 instead, the reason being that the symmetry s might be realized projectively in the Hilbert space.

The most common example where this may happen is charge-conjugation c. We can illustrate this in U(1)₁ Chern-Simons theory, the simplest theory where this phenomenon

¹¹²Staking SO(N)₁ for odd N to a 3d theory has the same effect as stacking to a 2d theory the trivial spin TQFT known as the Arf-invariant, which changes the sign of the partition with odd spin structure.

occurs. This is an invertible spin TQFT, which means that it has a unique state on any spin structure. This state is clearly fixed by \mathbf{c} but, interestingly, it has $\mathbf{c} = -1$ in the odd-spin-structure Hilbert space $\mathcal{H}_{\text{R-R}}$. We can show this as follows. The bosonic parent theory is U(1)₄ Chern-Simons theory, which has four states, labeled by q = 0, 1, 2, 3. The U(1)₁ theory is obtained by condensing the fermion ψ , which has q = 2. The Ramond lines are easily checked to be q = 1, 3, and they are paired by fusion with ψ into a single two-dimensional orbit, since $1 \times 2 = 3$. Thus, the unique state in the R-R sector is (cf. (6.3.7))

$$|1\rangle_{\rm spin} = \frac{1}{\sqrt{2}}(|1\rangle - |3\rangle).$$
 (6.3.27)

This indeed satisfies $c|1\rangle_{spin} = -|1\rangle_{spin}$, inasmuch as $c: q \mapsto -q \mod 4$ in the bosonic parent, which exchanges $|1\rangle$ and $|3\rangle$.

6.4 Domain Wall TQFT Partition Functions

In this section we calculate partition functions twisted by a symmetry $s \in S_{\mathrm{TQFT}}$

$$\operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}}(-1)^{F}\mathsf{s} \tag{6.4.1}$$

of the Chern-Simons TQFTs we proposed emerge in the infrared of the domain wall theories (see section 6.1). Our calculations beautifully reproduce the results obtained in section 6.2. Namely, we will now demonstrate that the trace (6.4.1) agrees with the twisted Witten index on the *n*-domain wall W_n (cf. (6.2.10)) as computed in terms of the original 4*d* fields

$$\operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}}(-1)^F \mathbf{s} \equiv I_n^{\mathbf{s}} \,. \tag{6.4.2}$$

We identify each symmetry $s \in S$ in $4d \mathcal{N} = 1$ SYM with a symmetry $s \in S_{TQFT}$ in the infrared TQFT.

In $4d \ \mathcal{N} = 1$ SYM with gauge group $\operatorname{Sp}(N)$ our proposed domain wall theory corresponds to a Chern-Simons theory based on a group that is simple, connected, and simply-connected, whereas for SYM with $\operatorname{SU}(N)$, $\operatorname{Spin}(N)$ and G_2 gauge groups, the proposed infrared Chern-Simons theories are based on a group that is neither. We discuss both cases in turn.

Chern-Simons theory G_k , with G simple, connected and simply-connected is always a bosonic TQFT. These theories are made spin by tensoring with the trivial spin TQFT SO $(N)_1$. It follows from our discussion in section 6.3 that

$$\operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}^{G_k \times \mathrm{SO}(N)_1}} (-1)^F = (-1)^N \operatorname{tr}_{G_k}(\mathbf{1}), \qquad (6.4.3)$$

since all states have the same fermion parity – all bosonic, or all fermionic, depending on the parity of N. Therefore the partition function of $G_k \times SO(N)_1$ in (6.4.3) is, up to possibly a sign, the dimension of the Hilbert space of G_k Chern-Simons theory on the two-torus.

The states in the torus Hilbert space of G_k Chern-Simons theory are conformal blocks on the torus, which are labeled by the integrable representations of the corresponding affine lie algebra $\mathfrak{g}^{(1)}$ at level k [22, 319]. By definition, the representations of G that are integrable are those whose highest weight λ satisfies $(\lambda, \theta) \leq k$, with θ the highest root of G. Expanding the latter in a basis of simple coroots, and introducing an extended label $\lambda_0 := k - (\lambda, \theta)$, integrability can be expressed as

$$\sum_{i=0}^{r} \lambda_i a_i^{\vee} = k, \qquad \lambda_i \in \mathbb{Z}_{\ge 0}.$$
(6.4.4)

The dimension of the Hilbert space $\operatorname{tr}_{G_k}(1)$ is equal to the number of solutions to this equation.

Much like the discussion in section 6.2, where the Witten index on the domain wall was computed through an auxiliary system of free fermions, $\operatorname{tr}_{G_k}(\mathbf{1})$ has a nice combinatorial interpretation in terms of a system of free bosons in 0 + 1 dimensions. Indeed, the number of integrable representations $\operatorname{tr}_{G_k}(\mathbf{1})$ is the number of ways of creating a state of energy k from r+1 free bosons, each with energy a_i^{\vee} . Each boson is associated with a node in the extended Dynkin diagram $\mathfrak{g}^{(1)}$, and $\lambda_i \in \{0, 1, 2, ...\}$ in (6.4.4) corresponds to the occupation number of the *i*-th boson. Introducing a fugacity parameter q defines a generating function, which is the partition function of the bosons on the circle

$$Z(G,q) \equiv \sum_{k\geq 0} \operatorname{tr}_{G_k}(\mathbf{1}) q^k \,. \tag{6.4.5}$$

The partition function is thus

$$Z(G,q) = \prod_{i=0}^{r} (1 - q^{a_i^{\vee}})^{-1}.$$
(6.4.6)

The Chern-Simons trace $\operatorname{tr}_{G_k}(1)$ is the coefficient of q^k in (6.4.6).

In a similar fashion, we define the trace twisted by a symmetry $s \in S_{TQFT}$ of G_k Chern-Simons theory:

$$\operatorname{tr}_{G_k}(\mathsf{s}). \tag{6.4.7}$$

When $\mathbf{s} = \mathbf{c}$ is a zero-form symmetry, this corresponds to inserting a surface operator, i.e., the symmetry defect is supported on the whole spatial torus. On the other hand, if $\mathbf{s} = \mathbf{g}$ denotes a one-form symmetry, the symmetry defect is a line operator, and one must specify a homology cycle on the torus on which it is is supported. The states of G_k Chern-Simons are created by wrapping on a cycle Wilson lines labeled by integrable representations λ ; if \mathbf{g} is supported on the same cycle, it acts on the states via fusion:



Conversely, if **g** is supported on the dual cycle, it acts on the states via braiding:



where $\alpha_{\mathbf{g}}(\lambda)$ is the charge of λ under the center of G (cf. (6.2.7)). More generally, one can wrap a pair of symmetry defects on both cycles, but one can always conjugate such configuration via a modular transformation to either of the two options above. This operation, being a similarity transformation, does not affect the value of the trace. In other words, the value of $\operatorname{tr}_{G_k}(\mathbf{g})$ is independent of which cycle we define \mathbf{g} on.

When **s** is a symmetry of the classical action of G_k Chern-Simons theory, it is induced by an outer automorphism of the extended Dynkin diagram $\mathfrak{g}^{(1)}$, and it acts as a permutation of the nodes thereof. In that case, **s** induces an action on the system of bosons, which permutes them in the same way it permutes the nodes of the Dynkin diagram. As in the system of free fermions, the trace above can be obtained from the partition function of these bosons

$$Z^{\mathsf{s}}(G,q) \equiv \sum_{k \ge 0} \operatorname{tr}_{G_k}(\mathsf{s})q^k \,, \tag{6.4.10}$$

where $Z^{s}(G,q)$ denotes the bosonic partition function twisted by the permutation s. One can evaluate this partition function by the same methods as in section 6.2, i.e., by diagram folding or directly in the diagonal basis

$$Z^{\mathsf{s}}(G,q) = \prod_{i=0}^{r} (1 - \mathsf{s}_i q^{a_i^{\vee}})^{-1}, \qquad (6.4.11)$$

where \mathbf{s}_i are the eigenvalues of the permutation. Note that, for $\mathbf{s} = \mathbf{g}$ a one-form symmetry, diagram folding naturally corresponds to \mathbf{g} acting as a permutation, i.e. (6.4.8), while the diagonal action corresponds to \mathbf{g} acting via braiding, i.e. (6.4.9). Indeed, it is a well-known fact that an S modular transformation – which interchanges the two cycles – diagonalizes the fusion rules.

It should be noted that Chern-Simons theories can have "quantum symmetries". These are symmetries of the entire TQFT data that are not symmetries of the Lagrangian. Many explicit examples of these symmetries have been found in [4]. These symmetries permute the Wilson lines of the theory, in a way that does not necessarily correspond to a permutation of their Dynkin labels. As such, the free boson representation cannot be used to evaluate the twisted trace, but it must be computed from the action of the symmetry on the Hilbert space of the TQFT. That being said, we find that the symmetries S in the ultraviolet domain wall map to classical symmetries of the infrared Chern-Simons theories, and we can compute the twisted index using (6.4.11).

Some of our proposed domain wall TQFTs are Chern-Simons theory with a group G that is not connected and/or simply-connected, in which case the theory G_k (with k a set of integers that defines the Chern-Simons action) may depend on the spin structure of the underlying manifold. There are four distinct Hilbert spaces corresponding to the four spin structures on the spatial torus (see section 6.3), but our interest here is in the Hilbert space $\mathcal{H}_{\text{R-R}}$. Now G_k can have fermionic states, which correspond to once-punctured conformal blocks of the parent bosonic theory, and $(-1)^F$ is in general a non-trivial operator. Our goal is to compute

$$\operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}^{G_k}}((-1)^F \mathbf{s}) \equiv \operatorname{tr}_{G_k}((-1)^F \mathbf{s}), \qquad (6.4.12)$$

where we use the latter to simplify notation.

We shall next compute the twisted traces for all the Chern-Simons theories of interest. We begin by considering the simply-connected group Sp(n), and then we move on to the more subtle and interesting cases U(n), O(n). We finally make a few remarks concerning the exceptional groups. The remaining simply-connected groups SU(n), Spin(n) as well as SO(n)are studied in section 6.6.

6.4.1 $G = \operatorname{Sp}(n)$

The *n*-domain wall theory for $4d \mathcal{N} = 1$ SYM with $G = \operatorname{Sp}(N)$ is proposed to be $\operatorname{Sp}(n)_{N+1-n}$ Chern-Simons theory. Let us proceed to study the partition functions of $\operatorname{Sp}(n)_k$.

Consider the algebra $C_n = \mathfrak{sp}_n$. The comarks are all $a_i^{\vee} = 1$. Plugging this into (6.4.11) we obtain the generating function as

$$Z(Sp(n),q) = (1-q)^{-(n+1)}, \qquad (6.4.13)$$

and, by expanding, the untwisted trace

$$\operatorname{tr}_{\operatorname{Sp}(n)_k}(\mathbf{1}) = \binom{n+k}{k}.$$
(6.4.14)

This is the number of integrable representations of $\text{Sp}(n)_k$, that is, the dimension of the torus Hilbert space of this Chern-Simons theory.

Let us also compute the partition function twisted by the one-form symmetry $\Gamma = \mathbb{Z}_2$. This symmetry reverses the order of the extended labels, and the charged representations are the pseudo-real ones. Denoting by **g** the non-trivial element of \mathbb{Z}_2 , and using (6.4.11) and (6.2.30), we get the twisted partition function

$$Z^{\mathsf{g}}(\mathrm{Sp}(n), q) = (1-q)^{-\lfloor n/2 \rfloor - 1} (1+q)^{-\lceil n/2 \rceil}, \qquad (6.4.15)$$

and by expanding

$$\operatorname{tr}_{\operatorname{Sp}(n)_{k}}(\mathbf{g}) = \begin{cases} 0 & n \text{ odd, } k \text{ odd,} \\ \binom{(n+k-1)/2}{k/2} & n \text{ odd, } k \text{ even,} \\ \binom{(n+k-1)/2}{(k-1)/2} & n \text{ even, } k \text{ odd,} \\ \binom{(n+k)/2}{k/2} & n \text{ even, } k \text{ even.} \end{cases}$$
(6.4.16)

Note that Z^{g} is nothing but the untwisted partition function associated to the Dynkin diagram given by folding the original diagram by the one-form symmetry (6.2.33).

We are now ready to test our proposal. Recall that the conjectured infrared theory corresponding to the *n*-domain wall of Sp(N) SYM was $W_n = \text{Sp}(n)_k$, with k = N + 1 - n. Using this value of the level in (6.4.14) and (6.4.16) indeed reproduces the (twisted) Witten indices computed in the ultraviolet, cf. (6.2.32) and (6.2.36).

6.4.2 G = U(n)

The *n*-domain wall theory for $4d \mathcal{N} = 1$ SYM with G = SU(N) is proposed to be $U(n)_{N-n,N}$ Chern-Simons theory. Let us proceed to study the partition functions of $U(n)_{k,n+k}$.

The Chern-Simons gauge group is not simply connected. The theory is defined as

$$U(n)_{k,n+k} := \frac{SU(n)_k \times U(1)_{n(n+k)}}{\mathbb{Z}_n}, \qquad (6.4.17)$$

where \mathbb{Z}_n is the one-form symmetry generated by the line $\psi = [0, k, 0, \dots, 0] \otimes (n+k)$. Here and in what follows, $[\lambda_0, \lambda_1, \dots, \lambda_{n-1}]$ denotes the Dynkin labels of an $\mathrm{SU}(n)_k$ representation, and $(q) \in \mathbb{Z}$ the charge of a $\mathrm{U}(1)_{n(n+k)}$ representation. The spin of the generator is easily computed to be $h_{\psi} = \frac{k(n-1)}{2n} + \frac{n+k}{2n} = \frac{k+1}{2}$. This theory is spin if and only if k is even.

We now proceed to compute the relevant traces. As the theory can be a spin TQFT, the theory may contain fermionic states, and $(-1)^F$ will in general be a non-trivial operator, which we need to understand to compute $tr((-1)^F s)$. In other words, we have to identify which of the states of this Chern-Simons theory are bosons, and which are fermions.

Note that, unlike the general discussion of section 6.3, this theory is more conveniently presented as a \mathbb{Z}_n quotient rather than a \mathbb{Z}_2 quotient, so let us slightly generalize the discussion in that section to such quotients. In section 6.3 we argued that the bosonic states after a \mathbb{Z}_2 fermionic quotient are the length-2 orbits, while the fermions are the fixed-points. We now claim that the general statement for \mathbb{Z}_n fermionic quotients is that the bosonic states are the orbits of even length, while the fermions are the orbits of odd length.

To prove this claim, consider a general spin TQFT that can be written as

$$G_F = \frac{G_B}{\mathbb{Z}_n},\tag{6.4.18}$$

where G_B is some bosonic TQFT, and where \mathbb{Z}_n is a one-form symmetry generated by a fermion ψ . Since $h_{\psi^p} = ph_{\psi}$, ψ^p is a fermion if p is odd, and a boson if p is even. This means that n is necessarily even, because $\psi^n = \mathbf{1}$ is a boson.

The braiding phase with respect to ψ is always an *n*-th root of unity, it is the charge with respect to the \mathbb{Z}_n symmetry. This fact allows us to partition the lines of G_B into *n* equivalence classes according to their *n*-ality *j*, i.e., the value of braiding $B(\alpha, \psi) = e^{2\pi i j/n}$, $j = 0, 1, \ldots, n - 1$. The lines with j = 0 are the NS lines (so that $B(\alpha, \psi) = +1$), and those with j = n/2 are the R lines (so that $B(\alpha, \psi) = -1$). The rest of lines are projected out by the \mathbb{Z}_n quotient (unless we turn on a suitable background for the dual \mathbb{Z}_n zero-form symmetry). Furthermore, in each sector the lines are organized into \mathbb{Z}_n orbits,

$$\{\alpha, \psi\alpha, \psi^2\alpha, \dots, \psi^{|\alpha|-1}\alpha\},\tag{6.4.19}$$

where $|\alpha| \in [1, n]$ denotes the length of the orbit – the minimal integer such that $\psi^{|\alpha|} \times \alpha = \alpha$. An orbit is Majorana if and only if its length is odd, for then and only then it may absorb a fermion. Indeed, the conformal block with puncture $\psi^{|\alpha|}$ is non-vanishing only if $\psi^{|\alpha|} \times \alpha = \alpha$. In conclusion, the fermionic states in the R-R sector of the quotient theory G_F correspond to the orbits of G_B R-lines with an odd number of elements, as claimed.

We are now in position to study the theory $U(n)_{k,n+k}$. The discussion above has taught us how to identify fermionic states in the Hilbert space of the theory. Rather anticlimactically, we shall now argue that this theory has, in fact, no fermionic states at all! This means that the trace $tr(-1)^F$ actually just counts the number of states of the theory, much like in a bosonic theory. This explains why the counting of states in [46] matched the domain wall index – because all states are bosonic. This, importantly, is not always the case for other spin TQFTs, such as O(n) (see below, section 6.4.3).

Let us prove that the theory has no fermionic states. $U(n)_{k,n+k}$ is level-rank dual to $U(k)_{-n,-(n+k)}$ as a spin TQFT. Therefore, if either k or n is odd, the theory factorizes as a bosonic theory times an invertible spin TQFT, and so the theory clearly has no Majorana states. The only non-trivial case is, therefore, that of n, k both even, which we assume in what follows.

The theory in the numerator of the quotient description of $U(n)_{k,n+k}$ in (6.4.17) is bosonic (recall that $U(1)_K$ is spin for K odd and bosonic for K even; here K = n(n+k), which is even). The states of $U(n)_{k,n+k}$ are \mathbb{Z}_n orbits of $SU(n)_k \times U(1)_{n(n+k)}$ representations. If we manage to prove that there are no orbits of odd length, we succeed in proving that the theory has no Majorana states. In fact, we show that, more generally, all orbits have length-n, i.e., all orbits are long. This implies that the states correspond to conformal blocks with no punctures, i.e., all states are bosonic, $(-1)^F \equiv +1$. Write $\alpha = (R, q)$, where R is an $SU(n)_k$ representation, and $q \in [0, n(n+k))$ labels a $U(1)_{n(n+k)}$ representation. The abelian part of the condition $\psi^{|\alpha|} \times \alpha = \alpha$ reads

$$q + |\alpha|(n+k) = q \mod n(n+k),$$
 (6.4.20)

which can be written as $|\alpha| = 0 \mod n$, i.e., $|\alpha| = n$, as claimed. This proves that all orbits are long, which indeed implies the absence of Majorana lines.

Let us now use this information to compute the different $U(n)_{k,n+k}$ partition functions. The untwisted trace is the number of conformal blocks (in any of the 2³ spin structures). Counting this is a straightforward exercise in combinatorics: we have a factor of n(n + k)due to $U(1)_{n(n+k)}$, times a factor of $\binom{n+k-1}{k}$ due to $SU(n)_k$ (cf. (6.6.2)), and a factor of $1/n^2$ due to the quotient \mathbb{Z}_n (one factor of n is due to the projecting out of lines, and the other one because the neutral lines are organized into length-n orbits). All in all, the number of states – the untwisted trace – is

$$\operatorname{tr}_{\mathrm{U}(n)_{k,n+k}}(1) = \frac{n(n+k)}{n^2} \binom{n+k-1}{k} \equiv \binom{n+k}{k}.$$
(6.4.21)

This standard argument was already used in [46]. An important aspect of this computation, much overlooked in the literature, is that this equals $tr(-1)^F$ only because all the states have trivial fermion parity, which is nontrivially true in this theory. This shall not be the case in the orthogonal group O(n), where $tr(-1)^F$ does not just count the total number of states, but rather the bosons minus the fermions, both sets being typically nonempty.

Recall that the conjectured infrared TQFT corresponding to the *n*-domain wall of SU(N) is $W_n = U(n)_{k,n+k}$, with k = N - n. Using this value of the level in (6.4.21) indeed reproduces the Witten index computed in the ultraviolet, cf. (6.2.20).

We now proceed to computing the trace twisted by the charge conjugation symmetry c of $U(n)_{k,n+k}$. Consider first the case of odd k, where the theory is naturally bosonic. In this case, computing the trace amounts to counting the real representations of $U(n)_{k,n+k}$. A representation of $U(n)_{k,n+k}$ can be labeled by the pair (R,q), where R is an $SU(n)_k$ representation, and $q \in [0, n(n+k))$, subject to $|R| = q \mod n$, where |R| denotes the number of boxes in the Young diagram of R. Representations (R,q) and $(\sigma^{\ell} \cdot R, q + \ell(n+k))$, with $\sigma^{\ell} \cdot R$ the SU(n) representation with Dynkin labels $(\sigma^{\ell} \cdot \lambda)_i = \lambda_{i-\ell \mod n}$, are identified by \mathbb{Z}_n spectral flow.

The abelian charge q is correlated with the SU(n) representation. Indeed, if n is even and R is real modulo σ^{ℓ} , there is a single charge $q \in [0, n(n+k))$ that makes (R, q) real; if n is odd, there are two such charges.¹¹³ Therefore, the number of real representations in $U(n)_{k,n+k}$ is the number of representations of $SU(n)_k$ that are real up to the action of σ ,

¹¹³Namely, we are looking for solutions to $2q = \ell(n+k) \mod n(n+k)$. These are $q = \frac{\ell}{2}(n+k)$ and $q = \frac{\ell+n}{2}(n+k)$ (except for $(n,\ell) =$ (even, odd), where there is no solution; see also (6.4.23)). For *n* even, only one of these two solutions is valid, depending on the parity of |R| (recall that we require $q = |R| \mod n$).

divided by n (the length of the orbits), and multiplied by 2 if n is odd. Let us now count the $SU(n)_k$ representations.

An $\mathrm{SU}(n)_k$ representation $[\lambda_0, \lambda_1, \ldots, \lambda_{n-1}]$ is real up to the action of σ if $\lambda_i = \lambda_{\ell-i \mod n}$ for some ℓ . The number of such representations is the number of integer solutions to

$$2\lambda_{0} + 2\lambda_{1} + \dots + 2\lambda_{\lfloor (\ell-1)/2 \rfloor} + \left\{ \begin{array}{cc} 0 & \ell \text{ odd} \\ \lambda_{\ell/2} & \ell \text{ even} \end{array} \right\} + \\ + 2\lambda_{\ell+1} + \dots + 2\lambda_{\lfloor (n+\ell-1)/2 \rfloor} + \left\{ \begin{array}{cc} 0 & n+\ell \text{ odd} \\ \lambda_{(n+\ell)/2} & n+\ell \text{ even} \end{array} \right\} = k \,.$$

$$(6.4.22)$$

The number of solutions to this equation is

$$N_{\ell} = \begin{cases} \binom{(n+k)/2 - 1}{(k-1)/2} & n \text{ odd, } k \text{ odd,} \\ 0 & n \text{ even, } \ell \text{ odd, } k \text{ odd,} \\ 2\binom{(n+k-1)/2}{(k-1)/2} & n \text{ even, } \ell \text{ even, } k \text{ odd,} \\ \binom{(n+k)/2}{k/2} + \binom{(n+k)/2 - 1}{k/2 - 1} & n \text{ even, } \ell \text{ even, } k \text{ even,} \\ \binom{(n+k)/2 - 1}{k/2} & n \text{ even, } \ell \text{ odd, } k \text{ even,} \\ \binom{(n+k)/2 - 1}{k/2} & n \text{ even, } \ell \text{ odd, } k \text{ even,} \end{cases}$$

where, for future reference, we have also included the case of k even.

We now sum over all $\ell = 0, 1, \ldots, n-1$. For n odd, this just multiplies $\binom{(n+k)/2-1}{(k-1)/2}$ by n. If n is even, it multiplies $2\binom{(n+k-1)/2}{(k-1)/2}$ by n/2, because half the cases yield no solutions. Next, we divide by n (due to the quotient), and multiply by 2 if n is odd. This yields the number of real representations of $U(n)_{k,n+k}$ with k odd as¹¹⁴

$$\operatorname{tr}_{\mathrm{U}(n)_{k,n+k}}(\mathbf{c}) = \begin{cases} 2\binom{(n+k)/2 - 1}{(k-1)/2} & n \text{ odd, } k \text{ odd,} \\ \\ \binom{(n+k-1)/2}{(k-1)/2} & n \text{ even, } k \text{ odd.} \end{cases}$$
(6.4.24)

By plugging k = N - n in (6.4.24), the partition function reproduces the Witten index twisted by charge conjugation computed in the ultraviolet, cf. (6.2.24).

The case of k even is slightly more complicated because the theory is naturally spin. For n odd we can obtain the twisted trace from the k odd case by using level-rank duality

¹¹⁴Note that the expression for n odd is invariant under $n \leftrightarrow k$, as required by level-rank duality.

 $U(n)_{k,n+k} \leftrightarrow U(k)_{-n,-(n+k)}$. But for n, k both even, the theory is spin, and cannot be written as a bosonic theory times a trivial spin theory – at least not using the standard level-rank duality. Thus, we have to explicitly compute the trace of **c** in the R-R sector. This is non-trivial because, among other things, **c** may act as -1 on some states (see the discussion around (6.3.27)), and thus it is not enough to just count real representations.

A shortcut to compute the trace of c over the odd spin structure, for k, n both even, is to sum over all spin structures:

$$\frac{1}{2} \left(\sum_{\sigma} \operatorname{tr}_{\sigma} \mathbf{c} + \operatorname{tr}_{\sigma} (-1)^{F} \mathbf{c} \right) = \operatorname{tr}_{B} \mathbf{c} , \qquad (6.4.25)$$

where tr_B denotes the trace over the bosonic parent. From this expression, and noting that $(-1)^F$ is trivial in $\operatorname{U}(n)_{k,n+k}$ theories (due to the lack of Majorana lines), we can solve for the trace we are after:

$$\operatorname{tr}_{\operatorname{R-R}} \mathsf{c} = \operatorname{tr}_B \mathsf{c} - 3 \operatorname{tr}_{\operatorname{NS-NS}} \mathsf{c} \,. \tag{6.4.26}$$

Let us begin with the first term. As this is a trace over a bosonic Hilbert space, we are just to count real representations of $SU(n)_k \times U(1)_{n(n+k)}$. The first factor corresponds to $\ell = 0$ in (6.4.23), while the second factor has two real representations (namely, q = 0 and q = n(n+k)/2). The end result is

$$\operatorname{tr}_{B} \mathbf{c} = 2 \left[\binom{(n+k)/2}{k/2} + \binom{(n+k)/2 - 1}{k/2 - 1} \right].$$
(6.4.27)

Let us now compute the second term in (6.4.26). This is a trace over a fermionic Hilbert space, but over the NS-NS sector, and so we only have to count fixed-points, as they all contribute with $\mathbf{c} = +1$. In other words, the trace is just the number of real representations of $U(n)_{k,n+k}$, that is, the number of solutions to (6.4.22), summed over $\ell = 0, 1, \ldots, n-1$, and divided by n due to the quotient. Using (6.4.23), we get

$$\operatorname{tr}_{\text{NS-NS}} \mathbf{c} \equiv \frac{1}{n} \left(\frac{n}{2} \left[\binom{(n+k)/2}{k/2} + \binom{(n+k)/2 - 1}{k/2 - 1} \right] + \frac{n}{2} \binom{(n+k)/2 - 1}{k/2} \right) \\ = \frac{1}{2} \binom{(n+k)/2}{k/2} + \frac{1}{2} \binom{(n+k)/2 - 1}{k/2 - 1} + \frac{1}{2} \binom{(n+k)/2 - 1}{k/2} \right).$$
(6.4.28)

Plugging these two traces into (6.4.26), the twisted index, for n, k even, becomes

$$\operatorname{tr}_{\mathrm{U}(n)_{k,n+k}}(\mathsf{c}) = \binom{(n+k)/2 - 1}{n/2} - \binom{(n+k)/2 - 1}{k/2}.$$
(6.4.29)

Note that this is invariant under $n \leftrightarrow k$, as required by level-rank duality.¹¹⁵ This expression for the twisted partition function of $U(n)_{k,n+k}$ with k = N - n matches the twisted Witten index computed in the ultraviolet, cf. (6.2.24).

¹¹⁵That is, invariant up to a sign. This is due to the fact that c = -1 in the R-R sector of U(1)₁ (cf. (6.3.27)), and the level-rank pair has a difference in their framing anomaly equal to $nk + 1 \equiv 1 \mod 2$, cf. [28].

Finally, we briefly sketch the computation of the trace twisted by the one-form symmetry $\mathbf{g}^t \in \mathbb{Z}_{n+k}$ of $\mathrm{U}(n)_{k,n+k}$, where \mathbf{g} denotes a primitive root of unity, and $t \in [0, n+k)$. The states of $\mathrm{U}(n)_{k,n+k}$ are orbits of the form

$$\{(\sigma^{\ell} \cdot R, q + \ell(n+k))\}, \qquad (6.4.30)$$

where ℓ ranges from 0 to n-1. All the orbits are of length-n. The theory has a \mathbb{Z}_{n+k} one-form symmetry that acts as $q \mapsto q + tn$, where $t \in [0, n+k)$. A state is invariant if and only if this transformation cyclically permutes the elements of the orbit, i.e., if a representative (R, q) is mapped into itself up to spectral flow,

$$(R, q+tn) \equiv (\sigma^{\ell} \cdot R, q+\ell(n+k)).$$
(6.4.31)

It is clear that if tn is not of the form $\ell(n+k)$ for some $\ell \in \mathbb{Z}$, then no state is invariant, and the twisted trace vanishes. So let us assume that such an ℓ exists; it is clear that it is unique, so counting invariant orbits reduces to counting appropriate $SU(n)_k$ representations. More specifically, the number of invariant states is

$$\operatorname{tr}_{\mathrm{U}(n)_{k,n+k}}(\mathsf{g}^{t}) = \frac{n(n+k)}{n^{2}} \hat{N}_{\ell}, \qquad (6.4.32)$$

where n(n+k) denotes the number of states in $U(1)_{n(n+k)}$, and the factor of n^2 is due to the \mathbb{Z}_n quotient. \hat{N}_{ℓ} denotes the number of $SU(n)_k$ representations that satisfy $R = \sigma^{\ell} \cdot R$, with $\ell := tn/(n+k) \in \mathbb{Z}$.

Counting such $SU(n)_k$ representations is easy, because this is a simply-connected group, so the states are labelled by the Dynkin labels, $\lambda_0, \lambda_1, \ldots, \lambda_{n-1}$, which can be thought of a collection of independent bosons (cf. (6.4.11)). The most efficient way to count the representations that are invariant under σ^ℓ is to recall that the associated diagonal phase is just the charge under the center (6.2.18), which is a multiplicative phase, so the partition function factorizes:

$$\prod_{j=0}^{n-1} (1 - e^{2\pi i j\ell/n} q)^{-1} \equiv \sum_{k \ge 0} \hat{N}_{\ell} q^k , \qquad (6.4.33)$$

and therefore

$$\hat{N}_{\ell} = \binom{n+k-1}{k}_{e^{2\pi i\ell/n}}.$$
(6.4.34)

Recall that the q-binomial coefficient at a root of unity can be expressed as a regular binomial coefficient, cf. (6.2.29) and (6.2.27).

Putting everything together, the one-form symmetry twisted trace reads

$$\operatorname{tr}_{\mathrm{U}(n)_{k,n+k}}(\mathbf{g}^{t}) = \binom{n+k}{n}_{\mathbf{g}^{t}}$$
$$\equiv \begin{cases} \binom{\gcd(n+k,t)}{n \gcd(n+k,t)/(n+k)} & tn \equiv 0 \mod n+k, \\ 0 & \text{otherwise.} \end{cases}$$
(6.4.35)

It is easily checked that, if we plug k = N - n in (6.4.35), the twisted trace for $U(n)_{k,n+k}$ exactly reproduces the twisted Witten index of the *n*-domain wall of SU(N) computed in the ultraviolet, cf. (6.2.27).

6.4.3 G = O(n)

The *n*-domain wall theory for $4d \ \mathcal{N} = 1$ SYM with G = Spin(N) is proposed to be $O(n)_{N-2-n,N-n+1}^1$ Chern-Simons theory. Let us proceed to study the partition functions of $O(n)_{k,L}^1$.

The $O(n)_{k,L}^1$ Chern-Simons theory is defined as [113]

$$\mathcal{O}(n)_{k,L}^{1} := \frac{\mathcal{O}(n)_{k,0}^{1} \times (\mathbb{Z}_{2})_{L}}{\mathbb{Z}_{2}}, \qquad (6.4.36)$$

where $(\mathbb{Z}_2)_L \leftrightarrow \operatorname{Spin}(L)_{-1}$ denotes a \mathbb{Z}_2 gauge theory with twist L, and the quotient denotes the gauging of a diagonal \mathbb{Z}_2 one-form symmetry. The value of the level we shall be interested in is L = k + 3. On the other hand, the first factor is given by the following:

- If n is even, the theory O(n)¹_{k,0} is defined as the CM-orbifold of SO(n)_k. Here C denotes the charge-conjugation Z₂ zero-form symmetry that acts by permuting the last two Dynkin labels in SO(n), and M is the magnetic Z₂ zero-form symmetry that is dual to the gauged one-form Z₂ symmetry in the denominator of SO(n)_k ≡ Spin(n)_k/Z₂. As such, it permutes the lines that split in the quotient, i.e., the lines of Spin(n)_k that are fixed by fusion with the extending simple current.
- If n is odd, the group O(n) is a direct product of SO(n) and Z₂. The Chern-Simons theory O(n)¹_{k,0} itself does not necessarily factorize, because of the convention of which Z₂ subgroup the reflection represents. The choice in [113] was

$$O(n)_{k,0}^{1} := \begin{cases} \frac{\operatorname{Spin}(n)_{k} \times (\mathbb{Z}_{2})_{(k-2)(n-1)}}{\mathbb{Z}_{2}} & n \text{ odd, } k \text{ even,} \\ \\ \operatorname{SO}(n)_{k} \times (\mathbb{Z}_{2})_{(k-2)(n-1)} & n \text{ odd, } k \text{ odd.} \end{cases}$$
(6.4.37)

Let us compute the different traces in this theory. As above, the details depend sensitively on the parity of n and k, so we consider each case separately. **Even/Even.** We begin with the theory $\text{Spin}(2n)_{2k}$. Its integrable representations satisfy

$$\lambda_0 + \lambda_1 + 2(\lambda_2 + \dots + \lambda_{n-2}) + \lambda_{n-1} + \lambda_n = 2k, \qquad (6.4.38)$$

which has $\operatorname{tr}_{\operatorname{Spin}(2n)_{2k}}(1)$ solutions (cf. (6.6.12)).

We now construct $SO(2n)_{2k}$, i.e., we gauge a \mathbb{Z}_2 one-form symmetry, which acts as $\lambda_0 \leftrightarrow \lambda_1$ and $\lambda_{n-1} \leftrightarrow \lambda_n$. This is a bosonic quotient. The neutral representations satisfy $\lambda_{n-1} + \lambda_n = \text{even}$, which has

$$\mathbb{N} := \binom{n+k}{k} + 2\binom{n+k-1}{k-1} + \binom{n+k-2}{k-2}, \qquad (6.4.39)$$

solutions. These are divided into length-2 orbits, and fixed points. The former satisfy $\lambda_0 \neq \lambda_1 \vee \lambda_{n-1} \neq \lambda_n$, and the latter $\lambda_0 = \lambda_1 \wedge \lambda_{n-1} = \lambda_n$. The number of fixed points is

$$\mathbb{F} := \binom{n+k-2}{k},\tag{6.4.40}$$

and the number of length-2 orbits is $\frac{1}{2}(\mathbb{N} - \mathbb{F})$. Finally, the number of representations of $SO(2n)_{2k}$ is

$$\operatorname{tr}_{\mathrm{SO}(2n)_{2k}}(\mathbf{1}) = 2\mathbb{F} + \frac{1}{2}(\mathbb{N} - \mathbb{F}) \equiv 2\binom{n+k-2}{k} + \binom{n+k-2}{k-1} + (n \leftrightarrow k), \quad (6.4.41)$$

which is invariant under $n \leftrightarrow k$, as expected by level-rank duality. This also agrees with expression (6.6.15).

We now orbifold by CM, which acts by swapping the lines in 2F pairwise, and as $\lambda_{n-1} \leftrightarrow \lambda_n$. The representations that are fixed under CM are the subset of the length-2 orbits that satisfy either $\lambda_0 \neq \lambda_1 \wedge \lambda_{n-1} = \lambda_n$ or $\lambda_0 = \lambda_1 \wedge \lambda_{n-1} \neq \lambda_n$. In other words, the lines that satisfy either of

$$\lambda_0 + \lambda_1 + 2(\lambda_2 + \dots + \lambda_{n-2} + \lambda_{n-1}) = 2k, \qquad \lambda_0 \neq \lambda_1, 2(\lambda_1 + \lambda_2 + \dots + \lambda_{n-2}) + \lambda_{n-1} + \lambda_n = 2k, \qquad \lambda_{n-1} \neq \lambda_n.$$
(6.4.42)

By symmetry, both conditions have the same number of solutions. In total,

$$\mathbb{A} := 2 \left[\binom{n+k-1}{k} + \binom{n+k-2}{k-1} - \binom{n+k-2}{k} \right]$$
(6.4.43)

solutions. Note that these are length-2 orbits of $\text{Spin}(2n)_{2k}$, so the number of lines is $\mathbb{A}/2$.

The representations that are interchanged under CM are all of \mathbb{F} , plus the subset of the length-2 orbits that satisfy $\lambda_{n-1} + \lambda_n =$ even and $\lambda_{n-1} \neq \lambda_n$, minus the solutions to the second line in (6.4.42). The latter are

$$\mathbb{B} := \binom{n+k}{k} + 2\binom{n+k-1}{k-1} + \binom{n+k-2}{k-2} \\
-\binom{n+k-1}{k} - \binom{n+k-2}{k-1} - \frac{1}{2}\mathbb{A} \\
= 4\binom{n+k-2}{k-2}.$$
(6.4.44)

Note that \mathbb{F} are fixed points of \mathbb{Z}_2 , while \mathbb{B} are length-2 orbits, so the number of lines is $2\mathbb{F} + \mathbb{B}/2$. Adding the lines in \mathbb{A} we get $\frac{1}{2}(\mathbb{A} + \mathbb{B}) + 2\mathbb{F} \equiv \operatorname{tr}_{\operatorname{SO}(2n)_{2k}}(1)$, as one would expect.

Putting all these results together, we see that the number of twisted and untwisted lines in the orbifold is [320]

$$N_{\text{twisted}} = \mathbb{A} ,$$

$$N_{\text{untwisted}} = \mathbb{A} + \frac{1}{4}\mathbb{B} + \mathbb{F} ,$$

$$(6.4.45)$$

and so the theory has

$$\operatorname{tr}_{\mathcal{O}(2n)^{1}_{2k,0}}(\mathbf{1}) = N_{\text{twisted}} + N_{\text{untwisted}}$$
$$= -\frac{9}{8} \binom{n+k-2}{k} + 4 \binom{n+k-2}{k-1} + \frac{17}{8} \binom{n+k-2}{k-2} + (n \leftrightarrow k)$$
(6.4.46)

lines. This expression agrees with [113].

We now move on to $O(2n)_{2k,2k+3}^1$. This is obtained by taking the theory we just constructed, $O(2n)_{2k,0}^1$, tensoring with $Spin(2k+3)_{-1}$, and gauging a diagonal \mathbb{Z}_2 one-form symmetry:

$$O(2n)_{2k,2k+3}^{1} = \frac{O(2n)_{2k,0}^{1} \times \operatorname{Spin}(2k+3)_{-1}}{\mathbb{Z}_{2}}, \qquad (6.4.47)$$

where the quotient is fermionic.

Take the states of $O(2n)_{2k,0}^1$ as above, i.e., N_{twisted} and $N_{\text{untwisted}}$, and tensor by $\text{Spin}(2k + 3)_{-1} = \{\mathbf{1}, \sigma, \chi\}$. The NS and R lines are as follows:

We now quotient by the \mathbb{Z}_2 one-form symmetry. This symmetry maps $\mathbf{1} \leftrightarrow \chi$, and it fixes σ ; and, also, it permutes lines in \mathbb{A} pairwise, $a \leftrightarrow a'$, and it fixes those in $\frac{1}{4}\mathbb{B} + \mathbb{F}$. Therefore, in the NS sector it acts as

$$\begin{array}{c} \mathbb{A} \otimes \mathbf{1} \leftrightarrow \mathbb{A}' \otimes \chi \\ (\frac{1}{4}\mathbb{B} + \mathbb{F}) \otimes \mathbf{1} \leftrightarrow (\frac{1}{4}\mathbb{B} + \mathbb{F}) \otimes \chi \\ \mathbb{A} \otimes \sigma \leftrightarrow \mathbb{A}' \otimes \sigma \end{array}$$
(6.4.49)

which are all length-two orbits (recall that there are never fixed-points in the NS sector). Thus, the dimension of the Hilbert space is

$$\dim(\mathcal{O}(2n)^{1}_{2k,2k+3}) = \mathbb{A} + \left(\frac{1}{4}\mathbb{B} + \mathbb{F}\right) + \frac{1}{2}\mathbb{A}$$
$$\equiv \frac{1}{2}N_{\text{twisted}} + N_{\text{untwisted}}.$$
(6.4.50)

This corresponds to the trace of **1** over the Hilbert space on any of the spatial spin structures.

Consider now the R sector. The one-form symmetry acts as

$$\begin{array}{l} \mathbb{A} \otimes \mathbf{1} \leftrightarrow \mathbb{A}' \otimes \chi \\ (\frac{1}{4}\mathbb{B} + \mathbb{F}) \otimes \sigma \leftrightarrow (\frac{1}{4}\mathbb{B} + \mathbb{F}) \otimes \sigma \\ \mathbb{A} \otimes \sigma \leftrightarrow \mathbb{A}' \otimes \sigma , \end{array}$$

$$(6.4.51)$$

and so all of \mathbb{A} are in length-two orbits, while all of $\frac{1}{4}\mathbb{B} + \mathbb{F}$ are fixed-points. Thus, the number of fermions and bosons is

$$N_{\text{boson}} = \mathbb{A} + \frac{1}{2}\mathbb{A} \equiv \frac{3}{2}N_{\text{twisted}},$$

$$N_{\text{fermion}} = \frac{1}{4}\mathbb{B} + \mathbb{F} \equiv N_{\text{untwisted}} - N_{\text{twisted}}.$$
(6.4.52)

Note that $N_{\text{boson}} + N_{\text{fermion}}$ agrees with the dimension of the Hilbert space as computed in the NS sector (cf. (6.4.50)). On the other hand, the trace in the odd spin structure, weighted by fermion parity, is $N_{\text{boson}} - N_{\text{fermion}}$:

$$\operatorname{tr}_{\mathcal{O}(2n)^{1}_{2k,2k+3}}(-1)^{F} \equiv \frac{5}{2}N_{\text{twisted}} - N_{\text{untwisted}}.$$
 (6.4.53)

As a consistency check, recall that one can also express the fermionic trace as $\operatorname{tr}_{\mathcal{H}_{\mathrm{R-R}}}(-1)^F = 2 \operatorname{dim}(\mathcal{H}_B) - 7 \operatorname{dim}(\mathcal{H}_F)$ (cf. (6.3.20)). The dimension of the bosonic Hilbert space is

$$\dim(\mathcal{O}(2n)^{1}_{2k,0} \times \operatorname{Spin}(2k+3)_{-1}) \equiv 3(N_{\text{untwisted}} + N_{\text{twisted}}), \qquad (6.4.54)$$

while the dimension of the fermionic Hilbert space is half the number of lines, i.e., $\frac{1}{2}(2N_{\text{untwisted}} + N_{\text{twisted}})$. Thus,

$$tr_{O(2n)_{2k,2k+3}^{1}}(-1)^{F} = 6N_{twisted} + 6N_{untwisted} - \frac{7}{2}N_{twisted} - 7N_{untwisted}, \qquad (6.4.55)$$

which indeed matches the expression above.

Recall that the conjectured infrared theory corresponding to the *n*-domain wall of Spin(N) was $W_n = O(n)_{k,k+3}^1$, with k = N - 2 - n. Using this value of the level in (6.4.53) indeed reproduces the Witten indices computed in the ultraviolet, cf. (6.2.46).

Odd/Odd. We consider

$$O(2n+1)_{2k+1,2k+4}^{1} = SO(2n+1)_{2k+1} \times (\mathbb{Z}_{2})_{2(n+k)}.$$
(6.4.56)

As the theory is a tensor product, the traces factorize:

$$\operatorname{tr}_{\mathcal{O}(2n+1)_{2k+1,2k+4}^{1}}(\mathcal{O}_{1}\otimes\mathcal{O}_{2}) \equiv \operatorname{tr}_{\mathcal{SO}(2n+1)_{2k+1}}(\mathcal{O}_{1})\cdot\operatorname{tr}_{(\mathbb{Z}_{2})_{2(n+k)}}(\mathcal{O}_{2}).$$
(6.4.57)

For example, the \mathbb{Z}_2 gauge theory has four states, all bosonic, $\operatorname{tr}_{(\mathbb{Z}_2)_{2(n+k)}}(-1)^F \equiv 4$, which means that the untwisted index is

$$\operatorname{tr}_{\mathcal{O}(2n+1)_{2k+1,2k+4}^{1}}(-1)^{F} = 4\left[\binom{n+k}{k} - 2\binom{n+k-1}{k}\right], \qquad (6.4.58)$$

where we have used the trace of $SO(2n+1)_{2k+1}$ as given in (6.6.10).

Similarly, the index twisted by the zero-form symmetry \mathbf{c} has $\operatorname{tr}_{(\mathbb{Z}_2)_{2(n+k)}}(\mathbf{c}) \equiv 2$, where \mathbf{c} acts by permuting the two spinors (this is the only zero-form symmetry of this \mathbb{Z}_2 gauge theory, cf. [4, 321]; it fixes both the identity and the vector). On the other hand, the only zero-form symmetry of $\operatorname{SO}(2n+1)_{2k+1}$ is fermion parity,¹¹⁶ and there is in fact a natural identification $\mathbf{c} = (-1)^F$ (cf. [113]). Thus, the \mathbf{c} -twisted trace weighted by fermion parity actually computes the untwisted trace, with antiperiodic (NS) boundary conditions on the time circle:

$$\operatorname{tr}_{\mathrm{SO}(2n+1)_{2k+1}}((-1)^F \mathbf{c}) \equiv \operatorname{tr}_{\mathrm{SO}(2n+1)_{2k+1}}(\mathbf{1}) \equiv \binom{n+k}{k},$$
 (6.4.59)

where we have used (6.6.9). All in all, the twisted trace of $O(2n+1)^{1}_{2k+1,2k+4}$ is

$$\operatorname{tr}_{\mathcal{O}(2n+1)_{2k+1,2k+4}^{1}}((-1)^{F}\mathsf{c}) = 2\binom{n+k}{k}.$$
(6.4.60)

The index twisted by the one-form symmetry is also straightforward. This symmetry is \mathbb{Z}_2^2 for $O(4n+1)_{4k+1}$ and $O(4n+3)_{4k+3}$, and \mathbb{Z}_4 for $O(4n+1)_{4k+3}$ and $O(4n+3)_{4k+1}$. These correspond to fusion with the abelian anyons of $\operatorname{Spin}(L)_{-1}$, with $L = 0 \mod 4$ and $L = 2 \mod 4$ respectively, which indeed have a $\mathbb{Z}_2^2/\mathbb{Z}_4$ fusion algebra. As abelian fusion has no fixed-points, all the twisted traces vanish:

$$\begin{aligned} \operatorname{tr}_{\mathcal{O}(2n+1)_{2k+1,2k+4}^{1}}((-1)^{F}\mathsf{g}_{1}\mathsf{g}_{2}) &\equiv 0, \\ \operatorname{tr}_{\mathcal{O}(2n+1)_{2k+1,2k+4}^{1}}((-1)^{F}\mathsf{g}) &\equiv 0, \end{aligned}$$
(6.4.61)

where $(\mathbf{g}_1, \mathbf{g}_2) \in \mathbb{Z}_2^2$ and $\mathbf{g} \in \mathbb{Z}_4$, respectively.

Recall that the conjectured infrared theory corresponding to the *n*-domain wall of Spin(N) was $W_n = O(n)_{k,k+3}^1$, with k = N - 2 - n. Using this value of the level in (6.4.58), (6.4.60), (6.4.61) indeed reproduces the Witten indices computed in the ultraviolet, cf. (6.2.46), (6.2.49), (6.2.53), (6.2.56).

Odd/Even & Even/Odd. We only need to consider one; the other follows by the level-rank duality. Take

$$O(2n+1)_{2k,0}^{1} = \frac{\operatorname{Spin}(2n+1)_{2k} \times (\mathbb{Z}_{2})_{4n(k-1)}}{\mathbb{Z}_{2}}, \qquad (6.4.62)$$

where the gauged one-form symmetry is generated by $a \otimes e$, where a = [0, 2k, 0, ..., 0] and e is the electric line of the toric code. This is a bosonic quotient.

¹¹⁶The Dynkin diagram of SO(N) for N odd has no reflection symmetries, i.e., its outer automorphism group is trivial. Thus, the zero-form symmetries of SO(N), if any, must be due to the global structure of the group, as its algebra has no symmetries. Indeed, the zero-form symmetry comes from $\pi_1(SO(N)) = \mathbb{Z}_2$, but this is just the magnetic dual to the gauged \mathbb{Z}_2 one-form symmetry, which means that the magnetic symmetry is formally just $(-1)^F$. If we were to gauge this symmetry, we would recover Spin(N).

The one-form symmetry acts as $\lambda_0 \leftrightarrow \lambda_1$ and $e: m \leftrightarrow em$. The neutral lines are of the form

$$\lambda \otimes \mathbf{1}, \quad \lambda \otimes \mathbf{e}, \qquad \lambda_n = \text{even}$$

$$\lambda \otimes \mathbf{m}, \quad \lambda \otimes \text{em}, \qquad \lambda_n = \text{odd}.$$
 (6.4.63)

Note that there are no fixed points, and all orbits are of length 2:

$$\{\lambda \otimes \mathbf{1}, (a \times \lambda) \otimes \mathbf{e}\}, \{\lambda \otimes \mathbf{m}, (a \times \lambda) \otimes \mathbf{em}\}.$$
 (6.4.64)

Therefore, a set of representatives can be taken as $\lambda_{\text{tensor}} \otimes \mathbf{1}$ and $\lambda_{\text{spinor}} \otimes \mathbf{m}$. In what follows we drop the second label, as it is correlated with λ in a unique way. The number of tensors and spinors is (cf. (6.6.5))

$$N_{\text{tensor}} = \binom{n+k-1}{k-1} + \binom{n+k}{k}, \qquad N_{\text{spinor}} = 2\binom{n+k-1}{k-1}.$$
(6.4.65)

We now tensor the theory by a factor of $\text{Spin}(2k+3)_{-1} = \{\mathbf{1}, \sigma, \chi\}$, and gauge the fermionic one-form symmetry generated by $f = a \otimes \chi$. The Ramond sector requires $h_{\alpha \times f} = h_{\alpha} \mod 1$, which means that the lines are

$$(\lambda_{\text{tensor}}, \sigma), \qquad (\lambda_{\text{spinor}}, \mathbf{1} \text{ or } \chi).$$
 (6.4.66)

Note that only the former can be a fixed-point under the fermionic quotient, inasmuch as $\chi \times \sigma = \sigma$ while $\chi: \mathbf{1} \leftrightarrow \chi$. In particular, the fixed-points are

$$\lambda_{\text{tensor}}, \qquad \lambda_0 = \lambda_1 \,, \tag{6.4.67}$$

while the rest of lines are all in length-2 orbits. The fixed-points satisfy $\lambda_1 + \lambda_2 + \cdots + \lambda_{n-1} + \lambda_n/2 = k$, which has

$$\mathbb{F} := \binom{n+k-1}{k}, \qquad (6.4.68)$$

solutions. Thus, finally

$$\operatorname{tr}_{\mathcal{O}(2n+1)_{2k,2k+3}^{1}}(-1)^{F} = N_{\operatorname{spinor}} + \frac{1}{2}(N_{\operatorname{tensor}} - \mathbb{F}) - \mathbb{F}$$
$$= \frac{5}{2}\binom{n+k-1}{k-1} + \frac{1}{2}\binom{n+k}{k} - \frac{3}{2}\binom{n+k-1}{k}.$$
(6.4.69)

The trace over $O(2n)_{2k+1,2k+4}^1$ can be obtained by using the orthogonal level-rank duality $O(2n)_{2k+1,2k+4}^1 \leftrightarrow O(2k+1)_{-2n,-(2n+3)}^1$.

One can similarly compute the index twisted by the \mathbb{Z}_2 one-form symmetry, which acts via fusion with the electric line \mathbf{e} . The charged states are those that include the magnetic line \mathbf{m} , to wit, the spinors. In other words, the one-form symmetry correlates (gauge) spin and (spacetime) statistics, so that the states with $(-1)^F \mathbf{e} = +1$ are the tensor bosons and

spinor fermions, and states with $(-1)^F \mathbf{e} = -1$ are the spinor bosons and the tensor fermions. With this,

$$\operatorname{tr}_{\mathcal{O}(2n+1)_{2k,2k+3}^{1}}((-1)^{F}\mathbf{e}) = -N_{\operatorname{spinor}} + \frac{1}{2}(N_{\operatorname{tensor}} - \mathbb{F}) - \mathbb{F}$$

= $-\binom{n+k}{k}$. (6.4.70)

As above, the trace for $O(2n)_{2k+1,2k+4}^1$ is obtained by level-rank duality.

Recall that the conjectured infrared theory corresponding to the *n*-domain wall of Spin(N) was $W_n = O(n)_{k,k+3}^1$, with k = N - 2 - n. Using this value of the level in (6.4.69), (6.4.70) indeed reproduces the Witten indices computed in the ultraviolet, cf. (6.2.39), (6.2.42).

6.4.4 $G = G_2$

The 2-domain wall theory for $4d \mathcal{N} = 1$ SYM with $G = G_2$ is SO(3)₃ × S¹, where S¹ denotes then nonlinear sigma model with S¹ target space. We already proved in section 6.3.1 that the theory SO(3)₃ has vanishing Witten index, and since there is a unique vacuum of the S¹ sigma model on the torus, the infrared index vanishes. This matches the Witten index computed in the ultraviolet, which is given by the coefficient of q^2 in (6.1.20). Indeed, expanding this polynomial one finds that the index vanishes.

The domain wall with n = 1 (and n = 3, which is the anti-wall of n = 1) is addressed below.

6.4.5 Minimal Wall for Arbitrary Gauge Group

The n = 1 domain wall theory for $4d \mathcal{N} = 1$ SYM with arbitrary G is proposed to be G_{-1} Chern-Simons theory.

As G is simply-connected, the theory is naturally bosonic, and the trace $\operatorname{tr}_{G_{-1}}(-1)^F$ computes the dimension of the Hilbert space, that is, the number of integrable representations at level 1. In other words, the trace is the number of solutions to (6.4.4) with k = 1, namely

$$\sum_{i=0}^{r} \lambda_i a_i^{\vee} = 1 \,, \tag{6.4.71}$$

which, as in (6.2.59), requires $\lambda_i = 1$ for some *i* with $a_i^{\vee} = 1$, and $\lambda_j = 0$ for all $j \neq i$. Therefore, the trace is

$$\operatorname{tr}_{G_{-1}}(-1)^F = m_1 \tag{6.4.72}$$

where m_1 denotes the number of nodes in the Dynkin diagram of G that have comark equal to 1. This clearly reproduces the ultraviolet index (6.2.60), as required.

For simply-laced G, G_{-1} Chern-Simons theory is in fact an abelian TQFT, and all the lines generate one-form symmetries. The number of lines is the number of one-form symmetries, that is, the order of Γ , which indeed agrees with m_1 . Equivalently, it is known that simply-laced theories at level 1 admit a K-matrix representation, where one can take K as the Cartan matrix of \mathfrak{g} . The number of states is indeed det $(K) \equiv |\Gamma|$.¹¹⁷

One can define a zero-form twisted index for $(E_6)_{-1}$. The only node with comark 1 preserved by the charge conjugation symmetry of this theory is the extended node, and thus

$$\operatorname{tr}_{(E_6)_{-1}}(-1)^F = 1.$$
 (6.4.73)

Since $(E_6)_{-1}$ and $(E_7)_{-1}$ are abelian, twisting by a one-form symmetry has no fixed points and

$$\begin{aligned} & \operatorname{tr}_{(E_6)_{-1}}((-1)^F \mathbf{g}) = 0 & \mathbf{g} \in \Gamma = \mathbb{Z}_3 \\ & \operatorname{tr}_{(E_7)_{-1}}((-1)^F \mathbf{g}) = 0 & \mathbf{g} \in \Gamma = \mathbb{Z}_2 . \end{aligned}$$
 (6.4.74)

These reproduce the twisted Witten indices on the walls (6.2.63).

6.5 Concluding Remarks and Open Questions

In this paper we have proposed explicit 3d topological field theories on the domain walls of 4d $\mathcal{N} = 1$ SYM with gauge group G. We have found precise agreement between computations carried out in terms of the ultraviolet 4d degrees of freedom (gluons and gluinos) and the conjectured infrared topological 3d degrees of freedom. We have highlighted the importance in identifying the infrared of the domain wall theories of studying the Hilbert space of spin TQFTs, in particular the partition function in the R-R sector and identifying the fermionic states in the Hilbert space, as opposed to merely counting states. The nontrivial matching of the twisted Witten indices provides strong support for our proposal.

A heuristic argument can be made in favor of our proposal that the *n*-domain wall in $4d \ \mathcal{N} = 1$ SYM with gauge group G is the infrared of $3d \ \mathcal{N} = 1 \ G_{h/2-n}$ SYM (see equation (6.1.3)).¹¹⁸ Consider 4d SYM on $\mathbb{R}^3 \times S^1$ with the YM θ -angle linear in the S^1 coordinate and winding number n around the circle. This theory can be defined while preserving half of the supersymmetry.¹¹⁹ When the radius of the circle is large one can expect the theory to be gapped everywhere except at the location of the wall W_n . For small radius, the theory reduces to $3d \ \mathcal{N} = 1 \ G_{-n}$ SYM with an adjoint real multiplet (the scalar is compact, as it arises from reducing the gauge field along a circle). It was argued in [127] that with a suitable superpotential for the real multiplet, the multiplet gaps out and flows to

¹¹⁷Note that abelian systems typically have a very large number of zero-form symmetries [4], most of which are emergent in our picture, inasmuch as the ultraviolet theory only has C as its zero-form symmetry group. ¹¹⁸We would like to thank D. Gaiotto for an interesting discussion regarding this point.

¹¹⁹The Lagrangian of this theory can be written as $\mathcal{L} = \int d^2 \vartheta \ X W_{\alpha} W^{\alpha}$, where X is a background chiral multiplet and W_{α} the chiral gauge field strength. Re(X) determines the gauge coupling and Im(X) the θ -angle. The background Im(X) $\propto nx_3$ with $F_X \propto in$ preserves half of the supersymmetries. The background for F_X induces a mass term for the gaugino $\propto in\bar{\lambda}\gamma_{(5)}\gamma^3\lambda$, where λ is Majorana.

 $3d \mathcal{N} = 1 \ G_{h/2-n}$ SYM, where the shift is induced by integrating out the massive fermion in the real multiplet. Assuming that there is no phase transition as the size of the circle is reduced leads to the proposal (6.1.3). However, the lack of control over the superpotential upon reduction makes the argument suggestive but heuristic.

The n > 1 domain wall theories for the groups $G = F_4, E_6, E_7$ and E_8 remain to be discovered. Equivalently, the phase diagram of the corresponding $3d \mathcal{N} = 1 G_k$ SYM with k < h/2-1 remains elusive. We collect in section 6.7 the twisted partition functions computed in the ultraviolet for future reference. One strategy towards the identification of the infrared domain wall theory is to search for novel level-rank dualities in G_k Chern-Simons theories that go beyond the ones that follow from conformal embeddings. In general, level-rank dualities follow from embeddings into holomorphic theories (theories with only one state), and this approach could lead to suitable level-rank dualities and in turn to explicit proposals for the remaining $3d \mathcal{N} = 1 G_k$ SYM phase diagrams (and associated 4d domain walls).

In this paper we have made an intriguing connection between the Hilbert space of Chern-Simons theories on the torus and the Hilbert space of fermions in 0 + 1 dimensions labeled by the extended Dynkin diagram $\mathfrak{g}^{(1)}$ corresponding to a Lie group G. That is, the fermionic Hilbert space $\mathcal{H}_{\mathrm{F}}^n$ with energy n is isomorphic as super-vector spaces to the R-R Hilbert space of a suitable spin TQFT, which we denote by TQFT_n

$$\mathcal{H}_{\rm F}^n \simeq \mathcal{H}_{\rm R-R}^{{\rm TQFT}_n} \,. \tag{6.5.1}$$

Consequently, the partition functions with periodic and antiperiodic boundary conditions on the time circle also match. Specifically we have established the correspondence (the $A_{N-1}^{(1)}$ case was studied by Douglas in [322])¹²⁰

$$\begin{array}{lcl}
A_{N-1}^{(1)} &\longleftrightarrow & \mathrm{U}(n)_{N-n,N} \\
B_{N}^{(1)} &\longleftrightarrow & \mathrm{O}(n)_{2N-1-n,2N-n+2}^{1} \\
C_{N}^{(1)} &\longleftrightarrow & \mathrm{Sp}(n)_{N+1-n} \\
D_{N}^{(1)} &\longleftrightarrow & \mathrm{O}(n)_{2N-2-n,2N-n+1}^{1} \\
G_{2}^{(1)} &\longleftrightarrow & \mathrm{U}(2)_{3n,2-n} .
\end{array}$$
(6.5.2)

Another route to constructing the domain walls for $G = F_4, E_6, E_7$ and E_8 is to identify the TQFT whose R-R Hilbert space on the torus is that of the collection of free fermions based on the corresponding affine Dynkin diagram $\mathbf{g}^{(1)}$.

6.6 Chern-Simons with Unitary and Orthogonal groups

In this section we compute several traces on the torus Hilbert space of Chern-Simons theories over simply-connected Lie groups. These traces are useful when studying more complicated

¹²⁰In writing this we use the duality $(G_2)_1 \leftrightarrow U(2)_{3,1}$ and the notation $U(2)_{6,0} \equiv SO(3)_3 \times S^1$.

theories over non-simply-connected groups.

6.6.1 G = SU(n)

Consider the algebra $A_{n-1} = \mathfrak{su}_n$. The comarks are all $a_i^{\vee} = 1$. Plugging this into (6.4.11) we get the generating function as

$$Z(SU(n),q) = (1-q)^{-n}, \qquad (6.6.1)$$

and, by expanding, the untwisted trace

$$\operatorname{tr}_{\mathrm{SU}(n)_k}(\mathbf{1}) = \binom{n+k-1}{k}.$$
(6.6.2)

This is the number of integrable representations of $SU(n)_k$, that is, the dimension of the torus Hilbert space of this Chern-Simons theory. This result will be useful when we discuss the Chern-Simons theory over the unitary group U(n), see section 6.4.2.

6.6.2 G = Spin(2n+1)

Consider the algebra $B_n = \mathfrak{so}_{2n+1}$. The comarks are $a_i^{\vee} = 1$ for i = 0, 1, n, and $a_i^{\vee} = 2$ for $i = 2, \ldots, n-1$. Plugging this into (6.4.11) we get the generating function as

$$Z(\operatorname{Spin}(2n+1), q) = (1-q)^{-3}(1-q^2)^{-(n-2)}, \qquad (6.6.3)$$

and, by expanding, the untwisted trace

$$\operatorname{tr}_{\operatorname{Spin}(2n+1)_{2k}}(\mathbf{1}) = \binom{n+k}{k} + 3\binom{n+k-1}{k-1},$$

$$\operatorname{tr}_{\operatorname{Spin}(2n+1)_{2k+1}}(\mathbf{1}) = \binom{n+k-1}{k-1} + 3\binom{n+k}{k}.$$

(6.6.4)

This is the number of integrable representations of $\text{Spin}(2n+1)_k$, that is, the dimension of the torus Hilbert space of this Chern-Simons theory. For future reference, it is also useful to break up the states into the tensors and spinors. In other words, we shall be interested in knowing how many of the states of Spin(2n+1) are tensorial representations, and how many are spinorial representations. These are defined by $\lambda_n = \text{even and } \lambda_n = \text{odd}$, respectively, which yields the following:

$$N_{\text{tensor}}^{\text{Spin}(2n+1)_{2k}} = \binom{n+k}{k} + \binom{n+k-1}{k-1},$$

$$N_{\text{spinor}}^{\text{Spin}(2n+1)_{2k}} = 2\binom{n+k-1}{k-1},$$

$$N_{\text{tensor}}^{\text{Spin}(2n+1)_{2k+1}} = 2\binom{n+k}{k},$$

$$N_{\text{spinor}}^{\text{Spin}(2n+1)_{2k+1}} = \binom{n+k}{k} + \binom{n+k-1}{k-1},$$
(6.6.5)

so that

$$\operatorname{tr}_{\operatorname{Spin}(2n+1)_{k}}(\mathbf{1}) \equiv N_{\operatorname{tensor}}^{\operatorname{Spin}(2n+1)_{k}} + N_{\operatorname{spinor}}^{\operatorname{Spin}(2n+1)_{k}}.$$
(6.6.6)

For a more interesting example, let us now compute the partition function of $SO(2n+1)_k =$ $Spin(2n+1)_k/\mathbb{Z}_2$, which corresponds to the algebra \mathfrak{so}_{2n+1} extended by the simple current $\chi = [0, k, 0, \ldots, 0]$. This current has spin $h_{\chi} = k/2$, and so the extension is fermionic for odd k. The current acts on a given representation $[\lambda_0, \lambda_1, \ldots, \lambda_n]$ as $\lambda_0 \leftrightarrow \lambda_1$.

Consider first the case of even k, so that $SO(2n + 1)_k$ makes sense as a bosonic theory. The extension has two effects: first, it projects out all the spinors, and second, it organizes the tensors into \mathbb{Z}_2 -orbits. Such an orbit may have length two or one; the latter corresponds to a fixed-point under spectral flow, i.e., to a tensor with $\lambda_0 = \lambda_1$, which splits into two primaries in the quotient. The number of fixed-points corresponds to the number of solutions to $\lambda_0 + \lambda_1 + 2(\lambda_2 + \cdots + \lambda_{n-1}) + \lambda_n = k$ with $\lambda_0 = \lambda_1$ and λ_n even, i.e., $\binom{n+k/2-1}{k/2}$. Therefore, the number of conformal blocks is

$$\operatorname{tr}_{\mathrm{SO}(2n+1)_{2k}}(\mathbf{1}) = \frac{1}{2} \left(N_{\operatorname{tensor}}^{\operatorname{Spin}(2n+1)_{2k}} - \binom{n+k-1}{k} \right) + 2\binom{n+k-1}{k}.$$
 (6.6.7)

Let now k be odd, which makes $SO(2n + 1)_k$ a spin theory. The total number of states is the same on every spin structure, so we shall count the bosons and fermions in the Ramond sector (which is the richest case, as only this sector may contain fermions). The total number of states is the sum, while the Witten index is the difference. In the Ramond sector, the quotient projects out the tensors, and it organizes the spinors into \mathbb{Z}_2 -orbits. The bosons are the length-two orbits, and the fermions are the fixed-points. The latter are the representations with $\lambda_0 + \lambda_1 + 2(\lambda_2 + \cdots + \lambda_{n-1}) + \lambda_n = k$ with $\lambda_0 = \lambda_1$ and λ_n odd, which has $\binom{n+(k-1)/2-1}{(k-1)/2}$ solutions. Thus, the number of bosons and fermions is

$$N_{\text{boson}}^{\text{SO}(2n+1)_{2k+1}} = \frac{1}{2} \left(N_{\text{spinor}}^{\text{Spin}(2n+1)_{2k+1}} - \binom{n+k-1}{k} \right),$$

$$N_{\text{fermion}}^{\text{SO}(2n+1)_{2k+1}} = \binom{n+k-1}{k},$$
(6.6.8)

from where it follows that

$$\operatorname{tr}_{\mathrm{SO}(2n+1)_{2k+1}}(\mathbf{1}) = \operatorname{tr}_{\mathrm{SO}(2n+1)_{2k+1}}(-1)^{F} = N_{\mathrm{boson}}^{\mathrm{SO}(2n+1)_{2k+1}} + N_{\mathrm{fermion}}^{\mathrm{SO}(2n+1)_{2k+1}} = \binom{n+k}{k},$$
(6.6.9)

for all spatial spin structures, except for the odd structure for which

$$\operatorname{tr}_{\mathrm{SO}(2n+1)_{2k+1}}(-1)^{F} = N_{\mathrm{boson}}^{\mathrm{SO}(2n+1)_{2k+1}} - N_{\mathrm{fermion}}^{\mathrm{SO}(2n+1)_{2k+1}} = \binom{n+k}{k} - 2\binom{n+k-1}{k}.$$
(6.6.10)

We see that tr(1) is invariant under $n \leftrightarrow k$, as required by level-rank duality. Similarly, $tr(-1)^F$ is invariant up to a sign, which is due to the difference in the framing anomalies (i.e., the precise level-rank duality [112] is $SO(2n+1)_{2k+1} \leftrightarrow SO(2k+1)_{-2n-1} \times SO((2n+1)(2k+1))_1$, with the invertible factor contributing with a global factor of $(-1)^{(2n+1)(2k+1)} \equiv -1$ to the trace, cf. (6.3.25)).

$6.6.3 \quad G = \mathrm{Spin}(2n)$

Consider the algebra $D_n = \mathfrak{so}_{2n}$. The comarks are $a_i^{\vee} = 1$ for i = 0, 1, n - 1, n, and $a_i^{\vee} = 2$ for $i = 2, \ldots, n - 2$. Plugging this into (6.4.11) we get the generating function as

$$Z(\operatorname{Spin}(2n), q) = (1 - q)^{-4} (1 - q^2)^{-(n-3)}, \qquad (6.6.11)$$

and, by expanding, the untwisted trace

$$\operatorname{tr}_{\operatorname{Spin}(2n)_{2k}}(\mathbf{1}) = \binom{n+k}{k} + 6\binom{n+k-1}{k-1} + \binom{n+k-2}{k-2},$$

$$\operatorname{tr}_{\operatorname{Spin}(2n)_{2k+1}}(\mathbf{1}) = 4\binom{n+k}{k} + 4\binom{n+k-1}{k-1}.$$

(6.6.12)

This is the number of integrable representations of $\text{Spin}(2n)_k$, that is, the dimension of the torus Hilbert space of this Chern-Simons theory. For future reference, it is also useful to break up the states into the tensors and spinors. In other words, we shall be interested in knowing how many of the states of Spin(2n) are tensorial representations, and how many are spinorial representations. These are defined by $\lambda_{n-1} + \lambda_n = \text{even and } \lambda_{n-1} + \lambda_n = \text{odd}$, respectively, which yields the following:

$$N_{\text{tensor}}^{\text{Spin}(2n)_{2k}} = \binom{n+k}{k} + 2\binom{n+k-1}{k-1} + \binom{n+k-2}{k-2},$$

$$N_{\text{spinor}}^{\text{Spin}(2n)_{2k}} = 4\binom{n+k-1}{k-1},$$

$$N_{\text{tensor}}^{\text{Spin}(2n)_{2k+1}} = N_{\text{spinor}}^{\text{Spin}(2n)_{2k+1}} = 2\binom{n+k}{k} + 2\binom{n+k-1}{k-1},$$
(6.6.13)

so that

$$\operatorname{tr}_{\operatorname{Spin}(2n)_k}(\mathbf{1}) \equiv N_{\operatorname{tensor}}^{\operatorname{Spin}(2n)_k} + N_{\operatorname{spinor}}^{\operatorname{Spin}(2n)_k}.$$
(6.6.14)

For a more interesting example, let us now compute the partition function $SO(2n)_k = Spin(2n)_k/\mathbb{Z}_2$, which corresponds to the algebra \mathfrak{so}_{2n} extended by the simple current $\chi = [0, k, 0, \ldots, 0]$. This current has spin $h_{\chi} = k/2$, and so the extension is fermionic for odd k. The current acts on a given representation $[\lambda_0, \lambda_1, \ldots, \lambda_n]$ as $\lambda_0 \leftrightarrow \lambda_1$ and $\lambda_{n-1} \leftrightarrow \lambda_n$.

Consider first the case of even k, so that $SO(2n)_k$ makes sense as a bosonic theory. The extension has two effects: first, it projects out all the spinors, and second, it organizes the

tensors into \mathbb{Z}_2 -orbits. Such an orbit may have length two or one; the latter corresponds to a fixed-point under spectral flow, i.e., to a tensor with $\lambda_0 = \lambda_1$ and $\lambda_{n-1} = \lambda_n$, which splits into two primaries in the quotient. The number of fixed-points corresponds to the number of solutions to $\lambda_0 + \lambda_1 + 2(\lambda_2 + \cdots + \lambda_{n-2}) + \lambda_{n-1} + \lambda_n = k$ with $\lambda_0 = \lambda_1$ and $\lambda_{n-1} = \lambda_n$, i.e., $\binom{n+k/2-2}{k/2}$. Therefore, the number of conformal blocks is

$$\operatorname{tr}_{\mathrm{SO}(2n)_{2k}}(\mathbf{1}) = \frac{1}{2} \left(N_{\operatorname{tensor}}^{\operatorname{Spin}(2n)_{2k}} - \binom{n+k-2}{k} \right) + 2\binom{n+k-2}{k}.$$
(6.6.15)

Let now k be odd, which makes $SO(2n)_k$ a spin theory. The number of states is the same on every spin structure, so we shall count the bosons and fermions in the Ramond sector (which is the richest case, as only this sector may contain fermions). The total number of states is the sum, while the Witten index is the difference. In the Ramond sector, the quotient projects out the tensors, and it organizes the spinors into \mathbb{Z}_2 -orbits. The bosons are the length-two orbits, and the fermions are the fixed-points. Note that the spinors have $\lambda_{n-1} + \lambda_n = \text{odd}$, which is incompatible with the fixed-point condition $\lambda_{n-1} = \lambda_n$, and so there are no fixed-points. Thus, the number of bosons and fermions is

$$N_{\text{boson}}^{\text{SO}(2n)_{2k+1}} = \frac{1}{2} N_{\text{spinor}}^{\text{Spin}(2n)_{2k+1}},$$

$$N_{\text{fermion}}^{\text{SO}(2n)_{2k+1}} = 0,$$
(6.6.16)

from where it follows that

$$\operatorname{tr}_{\mathrm{SO}(2n)_{2k+1}}(\mathbf{1}) = \operatorname{tr}_{\mathrm{SO}(2n)_{2k+1}}(-1)^F = N_{\mathrm{boson}}^{\mathrm{SO}(2n)_{2k+1}}, \qquad (6.6.17)$$

for all spatial spin structures. Note that the equality of $\operatorname{tr}(\mathbf{1}), \operatorname{tr}(-1)^F$ on all spin structures was in fact expected from the level-rank duality $\operatorname{SO}(2n)_{2k+1} \leftrightarrow \operatorname{SO}(2k+1)_{-2n}$, the *r.h.s.* being fermionic only due to a trivial $\operatorname{SO}(2n(2k+1))_1 = \{\mathbf{1}, \psi\}$ factor (which contains an even number of fermions, so not even the sign of $\operatorname{tr}(-1)^F$ may depend on the spin structure).

6.7 The Exceptional Groups

In this section we gather the different indices for the exceptional groups, whose domain wall theory is yet to be identified. Any given proposal for the dynamics of such walls ought to be consistent with the indices below. By particle-hole symmetry, the indices satisfy $I_n^{s} = \pm I_{h-n}^{s}$, and therefore we only show the first $\lceil h/2 \rceil$ indices, so as to avoid repetition.

We compute the untwisted indices, and the indices twisted by the zero-form and one-form symmetries (see table 6.1). The symmetries $c \in C = \mathbb{Z}_2$ and $g \in \Gamma = \mathbb{Z}_3$ act on the Dynkin

diagram of E_6 as follows:

The symmetry $\mathbf{g} \in \Gamma$ acts on E_7 as follows:

$$E_{7}^{(1)}: \qquad \stackrel{1}{\overset{2}{\longrightarrow}} \stackrel{3}{\overset{2}{\longrightarrow}} \stackrel{2}{\overset{1}{\longrightarrow}} \stackrel{1}{\longrightarrow} F_{4}^{(1)}: \qquad \stackrel{2}{\overset{4}{\longrightarrow}} \stackrel{6}{\overset{4}{\longrightarrow}} \stackrel{2}{\overset{2}{\longrightarrow}} \stackrel{2}{\overset{2}{\longrightarrow}} \stackrel{6}{\overset{4}{\longrightarrow}} \stackrel{2}{\overset{2}{\longrightarrow}} (6.7.2)$$

Using these diagrams we find:

• E_6 :

$$Z(q) = 1 - 3q + 7q^3 - 3q^4 - 6q^5 + \cdots$$

$$Z^{c}(q) = 1 - q - 2q^2 + q^3 + q^4 + 2q^5 + \cdots$$

$$Z^{g}(q) = 1 - 2q^3 + \cdots$$

(6.7.3)

• E_7 :

$$Z(q) = 1 - 2q - 2q^{2} + 4q^{3} + 3q^{4} - 7q^{6} - 4q^{7} + 5q^{8} + 4q^{9} + \cdots$$

$$Z^{g}(q) = 1 - 2q^{2} - q^{4} + 3q^{6} + q^{8} + \cdots$$
(6.7.4)

• E_8 :

$$Z(q) = 1 - q - 2q^{2} + q^{4} + 4q^{5} + q^{6} - 3q^{8} - 6q^{9} - q^{10} + 4q^{12} + 5q^{13} + 5q^{14} + \dots$$
(6.7.5)

• F_4 :

$$Z(q) = 1 - 2q - q^2 + 3q^3 + q^4 + \cdots$$
(6.7.6)

References

- Diego Delmastro, Davide Gaiotto, and Jaume Gomis. "Global anomalies on the Hilbert space". In: JHEP 11 (2021), p. 142. DOI: 10.1007/JHEP11(2021)142. arXiv: 2101.02218 [hep-th].
- [2] Diego Delmastro, Jaume Gomis, and Matthew Yu. "Infrared phases of 2d QCD". In: (Aug. 2021). arXiv: 2108.02202 [hep-th].
- [3] Changha Choi, Diego Delmastro, Jaume Gomis, and Zohar Komargodski. "Dynamics of QCD₃ with Rank-Two Quarks And Duality". In: *JHEP* 03 (2020), p. 078. DOI: 10.1007/JHEP03(2020)078. arXiv: 1810.07720 [hep-th].
- [4] Diego Delmastro and Jaume Gomis. "Symmetries of Abelian Chern-Simons Theories and Arithmetic". In: JHEP 03 (2021), p. 006. DOI: 10.1007/JHEP03(2021)006. arXiv: 1904.12884 [hep-th].
- [5] Diego Delmastro and Jaume Gomis. "Domain walls in $4d\mathcal{N} = 1$ SYM". In: *JHEP* 03 (2021), p. 259. DOI: 10.1007/JHEP03(2021)259. arXiv: 2004.11395 [hep-th].
- [6] Steven Weinberg. *The Quantum Theory of Fields*. Vol. 1 and 2. Cambridge University Press.
- [7] Gerard 't Hooft. "Naturalness, chiral symmetry, and spontaneous chiral symmetry breaking". In: NATO Sci. Ser. B 59 (1980). Ed. by Gerard 't Hooft, C. Itzykson, A. Jaffe, H. Lehmann, P.K. Mitter, I.M. Singer, and R. Stora, pp. 135–157. DOI: 10.1007/978-1-4684-7571-5_9.
- [8] E. Wigner. "Über die Operation der Zeitumkehr in der Quantenmechanik". In: Nachrichten von der Gesellschaft der Wissenschaften zu Göttingen, Mathematisch-Physikalische Klasse (1932), pp. 546–559. DOI: 10.1007/978-3-662-02781-3_15.
- [9] Iñaki García-Etxebarria and Miguel Montero. "Dai-Freed anomalies in particle physics".
 In: JHEP 08 (2019), p. 003. DOI: 10.1007/JHEP08(2019)003. arXiv: 1808.00009
 [hep-th].
- [10] F. Hirzebruch M. F. Atiyah. "Vector bundles and homogeneous spaces". In: Topological Library, pp. 423–454. DOI: 10.1142/9789814401319_0008.

- [11] Ryan Thorngren. "Anomalies and Bosonization". In: Commun. Math. Phys. 378.3 (2020), pp. 1775–1816. DOI: 10.1007/s00220-020-03830-0. arXiv: 1810.04414
 [cond-mat.str-el].
- [12] A A Migdal. "Recursion equations in gauge field theories". In: Zh. Eksp. Teor. Fiz., v. 69, no. 3, pp. 810-822 (Sept. 1975).
- [13] Edward Witten. "On quantum gauge theories in two dimensions". In: Communications in Mathematical Physics 141.1 (Oct. 1991), pp. 153–209. DOI: 10.1007/BF02100009.
- [14] Edward Witten. "Nonabelian Bosonization in Two-Dimensions". In: Commun. Math. Phys. 92 (1984). Ed. by M. Stone, pp. 455–472. DOI: 10.1007/BF01215276.
- [15] Giovanni Felder, K. Gawedzki, and A. Kupiainen. "Spectra of Wess-Zumino-Witten Models With Arbitrary Simple Groups". In: Commun. Math. Phys. 117 (1988), pp. 127– 158. DOI: 10.1007/BF01228414.
- [16] K. Gawedzki and A. Kupiainen. "Coset construction from functional integrals". In: *Nuclear Physics B* 320.3 (1989), pp. 625–668. ISSN: 0550-3213. DOI: https://doi. org/10.1016/0550-3213(89)90015-1.
- [17] Dimitra Karabali, Q-Han Park, Howard J. Schnitzer, and Zhu Yang. "A GKO Construction Based on a Path Integral Formulation of Gauged Wess-Zumino-Witten Actions". In: *Phys. Lett. B* 216 (1989), pp. 307–312. DOI: 10.1016/0370-2693(89)91120-9.
- [18] Dimitra Karabali and Howard J. Schnitzer. "BRST Quantization of the Gauged WZW Action and Coset Conformal Field Theories". In: Nucl. Phys. B 329 (1990), pp. 649–666. DOI: 10.1016/0550-3213(90)90075-0.
- [19] D. Kutasov and A. Schwimmer. "Universality in two-dimensional gauge theory". In: Nucl. Phys. B 442 (1995), pp. 447–460. DOI: 10.1016/0550-3213(95)00106-3. arXiv: hep-th/9501024.
- [20] Jonathan F. Schonfeld. "A Mass Term for Three-Dimensional Gauge Fields". In: Nucl. Phys. B 185 (1981), pp. 157–171. DOI: 10.1016/0550-3213(81)90369-2.
- [21] S Deser, R Jackiw, and S Templeton. "Topologically massive gauge theories". In: Annals of Physics 140.2 (1982), pp. 372–411. ISSN: 0003-4916. DOI: https://doi. org/10.1016/0003-4916(82)90164-6.
- [22] Edward Witten. "Quantum field theory and the Jones polynomial". In: Comm. Math. Phys. 121.3 (1989), pp. 351–399.
- [23] A. J. Niemi and G. W. Semenoff. "Axial-Anomaly-Induced Fermion Fractionization and Effective Gauge-Theory Actions in Odd-Dimensional Space-Times". In: *Phys. Rev. Lett.* 51 (23 1983), pp. 2077–2080. DOI: 10.1103/PhysRevLett.51.2077.

- [24] A. N. Redlich. "Parity Violation and Gauge Noninvariance of the Effective Gauge Field Action in Three-Dimensions". In: *Phys. Rev.* D29 (1984). [,2366(1983)], pp. 2366–2374. DOI: 10.1103/PhysRevD.29.2366.
- [25] Thomas Appelquist and Daniel Nash. "Critical Behavior in (2+1)-dimensional QCD". In: Phys. Rev. Lett. 64 (1990), p. 721. DOI: 10.1103/PhysRevLett.64.721.
- [26] L. V. Avdeev, G. V. Grigorev, and D. I. Kazakov. "Renormalizations in Abelian Chern-Simons field theories with matter". In: *Nucl. Phys.* B382 (1992), pp. 561–580. DOI: 10.1016/0550-3213(92)90659-Y.
- [27] Adi Armoni, Thomas T. Dumitrescu, Guido Festuccia, and Zohar Komargodski.
 "Metastable vacua in large-N QCD₃". In: *JHEP* 01 (2020), p. 004. DOI: 10.1007/ JHEP01(2020)004. arXiv: 1905.01797 [hep-th].
- [28] Po-Shen Hsin and Nathan Seiberg. "Level/rank Duality and Chern-Simons-Matter Theories". In: JHEP 09 (2016), p. 095. DOI: 10.1007/JHEP09(2016)095. arXiv: 1607.07457 [hep-th].
- [29] Zohar Komargodski and Nathan Seiberg. "A symmetry breaking scenario for QCD₃".
 In: *JHEP* 01 (2018), p. 109. DOI: 10.1007/JHEP01(2018)109. arXiv: 1706.08755
 [hep-th].
- [30] Jaume Gomis, Zohar Komargodski, and Nathan Seiberg. "Phases Of Adjoint QCD₃ And Dualities". In: SciPost Phys. 5.1 (2018), p. 007. DOI: 10.21468/SciPostPhys.5.
 1.007. arXiv: 1710.03258 [hep-th].
- [31] Xiao-Gang Wen. "Topological orders and edge excitations in fractional quantum Hall states". In: Advances in Physics 44.5 (1995), 405–473. ISSN: 1460-6976. DOI: 10.1080/00018739500101566.
- [32] Davide Gaiotto, Anton Kapustin, Nathan Seiberg, and Brian Willett. "Generalized Global Symmetries". In: JHEP 02 (2015), p. 172. DOI: 10.1007/JHEP02(2015)172. arXiv: 1412.5148 [hep-th].
- [33] Jennifer Cano, Meng Cheng, Michael Mulligan, Chetan Nayak, Eugeniu Plamadeala, and Jon Yard. "Bulk-edge correspondence in (2 + 1)-dimensional Abelian topological phases". In: *Phys. Rev.* B89.11 (2014), p. 115116. DOI: 10.1103/PhysRevB.89.115116. arXiv: 1310.5708 [cond-mat.str-el].
- [34] Clay Córdova, Po-Shen Hsin, and Nathan Seiberg. "Time-Reversal Symmetry, Anomalies, and Dualities in (2+1)d". In: SciPost Phys. 5 (2018), p. 006. DOI: 10.21468/ SciPostPhys.5.1.006. arXiv: 1712.08639 [cond-mat.str-el].
- [35] Robbert Dijkgraaf and Edward Witten. "Topological Gauge Theories and Group Cohomology". In: Commun. Math. Phys. 129 (1990), p. 393. DOI: 10.1007/BF02096988.

- [36] Davide Gaiotto and Anton Kapustin. "Spin TQFTs and fermionic phases of matter". In: Int. J. Mod. Phys. A 31.28n29 (2016). Ed. by Yuri L. Dokshitzer, Peter Levai, and Julia Nyiri, p. 1645044. DOI: 10.1142/S0217751X16450445. arXiv: 1505.05856
 [cond-mat.str-el].
- [37] Lakshya Bhardwaj, Davide Gaiotto, and Anton Kapustin. "State sum constructions of spin-TFTs and string net constructions of fermionic phases of matter". In: *JHEP* 04 (2017), p. 096. DOI: 10.1007/JHEP04(2017)096. arXiv: 1605.01640 [cond-mat.str-el].
- [38] Gregory W. Moore and Nathan Seiberg. "Classical and Quantum Conformal Field Theory". In: Commun. Math. Phys. 123 (1989), p. 177. DOI: 10.1007/BF01238857.
- [39] Erik Verlinde. "Fusion rules and modular transformations in 2D conformal field theory". In: Nuclear Physics B 300 (1988), pp. 360 -376. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(88)90603-7.
- [40] Davide Gaiotto, Anton Kapustin, Zohar Komargodski, and Nathan Seiberg. "Theta, Time Reversal, and Temperature". In: JHEP 05 (2017), p. 091. DOI: 10.1007/ JHEP05(2017)091. arXiv: 1703.00501 [hep-th].
- [41] Davide Gaiotto, Zohar Komargodski, and Nathan Seiberg. "Time-reversal breaking in QCD₄, walls, and dualities in 2 + 1 dimensions". In: JHEP 01 (2018), p. 110. DOI: 10.1007/JHEP01(2018)110. arXiv: 1708.06806 [hep-th].
- [42] Edward Witten. "Constraints on supersymmetry breaking". In: Nuclear Physics B 202.2 (1982), pp. 253 –316. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(82)90071-2.
- [43] Ian Affleck, Michael Dine, and Nathan Seiberg. "Dynamical supersymmetry breaking in four dimensions and its phenomenological implications". In: Nuclear Physics B 256 (1985), pp. 557 –599. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(85)90408-0.
- [44] M.A. Shifman and A.I. Vainshtein. "On gluino condensation in supersymmetric gauge theories with SU(N) and O(N) groups". In: *Nuclear Physics B* 296.2 (1988), pp. 445 –461. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(88)90680-3.
- [45] A.Yu. Morozov, M.A. Olshanetsky, and M.A. Shifman. "Gluino condensate in supersymmetric gluodynamics (II)". In: *Nuclear Physics B* 304 (1988), pp. 291 –310. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(88)90628-1.
- [46] Bobby Samir Acharya and Cumrun Vafa. "On domain walls of N=1 supersymmetric Yang-Mills in four-dimensions". In: (2001). arXiv: hep-th/0103011 [hep-th].

- [47] A. Armoni, M. Shifman, and G. Veneziano. "SUSY relics in one flavor QCD from a new 1/N expansion". In: *Phys. Rev. Lett.* 91 (2003), p. 191601. DOI: 10.1103/ PhysRevLett.91.191601. arXiv: hep-th/0307097 [hep-th].
- [48] Bruno Zumino. "Chiral anomalies and differential geometry: Lectures given at Les Houches, August 1983". In: Les Houches Summer School on Theoretical Physics: Relativity, Groups and Topology. Oct. 1983, pp. 1291–1322.
- [49] Raymond Stora. "Algebraic Structure and Topological Origin of Anomalies". In: NATO Sci. Ser. B 115 (1984). Ed. by Gerard 't Hooft, A. Jaffe, H. Lehmann, P.K. Mitter, I.M. Singer, and R. Stora.
- [50] C.G. Callan and J.A. Harvey. "Anomalies and fermion zero modes on strings and domain walls". In: *Nuclear Physics B* 250.1 (1985), pp. 427 –436. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(85)90489-4.
- [51] Xie Chen, Zheng-Cheng Gu, Zheng-Xin Liu, and Xiao-Gang Wen. "Symmetry-Protected Topological Orders in Interacting Bosonic Systems". In: Science 338.6114 (2012), pp. 1604–1606. DOI: 10.1126/science.1227224. arXiv: 1301.0861 [cond-mat.str-el].
- [52] Xie Chen, Zheng-Cheng Gu, Zheng-Xin Liu, and Xiao-Gang Wen. "Symmetry protected topological orders and the group cohomology of their symmetry group". In: *Phys. Rev. B* 87 (15 2013), p. 155114. DOI: 10.1103/PhysRevB.87.155114. URL: https://link.aps.org/doi/10.1103/PhysRevB.87.155114.
- [53] T. Senthil. "Symmetry Protected Topological phases of Quantum Matter". In: Ann. Rev. Condensed Matter Phys. 6 (2015), p. 299. DOI: 10.1146/annurev-conmatphys-031214-014740. arXiv: 1405.4015 [cond-mat.str-el].
- [54] Zheng-Cheng Gu and Xiao-Gang Wen. "Symmetry-protected topological orders for interacting fermions: Fermionic topological nonlinear σ models and a special group supercohomology theory". In: *Phys. Rev. B* 90.11 (2014), p. 115141. DOI: 10.1103/ PhysRevB.90.115141. arXiv: 1201.2648 [cond-mat.str-el].
- [55] Xiao-Gang Wen. "Classifying gauge anomalies through symmetry-protected trivial orders and classifying gravitational anomalies through topological orders". In: *Phys. Rev. D* 88.4 (2013), p. 045013. DOI: 10.1103/PhysRevD.88.045013. arXiv: 1303.1803 [hep-th].
- [56] Alexei Kitaev. "Homotopy-theoretic approach to SPT phases in action: Z16 classification of three-dimensional superconductors". In: Lecture notes available at ipam.ucla.edu. IPAM program Symmetry and Topology in Quantum Matter. Jan. 2015.

- [57] Xie Chen, Yuan-Ming Lu, and Ashvin Vishwanath. "Symmetry-protected topological phases from decorated domain walls". In: *Nature Communications* 5.1 (2014). ISSN: 2041-1723. DOI: 10.1038/ncomms4507. URL: http://dx.doi.org/10.1038/ ncomms4507.
- [58] Anton Kapustin. "Symmetry Protected Topological Phases, Anomalies, and Cobordisms: Beyond Group Cohomology". In: (2014). arXiv: 1403.1467 [cond-mat.str-el].
- [59] Anton Kapustin, Ryan Thorngren, Alex Turzillo, and Zitao Wang. "Fermionic symmetry protected topological phases and cobordisms". In: *Journal of High Energy Physics* 2015.12 (2015), pp. 1–21. ISSN: 1029-8479. DOI: 10.1007/JHEP12(2015)052. URL: https://doi.org/10.1007/JHEP12(2015)052.
- [60] Daniel S. Freed and Michael J. Hopkins. "Reflection positivity and invertible topological phases". In: (Apr. 2016). arXiv: 1604.06527 [hep-th].
- [61] Chang-Tse Hsieh, Olabode Mayodele Sule, Gil Young Cho, Shinsei Ryu, and Robert G. Leigh. "Symmetry-protected topological phases, generalized Laughlin argument, and orientifolds". In: *Phys. Rev. B* 90 (16 2014), p. 165134. DOI: 10.1103/PhysRevB.90. 165134.
- [62] Edward Witten. "The "Parity" Anomaly On An Unorientable Manifold". In: *Phys. Rev.* B 94.19 (2016), p. 195150. DOI: 10.1103/PhysRevB.94.195150. arXiv: 1605.02391
 [hep-th].
- [63] Yuji Tachikawa and Kazuya Yonekura. "On time-reversal anomaly of 2+1d topological phases". In: *PTEP* 2017.3 (2017), 033B04. DOI: 10.1093/ptep/ptx010. arXiv: 1610.07010 [hep-th].
- [64] Maissam Barkeshli, Parsa Bonderson, Meng Cheng, Chao-Ming Jian, and Kevin Walker. "Reflection and Time Reversal Symmetry Enriched Topological Phases of Matter: Path Integrals, Non-orientable Manifolds, and Anomalies". In: Commun. Math. Phys. 374.2 (2019), pp. 1021–1124. DOI: 10.1007/s00220-019-03475-8. arXiv: 1612.07792 [cond-mat.str-el].
- [65] Charles Zhaoxi Xiong. "Minimalist approach to the classification of symmetry protected topological phases". In: J. Phys. A 51.44 (2018), p. 445001. DOI: 10.1088/1751-8121/aae0b1. arXiv: 1701.00004 [cond-mat.str-el].
- [66] Anton Kapustin and Ryan Thorngren. "Fermionic SPT phases in higher dimensions and bosonization". In: JHEP 10 (2017), p. 080. DOI: 10.1007/JHEP10(2017)080. arXiv: 1701.08264 [cond-mat.str-el].
- [67] Davide Gaiotto and Theo Johnson-Freyd. "Symmetry Protected Topological phases and Generalized Cohomology". In: JHEP 05 (2019), p. 007. DOI: 10.1007/JHEP05(2019) 007. arXiv: 1712.07950 [hep-th].

- [68] Qing-Rui Wang and Zheng-Cheng Gu. "Construction and classification of symmetry protected topological phases in interacting fermion systems". In: *Phys. Rev. X* 10.3 (2020), p. 031055. DOI: 10.1103/PhysRevX.10.031055. arXiv: 1811.00536 [cond-mat.str-el].
- [69] Kazuya Yonekura. "On the cobordism classification of symmetry protected topological phases". In: Commun. Math. Phys. 368.3 (2019), pp. 1121–1173. DOI: 10.1007/s00220-019-03439-y. arXiv: 1803.10796 [hep-th].
- [70] Edward Witten and Kazuya Yonekura. "Anomaly Inflow and the η-Invariant". In: The Shoucheng Zhang Memorial Workshop. Sept. 2019. arXiv: 1909.08775 [hep-th].
- [71] Meng Guo, Kantaro Ohmori, Pavel Putrov, Zheyan Wan, and Juven Wang. "Fermionic Finite-Group Gauge Theories and Interacting Symmetric/Crystalline Orders via Cobordisms". In: *Commun. Math. Phys.* 376.2 (2020), pp. 1073–1154. DOI: 10.1007/ s00220-019-03671-6. arXiv: 1812.11959 [hep-th].
- [72] Zheyan Wan and Juven Wang. "Higher anomalies, higher symmetries, and cobordisms I: classification of higher-symmetry-protected topological states and their boundary fermionic/bosonic anomalies via a generalized cobordism theory". In: Ann. Math. Sci. Appl. 4.2 (2019), pp. 107–311. DOI: 10.4310/AMSA.2019.v4.n2.a2. arXiv: 1812.11967 [hep-th].
- [73] Zheyan Wan and Juven Wang. "Beyond Standard Models and Grand Unifications: Anomalies, Topological Terms, and Dynamical Constraints via Cobordisms". In: *JHEP* 07 (2020), p. 062. DOI: 10.1007/JHEP07(2020)062. arXiv: 1910.14668 [hep-th].
- [74] Daniel Bulmash and Maissam Barkeshli. "Absolute anomalies in (2+1)D symmetryenriched topological states and exact (3+1)D constructions". In: *Phys. Rev. Res.* 2.4 (2020), p. 043033. DOI: 10.1103/PhysRevResearch.2.043033. arXiv: 2003.11553
 [cond-mat.str-el].
- [75] Edward Witten. "An SU(2) anomaly". In: *Physics Letters B* 117.5 (1982), pp. 324
 -328. ISSN: 0370-2693. DOI: https://doi.org/10.1016/0370-2693(82)90728-6.
- [76] Juven Wang, Xiao-Gang Wen, and Edward Witten. "A New SU(2) Anomaly". In: J. Math. Phys. 60.5 (2019), p. 052301. DOI: 10.1063/1.5082852. arXiv: 1810.00844
 [hep-th].
- [77] AtMa P O Chan, Jeffrey C Y Teo, and Shinsei Ryu. "Topological phases on non-orientable surfaces: twisting by parity symmetry". In: New Journal of Physics 18.3 (2016), p. 035005. ISSN: 1367-2630. DOI: 10.1088/1367-2630/18/3/035005. URL: http://dx.doi.org/10.1088/1367-2630/18/3/035005.

- [78] Chenjie Wang and Michael Levin. "Anomaly Indicators for Time-Reversal Symmetric Topological Orders". In: *Phys. Rev. Lett.* 119 (13 2017), p. 136801. DOI: 10.1103/ PhysRevLett.119.136801. arXiv: 1610.04624 [cond-mat.str-el].
- [79] Yuji Tachikawa and Kazuya Yonekura. "More on time-reversal anomaly of 2+1d topological phases". In: *Phys. Rev. Lett.* 119.11 (2017), p. 111603. DOI: 10.1103/ PhysRevLett.119.111603. arXiv: 1611.01601 [hep-th].
- [80] Lakshya Bhardwaj. "Unoriented 3d TFTs". In: JHEP 05 (2017), p. 048. DOI: 10.
 1007/JHEP05(2017)048. arXiv: 1611.02728 [hep-th].
- [81] Maissam Barkeshli and Meng Cheng. "Time-reversal and spatial-reflection symmetry localization anomalies in (2+1)-dimensional topological phases of matter". In: *Phys. Rev. B* 98.11 (2018), p. 115129. DOI: 10.1103/PhysRevB.98.115129. arXiv: 1706.09464 [cond-mat.str-el].
- [82] Alex Turzillo. "Diagrammatic State Sums for 2D Pin-Minus TQFTs". In: JHEP 03 (2020), p. 019. DOI: 10.1007/JHEP03(2020)019. arXiv: 1811.12654 [math.QA].
- [83] Ryohei Kobayashi. "Pin TQFT and Grassmann integral". In: JHEP 12 (2019), p. 014.
 DOI: 10.1007/JHEP12(2019)014. arXiv: 1905.05902 [cond-mat.str-el].
- [84] Kansei Inamura, Ryohei Kobayashi, and Shinsei Ryu. "Non-local Order Parameters and Quantum Entanglement for Fermionic Topological Field Theories". In: JHEP 01 (2020), p. 121. DOI: 10.1007/JHEP01(2020)121. arXiv: 1911.00653 [cond-mat.str-el].
- [85] Joe Davighi and Nakarin Lohitsiri. "The algebra of anomaly interplay". In: (Nov. 2020). arXiv: 2011.10102 [hep-th].
- [86] Lukasz Fidkowski and Alexei Kitaev. "The effects of interactions on the topological classification of free fermion systems". In: *Phys. Rev. B* 81 (2010), p. 134509. DOI: 10.1103/PhysRevB.81.134509. arXiv: 0904.2197 [cond-mat.str-el].
- [87] Max A. Metlitski, Lukasz Fidkowski, Xie Chen, and Ashvin Vishwanath. "Interaction effects on 3D topological superconductors: surface topological order from vortex condensation, the 16 fold way and fermionic Kramers doublets". In: (2014). arXiv: 1406.3032 [cond-mat.str-el].
- [88] Chong Wang and T. Senthil. "Interacting fermionic topological insulators/superconductors in three dimensions". In: *Phys. Rev.* B89.19 (2014). [Erratum: Phys. Rev.B91,no.23,239902(2015)], p. 195124. DOI: 10.1103/PhysRevB.89.195124, 10.1103/PhysRevB.91.239902. arXiv: 1401.1142 [cond-mat.str-el].
- [89] Edward Witten. "Fermion Path Integrals And Topological Phases". In: Rev. Mod. Phys. 88.3 (2016), p. 035001. DOI: 10.1103/RevModPhys.88.035001. arXiv: 1508.04715
 [cond-mat.mes-hall].
- [90] Clay Córdova, Kantaro Ohmori, Shu-Heng Shao, and Fei Yan. "Decorated Z₂ symmetry defects and their time-reversal anomalies". In: *Phys. Rev. D* 102.4 (2020), p. 045019. DOI: 10.1103/PhysRevD.102.045019. arXiv: 1910.14046 [hep-th].
- [91] Itamar Hason, Zohar Komargodski, and Ryan Thorngren. "Anomaly Matching in the Symmetry Broken Phase: Domain Walls, CPT, and the Smith Isomorphism". In: SciPost Phys. 8.4 (2020), p. 062. DOI: 10.21468/SciPostPhys.8.4.062. arXiv: 1910.14039 [hep-th].
- [92] Zheng-Cheng Gu and Michael Levin. "The effect of interactions on 2D fermionic symmetry-protected topological phases with Z2 symmetry". In: *Phys. Rev. B* 89 (2014), p. 201113. DOI: 10.1103/PhysRevB.89.201113. arXiv: 1304.4569 [cond-mat.str-el].
- [93] Chenjie Wang, Chien-Hung Lin, and Zheng-Cheng Gu. "Interacting fermionic symmetry-protected topological phases in two dimensions". In: *Phys. Rev. B* 95.19 (2017), p. 195147. DOI: 10.1103/PhysRevB.95.195147. arXiv: 1610.08478 [cond-mat.str-el].
- [94] Ying-Hsuan Lin and Shu-Heng Shao. "Anomalies and Bounds on Charged Operators".
 In: *Phys. Rev. D* 100.2 (2019), p. 025013. DOI: 10.1103/PhysRevD.100.025013.
 arXiv: 1904.04833 [hep-th].
- [95] Xiao-Liang Qi, Taylor L. Hughes, S. Raghu, and Shou-Cheng Zhang. "Time-Reversal-Invariant Topological Superconductors and Superfluids in Two and Three Dimensions". In: *Phys. Rev. Lett.* 102 (18 2009), p. 187001. DOI: 10.1103/PhysRevLett.102. 187001.
- [96] Abhishodh Prakash and Juven Wang. "Boundary supersymmetry of 1+1 d fermionic SPT phases". In: (Nov. 2020). arXiv: 2011.12320 [cond-mat.str-el].
- [97] Alex Turzillo and Minyoung You. "Supersymmetric boundaries of one-dimensional phases of fermions beyond SPTs". In: (Dec. 2020). arXiv: 2012.04621 [cond-mat.str-el].
- [98] Abhishodh Prakash and Juven Wang. "Unwinding fermionic symmetry-protected topological phases: Supersymmetry extension". In: *Phys. Rev. B* 103.8 (2021), p. 085130.
 DOI: 10.1103/PhysRevB.103.085130. arXiv: 2011.13921 [cond-mat.str-el].
- [99] Aleksey Cherman, Theodore Jacobson, Yuya Tanizaki, and Mithat Unsal. "Anomalies, a mod 2 index, and dynamics of 2d adjoint QCD". In: *SciPost Phys.* 8.5 (2020), p. 072. DOI: 10.21468/SciPostPhys.8.5.072. arXiv: 1908.09858 [hep-th].
- [100] Nathanan Tantivasadakarn. "Dimensional Reduction and Topological Invariants of Symmetry-Protected Topological Phases". In: *Phys. Rev. B* 96.19 (2017), p. 195101.
 DOI: 10.1103/PhysRevB.96.195101. arXiv: 1706.09769 [cond-mat.str-el].
- [101] Justin Kaidi, Julio Parra-Martinez, Yuji Tachikawa, and Arun Debray. "Topological Superconductors on Superstring Worldsheets". In: *SciPost Phys.* 9 (2020), p. 10. DOI: 10.21468/SciPostPhys.9.1.010. arXiv: 1911.11780 [hep-th].

- [102] Douglas Stanford and Edward Witten. "JT Gravity and the Ensembles of Random Matrix Theory". In: (July 2019). arXiv: 1907.03363 [hep-th].
- [103] Alex Turzillo and Minyoung You. "Fermionic matrix product states and one-dimensional short-range entangled phases with antiunitary symmetries". In: *Phys. Rev. B* 99.3 (2019), p. 035103. DOI: 10.1103/PhysRevB.99.035103. arXiv: 1710.00140 [cond-mat.str-el].
- [104] Alexei Kitaev. "A simple model of quantum holography". In: Video talks: Part I and Part II. Fundamental Physics Prize Symposium. 2015.
- [105] Subir Sachdev and Jinwu Ye. "Gapless spin-fluid ground state in a random quantum Heisenberg magnet". In: *Phys. Rev. Lett.* 70 (21 1993), pp. 3339-3342. DOI: 10.1103/PhysRevLett.70.3339. URL: https://link.aps.org/doi/10.1103/PhysRevLett.70.3339.
- [106] Yi-Zhuang You, Andreas W. W. Ludwig, and Cenke Xu. "Sachdev-Ye-Kitaev model and thermalization on the boundary of many-body localized fermionic symmetryprotected topological states". In: *Physical Review B* 95.11 (2017). ISSN: 2469-9969. DOI: 10.1103/physrevb.95.115150.
- [107] Jan Behrends and Benjamin Béri. "Supersymmetry in the Standard Sachdev-Ye-Kitaev Model". In: *Phys. Rev. Lett.* 124 (23 2020), p. 236804. DOI: 10.1103/PhysRevLett.124. 236804. URL: https://link.aps.org/doi/10.1103/PhysRevLett.124.236804.
- [108] Lukasz Fidkowski, Xie Chen, and Ashvin Vishwanath. "Non-Abelian Topological Order on the Surface of a 3D Topological Superconductor from an Exactly Solved Model". In: *Phys. Rev.* X3.4 (2013), p. 041016. DOI: 10.1103/PhysRevX.3.041016. arXiv: 1305.5851 [cond-mat.str-el].
- [109] Yi-Zhuang You and Cenke Xu. "Symmetry-protected topological states of interacting fermions and bosons". In: *Physical Review B* 90.24 (2014). ISSN: 1550-235X. DOI: 10.1103/physrevb.90.245120. URL: http://dx.doi.org/10.1103/PhysRevB.90.245120.
- [110] Yuji Tachikawa and Kazuya Yonekura. "Gauge interactions and topological phases of matter". In: *PTEP* 2016.9 (2016), 093B07. DOI: 10.1093/ptep/ptw131. arXiv: 1604.06184 [hep-th].
- [111] Edward Witten and Nathan Seiberg. "Gapped boundary phases of topological insulators via weak coupling". In: *Progress of Theoretical and Experimental Physics* 2016.12 (Nov. 2016). ISSN: 2050-3911. DOI: 10.1093/ptep/ptw083.
- [112] Ofer Aharony, Francesco Benini, Po-Shen Hsin, and Nathan Seiberg. "Chern-Simonsmatter dualities with SO and USp gauge groups". In: JHEP 02 (2017), p. 072. DOI: 10.1007/JHEP02(2017)072. arXiv: 1611.07874 [cond-mat.str-el].

- [113] Clay Córdova, Po-Shen Hsin, and Nathan Seiberg. "Global Symmetries, Counterterms, and Duality in Chern-Simons Matter Theories with Orthogonal Gauge Groups". In: SciPost Phys. 4.4 (2018), p. 021. DOI: 10.21468/SciPostPhys.4.4.021. arXiv: 1711.10008 [hep-th].
- Shinsei Ryu and Shou-Cheng Zhang. "Interacting topological phases and modular invariance". In: *Physical Review B* 85.24 (2012). ISSN: 1550-235X. DOI: 10.1103/ physrevb.85.245132. URL: http://dx.doi.org/10.1103/PhysRevB.85.245132.
- [115] Xiao-Liang Qi. "A new class of (2 + 1)-dimensional topological superconductors with Z₈ topological classification". In: New Journal of Physics 15.6 (2013), p. 065002. ISSN: 1367-2630. DOI: 10.1088/1367-2630/15/6/065002. URL: http://dx.doi.org/10.1088/1367-2630/15/6/065002.
- [116] Hong Yao and Shinsei Ryu. "Interaction effect on topological classification of superconductors in two dimensions". In: *Physical Review B* 88.6 (2013). ISSN: 1550-235X. DOI: 10.1103/physrevb.88.064507. URL: http://dx.doi.org/10.1103/PhysRevB. 88.064507.
- [117] Andreas Karch, David Tong, and Carl Turner. "A Web of 2d Dualities: Z₂ Gauge Fields and Arf Invariants". In: *SciPost Phys.* 7 (2019), p. 007. DOI: 10.21468/SciPostPhys. 7.1.007. arXiv: 1902.05550 [hep-th].
- Tian Lan, Liang Kong, and Xiao-Gang Wen. "Classification of (2+1)-dimensional topological order and symmetry-protected topological order for bosonic and fermionic systems with on-site symmetries". In: *Physical Review B* 95.23 (2017). ISSN: 2469-9969. DOI: 10.1103/physrevb.95.235140. URL: http://dx.doi.org/10.1103/PhysRevB.95.235140.
- [119] Robert Usher. Fermionic 6j-symbols in superfusion categories. 2016. arXiv: 1606. 03466 [math.QA].
- [120] David Aasen, Ethan Lake, and Kevin Walker. "Fermion condensation and super pivotal categories". In: J. Math. Phys. 60.12 (2019), p. 121901. DOI: 10.1063/1.5045669. arXiv: 1709.01941 [cond-mat.str-el].
- [121] Jiaqi Lou, Ce Shen, Chaoyi Chen, and Ling-Yan Hung. "A (Dummy's) Guide to Working with Gapped Boundaries via (Fermion) Condensation". In: (July 2020). arXiv: 2007.10562 [hep-th].
- [122] Miao Li and Ming Yu. "Braiding Matrices, Modular Transformations and Topological Field Theories in (2+1)-dimensions". In: Commun. Math. Phys. 127 (1990), p. 195. DOI: 10.1007/BF02096502.

- [123] Ryohei Kobayashi. "Anomaly constraint on chiral central charge of (2+1)d topological order". In: *Phys. Rev. Res.* 3.2 (2021), p. 023107. DOI: 10.1103/PhysRevResearch.3.
 023107. arXiv: 2101.01018 [cond-mat.str-el].
- [124] D. J. Gross and Frank Wilczek. "Asymptotically Free Gauge Theories I". In: Phys. Rev. D 8 (1973), pp. 3633–3652. DOI: 10.1103/PhysRevD.8.3633.
- [125] H. David Politzer. "Reliable Perturbative Results for Strong Interactions?" In: *Phys. Rev. Lett.* 30 (1973). Ed. by J. C. Taylor, pp. 1346–1349. DOI: 10.1103/PhysRevLett. 30.1346.
- [126] Thomas Appelquist, Daniel Nash, and L. C. R. Wijewardhana. "Critical Behavior in (2+1)-Dimensional QED". In: *Phys. Rev. Lett.* 60 (1988), p. 2575. DOI: 10.1103/ PhysRevLett.60.2575.
- [127] Vladimir Bashmakov, Jaume Gomis, Zohar Komargodski, and Adar Sharon. "Phases of $\mathcal{N} = 1$ theories in 2 + 1 dimensions". In: *JHEP* 07 (2018), p. 123. DOI: 10.1007/JHEP07(2018)123. arXiv: 1802.10130 [hep-th].
- [128] Francesco Benini and Sergio Benvenuti. " $\mathcal{N} = 1$ dualities in 2+1 dimensions". In: *JHEP* 11 (2018), p. 197. DOI: 10.1007/JHEP11(2018)197. arXiv: 1803.01784 [hep-th].
- [129] Changha Choi, Martin Roček, and Adar Sharon. "Dualities and Phases of 3DN = 1 SQCD". In: JHEP 10 (2018), p. 105. DOI: 10.1007/JHEP10(2018)105. arXiv: 1808.02184 [hep-th].
- [130] Changha Choi. "Phases of Two Adjoints QCD₃ And a Duality Chain". In: *JHEP* 04 (2020), p. 006. DOI: 10.1007/JHEP04(2020)006. arXiv: 1910.05402 [hep-th].
- [131] Edward Witten. " θ Vacua in Two-dimensional Quantum Chromodynamics". In: Nuovo Cim. A 51 (1979), p. 325. DOI: 10.1007/BF02776593.
- [132] Sidney R. Coleman. "More About the Massive Schwinger Model". In: Annals Phys. 101 (1976), p. 239. DOI: 10.1016/0003-4916(76)90280-3.
- [133] Matthew Yu. "Symmetries and Anomalies of (1+1)d Theories: 2-groups and Symmetry Fractionalization". In: (Oct. 2020). arXiv: 2010.01136 [hep-th].
- [134] Clay Córdova, Daniel S. Freed, Ho Tat Lam, and Nathan Seiberg. "Anomalies in the Space of Coupling Constants and Their Dynamical Applications I". In: SciPost Phys. 8.1 (2020), p. 001. DOI: 10.21468/SciPostPhys.8.1.001. arXiv: 1905.09315
 [hep-th].
- [135] Eric Sharpe. "Undoing decomposition". In: Int. J. Mod. Phys. A 34.35 (2020), p. 1950233. DOI: 10.1142/S0217751X19502336. arXiv: 1911.05080 [hep-th].
- [136] Zohar Komargodski, Kantaro Ohmori, Konstantinos Roumpedakis, and Sahand Seifnashri. "Symmetries and strings of adjoint QCD₂". In: JHEP 03 (2021), p. 103. DOI: 10.1007/JHEP03(2021)103. arXiv: 2008.07567 [hep-th].

- [137] David Kutasov. "Two-dimensional QCD coupled to adjoint matter and string theory".
 In: Nucl. Phys. B 414 (1994), pp. 33–52. DOI: 10.1016/0550-3213(94)90420-0. arXiv: hep-th/9306013.
- [138] N. D. Mermin and H. Wagner. "Absence of Ferromagnetism or Antiferromagnetism in One- or Two-Dimensional Isotropic Heisenberg Models". In: *Phys. Rev. Lett.* 17 (26 1966), pp. 1307–1307. DOI: 10.1103/PhysRevLett.17.1307.
- [139] Sidney R. Coleman. "There are no Goldstone bosons in two-dimensions". In: Commun. Math. Phys. 31 (1973), pp. 259–264. DOI: 10.1007/BF01646487.
- [140] Xie Chen, Zheng-Cheng Gu, and Xiao-Gang Wen. "Complete classification of onedimensional gapped quantum phases in interacting spin systems". In: *Phys. Rev. B* 84 (23 2011), p. 235128. DOI: 10.1103/PhysRevB.84.235128. URL: https: //link.aps.org/doi/10.1103/PhysRevB.84.235128.
- [141] I. Affleck. "On the Critical Behavior of Two-dimensional Systems With Continuous Symmetries". In: *Phys. Rev. Lett.* 55 (1985), p. 1355. DOI: 10.1103/PhysRevLett.55. 1355.
- [142] A. B. Zamolodchikov. "Irreversibility of the Flux of the Renormalization Group in a 2D Field Theory". In: JETP Lett. 43 (1986), pp. 730–732.
- [143] G. 't Hooft. "A two-dimensional model for mesons". In: Nuclear Physics B 75.3 (1974), pp. 461-470. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(74)90088-1.
- [144] Hans-Christian Pauli and Stanley J. Brodsky. "Solving field theory in one space and one time dimension". In: *Phys. Rev. D* 32 (8 1985), pp. 1993–2000. DOI: 10.1103/ PhysRevD.32.1993.
- [145] Hans-Christian Pauli and Stanley J. Brodsky. "Discretized light-cone quantization: Solution to a field theory in one space and one time dimension". In: *Phys. Rev. D* 32 (8 1985), pp. 2001–2013. DOI: 10.1103/PhysRevD.32.2001.
- Stanley J. Brodsky, Hans-Christian Pauli, and Stephen S. Pinsky. "Quantum chromodynamics and other field theories on the light cone". In: *Phys. Rept.* 301 (1998), pp. 299–486. DOI: 10.1016/S0370-1573(97)00089-6. arXiv: hep-ph/9705477.
- [147] Ross Dempsey, Igor R. Klebanov, and Silviu S. Pufu. "Exact Symmetries and Threshold States in Two-Dimensional Models for QCD". In: (Jan. 2021). arXiv: 2101.05432 [hep-th].
- [148] Wenjie Ji, Shu-Heng Shao, and Xiao-Gang Wen. "Topological Transition on the Conformal Manifold". In: *Phys. Rev. Res.* 2.3 (2020), p. 033317. DOI: 10.1103/ PhysRevResearch.2.033317. arXiv: 1909.01425 [cond-mat.str-el].

- [149] Hirotaka Sugawara. "A Field Theory of Currents". In: *Phys. Rev.* 170 (5 1968), pp. 1659–1662. DOI: 10.1103/PhysRev.170.1659.
- [150] P. Di Francesco, P. Mathieu, and D. Senechal. *Conformal Field Theory*. Graduate Texts in Contemporary Physics. New York: Springer-Verlag, 1997. ISBN: 978-0-387-94785-3, 978-1-4612-7475-9. DOI: 10.1007/978-1-4612-2256-9.
- [151] Dieter Zeppenfeld. "Two-dimensional QCD in the Temporal Gauge". In: Nucl. Phys. B 247 (1984), pp. 125–156. DOI: 10.1016/0550-3213(84)90376-6.
- [152] Edwin Langmann and Gordon W. Semenoff. "Massless QCD in (1+1)-dimensions and gauge covariant Sugawara construction". In: *Phys. Lett. B* 341 (1994), pp. 195–204.
 DOI: 10.1016/0370-2693(94)90310-7. arXiv: hep-th/9404159.
- P. Goddard and D. Olive. "Kac-Moody algebras, conformal symmetry and critical exponents". In: Nuclear Physics B 257 (1985), pp. 226-252. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(85)90344-X.
- P. Goddard, W. Nahm, and David I. Olive. "Symmetric Spaces, Sugawara's Energy Momentum Tensor in Two-Dimensions and Free Fermions". In: *Phys. Lett. B* 160 (1985), p. 111. DOI: 10.1016/0370-2693(85)91475-3.
- [155] Elie Cartan. "Sur une classe remarquable d'espaces de Riemann I & II". In: Bulletin de la Société Mathématique de France 54, 55 (1926-1927). DOI: 10.24033/bsmf.1105, 10.24033/bsmf.1113.
- [156] A. N. Schellekens and N. P. Warner. "Conformal subalgebras of Kac-Moody algebras". In: *Phys. Rev. D* 34 (10 1986), pp. 3092–3096. DOI: 10.1103/PhysRevD.34.3092.
- F. Alexander Bais and Peter G. Bouwknegt. "A classification of subgroup truncations of the bosonic string". In: Nuclear Physics B 279.3 (1987), pp. 561–570. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(87)90010-1.
- [158] R.C. Arcuri, J.F. Gomes, and D.I. Olive. "Conformal subalgebras and symmetric spaces". In: *Nuclear Physics B* 285 (1987), pp. 327–339. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(87)90342-7.
- [159] Mikhail Isachenkov, Ingo Kirsch, and Volker Schomerus. "Chiral Primaries in Strange Metals". In: Nucl. Phys. B 885 (2014), pp. 679–712. DOI: 10.1016/j.nuclphysb. 2014.06.004. arXiv: 1403.6857 [hep-th].
- [160] Daniel G. Robbins, Eric Sharpe, and Thomas Vandermeulen. "A generalization of decomposition in orbifolds". In: (Jan. 2021). arXiv: 2101.11619 [hep-th].
- [161] D. Robbins, E. Sharpe, and T. Vandermeulen. "Quantum symmetries in orbifolds and decomposition". In: (July 2021). arXiv: 2107.12386 [hep-th].
- [162] Doron Gepner. "Field Identification in Coset Conformal Field Theories". In: Phys. Lett. B 222 (1989), pp. 207–212. DOI: 10.1016/0370-2693(89)91253-7.

- [163] A. N. Schellekens and S. Yankielowicz. "Field Identification Fixed Points in the Coset Construction". In: Nucl. Phys. B 334 (1990), pp. 67–102. DOI: 10.1016/0550– 3213(90)90657-Y.
- [164] Kentaro Hori. "On global aspects of gauged Wess-Zumino-Witten model". Other thesis. Jan. 1994. arXiv: hep-th/9402019.
- [165] Ingo Runkel and Gérard M. T. Watts. "Fermionic CFTs and classifying algebras". In: JHEP 06 (2020), p. 025. DOI: 10.1007/JHEP06(2020)025. arXiv: 2001.05055
 [hep-th].
- [166] Chang-Tse Hsieh, Yu Nakayama, and Yuji Tachikawa. "On fermionic minimal models". In: (Feb. 2020). arXiv: 2002.12283 [cond-mat.str-el].
- [167] Justin Kulp. "Two More Fermionic Minimal Models". In: (Mar. 2020). arXiv: 2003. 04278 [hep-th].
- [168] Philip Boyle Smith. "Boundary States and Anomalous Symmetries of Fermionic Minimal Models". In: (Feb. 2021). arXiv: 2102.02203 [hep-th].
- Simon Dalley and Igor R. Klebanov. "String spectrum of (1+1)-dimensional large N QCD with adjoint matter". In: *Phys. Rev. D* 47 (1993), pp. 2517-2527. DOI: 10.1103/PhysRevD.47.2517. arXiv: hep-th/9209049.
- [170] Joshua Boorstein and David Kutasov. "Symmetries and mass splittings in QCD in two-dimensions coupled to adjoint fermions". In: Nucl. Phys. B 421 (1994), pp. 263– 277. DOI: 10.1016/0550-3213(94)90328-X. arXiv: hep-th/9401044.
- [171] David J. Gross, Akikazu Hashimoto, and Igor R. Klebanov. "The Spectrum of a large N gauge theory near transition from confinement to screening". In: *Phys. Rev. D* 57 (1998), pp. 6420–6428. DOI: 10.1103/PhysRevD.57.6420. arXiv: hep-th/9710240.
- [172] Emanuel Katz, Gustavo Marques Tavares, and Yiming Xu. "Solving 2D QCD with an adjoint fermion analytically". In: JHEP 05 (2014), p. 143. DOI: 10.1007/JHEP05(2014) 143. arXiv: 1308.4980 [hep-th].
- [173] Sergei Dubovsky. "A Simple Worldsheet Black Hole". In: JHEP 07 (2018), p. 011.
 DOI: 10.1007/JHEP07(2018)011. arXiv: 1803.00577 [hep-th].
- [174] Victor G Kač and Minoru Wakimoto. "Modular and conformal invariance constraints in representation theory of affine algebras". In: Advances in Mathematics 70.2 (1988), pp. 156–236. ISSN: 0001-8708. DOI: https://doi.org/10.1016/0001-8708(88) 90055-2.
- [175] Koji Hasegawa. "Spin Module Versions of Weyl's Reciprocity Theorem for Classical Kac-Moody Lie Algebras. An Application to Branching Rule Duality". In: *Publications* of the Research Institute for Mathematical Sciences 25.5 (1989), pp. 741–828. DOI: 10.2977/prims/1195172705.

- [176] S.G. Naculich, H.A. Riggs, and H.J. Schnitzer. "Group-level duality in WZW models and Chern-Simons theory". In: *Physics Letters B* 246.3 (1990), pp. 417–422. ISSN: 0370-2693. DOI: https://doi.org/10.1016/0370-2693(90)90623-E.
- [177] D. Verstegen. "Conformal embeddings, rank-level duality and exceptional modular invariants". In: Communications in Mathematical Physics 137.3 (1991), pp. 567 –586.
 DOI: cmp/1104202741.
- [178] Tomoki Nakanishi and Akihiro Tsuchiya. "Level-rank duality of WZW models in conformal field theory". In: *Communications in Mathematical Physics* 144.2 (1992), pp. 351 –372. DOI: cmp/1104249321.
- [179] F. Levstein and J. I. Liberati. "Branching rules for conformal embeddings". In: Communications in Mathematical Physics 173.1 (1995), pp. 1–16. DOI: cmp/1104274518.
- [180] Takuya Okuda, Koichi Saito, and Shuichi Yokoyama. "U(1) spin Chern-Simons theory and Arf invariants in two dimensions". In: Nucl. Phys. B 962 (2021), p. 115272. DOI: 10.1016/j.nuclphysb.2020.115272. arXiv: 2005.03203 [hep-th].
- [181] V. G. Kač. "Contravariant form for infinite-dimensional Lie algebras and superalgebras". In: *Group Theoretical Methods in Physics*. Ed. by Wolf Beiglböck, Arno Böhm, and E. Takasugi. Berlin, Heidelberg: Springer Berlin Heidelberg, 1979, pp. 441–445. ISBN: 978-3-540-35345-4.
- [182] A.A. Belavin, A.M. Polyakov, and A.B. Zamolodchikov. "Infinite conformal symmetry in two-dimensional quantum field theory". In: *Nuclear Physics B* 241.2 (1984), pp. 333– 380. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(84)90052-X.
- [183] Daniel Friedan, Zongan Qiu, and Stephen Shenker. "Superconformal invariance in two dimensions and the tricritical Ising model". In: *Physics Letters B* 151.1 (1985), pp. 37–43. ISSN: 0370-2693. DOI: https://doi.org/10.1016/0370-2693(85)90819-6.
- [184] J. L. Petersen. "Conformal and Superconformal two-dimensional field theories". In: 19th International Symposium: Special Topics in Gauge Field Theories. Dec. 1985.
- [185] P. Goddard, A. Kent, and David I. Olive. "Virasoro Algebras and Coset Space Models". In: *Phys. Lett. B* 152 (1985), pp. 88–92. DOI: 10.1016/0370-2693(85)91145-1.
- [186] Wayne Boucher, Daniel Friedan, and Adrian Kent. "Determinant Formulae and Unitarity for the N=2 Superconformal Algebras in Two-Dimensions or Exact Results on String Compactification". In: *Phys. Lett. B* 172 (1986), p. 316. DOI: 10.1016/0370-2693(86)90260-1.
- [187] P. Di Vecchia, J. L. Petersen, M. Yu, and H. B. Zheng. "Explicit Construction of Unitary Representations of the N=2 Superconformal Algebra". In: *Phys. Lett. B* 174 (1986), pp. 280–284. DOI: 10.1016/0370-2693(86)91099-3.

- [188] P. Christe and F. Ravanini. " $G_N \otimes G_L/G_{N+L}$ Conformal Field Theories and Their Modular Invariant Partition Functions". In: International Journal of Modern Physics A 4.4 (Jan. 1989), pp. 897–920. DOI: 10.1142/S0217751X89000418.
- [189] Jürgen Fuchs, Bert Schellekens, and Christoph Schweigert. "The resolution of field identification fixed points in diagonal coset theories". In: Nuclear Physics B 461.1 (1996), pp. 371–404. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(95)00623-0.
- [190] Yoichi Kazama and Hisao Suzuki. "Characterization of N=2 superconformal models generated by the coset space method". In: *Physics Letters B* 216.1 (1989), pp. 112–116. ISSN: 0370-2693. DOI: https://doi.org/10.1016/0370-2693(89)91378-6.
- [191] Mendel Nguyen, Yuya Tanizaki, and Mithat Ünsal. "Non-invertible 1-form symmetry and Casimir scaling in 2d Yang-Mills theory". In: (Apr. 2021). arXiv: 2104.01824 [hep-th].
- [192] Ignatios Antoniadis and Constantin Bachas. "Conformal Invariance and Parastatistics in Two-dimensions". In: Nucl. Phys. B 278 (1986), pp. 343–352. DOI: 10.1016/0550– 3213(86)90217–8.
- [193] Ian Affleck. "Exact Critical Exponents for Quantum Spin Chains, Nonlinear Sigma Models at Theta = pi and the Quantum Hall Effect". In: Nucl. Phys. B 265 (1986), pp. 409–447. DOI: 10.1016/0550-3213(86)90167-7.
- [194] P. Goddard, A. Kent, and David I. Olive. "Unitary Representations of the Virasoro and Supervirasoro Algebras". In: Commun. Math. Phys. 103 (1986), pp. 105–119. DOI: 10.1007/BF01464283.
- [195] Victor G Kač. "Infinite-dimensional Lie algebras and Dedekind's η-function". In: Functional Analysis and Its Applications 8 (1974), pp. 68–70. DOI: https://doi.org/ 10.1007/BF02028313.
- [196] Robert Feger, Thomas W. Kephart, and Robert J. Saskowski. "LieART 2.0 A Mathematica application for Lie Algebras and Representation Theory". In: Comput. Phys. Commun. 257 (2020), p. 107490. DOI: 10.1016/j.cpc.2020.107490. arXiv: 1912.10969 [hep-th].
- [197] Naoki Yamatsu. "Finite-Dimensional Lie Algebras and Their Representations for Unified Model Building". In: (Nov. 2015). arXiv: 1511.08771 [hep-ph].
- [198] Tatsuhiro Misumi, Yuya Tanizaki, and Mithat Ünsal. "Fractional θ angle, 't Hooft anomaly, and quantum instantons in charge-q multi-flavor Schwinger model". In: JHEP 07 (2019), p. 018. DOI: 10.1007/JHEP07(2019)018. arXiv: 1905.05781 [hep-th].

- [199] Nikhil Karthik and Rajamani Narayanan. "Bilinear condensate in three-dimensional large-N_c QCD". In: Phys. Rev. D94.4 (2016), p. 045020. DOI: 10.1103/PhysRevD.94. 045020. arXiv: 1607.03905 [hep-lat].
- [200] Nikhil Karthik and Rajamani Narayanan. "Scale-invariance and scale-breaking in parity-invariant three-dimensional QCD". In: *Phys. Rev.* D97.5 (2018), p. 054510. DOI: 10.1103/PhysRevD.97.054510. arXiv: 1801.02637 [hep-th].
- [201] A. J. Niemi and G. W. Semenoff. "Axial Anomaly Induced Fermion Fractionization and Effective Gauge Theory Actions in Odd Dimensional Space-Times". In: *Phys. Rev. Lett.* 51 (1983), p. 2077. DOI: 10.1103/PhysRevLett.51.2077.
- [202] A. N. Redlich. "Gauge Noninvariance and Parity Violation of Three-Dimensional Fermions". In: *Phys. Rev. Lett.* 52 (1984). [,364(1983)], p. 18. DOI: 10.1103/ PhysRevLett.52.18.
- [203] Kristan Jensen and Andreas Karch. "Bosonizing three-dimensional quiver gauge theories". In: JHEP 11 (2017), p. 018. DOI: 10.1007/JHEP11(2017)018. arXiv: 1709.01083 [hep-th].
- [204] Kyle Aitken, Andrew Baumgartner, and Andreas Karch. "Novel 3d bosonic dualities from bosonization and holography". In: JHEP 09 (2018), p. 003. DOI: 10.1007/JHEP09(2018)003. arXiv: 1807.01321 [hep-th].
- [205] Anton Kapustin and Nathan Seiberg. "Coupling a QFT to a TQFT and Duality".
 In: JHEP 04 (2014), p. 001. DOI: 10.1007/JHEP04(2014)001. arXiv: 1401.0740
 [hep-th].
- [206] Tomoki Nakanishi and Akihiro Tsuchiya. "Level rank duality of WZW models in conformal field theory". In: Commun. Math. Phys. 144 (1992), pp. 351–372. DOI: 10.1007/BF02101097.
- [207] Cumrun Vafa and Edward Witten. "Parity Conservation in QCD". In: Phys. Rev. Lett. 53 (1984), p. 535. DOI: 10.1103/PhysRevLett.53.535.
- [208] Ian Affleck, Jeffrey A. Harvey, and Edward Witten. "Instantons and (Super)Symmetry Breaking in (2+1)-Dimensions". In: Nucl. Phys. B206 (1982), pp. 413–439. DOI: 10.1016/0550-3213(82)90277-2.
- [209] Cyril Closset, Thomas T. Dumitrescu, Guido Festuccia, Zohar Komargodski, and Nathan Seiberg. "Comments on Chern-Simons Contact Terms in Three Dimensions". In: JHEP 09 (2012), p. 091. DOI: 10.1007/JHEP09(2012)091. arXiv: 1206.5218
 [hep-th].

- [210] Cyril Closset, Thomas T. Dumitrescu, Guido Festuccia, Zohar Komargodski, and Nathan Seiberg. "Contact Terms, Unitarity, and F-Maximization in Three-Dimensional Superconformal Theories". In: JHEP 10 (2012), p. 053. DOI: 10.1007/JHEP10(2012) 053. arXiv: 1205.4142 [hep-th].
- [211] Clay Córdova and Thomas T. Dumitrescu. "Candidate Phases for SU(2) Adjoint QCD₄ with Two Flavors from $\mathcal{N} = 2$ Supersymmetric Yang-Mills Theory". In: (2018). arXiv: 1806.09592 [hep-th].
- [212] Zhen Bi and T. Senthil. "An Adventure in Topological Phase Transitions in 3+1-D: Non-abelian Deconfined Quantum Criticalities and a Possible Duality". In: (2018). arXiv: 1808.07465 [cond-mat.str-el].
- [213] Mohamed M. Anber and Erich Poppitz. "Two-flavor adjoint QCD". In: *Phys. Rev.* D98.3 (2018), p. 034026. DOI: 10.1103/PhysRevD.98.034026. arXiv: 1805.12290
 [hep-th].
- [214] Dam Thanh Son. "Is the Composite Fermion a Dirac Particle?" In: *Phys. Rev.* X5.3 (2015), p. 031027. DOI: 10.1103/PhysRevX.5.031027. arXiv: 1502.03446
 [cond-mat.mes-hall].
- [215] Andreas Karch and David Tong. "Particle-Vortex Duality from 3d Bosonization". In: Phys. Rev. X6.3 (2016), p. 031043. DOI: 10.1103/PhysRevX.6.031043. arXiv: 1606.01893 [hep-th].
- [216] Nathan Seiberg, T. Senthil, Chong Wang, and Edward Witten. "A Duality Web in 2+1 Dimensions and Condensed Matter Physics". In: Annals Phys. 374 (2016), pp. 395–433.
 DOI: 10.1016/j.aop.2016.08.007. arXiv: 1606.01989 [hep-th].
- [217] Jeff Murugan and Horatiu Nastase. "Particle-vortex duality in topological insulators and superconductors". In: JHEP 05 (2017), p. 159. DOI: 10.1007/JHEP05(2017)159. arXiv: 1606.01912 [hep-th].
- [218] Vadim S. Kaplunovsky, Jacob Sonnenschein, and Shimon Yankielowicz. "Domain walls in supersymmetric Yang-Mills theories". In: *Nucl. Phys.* B552 (1999), pp. 209–245.
 DOI: 10.1016/S0550-3213(99)00203-5. arXiv: hep-th/9811195 [hep-th].
- [219] Zohar Komargodski, Adar Sharon, Ryan Thorngren, and Xinan Zhou. "Comments on Abelian Higgs Models and Persistent Order". In: (2017). arXiv: 1705.04786 [hep-th].
- [220] Zohar Komargodski, Tin Sulejmanpasic, and Mithat Ünsal. "Walls, anomalies, and deconfinement in quantum antiferromagnets". In: *Phys. Rev.* B97.5 (2018), p. 054418.
 DOI: 10.1103/PhysRevB.97.054418. arXiv: 1706.05731 [cond-mat.str-el].
- [221] Patrick Draper. "Domain Walls and the CP Anomaly in Softly Broken Supersymmetric QCD". In: Phys. Rev. D97.8 (2018), p. 085003. DOI: 10.1103/PhysRevD.97.085003. arXiv: 1801.05477 [hep-th].

- [222] Adam Ritz and Ashish Shukla. "Domain wall moduli in softly-broken SQCD at $\bar{\theta} = \pi$ ". In: *Phys. Rev.* D97.10 (2018), p. 105015. DOI: 10.1103/PhysRevD.97.105015. arXiv: 1804.01978 [hep-th].
- [223] Riccardo Argurio, Matteo Bertolini, Francesco Bigazzi, Aldo L. Cotrone, and Pierluigi Niro. "QCD domain walls, Chern-Simons theories and holography". In: JHEP 09 (2018), p. 090. DOI: 10.1007/JHEP09(2018)090. arXiv: 1806.08292 [hep-th].
- [224] Mohamed M. Anber and Erich Poppitz. "Anomaly matching, (axial) Schwinger models, and high-T super Yang-Mills domain walls". In: JHEP 09 (2018), p. 076. DOI: 10. 1007/JHEP09(2018)076. arXiv: 1807.00093 [hep-th].
- [225] Adi Armoni and Vasilis Niarchos. "QCD₃ with Two-Index Quarks, Mirror Symmetry and Fivebrane anti-BIons near Orientifolds". In: (2018). arXiv: 1808.07715 [hep-th].
- [226] Stefano Bolognesi. "Skyrmions in Orientifold and Adjoint QCD". In: (2009). arXiv: 0901.3796 [hep-th].
- [227] Sachin Jain, Shiraz Minwalla, and Shuichi Yokoyama. "Chern Simons duality with a fundamental boson and fermion". In: JHEP 11 (2013), p. 037. DOI: 10.1007/ JHEP11(2013)037. arXiv: 1305.7235 [hep-th].
- [228] Francesco Benini. "Three-dimensional dualities with bosons and fermions". In: JHEP 02 (2018), p. 068. DOI: 10.1007/JHEP02(2018)068. arXiv: 1712.00020 [hep-th].
- [229] Kristan Jensen. "A master bosonization duality". In: JHEP 01 (2018), p. 031. DOI: 10.1007/JHEP01(2018)031. arXiv: 1712.04933 [hep-th].
- [230] Kyle Aitken, Andreas Karch, and Brandon Robinson. "Master 3d Bosonization Duality with Boundaries". In: JHEP 05 (2018), p. 124. DOI: 10.1007/JHEP05(2018)124. arXiv: 1803.08507 [hep-th].
- [231] Kyle Aitken, Andrew Baumgartner, Changha Choi, and Andreas Karch. "Generalization of QCD₃ symmetry-breaking and flavored quiver dualities". In: *JHEP* 02 (2020), p. 060. DOI: 10.1007/JHEP02(2020)060. arXiv: 1906.08785 [hep-th].
- [232] Francesco Benini and Sergio Benvenuti. " $\mathcal{N} = 1$ QED in 2 + 1 dimensions: dualities and enhanced symmetries". In: *JHEP* 05 (2021), p. 176. DOI: 10.1007/JHEP05(2021)176. arXiv: 1804.05707 [hep-th].
- [233] Davide Gaiotto, Zohar Komargodski, and Jingxiang Wu. "Curious Aspects of Three-Dimensional $\mathcal{N} = 1$ SCFTs". In: *JHEP* 08 (2018), p. 004. DOI: 10.1007/JHEP08(2018) 004. arXiv: 1804.02018 [hep-th].
- [234] Antonio Amariti and Davide Forcella. "Spin(7) duality for $\mathcal{N} = 1$ CS-matter theories". In: *JHEP* 07 (2014), p. 082. DOI: 10.1007/JHEP07(2014)082. arXiv: 1404.4052 [hep-th].

- [235] Andreas Karch, Brandon Robinson, and David Tong. "More Abelian Dualities in 2+1 Dimensions". In: JHEP 01 (2017), p. 017. DOI: 10.1007/JHEP01(2017)017. arXiv: 1609.04012 [hep-th].
- [236] Vladimir Bashmakov and Nicola Gorini. "Phases of $\mathcal{N} = 1$ Quivers in 2+1 Dimensions". In: (Sept. 2021). arXiv: 2109.11862 [hep-th].
- [237] Clay Córdova, Po-Shen Hsin, and Kantaro Ohmori. "Exceptional Chern-Simons-Matter Dualities". In: (2018). arXiv: 1812.11705 [hep-th].
- [238] Ofer Aharony. "Baryons, monopoles and dualities in Chern-Simons-matter theories". In: JHEP 02 (2016), p. 093. DOI: 10.1007/JHEP02(2016)093. arXiv: 1512.00161 [hep-th].
- [239] Cumrun Vafa and Edward Witten. "Eigenvalue Inequalities for Fermions in Gauge Theories". In: Commun. Math. Phys. 95 (1984), p. 257. DOI: 10.1007/BF01212397.
- [240] X. G. Wen. "Quantum field theory of many-body systems: From the origin of sound to an origin of light and electrons". In: Oxford, UK: Univ. Pr. (2004) 505 p (2004).
- [241] Eduardo H. Fradkin. "Field Theories of Condensed Matter Physics". In: Front. Phys. 82 (2013), pp. 1–852.
- [242] Lukasz Fidkowski and Alexei Kitaev. "Topological phases of fermions in one dimension". In: *Phys. Rev. B* 83 (7 2011), p. 075103. DOI: 10.1103/PhysRevB.83.075103. URL: https://link.aps.org/doi/10.1103/PhysRevB.83.075103.
- [243] Ari M. Turner, Frank Pollmann, and Erez Berg. "Topological phases of one-dimensional fermions: An entanglement point of view". In: *Phys. Rev. B* 83 (7 2011), p. 075102.
 DOI: 10.1103/PhysRevB.83.075102. URL: https://link.aps.org/doi/10.1103/PhysRevB.83.075102.
- [244] Xie Chen, Zheng-Cheng Gu, and Xiao-Gang Wen. "Classification of gapped symmetric phases in one-dimensional spin systems". In: *Phys. Rev. B* 83 (3 2011), p. 035107. DOI: 10.1103/PhysRevB.83.035107. URL: https://link.aps.org/doi/10.1103/PhysRevB.83.035107.
- [245] Norbert Schuch, David Pérez-García, and Ignacio Cirac. "Classifying quantum phases using matrix product states and projected entangled pair states". In: *Phys. Rev. B* 84 (16 2011), p. 165139. DOI: 10.1103/PhysRevB.84.165139. URL: https://link.aps.org/doi/10.1103/PhysRevB.84.165139.
- [246] Andrew M. Essin and Michael Hermele. "Classifying fractionalization: Symmetry classification of gapped Z₂ spin liquids in two dimensions". In: *Phys. Rev. B* 87 (10 2013), p. 104406. DOI: 10.1103/PhysRevB.87.104406. URL: https://link.aps.org/doi/10.1103/PhysRevB.87.104406.

- [247] Yuan-Ming Lu and Ashvin Vishwanath. "Classification and properties of symmetryenriched topological phases: Chern-Simons approach with applications to Z₂ spin liquids". In: *Phys. Rev. B* 93 (15 2016), p. 155121. DOI: 10.1103/PhysRevB.93.155121. URL: https://link.aps.org/doi/10.1103/PhysRevB.93.155121.
- [248] Ling-Yan Hung and Xiao-Gang Wen. "Quantized topological terms in weak-coupling gauge theories with a global symmetry and their connection to symmetry-enriched topological phases". In: *Phys. Rev. B* 87 (16 2013), p. 165107. DOI: 10.1103/PhysRevB. 87.165107. URL: https://link.aps.org/doi/10.1103/PhysRevB.87.165107.
- [249] Maissam Barkeshli, Parsa Bonderson, Meng Cheng, and Zhenghan Wang. "Symmetry, Defects, and Gauging of Topological Phases". In: (2014). arXiv: 1410.4540 [cond-mat.str-el].
- [250] Meng Cheng, Michael Zaletel, Maissam Barkeshli, Ashvin Vishwanath, and Parsa Bonderson. "Translational Symmetry and Microscopic Constraints on Symmetry-Enriched Topological Phases: A View from the Surface". In: *Phys. Rev. X* 6 (4 2016), p. 041068. DOI: 10.1103/PhysRevX.6.041068.
- [251] Edward Witten. Unpublished, 2016.
- [252] Francesco Benini, Clay Córdova, and Po-Shen Hsin. "On 2-Group Global Symmetries and their Anomalies". In: JHEP 03 (2019), p. 118. DOI: 10.1007/JHEP03(2019)118. arXiv: 1803.09336 [hep-th].
- [253] Po-Shen Hsin, Ho Tat Lam, and Nathan Seiberg. "Comments on One-Form Global Symmetries and Their Gauging in 3d and 4d". In: (2018). arXiv: 1812.04716 [hep-th].
- [254] Anton Kapustin and Ryan Thorngren. "Higher symmetry and gapped phases of gauge theories". In: (2013). arXiv: 1309.4721 [hep-th].
- [255] Po-Shen Hsin. Unpublished, 2019.
- [256] Yasunori Lee and Yuji Tachikawa. "A study of time reversal symmetry of abelian anyons". In: *Journal of High Energy Physics* 2018.7 (2018), p. 90. ISSN: 1029-8479. DOI: 10.1007/JHEP07(2018)090.
- [257] Daniel S. Freed. "Anomalies and Invertible Field Theories". In: Proc. Symp. Pure Math. 88 (2014), pp. 25–46. DOI: 10.1090/pspum/088/01462. arXiv: 1404.7224 [hep-th].
- [258] Gregory W. Moore and Nathan Seiberg. "Lectures on RCFT". In: 1989 Banff NATO ASI: Physics, Geometry and Topology Banff, Canada, August 14-25, 1989. [,1(1989)].
 1989, pp. 1–129.
- [259] V. G. Turaev. "Quantum invariants of knots and three manifolds". In: De Gruyter Stud. Math. 18 (1994), pp. 1–588.

- [260] Alexei Kitaev. "Anyons in an exactly solved model and beyond". In: Annals of Physics 321.1 (2006). January Special Issue, pp. 2–111. ISSN: 0003-4916. DOI: https://doi. org/10.1016/j.aop.2005.10.005. arXiv: cond-mat/0506438 [cond-mat.mes-hall].
- [261] Parsa Bonderson, Chetan Nayak, and Xiao-Liang Qi. "A time-reversal invariant topological phase at the surface of a 3D topological insulator". In: *Journal of Statistical Mechanics: Theory and Experiment* 2013.09 (2013), P09016. DOI: 10.1088/1742-5468/2013/09/P09016.
- [262] Dmitriy Belov and Gregory W. Moore. "Classification of Abelian spin Chern-Simons theories". In: (2005). arXiv: hep-th/0505235 [hep-th].
- [263] Spencer D. Stirling. "Abelian Chern-Simons theory with toral gauge group, modular tensor categories, and group categories". PhD thesis. Texas U., Math Dept., 2008. arXiv: 0807.2857 [hep-th].
- [264] Anton Kapustin and Natalia Saulina. "Topological boundary conditions in abelian Chern-Simons theory". In: Nucl. Phys. B845 (2011), pp. 393–435. DOI: 10.1016/j. nuclphysb.2010.12.017. arXiv: 1008.0654 [hep-th].
- [265] V V Nikulin. "Integral Symmetric Bilinear Forms and some of their Applications". In: Mathematics of the USSR-Izvestiya 14.1 (1980), pp. 103–167. DOI: 10.1070/ im1980v014n01abeh001060.
- [266] Bill Semus and Sam Smith. On the Structure of the Automorphism Group of Some Finite Groups. http://people.sju.edu/~smith/Current_Courses/autG.pdf. 2016.
- [267] Jasha Sommer-Simpson. "Automorphism Groups For Semidirect Products of Cyclic Groups". In: (2013).
- [268] Ivan Niven, Hugh L Montgomery, and Herbert S Zuckerman. An introduction to the theory of numbers. English. 5th ed. New York: Wiley, 1991. ISBN: 0471625469.
- [269] Peter Stevenhagen. "A Density Conjecture for the Negative Pell Equation". In: Computational Algebra and Number Theory. Ed. by Wieb Bosma and Alf van der Poorten. Dordrecht: Springer Netherlands, 1995, pp. 187–200. ISBN: 978-94-017-1108-1. DOI: 10.1007/978-94-017-1108-1_13.
- [270] Wieb Bosma and Peter Stevenhagen. "Density computations for real quadratic units". In: Math. Comput. 65 (1996), pp. 1327–1337.
- [271] Richard A. Mollin. Fundamental Number Theory with Applications. 2nd. Chapman & Hall/CRC, 2008. ISBN: 1420066595, 9781420066593.
- [272] Parsa Bonderson, Kirill Shtengel, and J. K. Slingerland. "Interferometry of non-Abelian Anyons". In: Annals Phys. 323 (2008), pp. 2709–2755. DOI: 10.1016/j.aop.2008.01.
 012. arXiv: 0707.4206 [quant-ph].

- [273] Juan Martin Maldacena, Gregory W. Moore, and Nathan Seiberg. "D-brane charges in five-brane backgrounds". In: JHEP 10 (2001), p. 005. DOI: 10.1088/1126-6708/ 2001/10/005. arXiv: hep-th/0108152 [hep-th].
- [274] Yuan-Ming Lu and Ashvin Vishwanath. "Theory and classification of interacting 'integer' topological phases in two dimensions: A Chern-Simons approach". In: *Phys. Rev.* B86.12 (2012). [Erratum: Phys. Rev.B89,no.19,199903(2014)], p. 125119. DOI: 10.1103/PhysRevB.86.125119,10.1103/PhysRevB.89.199903. arXiv: 1205.3156 [cond-mat.str-el].
- [275] Jürgen Fuchs, Jan Priel, Christoph Schweigert, and Alessandro Valentino. "On the Brauer Groups of Symmetries of Abelian Dijkgraaf-Witten Theories". In: Commun. Math. Phys. 339.2 (2015), pp. 385–405. DOI: 10.1007/s00220-015-2420-y. arXiv: 1404.6646 [hep-th].
- [276] C.R. Leedham-Green, S. McKay, S. McKay, L.S.M.S.S. McKay, and London Mathematical Society. *The Structure of Groups of Prime Power Order*. London Mathematical Society monographs. Oxford University Press. ISBN: 9780198535485.
- [277] Saban Alaca and Kenneth S. Williams. Introductory Algebraic Number Theory. Cambridge University Press, 2003. DOI: 10.1017/CB09780511791260.
- [278] K. Hardy and K.S. Williams. "On the solvability of the diophantine equation $dV^2 2eVW dW^2 = 1$ ". In: *Pacific Journal of Mathematics* 124.1 (Jan. 1986), pp. 145–158.
- [279] H. F. Trotter. "On the Norms of Units in Quadratic Fields". In: Proceedings of the American Mathematical Society 22.1 (1969), pp. 198–201. ISSN: 00029939, 10886826.
 DOI: 10.2307/2036951.
- [280] G. H. Hardy and J. E. Littlewood. "Some problems of 'Partitio numerorum'; III: On the expression of a number as a sum of primes". In: Acta Math. 44 (1923), pp. 1–70. DOI: 10.1007/BF02403921.
- [281] Robert J. Lemke Oliver. "Almost-primes represented by quadratic polynomials". eng. In: Acta Arithmetica 151.3 (2012), pp. 241–261. URL: http://eudml.org/doc/279604.
- [282] Henryk Iwaniec. "Almost-primes represented by quadratic polynomials". In: Inventiones mathematicae 47.2 (1978), pp. 171–188. ISSN: 1432-1297. DOI: 10.1007/ BF01578070.
- Yu-An Chen and Anton Kapustin. "Bosonization in three spatial dimensions and a 2-form gauge theory". In: *Phys. Rev. B* 100.24 (2019), p. 245127. DOI: 10.1103/ PhysRevB.100.245127. arXiv: 1807.07081 [cond-mat.str-el].
- [284] A.N. Schellekens and S. Yankielowicz. "Simple Currents, Modular Invariants and Fixed Points". In: Int. J. Mod. Phys. A 5 (1990), pp. 2903–2952. DOI: 10.1142/ S0217751X90001367.

- [285] Gregory Moore and Nathan Seiberg. "Taming the conformal zoo". In: *Physics Letters B* 220.3 (1989), pp. 422 –430. ISSN: 0370-2693. DOI: https://doi.org/10.1016/0370-2693(89)90897-6.
- [286] C. Arf. "Untersuchungen über quadratische Formen in Körpern der Charakteristik
 2. (Teil I.)." In: Journal für die reine und angewandte Mathematik (Crelles Journal) 1941 (1941), pp. 148 –167.
- [287] Ying-Hsuan Lin and Shu-Heng Shao. "Duality Defect of the Monster CFT". In: J. Phys. A 54.6 (2021), p. 065201. DOI: 10.1088/1751-8121/abd69e. arXiv: 1911.00042 [hep-th].
- [288] Eddy Ardonne, Peter E. Finch, and Matthew Titsworth. *Classification of Metaplectic Fusion Categories*. 2016. arXiv: 1608.03762 [math.QA].
- [289] Marcos Mariño. "Chern-Simons theory and topological strings". In: *Rev. Mod. Phys.* 77 (2005), pp. 675–720. DOI: 10.1103/RevModPhys.77.675. arXiv: hep-th/0406005.
- [290] C. Blanchet and G. Masbaum. "Topological quantum field theories for surfaces with spin structure". In: *Duke Mathematical Journal* 82.2 (1996), pp. 229 -267. DOI: 10.1215/S0012-7094-96-08211-3. URL: https://doi.org/10.1215/S0012-7094-96-08211-3.
- [291] G. R. Dvali and Mikhail A. Shifman. "Domain walls in strongly coupled theories". In: *Phys. Lett.* B396 (1997). [Erratum: Phys. Lett.B407,452(1997)], pp. 64–69. DOI: 10.1016/S0370-2693(97)00808-3,10.1016/S0370-2693(97)00131-7. arXiv: hep-th/9612128 [hep-th].
- [292] Ofer Aharony, Nathan Seiberg, and Yuji Tachikawa. "Reading between the lines of four-dimensional gauge theories". In: JHEP 08 (2013), p. 115. DOI: 10.1007/ JHEP08(2013)115. arXiv: 1305.0318 [hep-th].
- [293] Yuji Tachikawa. "Magnetic discrete gauge field in the confining vacua and the supersymmetric index". In: JHEP 03 (2015), p. 035. DOI: 10.1007/JHEP03(2015)035. arXiv: 1412.2830 [hep-th].
- [294] Jaume Gomis, Zohar Komargodski, and Nathan Seiberg. Unpublished, 2017.
- [295] Vladimir Bashmakov, Francesco Benini, Sergio Benvenuti, and Matteo Bertolini.
 "Living on the walls of super-QCD". In: SciPost Phys. 6 (4 2019), p. 44. DOI: 10.
 21468/SciPostPhys.6.4.044.
- [296] Davide Gaiotto. "Kazama-Suzuki models and BPS domain wall junctions in N=1 SU(n) Super Yang-Mills". In: (2013). arXiv: 1306.5661 [hep-th].
- [297] Markus Dierigl and Alexander Pritzel. "Topological Model for Domain Walls in (Super-)Yang-Mills Theories". In: *Phys. Rev. D* 90.10 (2014), p. 105008. DOI: 10. 1103/PhysRevD.90.105008. arXiv: 1405.4291 [hep-th].

- [298] Adi Armoni, Amit Giveon, Dan Israel, and Vasilis Niarchos. "Brane Dynamics and 3D Seiberg Duality on the Domain Walls of 4D N=1 SYM". In: JHEP 07 (2009), p. 061. DOI: 10.1088/1126-6708/2009/07/061. arXiv: 0905.3195 [hep-th].
- [299] Martin Roček, Konstantinos Roumpedakis, and Sahand Seifnashri. "3D Dualities and Supersymmetry Enhancement from Domain Walls". In: JHEP 10 (2019), p. 097. DOI: 10.1007/JHEP10(2019)097. arXiv: 1904.02722 [hep-th].
- [300] Mohamed M. Anber and Erich Poppitz. "Domain walls in high-T SU(N) super Yang-Mills theory and QCD(adj)". In: JHEP 05 (2019), p. 151. DOI: 10.1007/JHEP05(2019) 151. arXiv: 1811.10642 [hep-th].
- [301] Juven Wang, Yi-Zhuang You, and Yunqin Zheng. "Gauge Enhanced Quantum Criticality and Time Reversal Domain Wall: SU(2) Yang-Mills Dynamics with Topological Terms". In: *Phys. Rev. Research.* 2 (2020), p. 013189. DOI: 10.1103/PhysRevResearch. 2.013189. arXiv: 1910.14664 [cond-mat.str-el].
- [302] Igor Bandos, Stefano Lanza, and Dmitri Sorokin. "Supermembranes and domain walls in $\mathcal{N} = 1$, D = 4 SYM". In: *JHEP* 12 (2019), p. 021. DOI: 10.1007/JHEP12(2019)021. arXiv: 1905.02743 [hep-th].
- [303] Edward Witten. "Supersymmetric index of three-dimensional gauge theory". In: (1999), pp. 156–184. DOI: 10.1142/9789812793850_0013. arXiv: hep-th/9903005 [hep-th].
- [304] Kyle Aitken, Aleksey Cherman, and Mithat Ünsal. "Dihedral symmetry in SU(N) Yang-Mills theory". In: Phys. Rev. D 100 (8 2019), p. 085004. DOI: 10.1103/PhysRevD. 100.085004.
- [305] Jurgen Fuchs, Bert Schellekens, and Christoph Schweigert. "From Dynkin diagram symmetries to fixed point structures". In: *Commun. Math. Phys.* 180 (1996), pp. 39–98.
 DOI: 10.1007/BF02101182. arXiv: hep-th/9506135 [hep-th].
- [306] E. Looijenga. "Root Systems and Elliptic Curves". In: Inventiones mathematicae 38 (1976), pp. 17–32. DOI: 10.1007/BF01390167.
- [307] Robert Friedman, John Morgan, and Edward Witten. "Vector bundles and F theory". In: Commun. Math. Phys. 187 (1997), pp. 679–743. DOI: 10.1007/s002200050154. arXiv: hep-th/9701162 [hep-th].
- [308] J. Fuchs. Affine Lie Algebras and Quantum Groups: An Introduction, with Applications in Conformal Field Theory. Cambridge Monographs on Mathematical Physics. Cambridge University Press, 1995. ISBN: 9780521484121.
- [309] Victor G. Kăc. Infinite-Dimensional Lie Algebras. Cambridge University Press, 1990. DOI: 10.1017/CB09780511626234.

- [310] D. Olive and Neil Turok. "The symmetries of Dynkin diagrams and the reduction of Toda field equations". In: Nuclear Physics B 215.4 (1983), pp. 470 –494. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(83)90256-0.
- [311] Denis Bernard. "String characters from Kac-Moody automorphisms". In: Nuclear Physics B 288 (1987), pp. 628 –648. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(87)90231-8.
- [312] C. Ahn and M. A. Walton. "Field identifications in coset conformal theories from projection matrices". In: *Phys. Rev. D* 41 (8 1990), pp. 2558–2562. DOI: 10.1103/ PhysRevD.41.2558.
- [313] Adam Ritz, Mikhail Shifman, and Arkady Vainshtein. "Counting domain walls in N=1 superYang-Mills". In: *Phys. Rev.* D66 (2002), p. 065015. DOI: 10.1103/PhysRevD.66. 065015. arXiv: hep-th/0205083 [hep-th].
- [314] Zheng-Cheng Gu, Zhenghan Wang, and Xiao-Gang Wen. "Lattice Model for Fermionic Toric Code". In: *Phys. Rev.* B90.8 (2014), p. 085140. DOI: 10.1103/PhysRevB.90. 085140. arXiv: 1309.7032 [cond-mat.str-el].
- [315] Anna Beliakova, Christian Blanchet, and Eva Contreras. "Spin Modular Categories". In: Quantum Topology 8 (Nov. 2014). DOI: 10.4171/QT/95. arXiv: 1411.4232 [math.GT].
- [316] F. A. Bais and J. K. Slingerland. "Condensate-induced transitions between topologically ordered phases". In: *Phys. Rev. B* 79 (4 2009), p. 045316. DOI: 10.1103/ PhysRevB.79.045316. arXiv: 0808.0627 [cond-mat.mes-hall].
- [317] I. S. Eliëns, J. C. Romers, and F. A. Bais. "Diagrammatics for Bose condensation in anyon theories". In: *Phys. Rev. B* 90 (19 2014), p. 195130. DOI: 10.1103/PhysRevB. 90.195130. arXiv: 1310.6001 [cond-mat.str-el].
- [318] Po-Shen Hsin and Shu-Heng Shao. "Lorentz Symmetry Fractionalization and Dualities in (2+1)d". In: SciPost Phys. 8 (2020), p. 018. DOI: 10.21468/SciPostPhys.8.2.018. arXiv: 1909.07383 [cond-mat.str-el].
- [319] Shmuel Elitzur, Gregory Moore, Adam Schwimmer, and Nathan Seiberg. "Remarks on the canonical quantization of the Chern-Simons-Witten theory". In: Nuclear Physics B 326.1 (1989), pp. 108 –134. ISSN: 0550-3213. DOI: https://doi.org/10.1016/0550-3213(89)90436-7.
- [320] Robbert Dijkgraaf, Cumrun Vafa, Erik P. Verlinde, and Herman L. Verlinde. "The Operator Algebra of Orbifold Models". In: Commun. Math. Phys. 123 (1989), p. 485. DOI: 10.1007/BF01238812.

- [321] A.Yu. Kitaev. "Fault-tolerant quantum computation by anyons". In: Annals of Physics 303.1 (2003), pp. 2 –30. ISSN: 0003-4916. DOI: https://doi.org/10.1016/S0003-4916(02)00018-0. arXiv: quant-ph/9707021 [quant-ph].
- [322] Michael R. Douglas. "Chern-Simons-Witten theory as a topological Fermi liquid". In: (Mar. 1994). arXiv: hep-th/9403119.
- [323] Antoine Van Proeyen. "Tools for supersymmetry". In: Ann. U. Craiova Phys. 9.I (1999), pp. 1–48. arXiv: hep-th/9910030.
- [324] nLab. Spin group. URL: https://ncatlab.org/nlab/show/spin+group.
- [325] Marc Henneaux and Claudio Teitelboim. *Quantization of Gauge Systems*. Princeton University Press.
- [326] P. A. M. Dirac. "Generalized Hamiltonian Dynamics". In: Canadian Journal of Mathematics 2 (1950), 129–148. DOI: 10.4153/CJM-1950-012-1.
- [327] W. Fulton and J. Harris. Representation Theory: A First Course. Graduate Texts in Mathematics. Springer New York, 1991. ISBN: 9780387974958.

Appendices.

Appendix A

Review and background.

In this appendix we review some basic facts about QFT that will be useful for the rest of this thesis.

A.1 Fermions in various dimensions.

We begin by collecting the properties of fermions in various dimensions. A useful reference is [323] (see also [324]).

In general dimension d = s + t, where s is the number of space dimensions and t is the number of time dimensions, the different fields are classified by the finite-dimensional irreducible representations of the Lorentz group Spin(s,t). Fermions are always assumed to transform according to the smallest spinorial representation, unless specified otherwise. Here, *spinorial* means that the representation is odd with respect to the central element $(-1)^F \in \text{Spin}(s,t)$, i.e., that it is charged under the \mathbb{Z}_2 subgroup that defines the extension $\text{SO}(s,t) \equiv \text{Spin}(s,t)/\mathbb{Z}_2$.

We take the mostly plus metric, with the first t components carrying -1 sign and the other s components a +1 sign. Classical fermions are valued in the algebra of Grassmann-odd numbers; consequently, quantum-mechanical fermions are operator fields that satisfy anti-commutation relations.

One dimension. In d = 0 + 1 (i.e., t = 1, s = 0) the Lorentz group is Spin $(1) = \mathbb{Z}_2$. There are two irreducible representations, both one-dimensional. The trivial representation classifies bosonic fields, and the sign representation fermionic fields. Therefore, fermions in d = 1 are described by one-dimensional spinors, $\psi(t)$. Lorentz transformations are just $(-1)^F: \psi(t) \mapsto -\psi(t)$, a map that commutes with the condition $\psi(t) = \psi(t)^*$. Therefore, the reality (Majorana) condition is compatible with Lorentz transformations, which means that spinors can be taken to be real.

In conclusion,

The minimal spinor in d = 1 + 0 is a real-valued one-dimensional field $\psi(t)$.

As Spin(1,0) is identical to Spin(0,1), a euclidean fermion $\psi(x)$ is formally identical to a Lorentzian one $\psi(t)$.

Two dimensions. In d = 1 + 1 (i.e., t = s = 1), the Lorentz group is Spin $(1, 1) \cong \mathbb{R}$, hence irreducible representations are one-dimensional. These representations are labelled by a *weight* $w \in \mathbb{R}$ such that, under a group element $\eta \in \mathbb{R}$, a field transforms as $\mathcal{O} \mapsto e^{w\eta}\mathcal{O}$. Here η is called the *rapidity* of the Lorentz transformation, and is related to the more common notation as $\cosh \eta \equiv 1/\sqrt{1-\beta^2}$, with β the speed.

Bosons are, by definition, fields with integral weight; and fermions are fields with halfintegral weight. The minimal half integers are $\pm 1/2$, hence we take fermions to transform as

$$\psi \mapsto e^{\pm \eta/2} \psi \tag{A.1.1}$$

If a spinor transforms with minus sign, it is said to be a *left-mover*, and if with plus sign, a *right-mover*.

Note that (A.1.1) commutes with complex conjugation, which means that fermions can be taken to be real.

In conclusion,

There are two types of minimal spinors in d = 1+1: left-moving fermions and right-moving fermions. They are both real-valued.

One can also derive these results via the standard gamma matrix approach (Clifford algebra). A basis of gamma matrices in d = 1 + 1 can be taken to be

$$\gamma^0 = i\sigma_y, \qquad \gamma^1 = \sigma_x \tag{A.1.2}$$

which satisfy $(\gamma^0)^2 = -1$, $(\gamma^1)^2 = +1$ and $\gamma^0 \gamma^1 + \gamma^1 \gamma^0 = 0$. These matrices act on twodimensional fermions as

$$\begin{pmatrix} \psi_1 \\ \psi_2 \end{pmatrix} \mapsto e^{-\frac{1}{4}\omega_{\mu\nu}\gamma^{\mu}\gamma^{\nu}} \begin{pmatrix} \psi_1 \\ \psi_2 \end{pmatrix}$$

$$= \begin{pmatrix} e^{-\omega/2} & 0 \\ 0 & e^{+\omega/2} \end{pmatrix} \begin{pmatrix} \psi_1 \\ \psi_2 \end{pmatrix}$$
(A.1.3)

Note that this transformation commutes with the "third" gamma matrix $\gamma_{\star} = \gamma_0 \gamma_1 \equiv \sigma_z$. Therefore, one may project to eigenspaces of γ_{\star} in a way that is consistent with Lorentz transformations. As $\gamma_{\star}^2 = +1$, the eigenvalues are ± 1 , and the reducibility condition reads $\gamma_{\star}\psi = \pm\psi$. A spinor with sign +1 is said to be a left-mover, and one with sign -1 a right-mover. We thus recover the previous classification under the renaming $\psi_1 \rightarrow \psi_L$, $\psi_2 \rightarrow \psi_R$, with $\omega \equiv \eta$. The euclidean Lorentz group d = 2 + 0 (i.e., t = 0, s = 2) is instead Spin(2) $\cong U(1)$. Representations are still labelled by a weight $w \in \mathbb{R}$, with transformation law $\mathcal{O} \mapsto e^{i\omega\eta}\mathcal{O}$. Left-movers are fields with w = -1/2, and right-movers fields with w = +1/2. Note that Lorentz transformations no longer commute with complex conjugation, and therefore these two types of fermions are now complex. Naturally, one can break up these fermions into their real and imaginary parts, and then the minimal fermions are real and two-dimensional.

The euclidean gamma matrices can be taken as

$$\gamma^0 = \sigma_y, \qquad \gamma^1 = \sigma_x \tag{A.1.4}$$

where now both matrices square to +1. A Lorentz transformation now becomes

$$\begin{pmatrix} \psi_1 \\ \psi_2 \end{pmatrix} \mapsto \begin{pmatrix} e^{i\omega/2} & 0 \\ 0 & e^{-i\omega/2} \end{pmatrix} \begin{pmatrix} \psi_1 \\ \psi_2 \end{pmatrix}$$
(A.1.5)

and the chirality matrix is $\gamma_* = i\gamma^0\gamma^1 = \sigma_z$. Hence, the irreducible spinors satisfy $\gamma_*\psi = \pm\psi$, and we recover the left- and right-movers above. Instead, one can work in a basis where the gamma matrices are real, for example

$$\gamma^0 = \sigma_x, \qquad \gamma^1 = \sigma_z \tag{A.1.6}$$

in which case Lorentz transformations read

$$\begin{pmatrix} \psi_1 \\ \psi_2 \end{pmatrix} \mapsto \begin{pmatrix} \cos \omega/2 & \sin \omega/2 \\ -\sin \omega/2 & \cos \omega/2 \end{pmatrix} \begin{pmatrix} \psi_1 \\ \psi_2 \end{pmatrix}$$
(A.1.7)

This now commutes with complex conjugation, and therefore the spinor can be taken to be real. But in this basis the chirality matrix is $\gamma_{\star} = \sigma_y$, whose eigenvectors are complex, and therefore one cannot impose the chirality condition $\gamma_{\star}\psi = \pm\psi$ at the same time as the reality condition $\psi = \psi^*$. Euclidean spinors in d = 2 can be taken to be either Weyl or Majorana, but not both.

Three dimensions. In d = 2 + 1 (i.e., t = 1, s = 2), the Lorentz group is Spin $(2, 1) \cong$ SL $(2, \mathbb{R})$, hence the irreducible representations are labelled by a half-integer $j \in \mathbb{Z}_{\geq 0}$, such that the dimension is 2j + 1. The minimal representation has j = 1/2, i.e., it is two-dimensional. Under an element $\begin{pmatrix} a & b \\ c & d \end{pmatrix} \in SL(2, \mathbb{R})$ (with ad - bc = 1), a spinor transforms as

$$\begin{pmatrix} \psi_1 \\ \psi_2 \end{pmatrix} \mapsto \begin{pmatrix} a & b \\ c & d \end{pmatrix} \begin{pmatrix} \psi_1 \\ \psi_2 \end{pmatrix}$$
(A.1.8)

Note that this is real, hence one can impose the reality condition $\psi = \psi^*$ consistently with Lorentz transformations.

In conclusion,

The minimal spinor in d = 2 + 1 has two components, and it is real-valued.

One can also establish these conclusions by looking at the gamma matrices. As d is odd, one can obtain a basis of gamma matrices by adjoining the chirality matrix γ_{\star} to the d = 1 + 1matrices, i.e., one can take

$$\gamma^0 = i\sigma_y, \qquad \gamma^1 = \sigma_x, \qquad \gamma^2 = \sigma_z$$
 (A.1.9)

(These matrices generate the algebra $\mathfrak{su}(1,1)$; this is nothing but the statement that $SL(2,\mathbb{R}) \cong$ SU(1,1).) A Lorentz transformation now reads

$$\begin{pmatrix} \psi_1 \\ \psi_2 \end{pmatrix} \mapsto e^{-\frac{1}{4}\omega_{\mu\nu}\gamma^{\mu}\gamma^{\nu}} \begin{pmatrix} \psi_1 \\ \psi_2 \end{pmatrix}$$

$$= \left[\cosh\Omega \mathbf{1}_2 + \frac{\sinh\Omega}{\Omega} (\omega_{2,0}\sigma_x + i\omega_{2,1}\sigma_y - \omega_{1,0}\sigma_z) \right] \begin{pmatrix} \psi_1 \\ \psi_2 \end{pmatrix}$$
(A.1.10)

where $\Omega := \sqrt{\omega_{1,0}^2 + \omega_{2,0}^2 - \omega_{2,1}^2}$. Note that this matrix has unit determinant and three free parameters $\omega_{1,0}, \omega_{2,0}, \omega_{2,1}$, hence it can be written as $\begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \mathrm{SL}(2,\mathbb{R})$, as above.

If we look instead at the euclidean Lorentz group Spin(3) = SU(2), the minimal spinor is the two-dimensional representation of SU(2), which is pseudo-real. One cannot impose a reality condition on the fermion (such a condition can be imposed on the direct sum of two fermions, hence the field effectively has four real degrees of freedom).

Four dimensions. In d = 3 + 1 (i.e., t = 1, d = 3), the Lorentz group is Spin $(3, 1) \cong$ SL $(2, \mathbb{C})$, hence the irreducible representations are labelled by two half-integers $(j_1, j_2) \in \mathbb{Z}^2_{\geq 0}$, whose dimension is $(2j_1 + 1)(2j_2 + 1)$. The minimal representations are $(j_1, j_2) = (1/2, 0)$ and $(j_1, j_2) = (0, 1/2)$, both two-dimensional. Under an element $X = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in SL(2, \mathbb{C})$ (with ad - bc = 1), the two irreducible spinors transform as

$$\psi \mapsto X\psi, \quad \text{or} \quad \psi \mapsto X^*\psi$$
 (A.1.11)

(The map $X \mapsto X^*$ is the unique outer automorphism of $SL(2, \mathbb{C})$; it interchanges $j_1 \leftrightarrow j_2$. It is more common to let the second action be $\psi \mapsto (X^{\dagger})^{-1}\psi$, which is an equivalent automorphism obtained by conjugating by the epsilon tensor, i.e., $\sigma_y X^* \sigma_y \equiv (X^{\dagger})^{-1}$.)

The matrix X is generically complex, and therefore Lorentz transformations do not commute with a reality condition $\psi^* = \psi$; the fermions are necessarily complex. That being said, note that if ψ transforms according to $\psi \mapsto X\psi$, then ψ^* transforms as $\psi^* \mapsto X^*\psi$. In this sense, we do not need to keep track of the two possibilities above; either can be obtained from the other by complex-conjugating the fermion. In particle-physics terminology, one can write down 3 + 1 dimensional theories using left-handed spinors alone.

In conclusion,

The minimal spinor in d = 3 + 1 has two components, and it is complex-valued.

Naturally, one can break such spinor into its real and imaginary parts, so the minimal spinor can also be taken to have four real-valued components.

In terms of the gamma matrix approach, this is reproduced as follows. The gamma matrices can be taken as

$$\gamma^{\mu} = \begin{pmatrix} 0 & \sigma^{\mu} \\ \bar{\sigma}^{\mu} & 0 \end{pmatrix} \tag{A.1.12}$$

where $\sigma^{\mu} = (\mathbf{1}_2, \vec{\sigma})$ and $\bar{\sigma}^{\mu} = (\mathbf{1}_2, -\vec{\sigma})$. The "fifth" gamma matrix is $\gamma_{\star} = -i\gamma^0\gamma^1\gamma^2\gamma^3 = \mathbf{1}_2 \oplus (-\mathbf{1}_2)$.

As $\gamma_{\star}^2 = \pm 1$, we can consider the eigenspaces $\gamma_{\star} = \pm 1$, which yield two-component complex fermions. Instead, if we change basis via the unitary matrix $\frac{1}{2}(\sigma_z \otimes \sigma_z - i\sigma_z \otimes \sigma_x - i\sigma_x \otimes \sigma_x - \sigma_x \otimes \sigma_z)$, the gamma matrices become purely imaginary, whence Lorentz transformations are real, and the Majorana condition $\psi^* = \psi$ can be imposed.

If we look instead at euclidean spinors (i.e., t = 0, s = 4), the Lorentz group becomes $\text{Spin}(4) \cong \text{SU}(2) \times \text{SU}(2)$. This group is just two copies of the Lorentz group in d = 3 + 0 dimensions, so the properties of spinors are immediate.

A.2 Fermion kinetic term.

Here we discuss a few important properties of the fermion kinetic term

$$\mathcal{L} = i\bar{\psi}\gamma^{\mu}\partial_{\mu}\psi \tag{A.2.1}$$

where $\bar{\psi} = \psi^{\dagger} \gamma^{0}$. We only consider Lorentzian fermions here.

The gamma matrices satisfy $\bar{\gamma}^{\mu} = \gamma^{\mu}$ (i.e., the time-like component is anti-hermitian and the space-like components are hermitian); therefore, $\bar{\psi}\gamma^{\mu}\partial_{\mu}\psi$ is anti-hermitian and the inclusion of *i* renders the Lagrangian real-valued, as it should be.

Note that in d = 0 + 1 and d = 3 + 1 we could write the gamma matrices as purely imaginary, while in d = 1 + 1 and d = 2 + 1 we could find matrices that were purely real. In general we write $\gamma^{\mu*} = \eta \gamma^{\mu}$, where $\eta = \pm 1$ depends on the number of spacetime dimensions. (This identity is valid in a suitable basis where γ^{μ} is either real or imaginary; for general bases the condition is replaced by $\gamma^{\mu*} = \eta B \gamma^{\mu} B^{-1}$ for a certain matrix B [323].)

Time-reversal. A symmetry of (A.2.1) that shall play a special role in this work is time-reversal. Such transformation takes $t \to -t$, hence it is anti-hermitian (as *iHt* must be invariant). The most general anti-linear transformation is

$$\mathsf{T} \colon \psi(t) \mapsto M\psi(-t) \tag{A.2.2}$$

for some unitary matrix M. Performing this transformation in the fermion Lagrangian yields the condition $M^{\dagger}\gamma^{0*}\gamma^{\mu*}M = \gamma^{\mu}\gamma^{0}$. In the Majorana basis where $\gamma^{\mu*} = \eta\gamma^{\mu}$ this becomes $M^{\dagger}\gamma^{0}\gamma^{\mu}M = \gamma^{\mu}\gamma^{0}$, from where it follows that $\gamma^{0}M$ commutes with $\gamma^{0}\gamma^{\mu}$. In odd dimensions this requires $\gamma^{0}M$ to be proportional to the unit matrix, while in even dimensions it must be a linear combination of the identity matrix and the chirality matrix γ_{\star} .

Multiplication by the chirality matrix constitutes an independent, unitary symmetry by itself, so if we count unitary symmetries separately, we do not need to count both the unit matrix and the chirality matrix in T; either of them can be obtained from the other by composing with the unitary symmetry generated by the chirality matrix. Therefore, for even and odd d, the basic time-reversal transformation is $M \propto \gamma^0$. Finally, in order to preserve the reality properties of ψ one must have $M^* = M$, i.e., $M = \pm \eta^{1/2} \gamma^0$. Again, multiplication by -1 is nothing but the unitary symmetry $(-1)^F : \psi \mapsto -\psi$, so if we count this separately we can take without loss of generality, say, the plus sign. In conclusion,

Time-reversal symmetry acts on fermions as

$$\mathsf{T} \colon \psi(t) \mapsto \eta^{1/2} \gamma^0 \psi(-t) \tag{A.2.3}$$

Note that $\mathsf{T}^2 = \gamma^{0*} \gamma^0 \equiv -\eta$. Therefore, if $\eta = 1$ this becomes $\mathsf{T}^2 = (-1)^F$ while if $\eta = -1$ it reads $\mathsf{T}^2 = 1$.

Of course, if the spinors carry internal indices (either flavor or gauge), then the most general transformation involves an orthogonal transformation on those indices.

Commutation relations. Another piece of important information contained in the fermion kinetic term is the canonical anti-commutation relations. In fact, these relations depend only on the time-derivatives, so it is enough to look at the first term only:

$$\mathcal{L} = i\psi^{\dagger}\partial_t\psi + \cdots \tag{A.2.4}$$

As far as the canonical formalism is concerned, we can focus without loss of generalization on the one-dimensional case d = 1; the generalization to higher d just requires adding the spatial dependence \vec{x} throughout, and suitable spinorial indices. Finally, by breaking complex fermions into their real and imaginary parts, we can take all fermions to be real, $\psi^* = \psi$. Real fields are often rescaled as $\psi \to \psi/\sqrt{2}$ so that the kinetic term carries the standard factor of 1/2 in front. With this in mind, the Lagrangian reads

$$\mathcal{L} = \frac{i}{2}\psi\partial_t\psi \tag{A.2.5}$$

The conjugate momentum is

$$\pi = \frac{\partial \mathcal{L}}{\partial \dot{\psi}} = \frac{i}{2}\psi \tag{A.2.6}$$

(Here we use a right-derivative.) As we cannot use this relation to solve for $\dot{\psi}$, it is to be regarded as a constraint, [325]

$$\chi = \pi - \frac{i}{2}\psi \tag{A.2.7}$$

which must vanish on-shell, $\chi \approx 0$.

Note that the Poisson bracket of the constraints is $\{\chi, \chi\} = -i$, and hence the Dirac bracket [326] is

$$\{\psi,\psi\}_D \equiv \{\psi,\psi\} - \{\psi,\chi\}\frac{1}{-i}\{\chi,\psi\} \equiv -i$$
 (A.2.8)

The quantum theory is obtained by promoting the Dirac bracket into an anti-commutator,

$$\{\psi,\psi\} \to i\{\psi,\psi\}_D \equiv 1 \tag{A.2.9}$$

The generalization to higher dimensions is straightforward:

The canonical anti-commutation relations for free fermions is

$$\{\psi_{\alpha,i}(t,\vec{x}),\psi_{\beta,j}(t,\vec{y})\} = \delta_{\alpha,\beta}\delta_{ij}\delta(\vec{x}-\vec{y})$$
(A.2.10)

where α, β are spinor indices and i, j are flavor indices. Complex fields can be written as the sum of two real fields, $\Psi = \frac{1}{\sqrt{2}}(\psi_1 + i\psi_2)$, hence their canonical anti-commutators read

$$\{\Psi_{\alpha,i}(t,\vec{x}),\Psi_{\beta,j}(t,\vec{y})\} = 0, \qquad \{\Psi_{\alpha,i}(t,\vec{x}),\Psi_{\beta,j}^{*}(t,\vec{y})\} = \delta_{\alpha,\beta}\delta_{ij}\delta(\vec{x}-\vec{y})$$
(A.2.11)

Canonical quantization. Here we describe the quantization of a system described by real fermions ψ_i subject to the canonical anti-commutation relations

$$\{\psi_i, \psi_j\} = \delta_{ij} \tag{A.2.12}$$

where i, j = 1, 2, ..., n. We can think of the fields ψ_i as describing actual 0 + 1 dimensional fermions, or the space of zero-modes for higher-dimensional fermions on a compact spatial manifold. This is also the rank-*n* real Clifford algebra so the fermions can also be regarded as the gamma matrices in *n* euclidean dimensions.

We begin by considering the case n = 1. Here the algebra is $\psi^2 = 1/2$, which has two one-dimensional representations, namely

$$\psi = \pm 1/\sqrt{2} \tag{A.2.13}$$

They are clearly inequivalent, and any other representation is a direct sum of these. Note also that neither of these two representations admits an action of $(-1)^F : \psi \mapsto -\psi$. The operator $(-1)^F$ should be represented as a matrix that anti-commutes with ψ , which is impossible

for 1×1 representations. This means that the vector space that realizes either of the 1×1 dimensional representations of the canonical algebra is not a \mathbb{Z}_2 -graded vector space; the system does not admit a good notion of fermion parity. One could consider, instead, the direct sum of these two representations, which now does admit an action of $(-1)^F$, but the corresponding vector space is not an irreducible module for the canonical algebra.

Consider now the n = 2 case, with algebra $\psi_1^2 = \psi_2^2 = 1/2$ and $\psi_1\psi_2 + \psi_2\psi_1 = 0$. It is clear that there are no 1×1 representations, but there is an obvious candidate for a 2×2 representation, namely

$$\psi_1 = \frac{1}{\sqrt{2}}\sigma_x, \qquad \psi_2 = \frac{1}{\sqrt{2}}\sigma_y \tag{A.2.14}$$

These are hermitian matrices that square to 1/2 and anti-commute with each other. There are more options, such as using σ_z instead of σ_x , but it is not hard to convince oneself that any other representation is equivalent to the one above. Furthermore, this representation admits an action of $(-1)^F$, namely $(-1)^F = \sigma_z$.

We can now tackle the general case. As the fermions are all independent, the Hilbert space of n + m fermions is the tensor product of the Hilbert space of n fermions and the Hilbert space of m fermions. Hence, given a representation of n fermions, we can obtain one for n + 2 fermions as follows. The first n matrices can be taken as

$$\psi_i^{(n+2)} = \psi_i^{(n)} \otimes A, \qquad i = 1, 2, \dots, n$$
 (A.2.15)

while the last two as

$$\psi_{n+1}^{(n+2)} = \mathbf{1} \otimes B, \qquad \psi_{n+2}^{(n+2)} = \mathbf{1} \otimes C \tag{A.2.16}$$

for some matrices A, B, C that realize a representation of the n = 2 case, e.g., $A = \sigma_z$, $B = \frac{1}{\sqrt{2}}\sigma_x$, $C = \frac{1}{\sqrt{2}}\sigma_y$. It is easily checked that the matrices $\psi_i^{(n+2)}$ satisfy the canonical anti-commutation relations if $\psi_i^{(n)}$ do. Finally, given a representation for $(-1)^F$ on the *n* fermions, one can obtain one for the n + 2 fermions via

$$(-1)^{F}|^{(n+2)} = (-1)^{F}|^{(n)} \otimes D$$
 (A.2.17)

for some matrix D. Requiring that $(-1)^F$ anti-commutes with all $\psi^{(n+2)}$ and squares to +1, we get the conditions

$$D^2 = 1,$$
 $[A, D] = 0,$ $\{D, B\} = \{D, C\} = 0$ (A.2.18)

with solution $D = \sigma_z$. Note that A, B, C, D are 2×2 matrices, which means that going from n to n + 2 doubles the number of states; and, as $D = \sigma_z$ has one +1 eigenvalue and one -1 eigenvalue, $n \to n + 2$ adds as many bosons as it adds fermions.

The conclusion is that, if a suitable representation of n fermions exists, then there is also one for n + 2 fermions, whose dimension is twice as large. The converse is obviously also true, just by looking at a subset of the operators. By counting degrees of freedom it is not hard to convince oneself that these are irreducible representations, and with a little bit more work one can show that the construction exhausts all the representations, up to unitary equivalence. So to summarize,

- The real Clifford algebra of odd rank has two inequivalent irreducible representations, of dimension $2^{(n-1)/2}$. Neither of these representations admits an action of $(-1)^F$ on it, so it is not a \mathbb{Z}_2 -graded vector space. The direct sum of these two representations does admit an action of $(-1)^F$, but the module does not furnish an irreducible representation of the algebra.
- The real Clifford algebra of even rank has a unique irreducible representation, of dimension 2^{n/2}. This representation admits an action of (−1)^F on it, so the module is a Z₂-graded vector space. The module contains 2^{n/2−1} even states (bosons) and 2^{n/2−1} odd states (fermions).

In either case, a possible choice for the representation matrices is

$$\psi^{(1)} = \pm \frac{1}{\sqrt{2}} \qquad \psi_1^{(2)} = \frac{1}{\sqrt{2}} \sigma_x \qquad \psi_i^{(n+2)} = \psi_i^{(n)} \otimes \sigma_z, \qquad i \le n$$
$$\psi_2^{(2)} = \frac{1}{\sqrt{2}} \sigma_y \qquad \psi_{n+1}^{(n+2)} = \frac{1}{\sqrt{2}} \mathbf{1} \otimes \sigma_x \qquad (A.2.19)$$
$$\psi_{n+2}^{(n+2)} = \frac{1}{\sqrt{2}} \mathbf{1} \otimes \sigma_y$$

with $(-1)^F \big|^{(n+2)} = (-1)^F \big|^{(n)} \otimes \sigma_z$, if *n* is even.

It is useful to rederive these conditions from a different point of view. When the number of fermions is even, one can break them up into two groups and use them to define complex fermions, say

$$\Psi_i := \frac{1}{\sqrt{2}} (\psi_{2i-1} + i\psi_{2i}), \qquad i = 1, 2, \dots, n/2$$
(A.2.20)

whose commutators are

$$\{\Psi_i, \Psi_j\} = 0, \qquad \{\Psi_i, \Psi_j^*\} = \delta_{ij}$$
 (A.2.21)

Note that now $\Psi_i^2 = 0$, which means that it is consistent to define the vacuum $|0\rangle$ as the state annihilated by all the fermions:

$$\Psi_i|0\rangle = 0 \tag{A.2.22}$$

The rest of states are now defined by acting with Ψ_i^* on $|0\rangle$. In particular, given an arbitrary state $|v\rangle$ created by these *n* fermions, the space of states for n+2 fermions is either of the form $|v\rangle$ or $\Psi_{n/2+1}^*|v\rangle$, which explains why the dimension is doubled when going from *n* to n+2, and why half of the new states are bosons and the other half are fermions. Furthermore, these two sets of states are identical as far as the first *n* fermions is concerned, except that $\Psi_{n/2+1}^*$

anti-commutes with such fermions, which explains the general structure $\psi_i^{(n+2)} = \psi_i^{(n)} \otimes \sigma_z$; similarly, given that the last two fermions can be written as $\psi_{n+1} = \frac{1}{\sqrt{2}}(\Psi_{n/2+1} + \Psi_{n/2+1}^*)$ and $\psi_{n+2} = \frac{1}{\sqrt{2}}(\Psi_{n/2+1} - \Psi_{n/2+1}^*)$, this also explains the appearance of the Pauli matrices σ_x, σ_y in $\psi_{n+1}^{(n+2)} = \frac{1}{\sqrt{2}} \mathbf{1} \otimes \sigma_x$ and $\psi_{n+2}^{(n+2)} = \frac{1}{\sqrt{2}} \mathbf{1} \otimes \sigma_y$. Finally, these considerations also explain why it is impossible to define a suitable Hilbert space for odd n: the complex fields Ψ are only defined when n is even.

One final perspective that is also quite illuminating is to try and solve the system via the path integral. Generically, a path integral on a time-circle with radius β and anti-periodic boundary conditions around it computes the trace $\operatorname{tr}(e^{-\beta H})$, while the same path integral with periodic boundary conditions computes $\operatorname{tr}((-1)^F e^{-\beta H})$. In our system the Hamiltonian vanishes and therefore these path integrals compute the total number of states in the Hilbert space, and the number of bosons minus the number of fermions, respectively.

The path integral is easily evaluated:

$$\operatorname{Pf}(i\partial_t)^n = \prod_{\lambda} \lambda^n \tag{A.2.23}$$

where λ are the eigenvalues of $i\partial_t$ with the appropriate boundary conditions. In particular, $\lambda = k + 1/2$ or $\lambda = k$, with $k \in \mathbb{Z}_{\geq 0}$, if the boundary conditions are anti-periodic or periodic, respectively. For periodic boundary conditions the product vanishes due to the zero mode k = 0. For anti-periodic boundary conditions the product can be computed using zeta-regularization,

$$\prod_{\lambda} \lambda^{n} := e^{-n\zeta_{i\partial_{t}}^{\prime}(0)}, \qquad \zeta_{i\partial_{t}}(s) := \sum_{\lambda} \lambda^{-s}$$
(A.2.24)

where in our case $\zeta_{i\partial_t}(s) = \sum_{k\geq 0} (k+1/2)^{-s} \equiv (2^s-1)\zeta(s)$, whence $\zeta'_{i\partial_t}(0) = -\frac{1}{2}\log 2$. Collecting results, we conclude that

$$\operatorname{tr}(1) = 2^{n/2}, \qquad \operatorname{tr}(-1)^F = 0$$
 (A.2.25)

This is consistent with our previous observations. The Hilbert space is only well-defined when n is even, in which case the total number of states is $2^{n/2}$, and half of these are bosons and the other half fermions. For odd n the would-be dimension $2^{n/2}$ is fractional.

In order to get a non-zero result for the periodic case we must insert fermions to compensate for the zero-modes. In particular, the (unnormalized) correlation function

$$\langle \psi_1 \psi_2 \cdots \psi_n \rangle$$
 (A.2.26)

is non-vanishing. For *n* even this is a bosonic operator, so a non-zero vacuum expectation value is perfectly fine; but for *n* odd this is a fermion, so its vacuum expectation value is inconsistent with the symmetry $(-1)^F$, under which it is charged, which means that this symmetry is not realized on the Hilbert space.

Fermion Hilbert space. We just reviewed the construction of the Hilbert space for a system of n free d = 1 Majorana fermions. One thing we noticed was that, when n is odd, this Hilbert space is somewhat ill-defined. Here we would like to be more explicit about what exactly goes wrong. Note that this is a question of practical interest, since given any system in d dimensions that has non-trivial fermionic zero-modes – and these are surprisingly common – the full Hilbert space will contain a subsector isomorphic d = 1 free fermions, and hence the lack of Hilbert space is inherited to the full theory.

The general conclusion from the previous discussion was that, when n is odd, there are two inequivalent irreducible representations of the canonical algebra, and neither admits an action of $(-1)^F$, i.e., an operator that anti-commutes with all the fermions. The direct sum of these two representations does admit an action, but this representation is no longer irreducible. This is not a good thing, for several reasons. The one that is most pertinent given the theme of this thesis is the fact that an action on the canonical operators does not induce a unique action on the Hilbert space. In other words, if we define some operation via its action on the fields, $\mathcal{O}\psi\mathcal{O}^{-1} := \psi'$, and the Hilbert space realizes an irreducible representation of the canonical algebra, the action of \mathcal{O} on the Hilbert space is uniquely specified by this information, while if the representation is reducible, the action of \mathcal{O} on it is not unique.

This is easily illustrated by considering the action of time-reversal, which is a recurring symmetry in this thesis, so it is useful to use it again here. Time-reversal is defined by its action on the fermions as $\mathsf{T}\psi(t)\mathsf{T}^{-1} := \psi(-t)$ (cf. (A.2.3)). When n is even, this fixes an action of T on the Hilbert space uniquely, up to an irrelevant global phase. For example, for n = 2 we showed that the irreducible representation could be taken as $\psi_1 = \sigma_x/\sqrt{2}$ and $\psi_2 = \sigma_y/\sqrt{2}$. Time-reversal should act as a certain 2×2 matrix τ which, by definition, satisfies $\tau\psi_i = \psi_i^*\tau$, which has solution $\tau \propto \sigma_x$, which is unique (up to an overall phase). On the other hand, when n is odd, this definition of time-reversal does not specify an operator on the Hilbert space. For example, when n = 1, the irreducible representations of the Hilbert space were $\psi = \pm 1/\sqrt{2}$, and the reducible representation that admits an action of $(-1)^F$ was the direct sum of these, namely $\psi = \sigma_z/\sqrt{2}$. Time-reversal should satisfy $\tau\psi = \psi^*\tau$, which now has general solution $\tau = \text{diag}(z_1, z_2)$, where z_i are two arbitrary phases. Here we see that the operation is not uniquely defined (and the undetermined parameters are not just a global phase).

Extrapolating to the general case, an action on the algebra of operators induces a unique action on the Hilbert space if and only if such space realizes an irreducible representation of the canonical commutation relations. In the case of d = 1 fermions (or zero-modes for general d), the symmetries have a unique action on the Hilbert space if and only if the number of fermions is even. When this number is odd, the action depends on several arbitrary choices.

That being said, there are some aspects of the action of T (or other operations in the general case) that are in fact specified, even if the representation is reducible. For example, in the n = 1 case above, we saw that $\tau = \text{diag}(z_1, z_2)$, where z_i are two arbitrary phases.

Note that $\mathsf{T}^2 = \tau \tau^* \equiv 1$, and therefore time-reversal squares to +1 for any choice of z_i . So even if the action of T is not uniquely defined, its square is. On the other hand, this system has $(-1)^F = \sigma_x$, and one can check that, if we choose $z_1 = \pm z_2$, then $(-1)^F \mathsf{T} = \pm \mathsf{T}(-1)^F$. Therefore, whether time-reversal commutes or anti-commutes with fermion parity can be fixed by an arbitrary choice. For arbitrary odd n, the pattern is

where $[A, B]_{\pm} = AB \pm BA$. For a given n, the sign on the second column can be chosen arbitrarily, but this choice then fixes the sign in the third column (i.e., the two signs in a given row are not independent). A common choice is to let T and $(-1)^F$ always commute, in which case one gets $T^2 = +1$ for $n = 1, 7 \mod 8$, and $T^2 = -1$ for $n = 3, 5 \mod 8$ (see e.g. [242]). Another useful choice (which we implicitly make in chapter 1) is $T(-1)^F = (-1)^{(n-1)/2}(-1)^F T$, in which case $T^2 = (-1)^{(n-1)(n-3)/8}$; this is useful because, in this case, the second column measures the second binary digit of n and the third column its first binary digit. In other words, given $n = n_0 + 2n_1 + 4n_2 \mod 8$, the digit n_0 tells us whether $(-1)^F$ exists; n_1 tells us whether time-reversal commutes or anti-commutes with $(-1)^F$; and n_2 whether time-reversal squares to +1 or -1.

Appendix B Useful data for Lie groups and representations.

In this appendix we collect some useful results concerning simple Lie groups and their representations. These groups arise as either the symmetry group or gauge group of a certain theory, and the representation specifies how it acts on the fields. Specifically, given a group G and a representation R, the group element $g \in G$ acts on a field ψ as $\psi \mapsto R(g)\psi$, where we think of ψ as a column vector and R(g) as a matrix.

A very useful physics-oriented reference is [150]. A more technical but still readable source is [327]. For quick computations one can use e.g. the Mathematica package LieART [196] or simply look at the extensive tables in [197].

It is a key result of Lie-representation theory that an arbitrary irreducible finite-dimensional representation can be labelled via rank(G) non-negative integers, known as the *Dynkin labels* of the representation. These integers $(\lambda_1, \lambda_2, \ldots, \lambda_{\operatorname{rank}(G)})$ are the coordinates of the highest weight of R with respect to the basis of fundamental weights. One can also label the representation via a Young diagram $Y = Y(\lambda)$, which is a diagram that has λ_i columns of i boxes, aligned on top and ordered by column height from left (largest column) to right (smallest column). For example the labels $\lambda = (2, 0, 1)$ yield a diagram with two columns with a single box, and one column with three boxes, i.e., $Y = \prod_{i=1}^{n} N_i$.

Given a representation R, two useful group-theoretic objects are its dimension dim(R)and its Dynkin index T(R). The dimension is just the size of the matrix R(g), i.e., the number of components of ψ . The Dynkin index is defined as $\delta^2 \operatorname{tr} R(g) = T(R)\delta^2 \operatorname{tr} g$, where δ denotes the differential at the origin g = 1, and tr the matrix trace. In general it will be useful to count real fields, so dim(R) and T(R) are rescaled by a factor of 2 if the field ψ is complex.

To be concrete we shall mostly concentrate on G = SU(N), SO(N), Sp(N), although we will make some remarks about other groups below. As these groups admit a natural definition in terms of matrices, they have an obvious representation, viz. R(g) = g, the fundamental

representation. There is also the *anti-fundamental* representation $R(g) = g^*$, although it is often useful to think of "anti" fields as being row vectors instead of column, i.e., fundamental indices are upper indices and anti-fundamental indices are lower indices; in this convention, the anti-fundamental representation acts from the right and it reads $R(g) = g^{\dagger}$.

The representation R(g) = g is clearly faithful, hence any representation of G = SU(N), SO(N), Sp(N) is contained in the tensor product of some number r_1 of fundamentals, and r_2 anti-fundamentals. The number $r = r_1 + r_2$ is the *rank* of the representation. In other words, an arbitrary representation is of the form

$$R(g): \psi^{i_{1}i_{2}\dots i_{r_{1}}}{}_{i'_{1}i'_{2}\dots i'_{r_{2}}} \mapsto g^{i_{1}}{}_{j_{1}} g^{i_{2}}{}_{j_{2}}\cdots g^{i_{r_{1}}}{}_{j_{r_{1}}}$$

$$\times (g^{\dagger})^{j'_{1}}{}_{i'_{1}} (g^{\dagger})^{j'_{2}}{}_{i'_{2}}\cdots (g^{\dagger})^{j'_{r_{2}}}{}_{i'_{r_{2}}}$$

$$\times \psi^{j_{1}j_{2}\dots j_{r_{1}}}{}_{j'_{1}j'_{2}\dots j'_{r_{2}}}$$
(B.0.1)

where $g \in G$. (By contrast, the fundamental representation of Spin(N) is not faithful and thus it does not generate the set of spinorial representations; in general the lack of faithfulness of a given representation is conveniently captured by its charge under the center of the group.)

In the case of SO(N) and Sp(N) we have the invariant symbols δ^{ij} and Ω^{ij} which allow us to raise and lower indices, which means that we can take without loss of generality $r_2 = 0$, so that all indices are upper indices. The group SU(N) has invariant symbol δ^i_j , which does not raise or lower indices, so fundamental and anti-fundamental indices are truly distinct. That being said, this group also has invariant symbol $\epsilon^{i_1i_2\cdots i_N}$, whence we can exchange ranti-symmetrized fundamental indices for N - r anti-symmetrized anti-fundamental ones. In other words, we can replace in its Young diagram any column with $N + \ell$ boxes for one with ℓ boxes. It is notationally convenient to extend this to negative ℓ so, for example, a column of N - 1 boxes is equivalent to one with "-1 boxes", which are drawn below the Young diagram. Specifically, we draw the usual Young diagram for the upper indices, and an inverted diagram for the lower indices, such that for example the representation $\psi^i_{i'}$ is drawn as \square , with the "anti-box" on the left standing for a column with N - 1 regular boxes.

A given representation can be either complex or self-conjugate. The latter can be further subdivided into real and pseudo-real representations. We denote the field over which the representation is defined as \mathbb{K} . Complex, real, and pseudo-real representations have $\mathbb{K} = \mathbb{C}, \mathbb{R}, \mathbb{H}$, respectively. Note that complex representations occur only when the Dynkin diagram of G has a \mathbb{Z}_2 reflection symmetry, i.e., A_n and D_n (and also E_6). This outer automorphism acts on the Dynkin labels of the representation as it acts on the nodes of the diagram.

In order for R to be irreducible we must impose suitable (anti)symmetrization on all the indices. Finally, we must subtract traces using invariant symbols, if possible. We next summarize the basic properties of all the representations of SU(N), SO(N), Sp(N) up to rank-4.

B.1 Summary of representations.

SU(N): For a given representation with Dynkin labels $(\lambda_1, \lambda_2, \ldots, \lambda_{N-1})$, the complex conjugate representation is the representation with reversed Dynkin labels: $(\lambda_{N-1}, \lambda_{N-2}, \ldots, \lambda_1)$. The representation is self-conjugate if and only if it is equal to its conjugate representation (i.e., if the Dynkin labels form a palindrome). A representation has the same dimension and index as its conjugate. A self-conjugate representation is pseudo-real if $N = 2 \mod 4$ and $\lambda_{N/2}$ is odd; otherwise it is real. (In what follows K denotes the "generic case", i.e., valid for most N; for some specific N a representation labelled as \mathbb{C} may in fact be \mathbb{R} or \mathbb{H} , such as e.g. the rank-r anti-symmetric for N = 2r).

The center of the group is $Z(SU(n)) = \mathbb{Z}_N$, and it acts on an arbitrary representation as $\psi \mapsto \mu^{r_1-r_2}\psi$, which can also be written in terms of Dynkin labels as $r_1 - r_2 = \lambda_1 + 2\lambda_2 + 3\lambda_3 + \cdots + (N-1)\lambda_{N-1} \mod N$; this number also agrees with the number of boxes minus the number of anti-boxes.

Rank 1: The representation is the fundamental representation:

Y	λ	$\dim_{\mathbb{R}}(\lambda)$	$T_{\mathbb{R}}(\lambda)$	\mathbb{K}
	$(1,0,\ldots,0)$	2N	1	\mathbb{C}

Under $\mathfrak{su}_{A+B} \to \mathfrak{su}_A \oplus \mathfrak{su}_B$, this representation decomposes as

$$\Box \mapsto \Box \otimes \bullet + \bullet \otimes \Box \tag{B.1.1}$$

Rank 2: We have the symmetric and anti-symmetric, which satisfy $\psi^{ij} = +\psi^{ji}$ and $\psi^{ij} = -\psi^{ji}$:

Y	λ	$\dim_{\mathbb{R}}(\lambda)$	$T_{\mathbb{R}}(\lambda)$	\mathbb{K}
	$(2, 0, 0, \ldots, 0)$	N(N+1)	N+2	\mathbb{C}
\square	$(0,1,0,\ldots,0)$	N(N-1)	N-2	\mathbb{C}

We also have a traceless representation, namely the adjoint, which satisfies $(\psi_j^i)^{\dagger} = \psi_i^j$ and $\psi_i^i = 0$:

$$Y \qquad \lambda \qquad \dim_{\mathbb{R}}(\lambda) \quad T_{\mathbb{R}}(\lambda) \quad \mathbb{K}$$

$$\Box \qquad (1, 0, \dots, 0, 1) \qquad N^2 - 1 \qquad N \qquad \mathbb{R}$$
Under $\mathfrak{su}_{A+B} \to \mathfrak{su}_A \oplus \mathfrak{su}_B$, these representations decompose as

$$\Box \mapsto \Box \otimes \bullet + \Box \otimes \Box + \bullet \otimes \Box$$

$$\Box \mapsto \Box \otimes \bullet + \Box \otimes \Box + \bullet \otimes \Box$$

$$\Box \mapsto \Box \otimes \bullet + \Box \otimes \Box + \bullet \otimes \Box$$

$$\Box \mapsto \Box \otimes \bullet + \overline{\Box} \otimes \Box + \bullet \otimes \bullet + \Box \otimes \overline{\Box} + \bullet \otimes \Box$$
(B.1.2)

Rank 3: We have the symmetric, anti-symmetric and mixed. The first one is completely symmetric with respect to its three indices, the second one completely anti-symmetric, and the third one satisfies $\psi^{ijk} = \psi^{jik}$ and $\psi^{ijk} + \psi^{kji} + \psi^{ikj} = 0$:

Y	λ	$\dim_{\mathbb{R}}(\lambda)$	$T_{\mathbb{R}}(\lambda)$	\mathbb{K}
	$(3, 0, 0, 0, \dots, 0)$	$\frac{1}{3}N(N+1)(N+2)$	$\frac{1}{2}(N+2)(N+3)$	\mathbb{C}
	$(0, 0, 1, 0, \dots, 0)$	$\frac{1}{3}N(N-1)(N-2)$	$\frac{1}{2}(N-2)(N-3)$	\mathbb{C}
₽	$(1, 1, 0, 0, \dots, 0)$	$\frac{2}{3}N(N+1)(N-1)$	$N^{2} - 3$	\mathbb{C}

We also have two traceless representations:

Y

$$\lambda$$
 $\dim_{\mathbb{R}}(\lambda)$
 $T_{\mathbb{R}}(\lambda)$
 \mathbb{K}
 \square
 (2,0,0,...,0,1)
 $N(N-1)(N+2)$
 $\frac{1}{2}(N+2)(3N-1)$
 \mathbb{C}
 \square
 (0,1,0,...,0,1)
 $N(N-2)(N+1)$
 $\frac{1}{2}(N-2)(3N+1)$
 \mathbb{C}

These satisfy $\psi^{ij}{}_k = +\psi^{ji}{}_k$ and $\psi^{ij}{}_k = -\psi^{ji}{}_k$, respectively, together with $\psi^{ij}{}_j = 0$.

Under $\mathfrak{su}_{A+B} \to \mathfrak{su}_A \oplus \mathfrak{su}_B$, these representations decompose as

Rank 4: The representations are as follows:

Y	λ	$\dim_{\mathbb{R}}(\lambda)$	$T_{\mathbb{R}}(\lambda)$	\mathbb{K}
	$(4, 0, 0, 0, 0, \dots, 0)$	$\frac{1}{12}N(N+1)(N+2)(N+3)$	$\frac{1}{6}(N+2)(N+3)(N+4)$	\mathbb{C}
	$(2, 1, 0, 0, 0, \dots, 0)$	$\frac{1}{4}N(N-1)(N+1)(N+2)$	$\frac{1}{2}(N+2)(N^2+N-4)$	\mathbb{C}
	$(1, 0, 1, 0, 0, \dots, 0)$	$\frac{1}{4}N(N-2)(N-1)(N+1)$	$\frac{1}{2}(N-2)(N^2-N-4)$	\mathbb{C}
	$(0, 2, 0, 0, 0, \dots, 0)$	$\frac{1}{6}N^2(N-1)(N+1)$	$\frac{1}{3}N(N-2)(N+2)$	\mathbb{C}
	$(0, 0, 0, 1, 0, \dots, 0)$	$\frac{1}{12}N(N-1)(N-2)(N-3)$	$\frac{1}{6}(N-4)(N-3)(N-2)$	\mathbb{C}

which satisfy

$$\begin{split} \psi^{ijk\ell} &= \psi^{(ijk\ell)} \\ \psi^{ijk\ell} &= \psi^{(ijk)\ell}, \quad \psi^{ijk\ell} + \psi^{\ell jki} + \psi^{i\ell kj} + \psi^{ij\ell k} = 0 \\ \psi^{ijk\ell} &= \psi^{[ijk]\ell}, \quad \psi^{ijk\ell} - \psi^{\ell jki} - \psi^{i\ell kj} - \psi^{ij\ell k} = 0 \\ \psi^{ijk\ell} &= \psi^{[ij]k\ell} = \psi^{ij[k\ell]} = \psi^{k\ell ij}, \quad \psi^{ijk\ell} + \psi^{ik\ell j} + \psi^{i\ell jk} = 0 \\ \psi^{ijk\ell} &= \psi^{[ijk\ell]} \end{split}$$
(B.1.4)

respectively. We also have representations with one trace removed:

which satisfy the symmetry properties of the corresponding rank-3 representation (such as e.g. $\psi^{ijk}_{\ell} = \psi^{(ijk)}_{\ell}$ and $\psi^{ijk}_{i} = 0$ for the first representations), and representations with two traces removed:

Y	λ	$\dim_{\mathbb{R}}(\lambda)$	$T_{\mathbb{R}}(\lambda)$	\mathbb{K}
	$(0, 1, 0, \dots, 0, 1, 0)$	$\frac{1}{4}N^2(N-3)(N+1)$	$N^{2}(N-3)$	\mathbb{R}
	$(2, 0, 0, \dots, 0, 1, 0)$	$\frac{1}{2}(N-2)(N-1)(N+1)(N+2)$	2N(N-2)(N+2)	\mathbb{C}
	$(0, 1, 0, \dots, 0, 0, 2)$	$\frac{1}{2}(N-2)(N-1)(N+1)(N+2)$	2N(N-2)(N+2)	\mathbb{C}
	$(2, 0, 0, \dots, 0, 0, 2)$	$\frac{1}{4}N^2(N-1)(N+3)$	$N^{2}(N+3)$	\mathbb{R}

which satisfy the symmetry properties of the corresponding pair of rank-2 representations, together with the appropriate hermiticity condition, if real. (For example, the first representation satisfies $\psi^{ij}_{k\ell} = \psi^{[ij]}_{k\ell} = \psi^{ij}_{[k\ell]}$ and $\psi^{ij}_{i\ell} = 0$, together with $(\psi^{ij}_{k\ell})^{\dagger} = \psi^{k\ell}_{ij})$.



SO(N): Consider a representation $(\lambda_1, \lambda_2, \ldots, \lambda_{\lfloor N/2 \rfloor})$ of Spin(N). It is self-conjugate except if N = 4n + 2, in which case its conjugate is obtained by permuting the last two labels: $(\lambda_1, \lambda_2, \ldots, \lambda_{2n-1}, \lambda_{2n+1}, \lambda_{2n})$. A self-conjugate representation is pseudo-real if N = 3, 5 mod 8 and $\lambda_{(N-1)/2}$ is odd, or if $N = 2 \mod 8$ and $\lambda_{(N/2)-1} + \lambda_{N/2}$ is odd; otherwise it is real. A representation of Spin(N) descends to a representation of SO(N) if and only if N is odd and $\lambda_{(N-1)/2}$ is even, or if N is even and $\lambda_{(N/2)-1} + \lambda_{N/2}$ is even.

The symmetry properties of the tensors is the same as in the SU(N) case, except that we subtract a trace for every pair of symmetrized indices. All the representations are real. Also, SO(N) has no center if N is odd, and a \mathbb{Z}_2 center if N is even. In the latter case it acts as $\psi \mapsto (-1)^r \psi$; this number also agrees with the number of boxes.

Rank 1: The representation is the fundamental representation:

Y	λ	$\dim_{\mathbb{R}}(\lambda)$	$T_{\mathbb{R}}(\lambda)$	\mathbb{K}
	$(1,0,\ldots,0)$	N	1	\mathbb{R}

Rank 2: We have the symmetric and anti-symmetric, which satisfy $\psi^{ij} = +\psi^{ji}$ and $\psi^{ij} = -\psi^{ji}$, respectively. Also, the symmetric is traceless: $\psi^{ij}\delta_{ij} = 0$. With this, the representations are as follows:

Y	λ	$\dim_{\mathbb{R}}(\lambda)$	$T_{\mathbb{R}}(\lambda)$	\mathbb{K}
	$(2, 0, 0, \dots, 0)$	$\frac{1}{2}(N-1)(N+2)$	N+2	\mathbb{R}
\square	$(0,1,0,\ldots,0)$	$\frac{1}{2}N(N-1)$	N-2	\mathbb{R}

Rank 3: We have the symmetric and anti-symmetric, and mixed:

Y	λ	$\dim_{\mathbb{R}}(\lambda)$	$T_{\mathbb{R}}(\lambda)$	K
	$(3, 0, 0, 0, \dots, 0)$	$\frac{1}{6}N(N-1)(N+4)$	$\frac{1}{2}(N+1)(N+4)$	\mathbb{R}
₽	$(1, 1, 0, 0, \dots, 0)$	$\frac{1}{3}N(N-2)(N+2)$	(N-2)(N+2)	\mathbb{R}
	$(0, 0, 1, 0, \dots, 0)$	$\frac{1}{6}N(N-1)(N-2)$	$\frac{1}{2}(N-2)(N-3)$	$\mathbb R$

which satisfy the same symmetry properties as the SU(N) representations, but we also impose tracelessness for every pair of symmetrized indices (e.g., the first representation satisfies $\psi^{ijk} = \psi^{(ijk)}$ and $\psi^{ijk} \delta_{ij} = 0$).

Rank 4: We have

Y	λ	$\dim_{\mathbb{R}}(\lambda)$	$T_{\mathbb{R}}(\lambda)$	\mathbb{K}
	$(4, 0, 0, 0, 0, \dots, 0)$	$\frac{1}{24}N(N-1)(N+1)(N+6)$	$\frac{1}{6}(N+1)(N+2)(N+6)$	\mathbb{R}
$\square \square$	$(2, 1, 0, 0, 0, \dots, 0)$	$\frac{1}{8}(N-1)(N-2)(N+1)(N+4)$	$\frac{1}{2}(N-2)(N+1)(N+4)$	\mathbb{R}
	$(1, 0, 1, 0, 0, \dots, 0)$	$\frac{1}{8}N(N-1)(N-3)(N+2)$	$\frac{1}{2}(N-2)(N-3)(N+2)$	\mathbb{R}
	$(0, 2, 0, 0, 0, \dots, 0)$	$\frac{1}{12}N(N-3)(N+1)(N+2)$	$\frac{1}{3}(N-3)(N+1)(N+2)$	\mathbb{R}
	$(0, 0, 0, 1, 0, \dots, 0)$	$\frac{1}{24}N(N-1)(N-2)(N-3)$	$\frac{1}{6}(N-2)(N-3)(N-4)$	\mathbb{R}

which again satisfy the same symmetry properties of the corresponding SU(N) representation, with traces removed.

The representations decompose under $\mathfrak{so}_{A+B} \to \mathfrak{so}_A \oplus \mathfrak{so}_B$ the same as in \mathfrak{su}_N , with the following extra representations:

$$\Box \mapsto \dots + \bullet \otimes \bullet$$

$$\Box \mapsto \dots + \Box \otimes \bullet + \bullet \otimes \Box$$

$$\Box \mapsto \dots + \Box \otimes \bullet + \bullet \otimes \Box$$

$$\Box \mapsto \dots + \Box \otimes \bullet + \Box \otimes \Box + \bullet \otimes \bullet + \bullet \otimes \Box$$

$$\Box \mapsto \dots + \Box \otimes \bullet + \Box \otimes \Box + \bullet \otimes \Box$$

$$+ \Box \otimes \bullet + \Box \otimes \Box + \bullet \otimes \Box$$

$$+ \Box \otimes \bullet + \Box \otimes \Box + \bullet \otimes \Box$$

$$\Box \mapsto \dots + \Box \otimes \bullet + \Box \otimes \Box + \bullet \otimes \Box$$

$$\Box \mapsto \dots + \Box \otimes \bullet + \Box \otimes \Box + \bullet \otimes \Box$$

$$\Box \mapsto \dots + \Box \otimes \bullet + \Box \otimes \Box + \bullet \otimes \Box$$

Sp(*N*): All the representations $(\lambda_1, \lambda_2, ..., \lambda_N)$ are self-conjugate. A given representation is pseudo-real if $\lambda_1 + \lambda_3 + \lambda_5 + ...$ is odd, and real otherwise; equivalently, it is pseudo-real if its rank is odd, and real otherwise. The representations are as follows:

Rank 1: The representation is the fundamental representation:

$$\begin{array}{cccc} Y & \lambda & \dim_{\mathbb{R}}(\lambda) & T_{\mathbb{R}}(\lambda) & \mathbb{K} \\ \\ \hline \Box & (1,0,\ldots,0) & 2N & 1 & \mathbb{H} \end{array}$$

Rank 2: We have the symmetric and anti-symmetric. The latter is Ω -traceless: $\psi^{ij}\Omega_{ij} = 0$. Thus,

Y	λ	$\dim_{\mathbb{R}}(\lambda)$	$T_{\mathbb{R}}(\lambda)$	K
	$(2, 0, 0, \ldots, 0)$	N(2N+1)	N+1	\mathbb{R}
\square	$(0,1,0,\ldots,0)$	(N-1)(2N+1)	N-1	\mathbb{R}

Rank 3: We have the symmetric, anti-symmetric, and mixed:

Y	λ	$\dim_{\mathbb{R}}(\lambda)$	$T_{\mathbb{R}}(\lambda)$	\mathbb{K}
	$(3, 0, 0, 0, \dots, 0)$	$\frac{2}{3}N(N+1)(2N+1)$	(N+1)(2N+3)	\mathbb{H}
₽	$(1, 1, 0, 0, \dots, 0)$	$\frac{8}{3}N(N-1)(N+1)$	4(N-1)(N+1)	\mathbb{H}
\square	$(0, 0, 1, 0, \dots, 0)$	$\frac{2}{3}N(N-2)(2N+1)$	(N-2)(2N-1)	\mathbb{H}

which have the same symmetry properties as the SU(N) representations, but with a tracelessness condition on every pair of anti-symmetrized indices (e.g., the last representation satisfies $\psi^{ijk} = \psi^{[ijk]}$ and $\psi^{ijk}\Omega_{ij} = 0$).

Rank 4:

Y	λ	$\dim_{\mathbb{R}}(\lambda)$	$T_{\mathbb{R}}(\lambda)$	\mathbb{K}
	$(4, 0, 0, 0, 0, \dots, 0)$	$\frac{1}{6}N(N+1)(2N+1)(2N+3)$	$\frac{1}{3}(N+1)(N+2)(2N+3)$	\mathbb{R}
	$(2, 1, 0, 0, 0, \dots, 0)$	$\frac{1}{2}N(N-1)(2N+1)(2N+3)$	(N-1)(N+1)(2N+3)	\mathbb{R}
	$(1, 0, 1, 0, 0, \dots, 0)$	$\frac{1}{2}(N-2)(N+1)(2N-1)(2N+1)$	(N-2)(N+1)(2N-1)	\mathbb{R}
	$(0, 2, 0, 0, 0, \dots, 0)$	$\frac{1}{3}N(N-1)(2N-1)(2N+3)$	$\frac{1}{3}(N-1)(2N-1)(2N+3)$	\mathbb{R}
	$(0, 0, 0, 1, 0, \dots, 0)$	$\frac{1}{6}N(N-3)(2N-1)(2N+1)$	$\frac{1}{3}(N-3)(N-1)(2N-1)$	\mathbb{R}

where again the symmetry properties are identical to those of the SU(N) representations, with traced removed as appropriate.

The representations decompose under $\mathfrak{sp}_{A+B} \to \mathfrak{sp}_A \oplus \mathfrak{sp}_B$ the same as in \mathfrak{su}_N , with the following extra representations:

B.2 Group characters.

In this section we write the characters of the basic representations of SU(N), SO(N), Sp(N). The group characters are defined as

$$\chi_R(g) \equiv \operatorname{tr}_R(g) := \operatorname{tr} R(g) \tag{B.2.1}$$

where in the last expression tr denotes the regular matrix trace. Note that, by definition, characters are class functions, i.e., they only depend on the conjugacy class of g. In other words, we can assume without loss of generality that g is inside a maximal torus of G, which in practice essentially means that it is diagonal. Hence, $\chi_R(g)$ can be expressed as a certain polynomial (of degree r) in the eigenvalues of g.

The most fundamental character is the character of the fundamental representation, namely $\chi_{\Box}(g) \equiv \operatorname{tr} g$. This character can be expressed as a sum of eigenvalues of g. Any other character can be expressed as a certain linear combination of monomials of the form $\chi_{\Box}(g^m)^n$ for integers m, n. In practice, one can use e.g. the Weyl character formula [327], which gives an explicit expression of χ_R in terms of the eigenvalues of g, which can then be reorganized as a linear combination of the monomials $\chi_{\Box}(g^m)^n$. In what follows we present the end result of this little exercise for the first few representations of G. • $\mathrm{SU}(N)$

$$tr_{\Box}(g) = tr(g)$$

$$tr_{\Box}(g) = \frac{1}{2}(+tr(g^{2}) + tr(g)^{2})$$

$$tr_{\Box}(g) = \frac{1}{2}(-tr(g^{2}) + tr(g)^{2})$$

$$tr_{\Box}(g) = \frac{1}{3!}(+2tr(g^{3}) + 3tr(g^{2})tr(g) + tr(g)^{3})$$

$$tr_{\Box}(g) = \frac{1}{3!}(-tr(g^{3}) + tr(g)^{3})$$

$$tr_{\Box}(g) = \frac{1}{3!}(+2tr(g^{3}) - 3tr(g^{2})tr(g) + tr(g)^{3})$$

$$tr_{\Box}(g) = \frac{1}{4!}(+6tr(g^{4}) + 8tr(g^{3})tr(g) + 3tr(g^{2})^{2} + 6tr(g^{2})tr(g)^{2} + tr(g)^{4})$$

$$tr_{\Box}(g) = \frac{1}{8}(-2tr(g^{4}) + 2tr(g^{2})tr(g)^{2} - tr(g^{2})^{2} + tr(g)^{4})$$

$$tr_{\Box}(g) = \frac{1}{12}(-4tr(g^{3})tr(g) + 3tr(g^{2})^{2} + tr(g)^{4})$$

$$tr_{\Box}(g) = \frac{1}{8}(+2tr(g^{4}) - 2tr(g^{2})tr(g)^{2} - tr(g^{2})^{2} + tr(g)^{4})$$

$$tr_{\Box}(g) = \frac{1}{4!}(-6tr(g^{4}) + 8tr(g^{3})tr(g) + 3tr(g^{2})^{2} + tr(g)^{4})$$

We also have the representations with traces removed:

$$\begin{aligned} \operatorname{tr}_{\square}(g) &= \operatorname{tr}(g)\operatorname{tr}(g^{-1}) - 1 \\ \operatorname{tr}_{\square}(g) &= \frac{1}{2}(+\operatorname{tr}(g)^{2}\operatorname{tr}(g^{-1}) + \operatorname{tr}(g^{2})\operatorname{tr}(g^{-1}) - 2\operatorname{tr}(g)) \\ \operatorname{tr}_{\square}(g) &= \frac{1}{2}(+\operatorname{tr}(g)^{2}\operatorname{tr}(g^{-1}) - \operatorname{tr}(g^{2})\operatorname{tr}(g^{-1}) - 2\operatorname{tr}(g)) \\ \operatorname{tr}_{\square}(g) &= \frac{1}{6}(+\operatorname{tr}(g)^{3}\operatorname{tr}(g^{-1}) + 3\operatorname{tr}(g^{2})\operatorname{tr}(g)\operatorname{tr}(g^{-1}) \\ &+ 2\operatorname{tr}(g^{3})\operatorname{tr}(g^{-1}) - 3\operatorname{tr}(g)^{2} - 3\operatorname{tr}(g)^{2}) \\ \operatorname{tr}_{\square}(g) &= \frac{1}{3}(+\operatorname{tr}(g)^{3}\operatorname{tr}(g^{-1}) - \operatorname{tr}(g^{3})\operatorname{tr}(g^{-1}) - 3\operatorname{tr}(g)^{2}) \\ \operatorname{tr}_{\square}(g) &= \frac{1}{6}(+\operatorname{tr}(g)^{3}\operatorname{tr}(g^{-1}) - 3\operatorname{tr}(g^{2})\operatorname{tr}(g)\operatorname{tr}(g^{-1}) \\ &+ 2\operatorname{tr}(g^{3})\operatorname{tr}(g^{-1}) + 3\operatorname{tr}(g^{2}) - 3\operatorname{tr}(g)^{2}) \\ \operatorname{tr}_{\square}(g) &= \frac{1}{4}(+\operatorname{tr}(g)^{2}\operatorname{tr}(g^{-1})^{2} - \operatorname{tr}(g^{2})\operatorname{tr}(g^{-1})^{2} - \operatorname{tr}(g)^{2}\operatorname{tr}(g^{-2}) \\ &- 4\operatorname{tr}(g)\operatorname{tr}(g^{-1}) + \operatorname{tr}(g^{2})\operatorname{tr}(g^{-2})) \\ \operatorname{tr}_{\square}(g) &= \frac{1}{4}(-\operatorname{tr}(g)^{2}\operatorname{tr}(g^{-1})^{2} - \operatorname{tr}(g^{2})\operatorname{tr}(g^{-1})^{2} + \operatorname{tr}(g)^{2}\operatorname{tr}(g^{-2}) \\ &+ \operatorname{tr}(g^{2})\operatorname{tr}(g^{-1})^{2} + \operatorname{tr}(g^{2})\operatorname{tr}(g^{-1}) - 1 \\ \operatorname{tr}_{\square}(g) &= \frac{1}{4}(+\operatorname{tr}(g)^{2}\operatorname{tr}(g^{-1})^{2} + \operatorname{tr}(g^{2})\operatorname{tr}(g^{-1})^{2} + \operatorname{tr}(g)^{2}\operatorname{tr}(g^{-2}) \\ &+ \operatorname{tr}(g^{2})\operatorname{tr}(g^{-2}) - 4\operatorname{tr}(g)\operatorname{tr}(g^{-1})) - 1 \end{aligned}$$

As a simple check, it is easy to see that these characters are consistent with the decompositions under $\mathfrak{su}_{A+B} \to \mathfrak{su}_A \oplus \mathfrak{su}_B$ in (B.1.3), e.g.,

$$\operatorname{tr}_{\Box}(g_1 \oplus g_2) = \operatorname{tr}_{\Box}(g_1) + \operatorname{tr}_{\Box}(g_1) \operatorname{tr}_{\Box}(g_2) + \operatorname{tr}_{\Box}(g_2)$$
(B.2.4)

as expected from $\Box \to \Box \otimes \bullet + \Box \otimes \Box + \bullet \otimes \Box$. Similarly, the characters are also consistent with the dimensions of the representations, e.g.,

$$\operatorname{tr}_{\square}(\mathbf{1}) = \frac{1}{3}(-N+N^3)$$
 (B.2.5)

as expected from $\dim_{\mathbb{R}}(\square) = \frac{2}{3}N(N+1)(N-1).$

We can also use the character to compute the index. For example, if we denote by δ the differential at g = 1, then

$$\delta^2 \operatorname{tr}_{\Box}(1) = \frac{1}{3} (-9 + 3 \operatorname{tr}(1)^2) \delta^2 \operatorname{tr}(1)$$
(B.2.6)

whence $T_{\mathbb{R}}(\square) = -3 + N^2$, as expected.

• SO(N)

The result is identical to the SU(N) case, with the following extra terms:

$$\operatorname{tr}_{\Box\Box}(g) = \cdots - 1$$

$$\operatorname{tr}_{\Box\Box}(g) = \cdots - \operatorname{tr}(g)$$

$$\operatorname{tr}_{\Box\Box}(g) = \cdots - \frac{1}{2}(\operatorname{tr}(g)^{2} + \operatorname{tr}(g^{2}))$$

$$\operatorname{tr}_{\Box\Box}(g) = \cdots - (\operatorname{tr}(g)^{2} - 1)$$

$$\operatorname{tr}_{\Box}(g) = \cdots - \frac{1}{2}(\operatorname{tr}(g)^{2} + \operatorname{tr}(g^{2}))$$

$$\operatorname{tr}_{\Box}(g) = \cdots - \frac{1}{2}(\operatorname{tr}(g)^{2} - \operatorname{tr}(g^{2}))$$

$$\operatorname{tr}_{\Box}(g) = \cdots - \frac{1}{2}(\operatorname{tr}(g)^{2} - \operatorname{tr}(g^{2}))$$

• $\operatorname{Sp}(N)$

The result is identical to the SU(N) case, with the following extra terms:

$$\operatorname{tr}_{\square}(g) = \cdots - 1$$

$$\operatorname{tr}_{\square}(g) = \cdots - \operatorname{tr}(g)$$

$$\operatorname{tr}_{\square}(g) = \cdots - \frac{1}{2}(\operatorname{tr}(g)^{2} + \operatorname{tr}(g^{2}))$$

$$\operatorname{tr}_{\square}(g) = \cdots - \frac{1}{2}(\operatorname{tr}(g)^{2} - \operatorname{tr}(g^{2}))$$

$$\operatorname{tr}_{\square}(g) = \cdots - (\operatorname{tr}(g)^{2} - 1)$$

$$\operatorname{tr}_{\square}(g) = \cdots - \frac{1}{2}(\operatorname{tr}(g)^{2} - \operatorname{tr}(g^{2}))$$

$$\operatorname{tr}_{\square}(g) = \cdots - \frac{1}{2}(\operatorname{tr}(g)^{2} - \operatorname{tr}(g^{2}))$$

• G_2

The maximal torus sits inside SO(7), say with

$$g = \mathcal{R}(\theta_1) \oplus \mathcal{R}(\theta_2) \oplus \mathcal{R}(\theta_3) \oplus 1, \qquad \theta_1 + \theta_2 + \theta_3 = 0$$
 (B.2.9)

where $\mathcal{R}(\theta) = \begin{pmatrix} \cos \theta & \sin \theta \\ -\sin \theta & \cos \theta \end{pmatrix}$. As G_2 has rank 2, the characters are polynomials in two variables. These are non-unique, as we can always add a polynomial that vanishes at g above,

e.g., $-\frac{1}{6}\operatorname{tr}(g)^3 + \frac{1}{2}\operatorname{tr}(g)^2 + \frac{1}{2}\operatorname{tr}(g^2)\operatorname{tr}(g) + \operatorname{tr}(g) + \frac{1}{2}\operatorname{tr}(g^2) - \frac{1}{3}\operatorname{tr}(g^3)$. All the expressions below are just one of the infinitely many possibilities.

The fundamental representation (1,0) has real dimension 7, and character

$$\chi_{(1,0)} = \operatorname{tr}(g) = 2(\cos\theta_1 + \cos\theta_2 + \cos\theta_3) + 1 \tag{B.2.10}$$

The next few representations have

$$\begin{split} \chi_{(0,1)} &= \frac{1}{2} \operatorname{tr}(g)^2 - \operatorname{tr}(g) - \frac{1}{2} \operatorname{tr}(g^2) \\ \chi_{(2,0)} &= \frac{1}{2} \operatorname{tr}(g)^2 + \frac{1}{2} \operatorname{tr}(g^2) - 1 \\ \chi_{(1,1)} &= \frac{1}{3} \operatorname{tr}(g)^3 - \operatorname{tr}(g)^2 - \frac{1}{3} \operatorname{tr}(g^3) + 1 \\ \chi_{(3,0)} &= \frac{1}{2} \operatorname{tr}(g)^2 + \operatorname{tr}(g^2) \operatorname{tr}(g) + \frac{1}{2} \operatorname{tr}(g^2) \\ \chi_{(0,2)} &= \frac{1}{4} \operatorname{tr}(g)^4 - \operatorname{tr}(g)^3 - \frac{1}{2} \operatorname{tr}(g^2) \operatorname{tr}(g)^2 \\ &\quad - \frac{1}{2} \operatorname{tr}(g)^2 + \operatorname{tr}(g) + \frac{1}{4} \operatorname{tr}(g^2)^2 - \frac{1}{2} \operatorname{tr}(g^2) \\ \chi_{(2,1)} &= \frac{1}{4} \operatorname{tr}(g)^4 - \operatorname{tr}(g)^3 - \frac{1}{2} \operatorname{tr}(g)^2 - \operatorname{tr}(g^2) \operatorname{tr}(g) \\ &\quad + 2 \operatorname{tr}(g) - \frac{1}{4} \operatorname{tr}(g^2)^2 + \frac{1}{2} \operatorname{tr}(g^2) \\ \chi_{(4,0)} &= -\frac{1}{4} \operatorname{tr}(g)^4 + \frac{4}{3} \operatorname{tr}(g)^3 + \operatorname{tr}(g^2) \operatorname{tr}(g)^2 - 3 \operatorname{tr}(g) \\ &\quad + \frac{1}{4} \operatorname{tr}(g^2)^2 - 2 \operatorname{tr}(g^2) + \frac{2}{3} \operatorname{tr}(g^3) \end{split}$$
(B.2.11)

all of which are real, and have real dimensions 14, 27, 64, 77, 77, 189, 182, respectively. For completeness, their indices are 4, 9, 32, 44, 55, 144, 156, respectively.

• F_4

The maximal torus is free of rank 4, and it sits inside SO(26), say with

$$g_4 = \mathcal{R}(0) \oplus \left(\bigoplus_{i=1}^4 \mathcal{R}(\theta_i)\right) \oplus \left(\bigoplus_{\pm} \mathcal{R}(\frac{1}{2}(\theta_1 \pm \theta_2 \pm \theta_3 \pm \theta_4))\right)$$
(B.2.12)

where the second sum runs over the $2^3 = 8$ choices of sign, for a total of $2 \times (8 + 4 + 1) = 26$ matrix elements. One can also realise the torus inside e.g. SO(52), with

$$g_1 = g_4 \oplus \mathcal{R}(0) \oplus \left(\bigoplus_{i < j, \pm} \mathcal{R}(\theta_i \pm \theta_j)\right)$$
 (B.2.13)

where the sum has $2 \times 6 = 12$ terms, for a total of $26 + 2 \times (12 + 1) = 52$.

These two realizations correspond to two of the four fundamental weights (0, 0, 0, 1), (1, 0, 0, 0), respectively, whose associated characters are

$$\chi_{(0,0,0,1)} = \operatorname{tr}(g_4) = 2\left(1 + \sum_{i=1}^4 \cos\theta_i + \sum_{\pm} \cos\frac{1}{2}(\theta_1 \pm \theta_2 \pm \theta_2 \pm \theta_4)\right)$$

$$\chi_{(1,0,0,0)} = \operatorname{tr}(g_1) = \operatorname{tr}(g_4) + 2\left(1 + \sum_{i < j, \pm} \cos(\theta_i \pm \theta_j)\right)$$
(B.2.14)

The next few representations are

$$\begin{aligned} \operatorname{tr}_{(1,0,0,0)} &= \operatorname{tr}(g_{1}) \\ \operatorname{tr}_{(0,0,0,1)} &= \operatorname{tr}(g_{4}) \\ \operatorname{tr}_{(0,1,0,0)} &= -\operatorname{tr}(g_{1}) + \frac{1}{2}\operatorname{tr}(g_{1})^{2} - \frac{1}{2}\operatorname{tr}(g_{1}^{2}) \\ \operatorname{tr}_{(0,0,1,0)} &= \frac{1}{2}(\operatorname{tr}(g_{4})^{2} - 2\operatorname{tr}(g_{1}) - \operatorname{tr}(g_{4}^{2})) \\ \operatorname{tr}_{(0,0,0,2)} &= \frac{1}{2}(-2 - 2\operatorname{tr}(g_{4}) + \operatorname{tr}(g_{4})^{2} + \operatorname{tr}(g_{4}^{2})) \\ \operatorname{tr}_{(1,0,0,1)} &= \frac{1}{6}(-3\operatorname{tr}(g_{4})^{2} + \operatorname{tr}(g_{4})^{3} + 12\operatorname{tr}(g_{1}) - 3\operatorname{tr}(g_{1})^{2} \\ &\quad + 3\operatorname{tr}(g_{4}^{2}) - 3\operatorname{tr}(g_{4})\operatorname{tr}(g_{4}^{2}) + 3\operatorname{tr}(g_{1}^{2}) + 2\operatorname{tr}(g_{4}^{3})) \\ \operatorname{tr}_{(2,0,0,0)} &= \frac{1}{2}(2\operatorname{tr}(g_{4}) - \operatorname{tr}(g_{4})^{2} + \operatorname{tr}(g_{1})^{2} - \operatorname{tr}(g_{4}^{2}) + \operatorname{tr}(g_{1}^{2})) \\ \operatorname{tr}_{(0,0,0,3)} &= \frac{1}{6}(-6\operatorname{tr}(g_{4})^{2} + \operatorname{tr}(g_{4})^{3} + 6\operatorname{tr}(g_{1}) \\ &\quad + 3\operatorname{tr}(g_{4})\operatorname{tr}(g_{4}^{2}) + 2\operatorname{tr}(g_{4}^{3})) \\ \operatorname{tr}_{(0,0,1,1)} &= 1 - \operatorname{tr}(g_{4}) - \frac{1}{2}\operatorname{tr}(g_{4})^{2} + \frac{1}{6}\operatorname{tr}(g_{4})^{3} - 2\operatorname{tr}(g_{1}) \\ &\quad + \frac{1}{2}\operatorname{tr}(g_{1})^{2} - \frac{1}{2}\operatorname{tr}(g_{4}^{2}) + \frac{1}{2}\operatorname{tr}(g_{4})\operatorname{tr}(g_{4}^{2}) \\ &\quad - \frac{1}{2}\operatorname{tr}(g_{1}^{2}) - \frac{2}{3}\operatorname{tr}(g_{4}^{3}) \\ \operatorname{tr}_{(1,0,1,0)} &= \operatorname{tr}(g_{4}) + \operatorname{tr}(g_{4})^{2} - \frac{1}{3}\operatorname{tr}(g_{4})^{3} + \frac{1}{24}\operatorname{tr}(g_{4})^{4} \\ &\quad - \frac{1}{2}\operatorname{tr}(g_{1}^{2})^{2} - \frac{1}{4}\operatorname{tr}(g_{4})^{2}\operatorname{tr}(g_{4}^{2}) + \frac{1}{8}\operatorname{tr}(g_{4}^{2})^{2} \\ &\quad - \frac{1}{2}\operatorname{tr}(g_{1}^{2}) + \frac{1}{3}\operatorname{tr}(g_{4}^{3}) + \frac{1}{3}\operatorname{tr}(g_{4})\operatorname{tr}(g_{4}^{3}) - \frac{1}{4}\operatorname{tr}(g_{4}) \\ \end{array} \right\}$$

• *E*₆

The maximal torus is free of rank 6, and can be realised inside SU(27), with

$$g_1 = \operatorname{diag}(e^{-\frac{2}{3}\theta_6}, e^{\frac{1}{3}\theta_6 \pm \theta_i}, e^{\frac{1}{2}(\pm\theta_1 \pm \theta_2 \pm \theta_3 \pm \theta_4 \pm \theta_5) - \frac{1}{6}\theta_6})$$
(B.2.16)

where in the last term we take all combinations of signs such that there is an odd number of negative signs. In total, there are $1 + 2 \times 5 + 16 = 27$ terms.

Another realisation of the maximal torus is inside SU(78), with

$$g_6 = \mathbf{1}_6 + \text{diag}(e^{\pm\theta_i \pm \theta_j}, e^{\frac{1}{2}(\pm\theta_1 \pm \theta_2 \pm \theta_3 \pm \theta_4 \pm \theta_5 \pm \theta_6)})$$
(B.2.17)

where i < j in the second term, and we take all signs combinations in the third term, such that the number of negative signs is odd. In total, there are $6 + 10 \times 4 + 32 = 78$ terms.

The first few characters are as follows:

$$\begin{split} \chi_{(1,0,0,0,0,0)} &= \operatorname{tr}(g_1) \\ \chi_{(0,1,0,0,0,0)} &= \frac{1}{2} \operatorname{tr}(g_1)^2 - \frac{1}{2} \operatorname{tr}(g_1^2) \\ \chi_{(0,0,1,0,0,0)} &= \frac{1}{6} (\operatorname{tr}(g_1)^3 - 3 \operatorname{tr}(g_1^2) \operatorname{tr}(g_1) + 2 \operatorname{tr}(g_1^3)) \\ \chi_{(0,0,0,0,0,1)} &= \operatorname{tr}(g_6) \\ \chi_{(0,0,0,0,2,0)} &= \frac{1}{2} (\operatorname{tr}(g_1^{-1})^2 + \operatorname{tr}(g_1^{-2}) - 2 \operatorname{tr}(g_1)) \\ \chi_{(1,0,0,0,1,0)} &= \operatorname{tr}(g_1) \operatorname{tr}(g_1^{-1}) - \operatorname{tr}(g_6) - 1 \\ \chi_{(1,0,0,0,1,0)} &= \frac{1}{2} \operatorname{tr}(g_1) \operatorname{tr}(g_3)^2 - \frac{1}{2} \operatorname{tr}(g_1) \operatorname{tr}(g_3^2) \\ &\quad - \frac{1}{6} \operatorname{tr}(g_1)^4 + \frac{1}{2} \operatorname{tr}(g_1^2) \operatorname{tr}(g_1)^2 - \frac{1}{3} \operatorname{tr}(g_1^3) \operatorname{tr}(g_1) \\ &\quad - \operatorname{tr}(g_1) + \frac{1}{2} \operatorname{tr}(g_1^{-2}) - \frac{1}{2} \operatorname{tr}(g_1^{-1})^2 \\ \chi_{(0,0,0,0,0,2)} &= -\frac{1}{2} \operatorname{tr}(g_1) \operatorname{tr}(g_3)^2 + \operatorname{tr}(g_1) \operatorname{tr}(g_3) + \frac{1}{2} \operatorname{tr}(g_1) \operatorname{tr}(g_3^2) + \operatorname{tr}(g_3^2) \\ &\quad - \operatorname{tr}(g_1) \operatorname{tr}(g_1^{-1}) + \frac{1}{6} \operatorname{tr}(g_1)^4 + \frac{1}{6} \operatorname{tr}(g_1)^3 - \frac{1}{2} \operatorname{tr}(g_1^2) \operatorname{tr}(g_1) \\ &\quad - \frac{1}{2} \operatorname{tr}(g_1^2) \operatorname{tr}(g_1)^2 + \frac{1}{3} \operatorname{tr}(g_1^3) \operatorname{tr}(g_1) + \frac{1}{3} \operatorname{tr}(g_1^3) + 2 \operatorname{tr}(g_3) \\ \chi_{(3,0,0,0,0)} &= \frac{1}{6} (-6 \operatorname{tr}(g_1) \operatorname{tr}(g_1^{-1}) + \operatorname{tr}(g_1)^3 + 3 \operatorname{tr}(g_1^2) \operatorname{tr}(g_1) \\ &\quad + 2 \operatorname{tr}(g_1^3) + 6 \operatorname{tr}(g_3)) \end{split}$$

• E_7

The maximal torus is free of rank 7, and can be realised inside Sp(28), with

$$g_6 = \operatorname{diag}(e^{\theta_i}, e^{\theta_i - \theta_7}, e^{\frac{1}{2}(\theta_1 + \theta_2 + \theta_3 + \theta_4 + \theta_5 + \theta_6 - 3\theta_7)}, e^{\frac{1}{2}(\pm \theta_1 \pm \theta_2 \pm \theta_3 \pm \theta_4 \pm \theta_5 \pm \theta_6 + \theta_7)}) \oplus \text{c.c.}$$
(B.2.19)

where i = 1, ..., 6 and, in the last term, we take all combinations of signs such that there are exactly two "+" signs. All in all, 6 + 6 + 1 + 15 = 28 terms.

One can also realise the maximal torus inside SO(133), with

$$g_{1} = \mathbf{1}_{7} \oplus \mathcal{R}(\theta_{7}) \oplus \mathcal{R}(\theta_{i} - \theta_{j}) \oplus \mathcal{R}(\theta_{i} + \theta_{j} - \theta_{7})$$

$$\oplus \mathcal{R}(\frac{1}{2}(\pm\theta_{1} \pm \theta_{2} \pm \theta_{3} \pm \theta_{4} \pm \theta_{5} \pm \theta_{6} + \theta_{7}))$$

$$\oplus \mathcal{R}(\frac{1}{2}(\pm\theta_{1} \pm \theta_{2} \pm \theta_{3} \pm \theta_{4} \pm \theta_{5} \pm \theta_{6} + 3\theta_{7}))$$
(B.2.20)

where in the first line $1 \le i < j \le 6$, and in the second line we take all combinations of signs such that there are either one or three "+" signs, and in the third line exactly one "+" sign. All in all, $7 + 2 \times (1 + 2 \times 15 + 26 + 6) = 133$ elements.

Finally, it is also convenient to introduce a second pseudo-real realisation of the torus, sitting inside Sp(456), this time using the seventh fundamental representation:

$$g_{7} = g_{6}^{\oplus 3} \oplus \\ \operatorname{diag}(e^{\theta_{i} \pm \theta_{j} \pm \theta_{k}}, e^{\theta_{i} + \theta_{j} + \theta_{k} - 2\theta_{7}}, e^{\pm \theta_{i} \pm \theta_{j} \pm \theta_{k} + \theta_{7}}, e^{\frac{1}{2}(\theta_{1} + \dots + \theta_{6} - 5\theta_{7})}, \\ e^{\frac{1}{2}(\pm \theta_{1} \pm \dots \pm \theta_{6} + 3\theta_{7})}, e^{\frac{1}{2}(\pm \theta_{1} \pm \dots \pm \theta_{i} \pm \dots \pm \theta_{6} + 3\theta_{7} - 3\theta_{j})}, e^{\frac{1}{2}(\pm \theta_{1} \pm \dots \pm \theta_{6} - \theta_{7} + 3\theta_{j})}, \\ e^{\frac{1}{2}(\theta_{1} + \dots + \theta_{6} - \theta_{7}) - 2\theta_{j}}, e^{\frac{1}{2}(\theta_{1} + \dots + \theta_{6} - \theta_{7})}, e^{\frac{1}{2}(\pm \theta_{1} \pm \dots \pm \theta_{6} + \theta_{7})}) \oplus \text{c.c.}$$
(B.2.21)

where $1 \le i < j < k \le 6$, and we take all the sign combinations for which the number of "+"s is as follows: in the first term, either zero or one; in the third term, either zero or one; in the fifth term, two; in the sixth term, one; in the seventh term, two; and finally in the tenth term, four. All in all, the number of terms is $3 \times 56 + 3 \times 20 + 1 \times 20 + 4 \times 20 + 1 + 15 + 6 \times 5 + 6 \times 10 + 6 + 1 + 15 = 456$.

The rest of characters can easily be expressed in terms of these. For example,

$$\begin{split} \chi_{(0,0,0,0,1,0)} &= \operatorname{tr}(g_{6}) \\ \chi_{(1,0,0,0,0,0)} &= \operatorname{tr}(g_{1}) \\ \chi_{(0,0,0,0,0,1)} &= \operatorname{tr}(g_{7}) \\ \chi_{(0,0,0,0,0,2,0)} &= +\frac{1}{2}\operatorname{tr}(g_{6}^{2}) + \frac{1}{2}\operatorname{tr}(g_{6})^{2} - \operatorname{tr}(g_{1}) \\ \chi_{(0,0,0,0,1,0,0)} &= -\frac{1}{2}\operatorname{tr}(g_{6}^{2}) + \frac{1}{2}\operatorname{tr}(g_{6})^{2} - 1 \\ \chi_{(1,0,0,0,0,0,0)} &= +\operatorname{tr}(g_{6})\operatorname{tr}(g_{1}) - \operatorname{tr}(g_{6}) - \operatorname{tr}(g_{7}) \\ \chi_{(2,0,0,0,0,0,0)} &= +\frac{1}{2}\operatorname{tr}(g_{6}^{2}) - \frac{1}{2}\operatorname{tr}(g_{6})^{2} + \frac{1}{2}\operatorname{tr}(g_{1}^{2}) + \frac{1}{2}\operatorname{tr}(g_{1})^{2} \\ \chi_{(0,1,0,0,0,0,0)} &= -\frac{1}{2}\operatorname{tr}(g_{1}^{2}) + \frac{1}{2}\operatorname{tr}(g_{1})^{2} - \operatorname{tr}(g_{1}) \\ \chi_{(0,0,0,0,0,0,0)} &= +\frac{1}{3}\operatorname{tr}(g_{6}^{3}) + \frac{1}{2}\operatorname{tr}(g_{6}^{2})\operatorname{tr}(g_{6}) - \operatorname{tr}(g_{6})\operatorname{tr}(g_{1}) + \frac{1}{6}\operatorname{tr}(g_{6})^{3} + \operatorname{tr}(g_{7}) \\ \chi_{(0,0,0,0,0,1,1)} &= +\frac{1}{2}\operatorname{tr}(g_{6}^{2}) + \operatorname{tr}(g_{6})\operatorname{tr}(g_{7}) - \frac{1}{2}\operatorname{tr}(g_{6})^{2} - \frac{1}{2}\operatorname{tr}(g_{1}^{2}) - \frac{1}{2}\operatorname{tr}(g_{1})^{2} + 1 \\ \chi_{(0,0,1,0,0,0)} &= +\frac{1}{3}\operatorname{tr}(g_{1}^{3}) - \frac{1}{2}\operatorname{tr}(g_{1}^{2})\operatorname{tr}(g_{1}) + \frac{1}{6}\operatorname{tr}(g_{1})^{3} - \operatorname{tr}(g_{1})^{2} + \operatorname{tr}(g_{1}) \\ \end{split}$$

of dimension 56, 133, 912, 1463, 1539, 6480, 7371, 8645, 24320, 27664, 40755, 365750, respectively.

 $\bullet~E_8$ The maximal torus is free of rank 8. We will not discuss this group here.